

# Long-time behavior of some Markov processes, and application to stochastic algorithms

Lucas Journel

## ► To cite this version:

Lucas Journel. Long-time behavior of some Markov processes, and application to stochastic algorithms. Mathematics [math]. Sorbonne Université, 2024. English. NNT: 2024SORUS109. tel-04689937

## HAL Id: tel-04689937 https://theses.hal.science/tel-04689937v1

Submitted on 6 Sep 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





## Sorbonne Université LJLL

Doctoral School École Doctorale Sciences Mathématiques de Paris Centre University Department Laboratoire Jacques-Louis Lions

Thesis defended by **Lucas JOURNEL** 

Defended on June 3, 2024

In order to become Doctor from Sorbonne Université

Academic Field Applied Mathematics

# Long-time behavior of some Markov processes, and application to stochastic algorithms

Thesis supervised by Pierre MONMARCHÉ

### **Committee members**

Referees	David Herzog Nicolas Champagnat	Associate Professor at Iowa state university Senior Researcher at Inria Nancy
Examiners	Arnaud Guyader Tony Lelievre Andreas Eberle Marylou Gabrié Mireille Bossy	Professor at Sorbonne université Professor at Ecole des points ParisTech Professor at Universität Bonn Associate Professor at Ecole polytechnique Senior Researcher at Inria Sophia Antipolis
Supervisor	Pierre Monmarché	Associate Professor at Sorbonne Université

Committee President

## COLOPHON

Doctoral dissertation entitled "Long-time behavior of some Markov processes, and application to stochastic algorithms", written by Lucas JOURNEL, completed on July 17, 2024, typeset with the document preparation system  $\mbox{IAT}_{\rm E} X$  and the yathesis class dedicated to these prepared in France.

- **Keywords:** probability theory, stochastic calculus, coupling methods, long time behavior of markov processes, poincaré inequality, stochastic algorithm, hypocoercivity.
- **Mots clés :** probabilités, calcul stochastique, méthodes de couplage, comportement en temps long de processus de markov, inégalité de poincaré, algorithme stochastique, hypocoercivité.

This thesis has been prepared at

### Laboratoire Jacques-Louis Lions

Sorbonne Université Campus Pierre et Marie Curie 4 place Jussieu 75005 Paris France

**a** +33 1 44 27 42 98 Web Site https://ljll.math.upmc.fr/



You can't be what you can't see.

Marian Wright Edelman

# Long-time behavior of some Markov processes, and application to stochastic algorithms

#### Abstract

This thesis is devoted to the study of the long-time behavior of some Markov processes arising in stochastic algorithms. These algorithms are variations of Monte Carlo methods, which constitute a class of techniques for sampling probability measures based on Markov processes. The thesis is structured into three parts, each addressing a different type of algorithm:

- 1. Chapters 2 and 3 focus on sampling Gibbs measures using two distinct kinetic processes.
- 2. Chapter 4 explores non-convex and high-dimensional optimization through simulated annealing, while Chapter 5 investigates optimization using a switched process.
- 3. Chapters 6, 7, and 8 delve into sampling quasi-stationary distributions based on the Fleming-Viot process, a Moran-type particle system.

The motivation for these algorithms, and thus much of the research conducted for this thesis, stems from molecular dynamics and statistics applications. By applying coupling and partial differential equation methods, we aim to establish convergence guarantees for these algorithms.

**Keywords:** probability theory, stochastic calculus, coupling methods, long time behavior of markov processes, poincaré inequality, stochastic algorithm, hypocoercivity.

#### Résumé

Cette thèse est consacrée à l'étude du comportement à long terme de certains processus de Markov apparaissant dans les algorithmes stochastiques. Ces algorithmes sont des variations autour des méthodes de Monte Carlo, qui constituent une classe de techniques d'échantillonnage de mesures de probabilité basées sur des processus de Markov. La thèse est structurée en trois parties, chacune traitant d'un type d'algorithme différent :

- 1. Les chapitres 2 et 3 portent sur l'échantillonnage des mesures de Gibbs à l'aide de deux processus cinétiques différents.
- 2. Le chapitre 4 traite de l'optimisation non-convexe en grande dimension par recuit simulé, tandis que le chapitre 5 étudie l'optimisation à l'aide d'un processus commuté.
- 3. Les chapitres 6, 7 et 8 traitent de l'échantillonnage de mesures quasi-stationnaires basé sur le processus de Fleming-Viot, un système de particules de type Moran.

La motivation pour ces algorithmes, et donc pour la plupart des recherches menées dans le cadre de cette thèse, provient de dynamique moléculaire et d'applications en statistiques. En appliquant des méthodes de couplage et d'équations aux dérivées partielles, nous établirons des garanties de convergence pour ces algorithmes.

Mots clés : probabilités, calcul stochastique, méthodes de couplage, comportement en temps long de processus de markov, inégalité de poincaré, algorithme stochastique, hypocoercivité.

Sorbonne Université – Campus Pierre et Marie Curie – 4 place Jussieu – 75005 Paris – France

Abstract

# Contents

A	bstr	ract	ix
С	onte	ents	xi
L	ist o	f Figures	xv
R	leme	erciement.	xvii
1	Intr 1.1 1.2 1.3	roductionMarkov processes: definitions and basic properties for Monte-Carlo purposesSome methods to study the long-time behavior of Markov processes1.2.1 The PDE methods: Entropy, $L^2$ , $H^k$ convergence1.2.2 Wasserstein distance and coupling methodsStochastic algorithms and contributions of this thesis1.3.1 Sampling Gibbs measures1.3.2 Optimisation1.3.3 A particle approximation for quasi-stationary distributions	1 2 5 5 11 18 18 23 27
Ι	Sa	ampling Gibbs measures	33
2	We Lar	ak error expansion of a numerical scheme with rejection for singular agevin process	35
	2.1	Introduction	35
	2.2	Mathematical setting and results	38
		2.2.1 Assumptions and mains results	38
		2.2.2 Comparison between numerical scheme with and without rejection	41
	2.3	Proof of Theorem 2.2.1	43
		2.3.1 $H^1$ -setting	45
		2.3.2 $H^k$ -setting	49
		2.3.3 Proof of Theorem 2.2.1	53
	2.4	Weak error expansion	53
		2.4.1 Finite-time expansion for the numerical scheme with rejection	54
		2.4.2 Expansion for the invariant measure of the numerical scheme with rejection	62
3	Lor	ng-time behavior of the velocity jump Langevin process	67
	3.1	Introduction	67
		3.1.1 Motivation	67

164

165

3.2	3.1.2 Mathematical setting3.1.3 NotationProof of Theorem 3.1.13.2.1 Invariance of $\mu$ 3.2.2 Exponential decay for the $H^1(\mu)$ -norm3.2.3 Exponential decay for the $H^k$ -norm	69 71 71 72 74 78
II	Optimisation	91
4 Co	nvergence of the kinetic annealing for general potentials	93
4.1	Introduction and main results	93
4.2	Main steps of the proofs	97
	4.2.1 Return to a compact set	98
	4.2.2 Position in a compact set	100
	4.2.3 Localization and convergence	103
	4.2.4 Full process on a compact space	105
	4.2.5 Non-convergence with fast cooling schedules	108
4.3	Auxiliary results	109
	4.3.1 Uniform energy bounds	109
	4.3.2 Small time regularisation	112
4.4	$L^2$ -hypocoercivity	114
4.5	$H^k$ -hypocoercivity	123
4.6	Faster than logarithmic schedules	129
$5 \mathrm{Sw}$	itched diffusion processes for non-convex optimization and saddle points	199
sea	rcn Introduction	100
0.1	5.1.1 Oromion	100
	a Li Uverview	199
	5.1.9 Deterministic coddle point algorithms	133
	5.1.2 Deterministic saddle point algorithms	133 135 126
50	5.1.2 Deterministic saddle point algorithms	133 135 136
5.2	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2 1 Nairy ISD and Nairy CAD	133 135 136 138
5.2	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD	133 135 136 138 138
5.2	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes	$     133 \\     135 \\     136 \\     138 \\     143 \\     143 $
5.2 5.3	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         Solution         Solu	$     133 \\     135 \\     136 \\     138 \\     138 \\     143 \\     143 \\     143 $
5.2 5.3	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians	$     133 \\     135 \\     136 \\     138 \\     138 \\     143 \\     143 \\     144 \\     144 $
5.2 5.3	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines	$     \begin{array}{r}       133 \\       135 \\       136 \\       138 \\       138 \\       143 \\       143 \\       144 \\       148 \\       152     \end{array} $
5.2 5.3	5.1.2 Deterministic saddle point algorithms5.1.3 The stochastic processesTheoretical analysis5.2.1 Noisy ISD and Noisy GAD5.2.2 Switched processesNumerical experiments5.3.1 First 2D model: a mixture of Gaussians5.3.2 A 2D model with singular lines5.3.3 Lennard-Jones clusters	$     133 \\     135 \\     136 \\     138 \\     143 \\     143 \\     144 \\     148 \\     152     $
5.2 5.3	5.1.2 Deterministic saddle point algorithms5.1.3 The stochastic processesTheoretical analysis5.2.1 Noisy ISD and Noisy GAD5.2.2 Switched processesNumerical experiments5.3.1 First 2D model: a mixture of Gaussians5.3.2 A 2D model with singular lines5.3.3 Lennard-Jones clusters	$     133 \\     135 \\     136 \\     138 \\     143 \\     143 \\     144 \\     144 \\     152     $
5.2 5.3 III	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines         5.3.3 Lennard-Jones clusters	133 135 136 138 138 143 143 144 148 152 <b>155</b>
5.2 5.3 III 6 Co	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines         5.3.3 Lennard-Jones clusters         Fleming-Viot processes         nvergence of a particle approximation for the quasi-stationary distribution	133 135 136 138 138 143 143 144 148 152 <b>155</b>
5.2 5.3 III 6 Co of a	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines         5.3.3 Lennard-Jones clusters         Fleming-Viot processes         nvergence of a particle approximation for the quasi-stationary distribution a diffusion process: uniform estimates in a compact soft case.	133 135 136 138 143 143 144 152 <b>155</b> 157
5.2 5.3 III 6 Cor of a 6.1	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines         5.3.3 Lennard-Jones clusters         Fleming-Viot processes         nvergence of a particle approximation for the quasi-stationary distribution         a diffusion process: uniform estimates in a compact soft case.         Introduction	133 135 136 138 143 143 144 152 <b>155</b> <b>157</b> 158
5.2 5.3 III 6 Coi 6.1	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines         5.3.3 Lennard-Jones clusters         Fleming-Viot processes         nvergence of a particle approximation for the quasi-stationary distribution         a diffusion process: uniform estimates in a compact soft case.         Introduction         6.1.1 The problem	133 135 136 138 138 143 143 144 148 152 <b>155</b> <b>157</b> 158 158
5.2 5.3 III 6 Cor 6.1	5.1.2 Deterministic saddle point algorithms         5.1.3 The stochastic processes         Theoretical analysis         5.2.1 Noisy ISD and Noisy GAD         5.2.2 Switched processes         Numerical experiments         5.3.1 First 2D model: a mixture of Gaussians         5.3.2 A 2D model with singular lines         5.3.3 Lennard-Jones clusters         Fleming-Viot processes         mvergence of a particle approximation for the quasi-stationary distribution         a diffusion process: uniform estimates in a compact soft case.         Introduction         6.1.1 The problem         Notations and conventions	133 135 136 138 138 143 143 144 152 <b>155</b> <b>155</b> <b>157</b> 158 158 158 159

		6.2.2 Long-time convergence	167
		6.2.3 Propagation of chaos	169
		6.2.4 Discrete to continuous time	170
		6.2.5 Conclusion	174
7	Uni	form convergence of the Fleming-Viot process in a hard killing metastable	
•	case	a	181
	7 1	Introduction	181
		7 1 1 Main Result	183
	7.2	Preliminary results	186
	1.2	7.2.1 Coupling of the non-killed diffusion	186
		7.2.1 Coupling of the non-kined diffusion	188
		7.2.3 Particles near the boundary	192
	7.3	Proofs of the main theorems	194
		7.3.1 Long time convergence	194
		7.3.2 Propagation of chaos	202
	7.4	Establishing Assumption 7.1.2	206
			-00
8	The	e Fleming-Viot process in a high-killing regime: the soft killing case.	<b>211</b>
	8.1	Introduction	212
		8.1.1 Motivation	212
		8.1.2 Mathematical setting and notation	213
		8.1.3 Mathematical setting and notation	213
		8.1.4 Fixed number of particles	214
		8.1.5 Increasing number of particles	216
	8.2	Proof of Theorem 8.1.1	217
		8.2.1 A time-scale separation result	219
		8.2.2 The initial condition	221
		8.2.3 The Poisson equation	224
		8.2.4 Proof of Theorem 8.1.1	229
	8.3	Proof of Theorem 8.1.2	233
	8.4	Proof of Theorem 8.1.3	237
в	Bibliography 243		

 $\operatorname{Contents}$ 

# List of Figures

$2.1 \\ 2.2$	Failure probability	$\begin{array}{c} 42 \\ 42 \end{array}$
<ul><li>4.1</li><li>4.2</li></ul>	Case with two local non-global minima $x$ and $y$ at the same energy level. If $c < b$ , then Assumption 4.1.2 holds by chosing either $\tilde{x} = x$ or $\tilde{x} = y$ and any $a \in (c, b)$ . If $c \in (b, D(x))$ , again Assumption 4.1.2 holds with $\tilde{x} \in \{x, y\}$ and any $a \in (b, D(x))$ . However, if $c = b$ , Assumption 4.1.2 does not hold with $\tilde{x} \in \{x, y\}$ Construction of $U^K$ from $U$ , with $\tilde{x}$ chosen so that $c^*(U) = D(\tilde{x})$	$96\\106$
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12$	$ \begin{array}{l} \text{Case } (s_x,s_y)=(3,1) \\ \text{Case } (s_x,s_y)=(1,3). \\ \text{Periodized potential.} \\ \text{Invariant measure of the ISP for } \nu=0.1, \ensuremath{\varepsilon}=0.05. \\ \text{Time to reach a minima from the other one on the periodized potential.} \\ \text{Level line of the potential } U_2. \\ \ensuremath{\varepsilon}=0.0. \\ \ensuremath{\varepsilon}=0.05. \\ \text{Case } U_2. \\ \ensuremath{\varepsilon}=0.05. \\ \text{Case } U_2. \\ \ensuremath{\varepsilon}=0.05. \\ \ensuremath{\varepsilon}=0.0$	$145 \\ 146 \\ 147 \\ 147 \\ 148 \\ 149 \\ 149 \\ 150 \\ 151 \\ 151 \\ 153 \\ 153$
6.1	Summary of the different results. The label of each arrow indicates the corollary or proposition where the corresponding quantitative convergence is stated. Vertical, horizontal and diagonal arrows correspond respectively to $t$ , $\gamma$ and $N$ going to their limit.	179
7.1	First, $U_0$ is the energy level on the boundary of $\mathcal{D}$ . Then $U_1$ is taken as low as possible with the constraint that Assumption 7.1.3 holds, namely it is such that there exist a neighborhood $\mathfrak{B}_1$ of $\partial \mathcal{D}$ over which $U \ge U_1$ and $ \nabla U $ is lower bounded by a positive constant. Then $U_2$ (resp. $U_3$ ) is taken slightly below $U_0$ (resp. $(U_1 + U_0)/2$ ) so that the length of the two grey lines in the figure is larger than $a$ , which is itself a constant larger than the critical depth $c_*$ . The condition that $c_* < (U_0 - U_1)/2$ means that it is possible to find $U_2, U_3$ and $a$ to fulfill these constraints.	207

xvi

# Remerciement.

Je ne serais pas arrivé au bout (voire même au début) de cette thèse sans la présence de nombreuses personnes. Je profite de cette soutenance pour vous remercier. Si mes remerciements ne suffisent pas à vous occuper durant ma présentation, vous pouvez toujours tenter de répondre à cette question qui m'a taraudé durant toute ma thèse : pourquoi il n'y a plus de mammouths ?

Merci d'abord à toi, Pierre Monmarché, pour avoir encadré cette thèse. J'ai énormément appris durant cette thèse, et tu as été une source inépuisable de problèmes de maths. J'ai apprécié que tu me laisses beaucoup de liberté dans tous mes choix de projets.

Merci à Nicolas Champagnat et David P. Herzog pour avoir accepté de rapporter cette thèse. Merci aussi à Arnaud Guyader, Mireille Bossy, Andreas Eberle, Marylou Gabrie d'avoir accepté de faire partie de mon jury.

Finalement, merci Tony de non seulement faire partie de mon jury de thèse, mais aussi d'avoir collaboré avec moi, de m'avoir invité à des conférences à plusieurs reprises, de m'avoir aidé à trouver un post-doc, et enfin pour toutes tes qualités humaines.

Merci aussi à tous ceux avec qui j'ai eu l'occasion de travailler : Julien, Matthias, Nicolaï. Ça a toujours été un immense plaisir de discuter de maths avec vous.

Merci à tous les membres du secrétarariat pour faire fonctionner ce laboratoire, et d'avoir résolu tous les problèmes que j'ai eus durant cette thèse.

Merci à tous les doctorants du LJLL (que je ne citerai pas par souci de concision, ou flemme), pour créer une si bonne ambiance dans ce laboratoire. Merci aussi évidemment aux doctorants qui traînent en courô pour me permettre de m'échapper du susmentionné LJLL.

Un immense merci à la team chauve (Nicolaï, PLB). J'ai énormément apprécié tous ces moments en conférences avec vous. PLB, sans toi j'en serais encore à essayer de m'inscrire en thèse.

Un merci spécial à ceux qui, hors labo, m'ont supporté et toujours soutenu. Merci Charlotte, Arthur, Nico, Cécile, Marine, Torvic, Lucas(2). Et finalement merci à ma famille, sans qui je ne serais même pas arrivé au niveau de la thèse.

En fait, la réponse, c'est qu'il n'y a plus de pappouths...

xviii

Remerciement.

# Chapter 1

# Introduction

### Outline of the current chapter

1.1	Markov processes: definitions and basic properties for Monte-	
	Carlo purposes	<b>2</b>
1.2	Some methods to study the long-time behavior of Markov pro-	
	cesses	5
	1.2.1 The PDE methods: Entropy, $L^2$ , $H^k$ convergence	5
	1.2.2 Wasserstein distance and coupling methods	11
1.3	Stochastic algorithms and contributions of this thesis	18
	1.3.1 Sampling Gibbs measures	18
	1.3.2 Optimisation	23
	1.3.3 A particle approximation for quasi-stationary distributions	27

The topic of this thesis is the application of the study of the long time behavior of Markov processes to the analysis of stochastic algorithms, at the intersection of PDE theory, probability, and applied mathematics. These algorithms are variations of Monte Carlo methods, which constitute a class of techniques for probability measure sampling based on Markov processes. In addition to sampling, stochastic algorithms can also be employed to model physical systems, or in non-convex and high-dimensional optimization problems that cannot be efficiently solved using deterministic algorithms. They offer a compelling compromise between obtaining high-quality results and reducing computation time. The theoretical study of the underlying Markov processes provides essential guarantees of convergence for such algorithms, which are of paramount importance.

The motivation for most of the projects undertaken during the preparation of this thesis stems from molecular dynamics and statistics, yet it can be applied to a broad spectrum of fields, including population dynamics, finance, and biology. Beyond their practical applications, the study of such algorithms gives rise to intriguing theoretical questions in the domain of coupling methods, hypocoercivity, and metastability. Since this thesis focuses on Markov processes in metric spaces (while non-Markovian processes are also of interest for stochastic algorithms, they are not the primary focus of this thesis), which constitute a particular class of stochastic processes, we will first introduce them in Section 1.1. This introduction will then be divided into two parts. Section 1.2 will present two families of methods for the study of the long time behavior of Markov processes. The final Section 1.3 will introduce some stochastic algorithms, as well as the various contributions of this thesis.

## 1.1 Markov processes: definitions and basic properties for Monte-Carlo purposes

The goal of Monte Carlo Markov Chains (MCMC) is to sample from a specified probability measure  $\mu$  using Markov chains or processes. To illustrate how this operates, let's introduce some notations and define Markov processes. For a more in-depth introduction to Markov processes, refer to [58].

**Notations:** Throughout this introduction,  $(\Omega, \mathcal{F})$  will denote a measurable space, and  $(E, \mathbf{d})$ a metric space referred to as the state space. We use  $\mathcal{B}$  for the Borel  $\sigma$ -algebra on E. As our focus is on dynamics within a state space, we denote by  $\mathcal{C}(\mathbb{R}_+, E)$  (resp.  $\mathcal{D}(\mathbb{R}_+, E)$ ) the set of continuous (resp., right-continuous with left limits at all points, or *cadlag*) functions from  $\mathbb{R}_+$ to E. These two sets are equipped with the product  $\sigma$ -algebra. Let  $\mathcal{M}^1(E)$  denotes the set of probability measures on any set E, and if  $\mu \in \mathcal{M}^1(E)$  is a measure on E, and  $h: E \to \mathbb{R}$ , denote

$$\mu(h) = \int_E h \mathrm{d}\mu.$$

Given  $d \in \mathbb{N}$ ,  $\mathcal{M}_d(\mathbb{R})$  denote the set of real valued matrix of size  $d \times d$ .

**Definition 1.1.1.** We call a continuous (resp. cadlag) Markov process a random variable

$$X: \Omega \to \mathcal{C}(\mathbb{R}_+, E), \quad resp. X: \Omega \to \mathcal{D}(\mathbb{R}_+, E),$$

with a family of probability measures  $(\mathbb{P}_x)_{x\in E}$  on  $(E, \mathcal{B})$  such that:

- $\mathbb{P}_x(X_0=x)=1.$
- For all  $x \in E$ ,  $0 \leq s_1 \leq \cdots \leq s_m \leq t \leq t_1 \leq \cdots \leq t_n$ ,  $(A_i)_{1 \leq i \leq m} \in \mathcal{B}$ ,  $(B_i)_{1 \leq i \leq n} \in \mathcal{B}$ :

$$\mathbb{P}_{x} \left( X_{s_{1}} \in A_{1}, \cdots, X_{s_{m}} \in A_{m}, X_{t_{1}} \in B_{1}, \cdots, X_{t_{n}} \in B_{n} | X_{t} \right) \\ = \mathbb{P}_{x} \left( X_{s_{1}} \in A_{1}, \cdots, X_{s_{m}} \in A_{m} | X_{t} \right) \mathbb{P}_{x} \left( X_{t_{1}} \in B_{1}, \cdots, X_{t_{n}} \in B_{n} | X_{t} \right).$$

Informally, a Markov process X is a random function from  $\mathbb{R}_+$  to E, such that, for all  $t \ge 0$ , knowing  $X_t$ , the future is independent of the past. In mathematical terms, it is equivalent to

$$\forall A \in \sigma(X_s, s \ge t), \forall B \in \sigma(X_s, s \le t), \mathbb{P}(AB|X_t) = \mathbb{P}(A|X_t) \mathbb{P}(B|X_t).$$

The probability  $\mathbb{P}_x$  corresponds to the Markov process with initial condition x. Additionally, for any probability measure  $\nu \in \mathcal{M}^1(E)$ , a process with initial condition  $\nu$  can be defined by

$$\mathbb{P}_{\nu} = \int_{E} \mathbb{P}_{x} \nu(\mathrm{d}x)$$

We denote  $\mathbb{E}_{\nu}$  the expectation under  $\mathbb{P}_{\nu}$  and simply  $\mathbb{E}_x$  when  $\nu = \delta_x$ . In all of this thesis, a Markov process will be denoted either X or Z. Let's start with a few examples that will arise at some point in this thesis. We will not delve into the construction of such processes or the proof of their Markovian nature, as these may be intricate subjects. Also, note that E is not necessarily complete or separable, as we will see below.

• The simplest Markov processes are jump processes on graphs. Let E be a (finite or countable) set, endowed with the trivial topology, namely  $\mathbf{d}(x, y) = \mathbb{1}_{x \neq y}$ . For all  $x, y \in E$ , let  $q(x, y) \ge 0$  be a jump rate. A Markov process X on E can be defined as follows: given some initial condition  $x \in E$ , let  $J_y$  be an exponential random variable with parameter q(x, y). Write

$$J^1 = \min_{y \in E} J_y, \quad Y^1 = \arg\min_{y \in E} J_y.$$

Then X jumps from a state x to a state y at rate q(x, y) in the sense that  $X_t = x$  for all  $t < J^1$  and  $X_{J^1} = Y^1$ . This defines X until  $J^1$ , and an induction argument then extends  $X_t$  for all  $t \ge 0$ . Such processes will appear exclusively in Chapter 8.

• (Itô-)Diffusion processes on ℝ<sup>d</sup>: this is most important example for this thesis, appearing in Chapters 2,4,6 and 7. They can be defined as the solution to the Itô equation:

$$dZ_t = b(Z_t)dt + \sigma(Z_t)dB_t, \qquad (1.1)$$

where B is a Brownian motion,  $b : \mathbb{R}^d \to \mathbb{R}^d$  is the drift, and  $\sigma : \mathbb{R}^d \to \mathcal{M}_d(\mathbb{R})$  is the diffusion coefficient. They are continuous Markov process, see [154]. While they can be defined more generally, this definition is sufficient for our purpose and encompasses the two most crucial examples for this thesis: the overdamped and kinetic Langevin processes (see Section 1.2 and 1.3).

• Switched diffusion process: fix a family of drifts  $(b_i)$  and diffusions  $(\sigma_i)$ ,  $1 \le i \le N$ , and a Markov chain I on the set  $[\![1, N]\!]$ . Then we say that Z is a switched diffusion if it solves:

$$\mathrm{d}Z_t = b_{I_t}(Z_t)\mathrm{d}t + \sigma_{I_t}(Z_t)\mathrm{d}B_t.$$

Z here is not a Markov process, but (Z, I) is. This represents a less classical class of Markov processes and will be discussed in Chapter 5.

• Piecewise deterministic Markov process (PDMP): fix some flow  $\varphi_t$  on a set E, jump rate  $\lambda: E \to \mathbb{R}_+$ , and jump kernel  $Q(x): E \to \mathcal{M}^1(E)$ . Then, X is a PDMP if, by defining the random times  $T_n$  by

$$T_{n+1} = \inf \left\{ t \ge T_n, \int_{T_n}^t \lambda(X_s) \mathrm{d}s \ge E_n \right\},\,$$

where  $(E_n)_{n \in \mathbb{N}}$  is a family of independent exponential random variable of parameter 1, we have that for all  $n \in \mathbb{N}$ :

$$X_t = \varphi_{t-T_n}(X_{T_n}), \forall t \in [T_n, T_{n+1}], X_{T_n} \sim Q(X_{T_n}).$$

The process X is deterministic between random times, at which it jumps randomly to another location. While such processes were more recently analyzed for sampling purposes and may not directly appear in the thesis, a mixture between PDMP and diffusion processes will be studied in Chapter 3.

For a given Markov process  $(X_t)$ , we define its semi-group  $(P_t)_{t\geq 0}$  by:

$$P_t f(x) = \mathbb{E}_x \left( f\left( X_t \right) \right),$$

for any  $f: E \to \mathbb{R}$  bounded and measurable. The process is completely characterized by its semigroup, making it a central object of study. The semi-group also acts on the set of probability measures  $\mathcal{M}^1(E)$  as follows:

$$\int f \mathrm{d} \left( \nu P_t \right) = \int P_t f \mathrm{d} \nu,$$

for any  $\nu \in \mathcal{M}^1(E)$ . By the definition of the semi-group,  $\nu P_t$  is the law of  $X_t$  under  $\mathbb{P}_{\nu}$ .

**Definition 1.1.2.** A measure  $\mu$  is said to be an invariant (or stationary) measure for the Markov process X if for all  $t \ge 0$ 

 $\mu P_t = \mu.$ 

There may exist no, one, or an infinite number of invariant measures. The goal of MCMC is, given a target distribution  $\mu$ , to design a Markov process X such that  $\mu$  is its only invariant measure, and X is ergodic, in one of the following senses:

$$\frac{1}{t} \int_0^t \delta_{X_s} \mathrm{d}s \underset{t \to \infty}{\to} \mu, \text{ weakly, almost surely}, \tag{1.2}$$

or

$$\mathcal{L}aw(X_t) \underset{t \to \infty}{\to} \mu$$
, weakly. (1.3)

In these cases, the invariant measure describes the behavior of the process as time goes to infinity. Reciprocally, the simulation of  $X_t$  (or of a trajectory  $(X_s)_{0 \leq s \leq t}$ ), for large time  $t \geq 0$ , yields an approximation of  $\mu$ . Hence, the analysis of such convergences (under different possible norms) will be the main topic of this thesis.

A powerful tool to study the evolution of the law of a Markov process (and thus its long-time behavior) is its so-called generator (or equivalently the generator of its semi-group).

**Definition 1.1.3.** Let  $\mathcal{E}$  be a Banach space of functions from E to  $\mathbb{R}$ , endowed with some norm  $\|\cdot\|_{\mathcal{E}}$ . The generator of  $P_t$  and its domain  $(L, \mathcal{D}(L))$  are defined by:

$$f \in \mathcal{D}(L), g = Lf \Leftrightarrow \lim_{t \to 0} \left\| \frac{P_t f - f}{t} - g \right\|_{\mathcal{E}} = 0.$$

The classical setting is to take  $\mathcal{E}$  as the set of bounded functions, endowed with the norm  $||g||_{\infty} = \sup_{E} |g|$ . In this case, for continuous or *cadlag* process, the generator completely determines the Markov process, and for all  $f \in \mathcal{D}(L)$ , we have the Kolmogorov equation:

$$\partial_t P_t f = L P_t f = P_t L f.$$

However, in many cases,  $P_t$  defines a semi-group in other Banach spaces, see Section 1.2.1. The minimal prerequisite on  $\mathcal{E}$  that we need here is that the set of smooth and compactly supported functions is a dense and stable by L subset of  $\mathcal{E}$ . The link between the semi-group and the generator (and resolvent) is the subject of the Hille-Yoshida theory, which lies outside the scope of this introduction (see [169]). Let's consider two examples:

• For a diffusion process (1.1) with Lipschitz continuous drift b and diffusion coefficient  $\sigma$ , we have  $C_c^2 \subset \mathcal{D}(L)$ , where  $C_c^k$  denotes the k times differentiable functions with compact support, and for  $f \in C^2$ :

$$Lf = b \cdot \nabla f + \frac{1}{2} \operatorname{tr} \left( \sigma^t \nabla^2 f \sigma \right).$$
(1.4)

• For a PDMP, we have  $\mathcal{C}_c^1 \subset \mathcal{D}(L)$ , and for  $f \in \mathcal{C}_c^1$ :

$$Lf(x) = \nabla\varphi(x) \cdot \nabla f(x) + \lambda(x) \int (f(y) - f(x)) Q(x, \mathrm{d}y).$$
(1.5)

The choice of  $\mathcal{E}$  depends on the topology in which the long-term convergence of the semi-group occurs and must be specified. Various topologies, such as weak, Wasserstein,  $L^2$ ,  $H^k$ , etc., are available, and each require a different method to prove convergence. Depending on the process, understanding which given norm or method is best suited is a critical part of the study.

Now, the remainder of this introduction is divided into two sections: first, Section 1.2 will elaborate on the different methods used to study the convergence of Markov processes towards their invariant probability measure; then, Section 1.3 will introduce the algorithms to which we will apply the previously presented methods, as well as the contributions of this thesis.

## 1.2 Some methods to study the long-time behavior of Markov processes

Two families of methods are used in this PhD to study the long-time behavior of Markov processes, giving explicit rates in the convergences (1.3). The first family of methods, presented in Section 1.2.1, corresponds to PDE methods, used for the convergence in entropy, and in  $L^2$  or Sobolev spaces. We will then delve into coupling methods in Section 1.2.2.

## **1.2.1** The PDE methods: Entropy, $L^2$ , $H^k$ convergence

#### Link between PDE and Markov process

The first family of methods to study the long-time evolution of the law of a Markov process are based on PDEs, and the use of the generator. As seen previously, the semi-group P of a Markov process which admits a generator L satisfies for  $f \in \mathcal{D}(L)$  the Kolmogorov (or Fokker-Planck) equation:

$$\partial_t P_t f = L P_t f = P_t L f. \tag{1.6}$$

Hence, if L is a differential operator, PDE methods can be employed to study the long-time behavior of the semi-group. There is also a dual formulation. Denote by l the Lebesgue measure,  $g_t$  the law of the Markov process, and  $L^*$  the dual of L in  $L^2(l)$ . Then, g is a weak solution to the PDE:

$$\partial_t g_t = L^* g_t. \tag{1.7}$$

This provides two points of view for the long-time convergence of a Markov process, depending on the context. For instance, equation (1.6) in Sobolev spaces, as done in Chapters 2 and 3, allows for estimates on  $P_t f$ , whereas equation (1.7) is better suited for the convergences in entropy (see below), or for the convergence in probability of the process studied in Chapter 4. In the remainder of this section, we will consider for simplicity diffusion processes with an invariant measure  $\mu$ , but other kinds of processes (such as PDMPs giving rise to integro-differential equations) can be considered. In those cases, we will have  $E = \mathbb{R}^d$ . We will have a look at two families of diffusions:

**Elliptic diffusions**: An elliptic diffusion Z is a diffusion in E that solves

$$\mathrm{d}Z_t = b(Z_t)\mathrm{d}t + \sigma(Z_t)\mathrm{d}B_t,$$

where the diffusion matrix satisfies

$$\sigma(x)^T \sigma(x) \ge c I_d,$$

where  $\sigma^T$  denote the adjoint of  $\sigma$ , for some c > 0 and all  $x \in E$ , in the sense that  $\sigma^T \sigma - cI_d$  is everywhere positive definite. The generator of these diffusions has a regularizing effect, making them suitable for PDE methods.

**Proposition 1.2.1.** Suppose that  $b \in L^p_{loc}$  for some p > d + 2, and  $\sigma^T \sigma$  and its inverse are bounded. Then the law of  $Z_t$  admits a Holder continuous density  $g_t$  with respect to the Lebesgue measure. If, in addition, we suppose b and  $\sigma$  Lipschitz continuous and  $C^p$ , then  $g_t \in C^p$  and  $P_t f \in C^p$ .

Proofs of those statements can be found in [18, 102]. The most important example of an elliptic diffusion in this thesis is given by the so-called overdamped Langevin process

$$\mathrm{d}X_t = -\nabla U(X_t)\mathrm{d}t + \sqrt{2\beta^{-1}}\mathrm{d}B_t,\tag{1.8}$$

where  $U : \mathbb{R}^d \to \mathbb{R}_+$  will usually be taken as a smooth function. This example is crucial as it has an explicit invariant measure given by:

$$\nu_{\beta}(\mathrm{d}x) = Z_{\beta}^{-1} e^{-\beta U(x)} \mathrm{d}x, \qquad (1.9)$$

where  $Z_{\beta}$  is a normalisation constant.

**Hypoelliptic diffusions**: Since the law of an elliptic diffusion with smooth coefficients at a given time has a smooth density, it is natural to wonder which other kind of generator will generate a smooth density, and hence be adapted to PDE methods. This corresponds to the notion of a parabolic hypoelliptic operator.

**Definition 1.2.1.** We say that L is parabolic hypoelliptic or that  $\partial_t - L$  is hypoelliptic if for all  $u : \mathbb{R}_+ \times \mathbb{R}^d \to \mathbb{R}$ :

$$\partial_t u - L u \in \mathcal{C}^\infty \implies u \in \mathcal{C}^\infty.$$

If L is the generator of the semi-group P and is parabolic hypoelliptic, then Equation (1.6) would yield the smoothness of the semi-group. Let's now give a criterion for parabolic hypoellipticity. Given two vector fields A, and B on  $\mathbb{R}^d$ , we denote their Lie bracket as

$$[A,B](x) = DB(x)A(x) - DA(x)B(x),$$

where  $(DA)_{i,j} = \partial_j A_i$ .

**Definition 1.2.2.** For a family of vector fields  $(A_i)_{0 \le i \le m}$  on  $\mathbb{R}^d$ , define

$$V_0 = \{A_i, i \in [\![1,m]\!]\}, V_{k+1} = V_k \cup \{[B,A_i], B \in V_k, i \in [\![0,m]\!]\}.$$

We say that  $(A_i)_{0 \le i \le m}$  satisfies the Hörmander's bracket condition if

$$Span\left(\bigcup_{k=0}^{\infty}V_{k}\right)=\mathbb{R}^{d}.$$

**Proposition 1.2.2** (Hörmander's theorem). Identify any vector field with the corresponding differential form. Then if  $L = A_0 + \sum_{i=1}^{m} A_i^2$ , the  $(A_i)_{0 \le i \le m}$  are bounded, and satisfy the Hörmander's bracket condition 1.2.2, then L is parabolic hypoelliptic.

Proof of this theorem and variations can be found in [77, 28]. The most important example of a hypoelliptic process for this thesis is the Langevin one:

$$\begin{cases} dX_t = Y_t dt, \\ dY_t = -\nabla U(X_t) dt - \gamma Y_t dt + \sqrt{2\gamma\beta^{-1}} dB_t. \end{cases}$$
(1.10)

Here,  $d = 2n, X \in \mathbb{R}^n$  and  $Y \in \mathbb{R}^n$  are not Markov processes, but (X, Y) is, and the fact that the noise only acts on Y makes this process non-elliptic. Its generator is given by

$$L = y \cdot \nabla_x - \nabla \cdot \nabla_y - \gamma y \cdot \nabla_y + \gamma \beta^{-1} \Delta_y = A_0 + \gamma \beta^{-1} \sum_{i=1}^n \partial_{y_i}^2.$$
(1.11)

We have the commutator

$$[\partial_{y_i}, y \cdot \nabla_x] = -\partial_{x_i}$$

Hence, when  $\nabla U$  is bounded, (X, Y) is hypoelliptic, as

$$Span(V_0 \cup V_1) = \mathbb{R}^d.$$

Under mild assumptions, the Langevin process admits a unique invariant probability measure given by

$$\mu_{\beta}(\mathrm{d}x,\mathrm{d}y) = \frac{e^{-\beta H(x,y)}}{\sqrt{\pi\beta^{-1}}^d Z_{\beta}} \mathrm{d}x\mathrm{d}y, \qquad (1.12)$$

n

where  $Z_{\beta}$  is the same normalisation constant as in (1.9). Many other hypoelliptic processes exist, such as the Generalised Langevin process [168, 132], or the chains of oscillators [125].

As we will see in Section 1.3, we are not only interested in the smoothness of the law or of  $P_t f$ , as it is not sufficient for the study of the stochastic algorithm. Since we study Markov processes that possess an invariant measure  $\mu$ , we may introduce the following Hilbert spaces:

$$L^{2}(\mu) = \left\{ f: E \to \mathbb{R}, \text{ measurable}, \int_{E} f^{2} \mathrm{d}\mu < \infty \right\},$$

or

$$H^{k}(\mu) = \left\{ f \in L^{2}(\mu), \forall \alpha \in \mathbb{N}^{d}, |\alpha|_{1} \leq k, \int_{E} |\partial^{\alpha} f|^{2} \, \mathrm{d}\mu < \infty \right\}.$$

for some  $k \in \mathbb{N}$ . Both are Hilbert spaces, with scalar product

$$\langle f,g\rangle_{L^2(\mu)}=\int_E fg\mathrm{d}\mu,\quad \langle f,g\rangle_{H^k(\mu)}=\sum_{\alpha\in\mathbb{N}^d, |\alpha|_1\leqslant k}\int_E\partial^\alpha f\partial^\alpha g\mathrm{d}\mu.$$

In many cases, we are interested in  $L^{\infty}$  bounds on  $P_t f$ , and Sobolev embedding combined with convergence in  $H^k(\mu)$ , will be the means to achieve such bounds, as discussed in Chapters 2, 3 and 4. The remainder of Section 1.2.1 will be devoted to presenting the usual methods for proving convergence in such spaces, employing functional inequalities. We will first delve into the elliptic and reversible case before introducing Gamma calculus for the study of the long-time behavior of non-elliptic processes.

#### The elliptic/reversible case

In order to later justify our computations, we enforce the following

**Assumption 1.2.1.**  $\mathcal{E}$  is a Hilbert space, the Markov process X defines a semi-group  $P_t : \mathcal{E} \to \mathcal{E}$ , and there exists a dense subset  $\mathcal{A} \subset \mathcal{E}$  such that  $\mathcal{A} \subset \mathcal{D}(L)$  and  $h \in \mathcal{A} \implies h^2 \in \mathcal{A}$ .

In all the spaces  $\mathcal{E}$  we will consider, we will have that  $\mathcal{C}_c^{\infty}$  (the set of smooth functions with compact support) is a dense subset of  $\mathcal{E}$ . Moreover, in the case of diffusions with regular

coefficients, we have  $\mathcal{C}_c^{\infty} \subset \mathcal{D}(L)$ , and hence we may think of  $\mathcal{A}$  as  $\mathcal{C}_c^{\infty}$ . In this section, we also suppose that the process X is reversible:

**Assumption 1.2.2.**  $\mu$  denotes an invariant measure for the semi-group  $(P_t)_{t \ge 0}$ ,  $\mathcal{E} \subset L^2(\mu)$ , and X is reversible with respect to  $\mu$ , in the sense that for all  $t \ge 0$ ,  $P_t$  is self-adjoint in  $L^2(\mu)$ .

Define for  $h \in \mathcal{A}$  the so-called *carré du champ* 

$$\Gamma(h) = \frac{1}{2} \left( L\left(h^2\right) - 2hLh \right).$$
(1.13)

This definition is motivated by the following formal computation:

$$\partial_t \int_E \left( P_t f - \mu \left( P_t f \right) \right)^2 \mathrm{d}\mu = 2 \int_E \left( P_t f - \mu \left( P_t f \right) \right) L P_t f \mathrm{d}\mu - \int_E L \left( P_t f \right)^2 \mathrm{d}\mu = -2 \int_E \Gamma(P_t f) \mathrm{d}\mu.$$

**Definition 1.2.3.** We say that  $\mu$  satisfies a Poincaré inequality (with respect to  $\Gamma$ ) if there exists  $\lambda > 0$  such that:

$$\lambda \int_{E} \left(h - \mu(h)\right)^{2} d\mu \leqslant \int_{E} \Gamma(h) d\mu, \qquad (1.14)$$

for all  $h \in \mathcal{A}$ .

In the reversible case, this Poincaré inequality is related to the so-called spectral gap of the process, since  $\lambda$  can be taken as the greatest non-zero eigenvalue of L. Hence, it is only natural that it is related to the long time behavior of the semi-group, as shown by the following:

**Proposition 1.2.3.** Under Assumption 1.2.1, the existence of a Poincaré inequality is equivalent to: for all  $t \ge 0$ 

$$\|P_t f - \mu(f)\|_{L^2(\mu)} \leqslant e^{-2\lambda t} \|f - \mu(f)\|_{L^2(\mu)}, \qquad (1.15)$$

for all  $f \in L^2(\mu)$ .

Hence, Poincaré inequality gives us convergences towards equilibrium and even speed of convergence through  $\lambda$ . Take the example of the overdamped Langevin process (1.8). In this case:

$$\Gamma(h) = \beta^{-1} |\nabla h|^2$$

and the stationary measure is the Gibbs measure (1.9). It can be shown that this probability measure satisfies the Poincaré inequality

$$\lambda \int_{\mathbb{R}^d} \left(h - \nu_\beta(h)\right)^2 \mathrm{d}\nu_\beta \leqslant \int_{\mathbb{R}^d} |\nabla h|^2 \mathrm{d}\nu_\beta$$

if for instance  $U \in \mathcal{C}^2(\mathbb{R}^d)$  and

$$\inf |\nabla U|^2 - \Delta U > -\infty,$$

yielding exponential convergence in  $L^2(\nu_{\beta})$ . Estimates on the Poincaré constant, and more particularly its dependency on the dimension and the temperature  $\beta^{-1}$ , are of great importance to analyse the speed of convergence of these Markov processes and their application to stochastic algorithms, as we will see in Section 1.3. The *carré du champs* also yields local information on the semi-group, but those will not be of interest in this thesis. Moreover, other functional inequalities exist, like the log-Sobolev and the Sobolev inequalities, which yield convergences in other spaces, as well as other properties such as hyper and ultracontractivity, see [6]. For instance, using a log-Sobolev inequality instead of a Poincaré inequality, one could prove the convergence in entropy, where the entropy of a probability measure  $\nu$  with respect to  $\mu$  is defined as:

Ent 
$$(\nu|\mu) = \begin{cases} \int \ln\left(\frac{\mathrm{d}\nu}{\mathrm{d}\mu}\right) \mathrm{d}\nu & \text{if } \nu \ll \mu, \\ +\infty & \text{else.} \end{cases}$$

This does not define a distance, but the entropy is particularly well-adapted to particle systems or high dimensional settings, as this quantity scales better than  $L^p$  distances with the dimension and is an upper bound for some Wasserstein distances, that will be defined in Section 1.2.2 (such as the total variation distance with the Pinsker inequality).

The method presented in this Section does not require reversibility, but only a Poincaré inequality (1.14). It however fails as such in the case of kinetic process. Let's consider the case of the Langevin process (1.10). Its *carré du champs* is expressed as:

$$\Gamma(h) = \gamma \beta^{-1} |\nabla_y h|^2$$

Considering a function h that only depends on x, an inequality of the form:

$$\lambda \int_{\mathbb{R}^{2d}} \left(h - \mu_{\beta}(h)\right)^2 \mathrm{d}\mu_{\beta} \leqslant \int_{\mathbb{R}^{2d}} |\nabla_{y}h|^2 \mathrm{d}\mu_{\beta}$$

cannot hold with  $\lambda > 0$ . According to Proposition 1.2.3, it is not possible to achieve convergence as for the overdamped Langevin process. The next section is dedicated to adapting the previously discussed method to kinetic processes.

#### Hypocoercivity and generalized Gamma calculus

The Langevin process is not coercive, as in Proposition 1.2.3, but is hypocoercive, a term introduced in [163].

**Definition 1.2.4.** A semi-group P with invariant measure  $\mu$  is said to be hypocoercive on  $\mathcal{E}$  if there exist C > 1 and  $\lambda > 0$  such that:

$$\|P_t f - \mu(f)\|_{\mathcal{E}} \leqslant C e^{-\lambda t} \|f - \mu(f)\|_{\mathcal{E}}, \qquad (1.16)$$

for all  $f \in \mathcal{E}$ .

If C = 1, then we revert to the case of Section 1.2.1, and the process is simply coercive. However, when C > 1, this convergence is not equivalent to the Poincaré inequality with respect to the *carré du champ*. Contrary to the convergence (1.15), this property is invariant by replacing the distance by an equivalent one on  $\mathcal{E}$ . In fact, Villani developed in his famous memoir [163] a method to show hypocoercive convergence by using a modified norm. This method has later been formalized using the so-called Gamma calculus [9, 134]. Demonstrating such convergence is usually equivalent to finding the correct distance for which the semi-group is actually coercive, as we will now see. We are interested in the evolution of quantities of the form:

$$\int_E \phi\left(P_t f\right) \mathrm{d}\mu,$$

for quadratic functionals  $\phi : \mathcal{C}^{\infty}(\mathcal{X}) \to \mathcal{C}^{\infty}(\mathcal{X})$  of the form:

$$\phi(h) = |A\nabla^{\alpha}h|^2 = \sum_{\alpha_1 + \dots + \alpha_{2d} = \alpha} |A_{\alpha_1,\dots,\alpha_{2d}}\partial_{z_1}^{\alpha_1}\dots\partial_{z_{2d}}^{\alpha_{2d}}h|^2,$$

for some  $\alpha \in \mathbb{N}$  and tensor A, as well as linear combination of such functionals. To this end, define:

$$\Gamma_{L,\phi}(h) = \frac{1}{2} \left( L(\phi(h)) - D_h \phi(h) Lh \right), \qquad (1.17)$$

where  $D_h \phi$  denote the differential operator of  $\phi$ . This definition is motivated by the following formal computation:

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{E} \phi\left(P_{t}f\right) \mathrm{d}\mu = -\int_{E} \Gamma_{L,\phi}\left(P_{t}f\right) \mathrm{d}\mu,\tag{1.18}$$

which uses that  $\mu$  is invariant for  $P_t$ . If we can get an inequality of the form

$$c \int_{E} \phi(h) \mathrm{d}\mu \leqslant \int_{E} \Gamma_{L,\phi}(h) \mathrm{d}\mu \tag{1.19}$$

for all smooth  $h \in \mathcal{A}$ , then this implies the convergence:

$$\int_{E} \phi\left(P_{t}f\right) \mathrm{d}\mu \leqslant e^{-ct} \int_{E} \phi\left(f\right) \mathrm{d}\mu.$$
(1.20)

Therefore, if  $\int_E \phi(f) d\mu$  is equivalent to the  $L^2$  or  $H^k$  distance, then our goal would be achieved. This tool is particularly well adapted for the study of quadratic functional such as the  $L^2$  or  $H^k$  norms, due to the following identity:

**Proposition 1.2.4.** If there exists  $A = (A_1, \dots, A_p) : \mathcal{C}^{\infty} \to (\mathcal{C}^{\infty})^p$  a linear operator such that  $\phi(h) = |Ah|^2$ , then

$$\Gamma_{L,\phi}(h) = \Gamma_{L,2}(Ah) + Ah \cdot [L,A]h,$$

where  $\Gamma_{L,2}(Ah) = \sum_{i=1}^{p} \Gamma(A_ih)$ ,  $\Gamma$  is the carré du champ defined in (1.13), and  $[L, A] = ([L, A_1], \cdots, [L, A_p])$ .

Let's now see how this applies to the  $L^2(\mu)$  and  $H^1(\mu)$  of the Langevin process, in the case of a bounded hessian potential:  $\|\nabla^2 U\|_{\infty} < \infty$ . Let's recall that under this condition, the Langevin process admits the measure  $\mu_{\beta}$  defined in (1.12) as its unique invariant measure, and its generator is given by:

$$L = y \cdot \nabla_x - \nabla U \cdot \nabla_y - \gamma y \cdot \nabla_y + \gamma \beta^{-1} \Delta_y.$$
(1.21)

We will show convergence in  $H^1$  using Gamma calculus. We have:

$$[L, \nabla_x] = \nabla^2 U \nabla_y, \quad [L, \nabla_y] = -\nabla_x + \gamma \nabla_y$$

Hence using Young inequality we have:

$$\begin{aligned} (\nabla_x - \nabla_y) g \cdot [L, \nabla_x - \nabla_y] g &= (\nabla_x - \nabla_y) g \cdot \left( \nabla^2 U \nabla_y + \nabla_x - \gamma \nabla_y \right) g \\ &\geqslant \frac{1}{2} |\nabla_x g|^2 - 2 \left( 1 + \gamma + \|\nabla^2 U\|_{\infty} \right)^2 |\nabla_y g|^2, \end{aligned}$$

as well as

$$\nabla_y g \cdot [L, \nabla_y] g = \nabla_y g \cdot (-\nabla_x + \gamma \nabla_y) g \ge -\frac{1}{4} |\nabla_x g|^2 - |\nabla_y g|^2.$$

Thus, with

$$\phi(h) = ch^2 + |(\nabla_x - \nabla_y)h|^2 + |\nabla_y h|^2, \ c = \gamma^{-1}\beta \left(2 + 2\left(1 + \gamma + \|\nabla^2 U\|_{\infty}\right)^2\right),$$

Proposition 1.2.4, the linearity of  $\Gamma_{\phi}$  with respect to  $\phi$ , as well as the positivity of  $\Gamma$  yield that

$$\Gamma_{\phi}(g) \ge c\Gamma(g) + (\nabla_x - \nabla_y) g \cdot [L, \nabla_x - \nabla_y] g + \nabla_y g \cdot [L, \nabla_y] g \ge \frac{1}{2} |\nabla g|^2.$$
(1.22)

The measure  $\mu_{\beta}$  satisfies the Poincaré inequality

$$\lambda \int_{\mathbb{R}^{2d}} \left(h - \mu_{\beta}(h)\right)^2 \mathrm{d}\mu_{\beta} \leqslant \int_{\mathbb{R}^{2d}} |\nabla h|^2 \mathrm{d}\mu_{\beta}$$

for some  $\lambda > 0$ , which, combined with inequality (1.22), yields

$$-\int_{\mathbb{R}^{2d}}\Gamma_{\phi}(g)\mathrm{d}\mu_{\beta}\leqslant -\kappa\int_{\mathbb{R}^{2d}}\phi(g)\mathrm{d}\mu_{\beta}$$

for some  $\kappa > 0$ . This inequality, at least formally, yields the  $H^1$  convergence:

$$\int_{\mathbb{R}^{2d}} \left(P_t f\right)^2 + |\nabla P_t f|^2 \mathrm{d}\mu_\beta \leqslant C e^{-ct} \int_{\mathbb{R}^{2d}} f^2 + |\nabla f|^2 \mathrm{d}\mu_\beta$$

for some C, c > 0 that can be made explicit, and all  $f \in H^1(\mu)$ . Actually, using an  $L^2 \to H^1$  hypoelliptic regularization [81]

$$\int_{\mathbb{R}^{2d}} \left( P_t f \right)^2 + |\nabla P_t f|^2 \mathrm{d}\mu_\beta \leqslant \frac{K}{t^\alpha} \int_{\mathbb{R}^{2d}} f^2 \mathrm{d}\mu_\beta, \quad \forall t \in ]0, 1]$$

for some  $K, \alpha > 0$ , this also yield the  $L^2$  convergence:

$$\int_{\mathbb{R}^{2d}} \left(P_t f\right)^2 \mathrm{d}\mu_\beta \leqslant C \frac{e^{-ct}}{\min(1,t)^{\alpha}} \int_{\mathbb{R}^{2d}} f^2 \mathrm{d}\mu_\beta.$$

These computations can be made rigorous using the Lumer-Phillips theorem. All computations are conducted on the semi-group's generator, and this theorem, rooted in the Hille-Yoshida theory, allows us to transfer a property from the generator to the semi-group. In the case of the Langevin process, Villani's method worked thanks to the bracket condition of hypoellipticity given in Definition 1.2.2. Regardless, some processes might exhibit hypocoercive convergence without being hypoelliptic, as is the case for some kinetic PDMPs. In these case, other methods exist to show hypocoercivity, such as the ones developed in [50, 1].

This method will be employed in Chapters 2, 3 and 4, in  $H^k$  spaces, to obtain either estimates on the semi-group or on the time-marginals of the law of the process under different sets of assumptions on the process. The proof of the inequality (1.19) will consistently rely on the use of the Poincaré inequality throughout this thesis. In the case of non-quadratic  $\phi$ , one could also use other functional inequalities such as the Log-Sobolev inequality when  $\phi(h) = h \ln(h)$ represents the entropy.

### 1.2.2 Wasserstein distance and coupling methods

The second family of methods we are going to look at are coupling methods, based on Wasserstein distances. These distances are typically introduced within the context of optimal transport and have the advantage that they don't require *a priori* the existence of an invariant probability measure, but will in fact be used for the proof of existence of such a measure, as we will discuss later on. Moreover, they are well-suited for studying the long-term behavior of Markov processes,

thanks to their probabilistic interpretation, even in cases where the generator is not a differential operator.

### The topology

**Definition 1.2.5.** For any two probability measures  $\nu_1, \nu_2$ , we define a coupling of  $\nu_1$  and  $\nu_2$  as  $(X^1, X^2)$  such that the law of  $X^1$  (resp.  $X^2$ ) is  $\nu_1$  (resp.  $\nu_2$ ).

Given a distance **d** on E,  $p \in \mathbb{N}$ , and  $\nu_1, \nu_2 \in \mathcal{M}^1(E)$ , we define the associated Wasserstein distance as follows:

$$W_{\mathbf{d},p}(\nu_1,\nu_2) = \left(\inf\left\{\mathbb{E}\left(\mathbf{d}^p(X^1,X^2)\right), (X^1,X^2) \text{ coupling of } \nu_1 \text{ and } \nu_2\right\}\right)^{1/p}.$$

Given a Markov semi-group P, we call a coupling of  $(\nu_1 P_t)_{t \ge 0}$  and  $(\nu_2 P_t)_{t \ge 0}$  a stochastic process  $(X_t^1, X_t^2)_{t \ge 0}$  such that  $(X_t^1)_{t \ge 0}$  and  $(X_t^2)_{t \ge 0}$  are Markov processes of semi-group  $(P_t)_{t \ge 0}$  and initial condition  $\nu_1$  for  $X^1$  and  $\nu_2$  for  $X^2$ .

When p = 1, it is also referred to as the Kantorovich distance. This definition yields a distance on the space:

$$\mathcal{M}_p^1(E) = \left\{ \nu \in \mathcal{M}^1(E), \int_E \mathbf{d}(x, x_0)^p \nu(\mathrm{d}x) < \infty \right\}$$

and this space is independent of the choice of  $x_0 \in E$ . We will be mostly interested in the case p = 1 and p = 2. If **d** is bounded, then all these distances are equivalent. A specific instance is when  $\mathbf{d}(x, y) = 2\mathbb{1}_{x \neq y}$  and p = 1, leading to the recovery of the total variation distance denoted by  $W_{\mathbf{d}}(\nu_1, \nu_2) = \|\nu_1 - \nu_2\|_{TV}$ . The Wasserstein distances satisfy the following properties:

**Proposition 1.2.5.** • If  $(E, \mathbf{d})$  is a Polish space, then  $(\mathcal{M}^1_p, W_{\mathbf{d},p})$  is also a Polish space.

- In this case, there exists an optimal coupling, meaning a coupling  $(X^1, X^2)$  of  $\nu_1$  and  $\nu_2$  such that  $W_{\mathbf{d}}(\nu_1, \nu_2) = \mathbb{E}(\mathbf{d}(X^1, X^2))$ .
- $\|\cdot\|_{TV}$  makes  $\mathcal{M}^1(E)$  complete for any set E, and there always exists an optimal coupling for this distance.

Proofs of these facts can be found in [164, Theorem 6.18 and Theorem 4.1]. These properties hold even though  $(E, 2\mathbb{1}_{x\neq y})$  is not separable as soon as E is uncountable. In the sequel, we will omit the dependence on **d**, and  $W_p$  will denote  $W_{\mathbf{d},p}$ . We will also refer to the law of the pair  $(X^1, X^2)$  as the optimal coupling.

Now, let's explain why the Wasserstein distance is well-suited to the study of the long-time behavior of Markov processes. The objective is to establish a coupling for all  $x, y \in E$  of  $\delta_x P_t$ and  $\delta_y P_t$ . Typically, such a coupling should ensure that  $X_s^1 = X_s^2 \implies X_t^1 = X_t^2, \forall t \ge s$ . When this happens, we say that the processes  $X^1$  and  $X^2$  have coupled at time s. Given that the Wasserstein distance is defined by an infimum, any coupling, not necessarily an optimal one, can provide an upper bound for the distance, as for all  $t \ge 0$  the following inequality holds:

$$W_{\mathbf{d}}(\nu_1 P_t, \nu_2 P_t) \leq \mathbb{E}\left(\mathbf{d}(X_t^1, X_t^2)\right).$$

More precisely we have:

**Proposition 1.2.6.** Suppose that  $(E, \mathbf{d})$  is a Polish space or that  $\mathbf{d}(x, y) = 2\mathbb{1}_{x\neq y}$ , and that there exist  $t_0 > 0$ ,  $0 < \kappa < 1$ ,  $\alpha > 0$  such that for all  $x, y \in E$ , there exists two couplings,  $(X^x, X^y)$ 

and  $(\bar{X}^x, \bar{X}^y)$ , of  $(\delta_x P_t)$  and  $(\delta_y P_t)$  such that for all  $t \ge 0$ ,  $(x, y) \mapsto (X_t^x, X_t^x)$  is measurable,

$$\mathbb{E}\left(\mathbf{d}\left(X_{t_0}^x, X_{t_0}^y\right)^p\right) \leqslant \kappa^p \mathbf{d}(x, y)^p,\tag{1.23}$$

and

$$\mathbb{E}\left(\mathbf{d}\left(X_{t}^{x}, X_{t}^{y}\right)^{p}\right) \leqslant \alpha^{p} \mathbf{d}(x, y)^{p}, \forall t \in [0, t_{0}].$$
(1.24)

Then the semi-group  $(P_t)$  admits a unique invariant measure  $\mu$ , and we have that for all  $\nu \in$  $\mathcal{M}^1_n(E)$ :

$$W_{p}\left(\nu P_{t},\mu\right)\leqslant Ce^{-ct}W_{p}\left(\nu,\mu\right),$$

for some constants C, c > 0 explicit in  $\kappa$ ,  $\alpha$  and  $t_0$ .

*Proof.* Fix  $\nu_1, \nu_2 \in \mathcal{M}^1(E)$ , and let  $(X_0, Y_0)$  be an optimal coupling of  $\nu_1, \nu_2$ . Denote  $X_t = X_t^{X_0}$ and  $Y_t = Y_t^{Y_0}$ .  $(X_{t_0}, Y_{t_0})$  is a coupling of  $\nu_1 P_{t_0}$  and  $\nu_2 P_{t_0}$ , and hence using the definition of the Wasserstein distance and conditioning we get:

$$\begin{split} W_p^p\left(\nu_1 P_{t_0}, \nu_2 P_{t_0}\right) &\leqslant \mathbb{E}\left(\mathbf{d}\left(X_{t_0}, Y_{t_0}\right)^p\right) = \int_{E \times E} \mathbb{E}\left(\mathbf{d}\left(X_{t_0}^x, Y_{t_0}^y\right)^p\right) \pi(\mathrm{d}x \mathrm{d}y) \\ &\leqslant \kappa^p \int_{E \times E} \mathbf{d}(x, y)^p \pi(\mathrm{d}x, \mathrm{d}y) = \kappa^p W_p^p\left(\nu_1, \nu_2\right). \end{split}$$

Thus, the application  $\nu \mapsto \nu P_{t_0}$  is a contraction on the complete space  $\mathcal{M}_p^1$ , and the Banach fixed point theorem yield that it admits a unique fixed point  $\mu \in \mathcal{M}_p^1(E)$ , and  $\mu$  is such that

$$W_{p}\left(\nu P_{t_{0}}^{k},\mu\right)\leqslant\kappa^{k}W_{p}\left(\nu,\mu\right).$$

An invariant measure of the semi-group is necessarily a fixed point of  $P_{t_0}$ . If there exists  $t \ge 0$ such that  $\mu P_t \neq \mu$ , then, by the semi-group property of P,  $\mu P_t$  would be a second fixed point of  $P_{t_0}$ . This implies that  $\mu$  is the unique invariant measure of the semi-group. Integrating Inequality (1.24) with respect to the initial condition implies that for all  $t \in [0, t_0]$ :

$$W_p\left(\nu_1 P_t, \nu_2 P_t\right) \leqslant C W_p\left(\nu_1, \nu_2\right).$$

Therefore, we may write

$$W_p(\nu P_t, \mu) = W_p(\nu P_t, \mu P_t) \leqslant \kappa^{\lfloor t/t_0 \rfloor} W_p(\nu P_{t-\lfloor t/t_0 \rfloor t_0}, \mu P_{t-\lfloor t/t_0 \rfloor t_0}) \leqslant \alpha \kappa^{\lfloor t/t_0 \rfloor} W_p(\nu, \mu)$$
  
ch concludes the proof with  $C = \alpha \kappa^{-1}$  and  $c = -t_0^{-1} \ln(\kappa)$ .

which concludes the proof with  $C = \alpha \kappa^{-1}$  and  $c = -t_0^{-1} \ln(\kappa)$ .

Constructing a coupling that satisfies (1.24) is typically straightforward, and the main challenges arise in building the coupling from (1.23).

#### A first example

Let's consider an example to illustrate how this works. Take the following SDE on  $\mathbb{R}^d$ :

$$\mathrm{d}X_t = b(X_t)\mathrm{d}t + \mathrm{d}B_t,\tag{1.25}$$

where B is a Brownian motion, and b satisfies the following one-sided Lipschitz condition:

$$\exists \kappa > 0, \, \forall x, y \in \mathbb{R}^d, \, (b(x) - b(y)) \cdot (x - y) \leqslant -\kappa |x - y|^2. \tag{1.26}$$

This condition implies that the deterministic drift tends to bring points in space closer together. It is for example satisfied by the gradient of convex functions. Let  $\mathbf{d}(x,y) = \|\cdot\|_2$  be the Euclidean norm. Fix  $x, y \in \mathbb{R}^d$  and a Brownian motion B. We can then define the so-called parallel coupling:

$$X_{t} = x + \int_{0}^{t} b(X_{s}) \,\mathrm{d}s + B_{t}, \qquad \qquad Y_{t} = y + \int_{0}^{t} b(Y_{s}) \,\mathrm{d}s + B_{t}. \tag{1.27}$$

They are both driven by the same Brownian motion. Ito's formula yields:

 $d|X_t - Y_t|_2^2 = 2(b(X_t) - b(Y_t)) \cdot (X_t - Y_t) dt \leq -2\kappa |X_t - Y_t|_2^2 dt,$ 

and hence

$$||X_t - Y_t||_2 \leqslant e^{-\kappa t} ||X_0 - Y_0||_2$$

which implies inequality (1.23). We can conclude that the SDE (1.25) defines a Markov process that admits a unique invariant measure, and that the law of this process converges exponentially fast towards this invariant measure at rate  $\kappa$  for the distance  $W_p$ , for all  $p \ge 1$ . In the case where (1.26) only holds outside a compact set, contraction in  $W_1$  was proved in [55] for a distance **d** equivalent to the euclidean one, using the famous reflection coupling introduced in [117]. In the case of non-elliptic processes, a mix between parallel and reflection coupling was used in [56] to prove convergence of the Langevin process (1.10). To my knowledge, it remains open to know if it is possible to construct a metric on  $\mathbb{R}^d$  and a coupling to prove the Wasserstein convergence for a process satisfying the Hörmander's bracket condition from Definition 1.2.2 and a contractivity assumption similar to (1.26) outside a compact set.

#### Convergence in weighted total variation and Lyapunov functions

We have shown in the previous section the convergence of the process (1.25) in Wasserstein distance  $W_p$  under a uniform one-sided Lipschitz condition on the drift. Now, we are interested in the convergence of the same process in total variation distance, i.e., considering  $\mathbf{d}(x, y) = 2\mathbb{1}_{x \neq y}$ . Denote the process by  $X^x$  when the initial condition is  $x \in \mathbb{R}^d$ , and let's suppose that b is Lipschitz continuous. Then, from Proposition 1.2.1, we get that the random variable  $X_t^x$  admits a density with respect to the Lebesgue measure, denoted by  $h_t(x, \cdot)$ , and the mapping  $(x, y) \mapsto h_t(x, y)$  is continuous and non-negative. Actually, using controllability arguments, one can show that it is positive. Suppose we have the additional condition

$$\inf_{x,y\in\mathbb{R}^d}h_{t_0}(x,y)>0,$$

which would hold if the diffusion was living in a compact space, like the periodic torus, or if a Sobolev inequality would hold, thus implying ultracontractivity. In that case, we would satisfy condition (1.23): the uniform creation of density implies exponential convergence towards a unique equilibrium in total variation distance. Indeed, we can construct an explicit coupling satisfying (1.23). Let's write  $c = \inf_{\mathbb{R}^d \times \tilde{K}} h_{t_0}$ , where  $\tilde{K}$  is a compact set of Lebesgue measure 1, and denote the Lebesgue measure by l. Notice that

$$\mathcal{L}aw\left(X_{t_0}^x\right) = (1-c)\frac{\mathcal{L}aw\left(X_{t_0}^x\right) - cl\left(\cdot \cap \tilde{K}\right)}{1-c} + cl\left(\cdot \cap \tilde{K}\right).$$

Now, let's define random variables as follows:

- Z is a random variable with law  $l\left(\cdot \cap \tilde{K}\right)$ .,
- B is a Bernoulli random variable with parameter c,
- For all  $x \in \mathbb{R}^d$ ,  $Z^x$  is a random variable with law  $\frac{\mathcal{L}aw(X_{t_0}^x) cl(\cdot \cap \tilde{K})}{1-c}$ .

Then, the law of the random variable

$$X^x = \mathbb{1}_{B=1}Z + \mathbb{1}_{B=0}Z^x$$

is indeed  $\delta_x P_{t_0}$ , and we have

$$\|\delta_x P_{t_0} - \delta_y P_{t_0}\|_{TV} \leq 2\mathbb{P} \left( X^x \neq X^y \right) \leq 2\mathbb{P} \left( B = 0 \right) = 2(1-c) = (1-c) \|\delta_x - \delta_y\|_{TV}$$

for all  $x, y \in \mathbb{R}^d$ , corresponding to (1.23). More generally, a condition

$$\forall x \in E, \mathbb{P}_x \left( X_{t_0} \in \cdot \right) \ge c\theta, \tag{1.28}$$

for some probability measure  $\theta \in \mathcal{M}^1$ , called a Doeblin condition, yields condition (1.23), and hence convergence in total variation distance. However, when the state space is not compact, the condition (1.28) might not hold uniformly on E. In this case, we need the existence of a so-called Lyapunov function. In many cases of interest, the process goes back to a compact set on which such an inequality holds.

**Definition 1.2.6.** A function  $V : E \to [1, \infty)$  is called a Lyapunov function if V has compact level sets and there exist C > 0 and  $0 < \alpha < 1$  such that:

$$\mathbb{E}_x\left(V\left(X_{t_0}\right)\right) \leqslant (1-\alpha)V(x) + \alpha C,\tag{1.29}$$

for some  $t_0 > 0$ .

Informally speaking, the existence of a Lyapunov function is equivalent to the fact that the process returns exponentially fast to compact sets. The usual way to prove the existence of a Lyapunov function is to use the generator characterisation: (1.29) is implied by the fact that  $V \in \mathcal{D}(L)$  and:

$$LV \leqslant -cV + C,$$

for some C, c > 0. Suppose that the one-sided Lipschitz condition holds for the drift, but only 'at infinity': there exists some compact set  $K \subset \mathbb{R}^d$  such that:

$$\forall x \in \mathbb{R}^d \setminus K, \, b(x) \cdot x \leqslant -\kappa \|x\|_2.$$

Then, if  $V(x) = x^2 + 1$ , we have:

$$LV = -2b(x) \cdot x + 2d \leqslant -2\kappa V(x) + 2(\kappa + d),$$

and hence V is a Lyapunov function. Now, let's write

$$\mathbf{d}_{V,\beta}(x,y) = 2\mathbb{1}_{x\neq y}(1+\beta V(x)+\beta V(y)).$$

The addition of a Lyapunov function to a Doeblin condition yields a Harris-type theorem, which corresponds to the following result:

**Proposition 1.2.7.** Suppose that V is a Lyapunov function satisfying (1.29), and that there exists a probability measure  $\theta$  such that for all  $x \in \{V < C'\}$ ,

$$\mathbb{P}_x \left( X_{t_0} \in \cdot \right) \geqslant c\theta,$$

for some C' > 2C and c > 0. Then there exists an explicit  $\beta > 0$  such that  $\nu \mapsto \nu P_{t_0}$  is a contraction in the space  $(\mathcal{M}^1(E), W_{\mathbf{d}_{V,\beta},1})$ , and all conclusions of Proposition 1.2.6 hold. Moreover, we have the total variation convergence:

$$\|\nu P_t - \mu\|_{TV} \leqslant De^{-dt} \int V(x)\nu(dx),$$

for some D, d > 0.

This kind of method first sprung from the work of Meyn and Tweedie, see [128]. Proof of this statement and variations can be found in [52, 79].

#### Particle systems

In this thesis, we primarily apply coupling methods to mean-field particle systems. For a given vector  $\mathbf{x} = (x_1, \ldots, x_N) \in E^N$ , we define its empirical measure as:

$$\pi(\mathbf{x}) = \frac{1}{N} \sum_{i=1} \delta_{x_i}.$$

It is a probability measure that counts the number of point  $(x_i)$  in any given set. A mean field particle system is a Markov process  $\mathbf{X} = (X^1, \ldots, X^N)$  with state space  $E^N$ , where N is the number of particles, such that each particle  $X^i$  only interacts with the other through their empirical measure  $\pi(\mathbf{X}_t)$ . In Section 1.3, we will introduce the so-called Fleming-Viot process, which is an instance of mean field particle system used for sampling. Another widely studied system is the famous McKean-Vlasov particle system:

$$\mathrm{d}X_t^i = -\nabla U\left(X_t^i\right) - \frac{1}{N} \sum_{j=1}^N \nabla W\left(X_t^i - X_t^j\right) \mathrm{d}t + \mathrm{d}B_t.$$
(1.30)

This kind of process is interesting because it models systems made up of interacting agents.

**Definition 1.2.7.** We say that a (finite) sequence of random variable  $(X_i)_{1 \leq i \leq N}$  is exchangeable if for all permutation  $\sigma : [\![1,N]\!] \to [\![1,N]\!], (X_{\sigma(i)})_{1 \leq i \leq N}$  is equal in law to  $(X_i)_{1 \leq i \leq N}$ .

In particular, if  $(X_i)$  is exchangeable, then  $X_i$  and  $X_j$  have the same laws for all i, j. The next proposition gives us the convergence of the empirical measure of independent random variables:

**Proposition 1.2.8** ([62]). If  $(Z_i)$  is a sequence of independent random variables with common law  $\nu$  with moments of order q > 0:

$$\int_{\mathbb{R}^d} x^q \nu(dx) < \infty$$

Then for all  $1 , the following convergence holds in <math>W_p$  distance :

$$\pi\left((Z_i)_{1\leqslant i\leqslant N}\right) = \frac{1}{N}\sum_{i=1}^N \delta_{Z_i} \underset{N\to\infty}{\to} \nu,$$

where is Wasserstein distance is relative to any norm on  $\mathbb{R}^d$ .

One interesting feature of such mean field systems is that they satisfy the so-called mean field limit: the particles interacting through their empirical measure are not independent but define an exchangeable system. Therefore, one may hope that as N goes to infinity, the convergence of the empirical measure towards their common law remains true, as soon as it is true at the initial time. This is called propagation of chaos, and would yield that as the number of particles goes to infinity, instead of seeing the interaction with all others, each particle only sees a deterministic averaged interaction which corresponds to their common law. Hence, they would become independent, and converge towards a non-linear process (in example (1.30), the McKean-Vlasov process). For such processes, the goal in Chapters 6 and 7 will be to show that the particle system converges towards an equilibrium, at a rate that is independent of the number of particles, see Section 1.3.3. To this end, we need a distance that scales well with the number of particles, and which is close to the distance at hand. For convergences in total variation distances, we define:

$$\mathbf{d}_N(\mathbf{x}, \mathbf{y}) = \frac{1}{N} \sum_{i=1}^N \mathbf{1}_{x_i \neq y_i}$$

or in the case where one needs a Lyapunov function:

$$\mathbf{d}_N(\mathbf{x}, \mathbf{y}) = \frac{1}{N} \sum_{i=1}^N \mathbf{1}_{x_i \neq y_i} \left( 1 + \beta V(x_i) + \beta V(y_i) \right).$$

The following lemma is the main motivation behind those distances:

**Proposition 1.2.9.** Let  $\nu^N P_t^N$  be the law of the particle system  $\mathbf{X}_t$ , and denote by  $\nu^{k,N}$  the *k*-first particles marginal. Suppose that  $\mathbf{X}_0$  is exchangeable, and that there exist C, c > 0 such that

$$W_{\mathbf{d}_{N}}\left(\nu^{N}P_{t}^{N},\mu^{N}
ight)\leqslant Ce^{-ct}W_{\mathbf{d}_{N}}\left(\nu^{N},\mu^{N}
ight)$$

Then for all  $k \in [\![1, N]\!]$ , we have:

$$\|\left(\nu^N P_t^N\right)^k - \mu^{k,N}\|_{TV} \leqslant \frac{k}{N} C e^{-ct} W_{\mathbf{d}_N}\left(\nu^N, \mu^N\right).$$

*Proof.* A mean field particle system at time  $t \ge 0$  is exchangeable as soon as its initial condition is. Additionally, if  $\nu_1$  and  $\nu_2$  are two exchangeable probability measure on  $E^N$ , by denoting  $\nu_1^k$  $\nu_2^k$  their k particles marginal, we have:

$$\|\nu_1^k - \nu_2^k\|_{TV} \leqslant \frac{k}{N} W_{\mathbf{d}_N} \left(\nu_1, \nu_2\right).$$

Such distances were employed in [38] for a Fleming-Viot process and in [133] for general processes with mean-field jump interaction. A discrete measure cannot converge in total variation distance towards a measure with density, making the total variation distance unsuitable for the mean field limit, where one wants to show the convergence of the empirical measure. To achieve long-time convergence and uniform in-time propagation of chaos, a distance close to the  $W_1$
distance in  $\mathbb{R}^d$  must be used. For instance, the distance:

$$\mathbf{d}_N(\mathbf{x}, \mathbf{y}) = \frac{1}{N} \sum_{i=1}^N |x_i - y_i|,$$

was used in [76, 118].

# 1.3 Stochastic algorithms and contributions of this thesis

Now, let's delve into the presentation of some MCMC-type algorithms. The theoretical investigation of these algorithms is rooted in the analysis of the long-time behavior of certain Markov processes, as discussed in Section 1.2. We will explore three types of algorithms, corresponding to the three parts of the thesis: the sampling of Gibbs measures, stochastic optimization, and the sampling of quasi-stationary measures using Fleming-Viot particle systems.

#### 1.3.1 Sampling Gibbs measures

The first type of algorithm involves sampling Gibbs measures. Many physical systems can be described by a Hamiltonian (or energy) of the form:

$$H(x,y) = U(x) + \frac{1}{2}|y|^2, \qquad (1.31)$$

where  $U : \mathbb{R}^d \to \mathbb{R}_+$  (or  $U : \mathbb{T}^d \to \mathbb{R}_+$ , where  $\mathbb{T}^d$  is the *d*-dimensional torus) is a potential function, and  $|y|^2/2$  represents the kinetic energy. When the dimension *d* is large, obtaining the exact evolution of the system becomes infeasible, and a statistical description must be employed. In statistical physics, for systems with fixed temperature  $T = \beta^{-1} > 0$  and size, but that can exchange energy, the system is characterized by the Gibbs measure (1.12). The probability for the system to be in a state  $(x, y) \in \mathbb{R}^{2d}$  solely depends on its energy and we recall that it is given by

$$\mu_{\beta}(\mathrm{d}x,\mathrm{d}y) = \frac{e^{-\beta H(x,y)}}{\sqrt{\pi\beta^{-1}}^{d} Z_{\beta}} \mathrm{d}x \mathrm{d}y$$

One goal of molecular dynamic is to compute macroscopic quantities of physical systems defined by averages of the form

$$\mu_{\beta}(f) := \mathbb{E}(f(X, Y)), \quad (X, Y) \sim \mu_{\beta}, \tag{1.32}$$

for some function  $f : \mathbb{R}^d \to \mathbb{R}$ . One important example in chemistry is a system of  $N \gg 1$ particles with van der Waals interaction, which corresponds to a noble gas. In this case, d = 3Nis of the order of  $\mathcal{N}_A \approx 10^{23}$ , and  $x = (x_1, \ldots, x_N)$  represents the positions of the particles. For a more in-depth introduction to molecular dynamics, refer to [161], as well as [111] for a mathematical perspective. Sampling high-dimensional Gibbs measures is not exclusive to statistical physics; it is also of paramount importance in computational statistics and machine learning. For instance, suppose given independent and identically distributed samples  $(X_i, Y_i)_i$ from an unknown probability measure  $\mu^*$  on  $\mathbb{R}^d \times \mathbb{R}$ . We are interested in the best predictor  $f^*$ of Y knowing X:

$$f^* \in \arg\min \mathbb{E}\left(\left(Y_1 - f(X_1)\right)^2\right)$$

The solution to this problem is given by  $f^*(x) = \mathbb{E}(Y|X = x)$ , which is not accessible in practice. If we assume that  $f^*$  can be efficiently approximated by a linear functional, then given a matrix  $\theta$ , and writing  $U(\theta) = \mathbb{E}(||Y - \theta X||^2)$ , a random variable whose law is the Gibbs measure

$$\nu_{\beta}(\mathrm{d}\theta) = \frac{e^{-\beta U(\theta)}}{Z_{\beta}}\mathrm{d}\theta,$$

or its expectancy, are estimators of the solution, see [114, 98]. In this case, and many others, there is no kinetic energy, and we are only interested in the space variable marginal  $\nu_{\beta}$ , also referred to as Gibbs measure, of the previously defined Gibbs measure  $\mu_{\beta}$ .

The tasks involved in sampling such probability measures have three aspects:

- 1. Designing a Markov Process: The first task is to design a Markov process X that has the targeted probability measure as its unique invariant state and is ergodic. The rate of convergence of this Markov process should be as high as possible, and especially it should not approach 0 too rapidly as the dimension d goes to infinity.
- 2. Choosing a Discretization Scheme: A discretization scheme  $X^{\delta}$  for the Markov process needs to be chosen so that it can be implemented on a computer. Some processes are easier to discretize than others, but in most cases, this introduces bias. This bias must be considered in both the choice of the process and its numerical scheme.
- 3. Monte-Carlo Method for Approximation: Finally, a Monte-Carlo method is needed to approximate the expectations of the numerical scheme. If the target is the invariant measure  $\mu$  and the numerical scheme admits an invariant measure  $\mu^{\delta}$ , then the ergodic property

$$\lim_{T \to \infty} \frac{1}{T} \sum_{i=1}^{T} f\left(X_i^{\delta}\right) = \mu^{\delta}(f), \text{ almost surely},$$

can be used. The error is then given by the bias of the invariant measure, and by the fact that  $T \in \mathbb{N}$  is finite. Another solution, that allows also the computation of  $\mathbb{E}(f(X_t))$  for finite  $t \ge 0$ , is to compute a large number  $M \in \mathbb{N}$  of independent realisations of  $X_{\lfloor t/\delta \rfloor}^{\delta}$ , say  $X_{\lfloor t/\delta \rfloor}^{\delta,i}$ , and use that fact that under some integrability assumption:

$$\lim_{M \to \infty} \frac{1}{M} \sum_{i=1}^{M} f\left(X_{\lfloor t/\delta \rfloor}^{\delta,i}\right) = \mathbb{E}(f(X_t)), \text{ almost surely.}$$

This second Monte Carlo introduces three types of errors. The first error is finite time error, arising from the fact that we can only access  $\mathbb{E}(f(X_t))$  for finite  $t \ge 0$ . The second error is the Monte-Carlo error, stemming from the approximation of  $\mathbb{E}(f(X^{\delta}))$  (namely M is finite), and the third error results from the bias in the numerical scheme, where  $\mathbb{E}(f(X^{\delta}_t)) \neq \mathbb{E}(f(X_t))$ . The total error can typically be expressed in the form::

$$\mathbb{E}\left(\left|\frac{1}{M}\sum_{i=1}^{M}f(X_{i}^{\delta})-\mu_{\beta}(f)\right|\right) \leqslant C\left(e^{-ct}+\frac{1}{\sqrt{M}}+\delta^{\alpha}\right).$$

The Monte Carlo error could potentially be improved by using other estimators of  $\mathbb{E}(f(X_t))$  from the numerical scheme, such as Multilevel Monte Carlo methods (refer to [68]). However, the focus in the first two chapters of this thesis will be on the first and third errors.

There are many choice for the process used in the sampling, as we will see now. The most

commonly used process for sampling from  $\nu_{\beta}$  is the overdamped Langevin process:

$$\mathrm{d}X_t = -\nabla U(X_t)\mathrm{d}t + \sqrt{2\beta^{-1}}\mathrm{d}B_t,\tag{1.33}$$

where B is a Brownian motion, thus making X a diffusion process, see [110, 112] for its use in stochastic algorithms.

Assuming mild conditions on U, such as  $U \in \mathcal{C}^{\infty}$ ,  $\nabla U$  Lipschitz continuous, and satisfying

$$(\nabla U(x) - \nabla U(y)) \cdot (x - y) \leqslant -c|x - y|^2$$

for all  $x, y \in \mathbb{R}^d \setminus K$ , where K is some compact set and c > 0, it can be shown, using the results from Section 1.2, that this process is ergodic with respect to  $\nu_{\beta}$ . The specific type of convergence depends on the assumptions about U and the methods used.

The Euler-Maruyama scheme is a well-known discretization method for Langevin-type processes. It reads here:

$$X_{n+1}^{\delta} = X_n^{\delta} - \delta \nabla U(X_n^{\delta}) + \sqrt{2\beta^{-1}\delta}G_n$$

where  $(G_n)_n = (B_{(n+1)\delta} - B_{n\delta})_n$  a family of independent normal random variable. This discretization leads to the Unadjusted Langevin Algorithm (ULA). If  $\nabla U$  is Lipschitz continuous, it can be demonstrated that this numerical scheme converges strongly (in the sense of pathwise) and weakly with a convergence order of 1.

**Definition 1.3.1.** A numerical scheme is said to converges strongly with rate of convergence  $\alpha$  if for all T > 0, there exists C > 0 such that for all  $\delta$  small enough:

$$\sup_{0 \leqslant n \leqslant T/\delta} \mathbb{E}\left( |X_n^{\delta} - X_{n\delta}| \right) \leqslant C\delta^{\alpha}.$$

A numerical scheme is said to converges weakly with rate of convergence  $\alpha$  if for all  $t \ge 0$ , and f in a suitable class of function, there exists C > 0 such that:

$$\left|\mathbb{E}\left(f(X_{n}^{\delta})\right) - \mathbb{E}\left(f(X_{t})\right)\right| \leqslant C\delta^{\alpha},$$

where  $n\delta = t$ .

In general, the weak and strong order of convergence are not the same, as in the case of multiplicative noise where the strong order is 1/2. However, for the purposes of this thesis, only weak convergence is considered since the goal is the computation of averages. Proofs for weak convergence often rely on estimates of the semi-group, such as:

$$|\partial^{\alpha} P_t f(x)| \leqslant C(1+|x|^k), \tag{1.34}$$

for regular enough f and  $|\alpha|_1 < \alpha_{max}$ , along with moment bounds on the numerical scheme, see [156] for instance in the case of the ULA. It can get intricate as soon as the drift is not lipschitz continuous as in [123] where it is shown that the explicit Euler scheme might not be ergodic for non-globally Lipchitz vector field. See also [88] for the study of the failure of the Euler-Maruyama scheme. In such cases, alternative schemes need to be employed, as discussed in [78, 87, 89, 90, 30]. Discretization introduces bias in the invariant measure, which can be mitigated using a general method called metropolization. For example, in the case of the Euler scheme for the over-damped Langevin dynamics, the Metropolis Adjusted Langevin Algorithm (MALA) is:

• 
$$\bar{X}_{n+1}^{\delta} = X_n^{\delta} - \delta \nabla U(X_n^{\delta}) + \sqrt{2\beta^{-1}\delta}G_n$$

- Let U be uniform on [0, 1].
- If  $U \leq \min(1, e^{\beta(U(X_n^{\delta}) U(\bar{X}_{n+1}^{\delta}))})$ , then  $X_{n+1}^{\delta} = \bar{X}_{n+1}^{\delta}$ , else  $X_{n+1}^{\delta} = X_n^{\delta}$ .

This is the so-called Metropolis Adjusted Langevin Algorithm (MALA), and it is known to admit  $\nu_{\beta}$  as its invariant measure. Other schemes, such as the Milstein scheme [145, Chapter 10], can be employed to achieve better orders of convergence. Implicit numerical schemes are another option, but they may not be well-suited for high-dimensional settings. The well-posedness of an implicit numerical scheme is not guaranteed, and even if it were, it would likely be numerically costly in practice.

A general class of processes are kinetic processes, which are characterized by the equations

$$\mathrm{d}X_t = Y_t \mathrm{d}t,$$

where X is the position, and noise only acts on the speed Y. Kinetic processes can be utilized for sampling from the measure  $\nu_{\beta}$  using only its position variable. They are expected to exhibit better convergence rates than elliptic diffusions, displaying a ballistic behavior rather than the diffusive behavior of reversible processes.

To establish estimates on the semi-groups, similar to (1.34), necessary for proving weak convergence, hypocoercive methods, as discussed in Section 1.2.1, must be employed. One of the most physically relevant kinetic processes is the underdamped Langevin process, which models the movement of a particle in a potential U while in contact with a heat bath at temperature  $T = \beta^{-1} > 0$ . We recall that it solves:

$$\begin{cases} \mathrm{d}X_t = Y_t \mathrm{d}t, \\ \mathrm{d}Y_t = -\nabla U(X_t) \mathrm{d}t - \gamma Y_t \mathrm{d}t + \sqrt{2\gamma\beta^{-1}} \mathrm{d}B_t. \end{cases}$$

Under mild assumptions, this process is ergodic with respect to  $\mu_{\beta}$ , allowing for the sampling of the full Gibbs measure. A straightforward Euler-Maruyama scheme can be used for this process, but explicit splitting schemes are commonly employed in Molecular Dynamics simulations, better suited for Hamiltonian dynamics, see [107].

Another family of kinetic processes includes kinetic Piecewise Deterministic Markov Processes. PDMPs have the advantage that they can be simulated exactly, eliminating bias. In general, such processes admit  $\nu_{\beta} \otimes \pi$  as steady states, where  $\pi$  is a probability measure on the state space of the speed variable, see [138, 52]. An example of a kinetic PDMP is the Bouncy-Particle sampler, with a generator given by

$$Lf(x,y) = y \cdot \nabla_x f + (v \cdot \nabla U(x))_+ (f(x, R(x,y)) - f(x,y)) + \lambda \int_{\mathbb{R}^d} (f(x,y') - f(x,y)) \frac{e^{-v'^2/2}}{(2\pi)^{d/2}} \mathrm{d}v',$$

where  $\lambda > 0$  is a refreshment parameter and  $R(x, y) = y - 2\nabla U(x) \cdot v\nabla U(x)/|\nabla U(x)|^2$ . The first jump part imposes the Gibbs measure  $\mu_{\beta}$  as an invariant measure, and the refreshment is introduced for ergodicity. To study the long time behavior of PDMP's, the method developed in [163] is not adapted, as the *carré du champ* is not of gradient type. However other method can be applied, see [50, 1] for hypocoercivity, as well as coupling method described in Section 1.2.2.

There are many other processes used for sampling from a Gibbs measure. For instance, one can mention the Hamiltonian Monte Carlo (HMC) method [69, 26]. In this method, a trajectory

 $(X_t, Y_t)$  evolves using the Hamiltonian dynamics

$$\begin{cases} \mathrm{d}X_t = Y_t \mathrm{d}t, \\ \mathrm{d}Y_t = -\nabla U(X_t) \mathrm{d}t \end{cases}$$

and every time T, Y is re-sampled independently from the past according to a  $\mathcal{N}(0, \beta I_d)$  distribution. See also [27] for a review of various usable methods.

**Contributions** In Part I of this thesis, the focus will be on the weak error expansion of numerical schemes, also known as weak backward error analysis, for kinetic processes. This involves searching for series expansions in weak convergences, denoted as

$$\mathbb{E}_z(f(X_n^{\delta})) = \mathbb{E}_z(f(X_t)) + C_1(t)\delta^{\alpha} + \dots + C_k(t)\delta^{\alpha+k} + O(\delta^{\alpha+k+1}),$$

for all  $k \in \mathbb{N}$ , where  $n\delta = t$ . More particularly, we are interested in cases where the constants  $C_p$ , as well as the rest, are bounded in time, leading to uniform weak convergence over time. This expansion then also applies to the bias of the invariant measure. The analysis involves estimates such as

$$\partial^{\alpha} P_t f(x, y) \leqslant C e^{-ct} (1 + |x|^k + |y|^k),$$

for function satisfying  $\mu_{\beta}(f) = 0$ . The weak error expansion of numerical schemes was initially proven and numerically studied by Talay and Tubaro in [158]. Their work primarily focuses on the Euler-Maruyama and Milstein schemes for Elliptic SDEs with globally Lipschitz continuous coefficients. Talay also proved similar expansions for Hamiltonian systems with polynomial growth at infinity in [157], although using an implicit Euler scheme. Further studies on these expansions can be found in [41, 101, 100]. For explicit numerical schemes for the Langevin process with bounded potentials on the torus, refer to [107], and for Langevin processes with more general kinetic energies, see [153]. Uniform convergence in time of a numerical scheme is also explored in [3].

Chapter 2 is devoted to the sampling of Gibbs measure using the Langevin process in the case of a singular potential, as in the case of N particles within a confining potential and subject to Lennard-Jones interactions. Explicit numerical schemes are known to exhibit instability when the drift of the stochastic differential equation lacks Lipschitz continuity [123]. To address the issue of singularity, a novel numerical scheme that avoids instability phenomena and can be applied even in high-dimensional settings is proposed. Employing hypocoercive method adapted to the singular case as in [10], estimates on the semi-group of (1.10) are proven and used to show uniform in time weak error expansion of this new numerical scheme, as well as at stationary.

The most expensive part when simulating the process (1.10) is the computation of  $\nabla U$ . A common approach is to substitute this gradient with a stochastic version. The goal of Chapter 3 is to explore an alternative solution. The idea is to break down the potential as  $\nabla U = \sum_{i=0}^{K} F_i$ , where  $F_0$  represents the cheap to compute part (corresponding to the short range interaction in a gas of particles). We then consider the process:

$$\begin{cases} \mathrm{d}X_t = Y_t \mathrm{d}t, \\ \mathrm{d}Y_t = -F_0(X_t)\mathrm{d}t - \gamma Y_t \mathrm{d}t + \sqrt{2\gamma\beta^{-1}}\mathrm{d}B_t, \end{cases}$$

and the process jumps at rate  $\lambda_i$  according to some jump kernel  $q_i$  depending on  $F_i$  in a way that ensures that the equilibrium measure of the process is indeed the Gibbs measure. This a a mix between diffusion and piecewise deterministic Markov process, with the latter being a type of Markov process known for its advantageous simulation properties. This study also draws upon hypocoercive methods. <u>Perspectives</u>: Firstly, another process employed for sampling purposes in molecular dynamics is the generalized Langevin process [168]. It is a hypocoercive process exhibiting a higher degree of degeneracy than kinetic processes. It would be valuable to extend the computation in Sobolev spaces of any order for such a process to establish the weak convergence of numerical schemes. Additionally, exploring out-of-equilibrium processes is intriguing. When the gradient  $\nabla U$  in the Langevin process (1.10) is replaced by a general force F, the invariant measure is not explicit and may not satisfy a Poincaré inequality. Despite this, it remains important to efficiently sample such measures. Lastly, in the context of sampling Gibbs measures, a criterion for efficiency is the variance of the Monte Carlo estimator. This aspect could be further studied through a central limit theorem for the occupation measure process of the two processes analyzed in Chapters 2 and 3.

#### 1.3.2 Optimisation

Part II of this thesis delves into a problem closely related to the sampling of Gibbs measure: the optimization of non-convex functions. The focus is on a function, called the objective function in optimization but referred to as the potential in the thesis,  $U : \mathbb{R}^d \to \mathbb{R}$ , which is bounded from below. The goal is to address the problem of finding the minimum (or the minima) of U over  $\mathbb{R}^d$ . In various applications, non-convex potentials with multiple local minima, not necessarily confined within a compact set, are of interest. For example, in the context of Molecular Dynamics, U may represent the energy of a configuration of a physical system, and the minima correspond to its metastable states. In computational statistics, U could represent minus the logarithm of the posterior probability, and its minima indicate the maximum likelihood. A well-known benchmark problem for non-convex optimization discussed in this thesis is the mixture of Gaussian distributions. If  $(X_i)$  is a data set consisting of Gaussian variable with means  $\Sigma_i \theta$ , then the maximal likelihood is given by the minimum of a potential that takes the form

$$U(\theta) = \ln\left(\sum_{i=1}^{n} a_i \exp\left(-\lambda_i \left(\theta - \Sigma_i^{-1} X_i\right)^2\right)\right).$$
(1.35)

When dealing with a convex U, standard gradient descent or its many variants are often sufficient for finding the minimum. However, the challenge arises in the case of non-convex potentials, as typical gradient descent algorithms tend to converge to local minima. One approach would be to initiate multiple gradient descents from random initial points, but the basin of attraction for global minima is usually of lower dimension compared to the ambient space. As a result, the probability of starting in this set is quite low. Additionally, in high-dimensional settings, an exhaustive search is impossible. To overcome these challenges, new paradigms that takes into account the geometry of U are needed to explore the space. The goal is to devise algorithms that impose minimal assumptions on U, typically requiring

$$U \in \mathcal{C}^1(\mathbb{R}^d), \quad \lim_{\|x\| \to \infty} U(x) = +\infty.$$

However, these assumptions may not be sufficient for the proposed algorithms.

There are various methods for non-convex optimization, including the utilization of meanfield particle systems like (1.30) for optimizing neural networks [86, 67] or genetic algorithms based on Fleming-Viot-type particle systems, as discussed in Section 1.3.3, see [148]. However, in this thesis, we will focus on two specific solutions for this optimization problem. The first one is provided by simulated annealing, based on the observation that, under mild assumptions,

$$\nu_{\beta}\left(U>\min_{\mathbb{R}^d}U+\varepsilon\right)\underset{\beta\to\infty}{\to}0,$$

for all  $\varepsilon > 0$ , where  $\nu_{\beta}$  is the Gibbs measure given in (1.9). This implies that if X is an ergodic Markov process with invariant measure  $\nu_{\beta}$ , and if  $\beta$  is large enough, X will spend most of its times in the vicinity of the global minima. The simulated annealing algorithm involves gradually increasing  $\beta$  over time. Given a generator  $L_{\beta}$  which admits  $\nu_{\beta}$  as its invariant measure for all  $t \ge 0$ , and a cooling schedule  $\beta_t \xrightarrow[t \to \infty]{} \infty$ , the goal is to design an inhomogeneous Markov process  $X_t$  with generator  $L_{\beta_t}$ . If  $\beta$  increases slowly enough, the law of  $X_t$  will be close to its instantaneous equilibrium  $\nu_{\beta_t}$ , and X will converge in probability towards the set of global minima of U. The faster the temperature goes to 0, the more the Gibbs measure will charge the minimum. However, if  $\beta$  increases too fast, the process might become trapped in the vicinity of a local minimum with positive probability, leading to algorithm failure. This transition is known to occur with a logarithmic schedule: if  $\beta_t = \ln(1+t)/c$  and the process is based on the overdamped Langevin process:

$$\mathrm{d}X_t = -\nabla U(X_t)\mathrm{d}t + \sqrt{2\beta_t^{-1}}\mathrm{d}B_t, \qquad (1.36)$$

then under mild assumption on U (see below), convergence holds for  $c > c^*$  and fails for  $c < c^*$ , where  $c^*$  is an explicit constant known as the largest energy barrier of the potential

$$c^* = \sup_{x_1, x_2} c(x_1, x_2)$$

with

$$c(x_1, x_2) = \inf \left\{ \max_{0 \le t \le 1} U(\xi(t)) - U(x_1) - U(x_2) \right\}$$

where the infimum runs over  $\{\xi \in \mathcal{C}([0,1], \mathbb{R}^d), \xi(0) = x_1, \xi(1) = x_2\}.$ 

The phase transition is attributed to metastability, a crucial phenomenon in the study of lowtemperature processes, among other things. Let's revisit the process with fixed temperature (1.8), and consider a domain  $D \subset \mathbb{R}^d$  corresponding to the basin of attraction for the gradient descent from a point  $x_0 \in \mathbb{R}^d$ . We are concerned with the first exit time from the domain:

$$\tau = \inf \left\{ t \ge 0, X_t \notin D \right\}$$

This problem was investigated in [65], revealing that  $\tau$  is of the order  $e^{\beta(\min_{\partial D} U - \min_{D} U)}$ . When the temperature  $\beta^{-1}$  is zero, the process behaves as a simple gradient descent, and it never leaves D. However, over an extended period, the introduced noise causes the process to move away from the vicinity of local minima. The speed of convergence of the law of  $X_t$  towards its equilibrium  $\mu_{\beta}$  is related to the pace at which it visits space. The longer it takes to exit the basin of attraction of local minima, the slower the convergence becomes. This can be quantified using PDE and functional inequality as described in Section 1.2.1. Since  $c^*$  is the minimal energy elevation required to transition from any local minimum (or even any point), to a global minimum,  $e^{\beta c^*}$  represents the order of the time needed to explore space. The Poincaré inequality, introduced in Definition 1.2.3, is one way to quantify this phenomenon.

**Proposition 1.3.1** ([83, 126]). Under the assumption that either  $U : \mathbb{T}^d \to \mathbb{R}_+$  or  $U : \mathbb{R}^d \to \mathbb{R}_+$ 

and

$$\liminf_{|x|\to\infty}|\nabla U(x)|>0,\quad \inf_{\mathbb{R}^d}|\nabla U|^2-\Delta U>-\infty,$$

 $\nu_{\beta}$  satisfies a Poincaré inequality

$$\lambda(\beta) \int_{\mathbb{R}^d} f^2 d\nu_\beta \leqslant \int_{\mathbb{R}^d} |\nabla f|^2 d\nu_\beta,$$

for all  $f \in C^1$ ,  $\int f d\nu_\beta = 0$ , and  $\lambda(\beta)$  satisfies

$$\lim_{\beta \to \infty} \frac{1}{\beta} \ln(\lambda(\beta)) = -c^*.$$

Combined with Proposition 1.2.3, this yields that the speed of convergence is of order  $e^{\beta c^*}$ , which is exponential bad if  $c^* > 0$ . This situation occurs when U possesses multiple local minima. In the context of simulated annealing, understanding metastability and exit times is crucial for selecting appropriate cooling schedules that balance exploration and exploitation. The objective is to avoid becoming trapped in local minima while still converging to the global minimum as the temperature decreases. This factor will be decisive in the convergence of simulated annealing (1.36). The proof idea is usually the same. In the case of U defined on the periodic torus  $\mathbb{T}^d$ , it proceeds as follows. Let  $h_t$  be the density of  $X_t$  with respect to  $\nu_{\beta_t}$ . Assuming that  $h_t \in L^2(\nu_{\beta_t})$ , we have:

$$\mathbb{P}\left(U(X_t) > \min U + \varepsilon\right) \leqslant \|h_t\|_{L^2(\nu_{\beta_t})} \left(\nu_{\beta_t}(U > \min U + \varepsilon)\right)^{1/2}.$$
(1.37)

If we can show that  $t \mapsto \|h_t\|_{L^2(\nu_{\beta_t})}$  is bounded, then we would obtain the desired result. Formally we have:

$$\begin{aligned} \partial_t \|h_t - 1\|_{L^2(\nu_{\beta_t})}^2 &= -\beta_t \int_{\mathbb{T}^d} |\nabla h_t|^2 \mathrm{d}\nu_{\beta_t} + \beta_t' \int_{\mathbb{R}^d} (h_t - 1)^2 U \mathrm{d}\nu_{\beta_t} \\ &\leqslant -\beta_t \lambda(\beta_t) \int_{\mathbb{T}^d} (h_t - 1)^2 \mathrm{d}\nu_{\beta_t} + \beta_t' \|U\|_{\infty} \int_{\mathbb{T}^d} (h_t - 1)^2 \mathrm{d}\nu_{\beta_t}. \end{aligned}$$

In this simple case where U is bounded (because continuous on a compact set), this computation combined with the estimate on  $\lambda$  from Proposition 1.3.1 indicates that if  $\beta_t = \frac{\ln(1+t)}{c}$ , then  $\|h_t\|_{L^2(\nu_{\beta_*})}$  remains bounded as soon as  $c > c^*$ . To get the correct rate of convergence, one must actually replace the  $L^2$  norm of the density in (1.37) by an  $L^{\infty}$  norm and use the Sobolev inequality. Proof of the convergence of the overdamped Langevin simulated annealing for slow logarithmic cooling schedule and of the non-convergence for fast logarithmic cooling schedule was first established by Holley, Kusuoaka and Strook [84, 83], using Sobolev inequalities, for a potential U on a compact manifold. The case of  $\mathbb{R}^n$  has been studied by Chiang, Hwang and Sheu [37], Royer [150] and Miclo [129], under restrictive conditions on the behavior at infinity of U, in particular  $|\nabla U| \to +\infty$  at infinity. These conditions are related to the functional inequalities used in these works, in particular spectral gap and Nelson hypercontractivity inequalities. The question of reducing these assumptions in order to consider slowly-growing potentials has been addressed by Zitt in [171], essentially by replacing spectral gap inequalities by weaker functional inequalities. The results of Zitt apply for instance if, outside some ball,  $U(x) = |x|^{\alpha}$  with  $\alpha \in (0,1)$ . More recently, Fournier and Tardif in [64] and with Monmarché in [63] have been interested in somehow minimal conditions on the growth of U at infinity. In [63], they established that, for coercive potentials in the sense that  $x \cdot \nabla U(x) \ge 0$  for |x| large enough, there is a phase transition for  $U_{\alpha}(x) = \alpha \ln(1 + \ln(1 + |x|^2))$  at some value  $\alpha_*$  of  $\alpha$  (depending on the cooling schedule  $\beta$  and the dimension), i.e. there is convergence if  $\alpha > \alpha_*$  and non-convergence if  $\alpha < \alpha_*$ , which is related to the transient properties of Bessel processes. More generally, convergence of the annealing algorithm is also proven in [63] under conditions that allow arbitrarily slow growth. In [64], the convergence of the simulated annealing is established as soon as  $\lim_{|x|\to\infty} U(x) = \infty$  and  $\int_{\mathbb{R}^d} e^{-\alpha_0 U(x)} dx < \infty$  for some  $\alpha_0 > 0$ . Although it doesn't cover all the cases of [63] (notice indeed that the condition is not met for  $U_{\alpha}$  for any  $\alpha > 0$ ), this is a very simple and mild condition. One of the main differences of [64, 63] with respect to previous works is that the question of the recurrence of the process is treated separately from the question of convergence in probability to the minimum of U. Indeed, once recurrence is proven, it is essentially sufficient to use the known results of convergence in the compact case to conclude. Notice that, unfortunately, this localization argument does not provide a rate of convergence as did previous works. Note as well that the idea that the behavior of U at infinity is not so important already appears in [37, 150] (see in particular [37, Lemma 6.4]) where it is proven that it is sufficient (under the conditions enforced in these works) to prove the result in the case where  $U(x) = |x|^4$  for |x| large enough.

On the other hand, Monmarché studied in [137] the simulated annealing based on the kinetic Langevin process:

$$\begin{cases} \mathrm{d}X_t = Y_t \mathrm{d}t \\ \mathrm{d}Y_t = -\nabla U(X_t) \mathrm{d}t - \beta_t Y_t \mathrm{d}t + \sqrt{2} \mathrm{d}B_t \end{cases}$$

The use of this non reversible process is motivated by its better convergence properties with respect to the overdamped process, as discussed in Section 1.3.1, although, in the regime  $\beta \rightarrow +\infty$ , it doesn't reduce the critical value of c in the logarithmic regime. Convergence of the kinetic simulated annealing in established in [137] for slow logarithmic cooling schedules similar to the overdamped case, under restrictive conditions on U, namely U is essentially quadratic at infinity, and the Hessian of U is bounded. The proof is similar to the overdamped Langevin case except that establishing quantitative longtime convergence estimates (at a fixed  $\beta$ ) for the process toward its equilibrium rely on so-called hypocoercive methods presented in 1.2.1. The arguments of [137] have been adapted to Generalized Langevin processes in [31]. This simulated algorithm can also be understood as a deterministic dynamics that converges towards local minima, to which we added a noise that goes to zero with time. The same can thus be done with process that does not admit the Gibbs measure as an equilibrium as in [23].

**Contribution:** Chapter 4 of this thesis is based on an article published in *Electronic Journal* of *Probability* [95]. The goal was to combine the works [137] and [64] to show convergence of the kinetic Langevin simulated annealing for potential under minimal conditions on the growth of U at infinity when  $c > c^*$ . We showed as well failure of the algorithm when  $c < c^*$  using hypocoercivity methods in Sobolev spaces  $H^k$ , which seems to be a novel result for kinetic processes. Indeed, unlike the elliptic case, it remains an open problem to show that Sobolev inequality implies ultracontractivity, which was the main tool in the work [84, 83].

The previously mentioned Freidlin-Wentzell theory [65] also provides insights into the behavior of low-temperature diffusion processes, suggesting that such processes move from one potential well to another by passing through index-1 saddle points. One strategy for escaping the domain of attraction (for gradient descent) of a local minimum would hence be to actively search for saddle points. Building on this understanding, Chapter 5, which is based on an article published in *Statistics and computing* [96], we introduced a novel process for both optimisation and saddle point search in [96]. We conducted a numerical study of this new process on basic examples from both statistics and Molecular Dynamics, such as the Gaussian mixture (1.35), as well as addressed fundamental theoretical questions.

Let's motivate further this last contribution. In Molecular Dynamic, index-1 saddle points

are also of interest, as if the local minima of the energy represent the metastable states, they represent the transition pathways between metastable states of a physical system. However, saddle point search remains a less standard topic than optimisation, especially in probability. A first class of methods, based on reaction paths, requires the knowledge of two local minimizers, and then finds a path of minimal elevation between them, which passes through a saddle point see e.g. [70, 53]. A second class of methods, considered e.g. in [54, 121, 66, 115, 151, 17, 75] and references within and which will be the one used in the definition of the process of Chapter 5, relies on local walkers, analogous of the gradient descent for optimization, i.e. solutions of some ODEs for which saddle points of U are stable equilibrium. Given some  $x \in \mathbb{R}^d$ , write:

$$\lambda_1(x) \leqslant \lambda_2(x) \leqslant \cdots \leqslant \lambda_d(x)$$

for the ordered eigenvalues of  $\nabla^2 U(x)$ , and suppose that we are given  $v_1(x), \ldots, v_d(x) \in \mathbb{R}^d$  such that  $v_i(x)$  is an eigenvector of  $\nabla^2 U(x)$  associated to  $\lambda_i(x)$  (in other words, when the eigenvalues are not all simple, we assume that we have an arbitrary rule to select a given eigenbasis, for instance we take the basis obtained as the limit of the Jacobi algorithm). The idealised saddle dynamic (ISD) is the solution of

$$\dot{x}_t = -(I - 2v_1(x_t)v_1(x_t)^T) \nabla U(x_t).$$
(1.38)

In other words,  $x_t$  follows a gradient descent, except in the direction  $v_1(x_t)$  where a reflection is performed, i.e. the process follows a gradient ascent in this direction. Notice that in general  $x \mapsto v_1(x)$  is not continuous so the existence of this process is unclear and may be restricted to some parts of the space or to a finite time interval. One could also see a saddle point as a minimum of the function  $x \mapsto |\nabla U(x)|^2$  or variants, which was done in [51, 19]. By introducing noise into the dynamic (1.38), we define a new process for locating index-1 saddle points. Subsequently, by alternately switching between this process and a noisy gradient descent (specifically, the overdamped Langevin process 1.8), we formulate a new process for optimization.

<u>Perspectives</u>: It would be of interest to show convergence (or non-convergence) for simulated annealing based on other non-reversible processes, such as annealed piecewise deterministic Markov process (e.g., the Zig-Zag process or Bouncy Particle Sampler), as they offer certain simulation advantages, or non-conservative overdamped Langevin process. This last example may prove useful in the context of this last switched process. Several open questions also remain for this switched process: how long does it take to exit from a well in the low temperature limit? What about the speed of convergence towards its invariant measure? What is the limit of the invariant measure when  $\varepsilon \to 0$ ? What would be the long-term behavior of an annealed version of this process?

#### **1.3.3** A particle approximation for quasi-stationary distributions

The third and final type of algorithm involves sampling the quasi-stationary distribution using the Fleming-Viot process. Let's first motivate this purpose. In Molecular Dynamic, it is of interest to sample state-to-state dynamics. Assuming that we can partition space into metastable subsets of a given Markov process X through a mapping  $S : \mathbb{R}^d \to \mathbb{N}$ , the goal is to efficiently compute a trajectory  $S(X_t)$ . However, by definition, exiting a metastable state takes a very long time. To accelerate such exit, various algorithms, collectively known as kinetic Monte Carlo (kMC) methods, have been developed. One such algorithm is the parallel replica algorithm [104], which relies on the sampling of a quasi-stationary distribution (QSD). In all generality, it is defined as follow:

**Definition 1.3.2.** Given a Markov process X living in a domain D and a stopping time  $\tau$ , a QSD is a probability measure  $\mu$  that satisfies:

$$\mathbb{P}_{\mu}\left(X_t \in \cdot | \tau > t\right) = \mu,$$

for all  $t \ge 0$ .

This definition is similar to that of the stationary measure, and like stationary measures, QSDs describe the long-term behavior of the process but are conditioned on the fact that  $\tau > t$ . In kMC, this stopping time corresponds to the exit time of a metastable domain D:

$$\tau_{\partial D} = \inf \left\{ t \ge 0, X_t \notin D \right\},\tag{1.39}$$

where D is an open set. In many cases, it can be shown that there exists a unique QSD  $\mu \in \mathcal{M}^1(D)$ , and

$$\forall \nu \in \mathcal{M}^1(D), \lim_{t \to \infty} \mathbb{P}_{\nu} \left( X_t \in \cdot | \tau > t \right) = \mu$$

In metastable cases, this convergence is faster than the exit time of the domain D, and hence, for some intermediate time scale, the law of X is described by the QSD. Therefore, QSDs are another valuable tool for the study of metastability, in addition to the study of constants in functional inequalities, as seen in Proposition 1.3.1. Moreover, it can be shown that:

**Proposition 1.3.2.** If  $\mu$  is a QSD, then there exists  $\alpha > 0$  such that:

$$\forall t \ge 0; \ \mathbb{P}_{\mu}(\tau > t) = e^{-\alpha t}$$

This arises from the fact that a QSD is an eigenvector of a certain operator, and  $\alpha$  is thus an eigenvalue. This property is actually an equivalence in many cases, see [104, Proposition 4]. The idea behind the parallel replica algorithm is to wait for some time for the law of the process to reach its QSD, and then replace the process with N independent processes  $(X^i)_{1 \leq i \leq N}$ , all starting from the QSD. Denote

$$\tau_1^N = \inf\left\{t \ge 0, \exists i \in [\![1,N]\!], X_t^i \notin D\right\},\$$

and let  $I_1^N$  be the index of the particle that exited the domain at time  $\tau_1^N$ . The property of the QSD implies that  $(\tau_1^N, X_{\tau_1^N}^{I_1^N})$  is equal in law to  $(\tau/N, X_{\tau})$ , where X is started from the QSD. This implies that we can sample an exit event exactly in a time N times smaller, and the error is simply introduced by approximating the law of X with the QSD.

However, the conditioning introduces a non-linearity in the evolution of  $\mathbb{P}(X_t \in \cdot | \tau > t)$ , which adds complexity to the sampling process and introduces new phenomena compared to the study of stationary distributions. For instance, a linear combination of QSDs is not necessarily a QSD, implying that there might be a finite but greater than one number of QSDs (which is not possible for stationary measures). Before delving into the sampling methods, let's explore killed processes further.

Stopping times, also called killing times, are usually of two types: the so-called soft-killing and hard-killing. Hard-killing corresponds to the exit of a domain, and is usually challenging to study. Given some domain  $D \subset E$ , it is given by (1.39). On the other hand, soft-killing is a more regular case, and given some death rate  $\lambda : E \to \mathbb{R}_+$  and an exponential random variable  $E_{\lambda}$  with parameter 1, we define the killing time as:

$$\tau_{\lambda} = \inf \left\{ t \ge 0, E_{\lambda} \leqslant \int_0^t \lambda(X_s) \mathrm{d}s \right\}.$$

Using  $\tau$  as a general killing time, a killed process defines a non-conservative semi-group by:

$$P_t f(x) = \mathbb{E} \left( f(X_t) \mathbb{1}_{\tau > t} \right).$$

Similar to the conservative Markov semi-group,  $P_t$  is solution to a PDE. If the process is killed outside a domain D, then  $P_t f$  is the solution to the PDE with Dirichlet condition:

$$\begin{cases} \partial_t v = Lv, \\ v(0, \cdot) = f, \\ v|_{\partial D} = 0, \end{cases}$$
(1.40)

for f in a suitable class of function, and where L is the generator of the underlying Markov process. In the case of soft-killing, the semi-group is the solution to:

$$\begin{cases} \partial_t v = Lv - \lambda v, \\ v(0, \cdot) = f. \end{cases}$$
(1.41)

In any case, the considerations on long-time behavior of Markov process can be extended to killed processes. The QSD is a left-eigenvector of this semi-group. In the case where the process is symmetric with respect to some reference measure  $\mu$ , the QSD admits a density with respect to this measure which is a right eigenvectors of the semi-group. Usual spectral gap is now defined as the difference between the first and second eigenvalues, which gives the rate of convergences towards the QSD, and the first eigenvalue is the killing rate from Proposition 1.3.2. The soft-killing case is linked with Feynman-Kac semi-group, as we have the following equality:

$$P_t f(x) = \mathbb{E}_x \left( f(X_t) e^{-\int_0^t \lambda(X_s) \mathrm{d}s} \right).$$

The sampling of QSD is thus of interest in simulating the ground states of Schrödinger's operator. For further references on the study of QSDs, see [35, 47, 36, 32, 7, 60, 8] and references within.

In a manner akin to McKean-Vlasov diffusions (1.30), a mean-field particle system  $\mathbf{X}$ , known as the Fleming-Viot process, can be used to sample the QSD. Informally, it operates as follows: each of the  $N \in \mathbb{N}$  particles evolves independently, following the same dynamics as X, until one of them is killed. The killed process then selects another particle uniformly at random, branches into it, and subsequently evolves independently of the others from this new position. More rigorously: fix some initial condition  $\mu \in \mathcal{M}^1(D^N)$ . Let  $(I_n^i)_{1 \leq i \leq N, n \in \mathbb{N}}$  be a family of independent random variables, where for  $1 \leq i \leq N$ ,  $I_n^i$  is uniform on  $\{1, \ldots, N\} \setminus \{i\}$ . Let  $\mathbf{X}_0 = (X_0^1, \ldots, X_0^N)$  be distributed according to  $\mu$ , and  $(\overline{X}^i)$  be a family of independent Markov processes following the same dynamic as X, with  $\overline{X}_0^i = X_0^i$ . Set

$$\tau_1 = \min \tau_1^i,$$

where  $\tau_1^i$  is the killing time of  $\bar{X}^i$ . Then, denote by  $i_1$  the index of the particle which died at time  $\tau_1$ . It is uniquely defined almost surely if, for example, the death time has a density on  $\mathbb{R}_+$ , so that the probability that two particles hit the boundary at the same time is zero (this is true in most cases of interest). For  $i \neq i_1$ ,  $0 \leq t \leq \tau_1$ , or  $i = i_1$  and  $1 \leq t < \tau_1$ , simply let:

$$X_t^i = \bar{X}_t^i$$
 and  $X_{\tau_1}^{i_1} = \bar{X}_{\tau_1}^{I_1^{i_1}}$ .

This defines the process between times 0 and  $\tau_1$ . The process is then defined on  $(\tau_1, \infty)$  by

induction: if the process is defined up to time  $\tau_{n-1}$ , we define it between time  $\tau_{n-1}$  and  $\tau_n$  in the same way, with  $\mathbf{X}_0$  replaced by  $\mathbf{X}_{\tau_{n-1}}$ ,  $i_1$  by  $i_n$  the index of the particle that hits  $\partial D$  at time  $\tau_n$ , and  $I_1^{i_0}$  by  $I_n^{i_n}$ . Thus,  $(\tau_n)_n$  is the sequence of branching times of the process.

Under mild assumption, the process  $\mathbf{X} = (X^1, \ldots, X^N)$  is well-defined and does not explode in finite time, in the sense that  $\sup_n \tau_n = \infty$  almost surely. This holds true immediately in the soft-killing case as long as  $\|\lambda\|_{\infty} < \infty$ . Refer to [166, Theorem 2.1] and [14, 15] for diffusions killed at the boundary of a domain.

If X is a Markov process with generator L, killed at rate  $\lambda$ , then the Fleming-Viot process is a Markov process with generator:

$$\mathcal{L}f = \sum_{i=1}^{N} \left( L_i f + \frac{\lambda(x_i)}{N-1} \sum_{j \neq i} \left( f(\mathbf{x}^{i,j}) - f(\mathbf{x}) \right) \right),$$

for  $f: E^N \to \mathbb{R} \in \mathcal{D}(L)$ , where  $L_i f = L(x_i \mapsto f(x_1, ..., x_N))$  and  $\mathbf{x}_k^{i,j} = \mathbf{x}_k$  if  $k \neq j$  and  $\mathbf{x}_j^{i,j} = \mathbf{x}_i$ . A martingale formulation for a hard killed process is more challenging to define and can be found in [71, Proposition 1].

Fleming-Viot processes were first introduced in the work of Moran [139], in the study of population genetics models. Their application to the approximation of a QSD dates back to [25], where the authors study the case of a Brownian motion in a rectangle. They were interested in the following convergence, commonly known as propagation of chaos, which can be shown in many cases:

$$\lim_{N \to \infty} \frac{1}{N} \sum_{i=1}^{N} f(X_t^i) = \mathbb{P}\left(X_t \in \cdot | \tau > t\right), \qquad (1.42)$$

for f in a suitable class of functions, under certain assumptions on the initial conditions, thus affirming that the process may be used for sampling purposes in the case of conditioned process. Since then, many results have been proven in different cases and for various questions, including long-time convergence, propagation of chaos, and existence of the Fleming-Viot process.

In the case of a process in a countable state space, the study began with [59]. The Fleming-Viot process is well defined here as soon as the death rate is bounded, and the authors have shown, under several conditions, the uniqueness of the QSD, convergence toward this QSD, the ergodicity of the Fleming-Viot process, and the propagation of chaos for finite time and at equilibrium. In [73], the authors improved the propagation of chaos with a quantitative rate, introducing the  $\pi$ -return process. For processes in a finite state space, the propagation of chaos is proven for all times and for the stationary measure, with a stronger convergence in [5], and the convergence as  $N \to \infty$  was refined with a central limit theorem in [108].

In the case of processes in a general space, numerous results are available. Finite-time propagation of chaos is addressed in [24, 71, 165, 45], with central limit theorems as  $N \to \infty$  in [29, 48]. Then, uniform-in-time propagation of chaos and long-time convergence are established in [46, 44, 149]. The long-time convergence is established when the underlying process is a Brownian motion in [25].

Other methods of approximating a QSD have been developed in discrete and continuous cases in [12, 13, 11], based on self-interacting processes

In cases where multiple QSDs exist, one of them is defined as the minimal and corresponds to the greatest associated eigenvalue. The associated Fleming-Viot process is usually still an irreducible Markov process and hence possesses a unique invariant measure. It is expected that this invariant measure converges toward the minimal QSD, but this remains open in many cases. This problem has been addressed in [34, 72, 160]. **Contributions:** Part III of the thesis is dedicated to the study of the Fleming-Viot process. For practical purposes, when the objective is to compute averages against the QSD, we are interested in bounds of the form

$$\mathbf{d}(\pi_t,\mu) \leqslant C\left(\frac{1}{N^{\alpha}} + e^{-ct}\right),$$

for some distance **d** on the set of probability measure, where  $\pi$  is the empirical measure of the particle system, and  $\mu$  the target distribution. Such bounds allow for the separate choice of the number of particles and the time to achieve a specific level of error. To this end, Chapters 6 and 7 prove such bounds by establishing exponential ergodicity of the Fleming-Viot system at a rate independent of N in the sense that there exist C, c > 0 independent of N such that for all  $t \ge 0$ :

$$\|\nu P_t^N - \mu^N\|_{TV} \leqslant NCe^{-ct},\tag{1.43}$$

where  $P^N$  is the semi-group of the Fleming-Viot system with N particle, and  $\mu^N$  its invariant measure. The scaling in N in this convergence is what one would obtain in the case of a vector comprised of N independent processes.

Chapter 6, based on a paper published in *ESAIM: Probability and Statistics* [94], is concerned with the soft-killing case, where a numerical scheme for the system is also proposed. Chapter 7 studies the hard-killing regime for the overdamped Langevin process (1.8) in a metastable domain. Both works are based on the coupling method presented in Section 1.2.2. These works use a perturbation approach where the interaction is assumed to be small enough with respect to the mixing properties of the underlying Markov process, corresponding to the metastable regime, notably for kMC. On the other hand, Chapter 8 deals with a Fleming-Viot process based on a non-metastable system, namely for which killing occurs on a faster time scale than convergence to (quasi-)stationarity. In the simplest case of a countable state space, we show that in the limit where the minimum of the killing rate goes to infinity, instead of converging towards a deterministic measure, the empirical measure process converges towards a dynamic in the set of Dirac measures.

<u>Perspectives</u>: It would be of interest to construct a coupling that implies exponential ergodicity (1.43) outside of the case where mixing is faster than killing. This could be achieved by studying the variance of the particle system, as Chapter 8 suggests that in cases where the death rate is high, the empirical measure is close to a Dirac, indicating that the particles are close to one another.

Additionally, exploring whether exponential ergodicity, independent of N, remains true when the first eigenvalue is smaller than the spectral gap would be valuable. Studying scenarios where mixing might not occur in the diffusion process but could emerge through killing and re-sampling is also intriguing. This is relevant to situations resembling Schrödinger operators that arise in quantum physics, which are operators of the form  $\Delta + V$ , where V is a confining potential.

Another interesting question concerns the sampling of an exit event using the Fleming-Viot process. Starting from the QSD, the exit time of a domain follows an exponential distribution, independent of the event's location. The parallel replica algorithm relies on sampling such a random variable when the initial condition is the QSD. However, with the Fleming-Viot process, there is a bias in the sampling, as we can only access its stationary measure. Under this invariant measure, and with a finite number of particles, correlations exist between them, which are expected to go to 0 as N goes to infinity. This raises crucial questions for practitioners: if  $\tau^N$  is the first time a particle is killed in the Fleming-Viot process, what is the limiting behavior of  $(N\tau_1^N, X_{\tau_1^N})$ ? Given that the Fleming-Viot process enables the sampling of several killing events, does this result in a sequence of inter-event times  $N(\tau_k^N - \tau_{k-1}^N)$  that converges to a

Poisson process?

# Part I

# Sampling Gibbs measures

# Chapter 2

# Weak error expansion of a numerical scheme with rejection for singular Langevin process

#### Outline of the current chapter

<b>2.1</b>	Introduction	35
<b>2.2</b>	Mathematical setting and results	38
	2.2.1 Assumptions and mains results	38
	2.2.2 Comparison between numerical scheme with and without rejection $$ .	41
<b>2.3</b>	Proof of Theorem 2.2.1	<b>43</b>
	2.3.1 $H^1$ -setting	45
	2.3.2 $H^k$ -setting	49
	2.3.3 Proof of Theorem 2.2.1	53
<b>2.4</b>	Weak error expansion	<b>53</b>
	2.4.1 Finite-time expansion for the numerical scheme with rejection	54
	2.4.2 Expansion for the invariant measure of the numerical scheme with	
	$rejection \dots \dots$	62

**Abstract**: We show expansion à *la Talay-Tubaro* of a numerical scheme with rejection for the Langevin process in the case of a singular potential. In order to achieve this, we provide estimates on the associated semi-group of the process. The class of admissible potentials includes the Lennard-Jones interaction with confinement, which is an important potential in molecular dynamics and served as the primary motivation for this study.

# 2.1 Introduction

As explained in Section 1.3.1, we are interested here in the sampling of the Gibbs measure

$$\mu_{\beta}(\mathrm{d}x,\mathrm{d}y) = \frac{e^{-\beta H(x,y)}}{Z} \mathrm{d}x \mathrm{d}y, \qquad (2.1)$$

where

$$H(x,y) = U(x) + \frac{1}{2}|y|^2,$$
(2.2)

 $U: \mathbb{R}^d \to \mathbb{R}_+$  (or  $U: \mathbb{T}^d \to \mathbb{R}_+$ , where  $\mathbb{T}^d$  is the *d*-dimensional torus) is a potential function,  $|y|^2/2$  represents the kinetic energy, and Z is a normalization constant. One important example in chemistry is a system of  $N \gg 1$  particles with van der Waals interaction, which corresponds to a noble gas. In this case, d = 3N, and  $x = (x_1, \ldots, x_N)$  represents the positions of the particles. The potential can be written as

$$U(x) = \sum_{i=1}^{N} U_c(x_i) + \sum_{1 \le i \ne j \le N} U_i(|x_i - x_j|),$$

where  $U_c : \mathbb{R}^{3N} \to \mathbb{R}_+$  is a confining potential and for  $r \in \mathbb{R}^3$ ,

$$U_i(r) = 4\left(\frac{1}{|r|^{12}} - \frac{1}{|r|^6}\right).$$

The interaction part  $U_i$  is commonly known as the Lennard-Jones potential. In the case where the gas is confined in a volume D, one could then get its pressure by computing the average

$$P = \mu_{\beta}(f_P), \qquad f_P(x, y) = \frac{1}{3|D|} \sum_{i=1}^{N} \left( \frac{y_i^2}{m} - x_i \cdot \nabla_{x_i} U(x) \right),$$

where m > 0 is the mass of a particle. What makes this example interesting is the presence of singularities in the potential, as it can take infinite value, and the fact that the function  $f_P$  is not locally integrable.

For non-singular potentials, there are numerous numerical schemes available to simulated from the Langevin process

$$\begin{cases} dX_t = Y_t dt, \\ dY_t = -\nabla U(X_t) dt - \gamma Y_t dt + \sqrt{2\gamma\beta^{-1}} dB_t, \end{cases}$$
(2.3)

The goal here is to adapt such a scheme and prove a series expansion in the weak convergence for singular potential, focusing on the following one:

$$\begin{cases} \bar{X}_{n+1} = \bar{X}_n + \mathbb{1}_{E_{\delta}(\bar{X}_n, \bar{Y}_n, G_n) \in \mathcal{H}_{\delta,\ell}} \delta \bar{Y}_{n+1}, \\ \bar{Y}_{n+1} = \bar{Y}_n + \mathbb{1}_{E_{\delta}(\bar{X}_n, \bar{Y}_n, G_n) \in \mathcal{H}_{\delta,\ell}} \left( -\delta \nabla U(\bar{X}_n) - \delta \gamma \bar{Y}_n + \sqrt{2\gamma \beta^{-1} \delta} G_n \right) \end{cases}$$
(2.4)

for some time step  $\delta > 0$ , where  $(G_n)$  is a family of independent standard Gaussian random variable, and where for a small fixed parameter  $\ell > 0$ ,

$$\mathcal{H}_{\delta,\ell} = \left\{ \phi \leqslant \delta^{-\ell} \right\},\tag{2.5}$$

with

$$\phi(x,y) = H(x,y) + 4d\gamma\beta^{-1} \frac{y \cdot \nabla U(x)}{1 + |\nabla U(x)|^2},$$
(2.6)

and

$$E_{\delta}(x,y,g) = \left(x + \delta\left(y - \delta\nabla U(x) - \delta\gamma y + \sqrt{2\gamma\beta^{-1}\delta}g\right), y - \delta\nabla U(x) - \delta\gamma y + \sqrt{2\gamma\beta^{-1}\delta}g\right).$$

The definition of  $\phi$  stems from the fact that, under our assumptions, for all  $b < \beta$ ,  $e^{b\phi}$  is a Lyapunov function for the Langevin process (2.3) (see Proposition 2.3.3 below), as well as for this numerical scheme (as proved in Lemma 2.4.2). Hence, this numerical scheme is a version with rejection of the following (first-order) splitting scheme

$$\begin{cases} \tilde{X}_{n+1} = \tilde{X}_n + \delta \tilde{Y}_{n+1}, \\ \tilde{Y}_{n+1} = \tilde{Y}_n - \delta \nabla U(\tilde{X}_n) - \delta \gamma \tilde{Y}_n + \sqrt{2\gamma \beta^{-1} \delta} G_n, \end{cases}$$
(2.7)

but we only accept a step of the scheme if its Lyapunov function doesn't exceed some threshold depending on the time step. We could have considered a simple Euler-Maruyama scheme, but if the initial condition satisfies  $U(X_0 + \delta Y_0) = \infty$ , then the chain would not be well defined. The splitting scheme (2.7) is the simplest numerical scheme that is well-defined with probability 1 for all initial conditions, and under a one-sided Lipschitz condition on U, it can be shown that this Markov chain is ergodic. However, there is another problem, which is that even if the numerical scheme is almost-surely well defined,  $\tilde{X}_n$  has a density bounded below by the Lebesgue measure on all compact sets. Hence, if f is not locally integrable, then :

$$\mathbb{E}(f(\tilde{X}_n)) = \infty, \quad \forall n \ge 1.$$

This is also the case for all splitting schemes, and it poses two problems. Firstly, it restricts us to consider averages only against bounded functions, which excludes the computation of important quantities such as the mean energy for the Lennard-Jones potential or the pressure of noble gases. Secondly, it prevents the construction of a Lyapunov function, which is a critical step in the proof of weak convergence for numerical schemes. To address these issues, we introduced the numerical Scheme (2.4), which has better integrability properties, see Section 2.4. Any scheme used for bounded smooth potential (e.g. splitting schemes) can be adapted in a similar manner (namely with rejection) for singular potentials. We study this specific first-order splitting scheme for simplicity. In the context of singular potentials in high-dimensional spaces, an implicit scheme may be ill-defined, and in any case would have a prohibitive numerical cost. Hence, we consider only explicit schemes. Another possible numerical scheme would have been to use rejection at a threshold depending only on the energy, i.e. replacing  $\{\phi \leq \delta^{-\ell}\}$  by  $\{H \leq \delta^{-\ell}\}$ . However, since the Hamiltonian is not a Lyapunov function of the Langevin process, it is not clear that this numerical scheme admits  $e^{b\phi}$  as a Lyapunov function, and hence satisfies the integrability property needed for the proofs. Notice in particular that  $\nabla U$  is already computed for the drift, hence no supplementary computations are needed to check if  $\phi$  is low enough. While in practice, averages are often computed using schemes without rejection, there are cases where rejection could lead to better results. The discussion in Section 2.2.2 likely elaborates further on this aspect.

There are two approaches for showing the convergence of a numerical scheme towards the continuous process. The first one is strong convergence, which focuses on the convergence of trajectories. However, since our goal is the computation of averages, we are interested in weak convergence. For  $f : \mathbb{R}^d \to \mathbb{R}$  in a suitable class of functions and  $t \ge 0$ , we aim to show that:

$$\lim_{\delta \to 0} \mathbb{E}(f(\bar{X}_n)) = \mathbb{E}(f(X_t))$$

where  $n = \lfloor t/\delta \rfloor$ . Our objective is to get a series expansion of this convergence, commonly known as expansion à la Talay-Tubaro, from the seminal work [158]. Write  $(P_t)$  for the semi-group of the Langevin process (2.3) defined for regular and integrable enough function by:

$$P_t f(x, y) = \mathbb{E}_{(x, y)}(f(X_t, Y_t)).$$

Methods to show weak convergence and series expansion of such convergence rely on estimates on  $P_t f$  and its derivatives, typically of the form: for  $f : \mathbb{R}^d \to \mathbb{R}$ , which grows at most polynomially at infinity (as well as its derivative of all orders), for all multi-index  $\alpha \in \mathbb{N}^{2d}$ , there exists C, k > 0 such that:

$$|\partial^{\alpha} P_t f(x, y)| \leqslant C(1 + |x|^k + |y|^k).$$

Such estimates cannot hold in the singular setting. Therefore, the primary objective of this work is to establish equivalent estimates in the singular setting, which will be addressed in Section 2.3. Those estimates will then enable us to prove in Section 2.4 an expansion à la Talay-Tubaro for the numerical scheme (2.4), as well as for its invariant measure. In Section 2.2, we will present the assumptions, theorems, and comparisons with existing works.

## 2.2 Mathematical setting and results

#### 2.2.1 Assumptions and mains results

Fix a potential  $U: \mathbb{R}^d \to [0, \infty]$  and write

$$\mathcal{D} = \left\{ x \in \mathbb{R}^d, \quad U(x) < \infty \right\}, \quad \mathcal{X} = \mathcal{D} \times \mathbb{R}^d, \tag{2.8}$$

for the domain of definition of the process (X, Y) solution to equation (2.3). In all of this work, we will write  $z = (x, y) \in \mathcal{X}$  for the global variable, and  $|\cdot|$  for the  $l^2$ -norm on  $\mathbb{R}^d$ . For any Banach space  $\mathcal{B}$ , let  $\mathcal{C}^{\infty}(\mathcal{B}, \mathbb{R})$  denote the set of smooth functions from  $\mathcal{B}$  to  $\mathbb{R}$ ,  $\mathcal{C}^{\infty}_b(\mathcal{B}, \mathbb{R})$  the set of bounded smooth functions, and  $\mathcal{C}^{\infty}_c(\mathcal{B}, \mathbb{R})$  the set of smooth functions with compact support. Let  $\mathcal{M}^1(\mathbb{R}^{2d})$  denote the set of probability measures on  $\mathbb{R}^{2d}$ . We work under several sets of assumptions. First:

Assumption 2.2.1. •  $U \in \mathcal{C}^{\infty}(\mathcal{D}, \mathbb{R}_+)$ .

•  $\mathcal{D}$  is connected and the set:

$$\mathcal{D}_n = \left\{ x \in \mathbb{R}^d, \quad U(x) < n \right\}$$

is bounded for all  $n \in \mathbb{N}$ .

• U defines a Gibbs measure at temperature  $\beta$ :

$$\int_{\mathcal{D}} e^{-\beta U(x)} dx < \infty.$$

**Assumption 2.2.2.** • *The following limit holds:* 

$$\lim_{x \to \mathcal{D}^c} \frac{|\nabla^2 U(x)|}{|\nabla U(x)|^2} = 0.$$

• There exists  $c_0, c_{\infty}, d_0, d_{\infty} > 0$ ,  $\eta_0 \in \mathbb{R} \setminus [-1, 0]$ ,  $\eta_{\infty} > 1$  such that:

$$c_{\infty}U^{2-\frac{2}{\eta_{\infty}}} + d_{\infty} \leqslant |\nabla U|^2 \leqslant c_0 U^{2+\frac{2}{\eta_0}} + d_0$$

**Assumption 2.2.3.** For all  $\alpha \in \mathbb{N}^d$ , there exist  $C_{\alpha} > 0$ ,  $k_{\alpha} \in \mathbb{N}$  such that

$$|\partial^{\alpha} U| \leqslant U^{k_{\alpha}} + C_{\alpha}.$$

Let  $\mathcal{P}(U)$  denote the set of admissible functions for our theorems:

$$\mathcal{P}(U) = \left\{ f \in \mathcal{C}^{\infty}(\mathcal{X}, \mathbb{R}) | \forall \alpha \in \mathbb{N}^{2d}, \exists C, c > 0, c < \beta, |\partial^{\alpha} f| \leqslant C e^{cH} \right\}.$$

The first theorem gives the estimates on the semi-group necessary for the proof of weak convergence.

**Theorem 2.2.1.** Suppose Assumptions 2.2.1, 2.2.2 and 2.2.3. Then for all  $f \in \mathcal{P}(U)$ , and all multi-index  $\alpha \in \mathbb{N}^{2d}$ , there exist C, q > 0,  $b \in (\beta(1 - \frac{1}{2d}), \beta)$ , such that for all t > 0,  $z \in \mathcal{X}$ 

$$\left|\partial^{\alpha}(P_t f - \mu_{\beta}(f))(z)\right| \leqslant C e^{-qt} e^{bH(z)}.$$

This theorem allows for the proof of the two following theorems:

**Theorem 2.2.2.** Suppose Assumption 2.2.1, 2.2.2 and 2.2.3. Then there exists  $l_0 > 0$  such that for all  $0 < \ell < \ell_0$ ,  $f \in \mathcal{P}(U)$ , and  $t \ge 0$ , there exists a family  $(C_i(t))_i$  of explicit real numbers such that for all  $k \in \mathbb{N}$ :

$$\mathbb{E}_z(f(\bar{Z}_n)) = \mathbb{E}_z(f(Z_t)) + C_1(t)\delta + \dots + C_k(t)\delta^k + O(\delta^{k+1}),$$

where  $(\bar{Z}_n)_n = (\bar{X}_n, \bar{Y}_n)_n$  is the numerical scheme defined in (2.4),  $n\delta = t$ , and  $O(\delta^{k+1})$  is uniform in t.

**Theorem 2.2.3.** Suppose Assumption 2.2.1, 2.2.2 and 2.2.3. Then there exists  $\delta_0, \ell_0 > 0$  such that for all  $0 < \delta < \delta_0, 0 < \ell < \ell_0$ , the numerical scheme defined in (2.4) admits an unique invariant measure  $\mu_{\beta,\delta} \in \mathcal{M}^1(\mathbb{R}^{2d})$ . For all  $f \in \mathcal{P}(U)$ , there exists C, c > 0 such that for all  $z \in \mathcal{H}_{\delta,\ell}$ :

$$\left|\mathbb{E}_{z}\left(f\left(\bar{Z}_{n}\right)\right)-\mu_{\beta,\delta}(f)\right|\leqslant Ce^{-cn}$$

Moreover, there exists a family  $(\tilde{C}_i)_i$  of explicit real numbers such that for all  $k \in \mathbb{N}$ :

$$\mu_{\beta,\delta}(f) = \mu_{\beta}(f) + \hat{C}_1 \delta + \dots + \hat{C}_k \delta^k + O(\delta^{k+1}).$$

Let's comment on those assumptions and theorems. Assumptions 2.2.1 and 2.2.2 are derived from [10], on which this work is based. As explained in their work, Assumption 2.2.1 is the minimal requirement to ensure the pathwise well-posedness of the process (2.3), as well as for the Gibbs measure to be well-defined and a stationary measure of the Langevin process. Assumption 2.2.2 differs slightly from the one in [10], as they suppose that there exists  $\kappa > 0$  such that for all  $v \in \mathbb{R}^d$ 

$$\nabla^2 U(x)v| \leqslant \frac{\beta}{16d} |\nabla U(x)|^2 |v| + \kappa |v|, \qquad (2.9)$$

which would allow potentials that exhibit logarithmic singularities. Here, we forbid those potentials, following the assumption stated in [82]. This assumption plays a vital role in the construction of a Lyapunov function for both the continuous-time process and the numerical scheme. We need a Lyapunov function of order  $e^{bH}$  for any  $b < \beta$ , which cannot be achieved under (2.9). In this case, it would be possible that the bounds on the semi-group given by Theorem 2.2.1 are not in  $L^1(\mathcal{L}aw(\bar{Z}_n))$ . Since the proof of theorem 2.2.1 relies on computation in Sobolev spaces, and the use of Sobolev embedding, as in [170, 95], we need an additional assumption on the derivatives of the potential of all orders in order to carry out the computation. However, the Villani-type condition presented in [170] is not satisfied by singular potentials, as in the  $H^1$ case from [10]. Therefore, we impose Assumption 2.2.3. This set of assumptions encompasses any repulsive interaction that exhibits a sufficiently rapid explosion (at least algebraic), such as Lennard-Jones interaction, and Coulomb interaction as soon as the particles are living in  $\mathbb{R}^q$  for  $q \ge 3$ , coupled with an additional confinement potential. However, it's important to note that 2-dimensional Coulomb interactions  $(U_i(r) \propto \ln(r))$  do not satisfy these assumptions.

It is well-established in numerical probability that the proof of weak convergence relies on estimates of the kind given by Theorem 2.2.1. The proof of this theorem is based on Gamma calculus, see Section 1.2.1. This enables us to make series expansion at any order of the error induced by the numerical scheme at each step, with a remaining term of the form  $\delta^k \mathbb{E}\left(e^{bH(\bar{Z}_n)}\right)$ . This would be infinite in the case of a numerical scheme without any rejection mechanism like (2.7), but we will show that in the case of the numerical scheme (2.4), there is, under our assumptions, as in the continuous setting, a Lyapunov function of this order, uniformly over small time step. In particular, we may apply the method developed in [158] to get weak convergence as well as expansion à la Talay-Tubaro for this numerical scheme. It can be noticed from the proofs that the constants  $(C_k(t))_k$  and  $(\tilde{C}_k(t))_k$  are independent of  $\ell > 0$ , and can be expressed in the same way as if we considered the numerical scheme (2.7) with a regular potential, as in [107]. For instance, we have the following formula:

$$C_1(t) = \int_0^t \mathbb{E}_z \left( \psi(s, Z_{t-s}) \right) \mathrm{d}s,$$

where

$$\begin{split} \psi(t,z) &= (\nabla U + \gamma y) \cdot (\nabla_y/2 - \nabla_x) P_t f + \frac{1}{2} \partial_t^2 P_t f - \frac{1}{2} y \cdot \nabla_x \partial_t P_t f + \frac{1}{2} \left( \nabla U + \gamma y \right) \cdot \nabla_y^2 y \left( \nabla U + \gamma y \right) \\ &- \frac{1}{2} y \cdot \nabla_{x,y}^2 \left( \nabla U + \gamma y \right) + \gamma \beta^{-1} \sum_{i=1}^d \partial_{x_i} \partial_{y_i} P_t f - \frac{2}{3} \gamma \beta^{-1} \partial_t \Delta_y P_t f \\ &- \frac{7}{12} \gamma \beta^{-1} \left( \nabla U + \gamma y \right) \cdot \nabla_y \Delta_y P_t f + \frac{1}{3} \gamma \beta^{-1} y \cdot \nabla_x \Delta_y P_t f \\ &+ \frac{1}{2} \left( \gamma \beta^{-1} \right)^2 \sum_{i=1}^d \partial_{y_i}^4 P_t f + \frac{1}{6} \left( \gamma \beta^{-1} \right)^2 \sum_{i\neq j=1}^d \partial_{y_j}^2 \partial_{y_j}^2 P_t f. \end{split}$$

Besides, a corollary of Theorem 2.2.2 is the uniform in time weak convergence of the process. For a given  $f \in \mathcal{P}(U)$ , and  $z \in \mathbb{R}^{2d}$ , there exists C > 0 such that for all  $t \ge 0$ ,  $n\delta = t$ :

$$\left|\mathbb{E}_{z}(f(\bar{Z}_{n})) - \mathbb{E}_{z}(f(Z_{t}))\right| \leq C\delta.$$

Theorems 2.2.2 and 2.2.3 are also motivated by the following fact:

$$2\mathbb{E}\left(f\left(Z_{2n}^{\delta/2}\right)\right) - \mathbb{E}\left(f\left(Z_{n}^{\delta}\right)\right) = \mathbb{E}\left(f\left(Z_{t}\right)\right) - \frac{C_{2}}{2}\delta^{2} + O(\delta^{3}),$$

which yields a better order convergence. With arbitrary order expansion, it is possible to get any

order of convergence using a combination of  $\left(\mathbb{E}\left(f\left(Z_{2^{k}n}^{\delta/2^{k}}\right)\right)\right)$ . This is the so-called Romberg-Richardson interpolation. For numerical experiments on this improvement, see the work of Talay and Tubaro [158].

### 2.2.2 Comparison between numerical scheme with and without rejection

In practice, numerical studies often utilize splitting schemes with no rejection on the bounded torus. If the potential is defined on  $\mathbb{T}^d = (\mathbb{R}/\mathbb{Z})^d$ , then the Langevin process (2.3) would be defined on  $\mathbb{T}^d \times \mathbb{R}^d$ . All proofs would still be valid, preserving the theorems' validity. As discussed in the introduction, for the numerical scheme (2.7), it is not possible to take the expectation of unbounded functions, and an invariant probability measure may not exist. Additionally, it also remains an open problem as to whether finite time weak expansion would still hold under our assumptions for bounded functions and numerical schemes without rejection. Regarding the use of schemes without rejection in practical applications, it is justified by the fact that the continuous process does not reach the singularities. For any fixed time T > 0, one can select a time step  $\delta > 0$  small enough so that the numerical scheme without rejection does not get close to the singularities, and the computation of averages would yield reasonable results. However, for a fixed  $\delta$ , the scheme would approach arbitrarily close to the singularity in large time, leading to abnormally large steps. Whereas with the rejection mechanism, it is possible to choose larger time steps, resulting in a more computationally efficient process, and avoiding the issue of getting too close to the singularities. This advantage of the rejection mechanism makes it a preferable choice in certain practical scenarios, such as in the following 1-dimensional toy model:

$$U(x) = \frac{1}{x} + x^2,$$

for which we conducted the following numerical experiment: fix  $\beta^{-1} = 15$ ,  $\gamma = 1$ , and a final time T = 15000. Define the empirical averages by

$$n \mapsto S_n = \frac{1}{n} \sum_{k=0}^{n-1} U(Y_k),$$

where  $Y_k = \bar{X}_k$  or  $Y_k = \bar{X}_k$ . For different values of  $\delta$ , we simulate K = 1000 copies  $S^i$  of  $S_{\lfloor T/\delta \rfloor}$ . We plot the evolution of the proportion of copies that have more than 1% error:

$$\delta \mapsto \frac{1}{K} \sum_{i=1}^{K} \mathbb{1}_{S^i \notin [0.99\mu_\beta(U), 1.01\mu_\beta(U)]},$$

where we estimated  $\mu_{\beta}(U)$  using Wolfram Alpha. We get the result in Figure 2.1.

We can observe that for very small  $\delta$ , the scheme without rejection produces satisfactory results. Nonetheless, there exists an interval of values of  $\delta$  where the scheme with rejection significantly outperforms the other. The proportion of failures does not converge to 0 because of the error stemming from insufficiently long simulation times. For values of  $\delta$  that aren't small enough, the energy threshold cannot, to prevent explosion, be set high enough to ensure the convergence of the empirical averages to a value close to  $\mu_{\beta}(U)$  ( $\approx 9.035$ ). This behavior is illustrated in Figure 2.2, where we display a typical trajectory for both cases, with  $\delta = 10^{-2}$ .

For smaller  $\beta$ , the explosion phenomenon would be even more prominent because the variance of the Gaussian distribution is proportional to  $\gamma \delta \beta^{-1}$ . As  $\beta$  decreases, the variance increases,



Figure 2.1: Failure probability.



Figure 2.2:  $n \mapsto S_n, \, \delta = 10^{-2}$ .

making the numerical scheme more susceptible to divergent behavior and explosions in the trajectories. Hence, a smaller  $\beta$  exacerbates the challenge of handling the singularities and maintaining stable and accurate numerical computations. This unstable behavior could also arise in the case of a metastable process. The metastability imposes long simulation times, which could prove longer than the time needed for the numerical scheme to visit the singularities. The conclusion of this experiment/discussion is that the use of the rejection mechanism in practical applications may help to avoid divergent behavior and explosions, leading to more stable and accurate results.

An alternative approach to address the singularities is to employ a Metropolis-adjusted scheme, as the one presented in [136]. Similarly to our scheme, Metropolis-adjusted algorithms tend to reject transitions going too close to singularities, but the difference is that they also reject moves in low-energy regions, where the process spends most of its time. For the toy model discussed in this section, such a scheme exhibits remarkably high performance, with a success rate exceeding 99% for all considered time steps. However, in high-dimensional settings, Metropolis-adjusted schemes often result in numerous rejections and exhibit suboptimal performance. As a result, they are typically not used in the realms of molecular dynamics or statistics, even in the case of regular potentials.

# 2.3 Proof of Theorem 2.2.1

The proof of Theorem 2.2.1 relies on Sobolev embedding, and  $H^k$ -estimates. Similar results have been established for Langevin processes and close to quadratic potentials, in the sense that there exists  $0 < m \leq M$  such that  $mI_d \leq \nabla^2 U \leq MI_d$ , where  $I_d$  stands for the identity matrix of size d, in [157]. Here we will use a method previously employed in [170, 95]. For a given  $f \in \mathcal{P}(U)$ , our goal is to show bound of the form:

$$\int_{\mathcal{X}} |\partial^{\alpha} \left( P_t f e^{-bH} \right)|^2 \mathrm{d}z \leqslant C e^{-qt},$$

for some  $0 < b < \beta$  and all  $\alpha \in \mathbb{N}^{2d}$ . Sobolev embedding would then yield Theorem 2.2.1. To do so, we define a norm  $\|\cdot\|_k$  that dominates the  $H^k$ -Sobolev norm, but such that  $t \mapsto \|P_t f - \mu_\beta(f)\|_k$  converges exponentially fast to 0. However, contrary to the cited work, we need to take into account the singular potential by using a Lyapunov function. It was shown in [10] that the Langevin process (2.3) admits a family of Lyapunov functions  $(V_{0,\bar{b}})$  such that for all  $0 < \bar{b} < \beta/(2d), \varepsilon > 0$ :

$$C^{-1}e^{(1-\varepsilon)\bar{b}H} < V_{0,\bar{b}} < Ce^{(1+\varepsilon)\bar{b}H},$$

see [10, Equation 5.2] for its exact definition and Proposition 2.3.3 below for its main properties. Thanks to this family of Lyapunov functions, we may now define our modified Sobolev norms. Let  $\nabla_x^l \nabla_y^p h$  denote the vector of all derivative of h of order l on x and p on y:

$$\nabla_x^l \nabla_y^p h = \left\{ \partial_x^{\alpha_1} \partial_y^{\alpha_2} h \mid |\alpha_1| = l, \, |\alpha_2| = p \right\},$$

and its norm

$$|\nabla^l_x \nabla^p_y h|^2 = \sum_{|\alpha_1|=l; |\alpha_2|=p} |\partial^{\alpha_1}_x \partial^{\alpha_2}_y h|^2.$$

Fix  $0 < \overline{b} < \beta/(2d)$ , and  $k \in \mathbb{N}^*$ . Write  $V = V_{0,\overline{b}}^{1/k}$ , and for  $p \in [\![1,k]\!]$ ,  $W_p = V^p + \lambda$ , for some  $\lambda > 0$ . For  $h \in \mathcal{C}^{\infty}(\mathcal{X}, \mathbb{R})$ , denote:

$$\|h\|_{mH^{r,k},\bar{b}}^{2} = \int_{\mathcal{X}} h^{2} W_{k} \mathrm{d}\mu_{\beta} + \int_{\mathcal{X}} \sum_{p=1}^{r} \left( \sum_{i=0}^{p-1} \omega_{i,p} |\nabla_{x}^{i} \nabla_{y}^{p-i} h|^{2} + \omega_{p,p} |\left(\nabla_{x}^{p} - \xi \nabla_{x}^{p-1} \nabla_{y}\right) h|^{2} \right) (1 + \varepsilon_{p} W_{k-p}) \mathrm{d}\mu_{\beta},$$
(2.10)

where  $(\omega_{i,p})$  and  $(\varepsilon_p)$  are two families of positive parameters to be fixed later, and  $\xi = \frac{\gamma}{2} + \sqrt{\frac{\gamma^2}{4} + 1}$ . The exact value of  $\xi$  does not matter, as the important term is the scalar product  $\nabla_x^p \cdot \nabla_x^{p-1} \nabla_y$ . However this expression allows for simplified computation, see inequality (2.21). For the sequel of this work, we will drop the dependence in  $\bar{b}$  in the definition of the norm and simply write  $\|\cdot\|_{mH^{r,k}}$ . Write:

$$mH^k = \left\{ h \in \mathcal{L}^2(\mathcal{X}, \mathbb{R}), \quad \|h\|_{mH^{k,k}} < \infty \right\}.$$

The goal is to show that for  $f \in \mathcal{P}(U)$ , we may chose  $\bar{b}$  such that  $||P_t f - \mu_\beta(f)||_{mH^{r,k}}$  decreases exponentially fast along the dynamic.

**Theorem 2.3.1.** Under Assumptions 2.2.1, 2.2.2 and 2.2.3, for all  $f \in \mathcal{P}(U)$ ,  $k \in \mathbb{N}$ , there exists  $0 < \overline{b} < \beta/(2d)$  such that for all  $t \ge 0$ ,  $P_t f \in mH^k$ , and we have that there is  $q_k > 0$  such that:

$$||P_t f - \mu_\beta(f)||_{mH^{k,k}} \leqslant e^{-q_k t} ||f||_{mH^{k,k}}.$$
(2.11)

This section will be mostly about the proof of this theorem, which relies on Gamma calculus, as presented in 1.2.1. We start by proving  $H^1$  convergence in Section 2.3.1, which serves as an initial case for an induction argument which will prove Theorem 2.3.1 in Section 2.3.2. Finally, Theorem 2.2.1 will be proven is Section 2.3.3.

In the sequel, we denote by L the generator of the process (2.3) given by

$$L = y \cdot \nabla_x - \nabla U \cdot \nabla_y - \gamma y \cdot \nabla_y + \gamma \beta^{-1} \Delta_y, \qquad (2.12)$$

and by  $L^*$  its adjoint in  $L^2(\mu_\beta)$ , given by:

$$L^* = -y \cdot \nabla_x + \nabla U \cdot \nabla_y - \gamma y \cdot \nabla_y + \gamma \beta^{-1} \Delta_y$$

Recall the definition of the generalized *carré du champ* 

$$\Gamma_{\phi}(h) = \frac{1}{2} \left( L(\phi(h)) - D_h \phi(h) Lh \right), \qquad (2.13)$$

where  $\phi : \mathcal{C}^{\infty}(\mathcal{X}) \to \mathcal{C}^{\infty}(\mathcal{X})$  and  $D_h \phi$  denote the differential operator of  $\phi$ . Recall also the Proposition that allows for the computations of this generalized *carré du champ*.

**Proposition 2.3.1.** If there exists  $A = (A_1, \dots, A_p) : \mathcal{C}^{\infty} \to (\mathcal{C}^{\infty})^p$  a linear operator such that  $\phi(h) = |Ah|^2$ , then

$$\Gamma_{L,\phi}(h) = \Gamma_{L,2}(Ah) + Ah \cdot [L,A]h$$

where  $\Gamma_{L,2}(Ah) = \sum_{i=1}^{p} \Gamma(A_i h), \ \Gamma(h) = \Gamma_{L,h^2}(h), \ and \ [L, A] = ([L, A_1], \cdots, [L, A_p]).$ 

#### 2.3.1 $H^1$ -setting

We now adapt the introduction to the singular case, as done in [10], to serve as the initialization of our induction argument. However, the modified  $H^1$ -norm used in the cited work does not contain any Lyapunov function in the  $\dot{H}^1$  part (i.e. the term with derivative of order 1), and we need some for the induction. Hence we use a slightly different norm, given in (2.10), which becomes for r = 1:

$$\|h\|_{mH^{1,k}} = \int_{\mathcal{X}} \left( h^2 W_k + \omega \left( |\nabla_y h|^2 + |(\nabla_x - \xi \nabla_y) h|^2 \right) (1 + \varepsilon W_{k-1}) \right) \mathrm{d}\mu_\beta,$$
(2.14)

where  $\xi = \frac{\gamma}{2} + \sqrt{\frac{\gamma^2}{4} + 1}$ , and  $\varepsilon > 0$ , will be determined later. As in [10], we need a local Poincaré inequality:

**Proposition 2.3.2.** For all compact set  $K \subset \mathcal{X}$ ,  $\mu_{\beta}$  satisfies a Poincaré inequality on K: there exists  $\rho > 0$  such that for all  $f \in \mathcal{C}^{\infty}(\mathcal{X}, \mathbb{R})$ :

$$\int_{K} f^{2} d\mu_{\beta} \leqslant \rho \int_{K} |\nabla f|^{2} d\mu_{\beta} + \frac{1}{\mu_{\beta}(K)} \left( \int_{K} f d\mu_{\beta} \right)^{2}.$$
(2.15)

*Proof.* Proof of such inequality can be found in [6].

The functions 
$$W$$
 and  $V$  satisfy the following properties, which are described in [10, Theorem 4.15]:

**Proposition 2.3.3.** Under Assumptions 2.2.1 and 2.2.2, for all  $\bar{b} > 0$ , all  $k \in \mathbb{N}$ , there exists  $V_{0,\bar{b}} : \mathcal{X} \to [1,\infty)$  such that if  $V = V_{0,\bar{b}}^{1/k}$ ,  $W_p = V^p + \lambda$  for some  $\lambda > 0$ ,  $V : \mathcal{X} \to [1,\infty)$  is  $\mathcal{C}^2$ , and there exist  $\alpha, \sigma > 0$ ,  $J \subset \mathcal{X}$  such that:

• For all  $\varepsilon > 0$ , there exists C > 0 such that

$$C^{-1}e^{(1-\varepsilon)\bar{b}H} < V < Ce^{(1+\varepsilon)\bar{b}H}.$$
(2.16)

• J is compact, connected, and for all  $p \in [1, k]$ :

$$L^* V^p \leqslant -\alpha \frac{p}{k} V^p + \sigma \mathbb{1}_J.$$

• For all  $p \in \llbracket 1, k \rrbracket$ ,  $g \in \mathcal{C}_b^{\infty}(\mathcal{X}, \mathbb{R})$ , we have that  $LgW_p, gL^*W_p \in L^1(\mu_\beta)$  and

$$\int_{\mathcal{X}} Lg W_p d\mu_\beta = \int_{\mathcal{X}} g L^* W_p d\mu_\beta$$

•  $\forall (x, y) \in \mathbf{X},$ 

$$V^{k}(x,y) \ge \frac{2\sigma\mu_{\beta}(J^{c})}{\alpha\mu_{\beta}(J)}.$$
(2.17)

•  $\forall (x,y) \in \mathbf{X}, v \in \mathbb{R}^d$ ,

$$W_k(x,y)|v|^2 \ge \beta(\sigma\rho'+1)\left(\left(\frac{1}{2} + \frac{2}{\gamma^2}\right)|v|^2 + \frac{1}{2\gamma^2}|\nabla^2 Uv|^2\right),$$
(2.18)

where

$$\rho' = 4(1 + \xi^2)\rho/\gamma, \tag{2.19}$$

and  $\rho$  is the constant from the Poincaré inequality on J.

In particular, (2.16) yields that  $\lim_{H\to\infty} V = \infty$ , and (2.17) that the set J is not so large. In order to study the evolution of the norm (2.14) along the trajectories, we use Gamma calculus. All computations will be performed on functions  $h : \mathcal{X} \to \mathbb{R}$  such that

$$\int_{\mathcal{X}} h \mathrm{d}\mu_{\beta} = 0. \tag{2.20}$$

Recall the definition (2.13) of  $\Gamma_{\phi}$ . We have

$$[L, \nabla_x] = \nabla^2 U \nabla_y, \qquad [L, \nabla_y] = -\nabla_x + \gamma \nabla_y.$$

Hence from Proposition 2.3.1 we get that:

$$\Gamma_{|\nabla_y \cdot|^2}(h) \ge \gamma \beta^{-1} |\nabla_y^2 h|^2 - \nabla_y h \cdot (\nabla_x - \xi \nabla_y) h + (\gamma - \xi) |\nabla_y h|^2$$

and:

$$\begin{split} \Gamma_{|(\nabla_x - \xi \nabla_y) \cdot|^2}(h) &\geqslant \gamma \beta^{-1} |(\nabla_x \nabla_y - \xi \nabla_y^2) h|^2 + \xi |(\nabla_x - \xi \nabla_y) h|^2 \\ &+ \xi (\xi - \gamma) \nabla_y h \cdot (\nabla_x - \xi \nabla_y) h + \nabla^2 U \nabla_y h \cdot (\nabla_x - \xi \nabla_y) h. \end{split}$$

Finally, with our choice of  $\xi$  and  $\phi(h) = |\nabla_y h|^2 + |(\nabla_x - \xi \nabla_y)h|^2$ , one gets:

$$\begin{split} \Gamma_{\phi}(h) &= \Gamma_{|\nabla_{y} \cdot|^{2}}(h) + \Gamma_{|(\nabla_{x} - \xi \nabla_{y}) \cdot|^{2}}(h) \\ &\geqslant \gamma \beta^{-1} |\nabla_{y}^{2}h|^{2} + \gamma \beta^{-1} |(\nabla_{x} \nabla_{y} - \xi \nabla_{y}^{2})h|^{2} \\ &+ \gamma |(\nabla_{x} - \xi \nabla_{y})h|^{2} - \frac{2}{\gamma} |\nabla_{y}h|^{2} + \nabla^{2} U \nabla_{y}h \cdot (\nabla_{x} - \xi \nabla_{y})h |^{2} \end{split}$$

Young inequality then yields:

$$\Gamma_{\phi}(h) \ge \gamma \beta^{-1} |\nabla_y^2 h|^2 + \gamma \beta^{-1} |(\nabla_x \nabla_y - \xi \nabla_y^2) h|^2 + \frac{\gamma}{2} |(\nabla_x - \xi \nabla_y) h|^2 - \mathcal{R}(x, \nabla_y h), \qquad (2.21)$$

where

$$\mathcal{R}(x,v) = \frac{2}{\gamma}|v|^2 + \frac{|\nabla^2 Uv|^2}{2\gamma}.$$

Write:

$$H(t) = \|P_t f - \int_{\mathcal{X}} f \mathrm{d}\mu_\beta\|_{mH^{1,k}}.$$

Because of the Lyapunov function, we have additional terms in the derivative of H in comparison to the regular case. This formally reads:

$$H'(t) = \int_{\mathcal{X}} \left( (P_t f)^2 L^* W_k - \gamma \beta^{-1} |\nabla_y P_t f|^2 W_k \right) d\mu_\beta + \omega \varepsilon \int_{\mathcal{X}} \left( |\nabla_y P_t f|^2 + |(\nabla_x - \xi \nabla_y) P_t f|^2 \right) L^* W_{k-1} d\mu_\beta$$

$$-\omega \int_{\mathcal{X}} \left( \Gamma_{|\nabla_{y} \cdot|^{2}}(P_{t}f) + \Gamma_{|(\nabla_{x} - \xi \nabla_{y}) \cdot|^{2}}(P_{t}f) \right) \left( 1 + \varepsilon W_{k-1} \right) \mathrm{d}\mu_{\beta}.$$

Hence to conclude in the  $H^1$  setting, we need a control on the additional terms. This is the object of the next lemma, which is also the initialisation of the induction argument used in the next section for the  $H^k$  setting:

**Lemma 2.3.1.** For all  $k \in \mathbb{N}^*$ , under Assumptions 2.2.1 and 2.2.2, there exist  $q, \omega, \varepsilon > 0$  such that:

$$\int_{\mathcal{X}} \left( h^{2}L^{*}W_{k} - \gamma\beta^{-1} |\nabla_{y}h|^{2}W_{k} \right) d\mu_{\beta} 
+ \omega\varepsilon \int_{\mathcal{X}} \left( |\nabla_{y}h|^{2} + |(\nabla_{x} - \xi\nabla_{y})h|^{2} \right) L^{*}W_{k-1} d\mu_{\beta} 
- \omega \int_{\mathcal{X}} \left( \Gamma_{|\nabla_{y}\cdot|^{2}}(h) + \Gamma_{|(\nabla_{x} - \xi\nabla_{y})\cdot|^{2}}(h) \right) \left( 1 + \varepsilon W_{k-1} \right) d\mu_{\beta} 
\leq -q \left( \|h\|_{mH^{1,k}} + \int_{\mathcal{X}} W_{k-1} \left( |\nabla_{y}^{2}h|^{2} + |\nabla_{x}\nabla_{y}h|^{2} \right) d\mu_{\beta} \right), \quad (2.22)$$

for all  $h \in \mathcal{C}^{\infty}_{b}(\mathcal{X}, \mathbb{R})$ .

Notice that in contrast to inequality (4.8) of [10], we keep the higher order derivative for the induction, see Section 2.3.2.

*Proof.* We first treat the  $L^2$ -term using the Lyapunov property of V:

$$\int_{\mathcal{X}} h^2 L^* W_k \mathrm{d}\mu_\beta \leqslant -\frac{\alpha}{1+\lambda} \int_{\mathcal{X}} h^2 W_k \mathrm{d}\mu_\beta + \sigma \int_J h^2 \mathrm{d}\mu_\beta.$$

Using Cauchy-Schwarz inequality, inequality (2.17) and that h satisfies (2.20), one gets

$$\left(\int_{J} h \mathrm{d}\mu_{\beta}\right)^{2} = \left(\int_{J^{c}} h \mathrm{d}\mu_{\beta}\right)^{2} \leqslant \mu_{\beta}(J^{c}) \int_{\mathcal{X}} h^{2} \mathrm{d}\mu_{\beta} \leqslant \frac{\alpha \mu_{\beta}(J)}{2\sigma(1+\lambda)} \int_{\mathcal{X}} h^{2} W_{k} \mathrm{d}\mu_{\beta}.$$

The local Poincaré inequality (2.15) on J can then be written as follow:

$$\begin{split} \sigma \int_{J} h^{2} \mathrm{d}\mu_{\beta} &\leqslant \sigma \rho' \frac{\gamma}{2} \int_{J} (|(\nabla_{x} - \xi \nabla_{y})h|^{2} + |\nabla_{y}h|^{2}) \mathrm{d}\mu_{\beta} + \frac{\sigma}{\mu_{\beta}(J)} \left( \int_{J} h \mathrm{d}\mu_{\beta} \right)^{2} \\ &\leqslant \sigma \rho' \int_{\mathcal{X}} \frac{\gamma}{2} |(\nabla_{x} - \xi \nabla_{y})h|^{2} - \mathcal{R}(x, \nabla_{y}h) + \mathcal{R}(x, \nabla_{y}h) + |\nabla_{y}h|^{2} \mathrm{d}\mu_{\beta} \\ &+ \frac{\alpha}{2(1+\lambda)} \int_{\mathcal{X}} h^{2} W_{k} \mathrm{d}\mu_{\beta}, \end{split}$$

where  $\rho'$  was defined in (2.19). Using inequality (2.18), we get:

$$\int_{\mathcal{X}} \mathcal{R}(x, \nabla_y h) + \frac{\gamma}{2} |\nabla_y h|^2 \mathrm{d}\mu_\beta \leqslant \frac{\gamma \beta^{-1}}{\sigma \rho' + 1} \int_{\mathcal{X}} |\nabla_y h|^2 W_k \mathrm{d}\mu_\beta.$$

We also have from inequality (2.21):

$$\int_{\mathcal{X}} \frac{\gamma}{2} |(\nabla_x - c\nabla_y)h|^2 - \mathcal{R}(x, \nabla_y h) \mathrm{d}\mu_\beta \leqslant \int \Gamma_{|\nabla_y \cdot|^2}(h) + \Gamma_{|(\nabla_x - c\nabla_y) \cdot|^2}(h) \mathrm{d}\mu_\beta.$$

We treat the second line of (2.22) using the Lyapunov property as follow:

$$\begin{split} &\int_{\mathcal{X}} \left( |\nabla_{y}h|^{2}L^{*}W_{k-1} + |(\nabla_{x} - \xi\nabla_{y})h|^{2}L^{*}W_{k-1} \right) \mathrm{d}\mu_{\beta} \\ &\leqslant \int_{\mathcal{X}} \left( |\nabla_{y}h|^{2} + |(\nabla_{x} - \xi\nabla_{y})h|^{2} \right) \sigma \mathbb{1}_{J} \mathrm{d}\mu_{\beta} \\ &\leqslant \frac{2\sigma}{\gamma} \int_{\mathcal{X}} \left( \frac{\gamma}{2} |\nabla_{y}h|^{2} + \mathcal{R}(x, \nabla_{y}h) + \frac{\gamma}{2} |(\nabla_{x} - \xi\nabla_{y})h|^{2} - \mathcal{R}(x, \nabla_{y}h) \right) \mathrm{d}\mu_{\beta} \\ &\leqslant \frac{2\sigma}{\beta(1 + \sigma\rho')} \int_{\mathcal{X}} |\nabla_{y}h|^{2} W_{k} \mathrm{d}\mu_{\beta} + \frac{2\sigma}{\gamma} \int \Gamma_{|\nabla_{y}\cdot|^{2}}(h) + \Gamma_{|(\nabla_{x} - \xi\nabla_{y})\cdot|^{2}}(h) \mathrm{d}\mu_{\beta}. \end{split}$$

Next, the  $\varepsilon$ -term of the third line is bounded thanks to (2.21) as follow:

$$-\int_{\mathcal{X}} \left( \Gamma_{|\nabla_{y} \cdot|^{2}}(h) + \Gamma_{|(\nabla_{x} - \xi \nabla_{y}) \cdot|^{2}}(h) \right) W_{k-1} d\mu_{\beta}$$

$$\leq -\gamma \beta^{-1} \int_{\mathcal{X}} \left( |\nabla_{y}^{2}h|^{2} + |(\nabla_{x} \nabla_{y} - \xi \nabla_{y})h|^{2} \right) W_{k-1} d\mu_{\beta}$$

$$- \frac{\gamma}{2} \int_{\mathcal{X}} |(\nabla_{x} - \xi \nabla_{y})h|^{2} W_{k-1} d\mu_{\beta} + \int_{\mathcal{X}} \mathcal{R}(x, \nabla_{y}h) W_{k-1} d\mu_{\beta}.$$

Thanks to Assumption 2.2.2 we get that there is C > 0 such that  $|\nabla^2 U| \leq CV$ . This yields that

$$\begin{aligned} \mathcal{R}(x, \nabla_y h) W_{k-1} &\leq C' \left( |\nabla_y h|^2 + |\nabla^2 U \nabla_y h|^2 \right) W_{k-1} \\ &\leq C'' \left( |\nabla_y h|^2 + |\nabla_y h|^2 V^{1/(k-1)} \right) W_{k-1} \leq \kappa |\nabla_y h|^2 W_k \end{aligned}$$

and:

$$\begin{split} -\int_{\mathcal{X}} \left( \Gamma_{|\nabla_{y}\cdot|^{2}}(h) + \Gamma_{|(\nabla_{x}-\xi\nabla_{y})\cdot|^{2}}(h) \right) W_{k-1} \mathrm{d}\mu_{\beta} \\ \leqslant -\gamma\beta^{-1} \int_{\mathcal{X}} \left( |\nabla_{y}^{2}h|^{2} + |\nabla_{x}\nabla_{y}h|^{2} \right) W_{k-1} \mathrm{d}\mu_{\beta} \\ - \frac{\gamma}{2} \int_{\mathcal{X}} |(\nabla_{x}-\xi\nabla_{y})h|^{2} W_{k-1} \mathrm{d}\mu_{\beta} + \kappa \int_{\mathcal{X}} |\nabla_{y}h|^{2} W_{k} \mathrm{d}\mu_{\beta}. \end{split}$$

We then get the following upper bound for the left-hand side of equation (2.22):

$$\begin{split} &-\frac{\alpha}{2(1+\lambda)}\int_{\mathcal{X}}h^{2}W_{k}\mathrm{d}\mu_{\beta} \\ &\quad -\left(\gamma\beta^{-1}-\frac{2\varepsilon\sigma\beta^{-1}}{\sigma\rho'+1}-\kappa\varepsilon\omega\right)\int_{\mathcal{X}}|\nabla_{y}h|^{2}W_{k}\mathrm{d}\mu_{\beta} \\ &\quad -\varepsilon\omega\gamma\beta^{-1}\int_{\mathcal{X}}\left(|\nabla_{y}^{2}h|^{2}+|\nabla_{x}\nabla_{y}h|^{2}\right)W_{k-1}\mathrm{d}\mu_{\beta} \\ &\quad +\left(\frac{2\sigma\varepsilon\omega}{\gamma}+\sigma\rho'-\omega\right)\int_{\mathcal{X}}\Gamma_{|\nabla_{y}\cdot|^{2}}(h)+\Gamma_{|(\nabla_{x}-\xi\nabla_{y})\cdot|^{2}}(h)\mathrm{d}\mu_{\beta} \\ &\quad -\frac{\gamma\varepsilon\omega}{2}\int_{\mathcal{X}}|(\nabla_{x}-\xi\nabla_{y})h|^{2}W_{k-1}\mathrm{d}\mu_{\beta}. \end{split}$$

Taking  $\omega > \sigma \rho'$ ,  $\varepsilon$  small enough so that the second and fourth terms of the previous bounds are negative, and using the fact that  $W_{k-1} \leq W_k$  (because  $V \geq 1$ ) then concludes the proof.

Removing the second order terms, Lemma 2.3.1 formally yields:

$$H'(t) \leqslant -qH(t),$$

for some q > 0. Since  $W_k \ge 1$  for all k > 0, this would imply convergence in  $H^1$ -norm

$$||P_t f - \mu_\beta(f)||_{H^1} \leq C e^{-qt} ||f - \mu_\beta(f)||_{mH^{1,k}},$$

for some C > 0. This is only a formal computation, which we will justify in the proof of Theorem 2.3.1.

## 2.3.2 $H^k$ -setting

Building upon the result of the previous section, we may now perform computations in higherorder Sobolev spaces. From those computations arise derivatives of U of arbitrary orders, hence we now need to assume Assumption 2.2.3. Since the Lyapunov function satisfies  $V = e^{bH+o(H)}$ , for some  $0 < b < \beta$ , this assumption implies: for all  $\alpha \in \mathbb{N}^d$ , there exists  $\kappa_{\alpha} > 0$  such that

$$|\partial^{\alpha}U| \leqslant \kappa_{\alpha}V. \tag{2.23}$$

Recall the definition of the modified Sobolev norm from (2.10), fix some  $r \in [1, k]$ , and write:

$$H_{r,k}(t) = \|P_t f - \int_{\mathcal{X}} f \mathrm{d}\mu_\beta\|_{mH^{r,k}}.$$

We formally have:

$$H_{r,k}'(t) = \int_{\mathcal{X}} \left( (P_t f)^2 L^* W_k - \gamma \beta^{-1} |\nabla_y P_t f|^2 W_k \right) d\mu_\beta + \int_{\mathcal{X}} \sum_{p=1}^r \varepsilon_p \left( \sum_{i=0}^{p-1} \omega_{i,p} |\nabla_x^i \nabla_y^{p-i} P_t f|^2 + \omega_p | \left( \nabla_x^p - \xi \nabla_x^{p-1} \nabla_y \right) P_t f|^2 \right) L^* W_{k-p} d\mu_\beta - \int_{\mathcal{X}} \sum_{p=1}^r \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p-i}(P_t f) + \omega_p \Gamma_p(P_t f) \right) (1 + \varepsilon_p W_{k-p}) d\mu_\beta$$
(2.24)

where we wrote:

$$\Gamma_{l,p} = \Gamma_{|\nabla_x^l \nabla_y^p \cdot|^2},$$

and

$$\Gamma_p = \Gamma_{\left| \left( \nabla_x^p - \xi \nabla_x^{p-1} \nabla_y \right) \cdot \right|^2}$$

for any  $l, p \in \mathbb{N} \times \mathbb{N}^*$ . As in the previous section, we need to bound this derivative. This bound is obtained through an induction principle on  $r \in [\![1, k]\!]$ , and the repetitive use of Proposition 2.3.1. First we compute:

$$[L, \nabla^l_x \nabla^p_y] = \gamma \nabla^l_x \nabla^p_y - \mathbbm{1}_{p \ge 1} \nabla^{l+1}_x \nabla^{p-1}_y + \sum_{i=1}^l \binom{l}{i} \nabla^{i+1} U \otimes \nabla^{l-i}_x \nabla^p_y,$$

where the terms in the sum as to be understood as:

$$\begin{vmatrix} l \\ i \end{vmatrix} \nabla^{i+1} U \otimes \nabla^{l-i}_x \nabla^p_y = \left( \sum_{j=1}^d \sum_{k_s \leqslant l_s} \prod_{s=1}^d \binom{l_s}{k_s} \partial_{x_j} \partial^k U \partial_{y_j} \partial_x^{l-k} \partial_y^{p+1} \right)_{\sum l_j = l, \sum p_j = p}$$

Using Proposition 2.3.1, we get that:

$$\Gamma_{l,p}(h) = \Gamma(\nabla_x^l \nabla_y^p h) + \nabla_x^l \nabla_y^p h \cdot [L, \nabla_x^l \nabla_y^p]h.$$

If  $p \ge 2$ , we may then bound below using inequality (2.23):

$$\Gamma_{l,p}(h) \ge \gamma \beta^{-1} |\nabla_x^l \nabla_y^{p+1} h|^2 - \theta_{l,p} \left( |\nabla_x^l \nabla_y^p h|^2 + V \sum_{i=1}^l |\nabla_x^{l-i} \nabla_y^{p+1} h|^2 \right) - \frac{1}{4\gamma} |\nabla_x^{l+1} \nabla_y^{p-1}|^2, \quad (2.25)$$

for some  $\theta_{l,p} \ge 0$ . If p = 1, there is a term  $|\nabla_x^{l+1}h|^2$  in the commutator, which will be problematic, as we cannot control it by induction using the derivative of the lower-order terms. In order to take care of it, we fix a small parameter  $\eta_l > 0$ , use the inequality  $a \cdot b \le |a|^2/(2\eta) + \eta |b|^2/2$  and we bound below again using inequality (2.23):

$$\Gamma_{l,1}(h) \ge \gamma \beta^{-1} |\nabla_x^l \nabla_y^2 h|^2 - \theta_{l,1} \left( |\nabla_x^l \nabla_y^1 h|^2 + V \sum_{i=1}^l |\nabla_x^{l-i} \nabla_y^2 h|^2 \right) - \eta_l |(\nabla_x^{l+1} - \xi \nabla_x^l \nabla_y) h|^2.$$
(2.26)

As usual in hypocoercive computations for kinetic processes, the derivative of a Sobolev norm of order p will lack in its derivative a term of the form  $|\nabla_x^p h|^2$ . It will be the derivative of the cross term  $\nabla_x^p h \cdot \nabla_x^{p-1} \nabla_y$  in  $|(\nabla_x^p - c \nabla_x^{p-1} \nabla_y) h|^2$  that will give us this missing derivative. Indeed, using that  $\xi \ge \gamma$  and inequality (2.23), one has:

$$\nabla_x^p h \cdot [L, -\xi \nabla_x^{p-1} \nabla_y] h = \xi \nabla_x^p h \cdot \left( \nabla_x^p - \gamma \nabla_x^{p-1} \nabla_y - \sum_{i=1}^p \left| \stackrel{p}{i} \right| \nabla^{i+1} U \otimes \nabla_x^{p-i} \nabla_y \right) h$$
$$\geqslant \frac{\gamma}{2} |\nabla_x^p h|^2 - CV \left( \sum_{l=1}^p |\nabla_x^{p-l} \nabla_y h|^2 + \sum_{l=1}^{p-1} |\nabla_x^{p-1-l} \nabla_y^2 h|^2 \right). \quad (2.27)$$

We also have that

$$|\left(\nabla_y \nabla_x^p - \xi \nabla_x^{p-1} \nabla_y^2\right) h|^2 \geqslant \frac{1}{2} |\nabla_x^p \nabla_y h|^2 - \frac{1}{2} \xi |\nabla_x^{p-1} \nabla_y^2 h|^2$$

Hence after rearranging the terms, Proposition 2.3.1 again yields:

$$\Gamma_p(h) \ge \frac{\gamma \beta^{-1}}{2} |\nabla_x^p \nabla_y h|^2 + \frac{\gamma}{2} |\left(\nabla_x^p - \xi \nabla_x^{p-1} \nabla_y\right) h|^2 - \theta_{p,0} V\left(\sum_{l=1}^p |\nabla_x^{p-l} \nabla_y h|^2 + \sum_{l=0}^{p-1} |\nabla_x^{p-l-l} \nabla_y^2 h|^2\right),$$
(2.28)

for some  $\theta_{p,0} > 0$ . Thanks to those different lower bounds, we will be able to prove the next lemma which is the central part of this section.

**Lemma 2.3.2.** Fix  $k \in \mathbb{N}^*$ . There exists  $(\omega_{i,p})_{1 \leq i \leq p \leq k}$ ,  $(\varepsilon_p)_{1 \leq p \leq k}$ , all positive, such that for all  $r \in [\![1,k]\!]$ , there exists  $q_r > 0$  such that for all  $h \in C_b^{\infty}(\mathcal{X}, \mathbb{R})$ :

$$\int_{\mathcal{X}} \left( h^2 L^* W_k - \gamma \beta^{-1} |\nabla_y h|^2 W_k \right) d\mu_\beta 
+ \int_{\mathcal{X}} \sum_{p=1}^r \varepsilon_p \left( \sum_{i=0}^{p-1} \omega_{i,p} |\nabla_x^i \nabla_y^{p-i} h|^2 + \omega_p | \left( \nabla_x^p - \xi \nabla_x^{p-1} \nabla_y \right) h |^2 \right) L^* W_{k-p} d\mu_\beta 
- \int_{\mathcal{X}} \sum_{p=1}^r \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p-i}(h) + \omega_p \Gamma_p(h) \right) (1 + \varepsilon_p W_{k-p}) d\mu_\beta 
\leqslant -q_r \left( \|h\|_{mH^{r,k}} + \int_{\mathcal{X}} \sum_{i=0}^r |\nabla_x^i \nabla_y^{r+1-i} h|^2 W_{k-r} d\mu_\beta \right).$$
(2.29)

*Proof.* The proof is done by induction. The case r = 1 corresponds to Lemma 2.3.1. Let  $r \in [\![1, k]\!]$ , and we suppose the inequality (2.29) true for such an r. To prove the result at rank r + 1, we want to bound:

$$\begin{split} &-q_r \int_{\mathcal{X}} \left( h^2 W_k + \sum_{1 \leqslant i+j \leqslant r,j \geqslant 1} |\nabla_x^i \nabla_y^j h|^2 W_{k-(i+j)} + \sum_{i=1}^r |\left(\nabla_x^i - \xi \nabla_x^{i-1} \nabla_y\right) h|^2 W_{k-i} \right) \mathrm{d}\mu_\beta \\ &- q_r \int_{\mathcal{X}} \sum_{i=0}^r |\nabla_x^i \nabla_y^{r+1-i} h|^2 W_{k-r} \mathrm{d}\mu_\beta \\ &+ \varepsilon_{r+1} \int_{\mathcal{X}} \left( \sum_{i=0}^r \omega_{i,r+1} |\nabla_x^i \nabla_y^{r+1-i} h|^2 + \omega_{r+1} |\left(\nabla_x^{r+1} - \xi \nabla_x^r \nabla_y\right) h|^2 \right) L^* W_{k-(r+1)} \mathrm{d}\mu_\beta \\ &- \int_{\mathcal{X}} \left( \sum_{i=0}^r \omega_{i,r+1} \Gamma_{i,r+1-i}(h) + \omega_{r+1} \Gamma_{r+1}(h) \right) \left( 1 + \varepsilon_{r+1} W_{k-(r+1)} \right) \mathrm{d}\mu_\beta. \end{split}$$

First we bound using the Lyapunov property:

$$|\nabla^i_x\nabla^{r+1-i}_yh|^2L^*W_{k-(r+1)}\leqslant\sigma|\nabla^i_x\nabla^{r+1-i}_yh|^2.$$

The goal now is to check that for any order of derivative, we may chose  $\varepsilon_{r+1}$ ,  $(\omega_{i,r+1})$  and  $(\omega_{r+1})$  such that inequality (2.29) holds true with (r+1) instead of r. To this end, we will use the inequality (2.25),(2.26) and (2.28) on the  $\Gamma$ 's. The term of order r+2 of derivatives is:

$$-\gamma\beta^{-1}\int_{\mathcal{X}}\left(\sum_{i=0}^{r}\omega_{i,r+1}|\nabla_{x}^{i}\nabla_{y}^{r+2-i}h|^{2}+\frac{\omega_{r+1}}{2}|(\nabla_{x}^{r+1}\nabla_{y}-\xi\nabla_{x}^{r}\nabla_{y}^{2})h|^{2}(1+\varepsilon_{r+1}W_{k-(r+1)})\right)d\mu_{\beta},$$

which is already what we were looking for. This does not impose any condition on the coefficient, except the positivity of the  $(\omega)$ . The derivative of order r + 1 in x which was missing from the rank r, is:

$$\int_{\mathcal{X}} \left( \varepsilon_{r+1}\omega_{r+1}\sigma - \frac{\gamma\omega_{r+1}}{2} (1 + \varepsilon_{r+1}W_{k-(r+1)}) + \eta_r \omega_{r,r+1} (1 + \varepsilon_{r+1}W_{k-(r+1)}) \right) \\ + \left| \left( \nabla_x^{r+1} - \xi \nabla_x^r \nabla_y \right) h \right|^2 \mathrm{d}\mu_{\beta}.$$

We choose  $\eta_r = \frac{\gamma \omega_{r+1}}{8 \omega_{r,r+1}}$ , and  $0 < \varepsilon_{r+1} < \frac{\gamma}{8\sigma}$ , so that we are left with:

$$-\frac{\gamma\omega_{r+1}}{4}\int_{\mathcal{X}}|\left(\nabla_x^{r+1}-\xi\nabla_x^r\nabla_y\right)h|^2\left(1+\varepsilon_{r+1}W_{k-(r+1)}\right)\mathrm{d}\mu_{\beta}$$

For the other terms of order r+1, fix some  $i \in [0, r]$ . The term in  $|\nabla_x^i \nabla_y^{r+1-i} h|^2$  is:

$$\int_{\mathcal{X}} \left( -q_r W_{k-r} + \sigma \varepsilon_{r+1} \omega_{i,r+1} + \left( \frac{1}{(4\gamma)} + \theta_{i+1,r+1-(i+1)} V \right) \right) \left( 1 + \varepsilon_{r+1} W_{k-(r+1)} \right) \left| \nabla_x^i \nabla_y^j h \right|^2 \mathrm{d}\mu_{\beta}.$$

Using that  $VW_{k-(r+1)} \leq W_{k-r}(1+\lambda)$ , we may choose the  $(\omega_{i,r+1})$  small enough so that this last quantity is less than:

$$-\frac{q_r}{2}\int_{\mathcal{X}} W_{k-r} |\nabla_x^i \nabla_y^j h|^2 \mathrm{d}\mu_\beta.$$

The lower order terms are treated in the same way, and this concludes the induction and the proof.  $\hfill \Box$ 

Proof of Theorem 2.3.1. Instead of trying to justify the derivative (2.24), we apply a semi-group argument. Our goal is to apply Lumer-Phillips theorem to the operator  $L + (q_k/2)I$ , where Idenotes the identity operator of  $mH^k$ . This theorem can be stated as follows: an operator Aon a Hilbert space generates a contraction semi-group if and only if it is maximally dissipative, see [169, Chapter IX, p.250]. Fix  $0 < \bar{b} < \beta/(2d)$  and  $k \in \mathbb{N}$ .  $mH^k$  is a Hilbert space, with scalar product

$$\langle f,g \rangle_{mH^k}^2 = \int_{\mathcal{X}} fg W_k d\mu_\beta$$

$$+ \int_{\mathcal{X}} \sum_{p=1}^k \left( \sum_{i=0}^{p-1} \omega_{i,p} \nabla_x^i \nabla_y^{p-i} f \cdot \nabla_x^i \nabla_y^{p-i} g + \omega_{p,p} \left( \nabla_x^p - \xi \nabla_x^{p-1} \nabla_y \right) f \cdot \left( \nabla_x^p - \xi \nabla_x^{p-1} \nabla_y \right) g \right)$$

$$(1 + \varepsilon_p W_{k-p}) d\mu_\beta.$$

Lemma 2.3.2 and a density argument yield that the operator  $L + (q_k/2)I$ , is dissipative:

$$\forall h \in D(L), \left\langle (L + (q_k/2)I)h, h \right\rangle_{mH^k} \leqslant 0,$$

where D(L) denote the domain of L in  $mH^k$  defined by:

$$f \in \mathcal{D}(L), g = Lf \Leftrightarrow f \in mH^k, \lim_{t \to 0} \left\| \frac{P_t f - f}{t} - g \right\|_{mH^k} = 0$$

We are left to show that L + qI, for some  $q < q_k/2$ , is surjective. Fix  $q < q_k/2$ . Thanks to Lemma 2.3.2, we have that

$$\Lambda: (f,g) \mapsto \langle -(L+qI)f,g \rangle_{mH^k}$$

is coercive and continuous from  $(mH^k)^2$  to  $\mathbb{R}$ , as long as  $k \ge 2$ . Hence, we may apply Lax-

Milgram theorem to get that for all  $g \in mH^k$ , there exists  $f \in mH^k$  such that for all  $h \in mH^k$ , we have:

$$\left\langle -(L+qI)f,h\right\rangle _{mH^{k}}=\left\langle -g,h\right\rangle _{mH^{k}},$$

which implies that f is a solution to the equation

$$(L+qI)f = g,$$

and  $L+(q_k/2)I$  is maximally dissipative. Lumer-Phillips theorem then yields that the semi-group generated by  $L+q_k/2$  is a contraction on  $mH^k$ : for all  $f \in mH^k$ 

$$||e^{q_k t/2} P_t f||_{mH^k} \leq ||f||_{mH^k}.$$

For all  $f \in \mathcal{P}(U)$ , we may fix  $\overline{b} > 0$  small enough so that  $f \in mH^k$ , and this concludes the proof.

#### 2.3.3 Proof of Theorem 2.2.1

The proof of Theorem 2.2.1, based on Theorem 2.3.1, uses the Lyapunov structure of V.

Proof of theorem 2.2.1. Fix some  $f \in \mathcal{P}(U)$ . Thanks to Proposition 2.3.3, we may fix the parameter  $\bar{b}$  from the Lyapunov function such that  $||f||_{mH^k} < \infty$ . For all  $k \in \mathbb{N}$ , we can apply Theorem 2.3.1 to get some C > 0 such that:

$$\|P_t f\|_{mH^k} \leqslant C e^{-q_k t}.$$

Fix  $0 < b < \beta$ ,  $C, \varepsilon > 0$  such that  $e^{(\beta - (1 - \varepsilon)b)H} \leq CV_0$ , for some C > 0. Thanks to Assumption 2.2.3, we may write for all  $\alpha \in \mathbb{N}^{2d}$ :

$$\begin{split} \int_{\mathcal{X}} \left| \partial^{\alpha} \left( P_t f e^{-bH} \right) \right|^2 &\leq C \sum_{|\sigma| \leq |\alpha|} \int_{\mathcal{X}} \left| \partial^{\sigma} P_t f \right|^2 \left| \partial^{\alpha - \sigma} e^{-bH} \right|^2 \\ &\leq C' \sum_{|\sigma| \leq |\alpha|} \int_{\mathcal{X}} \left| \partial^{\sigma} P_t f \right|^2 e^{-(1 - \varepsilon)bH} \\ &= C' \sum_{|\sigma| \leq |\alpha|} \int_{\mathcal{X}} \left| \partial^{\sigma} P_t f \right|^2 e^{(\beta - (1 - \varepsilon)b)H} e^{-\beta H} \\ &\leq C'' \|P_t f\|_{mH^{|\alpha|}} \\ &\leq C'' e^{-q_{|\alpha|}t}. \end{split}$$

We conclude with Sobolev embedding.

### 2.4 Weak error expansion

The main ingredient in the proof of a Talay-Tubaro expansion for a numerical scheme (or even simply weak convergence) are the estimates given in Theorem 2.2.1. It allows to control the error (in law) made by each step of the numerical scheme, uniformly in time. Hence we are now able to prove Theorem 2.2.2 and Theorem 2.2.3. This section is divided into two parts. The first one proves the existence of a Lyapunov function of order  $e^{bH}$ , for all  $0 < b < \beta$ , as well as Theorem 2.2.2, about finite time expansion of the numerical scheme (2.4). The second part shows
the existence of a unique stationary measure for the numerical scheme, as well as Theorem 2.2.3 for the Talay-Tubaro expansion of this stationary measure.

#### 2.4.1 Finite-time expansion for the numerical scheme with rejection

The reason why we are interested in the scheme (2.4) instead of the scheme (2.7) is Lemma 2.4.2. To bound the error made by the numerical scheme, and hence to prove Theorem 2.2.2, we need uniform in  $n \in \mathbb{N}$  and  $\delta > 0$  integrability. To get this integrability, we show that the Lyapunov structure of the continuous process described in [82] still holds after discretization. This comes from the rejection mechanism that prevents the scheme from seeing the part of space where the gradient of the potential and its derivatives are too big compared to  $\delta$ , which is a neighborhood of the singularities and of infinity. Recall the definition of  $\mathcal{H}_{\delta,\ell}$  from (2.5), and denote  $\zeta = 4d\gamma\beta^{-1}$ . To prove Lemma 2.4.2, we first need to introduce a regularized version of the potential  $\hat{U}$  and a modified kinetic energy  $\hat{W}$ , along with a numerical scheme corresponding to the counterpart of the scheme (2.7) for the new Hamiltonian  $\hat{H}(x, y) = \hat{U}(x) + \hat{W}(y)$ . A comparison between this new numerical scheme with the previously considered one will lead to the result.

**Lemma 2.4.1.** Under Assumptions 2.2.1 and 2.2.2, there exist  $\delta_0, \ell_0 > 0$  such that for all  $0 < \delta < \delta_0$  and  $0 < \ell < \ell_0$ , there exist

$$\hat{U}, \hat{W} : \mathbb{R}^d \to \mathbb{R}_+$$

satisfying

- $\hat{U}$  and  $\hat{W}$  are bounded.
- $\hat{U} = U$ ,  $\hat{W} = |\cdot|^2/2$  on  $\mathcal{H}_{\delta,\ell}$ .
- For all  $(x, y) \in \mathcal{H}_{\delta, \ell}^c$

$$\hat{\phi}(x,y) = \hat{U}(x) + \hat{W}(y) + \zeta \frac{\nabla \hat{W} \cdot \nabla \hat{U}(x)}{1 + |\nabla \hat{U}(x)|^2} \ge 1/\delta^{\ell}.$$

• For all  $\alpha \in \mathbb{N}^{2d}$ ,  $1 \leq |\alpha|_1 \leq 4$ , there exists  $C_{\alpha} > 0$ ,  $\ell_{\alpha} < 1/2$  such that:

$$|\partial^{\alpha} \hat{U}|, |\partial^{\alpha} \hat{W}| \leqslant \frac{C_{\alpha}}{\delta^{\ell_{\alpha}}}.$$
(2.30)

*Proof.* For all  $0 < \ell < 1$ ,  $0 < \delta < 1$ , fix a smooth, increasing and concave function

$$g_{\delta,\ell}:\mathbb{R}_+\to\mathbb{R}_+$$

such that

$$g_{\delta,\ell}(t) = t, \forall t \in [0, 4\delta^{-\ell}]; \quad g'_{\delta,\ell}(t) \leqslant \frac{2(4\delta^{-\ell})^2}{t^2}, \quad |g^{(p)}_{\delta,\ell}(t)| \leqslant \frac{C}{t^{p+1}}, \quad \forall t \ge 4\delta^{-\ell}, \, p \in [\![2, 4]\!],$$

for some C > 0. Then define

$$\hat{U} = g_{\delta,\ell}(U), \quad \hat{W} = g_{\delta,\ell}(|\cdot|^2/2).$$

Let's show that  $\hat{U}$  and  $\hat{W}$  satisfy the desired conditions. First, the conditions on  $g_{\delta,\ell}$  yield that

$$g_{\delta,\ell} \leqslant 4\delta^{-\ell} + \int_{4\delta^{-\ell}}^{\infty} \frac{2(4\delta^{-\ell})^2}{t^2} \mathrm{d}t = 12\delta^{-\ell},$$

so that  $\hat{U}$  and  $\hat{W}$  are bounded. For all  $(x, y) \in \mathcal{H}_{\delta, \ell}$ , we have using Young's inequality:

$$\delta^{-\ell} \geqslant U(x) + \frac{|y|^2}{2} + \zeta \frac{y \cdot \nabla U(x)}{1 + |\nabla U(x)|^2} \geqslant \frac{1}{2} H(x, y) - \zeta^2,$$

so that  $H \leq 2\delta^{-\ell} + 2\zeta^2$ . As soon as  $\delta^{-\ell} \geq \zeta^2$ , we have that for all  $(x, y) \in \mathcal{H}_{\delta,\ell}$ ,  $U(x) \leq 4\delta^{-\ell}$ and  $|y|^2/2 < 4\delta^{-\ell}$ , yielding  $\hat{U}(x), \hat{W}(y) = U(x), |y|^2/2$  and the second point.

The definition of the modified kinetic energy implies that  $\nabla \hat{W}(y) = g'_{\delta,\ell}(|y|^2/2)y$ , and the conditions on  $g_{\delta,\ell}$  then yield

$$\sup_{y \in \mathbb{R}^d} |\nabla \hat{W}(y)| \leqslant \max\left(\sqrt{8\delta^{-\ell}}, \sup_{s \geqslant \sqrt{8\delta^{-\ell}}} g'(s)s\right) \leqslant 16\delta^{-\ell/2}.$$

To prove the third point of the lemma, we distinguish two cases: if  $(x, y) \notin \mathcal{H}_{\delta,\ell}$  satisfies  $U(x) < 4\delta^{-\ell}$  and  $|y|^2/2 < 4\delta^{-\ell}$ , then  $\hat{\phi}(x, y) = \phi(x, y) \ge \delta^{-\ell}$  by definition of  $\mathcal{H}_{\delta,\ell}$ . If  $U(x) > 4\delta^{-\ell}$  or  $|y|^2/2 > 4\delta^{-\ell}$ , we have

$$\hat{\phi}(x,y) \ge 4\delta^{-\ell} - 16\zeta\delta^{-\ell/2} \ge \delta^{-\ell}$$

for  $\delta$  small enough depending on  $\ell > 0$ . The last point is a direct consequence of the conditions imposed on  $g_{\delta,\ell}$  and the definitions of  $\hat{U}$  and  $\hat{W}$ .

We are now able to prove:

**Lemma 2.4.2.** Under Assumptions 2.2.1 and 2.2.2, there exist  $\delta_0, \ell_0 > 0$  such that for all  $0 < b < \beta$ , there exists  $V_b : \mathcal{X} \to \mathbb{R}_+$  such that:

- $\inf_{\mathcal{X}} V_b > 0.$
- For all  $\varepsilon > 0$ , there exists C > 0 such that

$$C^{-1}e^{(1-\varepsilon)bH} \leqslant V_b \leqslant Ce^{(1+\varepsilon)bH}.$$

• There exists  $\alpha, K > 0$  such that for all  $0 < \delta < \delta_0, 0 < \ell < \ell_0, (x, y) \in \mathcal{H}_{\delta, \ell}$ :

$$\mathbb{E}_{(x,y)}\left(V_b(\bar{X}_1,\bar{Y}_1)\right) \leqslant (1-\alpha\delta)V_b(x,y) + \alpha\delta K.$$

*Proof.* Fix  $0 < b < \beta$  and write:

$$V_b(x,y) = \exp\left(b\phi(x,y)\right), \quad \hat{V}_b(x,y) = \exp\left(b\hat{\phi}(x,y)\right),$$

where  $\phi$  was defined in (2.6) and  $\hat{\phi}$  in Lemma 2.4.1. It is different from the Lyapunov function of [82] in that we need in the denominator of  $\phi$  the term  $1 + |\nabla U(x)|^2$  instead of  $|\nabla U(x)|^2$ . However, the proof written in [82] can be immediately adapted to cover the case we consider, and it yields that there exists a compact set  $K_1 \subset \mathbb{R}^{2d}$  such that for all  $(x, y) \notin K_1$ :

$$\frac{LV_b(x,y)}{V_b(x,y)} \leqslant -c, \tag{2.31}$$

where L is the generator defined in (2.12). For any  $\varepsilon > 0$ , we have

$$\left|\zeta \frac{y \cdot \nabla U(x)}{1 + |\nabla U(x)|^2}\right| \leqslant \varepsilon |y|^2 / 2 + \frac{\zeta^2}{2\varepsilon} \times \frac{|\nabla U(x)|^2}{1 + |\nabla U(x)|^2} \leqslant \varepsilon H + C,$$

for some C > 0, and this implies the first two points. Now all that remains to do to show the third point is to show that inequality (2.31) remains true for the numerical scheme, as it stays in the region of space where  $U = O(\delta^{-\ell})$ . Recall the definition of  $\hat{U}$  and  $\hat{W}$  from Lemma 2.4.1. We introduce the following numerical scheme, counterpart of the scheme (2.7) for the Hamiltonian  $\hat{H}$ :

$$\begin{cases} X_{n+1} = X_n + \delta \nabla W(Y_n), \\ \hat{Y}_{n+1} = \hat{Y}_n - \delta \nabla \hat{U}(\hat{X}_n) - \delta \gamma \nabla \hat{W}(\hat{Y}_{n+1}) + \sqrt{2\gamma\beta^{-1}\delta}G_n, \end{cases}$$
(2.32)

where  $(G_n)$  is the same family of Gaussian random variable as in the definition of the scheme (2.4). On the event  $\{(\bar{X}_1, \bar{Y}_1) \neq (\hat{X}_1, \hat{Y}_1)\}$ , the step of  $(\bar{X}_1, \bar{Y}_1)$  was rejected,  $(\hat{X}_1, \hat{Y}_1) \notin \mathcal{H}_{\delta,\ell}$ , and we have using the third point of Lemma 2.4.1

$$\hat{V}_b(\bar{X}_1, \bar{Y}_1) = \hat{V}_b(x, y) \leqslant e^{b\delta^{-\ell}} \leqslant \hat{V}_b(\hat{X}_1, \hat{Y}_1).$$
(2.33)

Suppose we have K > 0, and  $0 < \alpha < 1$  both independent of  $\delta$  such that:

$$\mathbb{E}_{(x,y)}\left(\hat{V}_b(\hat{X}_1,\hat{Y}_1)\right) \leqslant (1-\alpha\delta)\hat{V}_b(x,y) + \alpha\delta K,$$

for all  $(x, y) \in \mathcal{H}_{\delta, \ell}$ . Then, using that for all  $(x, y) \in \mathcal{H}_{\delta, \ell}$ ,  $V_b(x, y) = \hat{V}_b(x, y)$ , and inequality (2.33), we would have for all  $(x, y) \in \mathcal{H}_{\delta, \ell}$ :

$$\mathbb{E}_{(x,y)}\left(V_b(\bar{X}_1,\bar{Y}_1)\right) = \mathbb{E}\left(\hat{V}_b(\bar{X}_1,\bar{Y}_1)\right) \leqslant \mathbb{E}\left(\hat{V}_b(\hat{X}_1,\hat{Y}_1)\right)$$
$$\leqslant (1-\alpha\delta)\hat{V}_b(x,y) + \alpha\delta K = (1-\alpha\delta)V_b(x,y) + \alpha\delta K.$$

Thus we only have to show that the Lyapunov structure holds (partially) true for the regularized numerical scheme (2.32). In the sequel of this proof, the notation  $o(\delta)$  will denote a quantity uniform in  $(x, y) \in \mathbb{R}^{2d}$ :

$$f_{\delta}(x,y) = o(\delta) \Leftrightarrow \lim_{\delta \to 0} \sup_{(x,y) \in \mathbb{R}^{2d}} \frac{|f_{\delta}(x,y)|}{\delta} = 0.$$

We have:

$$\begin{split} \hat{V}_b(\hat{X}_1, \hat{Y}_1) - \hat{V}_b(x, y) &= \hat{V}_b(x, y) \left( \hat{\phi}(\hat{X}_1, \hat{Y}_1) - \hat{\phi}(x, y) \right) \\ &+ \frac{1}{2} \hat{V}_b(x, y) \left( \hat{\phi}(\hat{X}_1, \hat{Y}_1) - \hat{\phi}(x, y) \right)^2 + \hat{V}_b(x, y) \sum_{p \ge 3} \frac{1}{p!} \left( \hat{\phi}(\hat{X}_1, \hat{Y}_1) - \hat{\phi}(x, y) \right)^p. \end{split}$$

A Taylor expansion on  $\hat{\phi}$  yields that there exists  $(\theta_1, \theta_2) \in \mathbb{R}^{2d}$ , random, such that:

$$\hat{\phi}(\hat{Z}_1) = \hat{\phi}(z) + (\hat{Z}_1 - z) \cdot \nabla \hat{\phi}(z) + \frac{1}{2} \nabla^2 \hat{\phi}(z) (\hat{Z}_1 - z)^{\circ 2} + \frac{1}{6} \nabla^3 \hat{\phi}(\theta_1, \theta_2) (\hat{Z}_1 - z)^{\circ 3}, \quad (2.34)$$

where z = (x, y) and  $\hat{Z}_1 = (\hat{X}_1, \hat{Y}_1)$ . Condition (2.30) yields that for  $p \in \{1, 2\}$ :

$$\mathbb{E}\left(\left(\nabla^3\hat{\phi}(\theta_1,\theta_2)(\hat{Z}_1-z)^{\circ 3}\right)^p\right)=o(\delta).$$

Moreover, we have

$$(\hat{Z}_1 - z) \cdot \nabla \hat{\phi}(z) = \delta \nabla \hat{W}(y) \cdot \nabla_x \hat{\phi} - \delta \nabla \hat{U}(x) \cdot \nabla_y \hat{\phi} - \delta \gamma \nabla \hat{W}(y) \cdot \nabla_y \hat{\phi} + \sqrt{2\gamma \beta^{-1} \delta} G_1 \cdot \nabla_y \hat{\phi}(x, y),$$
  
and

$$\frac{1}{2}\nabla^2 \hat{\phi}(z) (\hat{Z}_1 - z)^{\circ 2} = \gamma \beta^{-1} \delta G_1 \cdot \nabla_y^2 \hat{\phi}(x, y) G_1 + o(\delta).$$

Thus, taking the expectation of equation (2.34) yields

$$\begin{split} \mathbb{E}\left(\hat{\phi}(\hat{X}_{1},\hat{Y}_{1})-\hat{\phi}(x,y)\right) \\ &=\delta\nabla\hat{W}(y)\cdot\nabla_{x}\hat{\phi}-\delta\nabla\hat{U}(x)\cdot\nabla_{y}\hat{\phi}-\delta\gamma\nabla\hat{W}(y)\cdot\nabla_{y}\hat{\phi}+\delta\gamma\beta^{-1}\Delta_{y}\hat{\phi}+o(\delta), \end{split}$$

as well as

$$\mathbb{E}\left(\left(\hat{\phi}(\hat{X}_1,\hat{Y}_1)-\hat{\phi}(x,y)\right)^2\right)=2\delta\gamma\beta^{-1}|\nabla_y\phi|^2+o(\delta).$$

In the same spirit, the bounds (2.30) and equation (2.34) yield:

$$\mathbb{E}\left(\left(\hat{\phi}(\hat{X}_{1},\hat{Y}_{1})-\hat{\phi}(x,y)\right)^{p}\right) \leqslant \sum_{\substack{k=0\\k \text{ odd}}}^{p} \binom{p}{k} (C\sqrt{\delta})^{k} \frac{k!}{2^{k/2}(k/2)!} (C\delta^{1-1/\ell})^{p-k}$$
$$\leqslant C^{p}\sqrt{\delta}^{p} p! \sum_{\substack{k=0\\k \text{ odd}}}^{p} \frac{1}{(p-k)!(k/2)!2^{k/2}}$$
$$\leqslant C(C\delta)^{p/2} p!.$$

Hence for  $\delta < C^{-1}$ :

$$\mathbb{E}\left(\sum_{p \ge 3} \frac{1}{p!} \left(\hat{\phi}(\hat{X}_1, \hat{Y}_1) - \hat{\phi}(x, y)\right)^p\right) \leqslant C \frac{(C\delta)^{3/2}}{1 - C\delta} = o(\delta).$$

Finally, using that  $\hat{U} = U$  and  $\hat{W} = |\cdot|^2/2$  on  $\mathcal{H}_{\delta,\ell}$ , we get that for all  $(x, y) \in \mathcal{H}_{\delta,\ell}$ :

$$\mathbb{E}_{(x,y)}\left(\hat{V}_b(\hat{X}_1,\hat{Y}_1)\right) - \hat{V}_b(x,y) = \hat{V}_b(x,y)\left(\delta \frac{LV_b(x,y)}{V_b(x,y)} + o(\delta)\right),$$
(2.35)

where  $o(\delta)$  is uniform in (x, y). Hence, from equation (2.31), we have  $\delta_0 > 0$  such that for all  $(x, y) \in \mathcal{H}_{\delta, \ell} \setminus K_1, \, \delta < \delta_0$ :

$$\mathbb{E}_{(x,y)}\left(\hat{V}_b(\hat{X}_1,\hat{Y}_1)\right) \leqslant (1-\alpha\delta)V_b(x,y),$$

for some  $\alpha > 0$ . The map  $LV_b/V_b$  is continuous on the compact set  $K_1$ , hence is bounded, and

inequality (2.35) yields that there exists C > 0 such that for  $(x, y) \in K_1$ :

$$\mathbb{E}_{(x,y)}\left(\hat{V}_b(\hat{X}_1,\hat{Y}_1)\right) \leqslant V_b(x,y) + C\delta$$

which concludes the proof.

This Lyapunov structure has an immediate corollary: the numerical scheme will stay away from the singularities uniformly on  $\delta$ , even though its forbidden area is a level set depending on  $\delta$ .

**Corollary 2.4.1.** Under Assumptions 2.2.1 and 2.2.2, there exist  $\delta_0, \ell_0 > 0$  such that for all  $0 < b < \beta, 0 < \ell < \ell_0, z \in \mathcal{X}$ :

$$\sup_{0<\delta<\delta_0}\sup_{n\in\mathbb{N}}\mathbb{E}_z\left(e^{bH(\bar{Z}_n)}\right)<\infty.$$

Moreover, for all  $z \in \mathcal{X}$ , there exists C, c > 0 such that for all a > 0,  $0 < \delta < \delta_0$ ,  $0 < \ell < \ell_0$  and  $n \in \mathbb{N}$ :

$$\mathbb{P}_z\left(H(\bar{Z}_n) \geqslant a\right) \leqslant Ce^{-ca}.$$

*Proof.* Fix  $b < b' < \beta$ . The second Lyapunov property of Lemma 2.4.2 implies that:

$$\mathbb{E}_{z}\left(e^{bH\left(\bar{X}_{n},\bar{Y}_{n}\right)}\right) \leqslant C\mathbb{E}_{z}\left(V_{b'}\left(\bar{X}_{n},\bar{Y}_{n}\right)\right),$$

for some C > 0, whereas an induction argument using the third property yields

$$\mathbb{E}_{z}\left(V_{b'}\left(\bar{X}_{n}, \bar{Y}_{n}\right)\right) \leqslant \max\left(V_{b'}(x, y), K\right).$$

Both inequality combined then gives the first point of the Corollary. The second point is a direct application of Markov inequality.  $\hfill \Box$ 

The next lemma will allow us to do series expansion of  $P_t(\overline{Z}_n)$  to prove Theorem 2.2.2. We write for  $z_1, z_2 \in \mathbb{R}^{2d}$ ,  $[z_1, z_2] = \{tz_1 + (1-t)z_2, t \in [0,1]\}$ .

**Lemma 2.4.3.** For all  $\kappa > 0$ , there exists  $C, \delta_0 > 0$  such that for all  $0 < \delta < \delta_0$ ,  $z_1, z_2 \in \{H \leq \kappa \ln 1/\delta\}$ , and  $|z_1 - z_2| \leq \delta^{\frac{2}{3}}$ , all  $z \in [z_1, z_2]$ :

$$H(z) \leqslant H(z_1) + C$$

*Proof.* Fix  $z_1, z_2 \in \{H \leq \kappa \ln 1/\delta\}$ , such that  $|z_1 - z_2| \leq \delta^{\frac{2}{3}}$ . We have that  $[z_1, z_2] \subset B(z_1, \delta^{\frac{2}{3}})$ . We first show that

$$B(z_1,\delta^{\frac{2}{3}}) \subset \{H \leqslant \kappa' \ln 1/\delta\},\$$

for some  $\kappa' > 0$ . To this end, fix  $z \in B(z_1, \delta^{\frac{2}{3}})$ , and write for  $0 \leq s \leq 1$ :

$$\varphi(s) = U(z_1 + s(z - z_1)).$$

Thanks to Assumption 2.2.1, we have that  $\varphi$  is finite and smooth for  $s < s_0$  small enough. We show at the same time that we may have  $s_0 = 1$  and that  $\varphi(1) \leq \kappa' \ln(1/\delta)$ . We have:

$$\varphi'(s) = \nabla U(z_1 + s(z - z_1)) \cdot (z - z_1)$$

$$\leq |\nabla U(z_1 + s(z - z_1))||(z - z_1)| \leq \delta^{\frac{2}{3}} \sqrt{c_0 \varphi^{2 + \frac{2}{\eta_0}} + d_0} \leq \delta^{\frac{2}{3}} \sqrt{c_0} \left(\varphi + \left(\frac{d_0}{c_0}\right)^{\frac{\eta_0}{2 + 2\eta_0}}\right)^{1 + \frac{1}{\eta_0}}.$$

Hence there is  $c_1, c_2 > 0$  independent of  $z_1, z$  and  $\delta$  such that for  $s \leq s_0$ :

$$\frac{\varphi'(s)}{\left(\varphi+c_2\right)^{1+\frac{1}{\eta_0}}} \leqslant c_1 \delta^{\frac{2}{3}}.$$

Integrating this last inequation yields:

$$\frac{\eta_0}{\left(\varphi(s)+c_2\right)^{1/\eta_0}} \geqslant \frac{\eta_0}{\left(\varphi(0)+c_2\right)^{1/\eta_0}} - c_1 \delta^{\frac{2}{3}} s.$$

Whether  $\eta_0 < 0$  or  $\eta_0 > 1$ , because  $\varphi(0) \leq \kappa \ln(1/\delta)$ , we have that there exists  $\delta_0, \kappa' > 0$  such that for all  $0 < \delta < \delta_0, \varphi(1) \leq \kappa' \ln(1/\delta)$ . Hence using Assumption 2.2.2 again, we have that

$$\forall z \in \mathcal{B}(z_1, \delta^{\frac{2}{3}}), |\nabla U(z)| \leq \kappa'' (\ln(1/\delta))^{1+\frac{1}{\eta_0}},$$

for some  $\kappa'' > 0$ . Using a Taylor expansion on  $\varphi$ , we get that for all  $z_1 \in \{H \leq \kappa \ln 1/\delta\}$  and  $z \in B(z_1, \delta^{\frac{2}{3}})$ :

$$U(z) \leq U(z_1) + \kappa'' \delta^{\frac{2}{3}} (\ln 1/\delta)^{\frac{1}{\eta_0}}$$

and this concludes the proof with  $C = \sup_{0 < \delta < \delta_0} \kappa'' \delta^{\frac{2}{3}} (\ln 1/\delta)^{\frac{1}{\eta_0}}$ .

Proof of theorem 2.2.2. As in [158], we only prove the case k = 1 for simplicity, since for k > 1, the proof is the same. Fix some function  $f \in \mathcal{P}(U)$ , and write  $u(t, x, y) = P_t f(x, y)$ . Recall that u is solution to:

$$\begin{cases} \partial_t u = Lu, \\ u(t = 0, \cdot) = f, \end{cases}$$

$$(2.36)$$

where L is the generator defined by (2.12). For the sequel of the proof, fix t > 0. For all  $n \in \mathbb{N}$ ,  $\delta > 0$  is such that  $n\delta = t$ . We can write the total error made by the numerical scheme as:

$$\mathbb{E}_{z}\left(f\left(\bar{Z}_{n}\right)\right) - \mathbb{E}_{z}\left(f\left(Z_{t}\right)\right) = \mathbb{E}_{z}\left(u\left(0,\bar{Z}_{n}\right) - u\left(n\delta,z\right)\right)$$
$$= \sum_{p=0}^{n-1} \mathbb{E}_{z}\left(u\left(p\delta,\bar{Z}_{n-p}\right) - u\left((p+1)\delta,\bar{Z}_{n-(p+1)}\right)\right).$$

To make a series expansion of the expectations in the sum, we need to distinguish between three cases. Fix some  $\kappa > 0$  to be determined later,  $n, p \in \mathbb{N}$ ,  $p \leq n$ , and denote:

$$\mathcal{A}_{1} = \left\{ Z_{n-p}, Z_{n-(p+1)} \in \left\{ H \leqslant \kappa \ln(1/\delta) \right\}, |Z_{n-p} - Z_{n-(p+1)}| \leqslant \delta^{2/3} \right\}, \\ \mathcal{A}_{2} = \left\{ Z_{n-p}, Z_{n-(p+1)} \in \left\{ H \leqslant \kappa \ln(1/\delta) \right\}, |Z_{n-p} - Z_{n-(p+1)}| \geqslant \delta^{2/3} \right\}, \\ \mathcal{A}_{3} = \left\{ Z_{n-p} \notin \left\{ H \leqslant \kappa \ln(1/\delta) \right\} \right\} \cup \left\{ Z_{n-(p+1)} \notin \left\{ H \leqslant \kappa \ln(1/\delta) \right\} \right\}.$$

The event  $\mathcal{A}_1$  is the event on which we may do a series expansion of

$$u\left(p\delta,\bar{Z}_{n-p}\right) - u\left((p+1)\delta,\bar{Z}_{n-(p+1)}\right),\,$$

thanks to Lemma 2.4.3. Let's first show that the two other event have very low probability as  $\delta$  goes to 0. Theorem 2.2.1 yields the existence of some  $0 < b < \beta$ , C, q > 0 such that:

$$|u - \int_{\mathcal{X}} f \mathrm{d}\mu_{\beta}| \leqslant C e^{-qt} e^{bH}.$$

Fix  $q_1, q_2 \ge 1$  such that

$$1/q_1 + 1/q_2 = 1, \qquad bq_1 < \beta.$$

Using Holder inequality we get:

$$\begin{aligned} \left| \mathbb{E}_{z} \left( \left( u \left( p\delta, \bar{Z}_{n-p} \right) - u \left( (p+1)\delta, \bar{Z}_{n-(p+1)} \right) \right) \mathbb{1}_{\mathcal{A}_{2}} \right) \right| \\ & \leq C e^{-qp\delta} \left( \left( \mathbb{E}_{z} \left( e^{bq_{1}H\left( \bar{Z}_{n-p} \right)} \right) \right)^{1/q_{1}} + \left( \mathbb{E}_{z} \left( e^{bq_{1}\bar{H}\left( Z_{n-(p+1)} \right)} \right) \right)^{1/q_{1}} \right) \left( \mathbb{P} \left( \mathcal{A}_{2} \right) \right)^{1/q_{2}}. \end{aligned}$$

On the event  $\{H(\overline{Z}_{n-(p+1)}) \leq \kappa \ln(1/\delta)\}$ , there is some c > 0 such that

$$\left|\nabla U\left(\bar{X}_{n-(p+1)}\right)\right| + \left|Y_{n-(p+1)}\right| \leqslant C\left(\ln(1/\delta)\right)^{\epsilon}$$

and

$$\left|\bar{Z}_{n-k} - \bar{Z}_{n-(k+1)}\right| \leq C\delta \ln(1/\delta)^c + \sqrt{\delta} |G|$$

where G is a standard Gaussian random variable. Hence for  $\delta$  small enough, there is c, C > 0 such that:

$$\mathbb{P}(\mathcal{A}_2) \leqslant \mathbb{P}\left(G \ge c\delta^{1/6}\right) \leqslant Ce^{-c\delta^{1/6}},$$

and finally

$$\mathbb{E}_{z}\left(\left(u\left(p\delta,\bar{Z}_{n-p}\right)-u\left((p+1)\delta,\bar{Z}_{n-(p+1)}\right)\right)\mathbb{1}_{\mathcal{A}_{2}}\right)\right|\leqslant Ce^{-c\delta^{1/6}}$$

Next, using Holder's inequality again:

$$\mathbb{E}_{z}\left(\left(u\left(p\delta,\bar{Z}_{n-p}\right)-u\left((p+1)\delta,\bar{Z}_{n-(p+1)}\right)\right)\mathbb{1}_{\mathcal{A}_{3}}\right)$$
  
$$\leq Ce^{-qp\delta}\left(\left(\mathbb{E}\left(e^{bq_{1}H\left(\bar{Z}_{n-p}\right)}\right)\right)^{1/q_{1}}+\left(\mathbb{E}\left(e^{bq_{1}\bar{H}\left(Z_{n-(p+1)}\right)}\right)\right)^{1/q_{1}}\right)\left(\mathbb{P}\left(\mathcal{A}_{3}\right)\right)^{1/q_{2}}.$$

Fix  $b' < \beta$ . We have, using Markov inequality:

$$\mathbb{P}(\mathcal{A}_3) \leqslant 2\sup_n \mathbb{P}\left(e^{b'H(Z_n)} \geqslant e^{-b'\kappa\ln(1/\delta)}\right) \leqslant 2\sup_n \mathbb{E}(e^{b'H(Z_n)})\delta^{b'\kappa} = C\delta^{b'\kappa}.$$

We can now fix  $\kappa > 3q_2/b'$  to get that:

$$\mathbb{E}_{z}\left(\left(u\left(p\delta,\bar{Z}_{n-p}\right)-u\left((p+1)\delta,\bar{Z}_{n-(p+1)}\right)\right)\mathbb{1}_{\mathcal{A}_{3}}\right)=O(\delta^{3}).$$

On the event  $\mathcal{A}_1$ , we may use Lemma 2.4.3 to make a series expansion of u. For  $0 \leq s \leq 1$ , write

$$\varphi(s) = u\left((p+1-s)\delta, \bar{Z}_{n-(p+1)} + s(\bar{Z}_{n-p} - \bar{Z}_{n-(p+1)})\right).$$

Then:

$$(\varphi(1)-\varphi(0))\mathbb{1}_{\mathcal{A}_1} = \left(\sum_{r=1}^5 \frac{\varphi^{(r)}(0)}{r!} + \frac{\varphi^{(6)}(\theta)}{6!}\right)\mathbb{1}_{\mathcal{A}_1},$$

for some  $0 \leq \theta \leq 1$ . Theorem 2.2.1 yields that there is C > 0,  $0 < b < \beta$  such that:

$$|\varphi^{(6)}(\theta)| \mathbb{1}_{\mathcal{A}_1} \leqslant C\left(\delta^3 G^6 + \delta^{7/2} G^5 \ln(1/\delta) + \dots + \delta^6 (\ln(1/\delta))^6\right) e^{bH(\bar{Z})} \mathbb{1}_{\mathcal{A}_1}$$

for some  $\bar{Z} \in [\bar{Z}_{n-p}, \bar{Z}_{n-(p+1)}]$ . However, Lemma 2.4.3 also yields that

$$H(\bar{Z})\mathbb{1}_{\mathcal{A}_1} \leqslant H\left(\bar{Z}_{n-p}\right) + C$$

Hence, Holder's inequality yields:

$$\mathbb{E}\left(|\varphi^{(6)}(\theta)|\mathbb{1}_{\mathcal{A}_1}\right) \leqslant C\delta^3 \sup_n \mathbb{E}\left(e^{q_1 b H\left(\bar{Z}_n\right)}\right)^{\frac{1}{q_1}} = C'\delta^3$$

Using Holder's inequality as in the bound for the event  $\mathcal{A}_2$  and  $\mathcal{A}_3$ , we get that:

$$\mathbb{E}\left(\mathbb{1}_{\mathcal{A}_1}\sum_{r=1}^5\frac{\varphi^{(r)}(0)}{r!}\right) = \mathbb{E}\left(\sum_{r=1}^5\frac{\varphi^{(r)}(0)}{r!}\right) + O(\delta^3) = \delta^2\mathbb{E}\left(\psi\left((p+1)\delta, \bar{Z}_{n-(p+1)}\right)\right) + O(\delta^3),$$

where

$$\begin{split} \psi(s,z) &= (\nabla U + \gamma y) \cdot (\nabla_y/2 - \nabla_x) \, u + \frac{1}{2} \partial_t^2 u - \frac{1}{2} y \cdot \nabla_x \partial_t u + \frac{1}{2} \left( \nabla U + \gamma y \right) \cdot \nabla_y^2 y \left( \nabla U + \gamma y \right) \\ &- \frac{1}{2} y \cdot \nabla_{x,y}^2 \left( \nabla U + \gamma y \right) + \gamma \beta^{-1} \sum_{i=1}^d \partial_{x_i} \partial_{y_i} u - \frac{2}{3} \gamma \beta^{-1} \partial_t \Delta_y u \\ &- \frac{7}{12} \gamma \beta^{-1} \left( \nabla U + \gamma y \right) \cdot \nabla_y \Delta_y u + \frac{1}{3} \gamma \beta^{-1} y \cdot \nabla_x \Delta_y u \\ &+ \frac{1}{2} \left( \gamma \beta^{-1} \right)^2 \sum_{i=1}^d \partial_{y_i}^4 u + \frac{1}{6} \left( \gamma \beta^{-1} \right)^2 \sum_{i \neq j=1}^d \partial_{y_j}^2 \partial_{y_j}^2 u. \end{split}$$

The expression of  $\psi$  is found using the fact that u solves (2.36). Finally, we have that:

$$\mathbb{E}_{z}\left(f\left(\bar{Z}_{n}\right)\right) - \mathbb{E}_{z}\left(f\left(Z_{t}\right)\right) = \delta^{2}\mathbb{E}\left(\sum_{p=1}^{n}\psi\left(p\delta,\bar{Z}_{n-p}\right)\right) + O(\delta^{2}).$$

The rest follows as in the proof of Talay and Tubaro [158], which we write for the sake of completeness. As  $\psi$  is a function of the derivative of u, Theorem 2.2.1 yields that:

$$\left|\mathbb{E}_{z}\left(f\left(\bar{Z}_{n}\right)\right) - \mathbb{E}_{z}\left(f\left(Z_{t}\right)\right)\right| \leqslant C\delta^{2}\left(\sum_{p=1}^{n}e^{-qp\delta}\mathbb{E}_{z}\left(e^{bH(\bar{Z}_{n-p})}\right) + 1\right) \leqslant C\delta^{2}\left(\frac{1}{\delta} + 1\right) \leqslant C\delta.$$

$$(2.3)$$

(2.37) The function  $s \to \mathbb{E}_z (\psi(z, Z_{t-s}))$  is smooth and is bounded, hence the weak convergence (2.37) and classical Riemann summation results yield that:

$$\left| \delta \mathbb{E} \left( \sum_{p=1}^{n} \psi \left( p \delta, \bar{Z}_{n-p} \right) \right) - \int_{0}^{t} \mathbb{E}_{z} \left( \psi(z, Z_{t-s}) \, \mathrm{d}s \right| = O(\delta),$$

which concludes the proof.

## 2.4.2 Expansion for the invariant measure of the numerical scheme with rejection

To establish an expansion of the invariant measure of the numerical scheme, we first need to prove its existence, and ergodicity of the scheme. For this purpose, we define a metric on  $\mathcal{M}^1(\mathbb{R}^{2d})$ . Fix some  $0 < b < \beta$ , and write for  $z_1, z_2 \in \mathbb{R}^{2d}$ :

$$\mathbf{d}(z_1, z_2) = \mathbb{1}_{z_1 \neq z_2} \left( 1 + V_b(z_1) + V_b(z_2) \right),$$

where  $V_b$  is the Lyapunov function from Lemma 2.4.2. For any two probability measures  $\mu, \nu$ , we call  $(Z_1, Z_2)$  a coupling of  $\mu$  and  $\nu$  if the law of  $Z_1$  (resp.  $Z_2$ ) is  $\mu$  (resp.  $\nu$ ). From the distance **d** on  $\mathbb{R}^{2d}$ , we define the corresponding Kantorovich distance on  $\mathcal{M}^1(\mathbb{R}^{2d})$  by:

$$W_{\mathbf{d}}(\mu,\nu) = \inf \left\{ \mathbb{E} \left( \mathbf{d}(Z_1, Z_2) \right), (Z_1, Z_2) \text{ coupling of } \mu \text{ and } \nu \right\}.$$

This distance makes  $\mathcal{M}^1(\mathbb{R}^{2d})$  complete. Hence, if we show that the map  $\mathcal{L}aw(z) \mapsto \mathcal{L}aw(\bar{Z}_k)$ , where  $\bar{Z}_k$  is the numerical scheme given by (2.4) with initial condition z, for some  $k \in \mathbb{N}$ , is a contraction, then this would imply the existence of a unique stationary measure, as well as exponentially fast convergence towards it for this Kantorovitch distance. It can be shown that convergence for such a distance would imply ergodicity of the numerical scheme, in the sense of the convergence of the average

$$\lim_{n \to \infty} \frac{1}{n} \sum_{k=1}^{n} \mathbb{E}_{(x,y)} \left( f\left(\bar{X}_k, \bar{Y}_k\right) \right) = \mu(f).$$
(2.38)

Write  $K_2 = \{H \leq K\}$ , where K is the constant from Lemma 2.4.2. Because  $V_b$  is a Lyapunov function, we only need to show that there exists c > 0,  $k \in \mathbb{N}$ , such that for all  $z \in K_2$ , we have the following Doeblin condition:

$$\mathbb{P}_{z}\left(\bar{Z}_{k}\in\cdot\right)\geqslant c\lambda\left(\cdot\cap K_{2}\right),\tag{2.39}$$

where  $\lambda$  stands for the Lebesgue measure, see for example [52, Theorem 24]. In other words, we want to show that the numerical scheme creates density.

**Lemma 2.4.4.** Under Assumptions 2.2.1 and 2.2.2, there exists  $\delta_0, \ell_0 > 0$  such that for all  $0 < \delta < \delta_0, 0 < \ell < \ell_0$ , the numerical scheme (2.4) admits a unique stationary measure  $\mu_{\beta,\delta} \in \mathcal{M}^1(\mathbb{R}^{2d})$ , and for all  $(x, y) \in \mathcal{H}_{\delta,\ell}$  and  $f : \mathcal{H}_{\delta,\ell} \to \mathbb{R}$  continuous:

$$\lim_{n \to \infty} \frac{1}{n} \sum_{k=1}^{n} \mathbb{E}_{(x,y)} \left( f\left( \bar{X}_{k}, \bar{Y}_{k} \right) \right) = \mu_{\beta,\delta} \left( f \right).$$

*Proof.* As explained before, it suffices to show the so-called Doeblin condition (2.39). To achieve such inequality, we first begin by showing, using a controllability argument, that this is locally true, in the sense that: there exits c > 0 such that for all  $z \in K_2$ ,  $z_0 \in B(z, \delta)$  and  $A \subset B(z, \delta)$ ,

$$\mathbb{P}_{z_0}\left(\overline{Z}_2 \in A\right) \geqslant c\lambda(A).$$

To this end, we want to show that there exists  $(g_0, g_1)$  such that if  $\sqrt{2\gamma\beta^{-1}\delta}(G_0, G_1) = (g_0, g_1)$ and  $\bar{Z}_0 = z_0$ , then  $\bar{Z}_2 = z_1$ . Since Gaussian random variables have a density with respect to the Lebesgue measure, then if  $z_1 \mapsto (g_0, g_1)$  is a diffeomorphism, the result will hold. Fix  $z_0, z_1 \in B(z, \delta)$  and write:

$$F(x,y) = (x + \delta (y - \delta \nabla U(x) - \delta \gamma y), y - \delta \nabla U(x) - \delta \gamma y),$$

and  $\mathbf{G}(g_0, g_1) = (\mathbf{G}_1(g_0, g_1), \mathbf{G}_2(g_0, g_1))$  where

$$\mathbf{G}_1(g_0, g_1) = \delta(2 - \delta\gamma)g_0 - \delta^2 \nabla U \left(x_0 + \delta \left(y_0 - \delta U(x_0) - \delta\gamma y_0 + g_0\right)\right) + \delta g_1,$$

and

$$\mathbf{G}_{2}(g_{0},g_{1}) = (1-\delta\gamma)g_{0} - \delta\nabla U \left(x_{0} + \delta \left(y_{0} - \delta U(x_{0}) - \delta\gamma y_{0} + g_{0}\right)\right) + g_{1}$$

Now, the goal is to show that there exists  $g_0, g_1 \in \mathbb{R}^d$  such that:

$$H\left(F(z_0) + (\delta g_0, g_0)\right) \leqslant \delta^{-\ell},\tag{2.40}$$

and

$$\mathbf{G}(g_0, g_1) = z_1 - z_0 + \left(\delta^2 (2 - \gamma \delta) \nabla U(x_0) - \delta (2 - 3\delta\gamma + (\delta\gamma)^2) y_0, \delta (1 - \gamma \delta) \nabla U(x_0) + (2\gamma \delta - (\gamma \delta)^2) y_0\right).$$
(2.41)

We have:

$$\mathbf{G} = \tilde{G} \left( I_d + \delta \bar{G} \right),$$

where  $\tilde{G}$  is the invertible matrix:

$$\tilde{G} = \begin{pmatrix} 2\delta I_d & \delta I_d \\ I_d & I_d \end{pmatrix}, \qquad \tilde{G}^{-1} = \begin{pmatrix} \delta^{-1}I_d & -I_d \\ -\delta^{-1}I_d & 2I_d \end{pmatrix},$$

and where  $\overline{G}$  and its derivative are bounded on  $B(0,\rho)$ , for all  $\rho > 0$ , uniformly with respect to  $\delta < \delta_0$  and  $z_0 \in K_2$ . Write  $h = I_d + \delta \overline{G}$ . The goal now is to show that for all  $\rho_2 > 0$ , there exists  $\rho_1 > 0$  such that h is a diffeomorphism from a neighborhood  $W \subset B(0,\rho_1)$  of 0 to  $B(h(0),\rho_2)$ , by following the proof of the local inverse theorem. To this end, for  $v, g \in \mathbb{R}^{2d}$ , write:

$$\phi_v(g) = g - (d_0 h)^{-1} (h(g) - v).$$

There exists  $\delta_0 > 0$  such that, for all  $z_0 \in K_2$  and  $0 < \delta < \delta_0$ , this function is well defined, i.e.  $d_0h$  is invertible. For all  $\rho_1 > 0$ , there exists  $\delta_0 > 0$  such that for all  $\delta < \delta_0$ ,  $g \in B(0, \rho_1)$ ,  $|||d_g\phi_v||| \leq 1/2$ . Hence for  $g, g' \in B_{\mathbb{R}^{2d}}(0, \rho_1)$ , we have

$$|\phi_v(g) - \phi_v(g')| \leq \frac{1}{2}|g - g'|.$$

For all  $\rho_2 > 0$ ,  $v \in B(h(0), \rho_2)$ , and  $\delta$  small enough we have that

$$|(d_0h)^{-1}(h(0)-v)| < 2\rho_2.$$

This implies that if  $\rho_1 > 4\rho_2$ , then for all  $g \in B(0, \rho_1)$ , we have  $\phi_v(g) \in B(0, \rho_1)$ :

$$|\phi_v(g)| \leq |\phi_v(g) - \phi_v(0)| + |\phi_v(0)| \leq \frac{1}{2}|g| + 2\rho_2 < \rho_1.$$

Hence we may apply Banach fixed point theorem to get that for all  $v \in B(0, \rho_2)$ , there exists  $g_v \in B(0, \rho_1)$  such that  $\phi_v(g_v) = g_v$ , or equivalently  $h(g_v) = v$ . The fact that  $v \mapsto g_v$  is smooth

is proved in the same way as in the proof of the inverse function theorem. Thus, for all  $\rho > 0$ , there exists  $\delta_0, r > 0$  such that for all  $z_0 \in K_2$ ,  $0 < \delta < \delta_0$ , there exists a bounded neighborhood  $W \subset B_{\mathbb{R}^{2d}}(0, r)$  of 0, such that G is a diffeomorphism from W to  $B_{\mathbb{R}^{2d}}(G(0, 0), \rho\delta)$ . We fix

$$\rho > 3 + 2 \sup_{(x,y) \in K_2} |y| + |\nabla U(x)|.$$

Write  $I(z_0, z_1)$  for the left hand side of Equation (2.41). For  $\delta$  small enough,  $I \in B(G(0, 0), \rho\delta)$ for all  $z_0, z_1 \in B(z, \delta)$ , so that Equation (2.41) has a solution  $G^{-1}(I(z_0, z_1))$ . The fact that  $W \subset B(0, r)$  yields condition (2.40) for  $\delta < \Delta/r$ , where  $\Delta$  is the distance between  $K_2$  and  $\mathcal{D}^c$ . Finally, because  $(G_0, G_1)$  has a positive density, for  $z_0 \in B(z, \delta)$  and  $A \subset B(z, \delta)$  we have:

$$\mathbb{P}_{z_0}\left(\bar{Z}_2 \in A\right) \ge \mathbb{P}\left((G_0, G_1) \in G^{-1} \circ I(\cdot, z_0)(A)\right)$$
$$\ge c \int_{G^{-1} \circ I(\cdot, z_0)(A)} \mathrm{d}z' = c \int_A |\det(\nabla G \circ I(z', z_0))|^{-1} \mathrm{d}z' \ge c'\lambda(A).$$

Now we are able to show condition (2.39). Fix  $\delta$  small enough so that the local Doeblin condition holds true. Since  $K_2$  is compact, there exists  $z_1, \dots, z_N$  such that  $K_2 = \bigcup_{i=1}^N B(z_i, \delta)$ . For all measurable  $A \subset K_2$ , we have that

$$\lambda(A) \leqslant \sum_{i=1}^{N} \lambda(A \cap B(z_i, \delta)).$$

Hence we may restrict ourselves to  $A \subset B(z_j, \delta)$ , for some  $j \in [\![1, N]\!]$ . Up to increasing the value of K, we may suppose that  $K_2$  is connected. In this case, there exists  $k \in \mathbb{N}$  and  $\varepsilon > 0$  such that for all  $z \in B(z_i, \delta)$  and  $A \subset B(z_j, \delta)$ , there exists a finite sequence  $z'_0, z'_1, \ldots, z'_k \in K_2$  satisfying the following condition:  $z'_0 = z, z'_k \in B(z_j, \delta)$ , and for all  $0 \leq p \leq k - 1$ , there is  $i_p$  such that  $B(z_p, \varepsilon), B(z_{p+1}, \varepsilon) \subset B(z_{i_p}, \delta)$ . Hence we have:

$$\mathbb{P}\left(Z_{2(k+1)} \in A\right) \ge \mathbb{P}\left(Z_{2p} \in B(z'_p, \varepsilon), 1 \le p \le k, Z_{2(k+1)} \in A\right) \ge (cw)^k \lambda(A),$$

where w is the volume of  $B_{\mathbb{R}^{2d}}(0,\varepsilon)$ , and this concludes the proof.

All function  $f \in \mathcal{P}(U)$  are continuous on  $\mathcal{H}_{\delta,\ell}$ , thus we do indeed have the convergence (2.38) for such function. To prove Theorem 2.2.3, we first need weak convergence of the stationary measure of the numerical scheme, for functions that depends on time.

**Lemma 2.4.5.** Let  $g \in C^{\infty}(\mathbb{R}_+ \times \mathbf{X}, \mathbb{R})$  be such that for all  $\alpha \in \mathbb{N}^{2d}$ , there exists  $C_{\alpha}, q_{\alpha} > 0$  and  $b_{\alpha} < \beta$  such that:

$$\left|\partial^{\alpha}g(t,z)\right| \leqslant C_{\alpha}e^{-q_{\alpha}t}e^{b_{\alpha}H(z)}$$

Under Assumption 2.2.1, 2.2.2 and 2.2.3, there exists C > 0 such that for  $\delta$  small enough:

$$|\mu_{\beta,\delta}(g(t,\cdot)) - \mu_{\beta}(g(t,\cdot))| \leq Ce^{-qt}\delta.$$

*Proof.* The proof is very close to the proof Theorem 2.2.2, hence we will omit some details. We fix g satisfying the assumptions of Lemma 2.4.5, t > 0, and we write  $u(s, z) = \mathbb{E}_z(g(t, Z_s))$ , where Z = (X, Y) is the continuous process (2.3).  $g(t, \cdot) \in \mathcal{P}(U)$ , and we have for any  $n \in \mathbb{N}$ :

$$\frac{1}{n}\sum_{k=1}^{n}\mathbb{E}_{z}\left(u\left(0,\bar{Z}_{k}\right)-u\left(k\delta,z\right)\right) = \frac{1}{n}\sum_{k=1}^{n}\sum_{p=0}^{k-1}\mathbb{E}_{z}\left(u\left(p\delta,\bar{Z}_{k-p}\right)-u\left((p+1)\delta,\bar{Z}_{k-(p+1)}\right)\right).$$
 (2.42)

Lemma 2.4.4 yields that:

$$\frac{1}{n}\sum_{k=1}^{n}\mathbb{E}_{z}\left(u\left(0,\bar{Z}_{k}\right)\right)=\frac{1}{n}\sum_{k=1}^{n}\mathbb{E}_{z}\left(g\left(t,\bar{Z}_{k}\right)\right)\underset{n\to\infty}{\rightarrow}\mu_{\beta,\delta}(g(t,\cdot))$$

The continuous process is also ergodic, as shown by Theorem 2.3.1. Hence we have:

$$\frac{1}{n}\sum_{k=1}^{n}\mathbb{E}_{z}\left(u\left(k\delta,z\right)\right) = \frac{1}{n}\sum_{k=1}^{n}P_{k\delta}g(t,\cdot)(z) \xrightarrow[n \to \infty]{} \mu_{\beta}(g(t,\cdot)).$$

As in the proof of Theorem 2.2.2, a Taylor expansion yields:

$$\left|\sum_{p=0}^{k-1} \mathbb{E}_z \left( u \left( p\delta, \bar{Z}_{k-p} \right) - u \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) \right| \leqslant C e^{-qt} \delta,$$

and letting n go to infinity in (2.42) concludes the proof.

Proof of Theorem 2.2.3. Fix  $f \in \mathcal{P}(U)$  and write  $u(t, z) = \mathbb{E}_z(f(Z_t))$ , where Z = (X, Y) is the continuous process (2.3). We write a Taylor expansion of higher order than in the proof of Lemma 2.4.5 to get:

$$\sum_{p=0}^{k-1} \mathbb{E}_z \left( u \left( p\delta, \bar{Z}_{k-p} \right) - u \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) = \delta^2 \sum_{p=0}^{k-1} \mathbb{E}_z \left( \psi \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) + O(\delta^2),$$

where the term  $O(\delta^2)$  is independent of k. Hence we may write:

$$\frac{1}{n} \sum_{k=1}^{n} \sum_{p=0}^{k-1} \mathbb{E}_z \left( u \left( p\delta, \bar{Z}_{k-p} \right) - u \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) \\ = \frac{1}{n} \sum_{k=1}^{n} \sum_{p=0}^{k-1} \delta^2 \mathbb{E}_z \left( \psi \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) + O(\delta^2) \\ = \frac{\delta^2}{n} \sum_{i=1}^{n} \sum_{j=0}^{n} \mathbb{E}_z \left( \psi \left( (i\delta, \bar{Z}_j) \right) - \frac{\delta^2}{n} \sum_{k=1}^{n} \sum_{p=k}^{n} \mathbb{E}_z \left( \psi \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) + O(\delta^2).$$

Using Theorem 2.2.1 and Corollary 2.4.1, we bound:

$$\left| \frac{1}{n} \sum_{k=1}^{n} \sum_{p=k}^{n} \mathbb{E}_{z} \left( \psi \left( (p+1)\delta, \bar{Z}_{k-(p+1)} \right) \right) \right| \leq \frac{1}{n} \sum_{k=1}^{n} \sum_{p=k}^{n} \mathbb{E}_{z} \left( Ce^{-q(p+1)\delta} e^{bH\left(\bar{Z}_{k-(p+1)}\right)} \right)$$
$$\leq \frac{C'}{n} \sum_{k=1}^{n} \sum_{p=k}^{\infty} e^{-q(p+1)\delta}$$
$$\leq \frac{C''}{n} \sum_{k=1}^{n} e^{-qk\delta} \underset{n \to \infty}{\to} 0.$$

Ergodicity of the numerical scheme yields that

$$\frac{1}{n}\sum_{j=0}^{n}\mathbb{E}_{z}\left(\psi\left((i\delta,\bar{Z}_{j})\right)\underset{n\to\infty}{\rightarrow}\mu_{\beta,\delta}\left(\psi(i\delta,\cdot)\right),\right.$$

and Theorem 2.2.1 and Corollary 2.4.1 that:

$$\left|\frac{1}{n}\sum_{j=0}^{n}\mathbb{E}_{z}\left(\psi\left((i\delta,\bar{Z}_{j})\right)\right)\right|\leqslant Ce^{-qi\delta}.$$

Hence the dominated convergence theorem yields that:

$$\frac{1}{n}\sum_{i=1}^{n}\sum_{j=0}^{n}\mathbb{E}_{z}\left(\psi\left((i\delta,\bar{Z}_{j})\right)\underset{n\to\infty}{\rightarrow}\sum_{i\geqslant0}\mu_{\beta,\delta}\left(\psi(i\delta,\cdot)\right).$$

Now Lemma 2.4.5 and Riemann sum tells us that:

$$\sum_{i \ge 0} \mu_{\beta,\delta} \left( \psi(i\delta, \cdot) \right) = \int_0^\infty \mu_\beta(\psi(t, \cdot)) dt + O(\delta).$$

Letting n go to infinity in Equation (2.42) (with  $g(t, \cdot) = f$ ) then concludes the proof.

### Chapter 3

# Long-time behavior of the velocity jump Langevin process

#### Outline of the current chapter

3.1	Introduction	67
	3.1.1 Motivation	67
	3.1.2 Mathematical setting	69
	3.1.3 Notation	71
<b>3.2</b>	Proof of Theorem 3.1.1	71
	3.2.1 Invariance of $\mu$	72
	3.2.2 Exponential decay for the $H^1(\mu)$ -norm	74
	3.2.3 Exponential decay for the $H^k$ -norm	78

**Abstract**: We study the long-time convergence in Sobolev spaces of the velocity jump Langevin process, which is a process designed to sample from the Gibbs measure with a reduced number of computation of the gradient of the potential compared to the Langevin process.

#### 3.1 Introduction

#### 3.1.1 Motivation

As in the previous chapter, we are interested in the sampling of the Gibbs measure

$$\mu = \frac{1}{\mathcal{Z}_{\mu}} \exp(-\beta H(x, v)) \mathrm{d}x \mathrm{d}v \tag{3.1}$$

where  $\mathcal{Z}_{\mu} = \int_{\mathbb{R}^{2d}} \exp(-\beta H(x, v)) dx dv$  is a normalization constant. A popular process that allows for the sampling of the Gibbs measure is the kinetic Langevin diffusion, defined as the solution of the following SDE:

$$\begin{cases} dX_t = V_t dt \\ dV_t = -\nabla U(X_t) dt - \gamma V_t dt - \sqrt{2\beta^{-1}\gamma} dW_t , \end{cases}$$
(3.2)

where  $(W_t)_{t\geq 0}$  is a standard Brownian motion in  $\mathbb{R}^d$  and  $\gamma > 0$  is a friction parameter. When simulating this dynamic, the most expensive part is the computation of the forces  $\nabla U$ . A common approach is to substitute this gradient with a stochastic version, as in the stochastic gradient descent in optimization, see [20, 146, 92], or to treat various parts of the potentials with different time steps (multi-time-step methods), see [162]. We are interested here in an alternative approach that involves replacing (3.2) with a hybrid model combining a classical Langevin diffusion and a piecewise deterministic Markov process. This hybrid model still samples from  $\mu$  but can be simulated using a numerical splitting scheme that requires fewer gradient computations per time step.

The idea involves decomposing the potential as

$$\nabla U = \sum_{i=0}^{K} F_i \,,$$

where  $F_0$  denotes the computationally inexpensive component (typically the short-range forces exhibiting fast variation), while the  $F_i$  terms, for  $i \ge 1$ , represent long-range forces, which are computationally more demanding than  $F_0$  (as each atom interacts with all others through these forces, unlike the short-range ones). Then, define the so-called velocity jump Langevin process  $(X_t, V_t)_{t\ge 0}$  as the Langevin diffusion process associated with the force  $F_0$ :

$$\begin{cases} dX_t = V_t dt \\ dV_t = -F_0(X_t) dt - \gamma V_t dt - \sqrt{2\beta^{-1}\gamma} dW_t \end{cases}$$

with additional jumps at rate  $\lambda_i(x, v)$ , corresponding to a jump kernel  $q_i(x, v, dv')$  depending on  $F_i$  in such a way that ensures that the equilibrium measure of the process is indeed the Gibbs measure  $\mu$ . The Langevin velocity jump process can be simulated using a splitting scheme similar to the BAOAB scheme introduced by Leimkuhler, where the parts corresponding to the free transport, the forces, the friction/dissipation, and the jumps are simulated separately, see [21, 107, 105, 106]. More precisely, the generator  $\mathcal{L}$  will be decomposed as a sum of Markov generators

$$\mathcal{L} = \mathcal{L}_A + \mathcal{L}_B + \mathcal{L}_O + \mathcal{L}_J \,,$$

each corresponding to a part of the dynamic, see Section 3.1.2 below. The process can then be discretized base on the Trotter/Strang formula

$$e^{t\mathcal{L}} = e^{\frac{t}{2}\mathcal{L}_B} e^{\frac{t}{2}\mathcal{L}_J} e^{\frac{t}{2}\mathcal{L}_A} e^{t\mathcal{L}_O} e^{\frac{t}{2}\mathcal{L}_A} e^{\frac{t}{2}\mathcal{L}_J} e^{\frac{t}{2}\mathcal{L}_B} + \mathop{o}_{t\to 0}(t^2) \,.$$

The simulation of the jump part of the process is based on the thinning method, see [113, 116]. Suppose that  $\lambda_i \leq \overline{\lambda_i}$  for  $1 \leq i \leq K$ , and denote  $\overline{\lambda} = \sum_i \overline{\lambda_i}$ . A direct computation on the generator shows that the jumps can be exactly simulated this way: starting from (x, v),

- 1. Draw  $\mathcal{E}$  a standard exponential random variable, and let  $T = \mathcal{E}/\overline{\lambda}$  be the next jump time proposal.
- 2. Draw I in  $[\![1,K]\!]$  such that  $\mathbb{P}(I=i) = \overline{\lambda_i}/\overline{\lambda}$ . Propose a jump of type I at time T.
- 3. Accept the jump with probability  $\lambda_I(x, v)/\overline{\lambda_I}$ , in which case the velocity is re-sampled at time T according to  $q_I(x, v, dv')$ , otherwise the velocity at time T is simply v (there is no jump).

Unlike a traditional numerical scheme for the Langevin diffusion (like BAOAB), where the gradient  $\nabla U$  is computed at every time step, here we only need to evaluate  $F_i$  when a jump of type *i* is proposed (at step 3 above), which does not occur at every time step for every *i*. Moreover, the jump rate and jump kernel, first defined in [138], will depend on a parameter, in such a way that when this parameter goes to 0, the velocity jump Langevin process converges to the classical Langevin diffusion (in the sense that the distributions of the processes are close in the space of cadlag trajectories endowed with the Skorohod topology, see [138, Theorem 3.6]). As a consequence, in addition to the sampling of the Gibbs measure, dynamical properties such as the diffusion constants may be computed using this process.

Motivated by a weak error expansion à la Talay-Tubaro of a splitting scheme for the velocity jump Langevin process as in Chapter 2, we prove convergence of its semi-group in Sobolev space of arbitrary order. The derivation of estimates on the semi-group and weak error expansion is left for future work.

#### 3.1.2 Mathematical setting

Fix some smooth  $U: \mathbb{R}^d \to \mathbb{R}$ . We will work under the following set of assumptions.

**Assumption 3.1.1.** • The Gibbs measure is well defined

$$\int_{\mathbb{R}^d} e^{-U(x)} dx < \infty.$$

- $U = U_0 + U_1$  where the derivatives of  $U_1$  of all order are bounded.
- $\nabla^2 U_0$  as well as its derivatives of all order are bounded, and there exist  $\kappa > 0$  and a compact set  $\mathcal{K} \subset \mathbb{R}^d$  such that for all  $x \notin \mathcal{K}$

$$-x \cdot \nabla U_0(x) \leqslant -\kappa |x|^2.$$

•  $\Psi \in \mathcal{C}^{\infty}(\mathbb{R}^d, \mathbb{R}_+)$  is a smooth non-negative function such that  $\Psi'$  and all its derivative are bounded, and

$$\Psi(s) - \Psi(-s) = s. \tag{3.3}$$

For simplicity, in the remainder of this work, we fix  $\beta = 1$ , and consider a particular decomposition of the forces. Let us denote  $(e_i)_{1 \le i \le d}$  the canonical basis of  $\mathbb{R}^d$ , let K = d, and

$$F_0 = \nabla U_0, \quad F_i = \partial_i U_1 e_i, \, i \in [\![1, d]\!].$$
 (3.4)

In other words, we write U as  $U_0 + U_1$  and decompose  $\nabla U_1$  in a similar fashion as in the Zig-Zag process:

$$\nabla U_1(x) = \sum_{i=1}^d \partial_i U_1(x) e_i \,.$$

Fix a friction parameter  $\gamma > 0$  and a constant  $\theta > 0$ . Write  $\mathcal{C}_c^{\infty}(\mathbb{R}^{2d})$  for the set of smooth function with compact support from  $\mathbb{R}^{2d}$  to  $\mathbb{R}$ . The Markov process we are going to study is the so-called velocity jump Langevin process, whose Markov generator is given by

$$\mathcal{L} = \mathcal{L}_H + \mathcal{L}_D + \mathcal{L}_J$$

where for any function  $h \in \mathcal{C}_c^{\infty}(\mathbb{R}^{2d})$ :

$$\mathcal{L}_{H}h(x,v) = v \cdot \nabla_{x}h(x,v) - \nabla U_{0}(x) \cdot \nabla_{v}h(x,v) \quad \text{(Hamiltonian part)}$$
$$\mathcal{L}_{D}h(x,v) = -\gamma v \cdot \nabla_{v}h(x,v) + \gamma \Delta_{v}h(x,v) \quad \text{(diffusion)}$$
$$\mathcal{L}_{J}h(x,v) = \sum_{i=1}^{d} \mathcal{L}_{J}^{i}h(x,v) \quad \text{(jumps)}$$

where

$$\mathcal{L}_J^i h(x,v) = \lambda_i(x,v) \int_{\mathbb{R}^d} \left( h(x,v') - h(x,v) \right) q_i(x,v,\mathrm{d}v'),$$

$$\lambda_i(x,v) \int_{\mathbb{R}^d} h(x,v') q_i(x,v,dv') = \frac{1+\theta^2}{\theta^2} \mathbb{E}\left[h\left(x,v-\frac{2\theta}{1+\theta^2}\left(\theta v_i+G\right)e_i\right)\Psi\left(\frac{\theta\partial_i U_1(x)}{1+\theta^2}\left(\theta v_i+G\right)\right)\right],$$

and  $G \sim \mathcal{N}(0,1)$  is a one-dimensional standard Gaussian variable. In the definition above,  $q_i(x, v, \cdot)$  is a probability measure for all  $(x, v) \in \mathbb{R}^{2d}$ , in other words  $\lambda_i$ , for all  $i \in [\![1, d]\!]$ , is obtained by taking h = 1 in the formula above. As explained above, this is the generator of a Langevin process with drift given by  $U_0$ , and that jumps at rate  $\lambda_i$  according to the kernel  $q_i$ . Let  $(X_t, V_t)$  be a Markov process with generator  $\mathcal{L}$ , and let  $(P_t)_{t \ge 0}$  be the semi-group associated to this process

$$P_t f(x, v) = \mathbb{E}_{(x,v)} \left[ f(X_t, V_t) \right]$$

defined for  $f : \mathbb{R}^{2d} \to \mathbb{R}$  bounded. Fix  $k \in \mathbb{N}$  and define the Sobolev space of order k by

$$H^{k}(\mu) = \left\{ f: \mathbb{R}^{2d} \to \mathbb{R} \text{ measurable}, \sum_{i+j \leqslant k} \int_{\mathbb{R}^{2d}} |\nabla^{i}_{x} \nabla^{j}_{y} f|^{2} \mathrm{d}\mu < \infty \right\},$$

where we denoted

$$|\nabla_x^i \nabla_v^j f|^2 = \sum_{|\alpha_1|=i; |\alpha_2|=j} |\partial_x^{\alpha_1} \partial_v^{\alpha_2} f|^2, \quad |\alpha_p| = \sum_{i=1}^d (\alpha_p)_i.$$

Write as well

$$\|f\|_{H^k(\mu)} = \sum_{i+j \leqslant k} \int_{\mathbb{R}^{2d}} |\nabla^i_x \nabla^j_y f|^2 \mathrm{d}\mu$$

the Sobolev norm for  $f \in H^k(\mu)$ , and

$$\mu(f) = \int_{\mathbb{R}^{2d}} f \mathrm{d}\mu.$$

**Theorem 3.1.1.** Suppose Assumptions 3.1.1. For all  $k \in \mathbb{N}$ , there exist  $C_k, \rho_k > 0$  such that for all  $f \in H^k(\mu)$  and  $t \ge 0$ 

$$||P_t f - \mu(f)||_{H^k(\mu)} \leq C_k e^{-\rho_k t} ||f - \mu(f)||_{H^k(\mu)}.$$

Let's make a few comments on the assumptions and the theorem. First, we take  $\theta > 0$  as a positive constant for simplicity, but it can actually be taken as a function of x (for instance  $\theta(x) = \theta_0 |\nabla U_1(x)|$  with  $\theta_0 > 0$ ). As explained in [138], the case  $\theta = \infty$  corresponds to the bouncy particle sampler (see [22]), whereas when  $\theta \to 0$ , we retrieve the kinetic Langevin process. Secondly, the condition (3.3) on  $\Psi$  is here to ensure the invariance of  $\mu$ . In the Zigzag process, the jump rate is usually taken such that  $\Psi(s) = \max(0, s)$ , which indeed satisfies the equality (3.3). However, the computations in Sobolev spaces require that the generator has smooth coefficients, which prevent such a  $\Psi$ . An admissible example is given by  $\Psi(s) = a \log(\exp(s/a) + 1)$  for any a > 0. Finally, the choice of the decomposition (3.4) is made in order to avoid singular terms of the form  $\nabla U/|\nabla U|$  in the hypocoercivity computations. As explained in Section 3.1.1, this theorem is a first step in proving uniform bounds on the semi-group of the velocity jump process, in order to show weak error expansion of numerical schemes. The main addition with respect to Chapter 2 is the treatment of the jump terms.

#### 3.1.3 Notation

In all the following,  $|\cdot|$  denotes the euclidean norm in  $\mathbb{R}^d$  as well as the Frobenius norm for matrices, and  $\cdot$  the standard dot product on  $\mathbb{R}^d$  or the Frobenius scalar product when it's applied on matrices. For any operators A and B, we denote [A, B] = AB - BA their commutator. In some computations, it will be useful to work with  $\rho = \frac{1-\theta^2}{1+\theta^2} \in (-1,1)$  instead of  $\theta$ . In the remainder of this work, we will denote by C various constants that may change from line to line.

### 3.2 Proof of Theorem 3.1.1

The proof of Theorem 3.1.1 relies on hypocoercive computations à la Villani in Sobolev spaces. The velocity jump Langevin process is hypocoercive, in the sense that the generator of this process is not coercive for the usual Sobolev norm. To prove exponential convergence to equilibrium, for a given  $k \in \mathbb{N}$ , we are going to introduce a norm equivalent to the classical Sobolev norm, defined for  $f \in H^k(\mu)$  by

$$\mathcal{N}_{k}(f) = \int_{\mathbb{R}^{2d}} f^{2} \mathrm{d}\mu + \sum_{p=1}^{k} \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} |\nabla_{x}^{i} \nabla_{v}^{p-i} f|^{2} + \omega_{p,p} |(\nabla_{x}^{p} - \nabla_{x}^{p-1} \nabla_{v}) f|^{2} \right) \mathrm{d}\mu$$
(3.5)

where  $(\omega_{i,p})_{p \in \mathbb{N}, 1 \leq i \leq p}$  are positive numbers. In the remainder of this work, we are going to refer to this norm as the modified Sobolev norm. Following [163], we are going to show that we may choose  $(\omega_{i,p})_{p \in \mathbb{N}, 1 \leq i \leq p}$  such that for  $f \in H^k$ , we have

$$\partial_t \mathcal{N}_k \left( P_t f - \mu(f) \right) \leqslant -\rho_k \mathcal{N}_k \left( P_t f - \mu(f) \right) ,$$

for some  $\rho_k > 0$ . Theorem 3.1.1 is a direct consequence of the following result:

**Theorem 3.2.1.** There exist positive  $(\omega_{i,p})_{p \in \mathbb{N}, 1 \leq i \leq p}$  and  $(\rho_k)_{k \in \mathbb{N}^*}$  such that for all  $k \in \mathbb{N}^*$ ,  $f \in H^k(\mu)$  and  $t \ge 0$ ,

$$\mathcal{N}_k(P_t f - \mu(f)) \leqslant e^{-\rho_k t} \mathcal{N}(f - \mu(f)),$$

where  $\mathcal{N}_k$  was defined in (3.5).

The proof of this theorem relies on an induction argument and Gamma calculus, as presented in Section 1.2.1. Fix  $\phi : \mathcal{C}_c^{\infty}(\mathbb{R}^{2d}) \to \mathcal{C}_c^{\infty}(\mathbb{R}^{2d})$ . Recall the definition of the generalized *carré du*  champ

$$\Gamma_{\phi}(h) = \frac{1}{2} \left( \mathcal{L}(\phi(h)) - D_h \phi(h) \mathcal{L}h \right), \qquad (3.6)$$

where  $D_h \phi$  denotes the differential operator of  $\phi$ . When  $\phi(h) = h^2$  we retrieve the usual carré du champ, which is simply denoted by  $\Gamma$ . We will show in Section 3.2.1 that  $\mu$  is indeed an invariant measure of the velocity jump process. This formally implies that:

$$\partial_t \mathcal{N}_k(P_t f - \mu(f)) = -\int_{\mathbb{R}^{2d}} \Gamma(P_t f) \mathrm{d}\mu - \sum_{p=1}^k \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p}(P_t f) + \omega_{p,p} \Gamma_{p,p}(P_t f) \right) \mathrm{d}\mu \,, \quad (3.7)$$

where, for i < p,

$$\Gamma_{i,p}(h) = \Gamma_{|\nabla_x^i \nabla_v^{p-i} \cdot|^2}(h) \quad \text{and} \quad \Gamma_{p,p}(h) = \Gamma_{|(\nabla_x^p - \nabla_x^{p-1} \nabla_v) \cdot|^2}(h) \,.$$

If we could show that for all  $h \in \mathcal{C}^{\infty}_{c}(\mathbb{R}^{2d})$ 

$$\int_{\mathbb{R}^{2d}} \Gamma(h) \mathrm{d}\mu + \sum_{p=1}^{k} \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p}(h) + \omega_{p,p} \Gamma_{p,p}(h) \right) \mathrm{d}\mu \ge \rho_k \mathcal{N}_k(h - \mu(h)) \,,$$

for some  $\rho_k > 0$ , then Theorem 3.2.1 would follow, at least formally, in the sense that we haven't justified the time derivatives. We will actually prove a slightly stronger result (see Proposition 3.2.5 below) for the induction argument. Recall also the proposition that allows for the computations of this generalized *carré du champ*.

**Proposition 3.2.1.** If there exists  $A = (A_1, \dots, A_p) : \mathcal{C}^{\infty} \to (\mathcal{C}^{\infty})^p$  a linear operator such that  $\phi(h) = |Ah|^2$ , then

$$\Gamma_{\phi}(h) = \Gamma(Ah) + Ah \cdot [\mathcal{L}, A]h$$

where  $\Gamma(Ah) = \sum_{i=1}^{p} \Gamma(A_i h)$  and  $[\mathcal{L}, A] = ([\mathcal{L}, A_1], \cdots, [\mathcal{L}, A_p]).$ 

For all i, j, i', j' such that i + j = i' + j', the scalar product here has to be understood as

$$\nabla^i_x \nabla^j_v h \cdot \nabla^{i'}_x \nabla^{j'}_v h := \sum_{|\alpha_1|=i} \sum_{|\alpha_2|=j'} \sum_{|\nu|=i'-i} (\partial^{\alpha_1}_x \partial^{\alpha_2+\nu}_v h) (\partial^{\alpha_1+\nu}_x \partial^{\alpha_2}_v h) \,.$$

We will also extensively use the fact that

$$[\mathcal{L}, A] = [\mathcal{L}_H, A] + [\mathcal{L}_J, A] + [\mathcal{L}_D, A].$$

In order to justify the discussion above, we will start by proving that the Gibbs measure is indeed invariant for the velocity jump Langevin process in Section 3.2.1. Section 3.2.2 will treat the case k = 1 which is the initialization of the induction argument. Section 3.2.3 will conclude with the proof of convergence in  $H^k$  space through Theorem 3.2.1.

#### **3.2.1** Invariance of $\mu$

The use of the velocity jump process for the sampling of  $\mu$ , as well as the use of hypocoercivity method rely on the fact that  $\mu$  is indeed an invariant measure of this process. Let us prove this property by computing the adjoint of  $\mathcal{L}$  in  $L^2(\mu)$ , denoted as  $\mathcal{L}^*$ . **Proposition 3.2.2.** Denote  $\mathcal{L}^*$  the adjoint of  $\mathcal{L}$  in  $L^2(\mu)$ . For all  $h \in \mathcal{C}^{\infty}_c(\mathbb{R}^{2d})$ , we have:

$$\mathcal{L}^* h = -v \cdot \nabla_x h + \nabla U_0 \cdot \nabla_v h - \gamma v \cdot \nabla_v h + \gamma \Delta_v h + \frac{1+\theta^2}{\theta^2} \sum_{i=1}^d \mathbb{E}\left[ (h(\cdot, V^i) - h) \Psi\left(\frac{\partial_i U_1}{2} (V_i^i - v_i)\right) \right],$$

with

$$V^{i} = v - \frac{2\theta}{1 + \theta^{2}} (\theta v_{i} + G)e_{i},$$

where G is a random variable with standard Gaussian distribution.

*Proof.* Let  $g, h \in \mathcal{C}^{\infty}_{c}(\mathbb{R}^{2d})$ . By integrating by parts, we get

$$\begin{split} \int_{\mathbb{R}^{2d}} g\mathcal{L}_{H} h \mathrm{d}\mu &= \mathcal{Z}_{\mu}^{-1} \int_{\mathbb{R}^{2d}} g(v \cdot \nabla_{x} h - \nabla U_{0} \cdot \nabla_{v} h) e^{-U(x)} e^{-|v|^{2}/2} \mathrm{d}x \mathrm{d}v \\ &= -\mathcal{Z}_{\mu}^{-1} \int_{\mathbb{R}^{2d}} h v \cdot \nabla_{x} (g e^{-U(x)}) e^{-|v|^{2}/2} \mathrm{d}x \mathrm{d}v \\ &\quad + \mathcal{Z}_{\mu}^{-1} \int_{\mathbb{R}^{2d}} h \nabla_{v} \cdot (g e^{-|v|^{2}/2} \nabla U_{0}(x)) e^{-U(x)} \mathrm{d}x \mathrm{d}v \\ &= -\int_{\mathbb{R}^{2d}} h v \cdot \nabla_{x} g \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} g h v \cdot \nabla U(x) \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} h \nabla U_{0}(x) \cdot \nabla_{v} g \mathrm{d}\mu \\ &\quad - \int_{\mathbb{R}^{2d}} g h v \cdot \nabla U_{0}(x) \mathrm{d}\mu \\ &= \int_{\mathbb{R}^{2d}} h(-v \cdot \nabla_{x} g + \nabla U_{0} \cdot \nabla_{v} g) \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} g h v \cdot \nabla U_{1} \mathrm{d}\mu \,. \end{split}$$

We also have

$$\begin{split} \int_{\mathbb{R}^{2d}} g\mathcal{L}_D h \mathrm{d}\mu &= \mathcal{Z}_{\mu}^{-1} \int_{\mathbb{R}^{2d}} g(-\gamma v \cdot \nabla_v h + \gamma \Delta_v h) e^{-U(x)} e^{-|v|^2/2} \mathrm{d}x \mathrm{d}v \\ &= \gamma \mathcal{Z}_{\mu}^{-1} \left( \int_{\mathbb{R}^{2d}} g \nabla_v (e^{-|v|^2/2}) \cdot \nabla_v h e^{-U(x)} \mathrm{d}x \mathrm{d}v + \int_{\mathbb{R}^{2d}} h \Delta_v (g e^{-|v|^2/2}) e^{-U(x)} \mathrm{d}x \mathrm{d}v \right) \\ &= \gamma \mathcal{Z}_{\mu}^{-1} \left( \int_{\mathbb{R}^{2d}} \nabla_v (g e^{-|v|^2/2}) \cdot \nabla_v h e^{-U(x)} \mathrm{d}x \mathrm{d}v - \int_{\mathbb{R}^{2d}} \nabla_v g \cdot \nabla_v h e^{-U(x)} e^{-|v|^2/2} \mathrm{d}x \mathrm{d}y \right) \\ &+ \int_{\mathbb{R}^{2d}} h \Delta_v (g e^{-|v|^2/2}) e^{-U(x)} \mathrm{d}x \mathrm{d}v \right) \\ &= -\gamma \int_{\mathbb{R}^{2d}} \nabla_v g \cdot \nabla_v h \mathrm{d}\mu = \int_{\mathbb{R}^{2d}} h \mathcal{L}_D g \mathrm{d}\mu \end{split}$$

For the jump part, denote  $k_0(x, v, \cdot)$  the law of the random variable  $V^i$ . Then  $k_0$  is reversible with respect to the standard Gaussian measure, in the sense that for all  $x \in \mathbb{R}^d$ ,

$$k_0(x, v, \mathrm{d}v')e^{-|v|^2/2}\mathrm{d}v = k_0(x, v', \mathrm{d}v)e^{-|v'|^2/2}\mathrm{d}v'.$$
(3.8)

Indeed, since  $V_j^i = v_j$  for  $j \neq i$  and

$$V_{i}^{i} = \frac{1 - \theta^{2}}{1 + \theta^{2}}v_{i} + \frac{2\theta}{1 + \theta^{2}}G = \rho v_{i} + \sqrt{1 - \rho^{2}}G,$$

where  $\rho = \frac{1-\theta^2}{1+\theta^2}$ , the coordinates of  $V^i$  are independent and each transition  $v_j \mapsto V_j^i$  is reversible for the standard one-dimensional Gaussian. Using this reversibility, and the fact that  $\Psi(s) - \Psi(-s) = s$ , we get that

$$\begin{split} \int_{\mathbb{R}^{2d}} g\mathcal{L}_{J}h \mathrm{d}\mu &= \frac{1+\theta^{2}}{\theta^{2}} \sum_{i=1}^{d} \int_{\mathbb{R}^{2d}} g\mathbb{E} \left[ (h(x,V^{i}) - h(x,v))\Psi\left(\frac{\partial_{i}U_{1}(x)}{2}(v_{i} - V_{i}^{i})\right) \right] \mathrm{d}\mu \\ &= \frac{1+\theta^{2}}{\theta^{2}} \sum_{i=1}^{d} \int_{\mathbb{R}^{2d}} h\mathbb{E} \left[ (g(x,V^{i}) - g)\Psi\left(-\frac{\partial_{i}U_{1}(x)}{2}(v_{i} - V_{i}^{i})\right) \right] \\ &\quad + \frac{1+\theta^{2}}{\theta^{2}} \sum_{i=1}^{d} \int_{\mathbb{R}^{2d}} gh\mathbb{E} \left[ \Psi\left(-\frac{\partial_{i}U_{1}(x)}{2}(v_{i} - V_{i}^{i})\right) - \Psi\left(\frac{\partial_{i}U_{1}(x)}{2}(v_{i} - V_{i}^{i})\right) \right] \mathrm{d}\mu \\ &= \frac{1+\theta^{2}}{\theta^{2}} \sum_{i=1}^{d} \int_{\mathbb{R}^{2d}} h\mathbb{E} \left[ (g(x,V^{i}) - g)\Psi\left(-\frac{\partial_{i}U_{1}}{2}(v_{i} - V_{i}^{i})\right) \right] \\ &\quad - \frac{1+\theta^{2}}{2\theta^{2}} \sum_{i=1}^{d} \int_{\mathbb{R}^{2d}} gh\partial_{i}U_{1}\mathbb{E} \left[ (v_{i} - V_{i}^{i}) \right] \mathrm{d}\mu \\ &= \frac{1+\theta^{2}}{\theta^{2}} \sum_{i=1}^{d} \int_{\mathbb{R}^{2d}} h\mathbb{E} \left[ (g(x,V^{i}) - g)\Psi\left(-\frac{\partial_{i}U_{1}}{2}(v_{i} - V_{i}^{i})\right) \right] - \int_{\mathbb{R}^{2d}} ghv \cdot \nabla U_{1} \mathrm{d}\mu \end{split}$$

Therefore, by summing the three terms, we get the result.

**Corollary 3.2.1.** Under Assumption 3.1.1, for all  $h \in \mathcal{C}^{\infty}_{c}(\mathbb{R}^{2d})$ 

$$\int_{\mathbb{R}^d \times \mathbb{R}^d} \mathcal{L}h\mu = 0$$

*Proof.* This is a direct consequence of the fact that  $\mathcal{L}^* 1 = 0$ .

Notice in particular that  $\mathcal{L}_D$  is symmetric in  $L^2(\mu)$ , whereas neither  $\mathcal{L}_H$  nor  $\mathcal{L}_J$  are left invariant by  $\mu$ , only  $\mathcal{L}_H + \mathcal{L}_J$  is.

### **3.2.2** Exponential decay for the $H^1(\mu)$ -norm

In this section, we prove convergence of the process in  $H^1(\mu)$ . The modified Sobolev norm here reads

$$\mathcal{N}_1(h) = \int_{\mathbb{R}^{2d}} h^2 \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} \omega_{0,1} |\nabla_v h|^2 + \omega_{1,1} |(\nabla_x - \nabla_v)h|^2 \mathrm{d}\mu,$$

and Equation (3.7) can be written:

$$\partial_t \mathcal{N}_1(P_t f - \mu(f)) = -\int_{\mathbb{R}^{2d}} \Gamma(P_t f) \mathrm{d}\mu - \int_{\mathbb{R}^{2d}} \left( \omega_{0,1} \Gamma_{0,1}(P_t f) + \omega_{1,1} \Gamma_{1,1}(P_t f) \right) \mathrm{d}\mu.$$

The initialisation of the induction argument to prove Theorem 3.2.1 is

**Lemma 3.2.1.** Under Assumption 3.1.1, there exist  $\omega_{0,1}, \omega_{1,1}, \rho_1$  such that for all  $h \in C_c^{\infty}(\mathbb{R}^{2d})$  satisfying  $\mu(h) = 0$ :

$$\int_{\mathbb{R}^{2d}} \Gamma(h) d\mu + \int_{\mathbb{R}^{2d}} \left( \omega_{0,1} \Gamma_{0,1}(h) + \omega_{1,1} \Gamma_{1,1}(h) \right) d\mu \ge \rho_1 \left( \mathcal{N}_1(h) + \int_{\mathbb{R}^{2d}} |\nabla_x \nabla_v h|^2 + |\nabla_v^2 h|^2 d\mu \right) \,.$$

The terms of order 2 are present to compensate future terms in the induction, but are not required for the the convergence in  $H^1(\mu)$ . This convergence relies on the so-called Poincare inequality

**Proposition 3.2.3.** Under Assumption 3.1.1, the Gibbs measure (3.1) satisfy a Poincaré inequality: there exists  $C_P > 0$  such that

$$\int_{\mathbb{R}^{2d}} \left(h - \mu(h)\right)^2 d\mu \leqslant C_P \int_{\mathbb{R}^{2d}} |\nabla h|^2 d\mu \,. \tag{3.9}$$

for all  $h \in \mathcal{C}^{\infty}_{c}(\mathbb{R}^{2d})$ .

*Proof.* Proof of such inequality can be found in [6].

In order to apply Proposition 3.2.1, we need to express the commutators. For the Hamiltonian part we get:

$$[\mathcal{L}_H, \nabla_x] = \nabla^2 U_0 \nabla_v, \quad [\mathcal{L}_H, \nabla_v] = -\nabla_x.$$

For the diffusion part we get:

$$[\mathcal{L}_D, \nabla_x] = 0, \quad [\mathcal{L}_D, \nabla_v] = \gamma \nabla_v.$$

Regarding the jump part, we have

**Lemma 3.2.2.** Under Assumption 3.1.1, there exists C > 0 such that for all  $h \in C_c^{\infty}(\mathbb{R}^{2d})$ :

$$\int_{\mathbb{R}^{2d}} |[\nabla_x, \mathcal{L}_J]h|^2 d\mu + \int_{\mathbb{R}^{2d}} |[\nabla_v, \mathcal{L}_J]h|^2 d\mu \leqslant C \left( \int_{\mathbb{R}^{2d}} |\nabla_v h|^2 d\mu + \int_{\mathbb{R}^{2d}} |\nabla_v^2 h|^2 d\mu \right) \,.$$

To prove Lemma 3.2.2, we need two intermediate results:

**Lemma 3.2.3.** For any function  $g \in \mathcal{C}^{\infty}_{c}(\mathbb{R}^{d})$ ,

$$\int_{\mathbb{R}^d} |v|^2 g^2(v) e^{-|v|^2/2} dv \leq 2d \int_{\mathbb{R}^d} g^2(v) e^{-|v|^2/2} dv + 4 \int_{\mathbb{R}^d} |\nabla_v g(v)|^2 e^{-|v|^2/2} dv.$$

*Proof.* This is a particular case of the Lemma A.24 of [163]. Notice that  $ve^{-|v|^2/2} = -\nabla e^{-|v|^2/2}$ . Hence, an integration by parts and Young's inequality yield, for any x,

$$\begin{split} \int_{\mathbb{R}^d} |v|^2 g^2(v) e^{-|v|^2/2} \mathrm{d}v &= -\int_{\mathbb{R}^d} g^2(v) v \cdot \nabla e^{-|v|^2/2} \mathrm{d}v \\ &= \int_{\mathbb{R}^d} \nabla \cdot (g^2(v) v) e^{-|v|^2/2} \mathrm{d}v \\ &= d \int_{\mathbb{R}^d} g^2(v) e^{-|v|^2/2} \mathrm{d}v + \int_{\mathbb{R}^d} 2g(v) \nabla g(v) \cdot v e^{-|v|^2/2} \mathrm{d}v \\ &\leqslant d \int_{\mathbb{R}^d} g^2(v) e^{-|v|^2/2} \mathrm{d}v + \int_{\mathbb{R}^d} 2|\nabla g(v)|^2 e^{-|v|^2/2} \mathrm{d}v \\ &+ \frac{1}{2} \int_{\mathbb{R}^d} |v|^2 g^2(v) e^{-|v|^2/2} \mathrm{d}v \,, \end{split}$$

and thus the result.

**Proposition 3.2.4** (Poincaré for Gaussians). For any function  $g \in C_c^{\infty}(\mathbb{R}^d)$ ,

$$\int_{\mathbb{R}^d} \left( g(v) - \frac{1}{(2\pi)^{d/2}} \int_{\mathbb{R}^d} h(w) e^{-|w|^2/2} dw \right)^2 e^{-|v|^2/2} dx \leqslant \int_{\mathbb{R}^d} |\nabla g(v)|^2 e^{-|v|^2/2} dv \,. \tag{3.10}$$

*Proof.* Proof of such inequality can be found in [6].

Proof of Lemma 3.2.2. For all  $i \in [\![1,d]\!]$ , denote

$$g_i(x,v,G) = \left(x,v - \frac{2\theta}{1+\theta^2} \left(\theta v_i + G\right) e_i\right), \quad \varphi_i(x,v,G) = \frac{\theta \partial_i U_1(x)}{1+\theta^2} \left(\theta v_i + G\right)$$

We then have

$$[\nabla_x, \mathcal{L}_J]h = \sum_{i=1}^d [\nabla_x, \mathcal{L}_J^i]h = \frac{1}{\theta} \sum_{i=1}^d \mathbb{E} \left[ (\theta v_i + G)(h(g_i(G)) - h)\Psi'(\varphi_i(G)) \right] \nabla \partial_i U_1 ,$$

and

$$[\nabla_v, \mathcal{L}_J]h = \sum_{i=1}^d [\nabla_v, \mathcal{L}_J^i]h = \sum_{i=1}^d \mathbb{E}\left[\partial_i U_1(h(g_i(G)) - h)\Psi'(\varphi_i(G)) - 2\partial_{v_i}h(g_i(G))\Psi(\varphi_i(G))\right]e_i.$$

Using the fact that  $\Psi'$ ,  $\nabla U_1$  and  $\nabla^2 U_1$  are bounded, there exists a constant C > 0 such that

$$|[\nabla_x, \mathcal{L}_J]h|^2 \leqslant C \sum_{i=1}^d \mathbb{E}[(v_i^2 + G^2)(h(g_i(G) - h)^2)],$$

and

$$|[\nabla_{v}, \mathcal{L}_{J}]h|^{2} \leq C \sum_{i=1}^{d} \mathbb{E}[(h(g_{i}(G)) - h)^{2} + (\partial_{v_{i}}h(g_{i}(G)))^{2}(1 + v_{i}^{2} + G^{2})]$$

Recall that by denoting  $k_0(x, v, \cdot)$  the law of the random variable

$$V^{i} = v - \frac{2\theta}{1 + \theta^{2}}(\theta v_{i} + G)e_{i}$$

where G is a random variable with standard normal distribution,  $k_0(x, v, \cdot)$  is reversible with respect to the standard Gaussian measure, see (3.8). By using this reversibility, and the fact that v being fixed,  $V_i^i \sim \mathcal{N}(\rho v_i, 1 - \rho^2)$ , we have that for any function  $h \in \mathcal{C}_c^{\infty}(\mathbb{R}^{2d})$ 

$$\begin{split} \int_{\mathbb{R}^{2d}} \mathbb{E}\left[ \left(1 + v_i^2 + G^2\right) \left(\partial_{v_i} h(g_i(G))\right)^2 \right] \mathrm{d}\mu &= \int_{\mathbb{R}^{2d}} (\partial_{v_i} h)^2 \left(1 + \mathbb{E}\left[ (V_i^i)^2 + \frac{(v_i^2 - \rho V_i^i)^2}{1 - \rho^2} \right] \right) \mathrm{d}\mu \\ &= \int_{\mathbb{R}^{2d}} (\partial_{v_i} h)^2 \left(2 + v_i^2\right) \mathrm{d}\mu \\ &\leqslant 4 \int_{\mathbb{R}^{2d}} \left( (\partial_{v_i} h)^2 + (\partial_{v_i}^2 h)^2 \right) \mathrm{d}\mu \,, \end{split}$$

where we applied Proposition 3.2.3 to with d = 1 to  $v_i \mapsto \partial_{v_i} h(x, v)$  in the last inequality. By

denoting  $\Pi_{v_i}$  the projection

$$\Pi_{v_i} h(x, v) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} h(x, v_1, \dots, v_{i-1}, w_i, v_{i+1}, \dots, v_d) e^{-w_i^2/2} \mathrm{d}w_i$$

we also have

$$\begin{split} \int_{\mathbb{R}^{2d}} \mathbb{E}\left[ (1+v_i^2+G^2)(h(g_i(G))-h)^2 \right] \mathrm{d}\mu &\leqslant 2 \int_{\mathbb{R}^{2d}} \mathbb{E}\left[ (1+v_i^2+G^2)((h(g_i(G))-\Pi_{v_i}h)^2+(h-\Pi_{v_i}h)^2) \right] \mathrm{d}\mu \\ &\leqslant 2 \int_{\mathbb{R}^{2d}} \left( 3+v_i^2+\mathbb{E}\left[ (V_i^i)^2+\frac{(v_i^2-\rho V_i^i)^2}{1-\rho^2} \right] \right) (h-\Pi_{v_i}h)^2 \mathrm{d}\mu \\ &\leqslant 2 \int_{\mathbb{R}^{2d}} (4+2v_i^2)(h-\Pi_{v_i}h)^2 \mathrm{d}\mu \\ &\leqslant 16 \int_{\mathbb{R}^{2d}} (h-\Pi_{v_i}h)^2 \mathrm{d}\mu + 16 \int_{\mathbb{R}^{2d}} (\partial_{v_i}h)^2 \mathrm{d}\mu \,, \end{split}$$

where we applied again Proposition 3.2.3 with d = 1 to  $v_i \mapsto h(x, v) - \prod_{v_i} (x, v)$  in the last inequality. The Poincaré inequality (3.10) can be written

$$\int_{\mathbb{R}^{2d}} (h - \Pi_{v_i} h)^2 \mathrm{d}\mu \leqslant \int_{\mathbb{R}^{2d}} |\partial_{v_i} h|^2 \mathrm{d}\mu \,,$$

and thus

Therefore, there exists C > 0 such that

$$\int_{\mathbb{R}^{2d}} |[\nabla, \mathcal{L}_J]h|^2 \mathrm{d}\mu \leqslant C \sum_{i=1}^d \int_{\mathbb{R}^{2d}} ((\partial_{v_i}h)^2 + |\nabla_v \partial_{v_i}h|^2) \mathrm{d}\mu = C \left( \int_{\mathbb{R}^{2d}} |\nabla_v h|^2 \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} |\nabla_v^2 h|^2 \mathrm{d}\mu \right),$$
which concludes the proof.

which concludes the proof.

We now have everything to prove Lemma 3.2.1.

Proof of Lemma 3.2.1. Fix  $\omega_{0,1} > \omega_{1,1} > 0$  that we are going to chose later. Let's first treat the derivative of the  $L^2$  norm. We have for all  $h \in \mathcal{C}^\infty_c(\mathbb{R}^{2d}), (x, v) \in \mathbb{R}^{2d}$ 

$$\Gamma(h)(x,v) = 2\gamma |\nabla_v h(x,v)|^2 + \sum_{i=1}^d \lambda_i(x,v) \int_{\mathbb{R}^d} (h(x,v') - h(x,v))^2 q_i(x,v,dv') \,. \tag{3.11}$$

This already implies that

$$\Gamma(h) \geqslant 2\gamma |\nabla_v h|^2$$

To get a lower bound on  $\Gamma_{0,1}(h)$  and  $\Gamma_{1,1}(h)$ , we use Proposition 3.2.1. Equation (3.11) yields:

$$\Gamma(\nabla_v h) \ge 2\gamma |\nabla_v^2 h|^2, \quad \Gamma((\nabla_x - \nabla_v) h) \ge 2\gamma |(\nabla_x \nabla_v - \nabla_v^2) h|^2,$$

so that using  $\omega_{0,1} > \omega_{1,1}$  and Young inequality:

$$\omega_{0,1}\Gamma(\nabla_v h) + \omega_{1,1}\Gamma((\nabla_x - \nabla_v)h) \ge 2\gamma(\omega_{0,1} - \omega_{1,1})|\nabla_v^2 h|^2 + \omega_{1,1}\left(\gamma|\nabla_v^2 h|^2 + 2\gamma/3|\nabla_x \nabla_v h|^2\right).$$

Let us now look at the commutators. Fix  $\varepsilon > 0$  and write using Young inequality and Lemma 3.2.2:

$$\begin{split} \int_{\mathbb{R}^{2d}} \nabla_v h \cdot [\mathcal{L}, \nabla_v] h \mathrm{d}\mu &= \int_{\mathbb{R}^{2d}} \nabla_v h \cdot (-\nabla_x h + \gamma \nabla_v + [\mathcal{L}_J, \nabla_v] h) \, \mathrm{d}\mu \\ &\geqslant -\int_{\mathbb{R}^{2d}} \left(\frac{1}{\varepsilon} + 1\right) |\nabla_v h|^2 + \frac{\varepsilon}{2} |(\nabla_x - \nabla_v) h|^2 + \frac{\varepsilon}{2} C \left(|\nabla_v h|^2 \mathrm{d}\mu + |\nabla_v^2 h|^2\right) \mathrm{d}\mu \,, \end{split}$$

for some C > 0. Using again Young inequality and Lemma 3.2.2, we get that there exists C > 0such that:

$$\int_{\mathbb{R}^{2d}} (\nabla_x - \nabla_v) h \cdot [\mathcal{L}, \nabla_x - \nabla_v] h d\mu = \int_{\mathbb{R}^{2d}} (\nabla_x - \nabla_v) h \cdot (\nabla^2 U_0 \nabla_v h + \nabla_x h - \gamma \nabla_v h + [\mathcal{L}_J, \nabla_x - \nabla_v] h) d\mu$$
$$\geqslant \int_{\mathbb{R}^{2d}} \frac{1}{2} |(\nabla_x - \nabla_v) h|^2 - C \left( |\nabla_v h|^2 + |\nabla_v^2 h|^2 \right) d\mu.$$

Now all is left to do is to choose  $\varepsilon$ ,  $\omega_{0,1}$  and  $\omega_{1,1}$ . We choose them such that

$$\frac{\omega_{1,1}}{\omega_{0,1}} \leqslant \frac{2\gamma - C\varepsilon}{2\gamma + C}, \quad \varepsilon < \min\left(\frac{\omega_{1,1}}{2\omega_{0,1}}, \frac{2\gamma}{C}\right), \quad C\omega_{1,1} + \omega_{0,1}\left(\frac{1}{\varepsilon} + 1 + \frac{C\varepsilon}{2}\right) \leqslant \gamma$$

so that

$$\begin{split} \int_{\mathbb{R}^{2d}} \Gamma(h) \mathrm{d}\mu &+ \int_{\mathbb{R}^{2d}} \left( \omega_{0,1} \Gamma_{0,1}(h) + \omega_{1,1} \Gamma_{1,1}(h) \right) \mathrm{d}\mu \\ &\geqslant \int_{\mathbb{R}^{2d}} \gamma |\nabla_v h|^2 + \frac{\omega_{1,1}}{4} |(\nabla_x - \nabla_y)h|^2 + \omega_{1,1} \gamma \left( |\nabla_v^2 h|^2 + \frac{2}{3} |\nabla_x \nabla_v h|^2 \right) \mathrm{d}\mu \\ &\geqslant \tilde{\rho}_1 \int_{\mathbb{R}^{2d}} \omega_{0,1} |\nabla_v h|^2 + \omega_{1,1} |(\nabla_x - \nabla_v)h|^2 + |\nabla_x \nabla_v h|^2 + |\nabla_v^2 h|^2 \mathrm{d}\mu \,, \end{split}$$

for some  $\tilde{\rho}_1 > 0$ . Since we consider h such that  $\mu(h) = 0$ , the Poincaré inequality (3.9) yields that there exists  $\rho_1 > 0$  such that

$$\begin{split} \tilde{\rho}_1 \int_{\mathbb{R}^{2d}} \omega_{0,1} |\nabla_v h|^2 + \omega_{1,1} |(\nabla_x - \nabla_v) h|^2 + |\nabla_x \nabla_v h|^2 + |\nabla_v^2 h|^2 \mathrm{d}\mu \geqslant \rho_1 \left( \mathcal{N}_1(h) + \int_{\mathbb{R}^{2d}} |\nabla_x \nabla_v h|^2 + |\nabla_v^2 h|^2 \mathrm{d}\mu \right) \,, \\ \text{which concludes the proof.} \end{split}$$

which concludes the proof.

#### Exponential decay for the $H^k$ -norm 3.2.3

Recall that we defined the modified Sobolev norm as

$$\mathcal{N}_k(h) := \int_{\mathbb{R}^{2d}} h^2 \mathrm{d}\mu + \sum_{p=1}^k \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} |\nabla_x^i \nabla_v^{p-i} h|^2 + \omega_{p,p} |(\nabla_x^p - \nabla_x^{p-1} \nabla_v) h|^2 \right) \mathrm{d}\mu$$

In order to control its derivative (3.7), we prove by induction the following lemma.

**Proposition 3.2.5.** Under Assumption 3.1.1, there exist  $(\omega_{i,p})_{1 \leq i \leq p}$  such that for all  $k \in \mathbb{N}$ ,

there exists  $\rho_k$  such that for all  $h \in \mathcal{C}^{\infty}_c(\mathbb{R}^{2d},\mathbb{R})$  such that  $\mu(h) = 0$ :

$$\int_{\mathbb{R}^{2d}} \Gamma(h) d\mu + \sum_{p=1}^{k} \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p}(h) + \omega_{p,p} \Gamma_{p,p}(h) \right) d\mu \ge \rho_k \left( \mathcal{N}_k(h) + \int_{\mathbb{R}^{2d}} \sum_{i=0}^{k} |\nabla_x^i \nabla_v^{k+1-i} h|^2 d\mu \right).$$

$$(3.12)$$

To prove this Lemma, we will need formulas for partial derivatives of products, provided by the Leibniz formula

**Proposition 3.2.6.** For all  $g, h \in C_c^{\infty}$  and  $\alpha_1, \alpha_2 \in \mathbb{N}^d$ ,

$$\partial_x^{\alpha_1} \partial_v^{\alpha_2}(gh) = \sum_{\nu_1 \leqslant \alpha_1, \nu_2 \leqslant \alpha_2} \binom{\alpha_1}{\nu_1} \binom{\alpha_2}{\nu_2} (\partial_x^{\nu_1} \partial_v^{\nu_2} g) (\partial_x^{\alpha_1 - \nu_1} \partial_v^{\alpha_2 - \nu_2} h)$$

where for  $\alpha = (\alpha_1, \ldots, \alpha_d)$  and  $\nu = (\nu_1, \ldots, \nu_d)$ ,

$$\begin{pmatrix} \alpha \\ \nu \end{pmatrix} = \prod_{i=1}^d \begin{pmatrix} \alpha_i \\ \nu_i \end{pmatrix},$$

and

$$\nu \leqslant \alpha \iff \forall i \in \llbracket 1, d \rrbracket, \nu_i \leqslant \alpha_i.$$

Let us look at the commutators. As in the  $H^1$  case, we will look separately the parts associated to the Hamiltonian dynamic  $\mathcal{L}_H$ , the diffusion  $\mathcal{L}_D$  and the jump process  $\mathcal{L}_J$ .

**Lemma 3.2.4** (Hamiltonian commutators). Let  $i \in [0, k]$ . There exists C > 0 such that the following holds. If i < k, we have

$$\nabla_x^i \nabla_v^{k+1-i} h \cdot [\mathcal{L}_H, \nabla_x^i \nabla_v^{k+1-i}] h \ge -|\nabla_x^i \nabla_v^{k+1-i} h|^2 - C \left( |\nabla_x^{i+1} \nabla_v^{k-i} h|^2 + \sum_{j=0}^{i-1} |\nabla_x^j \nabla_v^{k+2-i} h|^2 \right) \,.$$

In the case i = k, for all  $\eta > 0$ , we have

$$\nabla_x^k \nabla_v h \cdot [\mathcal{L}_H, \nabla_x^k \nabla_v] h \ge -\frac{\eta}{8} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v) h|^2 - C\left(1 + \frac{1}{\eta}\right) |\nabla_x^k \nabla_v h|^2 - C\sum_{j=0}^{k-1} |\nabla_x^j \nabla_v^2 h|^2.$$

Finally, we have

$$(\nabla_x^{k+1} - \nabla_x^k \nabla_v) h \cdot [\mathcal{L}_H, \nabla_x^{k+1} - \nabla_x^k \nabla_v] h \ge \frac{1}{2} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v) h|^2 - C \left( \sum_{j=0}^k |\nabla_x^j \nabla_v h|^2 + \sum_{j=0}^{k-1} |\nabla_x^j \nabla_v^2 h|^2 \right).$$

*Proof.* For all  $\alpha_1, \alpha_2 \in \mathbb{N}^d$  with  $|\alpha_1| = i$  and  $|\alpha_2| = k + 1 - i$ , where  $\alpha_2 = (\alpha_{2,1}, \ldots, \alpha_{2,d})$ , write

$$\alpha_2^j = (\alpha_{2,1}, \dots, \alpha_{2,j-1}, \alpha_{2,j} - 1, \alpha_{2,j+1}, \dots, \alpha_{2,d}),$$

and we have

$$\begin{aligned} \partial_x^{\alpha_1} \partial_v^{\alpha_2} \mathcal{L}_H h &= \sum_{j=1}^d \partial_x^{\alpha_1} \partial_v^{\alpha_2} \left( v_j \partial_{x_j} h - \partial_j U_0 \partial_{v_j} h \right) \\ &= \sum_{j=1}^d \partial_v^{\alpha_2} (v_j \partial_x^{\alpha_1} \partial_{x_j} h) - \partial_x^{\alpha_1} (\partial_j U_0 \partial_v^{\alpha_2} \partial_{v_j} h) \\ &= \sum_{j=1}^d \left( \sum_{\nu_2 \leqslant \alpha_2} \binom{\alpha_2}{\nu_2} \partial_v^{\nu_2} (v_j) \partial_v^{\alpha_2 - \nu_2} \partial_x^{\alpha_1} \partial_{x_j} h - \sum_{\nu_1 \leqslant \alpha_1} \binom{\alpha_1}{\nu_1} \partial_x^{\nu_1} \partial_j U_0 \partial_x^{\alpha_1 - \nu_1} \partial_v^{\alpha_2} \partial_{v_j} h \right) \\ &= \sum_{j=1}^d \left( \alpha_{2,j} \partial_v^{\alpha_2^j} \partial_x^{\alpha_1} \partial_{x_j} h + v_j \partial_v^{\alpha_2} \partial_x^{\alpha_1} \partial_{x_j} h - \sum_{\nu_1 \leqslant \alpha_1} \binom{\alpha_1}{\nu_1} \partial_x^{\nu_1} \partial_j U_0 \partial_x^{\alpha_1 - \nu_1} \partial_v^{\alpha_2} \partial_{v_j} h \right). \end{aligned}$$

Therefore,

$$[\mathcal{L}_H, \partial_x^{\alpha_1} \partial_v^{\alpha_2}]h = \sum_{j=1}^d \left( \sum_{\nu_1 \leqslant \alpha_1; \nu_1 \neq 0} \binom{\alpha_1}{\nu_1} \partial_x^{\nu_1} \partial_j U_0 \partial_x^{\alpha_1 - \nu_1} \partial_v^{\alpha_2} \partial_{v_j} h - \alpha_{2,j} \partial_v^{\alpha_j} \partial_x^{\alpha_1} \partial_{x_j} h \right)$$

Using the bound on the derivatives of  $U_0$ , this yields that there exists a C > 0 such that

$$\nabla_{x}^{i} \nabla_{v}^{k+1-i} h \cdot [\mathcal{L}_{H} \nabla_{x}^{i} \nabla_{v}^{k+1-i}] h \ge -|\nabla_{x}^{i} \nabla_{v}^{k+1-i} h|^{2} - C \left( |\nabla_{x}^{i+1} \nabla_{v}^{k-i} h|^{2} + \sum_{j=0}^{i-1} |\nabla_{x}^{j} \nabla_{v}^{k+2-i} h|^{2} \right).$$

If i = k, then for any  $\eta > 0$ , using the Young inequality

$$\sum_{\substack{|\alpha_1|=k;|\alpha_2|=1\\j\in[\![1,d]\!]}} (\partial_x^{\alpha_1} \partial_v^{\alpha_2} h)(\alpha_{2,j} \partial_v^{\alpha_2^j} \partial_x^{\alpha_1} \partial_{x_j} h) = \sum_{\substack{|\alpha_1|=k;j\in[\![1,d]\!]}} (\partial_x^{\alpha_1} \partial_{v_j} h)(\partial_x^{\alpha_1} \partial_{x_j} h)$$
$$= \sum_{\substack{|\alpha_1|=k;j\in[\![1,d]\!]}} (\partial_x^{\alpha_1} \partial_{v_j} h)(\partial_x^{\alpha_1} \partial_{x_j} - \partial_x^{\alpha_1} \partial_{v_j} + \partial_x^{\alpha_1} \partial_{v_j})h$$
$$\leqslant \left(1 + \frac{2}{\eta}\right) |\nabla_x^k \nabla_v h|^2 + \frac{\eta}{8} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v)h|^2.$$

which again implies that

$$\nabla_x^k \nabla_v h \cdot [\mathcal{L}_H, \nabla_x^k \nabla_v] h \ge -C\left(1 + \frac{1}{\eta}\right) |\nabla_x^k \nabla_v h|^2 - \frac{\eta}{8} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v) h|^2 - C\sum_{j=0}^{k-1} |\nabla_x^j \nabla_v^2 h|^2.$$

Similarly, for  $\beta$  a multi-index such that  $|\beta|=k$  and  $i\in [\![1,d]\!],$ 

$$\begin{aligned} [\mathcal{L}_{H},\partial_{x}^{\beta}\partial_{x_{i}}-\partial_{x}^{\beta}\partial_{v_{i}})]h &=\partial_{x}^{\beta}\partial_{x_{i}}h+\sum_{j=1}^{d}\sum_{\nu\leqslant\beta}\binom{\beta}{\nu}\partial^{\nu}\partial_{i,j}U_{0}\partial_{x}^{\beta-\nu}\partial_{v_{j}}h\\ &+\sum_{j=1}^{d}\sum_{\nu\leqslant\beta;\nu\neq0}\binom{\beta}{\nu}\partial^{\nu}\partial_{j}U_{0}\partial_{x}^{\beta-\nu}\partial_{x_{i}}\partial_{v_{j}}h-\sum_{j=1}^{d}\sum_{\nu\leqslant\beta;\nu\neq0}\binom{\beta}{\nu}(\partial^{\nu}\partial_{j}U_{0})(\partial_{x}^{\beta-\nu}\partial_{v_{i}}\partial_{v_{j}}h)\,,\end{aligned}$$

and thus, by applying Young inequality,

$$(\nabla_x^{k+1} - \nabla_x^k \nabla_v)h \cdot [\mathcal{L}_H, \nabla_x^{k+1} - \nabla_x^k \nabla_v]h \geqslant \frac{1}{2} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v)h|^2 - C\left(\sum_{j=0}^k |\nabla_x^j \nabla_v h|^2 + \sum_{j=0}^{k-1} |\nabla_x^j \nabla_v^2 h|^2\right),$$

which concludes the proof.

Regarding the diffusion part, we have

**Lemma 3.2.5.** For  $i \in [\![0, k]\!]$ ,

$$\nabla_x^i \nabla_v^{k+1-i} h \cdot [\mathcal{L}_D, \nabla_x^i \nabla_v^{k+1-i}] h = \gamma(k+1-i) |\nabla_x^i \nabla_v^{k+1-i} h|^2 \,,$$

and there exists C > 0 such that

$$(\nabla_x^{k+1} - \nabla_x^k \nabla_v)h \cdot [\mathcal{L}_D, \nabla_x^{k+1} - \nabla_x^k \nabla_v]h \ge -\frac{1}{8} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v)h|^2 - C|\nabla_x^k \nabla_v h|^2.$$

*Proof.* We have, for all  $\alpha_1, \alpha_2 \in \mathbb{N}^d$ ,

$$\partial_x^{\alpha_1} \partial_v^{\alpha_2} \mathcal{L}_D h(x,v) = -\gamma \sum_{j=1}^d \partial_v^{\alpha_2} (v_j \partial_x^{\alpha_1} \partial_{v_j} h) + \gamma \sum_{j=1}^d \partial_x^{\alpha_1} \partial_v^{\alpha_2} (\partial_{v_j}^2 h)$$
$$= -\gamma \sum_{j=1}^d v_j \partial_v^{\alpha_2} \partial_x^{\alpha_1} \partial_{v_j} h - \gamma \sum_{j=1}^d \alpha_{2,j} \partial_v^{\alpha_j} \partial_x^{\alpha_1} \partial_{v_j} h + \gamma \sum_{j=1}^d \partial_x^{\alpha_1} \partial_v^{\alpha_2} (\partial_{v_j}^2 h).$$

Therefore, if  $|\alpha_1| = i$  and  $|\alpha_2| = k + 1 - i$ , we have

$$[\mathcal{L}_D, \partial_x^{\alpha_1} \partial_v^{\alpha_2}]h(x, v) = \gamma \sum_{j=1}^d \alpha_{2,j} \partial_v^{\alpha_2} \partial_x^{\alpha_1} h = \gamma (k+1-i) \partial_x^{\alpha_1} \partial_v^{\alpha_2} h$$

and

$$\nabla_x^i \nabla_v^{k+1-i} h \cdot [\mathcal{L}_D, \nabla_x^i \nabla_v^{k+1-i}] h = \gamma(k+1-i) |\nabla_x^i \nabla_v^{k+1-i} h|^2.$$

Similarly, for all  $\beta \in \mathbb{N}^d$  and  $i \in \llbracket 1, d \rrbracket$ ,

$$\begin{aligned} (\partial_x^\beta \partial_{x_i} - \partial_x^\beta \partial_{v_i}) \mathcal{L}_D h(x, v) &= -\gamma \sum_{j=1}^d v_i \partial_x^\beta \partial_{x_i} \partial_{v_j} h + \gamma \sum_{j=1}^d \partial_{v_i} (v_j \partial_x^\beta \partial_{v_j} h) + \gamma \sum_{j=1}^d \partial_x^\beta \partial_{x_i} \partial_{v_j}^2 h - \gamma \sum_{j=1}^d \partial_x^\beta \partial_{v_i} \partial_{v_j}^2 h \\ &= -\gamma \sum_{j=1}^d v_j \partial_x^\beta \partial_{x_i} \partial_{v_j} h + \gamma \sum_{j=1}^d v_j \partial_{v_i} \partial_x^\beta \partial_{v_j} h + \gamma \partial_x^\beta \partial_{v_i} h \\ &+ \gamma \sum_{j=1}^d \partial_x^\beta \partial_{x_i} \partial_{v_j}^2 h - \gamma \sum_{j=1}^d \partial_x^\beta \partial_{v_i} \partial_{v_j}^2 h \end{aligned}$$

Therefore,

$$[\mathcal{L}_D, \partial_x^\beta \partial_{x_i} - \partial_x^\beta \partial_{v_i}]h(x, v) = \gamma \partial_x^\beta \partial_{v_i}h(x, v)$$

and finally applying Young inequality

$$\begin{aligned} (\nabla_x^{k+1} - \xi \nabla_x^k \nabla_v) h \cdot [\mathcal{L}_D, \nabla_x^{k+1} - \nabla_x^k \nabla_v] h &= \gamma \sum_{|\beta|=k; j \in \llbracket 1, d \rrbracket} (\partial_x^\beta \partial_{x_j} - \partial_x^\beta \partial_{v_j}) h \partial_x^\beta \partial_{v_j} h \\ &\geqslant \frac{1}{4} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v) h|^2 - C |\nabla_x^k \nabla_v h|^2 \,, \end{aligned}$$
or some  $C > 0.$ 

for some C > 0.

Finally, let us look at the jump part.

**Lemma 3.2.6.** For all  $k \in \mathbb{N}$ , there exists C > 0 such that for  $i \in [[0, k]]$  and  $\varepsilon > 0$ ,

$$\begin{split} \int_{\mathbb{R}^{2d}} \nabla_x^i \nabla_v^{k+1-i} h \cdot [\mathcal{L}_J, \nabla_x^i \nabla_v^{k+1-i}] h \, d\mu \geqslant -\frac{1}{\varepsilon} \int_{\mathbb{R}^{2d}} |\nabla_x^i \nabla_v^{k+1-i} h|^2 \, d\mu \\ &- C\varepsilon \int_{\mathbb{R}^{2d}} \left( \sum_{\substack{j \leqslant i \\ l \leqslant k+1-j}} |\nabla_x^j \nabla_v^l h|^2 + |\nabla_x^i \nabla_v^{k+2-i} h|^2 \right) \, d\mu \, . \end{split}$$

and

$$\begin{split} \int_{\mathbb{R}^{2d}} (\nabla_x^{k+1} - \nabla_k^k \nabla_v h) \cdot [\mathcal{L}_J, \nabla_x^{k+1} - \nabla_x^k \nabla_v] h \, d\mu \geqslant -\frac{1}{8} \int_{\mathbb{R}^{2d}} |\nabla_x^{k+1} - \nabla_x^k \nabla_v h|^2 \, d\mu \\ - C \left( \int_{\mathbb{R}^{2d}} \sum_{\substack{i \leqslant k \\ j \leqslant k+1-i}} |\nabla_x^i \nabla_v^j h|^2 \, d\mu + \int_{\mathbb{R}^{2d}} |\nabla_x^k \nabla_v^2 h|^2 \, d\mu \right) \, . \end{split}$$

To prove this Lemma, we need a generalization of Proposition 3.2.3:

**Lemma 3.2.7.** Let  $n \in \mathbb{N}^*$ . There exists C > 0 such that for any  $h \in \mathcal{C}^{\infty}_{c}(\mathbb{R})$ ,

$$\int_{\mathbb{R}} v^{2n} h^2(v) e^{-v^2/2} dv \leqslant C \sum_{k=0}^n \int_{\mathbb{R}} (h^{(k)}(v))^2 e^{-v^2/2} dv.$$

*Proof.* We prove the inequality by induction on n. The case n = 1 corresponds to Lemma 3.2.3. Suppose that the inequality holds for some  $n \in \mathbb{N}$ . Then

$$\begin{split} \int_{\mathbb{R}} v^{2n+2} h^2(v) e^{-v^2/2} \mathrm{d}v &= \int_{\mathbb{R}} v^{2n} (vh(v))^2 e^{-v^2/2} \mathrm{d}v \\ &\leqslant C \sum_{k=0}^n \int_{\mathbb{R}} ((vh(v))^{(k)})^2 e^{-v^2/2} \mathrm{d}v \\ &= C \sum_{k=0}^n \int_{\mathbb{R}} (kh^{(k-1)} + vh^{(k)})^2 e^{-v^2/2} \mathrm{d}v \\ &\leqslant C \sum_{k=0}^n 2 \int_{\mathbb{R}} (k^2 (h^{(k-1)})^2 + v^2 (h^{(k)})^2) e^{-v^2/2} \mathrm{d}v \\ &\leqslant 2C n^2 \sum_{k=0}^{n-1} \int_{\mathbb{R}} (h^{(k)}(v))^2 e^{-v^2/2} \mathrm{d}v + 4C \sum_{k=0}^n \int_{\mathbb{R}} (h^{(k)}(v))^2 e^{-v^2/2} \mathrm{d}v \end{split}$$

$$\begin{split} &+8C\sum_{k=1}^{n+1}\int_{\mathbb{R}}(h^{(k)})^2e^{-v^2/2}\mathrm{d}v\\ &\leqslant \widetilde{C}\sum_{k=0}^{n+1}\int_{\mathbb{R}}(h^{(k)})^2e^{-v^2/2}\mathrm{d}v\,, \end{split}$$

and hence it holds for n + 1.

Proof of Lemma 3.2.6. Recall that we denote

$$g_i(x, v, G) = (x, v - \frac{2\theta}{1 + \theta^2} \left(\theta v_i + G\right) e_i), \quad \varphi_i(x, v, G) = \frac{\theta \partial_i U_1(x)}{1 + \theta^2} \left(\theta v_i + G\right).$$

The functions  $(g_i)_i$  are smooth and we have for all  $i \in [\![1,d]\!]$ 

$$\partial_{v_j}g_i = \left(0, e_j - \frac{2\theta^2}{1+\theta^2}e_i frm[o] - -_{j=i}\right), \quad \partial_{x_j}g_i = (e_j, 0), \quad \partial_x^{\nu}\partial_x^{\alpha}g_i = 0 \quad \forall \left|\nu\right| + \left|\alpha\right| \geqslant 2.$$

As a consequence, for any  $\nu_1, \nu_2 \in \mathbb{N}^d$ , we have

$$\partial_x^{\nu_1} \partial_v^{\nu_2}(h(g_i)) = \left(1 - \frac{2\theta}{1 + \theta^2}\right)^{\nu_{2,i}} (\partial_x^{\nu_1} \partial_v^{\nu_2} h)(g_i).$$

The functions  $(\varphi_i)_i$  are smooth and we have for all  $i \in [\![1,d]\!]$ 

$$\partial_x^{\nu}\varphi_i = \frac{\theta}{1+\theta^2}(\theta v_i + G)\partial^{\nu}\partial_i U_1, \quad \partial_x^{\nu}\partial_{v_i}\varphi_i = \frac{\theta^2}{1+\theta^2}(\partial^{\nu}\partial_i U_1).$$

This yields that the only non-vanishing derivatives of  $\Psi(\varphi_i)$  are of the type  $\partial_x^{\alpha} \partial_{v_i}^k \Psi(\varphi_i)$ . For  $k \in \mathbb{N}$  and  $\alpha \in \mathbb{N}^d$ , we have:

$$\partial_{v_i}^k \Psi(\varphi_i) = \Psi^{(k)}(\varphi_i) \left(\frac{\theta^2}{1+\theta^2}\right)^k (\partial_i U_1)^k,$$

and

$$\partial_x^{\alpha} \partial_{v_i}^k \Psi(\varphi_i) = \left(\frac{\theta^2}{1+\theta^2}\right)^k \sum_{\nu \leqslant \alpha} \binom{\alpha}{\nu} \partial_x^{\nu} \Psi^{(k)}(\varphi_i) \partial_x^{\alpha-\nu} (\partial_i U_1)^k.$$

Therefore, using that  $\Psi(s) \leq c + |s|$  for some c > 0, and that  $\Psi'$  and  $U_1$  and their derivatives of all order are bounded, we get that for all  $k \in \mathbb{N}$  and  $\alpha \in \mathbb{N}^d$  there exists C > 0 such that:

$$|\partial_x^{\alpha} \partial_{v_i^k} \Psi(\varphi_i)| \leqslant C \left( 1 + |\theta v_i + G|^{|\alpha|} \right) \leqslant C' \left( 1 + |v_i|^{|\alpha|} + |G|^{|\alpha|} \right) \,.$$

Thus, for  $\alpha = (\alpha_1, \alpha_2) \in \mathbb{N}^{2d}$ , we have

$$\begin{split} [\partial^{\alpha}, \mathcal{L}_{J}^{i}]h &= \frac{1+\theta^{2}}{\theta^{2}} \left( \mathbb{E} \left[ \partial^{\alpha} \left[ (h(g_{i})-h)\Psi(\varphi_{i}) \right] \right] - \mathbb{E} \left[ (\partial^{\alpha}h(g_{i})-\partial^{\alpha}h)\Psi(\varphi_{i}) \right] \right) \\ &= \frac{1+\theta^{2}}{\theta^{2}} \left( \sum_{\nu < \alpha} \binom{\alpha}{\nu} \mathbb{E} \left[ (\partial^{\nu}(h(g_{i}))-\partial^{\nu}h)\partial^{\alpha-\nu}(\Psi(\varphi_{i})) \right] + \mathbb{E} \left[ (\partial^{\alpha}(h(g_{i}))-\partial^{\alpha}h(g_{i}))\Psi(\varphi_{i}) \right] \right) \\ &= \frac{1+\theta^{2}}{\theta^{2}} \sum_{\nu < \alpha} \binom{\alpha}{\nu} \mathbb{E} \left[ \left( \left( 1 - \frac{2\theta}{1+\theta^{2}} \right)^{\nu,i+d} \partial^{\nu}h(g_{i}) - \partial^{\nu}h \right) \partial^{\alpha-\nu}(\Psi(\varphi_{i})) \right] \\ &+ \frac{1+\theta^{2}}{\theta^{2}} \mathbb{E} \left[ \left( \left( 1 - \frac{2\theta}{1+\theta^{2}} \right)^{\alpha,i+d} - 1 \right) \partial^{\alpha}h(g_{i})\Psi(\varphi_{i}) \right] . \end{split}$$

As a consequence, there exists a  ${\cal C}>0$  such that

$$\begin{split} |[\partial^{\alpha}, \mathcal{L}_{J}^{i}]h|^{2} &\leq C\mathbb{E}[(h(g_{i}) - h)^{2}(1 + |v_{i}|^{2|\alpha|} + G^{2|\alpha|})] \\ &+ C\left(\sum_{0 < \nu < \alpha} \mathbb{E}\left[(\partial^{\nu}h(g_{i}))^{2}(1 + |v_{i}|^{2|\alpha-\nu|} + G^{2|\alpha-\nu|})\right] + (\partial^{\nu}h)^{2}(1 + |v_{i}|^{2|\alpha-\nu|})\right) \\ &+ C\mathbb{E}[(\partial^{\alpha}h(g_{i}))^{2}(1 + v_{i}^{2} + G^{2})]. \end{split}$$

As in the  $H^1$  case, the integrals with respect to  $\mu$  of the previous expectations can be bounded using the Poincaré inequality for the Gaussian measure, the reversibility of the law of  $V^i$  with respect to the Gaussian measure, as well as Lemma 3.2.7. Recall that thanks to the reversibility (3.8),

$$\int_{\mathbb{R}^{2d}} \mathbb{E}[(\partial^{\nu} h(g_i))^2] \mathrm{d}\mu = \int_{\mathbb{R}^{2d}} (\partial^{\nu} h)^2 \mathrm{d}\mu$$

Recall that  $V_i^i = \rho v_i + \sqrt{1 - \rho^2} G$  where G is a random variable with standard normal distribution. Hence, for any  $n \in \mathbb{N}^*$ ,

$$\mathbb{E}[(V_i^i)^{2n}] = \sum_{k=0}^{2n} \binom{2n}{k} (1-\rho^2)^{k/2} \mathbb{E}[G^k] \rho^{2n-k} v_i^{2n-k} \leqslant C \left(1+v_i^{2n}\right) \,.$$

Similarly we have:

$$\mathbb{E}[(v_i - \rho V_i^i)^{2n}] = \sum_{k=0}^{2n} \binom{2n}{k} (-1)^k \rho^k \mathbb{E}[(V_i^i)^k] v_i^{2n-k} \leqslant C \left(1 + v_i^{2n}\right)$$

Therefore, for all  $\nu \in \mathbb{N}^{2d}$  and  $n \in \mathbb{N}^*$ :

$$\int_{\mathbb{R}^{2d}} |v_i|^{2n} \mathbb{E}[|\partial^{\nu} h(g_i)|^2] \mathrm{d}\mu = \int_{\mathbb{R}^{2d}} \mathbb{E}[V_i^{2n}] |\partial^{\nu} h|^2 \mathrm{d}\mu \leqslant C \int_{\mathbb{R}^{2d}} \left(1 + v_i^{2n}\right) |\partial^{\nu} h|^2 \mathrm{d}\mu \,,$$

and using Lemma 3.2.7, we get:

$$\int_{\mathbb{R}^{2d}} |v_i|^{2n} \mathbb{E}[|\partial^{\nu} h(g_i)|^2] \mathrm{d}\mu \leqslant C \sum_{k=0}^n \int_{\mathbb{R}^{2d}} |\partial_{v_i}^k \partial^{\nu} h|^2 \mathrm{d}\mu.$$

Similarly we have

$$\begin{split} \int_{\mathbb{R}^{2d}} \mathbb{E}[G^{2n}|\partial^{\nu}h(g_i)|^2] \mathrm{d}\mu &= \frac{1}{(1-\rho^2)^n} \int_{\mathbb{R}^{2d}} |\partial^{\nu}h|^2 \mathbb{E}[(v_i - \rho V_i^i)^{2n}] \mathrm{d}\mu \\ &\leqslant C \int_{\mathbb{R}^{2d}} \left(1 + v_i^{2n}\right) |\partial^{\nu}h|^2 \mathrm{d}\mu \leqslant C' \sum_{k=0}^n \int_{\mathbb{R}^{2d}} |\partial_{v_i}^k \partial^{\nu}h|^2 \mathrm{d}\mu \,. \end{split}$$

Now bound

$$\begin{split} \int_{\mathbb{R}^{2d}} v_i^{2|\alpha|} \mathbb{E}[(h(g_i) - h)^2] \mathrm{d}\mu &\leq 2 \left( \int_{\mathbb{R}^{2d}} v_i^{2|\alpha|} \mathbb{E}[(h(g_i) - \Pi_{v_i}h)^2] \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} v_i^{2|\alpha|} (h - \Pi_{v_i}h)^2 \mathrm{d}\mu \right) \\ &= 2 \left( \int_{\mathbb{R}^{2d}} \mathbb{E}[(V_i^i)^{2|\alpha|}] (h - \Pi_{v_i}h)^2 \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} v_i^{2|\alpha|} (h - \Pi_{v_i}h)^2 \mathrm{d}\mu \right) \\ &\leq C \int_{\mathbb{R}^{2d}} \left( 1 + v_i^{2|\alpha|} \right) (h - \Pi_{v_i}h)^2 \mathrm{d}\mu \,. \end{split}$$

The Poincaré inequality (3.10) and Proposition 3.2.7 then yields

$$\int_{\mathbb{R}^{2d}} v_i^{2|\alpha|} \mathbb{E}[(h(g_i) - h)^2] \mathrm{d}\mu \leqslant C \sum_{k=0}^{|\alpha|} \int_{\mathbb{R}^{2d}} (\partial_{v_i}^k h)^2 \mathrm{d}\mu \,,$$

for some C > 0. Similarly we have

$$\begin{split} \int_{\mathbb{R}^{2d}} \mathbb{E}[G^{2|\alpha|}(h(g_i) - h)^2] \mathrm{d}\mu &\leq 2 \left( \int_{\mathbb{R}^{2d}} \mathbb{E}[G^{2|\alpha|}(h(g_i) - \Pi_{v_i}h)^2] \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} \mathbb{E}[G^{2|\alpha|}](h - \Pi_{v_i}h)^2 \mathrm{d}\mu \right) \\ &\leq C \left( \int_{\mathbb{R}^{2d}} \mathbb{E}[(v_i - \rho V_i^i)^{2|\alpha|}](h - \Pi_{v_i}h)^2 \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} (h - \Pi_{v_i}h)^2 \mathrm{d}\mu \right) \\ &\leq C \sum_{k=0}^{|\alpha|} \int_{\mathbb{R}^{2d}} (\partial_{v_i}^k h)^2 \mathrm{d}\mu \,, \end{split}$$

for some constant C > 0. Putting everything together, we get that for all  $\alpha \in \mathbb{N}^{2d}$ , there is C > 0 such that

$$\int_{\mathbb{R}^{2d}} |[\partial^{\alpha}, \mathcal{L}_{J}^{i}]h|^{2} \mathrm{d}\mu \leqslant C \left( \int_{\mathbb{R}^{2d}} (\partial_{v_{i}} \partial^{\alpha} h)^{2} \mathrm{d}\mu + \sum_{\nu \leqslant \alpha} \sum_{k=0}^{|\alpha - \nu|} \int_{\mathbb{R}^{2d}} (\partial_{v_{i}}^{k} \partial^{\nu} h)^{2} \mathrm{d}\mu \right).$$

For any  $i \in [\![0, k]\!]$  and  $\varepsilon > 0$ , Young inequality yields

$$\begin{split} \int_{\mathbb{R}^{2d}} \nabla_x^i \nabla_v^{k+1-i} h \cdot [\mathcal{L}_J, \nabla_x^i \nabla_v^{k+1-i}] h \mathrm{d}\mu \geqslant -\frac{1}{\varepsilon} \int_{\mathbb{R}^{2d}} |\nabla_x^i \nabla_v^{k+1-i} h|^2 \mathrm{d}\mu \\ &- C\varepsilon \int_{\mathbb{R}^{2d}} \left( \sum_{\substack{j \leqslant i \\ l \leqslant k+1-j}} |\nabla_x^j \nabla_v^l h|^2 + |\nabla_x^i \nabla_v^{k+2-i} h|^2 \right) \mathrm{d}\mu \,. \end{split}$$

To treat the term with derivative  $\nabla_x^{k+1} - \nabla_k^k \nabla_v$ , write for  $\alpha \in \mathbb{N}^d$ 

$$[\partial_x^{\alpha}, \mathcal{L}_J^i]h = \frac{1+\theta^2}{\theta^2} \sum_{\nu < \alpha} \binom{\alpha}{\nu} \mathbb{E} \left[ (\partial_x^{\nu} h(g_i) - \partial_x^{\nu} h) \partial_x^{\alpha-\nu} (\Psi(\varphi_i)) \right] \,,$$

so that no derivatives of order k+2 appear, nor derivatives of the form  $\nabla_x^{k+1}$ , and we have

$$\int_{\mathbb{R}^{2d}} |[\partial_x^{\alpha}, \mathcal{L}_J^i]h|^2 \mathrm{d}\mu \leqslant C \int_{\mathbb{R}^{2d}} \sum_{\nu < \alpha} \sum_{k=0}^{|\alpha - \nu|} (\partial_{v_i}^k \partial_x^\nu h)^2 \mathrm{d}\mu.$$

Hence, the only derivatives of order k + 2 that appear in  $[\mathcal{L}_J, \nabla_x^{k+1} - \nabla_x^k \nabla_v]h$  come from the term  $[\mathcal{L}_J, \nabla_x^k \nabla_v]h$  that we treated above. Using Young inequality, we get

$$\begin{split} \int_{\mathbb{R}^{2d}} (\nabla_x^{k+1} - \nabla_x^k \nabla_v h) \cdot [\mathcal{L}_J, \nabla_x^{k+1} - \nabla_x^k \nabla_v] h \mathrm{d}\mu \\ \geqslant -\frac{1}{8} \int_{\mathbb{R}^{2d}} |\nabla_x^{k+1} - \nabla_x^k \nabla_v h|^2 \mathrm{d}\mu - C \left( \int_{\mathbb{R}^{2d}} \sum_{\substack{i \leqslant k \\ i \leqslant k+1-i}} |\nabla_x^i \nabla_v^j h|^2 \mathrm{d}\mu - \int_{\mathbb{R}^{2d}} |\nabla_x^k \nabla_v^2 h|^2 \mathrm{d}\mu \right) \,, \end{split}$$

which concludes the proof.

Combining all the commutator terms, we get that for  $i \in [\![0, k-1]\!]$  there exists C > 0 such that for all  $\varepsilon > 0$  small enough

$$\begin{split} \int_{\mathbb{R}^{2d}} \nabla_x^i \nabla_v^{k+1-i} h \cdot [\mathcal{L}, \nabla_x^i \nabla_v^{k+1-i}] h \mathrm{d}\mu \\ \geqslant &- \frac{C}{\varepsilon} \int_{\mathbb{R}^{2d}} |\nabla_x^i \nabla_v^{k+1-i} h|^2 \mathrm{d}\mu - C \int_{\mathbb{R}^{2d}} |\nabla_x^{i+1} \nabla_v^{k-i} h|^2 \mathrm{d}\mu - C \sum_{j=0}^{i-1} \int_{\mathbb{R}^{2d}} |\nabla_x^j \nabla_v^{k+2-i} h|^2 \mathrm{d}\mu \\ &- C\varepsilon \int_{\mathbb{R}^{2d}} \left( \sum_{\substack{j \leqslant i \\ l \leqslant k+1-j}} |\nabla_x^j \nabla_v^l h|^2 + |\nabla_x^i \nabla_v^{k+2-i} h|^2 \right) \mathrm{d}\mu \,, \end{split}$$

as well as

$$\begin{split} \int_{\mathbb{R}^{2d}} \nabla_x^k \nabla_v h \cdot [\mathcal{L}, \nabla_x^k \nabla_v] h d\mu \geqslant \\ &- C \left( \frac{1}{\varepsilon} + \frac{1}{\eta} \right) \int_{\mathbb{R}^{2d}} |\nabla_x^k \nabla_v h|^2 d\mu - \frac{\eta}{8} \int_{\mathbb{R}^{2d}} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v) h|^2 d\mu - C \sum_{j=0}^{k-1} \int_{\mathbb{R}^{2d}} |\nabla_x^j \nabla_v^2 h|^2 d\mu \\ &- C \varepsilon \int_{\mathbb{R}^{2d}} \left( \sum_{\substack{j \leq k \\ l \leq k+1-j}} |\nabla_x^j \nabla_v^l h|^2 + |\nabla_x^k \nabla_v^2 h|^2 \right) d\mu \,, \end{split}$$

and finally

$$\begin{split} &\int_{\mathbb{R}^{2d}} (\nabla_x^{k+1} - \nabla_k^k \nabla_v h) \cdot [\mathcal{L}, \nabla_x^{k+1} - \nabla_x^k \nabla_v] h \mathrm{d}\mu \\ &\geqslant \frac{1}{4} \int_{\mathbb{R}^{2d}} |\nabla_x^{k+1} - \nabla_x^k \nabla_v h|^2 \mathrm{d}\mu - C \int_{\mathbb{R}^{2d}} \sum_{\substack{i \leq k \\ j \leq k+1-i}} |\nabla_x^i \nabla_v^j h|^2 \mathrm{d}\mu - C \left( \sum_{j=0}^k |\nabla_x^j \nabla_v h|^2 + \sum_{j=0}^k |\nabla_x^j \nabla_v^2 h|^2 \right) \,. \end{split}$$

Thanks to those expressions, we are now able to prove Proposition 3.2.5.

Proof of Proposition 3.2.5. The fact that inequality (3.12) holds for k = 1 is proven in Lemma 3.2.1. Let  $k \in \mathbb{N}$ , and suppose that inequality (3.12) holds for such k and some  $\rho_k > 0$  and  $\omega_{i,p} > 0$ ,  $0 \leq i \leq p \leq k$ . Fix some  $\omega_{i,k+1} > 0$ ,  $0 \leq i \leq k+1$ ,  $\varepsilon > 0$  small enough so that the previous formulas for the commutators hold, and choose  $\eta = \omega_{k+1,k+1}/\omega_{k,k+1}$ . By assumption we have

$$\int_{\mathbb{R}^{2d}} \Gamma(h) \mathrm{d}\mu + \sum_{p=1}^{k+1} \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p}(h) + \omega_{p,p} \Gamma_{p,p}(h) \right) \mathrm{d}\mu$$
  
$$\geq \rho_k \left( \mathcal{N}_k(h) + \int_{\mathbb{R}^{2d}} \sum_{i=0}^k |\nabla_x^i \nabla_v^{k+1-i} h|^2 \mathrm{d}\mu \right) + \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^k \omega_{i,k+1} \Gamma_{i,k+1}(h) + \omega_{k+1,k+1} \Gamma_{k+1,k+1}(h) \right) \mathrm{d}\mu.$$

By using Proposition 3.2.1, the bounds on the commutators and the fact that

$$\begin{split} \Gamma(\nabla_x^i \nabla_v^j h) &\geqslant 2\gamma |\nabla_x^i \nabla_v^{j+1} h|^2, \\ \Gamma((\nabla_x^{k+1} - \nabla_x^k \nabla_v) h) &\geqslant 2\gamma |(\nabla_x^{k+1} \nabla_v - \nabla_x^k \nabla_v^2) h|^2 \geqslant \gamma |\nabla_x^{k+1} \nabla_v h|^2 - 2\gamma |\nabla_x^k \nabla_v^2 h|^2 \end{split}$$

the term or order k + 2 are bounded below by

$$2\gamma \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^{k+2-i} h|^2 + \gamma \omega_{k+1,k+1} |\nabla_x^{k+1} \nabla_v h|^2 - (C_1 + 2\gamma) \omega_{k+1,k+1} |\nabla_x^k \nabla_v^2 h|^2 - C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^{k+2-i} h|^2 + \gamma \omega_{k+1,k+1} |\nabla_x^{k+1} \nabla_v h|^2 - (C_1 + 2\gamma) \omega_{k+1,k+1} |\nabla_x^k \nabla_v^2 h|^2 - C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^{k+2-i} h|^2 + \gamma \omega_{k+1,k+1} |\nabla_x^{k+1} \nabla_v h|^2 - (C_1 + 2\gamma) \omega_{k+1,k+1} |\nabla_x^k \nabla_v^2 h|^2 - C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^{k+2-i} h|^2 + \gamma \omega_{k+1,k+1} |\nabla_x^k \nabla_v h|^2 - C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^{k+2-i} h|^2 + \gamma \omega_{k+1,k+1} |\nabla_x^k \nabla_v^2 h|^2 - C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^k h|^2 - C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_x^i \nabla_v^k h|^2 + C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_v^k \nabla_v^k \nabla_v^k h|^2 + C_1 \varepsilon \sum_{i=0}^{k} \omega_{i,k+1} |\nabla_v^k \nabla_v^k \nabla_v^k h|^2 + C_1 \varepsilon \sum_{i=0}^{k} \omega_{i$$

for some constant  $C_1 > 0$ . We may bound from below the terms of order k+1 for  $\varepsilon$  small enough by

$$\rho_k \sum_{i=0}^k |\nabla_x^i \nabla_v^{k+1-i}h|^2 - \sup_{0 \leqslant i \leqslant k+1} \omega_{i,k+1} \frac{2C_2}{\varepsilon} \sum_{i=0}^k |\nabla_x^i \nabla_v^{k+1-i}h|^2 - C_2 \frac{\omega_{k,k+1}^2}{\omega_{k+1,k+1}} |\nabla_x^k \nabla_v h|^2 + \frac{\omega_{k+1,k+1}}{8} |(\nabla_x^{k+1} - \nabla_x^k \nabla_v)h|^2 + \frac{\omega_{k+1}}{8} |(\nabla_x^k \nabla_v)h|^2 + \frac{\omega_{k+1}}{8} |($$

for some constant  $C_2 > 0$ . Finally, the term of order at most k may be bounded from below by

$$\left(\rho_k \inf_{i+j\leqslant k} \omega_{i,i+j} - 3C_3 \sup_{0\leqslant i\leqslant k+1} \omega_{i,k+1}\right) \sum_{i+j\leqslant k} |\nabla_x^i \nabla_v^j h|^2 \,,$$

for some constant  $C_3 > 0$ . Set  $C_4 = \max(C_1, C_2, C_3)$  (which is independent from  $\varepsilon$  and the weights  $\omega_{i,j}$ ). Now, choose  $\varepsilon$  and  $\omega_{i,k+1}$  for  $i \in [\![1, k+1]\!]$  such that

$$\frac{\omega_{k+1,k+1}}{\omega_{k,k+1}} \leqslant \frac{\gamma - C_4 \varepsilon}{2\gamma + C_4}, \quad \varepsilon < \gamma/C_4, \quad \omega_{i,k+1} < \varepsilon \rho_k/4C_4, \\ 1 \leqslant i \leqslant k-1, \quad \omega_{k,k+1} \left(\frac{2}{\varepsilon} + \frac{\omega_{k,k+1}}{\omega_{k+1,k+1}}\right) < \rho_k/(2C_4)$$

and

$$6C_4 \sup_{0 \leqslant i \leqslant k+1} \omega_{i,k+1} < \rho_k \inf_{i+j \leqslant k} \omega_{i,i+j}$$

We get

$$\int_{\mathbb{R}^{2d}} \Gamma(h) \mathrm{d}\mu + \sum_{p=1}^{k+1} \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} \Gamma_{i,p}(h) + \omega_{p,p} \Gamma_{p,p}(h) \right) \mathrm{d}\mu$$

$$\geqslant \tilde{\rho}_{k+1} \left( \sum_{p=1}^{k+1} \int_{\mathbb{R}^{2d}} \left( \sum_{i=0}^{p-1} \omega_{i,p} |\nabla_x^i \nabla_v^{p-i} f|^2 + \omega_{p,p} |(\nabla_x^p - \nabla_x^{p-1} \nabla_v) f|^2 \right) + \sum_{i=0}^k |\nabla_x^i \nabla_v^{k+2-i} h|^2 \mathrm{d}\mu \right) + \sum_{i=0}^k |\nabla_x^i \nabla_v^{k+2-i} h|^2 \mathrm{d}\mu$$

for any

$$\tilde{\rho}_{k+1} < \min\left(\frac{\rho_k}{2}, \frac{\omega_{k+1,k+1}}{8}, \gamma \inf_{i+j \le k+1} \omega_{i,i+j}\right)$$

As in the  $H^1(\mu)$  case, an application of the Poincaré inequality (3.9) yields inequality (3.12), which concludes the induction.

Proof of Theorem 3.2.1. Our goal is to apply Lumer-Phillips theorem to the operator  $\mathcal{L} + \rho_k/2I$ , where I denotes the identity operator of  $H^k$ . This theorem can be stated as follows: an operator A on a Hilbert space generates a contraction semi-group if and only if it is maximally dissipative, see [169, Chapter IX, p.250]. Fix  $k \in \mathbb{N}$ . The scalar product

$$\langle f,g \rangle_k^2$$

$$= \int_{\mathbb{R}^{2d}} fg \mathrm{d}\mu + \int_{\mathbb{R}^{2d}} \sum_{p=1}^k \left( \sum_{i=0}^{p-1} \omega_{i,p} \nabla_x^i \nabla_v^{p-i} f \cdot \nabla_x^i \nabla_v^{p-i} g + \omega_{p,p} \left( \nabla_x^p - \nabla_x^{p-1} \nabla_v \right) f \cdot \left( \nabla_x^p - \nabla_x^{p-1} \nabla_v \right) g \right) \mathrm{d}\mu$$

generates a norm equivalent to the usual norm of  $H^k$ . Proposition 3.2.5 and a density argument yield that the operator  $\mathcal{L} + \rho_k/2I$ , is dissipative:

$$\forall h \in D(\mathcal{L}), \langle (\mathcal{L} + \rho_k/2I)h, h \rangle_k \leq 0,$$

where  $D(\mathcal{L})$  denote the domain of  $\mathcal{L}$  in  $H^k$  defined by:

$$f \in D(\mathcal{L}), g = \mathcal{L}f \Leftrightarrow f \in H^k, \lim_{t \to 0} \left\| \frac{P_t f - f}{t} - g \right\|_{H^k} = 0.$$

We are left to show that  $\mathcal{L} + \rho I$ , for some  $\rho < \rho_k/2$ , is surjective. Fix such a  $\rho < \rho_k/2$ . Thanks to Proposition 3.2.5, we have that

$$\Lambda: (f,g) \mapsto \langle -(\mathcal{L} + \rho I)f,g \rangle_k$$

is coercive and continuous from  $(H^k)^2$  to  $\mathbb{R}$  for  $k \ge 1$ . Hence, we may apply Lax-Milgram theorem to get that for all  $g \in H^k$ , there exists  $f \in H^k$  such that for all  $h \in H^k$ , we have:

$$\langle -(\mathcal{L}+\rho I)f,h\rangle_{H^k}=\langle -g,h\rangle_{H^k}\,,$$

which implies that f is a solution to the equation

$$(\mathcal{L} + \rho I)f = g,$$

and  $\mathcal{L} + \rho_k/2I$  is maximally dissipative. Lumer-Phillips Theorem then yields that the semi-group

generated by  $\mathcal{L} + \rho_k/2$  is a contraction on  $H^k$ : for all  $f \in H^k$ 

$$\mathcal{N}_k\left(e^{\rho_k t/2}(P_t f - \mu(f))\right) \leqslant \mathcal{N}_k\left(f - \mu(f)\right),$$

which concludes the proof.


CHAPTER 3. The velocity jump Langevin process

Part II Optimisation

## Chapter 4

# Convergence of the kinetic annealing for general potentials

#### Outline of the current chapter

4.1	Introduction and main results	93
4.2	Main steps of the proofs	97
	4.2.1 Return to a compact set	98
	4.2.2 Position in a compact set	100
	4.2.3 Localization and convergence	103
	4.2.4 Full process on a compact space	105
	4.2.5 Non-convergence with fast cooling schedules	108
4.3	Auxiliary results	109
	4.3.1 Uniform energy bounds	109
	4.3.2 Small time regularisation	112
4.4	$L^2$ -hypocoercivity	114
4.5	$H^k$ -hypocoercivity	123
4.6	Faster than logarithmic schedules	129

Abstract : The convergence of the kinetic Langevin simulated annealing is proven under mild assumptions on the potential U for slow logarithmic cooling schedules, which widely extends the scope of the previous results of [137]. Moreover, non-convergence for fast logarithmic and non-logarithmic cooling schedules is established. The results are based on an adaptation to non-elliptic non-reversible kinetic settings of a localization/local convergence strategy developed by Fournier and Tardif in [64] in the overdamped elliptic case, and on precise quantitative high order Sobolev hypocoercive estimates.

### 4.1 Introduction and main results

Given a potential  $U : \mathbb{R}^d \to \mathbb{R}$ , the goal of a simulated annealing procedure is to minimize U by designing a stochastic process  $(X_t)_{t \ge 0}$  whose law at time t is close to the probability density

proportional to  $e^{-\beta_t U(x)} dx$ , where  $\beta : \mathbb{R}_+ \to \mathbb{R}_+$  is called the cooling schedule. As  $\beta$  goes to infinity, this probability law concentrates around the global minimizers of U.

The most classical case is based on the overdamped Langevin process:

$$\mathrm{d}X_t = -\beta_t \nabla U(X_t) \mathrm{d}t + \sqrt{2} \mathrm{d}B_t \,,$$

where  $(B_t)_{t\geq 0}$  is a standard Brownian motion on  $\mathbb{R}^d$ . For a fixed  $\beta$ , the density  $e^{-\beta U(x)} dx$  is a stationary measure for this process. The idea is thus that, if  $\beta$  increases sufficiently slowly, the law of  $X_t$  gets and remains close to its instantaneous equilibrium. As a consequence, the convergence of the simulated annealing algorithm, in the sense of convergence in probability of  $U(X_t)$  toward min U as  $t \to +\infty$ , is related to the longtime convergence to equilibrium of the process at a fixed but high  $\beta$ . On the contrary, when  $\beta$  goes to infinity too fast, the algorithm is expected to fail with positive probability, i.e. the law of  $X_t$  is not expected to be close to  $e^{-\beta_t U}$ and  $U(X_t)$  to converge to min U.

On the other hand, the simulated annealing based on the kinetic Langevin process:

$$\begin{cases} dX_t = Y_t dt \\ dY_t = -\nabla U(X_t) dt - \beta_t Y_t dt + \sqrt{2} dB_t \end{cases}$$

was studied in [137]. The present work is concerned with the kinetic Langevin simulated annealing. The contributions with respect to [137] are the following. First, following the method of [64], the conditions on U are considerably weakened. Notice that, in the kinetic case, it means we can consider potentials that grow slower than those considered in [137], but also potentials that grow much faster (arbitrarily fast in fact), while the results of [137] require  $U(x)/(|x|^2+1)$  and  $\nabla^2 U$  to be bounded. Second, we prove the failure of the algorithm with fast cooling schedule, which was yet to be established in an hypocoercive case. Indeed, the failure of the overdamped Langevin simulated annealing is proven in [83] thanks to hypercontractivity results that are not available for the kinetic process, and the hypocoercive convergence results used in [137] to prove convergence are too weak to conclude about the failure of the algorithm in the fast cooling case (see the discussion at the beginning of Section 4.2.4). By proving the non-convergence of the algorithm under the same condition as in the overdamped case, we make rigorous the heuristic discussion in [137] according to which the kinetic process does not change the optimal condition on the cooling schedule. Third, as noted in [31], a technical truncation argument in [137], required for the rigorous computation of the modified entropy dissipation, is incorrect, and we have solved this issue (see Section 4.4 and more precisely Remark 4.4.1).

More precisely, we study the Markov process  $(Z_t)_{t\geq 0} = (X_t, Y_t)_{t\geq 0}$  on  $\mathbb{R}^{2d}$  that solves

$$\begin{cases} dX_t = Y_t dt \\ dY_t = -\nabla U(X_t) dt - \gamma_t Y_t dt + \sqrt{2\gamma_t \beta_t^{-1}} dB_t , \end{cases}$$

$$(4.1)$$

where  $\gamma : \mathbb{R}_+ \to \mathbb{R}_+$  is a friction parameter. We retrieve the settings of [137] with  $\gamma_t = \beta_t$ . We remark that the extension to Generalized Langevin processes as in [31] would not raise any particular difficulty, but we don't consider it for the sake of clarity. We will work under the following set of assumptions :

#### Assumption 4.1.1.

•  $U: \mathbb{R}^d \mapsto \mathbb{R}$  is a  $\mathcal{C}^{\infty}$  potential such that  $\min U = 0$ ,  $\lim_{|x| \to \infty} U(x) = \infty$ , and there exists

 $\alpha_0 > 0$  such that :

$$\int_{\mathbb{R}^d} e^{-\alpha_0 U(x)} \, dx < \infty$$

• The cooling schedule  $\beta : \mathbb{R}_+ \to \mathbb{R}_+$  is given by

$$\beta_t = \frac{\ln(e^{c\beta_0} + t)}{c},\tag{4.2}$$

for some parameters c > 0 and  $\beta_0 > 0$ .

- The friction  $\gamma : \mathbb{R}_+ \to \mathbb{R}_+$  is a  $\mathcal{C}^1$  function and there exists  $\kappa > 0$  such that for all  $t \ge 0$ ,  $\gamma_t \ge \kappa$  and  $\gamma'_t \le \frac{1}{\kappa(1+t)}$ . In particular there exists L such that  $\gamma_t \le L\beta_t$ .
- The critical height  $c^*$  of U is finite, where  $c^* = \sup_{x_1,x_2} c(x_1,x_2)$  and

$$c(x_1, x_2) = \inf \left\{ \max_{0 \le t \le 1} U(\xi(t)) - U(x_1) - U(x_2) \right\}$$

where the infimum runs over  $\{\xi \in \mathcal{C}([0,1], \mathbb{R}^d), \xi(0) = x_1, \xi(1) = x_2\}$ .

The condition  $\min U = 0$  is imposed for simplicity, it can always be enforced by changing U to  $U - \min U$ . The specific form of  $\beta$  is also made for simplicity, since it is known that, in order to study the convergence in probability for large time for simulated annealing algorithm, only logarithmic schedules are relevant. In particular, notice that, under Assumption 4.1.1, the time-shifted process  $(Z_{t_0+t})_{t\geq 0}$  for any  $t_0 \geq 0$  satisfies Assumption 4.1.1 with the same  $U, c, \kappa, L$  and with  $\beta_0$  replaced by  $\beta_{t_0}$ .

The critical height  $c^*$  represents the largest energy barrier the process has to cross in order to go from any local minimum to any global one. In the classical overdamped case, disregarding the question of the behavior of U at infinity, it is well known that, at least in the case where the global minimizer of U is unique, the algorithm converges if  $c > c^*$  (slow cooling) and has a positive probability to fail (i.e. to never visit a global minimum) if  $c < c^*$  (fast cooling). We retrieve this dichotomy in the kinetic case.

In the slow cooling case, we extend the results of [137]:

٢

**Theorem 4.1.1.** Under Assumption 4.1.1, assume furthermore that  $c > c^*$ . Then any solution  $(Z_t)_{t \ge 0}$  of (4.1) satisfies

$$\forall \delta > 0, \quad \mathbb{P}(H(Z_t) \ge \delta) \to 0.$$

Since  $H(x, v) = U(x) + |v|^2/2$ , this implies the convergence in probability of  $X_t$  to the set of global minimizers of U.

On the other hand, if  $c < c^*$ , then the process might remain stuck in a region that contains no global minimum of U. In fact, a slightly stronger condition is required. Indeed, it is possible that  $c^* > 0$  with all minima of U being global, in which case  $H(Z_t)$  may go to zero with fast cooling schedule, while the law of  $Z_t$  is not close to its local equilibrium. To be more precise, we need some additional definitions. For  $x, y \in \mathbb{R}^d$ , let

$$\tilde{c}(x,y) = \inf\left\{\max_{0 \le t \le 1} U(\xi(t)) - U(x)\right\}$$
(4.3)

where the infimum runs over  $\{\xi \in \mathcal{C}([0,1], \mathbb{R}^d), \xi(0) = x, \xi(1) = y\}$ . We define the depth of  $x \in \mathbb{R}^d$  by

$$\mathbf{D}(x) = \inf\left\{\tilde{c}(x,y), \, y \in \mathbb{R}^d, \, U(y) < U(x)\right\},\tag{4.4}$$



Figure 4.1: Case with two local non-global minima x and y at the same energy level. If c < b, then Assumption 4.1.2 holds by chosing either  $\tilde{x} = x$  or  $\tilde{x} = y$  and any  $a \in (c, b)$ . If  $c \in (b, D(x))$ , again Assumption 4.1.2 holds with  $\tilde{x} \in \{x, y\}$  and any  $a \in (b, D(x))$ . However, if c = b, Assumption 4.1.2 does not hold with  $\tilde{x} \in \{x, y\}$ .

with  $\inf \emptyset = \infty$ . It is clear that D(x) = 0 if x is not a local minimum of U. For  $x \in \mathbb{R}^d$  with D(x) > 0 and  $a \in (0, D(x))$ , we define the cup of bottom x and height a (this is the vocabulary of [80]) as

$$C(x,a) = \{ y \in \mathbb{R}^d, \ \tilde{c}(x,y) < a \}.$$

$$(4.5)$$

We want to discard pathological cases where there are two non-global local minima  $x_1, x_2$  with  $U(x_1) = U(x_2)$ ,  $D(x_1) = D(x_2) > c$  and  $c(x_1, x_2) = c$ , see Figure 4.1 (these cases do not prevent the result to hold, but the proof doesn't work directly, see Remark 4.1.1 below. Notice that the problem is not that there are two local minima with the same energy level and depth, which is a pretty common situation as soon as there are some symmetries in the system; the problem is that the elevation  $c(x_1, x_2)$  between them is *exactly* the parameter c chosen by the user in the cooling schedule which, now, is a very unlikely situation). Hence, we work under the following condition.

**Assumption 4.1.2.** Assumption 4.1.1 holds and there exist  $\tilde{x} \in \mathbb{R}^d$  with  $D(\tilde{x}) > c$  and  $a \in (c, D(\tilde{x}))$  such that for all  $y \in C(\tilde{x}, a)$ ,  $\tilde{c}(y, \tilde{x}) < c$ .

**Theorem 4.1.2.** Under Assumption 4.1.2, for all initial condition such that  $\mathbb{P}(X_0 \in C(\tilde{x}, c)) > 0$ and all  $\delta > 0$ , the solution of (4.1) is such that

$$\mathbb{P}\left(X_t \in \mathcal{C}(\tilde{x}, c+\delta) \ \forall t \ge 0\right) > 0.$$

If U has a finite number of minima and a unique global minimum, it is easily seen that there exists a non-global minimum x such that  $D(x) = c^*$ . More generally, since for all a < D(x) the minimum value of U over the cup  $C(\tilde{x}, a)$  is  $U(\tilde{x})$ , if there exists a non-global minimum x of U with depth D(x) > c, the previous results implies that, with positive probability,  $\inf_{t \ge 0} U(X_t) \ge U(\tilde{x}) > \min U$ . Moreover, even if the probability to start in  $C(\tilde{x}, c)$  is initially zero, due to the controllability of the process, it is positive for all positive times (see e.g. [137, proposition 5]). As a conclusion, we immediately get the following:

**Corollary 4.1.1.** Under Assumption 4.1.2, for all initial condition and all  $t_0 > 0$ ,

 $\mathbb{P}\left(U(X_t) \ge U(\tilde{x}) \ \forall t \ge t_0\right) > 0.$ 

Notice that, in practice, one can keep track of  $X_{s(t)}$  where

$$s(t) = \inf\{w \in [0, t], U(X_w) = \min_{w' \in [0, t]} U(X_{w'})\}$$

so that  $X_{s(t)}$  may converge to a minimizer of U even if  $X_t$  does not. However, our results show that this doesn't solve the issue of non-convergence for fast cooling schedules.

**Remark 4.1.1.** In Figure 4.1, in the case c = b (so that Assumption 4.1.2 do not hold) we cannot deduce from our results that the process stays stuck with positive probability in the cusp C(x,b) because, for any  $\delta > 0$ ,  $C(x,b+\delta)$  contains y (contrary to C(x,b)). In fact it is clear that the process can stay stuck with positive probability in  $C(x,b+\delta)$  for any  $\delta > 0$  (so that the conclusion of Corollary 4.1.1 also holds), but we cannot deduce it from our proof which requires that the critical depth within the cusp (i.e. for a suitable modification of the potential which only consider the local situation of this cusp, see Section 4.2.4 and Figure 4.2) is strictly smaller than c (in order to apply a variation of Theorem 4.1.1), while it is exactly b = c in this example. We refer to [130] where a fine analysis is conducted on a related question on finite graphs.

Finally we address the case of faster than logarithmic cooling schedules. For simplicity we restrict this study to the case of a constant friction parameter  $\gamma$  (although as discussed at the end of the proof it can be extended to the non-constant case with suitable conditions on  $\gamma$  depending on  $\beta$ ). In the following we do not assume that  $t \mapsto \beta_t$  is increasing, and possibly  $\beta_t = +\infty$  for some t.

**Theorem 4.1.3.** Assume that  $\gamma_t = \gamma$  is constant, that  $t \in \mathbb{R}_+ \mapsto \beta_t \in (0, +\infty]$  is piecewise continuous with  $\ln(t) = o(\beta_t)$  as  $t \to +\infty$ , that  $U \in \mathcal{C}^{\infty}(\mathbb{R}^d)$  and that  $x_* \in \mathbb{R}^d$  is a non-degenerate local minimum of U, i.e.  $\nabla U(x_*) = 0$  and  $\nabla^2 U(x_*) > 0$ . Then there exist C, r > 0 such that, denoting

$$\varepsilon_t = C\left(e^{-rt} + \sup_{s \ge t/2} \ln(s)\beta_s^{-1}\right),$$

the following holds. For all  $\delta > 0$  and all initial condition  $z_0 = (x_0, y_0) \in \mathbb{R}^{2d}$  with  $|x_0 - x_*| \leq \delta/2$ , the solution Z = (X, Y) of (4.1) is such that

$$\mathbb{P}_{z_0}\left(|X_t - x_*| \leq \min(\delta, \sqrt{\varepsilon_t}) \ \forall t \geq 0\right) > 0.$$

The rest of the paper is dedicated to the proofs of Theorems 4.1.1, 4.1.2 and 4.1.3. It is organised as follows. The main step of the proofs in the logarithmic case are exposed in Section 4.2, while technical intermediary results are postponed to Sections 4.3, 4.4 and 4.5. More precisely, Section 4.3.1 is dedicated to the proof of a uniform in time energy bound, which is the main ingredient in the proof that the process goes back infinitely many times to a compact set. A result of small-time conditional regularization is proven in Section 4.3.2, which is used to replace deterministic initial conditions by smooth distributions (with some quantitative bounds). Section 4.4 presents hypocoercivity estimates similar to those of [137], which are used to prove the convergence of the algorithm in the slow cooling case. In the fast logarithmic cooling case, similar estimates have to be established in higher order Sobolev norms, which is the topic of Section 4.5. Section 4.6 is dedicated to the faster than logarithmic case, with the proof of Theorem 4.1.3.

#### 4.2 Main steps of the proofs

The sketch of the proofs is the following.

The first point is to get uniform in time moment estimates. This would classically be done using Lyapunov arguments, but this would typically require some assumption on  $\nabla U$ , which we want to avoid. We adapt an argument from [64]. From these estimates, we get that  $\liminf_{t\to+\infty} H(Z_t)$  is almost surely finite, i.e. there exists a (random) compact set  $\{H \leq A\}$ which will be visited infinitely often by the process.

The second step is to prove that, for any A > 0, there exists A' > A such that, provided the process is in  $\{H \leq A\}$  at some time  $t_0$ , there is a probability at least, say, 1/4, that the process remains in  $\{H \leq A'\}$  for all times  $t \geq t_0$ . This is reminiscent of the study in [83] of fast cooling schedules, where it is proven that there is a positive probability that, starting in a potential well of depth larger than c, the process never climbs high enough to exit the well (as in Theorem 4.1.2). We will follow a similar proof, except that we have to check the dependency of the estimates with respect to  $t_0$  or, equivalently by taking  $t_0 = 0$ , to  $\beta_0$ . That way, we will conclude that, each time the process goes below A, it has a probability 1/4 to never go above A'again so that, if it goes below A infinitely often, eventually it will stay below A'.

Combining the two previous steps, we get that the process is almost surely bounded. It is thus sufficient to prove the convergence of the process when the position space is a compact torus, which is then similar to [137], but without the issue of the behavior of U at infinity.

The strategy for the non-convergence in the fast cooling case is similar to [83], the technical difficulties coming from the degeneracy of the process. Indeed the estimates used in [137] in the kinetic case to prove convergence are too weak to get that the process has a positive probability to stay forever below some energy level. On the other hand the estimates of [83] are based on hypercontractivity of elliptic diffusions on compact manifold. We are not aware of similar results for hypoelliptic diffusions, and thus we overcome this difficulty by working with higher Sobolev norms.

The faster than logarithmic case is similar and somehow simpler: in that case the process has a positive probability to converge to  $(x_*, 0)$ , it is thus sufficient to linearize  $\nabla U$  at this point and to study the corresponding Gaussian process.

#### 4.2.1 Return to a compact set

Diffusions defined by (4.1) are time-inhomogeneous Markov processes with generator :

$$L_t = y \cdot \nabla_x - (\gamma_t y + \nabla_x U) \cdot \nabla_y + \gamma_t \beta_t^{-1} \Delta_y.$$
(4.6)

First, let us check that the process is well-defined.

**Proposition 4.2.1.** Under Assumption 4.1.1, let  $z_0 \in \mathbb{R}^{2d}$ . There exists a unique process Z = (X, Y) that solves (4.1) with  $Z_0 = z_0$ . It is non-explosive (i.e. defined for all  $t \ge 0$ ) and, for all  $t \ge 0$ ,  $\mathbb{E}(H(Z_t)) \le H(z_0) + dLt$ .

*Proof.* Since the coefficients of the diffusion (4.1) are all smooth, there is existence and uniqueness until a time  $\xi$  of explosion. For all  $x, y \in \mathbb{R}^d$  and  $t \ge 0$ ,

$$L_t H(x, y) = \gamma_t \beta_t^{-1} d - \gamma_t |y|^2 \leq L d.$$

Considering for  $N \in \mathbb{N}$  the stopping time  $\tau_N = \inf \{t \ge 0; H(Z_t) \ge N\}$ , we get

$$\mathbb{E}(H(Z_{\tau_N \wedge t})) \leqslant H(z_0) + dLt$$

which implies

$$\mathbb{P}(\tau_N < t) = \mathbb{P}(H(Z_{\tau_N \wedge t}) \ge N) \leqslant \frac{H(z_0) + Ldt}{N} \xrightarrow[N \to \infty]{} 0,$$

hence  $\xi < +\infty$  almost surely, which concludes.

The first main point is to to strengthen the energy bound of Proposition 4.2.1 to a uniform in time estimate. For f a probability law or density on  $\mathbb{R}^{2d}$ , we write  $\mathbb{E}_f$  and  $\mathbb{P}_f$  expectations and probabilities with respect to the process Z solving (4.1) with initial condition  $Z_0$  distributed according to f. If  $f = \delta_z$  for some  $z \in \mathbb{R}^{2d}$ , we write  $\mathbb{E}_z$  and  $\mathbb{P}_z$  instead.

**Lemma 4.2.1.** Under Assumption 4.1.1, there exists  $b > \alpha_0$  that depends only on U such that the following holds. Provided  $\beta_0 \ge b$  then, for any  $C^{\infty}$  probability density  $f_0$  with compact support,

$$\sup_{t \ge 0} \mathbb{E}_{f_0} \left( H(Z_t) \right) \leqslant \frac{\kappa^{\beta_0}(f_0) + \ln(\mathcal{Z}_{\alpha_0})}{\beta_0 - \alpha_0}$$

where

$$\kappa^{\beta_0}(f_0) = \int_{\mathbb{R}^{2d}} f_0 \ln \left( 1 + f_0 e^{\beta_0 H} \right) \,, \qquad \mathcal{Z}_{\alpha_0} = \int_{\mathbb{R}^{2d}} e^{-\alpha_0 H}$$

The proof is postponed to Section 4.3.1. The next result enables the use of Lemma 4.2.1 when the initial condition is not smooth.

**Lemma 4.2.2.** Under Assumption 4.1.1 with  $c > c^*$ , fix A > 1 and  $\varepsilon > 0$ . Then there exist  $t^*, b_A, C_A^1, C > 0$  that do not depend on  $\beta_0$  such that, for all  $\beta_0 \ge b_A$  and  $z_0 \in \mathbb{R}^{2d}$  with  $H(z_0) \le A$ , the following holds. Writing  $\mathcal{B} = \{\sup_{t \le t^*} H(Z_t) \le A + 1\}$ ,

$$\mathbb{P}_{z_0}(\mathcal{B}) \ge 1 - \varepsilon.$$

Moreover, the law at time  $t^*$  of the process (4.1) with initial condition  $Z_0 = z_0$  conditioned on the event  $\mathcal{B}$  has a density  $f_{t^*}^c$  that satisfies

$$f_{t^*}^c \leqslant C e^{C_A^1 \beta_0} \mathbb{1}_{\{H \leqslant A+1\}}$$

This is proven in Section 4.3.2. The two previous lemmas yield the following.

**Proposition 4.2.2.** Under Assumption 4.1.1,  $\liminf_{t\to\infty} H(Z_t) < \infty$  almost surely.

*Proof.* For t large enough,  $\beta_t \ge b$  where  $b > \alpha_0$  is the constant from Lemma 4.2.1. Notice that, for  $t_0 \ge 0$ ,  $(Z_{t_0+t})_{t\ge 0}$  solves (4.1) except that  $\beta_0$  is replaced by  $\beta_{t_0}$  and  $\gamma$  is also time-shifted. Hence, by the Markov property, without loss of generality, we can assume that  $\beta_0 \ge b$ . Moreover, by conditioning with respect to the initial condition, it is sufficient to consider the case of a deterministic initial condition  $z_0 \in \mathbb{R}^{2d}$ .

Fix  $\varepsilon > 0$  and  $A = H(z_0)$ . From Lemma 4.2.2, there exist  $t^*, C' > 0$  such that  $\mathbb{P}_{z_0}(\mathcal{B}) \ge 1 - \varepsilon$ where  $\mathcal{B} = \{\sup_{t \le t^*} H(Z_t) \le A + 1\}$ , and such that the law of the process at time  $t^*$  and conditioned on  $\mathcal{B}$  has a density  $\tilde{f}_{t^*} \le C' \mathbb{1}_{\{H \le A + 1\}}$ . Let C > 0 and  $f_0$  be a  $\mathcal{C}^{\infty}$  probability density on  $\mathbb{R}^{2d}$  with compact support such that  $Cf_0 \ge C' \mathbb{1}_{\{H \le A + 1\}}$ . Then, denoting by  $(\mathcal{F}_t)_{t \ge 0}$ the filtration associated with  $(Z_t)_{t \ge 0}$ , by the strong Markov property and Lemma 4.2.1, for all  $t \ge 0$ ,

$$\mathbb{E}_{z_0}(H(Z_{t^*+t})|\mathcal{B}) = \mathbb{E}_{z_0}(\mathbb{E}(H(Z_{t^*+t})|\mathcal{F}_{t^*})|\mathcal{B}) \leqslant C\mathbb{E}_{f_0}(H(Z_t)) \leqslant C\frac{\kappa^{\beta_0}(f_0) + \ln(\mathcal{Z}_{\alpha_0})}{\beta_0 - \alpha_0}$$

As a consequence, by Fatou's Lemma,

$$\mathbb{E}_{z_0}\left(\liminf_{t\to\infty} H(Z_t)|\mathcal{B}\right) \leqslant \sup_{t\geqslant 0} \mathbb{E}_{z_0}(H(Z_t)|\mathcal{B}) < \infty.$$

Finally

$$\mathbb{P}_{z_0}\left(\liminf_{t\to\infty} H(Z_t) = \infty\right) \leqslant \mathbb{P}_{z_0}(\mathcal{B}^c) + \mathbb{P}_{z_0}\left(\liminf_{t\to\infty} H(Z_t) = \infty | \mathcal{B}\right) \leqslant \varepsilon,$$

which concludes since  $\varepsilon$  is arbitrary.

#### 4.2.2 Position in a compact set

As in [64], with Proposition 4.2.2 at hand, we can now focus on the behavior of the process when the position is in a compact set, ignoring the behavior of U at infinity. To this end, we fix some parameter K > 0, let  $L_K > 0$  be such that  $\{U \leq K\} \subset [-(L_K - 1), (L_K - 1)]^d$  and consider the torus  $M_K = (\mathbb{R}/2L_K\mathbb{Z})^d$  (i.e. we consider periodic boundary conditions, which will be technically simpler than e.g. reflecting boundary conditions). We now define a process  $(X_t^K, Y_t^K)$  on  $M_K \times \mathbb{R}^d$  and a potential  $U^K : M_K \mapsto \mathbb{R}$  to replace the initial one.

More precisely, write  $\theta_K : \mathbb{R}^d \mapsto M_K$  the canonical projection and  $\tilde{M}_K = [-L_K, L_K]^d$ , so that  $M_K = \theta_K(\tilde{M}_K)$ . For a non-negative function  $V : \mathbb{R}^d \to \mathbb{R}$ , we define the critical height  $c^*(V)$  with the same definition as  $c^*$  except that U is replaced by V.

Let  $\tilde{U}^K \in \mathcal{C}^{\infty}(\mathbb{R}^d)$  be equal to U on some open set  $O_K$  containing  $\{U \leq K\}$ , non-negative,  $2L_K$ -periodic, and such that  $c_K^* = c^*(\tilde{U}^K) \leq c^*$ . Such a function exists from [64, Notation 9]. We write  $U^K$  the corresponding function on  $M_K$ , given by  $U^K(\theta_K(x)) = \tilde{U}^K(x)$ , and  $H_K(x,y) = U^K(x) + |y|^2/2$ .

Finally, given the same Brownian motion as in (4.1), we consider  $Z^K = (X^K, Y^K)$  the process on  $M_K \times \mathbb{R}^d$  that solves

$$\begin{cases} \mathrm{d}X_t^K = Y_t^K \mathrm{d}t \\ \mathrm{d}Y_t^K = -\nabla_x U^K(X_t) \mathrm{d}t - \gamma_t Y_t^K \mathrm{d}t + \sqrt{2\gamma_t \beta_t^{-1}} \mathrm{d}B_t , \end{cases}$$
(4.7)

with  $(X_0^K, Y_0^K) = (\theta_K(X_0), Y_0)$ . Write  $\tau_K = \inf \{t \ge 0, X_t \notin O_K\}$ . Then, by design,

$$(U^{K}(X_{t}^{K}), Y_{t}^{K})_{t \leqslant \tau_{K}} = (U(X_{t}), Y_{t})_{t \leqslant \tau_{K}}.$$
(4.8)

For  $\beta > 0$ , write  $\mu_{\beta}^{K}$  the probability measure on  $M_{K} \times \mathbb{R}^{d}$  with density proportional to  $e^{-\beta H_{K}}$ . Lemma 4.2.3. Fix  $\delta, \alpha > 0$ . There exists C > 0 such that for all  $\beta > 0$ ,

$$\mu_{\beta}^{K} (H_{K} > \delta) \leqslant C e^{-\beta(\delta - \alpha)}.$$

Proof. For completeness, we recall the short proof of [137, section 3.1]. First,

$$\int_{M_K \times \mathbb{R}^d} e^{-\beta H_K(z)} \mathrm{d}z \ge \int_{\{H_K \leqslant \alpha\}} e^{-\beta H_K(z)} \mathrm{d}z \ge e^{-\beta \alpha}/C,$$

for some constant C > 0 because  $\{H^K \leq \alpha\}$  is compact. Likewise,  $\{\delta \leq H^K \leq \delta + |y|\}$  is compact and

$$\begin{split} \int_{\{H^K \ge \delta\}} e^{-\beta H_K(z)} \mathrm{d}z &\leqslant \int_{\{\delta \leqslant H_K \leqslant \delta + |y|\}} e^{-\beta H_K(z)} \mathrm{d}z + \int_{\{H_K \ge \delta + |y|\}} e^{-\beta H_K(z)} \mathrm{d}z \\ &\leqslant e^{-\beta \delta} \left( C + C \int_{\mathbb{R}^d} e^{-|y|} \mathrm{d}y \right) \end{split}$$

for some C > 0. The ratio of those two inequalities concludes the proof.

The goal of this section is to prove a similar result but with the law of  $Z_t^K$  instead of  $\mu_{\beta}^K$ , with an explicit dependency of the constants in term of  $\beta_0$ .

Denote by  $f_t^K$  the law of  $Z_t^K$ . Since  $\nabla U^K$  and its derivatives are all bounded, if  $f_0$  is smooth, so is  $f_t^K$  is for all  $t \ge 0$  (see Section 4.4 for details). Consider the relative density  $h_t^K = f_t^K / \mu_{\beta_t}^K$ . We start with a uniform in time quantitative bound of the norm of  $h_t^K$  in  $L^2\left(\mu_{\beta_t}^K\right)$ , for nice initial conditions.

**Proposition 4.2.3.** Under Assumption 4.1.1 with  $c > c^*$ , let  $K \ge 1$ . There exists  $b_K > 0$ , which depends on U and K, such that, if  $\beta_0 \ge b_K$  and  $f_0 \in \mathcal{C}^{\infty}(M_K \times \mathbb{R}^d)$  with compact support, then for all  $t \ge 0$ ,

$$\int_{M_K \times \mathbb{R}^d} (h_t^K)^2 d\mu_{\beta_t}^K \\ \leqslant 1 + \int_{M_K \times \mathbb{R}^d} \left( \left| (\nabla_x + \nabla_y) h_0^K \right|^2 + 4\sqrt{\gamma_0^{-1} \beta_0} (\|\nabla_x^2 U^K\|_{\infty} + 1 + \gamma_0)^2 (h_0^K)^2 \right) d\mu_0^K.$$

The proof of this proposition is postponed to Section 4.4.

**Lemma 4.2.4.** Under Assumption 4.1.1 with  $c > c^*$ , fix A > 1,  $\varepsilon = 1/4$  and consider  $\mathcal{B}, C_A^1, b_A, t^*$  as in Lemma 4.2.2. Set  $D_A = 2C_A^1 + A + 4 + 4c$  and  $K_A = D_A + 1$ . There exist  $C_A > 0$  and  $b'_A \ge b_A$  that do not depend on  $\beta_0$  such that, for all  $\beta_0 \ge b'_A$  and  $z_0 \in \mathbb{R}^{2d}$  with  $H(z_0) \le A$ , we have that,  $\mathbb{P}_{z_0}$ -a.s.,  $\sup_{t \in [0,t^*]} H_{K_A}\left(Z_t^{K_A}\right) \mathbb{1}_{\mathcal{B}} < D_A$ , and for all  $t \ge 0$ 

$$\mathbb{P}_{z_0}\left(\left.H_{K_A}\left(Z_{t+t^*}^{K_A}\right) \geqslant D_A\right|\mathcal{B}\right) \leqslant \frac{C_A}{(e^{c\beta_0}+t)^2}$$

*Proof.* The first point is clear since, by definition of  $\mathcal{B}$ ,  $\sup_{[0,t^*]} H_{K_A}(Z_t^{K_A}) \mathbb{1}_{\mathcal{B}} < A + 1 \leq D_A$ .

In the rest of the proof we denote by C different constants.

Let us introduce a function which will serve as a new initial condition. Let  $v_d$  and  $w_d$  be respectively the volumes of  $\{H \leq A + 1\}$  and  $\{H \leq A + 2\}$  in  $M_K \times \mathbb{R}^d$ , and  $f_0 \in \mathcal{C}^\infty$  be such that  $(2v_d)^{-1} \mathbb{1}_{\{H \leq A+1\}} \leq f_0 \leq 2(w_d)^{-1} \mathbb{1}_{\{H \leq A+2\}}$ . Notice that  $f_0$  does not depend on  $\beta_0$ , and that  $\nabla f_0$  is bounded since  $f_0$  is smooth on a compact set. As in the proof of Lemma 4.2.2, from Lemma 4.2.2 and the strong Markov property,

$$\mathbb{P}_{z_0}\left(H_{K_A}\left(Z_{t+t^*}^{K_A}\right) \geqslant D_A \mid \mathcal{B}\right) \leqslant 2v_d C e^{C_A^1 \beta_0} \mathbb{P}_{f_0}\left(H_{K_A}\left(Z_t^{K_A, \beta_{t^*}}\right) \geqslant D_A\right),$$

where  $Z^{K_A,\beta_{t^*}}$  is a process similar to  $Z^{K_A}$  except that  $\beta_0$  has been replaced by  $\beta_{t^*}$  and  $(\gamma_t)_{t\geqslant 0}$ by  $(\gamma_{t+t^*})_{t\geqslant 0}$ . Denote by  $h_t^*$  the relative density of the law of  $Z_t^{K_A,\beta_{t^*}}$  with respect to  $\mu_{\beta_{t+t^*}}^{K_A}$ . By the Cauchy-Schwartz inequality,

$$\mathbb{P}_{f_0}\left(H_{K_A}\left(Z_t^{K_A,\beta_{t^*}}\right) \ge D_A\right) = \int_{\{H_{K_A} \ge D_A\}} h_t^* \mathrm{d}\mu_{\beta_{t+t^*}}^{K_A}$$
$$\leqslant \sqrt{\int_{\mathbb{R}^d \times \mathbb{R}^d} (h_t^*)^2 \mathrm{d}\mu_{\beta_{t+t^*}}^{K_A}} \sqrt{\mu_{\beta_{t+t^*}}^{K_A} (H_{K_A} \ge D_A)}.$$
(4.9)

For the second term, applying Lemma 4.2.3 with  $\alpha = 1$ ,

$$\mu_{\beta_{t+t^*}}^{K_A}(H_{K_A} \ge D_A) \leqslant C e^{-\beta_{t+t^*}(D_A - 1)}$$

For the first term, Proposition 4.2.3 applies if  $\beta_0 \ge b_{K_A}$ , which yields

$$\begin{split} \int_{M_{K}\times\mathbb{R}^{d}} (h_{t}^{*})^{2} \,\mathrm{d}\mu_{\beta_{t+t}*}^{K_{A}} \\ &\leqslant 1 + \int_{M_{K}\times\mathbb{R}^{d}} \left( |(\nabla_{x} + \nabla_{y})h_{0}^{*}|^{2} + 4\sqrt{\gamma_{t^{*}}^{-1}\beta_{t^{*}}} (\|\nabla_{x}^{2}U^{K}\|_{\infty} + 1 + \gamma_{t^{*}})^{2} (h_{0}^{*})^{2} \right) \mathrm{d}\mu_{\beta_{t^{*}}}^{K_{A}} \\ &\leqslant C\beta_{t^{*}}^{\frac{5}{2}} \int_{\left\{ H_{K_{A}} \leqslant A+2 \right\}} \frac{1}{\mu_{\beta_{t^{*}}}^{K_{A}}(z)} \mathrm{d}z \end{split}$$

for some C > 0 that does not depend on  $\beta_0$ , where we used Assumption 4.1.1, that  $\beta_{t^*} \ge \beta_0$ ,  $h_0^K = f_0^K / \mu_{\beta_0}^K$ , and a uniform bound on  $f_0$  and  $\nabla f_0$ . For any  $\beta \ge 1$ , using that  $e^{-\beta U^{K_A}(x)} \le 1$ ,

$$\int_{M_{K_A}} e^{-\beta U^{K_A}(x)} \mathrm{d}x \leqslant (2L_{K_A})^d \qquad \text{and} \qquad \mathcal{Z}_\beta \int_{M_{K_A} \times \mathbb{R}^d} e^{-\beta H_{K_A}(z)} \mathrm{d}z \leqslant (2\sqrt{2\pi}L_{K_A})^d,$$

for all  $\beta > 0$ . Hence,

$$\int_{M_{K_A} \times \mathbb{R}^d} (h_t^*)^2 \, \mathrm{d}\mu_{\beta_{t+t^*}}^{K_A} \leqslant C \beta_t^{\frac{5}{2}} \mathcal{Z}_{\beta_{t^*}} \int_{\{H_{K_A} \leqslant A+2\}} e^{\beta_{t^*} H_{K_A}(z)} \mathrm{d}z \leqslant C' e^{\beta_{t^*} (A+3)} dz$$

Everything put together gives, if  $\beta_0 \ge \max(1, b_{K_A}, b_A)$ , using the monotonicity of  $t \mapsto \beta_t$ , we conclude with

$$\mathbb{P}\left(H_{K_A}\left(Z_{t+t^*}^{K_A}\right) \geqslant D_A \mid \mathcal{B}\right) \leqslant Ce^{-\frac{1}{2}\beta_{t^*+t}\left(D_A - 1 - 2C_A^1 - A - 3\right)} \leqslant Ce^{-2c\beta_t} = \frac{C}{(e^{c\beta_0} + t)^2}.$$

In fact, a similar proof already yields the convergence of the kinetic annealing on the compact torus:

**Proposition 4.2.4.** Under Assumption 4.1.1 with  $c > c^*$ , for all K > 1 and  $\delta > 0$ ,

$$\mathbb{P}(H_K(Z_t^K) > \delta) \xrightarrow[t \to +\infty]{} 0$$

Proof. As in the proof of Proposition 4.2.2, by the Markov property, without loss of generality we assume that  $\beta_0 \ge b_K$  and that the initial condition is a Dirac mass at some  $z_0 \in M_K \times \mathbb{R}^d$ . Fix  $\varepsilon > 0$ , let  $A = H(z_0)$  and, from Lemma 4.2.2 (applied to the process  $Z^K$  rather than Z, the proof is the same) let  $t_0, C > 0$  be such that the event  $\mathcal{B} = \{\forall t \in [0, t_0], H(Z^K) \le A + 1\}$ has a probability larger than  $1 - \varepsilon$  and the law of  $Z_{t_0}^K$  conditioned on  $\mathcal{B}$  has a density  $\tilde{f}_{t_0}^K \le C' \mathbb{1}_{\{H_K \le A+1\}}$ . Let C > 0 and  $f_0$  be a smooth probability density on  $M_K \times \mathbb{R}^d$  with compact support such that  $C' \mathbb{1}_{\{H_K \le A+1\}} \le Cf_0$ . We then have:

$$\mathbb{P}\left(H_{K}(Z_{t+t_{0}}^{K}) > \delta \mid \mathcal{B}\right) = \mathbb{E}\left(\mathbb{P}\left(H_{K}(Z_{t+t_{0}}^{K}) > \delta \mid \mathcal{F}_{t_{0}}\right) \mid \mathcal{B}\right) \leqslant C\mathbb{P}_{f_{0}}\left(H_{K}(\tilde{Z}_{t}) > \delta\right),$$

where  $\tilde{Z}$  is a process similar to  $Z^{K_A}$  except that  $\beta_0$  has been replaced by  $\beta_{t_0}$  and  $\gamma$  has been timeshifted. Denoting by  $\tilde{f}_t$  the law of  $\tilde{Z}_t$  (with initial condition  $f_0$ ), Proposition 4.2.3 means that  $t \mapsto \|\tilde{f}_t/\mu_{\beta_{t+t_0}}^K\|_{L^2(\mu_{\beta_{t+t_0}}^K)}$  is bounded. As in the previous proof, applying the Cauchy-Schwartz inequality and Lemma 4.2.3 with  $\alpha = \delta/2$  then yields :

$$\mathbb{P}_{f_0}^2\left(H_K(\tilde{Z}_t) > \delta\right) \leqslant C\mu_{\beta_{t+t_0}}^K(H_K > \delta) \leqslant Ce^{-\frac{\beta_t\delta}{2}} \to 0.$$

As a consequence,

$$\limsup_{t\to\infty} \mathbb{P}(H_K(Z_t^K) > \delta) \leqslant \mathbb{P}(\mathcal{B}^c) + \limsup_{t\to\infty} \mathbb{P}\left(H_K(Z_t^K) > \delta \mid \mathcal{B}\right) \leqslant \varepsilon,$$

which concludes since  $\varepsilon$  is arbitrary.

#### 4.2.3 Localization and convergence

Building upon the results of the previous section, we can now prove the following.

**Proposition 4.2.5.** Under Assumption 4.1.1 with  $c > c^*$ , fix some A > 1. There exist  $b_A > 1$ ,  $K_A > A$  which depends on A, U, c, and  $\kappa$  but not  $\beta_0$  such that, for all  $\beta_0 \ge b_A$  and all initial condition  $z_0 \in \{H \le A\}$ ,

$$\mathbb{P}_{z_0}\left(\sup_{t\geq 0}H\left(Z_t\right)\leqslant K_A\right)\geqslant \frac{1}{4}.$$

*Proof.* It is enough to show the same result for the process  $Z^{K_A}$  since, from (4.8),

$$\begin{split} &\left\{\sup_{t\geq 0} H(Z_t)\leqslant K_A\right\}\\ &=\left\{\sup_{t\geq 0} H(Z_t)\leqslant K_A, (U^K(X_t^{K_A}),Y_t^{K_A})_{t\leqslant \tau_{K_A}}=(U(X_t),Y_t)_{t\leqslant \tau_{K_A}}, \tau_{K_A}=\infty\right\}\\ &=\left\{\sup_{t\geq 0} H_{K_A}(Z_t^{K_A})\leqslant K_A, (U^K(X_t^{K_A}),Y_t^{K_A})_{t\leqslant \tau_{K_A}}=(U(X_t),Y_t)_{t\leqslant \tau_{K_A}}, \tau_{K_A}=\infty\right\}\\ &=\left\{\sup_{t\geq 0} H_{K_A}(Z_t^{K_A})\leqslant K_A\right\}. \end{split}$$

We use the definitions and notations of Lemma 4.2.4. Let  $\Psi_A \in \mathcal{C}^{\infty}(\mathbb{R}_+, [0, 1])$  equal to 0 outside of  $[D_A, K_A]$  and such that  $\Psi_A\left(\frac{D_A+K_A}{2}\right) = 1$ , and let  $\Phi_A = \Psi_A \circ H^{K_A}$ . Then, there exists C > 0, independent of  $\beta$ , such that, for all  $t \ge 0$ ,  $|L_t \Phi_A| \le C(1 + \gamma_t) \mathbb{1}_{\{H^{K_A} \ge D_A\}}$ .

Recall the definition  $\mathcal{B} = \{ \sup_{t \leq t^*} H(Z_t) \leq A + 1 \}$ . From Ito's formula,

$$\Phi_A\left(Z_{t+t^*}^{K_A}\right)\mathbb{1}_{\mathcal{B}} = \left(M_t + R_t\right)\mathbb{1}_{\mathcal{B}}$$

where  $M_t$  is a local martingale and  $R_t = \int_{t^*}^t L_s \Phi_A(Z_s^{K_A}) ds$ . We then get that for  $\beta_0 \ge b'_A$ ,

$$\mathbb{E}\left(\sup_{t\geq 0}|R_t|\mathbb{1}_{\mathcal{B}}\right) \leqslant C \int_0^\infty (1+\gamma_s) \mathbb{E}\left(\mathbb{1}_{\left\{H^{K_A}\left(Z_{s+t^*}^{K_A}\right) \ge D_A\right\}} \middle| \mathcal{B}\right) \mathrm{d}s$$
$$\leqslant C \int_0^\infty (1+L\beta_s) \mathbb{P}\left(H\left(Z_{s+t^*}^{K_A}\right) \ge D_A \middle| \mathcal{B}\right) \mathrm{d}s$$
$$\leqslant C \int_0^\infty \frac{1+L\ln(e^{c\beta_0}+t)}{c(e^{c\beta_0}+t)^2} \mathrm{d}s.$$

We take  $\beta_0$  large enough to get that, by Markov's inequality,  $\mathbb{P}(\mathcal{E}|\mathcal{B}) \geq 3/4$ , where  $\mathcal{E} = \{\sup_{t\geq 0} |R_t| \leq 1/10\}$ . Now,

$$\mathcal{E} \cap \mathcal{B} \cap \left\{ \sup_{t \ge 0} H_{K_A}(Z_t^{K_A}) \ge K_A \right\} \subset \mathcal{E} \cap \mathcal{B} \cap \left\{ \sup_{t \ge 0} M_t \ge \frac{9}{10} \right\},$$

so that

$$\mathbb{P}\left(\sup_{t\geq 0}H_{K_{A}}\left(Z_{t}^{K_{A}}\right)\geq K_{A}\mid \mathcal{B}\right)\leqslant \mathbb{P}\left(\mathcal{E}^{c}\mid \mathcal{B}\right)+\mathbb{P}\left(\mathcal{E}, M_{t} \text{ up-crosses } \left[\frac{1}{10}, \frac{9}{10}\right]\mid \mathcal{B}\right)$$

Consider the stopping time

$$\sigma = \inf\left\{t \ge 0, M_t \notin \left[-\frac{1}{10}, \frac{11}{10}\right]\right\}$$

Then  $M_{t \wedge \sigma}$  is a bounded local martingale, hence a martingale. Moreover, since  $\Phi_A$  takes values in [0, 1],  $M_t = M_{t \wedge \sigma}$  for all  $t \ge 0$  on  $\mathcal{E} \cap \mathcal{B}$ . By Doob's up-crossing inequality, for any T > 0, the probability that  $M_t$  up-crosses [1/10, 9/10] before time T is bounded by  $\mathbb{E}(M_{T \wedge \sigma} - \frac{1}{10})/(8/10) \le \frac{1}{4}$ . As a consequence,

$$\mathbb{P}\left(\mathcal{E}, M_t \text{ up-crosses } \left[\frac{1}{10}, \frac{9}{10}\right] \mid \mathcal{B}\right) \leqslant \frac{1}{4}.$$

As a conclusion,

$$\mathbb{P}\left(\sup_{t\geq 0}H_{K_{A}}\left(Z_{t}^{K_{A}}\right)\geq K_{A}\right)$$
$$\leqslant \mathbb{P}\left(\mathcal{B}^{c}\right)+\mathbb{P}\left(\mathcal{E}^{c}|\mathcal{B}\right)+\mathbb{P}\left(\mathcal{E},M_{t} \text{ up-crosses } \left[\frac{1}{10},\frac{9}{10}\right] \mid \mathcal{B}\right)\leqslant \frac{3}{4}.$$

As announced, combining this result with Proposition 4.2.2 yields the following.

**Proposition 4.2.6.** Under Assumption 4.1.1 with  $c > c^*$ , almost surely,  $\sup_{t \ge 0} H(Z_t) < +\infty$ . *Proof.* For A > 0, let

$$\Omega_A = \left\{ \liminf_{t \to \infty} H(Z_t) < A \right\}.$$

It is sufficient to show that, for all A > 0,  $(Z_t)_{t \ge 0}$  is bounded on  $\Omega_A$  since, from Proposition 4.2.2,  $\mathbb{P}(\bigcup_{A \ge 0} \Omega_A) = 1$ . Hence, we fix A > 0.

Let  $b_A, K_A$  be as in Proposition 4.2.5 and  $S_0 = t_A$  where  $t_A$  satisfies  $\beta_{t_A} \ge b_A$ . By induction, for  $k \in \mathbb{N}$ , define  $T_k = \inf \{t \ge S_k; H(Z_t) \le A\}$  and  $S_{k+1} = \inf \{t \ge T_k; H(Z_t) \ge K_A\}$ . From Proposition 4.2.5 and the Markov property,

$$\mathbb{P}(S_{k+1} < \infty | S_k < \infty) = \mathbb{P}(S_{k+1} < \infty, T_k < \infty | S_k < \infty)$$
$$= \mathbb{E}(\mathbb{1}_{T_k < \infty} \mathbb{P}(S_{k+1} < \infty | \mathcal{F}_{T_k}) | S_k < \infty) \leqslant \frac{3}{4}$$

where we used that  $\beta_{T_k} \ge \beta_{t_A} \ge b_A$  for all  $k \in \mathbb{N}$ . This implies that a.s. there exists  $J \in \mathbb{N}$  such that  $S_J = \infty$ . On  $\Omega_A$ ,  $T_J$  is finite and  $\sup_{t \ge T_J} H(Z_t) \le K_A$ , which concludes.

We can now finally prove Theorem 4.1.1.

Proof of Theorem 4.1.1. Let  $\varepsilon > 0$  and, thanks to Proposition 4.2.6, let K > 1 be such that  $\mathbb{P}\left(\sup_{t\geq 0} H(Z_t) \geq K\right) \leq \varepsilon$ . Using Proposition 4.2.4, for  $t \geq 0$ ,

$$\mathbb{P}(H(Z_t) > \delta) \leqslant \varepsilon + \mathbb{P}\left(\sup_{t \ge 0} H(Z_t) \leqslant K; H(Z_t) > \delta\right)$$
$$= \varepsilon + \mathbb{P}\left(\sup_{t \ge 0} H_K\left(Z_t^K\right) \leqslant K; H_K\left(Z_t^K\right) > \delta\right)$$
$$\leqslant \varepsilon + \mathbb{P}\left(H_K\left(Z_t^K\right) > \delta\right) \xrightarrow[t \to +\infty]{} \varepsilon.$$

As a consequence,  $\limsup_{t\to\infty} \mathbb{P}(H(Z_t) > \delta) \leq \varepsilon$  for all  $\varepsilon > 0$ , which concludes.

#### 4.2.4 Full process on a compact space

One of the main point of the proof of Theorem 4.1.2 is essentially to get something of the form

$$\forall \delta > 0, \qquad \mathbb{P}_{z_0}\left(\sup_{t \geqq 0} H(Z_t) < c + \delta\right) > 0$$

for suitable initial conditions  $z_0$ . First, let us highlight some of the difficulties in order to motivate the rest of this section. Refining the proof of Proposition 4.2.5, we would obtain a similar result but with c replaced by 4c. The factor 4 is due to two things. First, in Lemma 4.2.4, we prove a bound of order  $1/t^2$  while the proof of Proposition 4.2.5 in fact only requires an integrable bound, i.e.  $1/t^{1+\delta}$  for any  $\delta > 0$  is enough. The second factor 2 is lost when the Cauchy-Schwartz inequality is used in (4.9). To solve this, the Cauchy-Schwartz inequality has to be replaced by the Hölder inequality, which means the  $L^2$  estimate of Proposition 4.2.3 is not sufficient and we need  $L^p$  estimates for all p > 2, or even better,  $L^{\infty}$  estimates (besides, in [137], the convergence is stated in relative entropy, which is weaker than  $L^p$  for any p > 1, which is why we said in the introduction that these results were not sufficient to conclude in the fast cooling case). In fact in the elliptic case the proof of [83] relies on  $L^{\infty}$  bounds. In order to get such bounds in an hypocoercive case, we will work with  $H^k$ -Sobolev norms for  $k \ge 1$ , as in the work [170] of Zhang, and then use Sobolev embeddings. This should be done with a correct control of the dependency in time of the constants, and to do so it is convenient to work with a process (both position and velocity) in a compact manifold. This is done by replacing the Hamiltonian  $H(x,y) = U(x) + |y|^2/2$  by  $H_K(x,y) = U^K(x) + W(y)$  where W is some periodic potential with  $W(y) = |y|^2/2$  below some threshold. This raises an issue in the dissipation of the hypocoercive modified entropy. Indeed, in the modified  $H^1$ -norm of Villani (and similarly in the modified  $H^k$ -norm of Zhang), the key point is that the missing coercivity in x is recovered through a term

$$abla_x h \cdot [
abla_y, y 
abla_x] h = |
abla_x h|^2$$

where [A, B] = AB - BA stands for the commutator of A and B. When H is replaced by  $H_K$ , this term becomes

$$abla_x h \cdot [
abla_y, 
abla W(y) 
abla_x] h = 
abla_x h \cdot 
abla^2 W 
abla_x h$$

which is negative at maxima of W. For this reason, and since anyway we are not really interested in the process above some energy threshold, we will add some Brownian noise in the position variable where  $W(y) \neq |y|^2/2$ .

As a conclusion, for these reasons, in this section, we consider a process  $Z^{K} = (X^{K}, Y^{K})$  on



Figure 4.2: Construction of  $U^K$  from U, with  $\tilde{x}$  chosen so that  $c^*(U) = D(\tilde{x})$ .

a periodic torus  $M_K^2$  solution of

$$\begin{cases} \mathrm{d}X_t^K = \nabla W(Y_t^K) \mathrm{d}t - \sigma(Y_t) \nabla U^K(X_t^K) \mathrm{d}t + \sqrt{2\sigma(Y_t^K)\beta_t^{-1}} \mathrm{d}\tilde{B}_t \\ \mathrm{d}Y_t^K = -\nabla U^K(X_t^K) \mathrm{d}t - \gamma_t \nabla W(Y_t^K) \mathrm{d}t + \sqrt{2\gamma_t\beta_t^{-1}} \mathrm{d}B_t \end{cases}$$
(4.10)

where B is as in (4.1),  $\tilde{B}$  is another independent d-dimensional Brownian motion, and  $\sigma$ , W and  $U^K$  are  $2L_K$ -periodic non-negative functions on  $\mathbb{R}^d$  (with  $M_K = (\mathbb{R}/2L_K\mathbb{Z})^d$ ). The term  $\sigma \nabla U^K$  has been added so that, for fixed  $\gamma, \beta$ , this new process admits the explicit stationary measure

$$\mu_{\beta}^{K} \propto e^{-\beta H_{K}(x,y)} \mathrm{d}x \mathrm{d}y$$

that satisfies a Poincaré inequality (see Section 4.5). Here and in all this section, when there is no ambiguity we identify  $2L_K$ -periodic functions on  $\mathbb{R}^d$  and functions on  $M_K$ . Let us now give the precise definition of  $L_K, \sigma, W$  and  $U^K$ .

In all this section we consider fixed  $U \in C^{\infty}(\mathbb{R}^d)$ ,  $\beta$ ,  $\gamma$ ,  $\tilde{x} \in \mathbb{R}^d$  and a > 0 satisfying Assumption 4.1.2. Setting K = a + 1, similarly to Section 4.2.2 we fix some  $L_K > \sqrt{2K} + 2$  large enough so that  $C(\tilde{x}, a) \subset [-L_K + 1, L_K - 1]^d$ . We consider a non-negative  $U^K \in C^{\infty}(M_K)$  with  $c^*(U^K) < c$  and such that, seen as a periodic function on  $\mathbb{R}^d$ ,  $U^K(x) = U(x) - U(\tilde{x})$  for all  $x \in C(\tilde{x}, a)$ . Such a function exists: indeed, since  $\tilde{c}$  is continuous,  $C(\tilde{x}, a)$  is a compact set, and then, under Assumption 4.1.2,  $\sup\{\tilde{c}(y,\tilde{x}), y \in C(\tilde{x}, a)\} < c$ . We can thus choose  $\delta \in (0, a - c)$  small enough so that  $\sup\{\tilde{c}(y,\tilde{x}) : y \in C(\tilde{x}, a)\} + 2\delta < c$  and, using that the boundary of  $C(\tilde{x}, a)$  is in  $\{U = U(\tilde{x}) + a\}$ , we let  $U^K$  be any  $C^{\infty}$  potential on  $[-L_K, L_K]^d$  with  $U^K(x) = u(x) - U(\tilde{x})$  on  $C(\tilde{x}, a), U^K(x) \in [a - \delta, a + \delta]$  for  $x \in [-L_K, L_K]^d \setminus C(\tilde{x}, a)$  with  $U^K(x) = a$  for all  $x \in [-L_K, L_K]^d \setminus [-L_K + 1/2, L_K + 1/2]$  so that we can extend it to a  $2L_K$ -periodic  $C^{\infty}$  potential on  $\mathbb{R}^d$ . The fact that  $c^*(U^K) < c$  is straightforward since a path from any  $y \in [-L_K, L_K]^d$  to  $\tilde{x}$  can be obtained by a straight line from y to some  $y_* \in C(\tilde{x}, a)$  with  $\tilde{c}(y, y_*) \leq 2\delta$  and then a path to  $\tilde{x}$ , where Assumption 4.1.2 is used. See Figure 4.2 for an illustration of the construction of  $U^K$ .

Concerning the new potential W for the velocity, first, write  $n = \sqrt{2K} + 1$  and  $m = \sqrt{2K} + 2$ . Then, fix some  $2L_K$ -periodic  $W_1 \in \mathcal{C}^{\infty}(\mathbb{R}, \mathbb{R})$  such that for all  $s \in [-m, m]$ ,  $W_1(s) = s^2/2$ ,  $W_1$  is symmetric on  $[-L_K, L_K]$ , and increasing on  $[0, L_K]$ . For  $y \in [-L_K, L_K]^d$ , we set  $W(y) = \sum_{i=1}^d W_1(y_i)$ . Similarly, fix some  $2L_K$ -periodic non-negative  $\sigma^0 \in \mathcal{C}^{\infty}(\mathbb{R}^d, \mathbb{R}_+)$  with  $\sigma^0(y) = 0$  for  $y \in [-n, n]^d$ ,  $\sigma^0(y) = \sigma^0_*$  for  $y \in [-L_K, L_K]^d \setminus [-m, m]^d$  for some constant  $\sigma^0_* > 0$ , and  $|\nabla \sigma^0|^2 \leq C \sigma^0$  for some C > 0. Set  $\sigma = r\sigma^0$  where r > 0 is chosen small enough that  $\|\nabla \sigma\|_{\infty} \|\nabla U^K\|_{\infty} \leq \frac{1}{2}$  and  $|\nabla \sigma|^2 \leq \sigma$ .

The useful properties of W and  $\sigma$  can be summarized as follows :

**Lemma 4.2.5.** • Seeing W as a periodic function from  $\mathbb{R}^d$  to  $\mathbb{R}$ , there exists  $m \in (\sqrt{2K}, L_K)$ such that  $W(y) = |y|^2/2$  for all  $y \in [-m, m]^d$ .

- On  $M_K$ , W has a unique local minimum, which is global.
- There exist  $n \in (\sqrt{2K}, m)$  and  $\sigma_* > 0$  such that, seeing  $\sigma$  as a periodic function from  $\mathbb{R}^d$  to  $\mathbb{R}_+$ ,  $\sigma(y) = \sigma_*$  for all  $y \in [-L_K, L_K]^d \setminus [-m, m]^d$  and  $\sigma(y) = 0$  for all  $y \in [-n, n]^d$ .
- $\sigma$  satisfies  $\|\nabla \sigma\|_{\infty} \|\nabla U^K\|_{\infty} \leq 1/2$  and  $|\nabla \sigma(y)|^2 \leq \sigma(y)$  for all  $y \in M_K$ .

We write  $H_K(x, y) = U^K(x) + W(y)$ ,

$$\mathcal{M} = \left\{ y \in M_K; \nabla^2 W(y) = I_d \right\}$$

and notice that, by construction, for  $y \in M_K \setminus \mathcal{M}$ ,  $\sigma(y) = \sigma_*$ . Moreover, if  $|y|^2/2 \leq K$  then  $W(y) = |y|^2/2$  and  $\sigma(y) = 0$ .

By construction of  $U^K$ , W and  $\sigma$ , if we consider Z and  $Z^K$  the solutions respectively of (4.1) and (4.10) with the same initial condition in  $C(\tilde{x}, c)$  and the same Brownian motion B, then  $H(Z_t) = H_K(Z_t^K)$  for all  $t \leq \tau := \inf\{s \geq 0, H(Z_s) \geq a\}$ . In particular, for  $\delta > 0$  small enough so that  $c + \delta < \inf\{U^K(x), x \in [-L_K, L_K]^d \setminus C(\tilde{x}, a)\}$ ,

$$\{H_K(Z_t^K) < c + \delta \ \forall t \ge 0\} \subset \{X_t \in \mathcal{C}(\tilde{x}, c + \delta) \ \forall t \ge 0\},$$

$$(4.11)$$

which means we are lead to prove that the left hand side has a positive probability.

Denote by  $f_t^K$  the law of the solution of (4.10), write  $\mu_{\beta}^K = \mathcal{Z}_{\beta}^{K^{-1}} e^{-\beta H_K(z)} dz$  where  $\mathcal{Z}_{\beta}^K$  makes  $\mu_{\beta}^K$  a probability measure, and let  $h_t^K = f_t^K / \mu_{\beta_t}^K$ . Similarly to Section 4.2.3, the key point is the following estimate, proven in Section 4.5.

**Proposition 4.2.7.** Let  $f_0^K \in \mathcal{C}^{\infty}(M_K \times M_K)$ . Under Assumption 4.1.2,  $t \mapsto \|h_t^K\|_{\infty}$  is bounded.

**Lemma 4.2.6.** Under Assumption 4.1.2, for all probability density  $f_0^K \in C^{\infty}(M_K \times M_K)$  and  $\delta > 0$ , there exists C > 0 such that for all  $t \ge 0$ ,

$$\mathbb{P}_{f_0^K}\left(H_K\left(Z_t^K\right) \ge c + \delta\right) \le \frac{C}{(1+t)^{1+\frac{\delta}{2c}}}.$$

*Proof.* Applying Proposition 4.2.7, for  $t \ge 0$ 

$$\mathbb{P}\left(H_{K}\left(Z_{t}^{K}\right) \ge c+\delta\right) = \int_{\{H_{K} \ge c+\delta\}} h_{t}^{K} \mathrm{d}\mu_{\beta_{t}}^{K}$$
$$\leqslant \sup_{s \ge 0} \|h_{s}^{K}\|_{\infty} \mu_{\beta_{t}}^{K}\left(H_{K} \ge c+\delta\right) \leqslant \frac{C}{(1+t)^{(1+\frac{\delta}{2c})}}$$

for some C > 0, where we used Lemma 4.2.3 with  $\alpha = \delta/2$ .

**Lemma 4.2.7.** Under Assumption 4.1.2, for all probability density  $f_0^K \in C^{\infty}(M_K \times M_K)$  and  $\delta > 0$ , there exists  $t_b > 1$  such that

$$\mathbb{P}_{f_0^K}\left(\sup_{t \ge t_b} H_K(Z_t^K) < c + \delta\right) > 0.$$

*Proof.* It is enough to show the result for  $\delta$  satisfying  $c + \delta < a$ . Let  $\mathcal{B}$  the event  $\{H(Z_{t_b}) < c\}$ . Let  $\psi_K : \mathbb{R} \mapsto [0, 1]$  be a  $\mathcal{C}_c^{\infty}$  function equal to zero outside of  $[c + \frac{\delta}{2}, c + \delta + 1]$  such that  $\psi_K < 1$  on  $[c + \frac{\delta}{2}, c + \delta[$  and  $\psi_K(c + \delta) = 1$  and write  $\Psi_K = \psi_K \circ H_K$ . We have :

$$\left\{ \sup_{t \ge t_b} H\left(Z_t^K\right) < c + \delta \right\} = \left\{ \sup_{t \ge t_b} \Psi_K\left(Z_t^K\right) < 1 \right\}.$$

There exists some constant C > 0 such that  $|L\Psi_K| \leq C(1+\beta_t) \mathbb{1}_{\{H_K \geq c+\frac{\delta}{2}\}}$ , and from Ito's formula we can write :

$$\Phi\left(Z_{t+t_b}^{K}\right)\mathbb{1}_{\mathcal{B}} = \left(M_t + R_t\right)\mathbb{1}_{\mathcal{B}}$$

where  $R_t = \int_{t_b}^t L \Psi_K \left( Z_s^K \right) \mathrm{d}s$ , and if  $t_b \ge t_0$ :

$$\mathbb{E}\left(\sup_{t\geqslant t_{b}}|R_{t}|\mathbb{1}_{\mathcal{B}}\right) \leqslant C \int_{t_{b}}^{\infty} (1+\beta_{s})\mathbb{E}\left(\mathbb{1}_{H^{K}\left(Z_{t}^{K}\right)\geqslant c+\frac{\delta}{2}} \mid \mathcal{B}\right) \mathrm{d}s$$
$$\leqslant C \int_{t_{b}}^{\infty} (1+\beta_{s})\mathbb{P}\left(H_{K}(Z_{s}^{K})\geqslant c+\frac{\delta}{2}|\mathcal{B}\right) \mathrm{d}s$$
$$\leqslant C \int_{t_{b}}^{\infty} \frac{1+\ln(1+t)}{(1+t)^{1+\frac{\delta}{\delta c}}} \mathrm{d}s$$

where the constant C depends only on  $U^K$  and its derivative, but not on  $t_b$ . Since  $(1 + \ln(1 + t))/(1+t)^{1+\frac{\delta}{8c}}$  is integrable, we can take  $t_b$  great enough so that the event  $E = \{\sup_{t \ge t_b} |R_t| \le \frac{1}{10}\}$  has probability at least  $\frac{3}{4}$ . On  $E \cap \mathcal{B}$ ,  $M_t$  takes value in  $\left[-\frac{1}{10}, \frac{11}{10}\right]$  because  $0 \le \psi_A \le 1$ . Using Doob's up-crossing inequality as in the proof of Lemma 4.2.5, we get that the probability of  $\Psi_K$  going to 1 knowing  $\mathcal{B}$  is less then  $\frac{1}{2}$ . We conclude by :

$$\mathbb{P}\left(\sup_{t\geq 0} H_{K_A}(Z_t^{K_A}) < c+\delta\right) \geq \mathbb{P}(\mathcal{B}) \times \mathbb{P}\left(\sup_{t\geq 0} H_{K_A}(Z_t^{K_A}) < c+\delta \mid \mathcal{B}\right) > 0.$$

#### 4.2.5 Non-convergence with fast cooling schedules

We are now ready to prove Theorem 4.1.2. In this section, Assumption 4.1.2 is enforced and we use the definitions and notations of Section 4.2.4.

We start with a result on the position of the process for small times, as well as a Doeblin-like condition, which will be proven in Section 4.3.2:

**Lemma 4.2.8.** For all  $t, \delta > 0$ , write :

$$\mathcal{B}_t = \{ X_s \in \mathcal{C}(\tilde{x}, c+\delta) \,\,\forall s \in [0, t] \} \,.$$

Then, for all  $z_0 = (x_0, y_0)$  with  $x_0 \in C(\tilde{x}, c)$ ,  $\mathbb{P}_{z_0}(\mathcal{B}_t) > 0$ , and more precisely for all compact set  $\mathcal{K}$  included in the interior of  $C(\tilde{x}, c + \delta) \times \mathbb{R}^d$ , there exists  $\varepsilon > 0$  such that

$$\mathbb{P}_{z_0}\left(\mathcal{B}_t, \ Z_t \in \cdot\right) \geqslant \varepsilon \ell\left(\cdot \cap \mathcal{K}\right)$$

where  $\ell$  stands for the Lebesgue measure.

Proof of Theorem 4.1.2. By conditioning on the initial condition, it is sufficient to prove the result with a fixed initial condition  $z_0 = (x_0, y_0)$  with  $x_0 \in C(\tilde{x}, c)$ . Moreover it is sufficient to prove the result for  $\delta > 0$  small enough. We consider a fixed  $\delta > 0$  such that  $c + \delta < \inf\{U^K(x), x \in [-L_K, L_K]^d \setminus C(\tilde{x}, a)\} \leq a$ , which means (4.11) holds and  $\{U^K \leq c + \delta\} = C(\tilde{x}, c + \delta)$  (seeing  $\{U^K \leq c + \delta\}$  as a subset of  $[-L_K, L_K]^d$ ).

Let  $f_0^K \in \mathcal{C}^{\infty}(M_K \times M_K)$  a probability density and  $t_b > 0$  as in Lemma 4.2.7 applied with  $\delta/2$  instead of  $\delta$  such that:

$$\mathbb{P}_{f_0^K}\left(\sup_{t \ge t_b} H_K(Z_t^K) \leqslant c + \delta/2\right) > 0.$$

In other words, denoting by  $\tilde{f}$  the law at time  $t_b$  of the process solution to equation (4.10) with initial condition  $f_0^K$  and conditioned to  $\{H^K(Z_{t_b}) \leq c + \delta/2\}$ , and  $(Z^{K,t_b})_{t \geq 0}$  the solution to equation (4.10) with initial condition  $\tilde{f}$  at time  $t_b$ ,

$$\mathbb{P}_{\tilde{f}}\left(\sup_{t\geq 0}H_K(Z_t^{K,t_b})\leqslant c+\delta/2\right)>0.$$

Let  $\mathcal{B}_{t_b}$  be as in Lemma 4.2.8. From Lemma 4.2.8 and the fact  $\tilde{f}$  has a bounded density on its support  $\{H^K \leq c + \delta/2\}$  (which we see as a subset of  $[-L_K, L_K]^{2d}$ ), there exists  $\varepsilon > 0$  such that

$$\mathbb{P}_{z_0}\left(B_{t_b}, \ Z_{t_b} \in \cdot\right) \geqslant \varepsilon f.$$

Finally, thanks to (4.11), we conclude that

$$\mathbb{P}_{z_0} \left( X_t \in \mathcal{C}(\tilde{x}, c+\delta) \; \forall t \ge 0 \right) \ge \mathbb{P}_{z_0} \left( \mathcal{B}_{t_b}, \; X_t \in \mathcal{C}(\tilde{x}, c+\delta) \; \forall t \ge t_b \right)$$
$$\ge \varepsilon \mathbb{P}_{\tilde{f}} \left( \sup_{t \ge 0} H(Z_t^{K, t_b}) < c+\delta \right) > 0.$$

-	-	-
L		
L		1

#### 4.3 Auxiliary results

#### 4.3.1 Uniform energy bounds

This section is dedicated to the proof of Lemma 4.2.1. First, we consider a family of approximation functions  $\eta_m \in \mathcal{C}_c^{\infty}$  for  $m \ge 1$  in order to justify some PDE computations below. Let

$$\Phi(s) = \begin{cases} e^{\frac{1}{s^2 - 1}} / \int e^{\frac{1}{u^2 - 1}} du & \text{for } s \in (-1, 1) \\ 0 & \text{for } s \in \mathbb{R} \setminus (-1, 1) \end{cases}$$

and, for  $m \ge 1$ ,  $\Phi_m(s) = \Phi(s/m)/m$ ,  $\nu_m = \mathbb{1}_{(-\infty,m^2]} * \Phi_m$  where \* denotes the convolution, and finally, for  $z \in \mathbb{R}^{2d}$ ,

$$\eta_m(z) = \nu_m \left( \ln \left( H(z) + 1 \right) \right) \,.$$

**Proposition 4.3.1.** Assume  $U \to +\infty$  at infinity and  $\nabla U$  is bounded.

- For all  $m \ge 1$ ,  $\eta_m \in \mathcal{C}^{\infty}(\mathbb{R}^d)$ , has a compact support, and satisfies  $0 \le \eta_m \le 1$ .
- For all  $z \in \mathbb{R}^{2d}$ ,  $\eta_m(z) \to 1$  as  $m \to \infty$ .
- There exists C > 0 such that  $L_t \eta_m \leq C \beta_t / m$  and  $|\nabla \eta_m| \leq C \beta_t / m$  for all  $m \geq 1$ .

*Proof.* This is [31, Lemma 4].

Proof of Lemma 4.2.1. We first show the result when  $\nabla U$  and its derivative are bounded, the general case being then obtained by approximating U.

Hence, suppose for now that  $\nabla U$  and its derivative are bounded. In this case, it can be shown that the law of the process at time t admits a bounded density  $f_t$  such that  $(t, z) \mapsto f_t(z) \in \mathcal{C}^{\infty}(\mathbb{R}_+ \times \mathbb{R}^{2d})$ , see the proof of Lemma 4.4.2. Define  $g_t(z) = f_t(z)e^{\beta_t H(z)}$  and  $u(t) = \mathbb{E}(H(Z_t))$ . Since f and U are smooth, so is g. Consider

$$N(t) = \int_{\mathbb{R}^{2d}} g_t(z) \ln(1 + g_t(z)) e^{-\beta_t H(z)} dz.$$

From the inequality  $\ln(1+x) \leq \ln(x) + 1/x$  for all x > 0,

$$N(t) \leqslant \int_{\mathbb{R}^{2d}} f_t(z) \ln(f_t(z)) \mathrm{d}x \mathrm{d}y + \beta_t \mathbb{E}(H(Z_t)) + \int_{\mathbb{R}^{2d}} e^{-\beta_t H(z)} \mathrm{d}z.$$

Since  $f_t$  is bounded on  $[0,T] \times \mathbb{R}^{2d}$ , and is in  $L^1(dz)$  for all  $t \ge 0$ ,  $t \mapsto \int_{\mathbb{R}^{2d}} f_t \ln(f_t)$  is locally bounded and so is N(t). In order to differentiate N, we introduce for  $m \ge 1$  the approximation

$$N_m(t) = \int_{\mathbb{R}^{2d}} \eta_m(z) g_t(z) \ln(1 + g_t(z)) e^{-\beta_t H(z)} dz.$$

Integrating by parts, we see that the dual of the generator  $L_t$  in  $L^2(e^{-\beta_t H})$  is

$$L_t^* = -y \cdot \nabla_x + (-\gamma_t y + \nabla_x U) \cdot \nabla_y + \gamma_t \beta_t^{-1} \Delta_y$$
(4.12)

Since  $f_t$  solves  $\partial_t f_t = L_t^*(f_t e^{\beta_t H}) e^{-\beta_t H}$ ,  $g_t$  solves  $\partial_t g_t = \beta'_t H g_t + L_t^* g_t$ . Using that  $\eta_m$  is compactly supported for all  $m \ge 1$ , we can differentiate

$$\begin{split} N'_{m}(t) &= \int_{\mathbb{R}^{2d}} \eta_{m} \bigg( -\beta'_{t} H g_{t} \ln(1+g_{t}) e^{-\beta_{t} H} + \partial_{t} g_{t} \ln(1+g_{t}) e^{-\beta_{t} H} + \frac{g_{t}}{1+g_{t}} \partial_{t} g_{t} e^{-\beta_{t} H} \bigg) \\ &= \int_{\mathbb{R}^{2d}} \eta_{m} L_{t}^{*} g_{t} \left( \ln(1+g_{t}) + \frac{g_{t}}{1+g_{t}} \right) e^{-\beta_{t} H} + \int_{\mathbb{R}^{2d}} \eta_{m} \frac{g_{t}^{2}}{1+g_{t}} \beta'_{t} H e^{-\beta_{t} H}. \end{split}$$

Now, using that  $g_t \ge 0$ ,

$$\int_{\mathbb{R}^{2d}} \eta_m \frac{g_t^2}{1+g_t} \beta_t' H e^{-\beta_t H} \leqslant \beta_t' \int_{\mathbb{R}^{2d}} f_t H = \beta_t' u(t).$$

Consider the carré du champ operator  $\Gamma_t$  associated to  $L^*_t$  given by

$$\Gamma_t(g_1, g_2) := \frac{1}{2} \left( L_t^*(g_1 g_2) - g_2 L_t^*(g_1) - g_1 L_t^*(g_2) \right) = \gamma_t \beta_t^{-1} \nabla_y g_1 \cdot \nabla_y g_2$$
(4.13)

for smooth  $g_1, g_2$ , and  $\Gamma_t(g) := \Gamma_t(g, g)$ . Using that  $e^{-\beta_t H}$  is invariant for  $L_t^*$  so that

$$\int L_t^* g e^{-\beta_t H} = 0$$

for all smooth g and the diffusion property

$$L_t^*(\phi(g)) = \phi'(g)L_t^*(g) + \phi''(g)\Gamma_t(g)$$

for any smooth  $\phi$ , we get

$$\begin{split} \int_{\mathbb{R}^{2d}} \eta_m L_t^* g_t \left( \ln(1+g_t) + \frac{g_t}{1+g_t} \right) e^{-\beta_t H} \\ &= \int_{\mathbb{R}^{2d}} \eta_m L_t^* g_t \left( \ln(1+g_t) + \frac{g_t}{1+g_t} \right) e^{-\beta_t H} - \int_{\mathbb{R}^{2d}} L_t^* (\eta_m g_t \ln(1+g_t)) e^{-\beta_t H} \\ &= -\int_{\mathbb{R}^d \times \mathbb{R}^d} \eta_m \Gamma_t(g_t) \left( \frac{1}{1+g_t} + \frac{1}{(1+g_t)^2} \right) e^{-\beta_t H} - \int_{\mathbb{R}^{2d}} L_t^* (\eta_m) g_t \ln(1+g_t) e^{-\beta_t H} \\ &- 2 \int_{\mathbb{R}^{2d}} \Gamma_t(\eta_m, g_t \ln(1+g_t)) e^{-\beta_t H} \,. \end{split}$$

The first term is negative (since  $\eta_m \Gamma_t(g_t) \ge 0$ ) and the others are bounded as follows:

$$\begin{split} -\int_{\mathbb{R}^d \times \mathbb{R}^d} L_t^*(\eta_m) g_t \ln(1+g_t) e^{-\beta_t H} &- 2 \int_{\mathbb{R}^d \times \mathbb{R}^d} \Gamma(\eta_m, g_t \ln(1+g_t)) e^{-\beta_t H} \\ &= \int_{\mathbb{R}^d \times \mathbb{R}^d} \eta_m L_t^*(g_t \ln(1+g_t)) e^{-\beta_t H} \\ &= \int_{\mathbb{R}^d \times \mathbb{R}^d} L_t(\eta_m) g_t \ln(1+g_t) e^{-\beta_t H} \mathrm{d}x \mathrm{d}y \\ &\leqslant \frac{C\beta_t}{m} N(t) \end{split}$$

thanks to Proposition 4.3.1. We conclude that for all  $m \geqslant 1, \, 0 \leqslant s \leqslant t$  :

$$N_m(t) - N_m(s) \leqslant \int_s^t \beta'_r u(r) + \frac{C\beta_r}{m} N(r) \mathrm{d}r.$$

By the monotone convergence theorem,  $N_m(t) \to N(t)$  as  $m \to \infty$ , hence :

$$N(t) - N(s) \leqslant \int_{s}^{t} \beta'_{r} u(r) \mathrm{d}r.$$

On the other hand, from the variational formula for the entropy:

$$\int_{\mathbb{R}^{2d}} f_t \ln f_t = \max\left\{\int_{\mathbb{R}^{2d}} f_t \ln g; \ g: \mathbb{R}^{2d} \mapsto \mathbb{R}_+, \ \int_{\mathbb{R}^{2d}} g = 1\right\},\$$

we get with  $g_0 = e^{-\alpha_0 H} / \mathcal{Z}_{\alpha_0}$ , where  $\mathcal{Z}_{\alpha_0} = \int e^{-\alpha_0 H}$ ,

$$N(t) \ge \beta_t u(t) + \int_{\mathbb{R}^{2d}} f_t \ln f_t \ge \beta_t u(t) + \int_{\mathbb{R}^{2d}} f_t \ln g_0 = (\beta_t - \alpha_0) u(t) - \ln(\mathcal{Z}_{\alpha_0}).$$

As a consequence, writing  $\phi(t) = N(t) + \ln(\mathcal{Z}_{\alpha_0})$  for  $t \ge 0$ ,

$$\phi(t) \leqslant \phi(0) + \int_0^t \frac{\beta'_s}{\beta_s - \alpha_0} \phi(s) \mathrm{d}s.$$

Then Gronwall's lemma gives  $\phi(t) \leq \phi(0) \frac{\beta_t - \alpha_0}{\beta_0 - \alpha_0}$  and using again that  $u(t) \leq \phi(t)/(\beta_t - \alpha_0)$  concludes the proof in the case where  $\nabla U$  and all its derivatives are bounded.

Let us now consider the general case, without any assumption on  $\nabla U$ . For n large enough, we fix some  $U_n \in \mathcal{C}^{\infty}(\mathbb{R}^d)$  equal to U on B(0,n), to |x| outside of B(0, n + 1) and such that for all  $x \in \mathbb{R}^d$ ,  $U_n(x) \ge \min(U(x), |x|) - 1$ . Let  $(Z_t^n)_{t \ge 0} = (X_t^n, Y_t^n)_{t \ge 0}$  be the diffusion defined by Equation (4.1) where U is replaced by  $U_n$  and starting from the same initial distribution  $f_0$ . By design,  $Z_t$  and  $Z_t^n$  are equal up to the time  $\tau_n = \inf\{t \ge 0; |X_t| \ge n\}$ . As Z does not explode in finite time,  $\lim_n \tau_n = \infty$  and, for all  $t \ge 0$ ,  $\lim_n H_n(Z_t^n) = H(Z_t)$  almost surely, where  $H_n(x, y) = U_n(x) + |y|^2/2$ . By Fatou's lemma, for all  $t \ge 0$ ,

$$\mathbb{E}(H(Z_t)) = \mathbb{E}\left(\liminf_{n \to \infty} H_n(Z_t^n)\right) \leqslant \liminf_{n \to \infty} \mathbb{E}(H_n(Z_t^n)) \leqslant \liminf_{n \to \infty} \frac{\kappa^{\beta_0, n}(f_0) + \ln(\mathcal{Z}_{\alpha_0}^n)}{\beta_0 - \alpha_0}$$

where  $\kappa^{\beta_0,n}(f_0)$  is define as  $\kappa^{\beta_0,n}(f_0)$  but with U replaced by  $U_n$ . Now, since  $f_0$  is compactly supported,  $\kappa^{\beta_0,n}(f_0)$  is independent of n for n large enough. Finally, using that  $e^{-\beta_0 U_n(x)} \leq e^{\alpha_0}(e^{-\alpha_0 U(x)} + e^{-\alpha_0|x|})$ , we can apply the Dominated Convergence Theorem to get that  $\mathcal{Z}_{\alpha_0}^n \to \mathcal{Z}_{\alpha_0}$  as  $n \to \infty$ , which concludes.

г	-	-	٦
L			I
L			I
•			

#### 4.3.2 Small time regularisation

Proof of Lemma 4.2.2. It is enough to show the first point for  $\nabla U$  bounded. Indeed, for all  $\hat{U}$  equal to U on  $\{U \leq A+1\}$ , and  $(\tilde{Z})_{t\geq 0}$  the corresponding process, we have the equality

$$\left\{\sup_{t\leqslant t^*} H(Z_t)\leqslant A+1\right\} = \left\{\sup_{t\leqslant t^*} \tilde{H}(\tilde{Z}_t)\leqslant A+1\right\}.$$

We only need a bound on  $\sup_{t \leq t^*} |Y_t - y_0|$  because :

$$\sup_{t\leqslant t^*} |U(X_t) - U(x_0)| \leqslant \|\nabla U\|_{\infty} t^* \sup_{t\leqslant t^*} |Y_t|.$$

From

$$Y_t = e^{-\int_0^t \gamma_s \mathrm{d}s} \left( y_0 + \int_0^t \nabla U(X_s) e^{\int_0^s \gamma_u \mathrm{d}u} \mathrm{d}s + \int_0^t \sqrt{2\gamma_s \beta_s^{-1}} e^{\int_0^s \gamma_u \mathrm{d}u} \mathrm{d}B_s \right) \,,$$

 $\sup_{s\leqslant t}|Y_s|\leqslant |y_0|+\|\nabla U\|_{\infty}t+\sup_{s\leqslant t}|W_s|\,,$ 

we get

where  $W_t = e^{-\int_0^t \gamma_s ds} \int_0^t \sqrt{2\gamma_s \beta_s^{-1}} e^{\int_0^s \gamma_u du} dB_s$ . Since  $(W_t)_{0 \leq t \leq 1}$  is a  $L^2$ -martingale, Doob's inequality implies

$$\mathbb{E}\left(\sup_{0\leqslant s\leqslant t}|W_s|^2\right)\leqslant 4\mathbb{E}(W_t^2)=4e^{-2\int_0^t\gamma_s\mathrm{d}s}\int_0^t 2\gamma_s\beta_s^{-1}e^{2\int_0^s\gamma_u\mathrm{d}u}\mathrm{d}s\leqslant 8Lt$$

Now we can take  $t_1^*$  such that  $t_1^* ||\nabla U||_{\infty} + \sup_{t \leq t_1^*} W_t \leq \frac{1}{8\sqrt{A}}$  with probability larger than  $1 - \varepsilon$ . Thus, with probability at least  $1 - \varepsilon$ , we have :

$$\sup_{t \leqslant t_1^*} |Y_t|^2 \leqslant |y_0|^2 + \frac{1}{4} + \frac{1}{32A}$$

Since A > 1, this is less than  $|y_0|^2 + \frac{1}{2}$  and we write  $t_2^* = t_1^* \wedge (2\|\nabla U\|_{\infty}\sqrt{1+|y_0|^2})^{-1}$ .

For the second point, fix some  $t^* \leq t_2^*$  and let, for  $t \leq t^*$ ,  $(\bar{X}_t, \bar{Y}_t)$  be the solution of the system  $(d\bar{X}_t - \bar{X}_t) dt$ 

$$\begin{cases} \mathrm{d}X_t = Y_t \mathrm{d}t \\ \mathrm{d}\bar{Y}_t = \sqrt{2\gamma_t \beta_t^{-1}} \mathrm{d}B_t \\ (X_0, Y_0) = (x_0, y_0), \end{cases}$$

 $\operatorname{let}$ 

$$N_t = -\int_0^t \frac{\sqrt{\beta_s}}{\sqrt{2\gamma_s}} \left(\nabla U(\bar{X}_s) + \gamma_s \bar{Y}_s\right) \mathrm{d}B_s$$

and  $\mathbb{Q} = e^{N_{t^*} - \frac{1}{2} \langle N \rangle_{t^*}} \mathbb{P}$ . Ito's formula gives for  $t \leqslant t^*$ :

$$\frac{\beta_t}{2\gamma_t}\bar{Y}_t \cdot (\nabla U(\bar{X}_t) + \frac{\gamma_t}{2}\bar{Y}_t) - \frac{\beta_0}{2\gamma_0}y_0 \cdot (\nabla U(x_0) + \frac{\gamma_0}{2}y_0) \\
= -N_t + \int_0^t \left(\frac{\beta_s}{2\gamma_s}\Delta U(\bar{X}_s)|\bar{Y}_s|^2 + \frac{\beta_s'}{2\gamma_s}\bar{Y}_s \cdot (\nabla U(\bar{X}_s) + \frac{\gamma_s}{2}\bar{Y}_s) - \frac{\beta_s\gamma_s'}{2\gamma_s^2}\bar{Y}_s \cdot \nabla U(\bar{X}_s) + \frac{\beta_s}{4}\right) \mathrm{d}s. \tag{4.14}$$

Using  $\beta' \leq c$  and  $\beta_t \leq \beta_0 + \frac{t^*}{c}$ , we then get  $\mathbb{1}_{\mathcal{B}}N_t \leq C(\beta_0 + 1)$  where C is independent from  $\beta_0$  and  $\gamma$  and depends only on  $\sup_{\{U \leq A+1\}} |\nabla U|$ ,  $\sup_{\{U \leq A+1\}} |\Delta U|$ ,  $\kappa$  and c.

Girsanov's theorem yields that, under the change of probability  $\mathbb{P} \to \mathbb{Q}$ ,  $(\bar{X}_t, \bar{Y}_t)_{t \leq t^*}$  is a solution to the original Equation (4.1) and, for all  $\phi \ge 0, t \leq t^*$ ,

$$\mathbb{E}(\phi(X_t, Y_t)\mathbb{1}_{\mathcal{B}}) = \mathbb{E}(e^{N_{t^*} - \frac{1}{2}\langle N \rangle_{t^*}} \phi(\bar{X}_t, \bar{Y}_t)\mathbb{1}_{\mathcal{B}}) \leqslant Ce^{C_A^1(\beta_0 + 1)} \mathbb{E}(\phi(\bar{X}_t, \bar{Y}_t)\mathbb{1}_{\mathcal{B}}).$$

It only remains to show that  $(\bar{X}_t, \bar{Y}_t)$  has a density bounded by some  $e^{C\beta_0}$ . As a Gaussian process, the density of  $(\bar{X}_t, \bar{Y}_t)$  is bounded by  $(2\pi)^{-d/2}/\det(Q_d)$  where  $Q_d$  is the covariance matrix of the process at time  $t^*$ . By independence,  $\det(Q_d) = (\det(Q_1))^d$ , and a straightforward computation yields

$$\det(Q_1) = 8 \int_0^{t^*} \gamma_s \beta_s^{-1} \mathrm{d}s \int_0^{t^*} (t^* - s) \int_0^s \gamma_u \beta_u^{-1} \mathrm{d}u \mathrm{d}s - 4 \left( \int_0^{t^*} \int_0^s \gamma_u \beta_u^{-1} \mathrm{d}u \mathrm{d}s \right)^2$$

Now, let  $\varpi, \lambda > 1$  be such that  $\varpi \lambda^2 < 4/3$  and take  $t^*$  small enough (depending only on c and

 $\kappa$ ) so that uniformly in  $\beta_0 \ge 1$ 

$$\beta_0 \leqslant \beta_{t^*} \leqslant \varpi \beta_0 \quad \text{and} \quad \lambda^{-1} \gamma_0 \leqslant \gamma_{t^*} \leqslant \lambda \gamma_0.$$

This choice ensures

$$\det(Q_1) \ge \kappa \beta_0^{-1} t^{*4} \left(\frac{4}{3\varpi\lambda} - \lambda^2\right)$$

and thus the result.

Proof of Lemma 4.2.8. The first part follows from the controlability of the process, see [137, proposition 5]. The second part follows from the fact the distribution of the process killed when it leaves  $C(\tilde{x}, c+\delta) \times \mathbb{R}^d$  solves a parabolic hypoelliptic Dirichhet problem, hence has a continuous positive density. In fact, in the time-homogeneous case, this is exactly [109, Theorem 2.20]. In our time-inhomogeneous case, we can proceed as follows. First, for some  $A' > \max(|y_0|, A)$  large enough, consider  $\mathcal{D}$  the interior of  $C(\tilde{x}, c+\delta) \times [-A', A']$  and  $\tau = \inf\{s \ge 0, Z_t \notin \mathcal{D}\}$ . Then, for all t > 0,

$$\mathbb{P}_{z_0}\left(\mathcal{B}_t, \ Z_t \in \cdot\right) \geqslant \mathbb{P}_{z_0}\left(\tau > t, \ Z_t \in \cdot\right) := p_t^{\mathcal{D}}(z_0, \cdot).$$

It is well-known that  $p_t^{\mathcal{D}}$  solves a parabolic equation with generator  $L_t$  on  $\mathcal{D}$ . Since  $L_t$  is hypoelliptic,  $p_t^{\mathcal{D}}(z_0, \cdot)$  has a continuous density. Finally, the coefficients of  $L_t$  being smooth and bounded on  $\mathcal{D}$ ,  $p_t^{\mathcal{D}}(z_0, \cdot)$  being not identically zero and the process being controllable, we can use the strong maximum principle of [155, Theorem 6.1] to deduce that  $p_t^{\mathcal{D}}(z_0, \cdot)$  cannot take the value 0 in  $\mathcal{D}$ . As a continuous positive function,  $p_t^{\mathcal{D}}(z_0, \cdot)$  is thus lower bounded by a positive constant over any compact subset of  $\mathcal{D}$ , which concludes.

## 4.4 $L^2$ -hypocoercivity

In this section we use the definitions and notations of Section 4.2.2

The hypocoercivity issue arises in the proof of Proposition 4.2.3 when computing the evolution of the  $L^2$ -norm of  $h_t^K$ . In the standard elliptic case, as in [83], one would simply differentiate this quantity and concludes with a Poincaré inequality of the form :

$$\int \left(h - \int h \mathrm{d}\mu\right)^2 \mathrm{d}\mu \leqslant C \int \Gamma(h) \mathrm{d}\mu$$

where  $\Gamma$  is the carré du champ associated to the process. However, in the kinetic case, as we saw in (4.13),  $\Gamma_t(f) = \gamma_t \beta_t^{-1} |\nabla_y f|^2$ , which means such an inequality cannot hold since, for non-constant functions of x, the left hand side is positive while the right hand side vanishes. For this reason, we work with a modified norm as in [163].

More precisely, at a formal level, the proof of Proposition 4.2.3 is the following: writing  $\phi_t(h) = |(\nabla_x + \nabla_y)h|^2 + \sigma_t h^2$  with  $\sigma_t = \frac{1}{2} + 2\sqrt{\gamma_t^{-1}\beta_t}(1 + ||\nabla U^K||_{\infty} + \gamma_t)^2$ , we introduce

$$\tilde{N}(t) = \int_{M_K \times \mathbb{R}^d} \phi_t \left( h_t^K - 1 \right) \mathrm{d}\mu_{\beta_t}^K \,, \qquad \tilde{I}(t) = \int_{M_K \times \mathbb{R}^d} \left| \nabla h_t^K \right|^2 \mathrm{d}\mu_{\beta_t}^K \,.$$

Differentiating  $\tilde{N}$ , one can (formally) check that

$$\tilde{N}'(t) \leqslant -\frac{1}{2}\tilde{I}(t) + C\beta'_t(1+\beta_t)\tilde{N}(t)$$

for some constant C > 0, the definition of  $\tilde{N}$  being motivated by the  $-\tilde{I}$  term in this inequality. Using a Poincaré inequality for  $\mu_{\beta_t}^K$  (with the full gradient rather than  $\Gamma_t$ ) with a constant  $\lambda(\beta_t)$  that scales as  $1/t^{c^*/c} \gg \beta'_t \beta_t$ , we get that  $\tilde{N}'(t) \leq 0$  for t large enough (or equivalently for all  $t \geq 0$  if  $\beta_0$  is large enough), hence  $\tilde{N}$  is bounded. We conclude the proof of Proposition 4.2.3 by bounding

$$\|h_t\|_{L^2(\mu_{\beta_t})} \leq 1 + \|h_t - 1\|_{L^2(\mu_{\beta_t})} \leq 1 + 2N(t)$$

In the remainder of this section, this formal proof is made rigorous and we give the details of the computations.

**Remark 4.4.1.** The proof of [137] is based on a similar argument but, as mentioned in the introduction, and as noticed by the authors of [31], it contains an error. The problem occurs when it comes to justify rigorously the derivation of  $\tilde{N}$ . In [137], a compactly-supported truncation function is added within the integral. This leads to additionnal terms in  $\partial_t \tilde{N}$ . One of these terms is said to be non-positive in [137, Lemma 16], which is false (there is a sign error). In [31], the authors add a small elliptic term to the dynamics, use elliptic regularity results to justify the computation and then let the small ellipticity parameter vanish afterwards. In this section, we make a correct version of the argument of [137], combining some bounds on the density of the process (Lemma 4.4.2 below) and some moment estimates (Lemma 4.4.1 below). Also, notice that, by comparison with [137, 31], we have already reduced the problem to a compact state space for the position x, the non-compact part of the dynamics only concerns the velocity.

First, we need a few preliminary lemmas. We start by stating the following Poincaré inequality (with the full gradient) :

**Proposition 4.4.1.** For  $U^K \in \mathcal{C}^{\infty}(M_K, \mathbb{R})$  there exists  $\lambda : \mathbb{R}_+ \to \mathbb{R}_+$  such that for all  $f \in \mathcal{C}^{\infty}(M_K \times \mathbb{R}^d)$  with compact support :

$$\lambda(\beta) \int_{M_K \times \mathbb{R}^d} \left( f - \int_{M_K \times \mathbb{R}^d} f d\mu_\beta^K \right)^2 d\mu_\beta^K \leqslant \int_{M_K \times \mathbb{R}^d} |\nabla f|^2 d\mu_\beta^K$$

and moreover

$$\lim_{\beta \to \infty} \frac{1}{\beta} \ln(\lambda(\beta)) = -c^*(U^K)$$

where  $c^*(U^K)$  is defined as  $c^*$  with U replaced by  $U^K$ .

*Proof.* The fact that the Gibbs probability measure  $\tilde{\mathcal{Z}}e^{-\beta U^{K}}$  satisfies a Poincaré inequality with a constant  $\lambda(\beta)$  satisfying

$$\lim_{\beta \to \infty} \frac{1}{\beta} \ln(\lambda(\beta)) = -c^*(U^K)$$

corresponds to [83, Theorem 1.14]. The Gaussian measure  $\mathcal{N}(0, \beta I)$  also satisfies a Poincaré inequality with constant  $\beta$ , and we conclude with the tensorization property of the Poincaré inequality, see [6, Proposition 4.3.1].

For  $\mu$  a probability measure,  $H^1(\mu)$  denotes the usual Sobolev space of functions in  $L^2(\mu)$  with derivative in  $L^2(\mu)$ .

**Proposition 4.4.2.** There exists C > 0 such that for all  $\beta > 1$  and  $g \in H^1(\mu_{\beta}^K)$ ,

$$\int y^2 g(x,y)^2 \mu_{\beta}^K(dxdy) \leqslant C\left(\int g(x,y)^2 \mu_{\beta}^K(dxdy) + \int |\nabla_y g(x,y)|^2 \mu_{\beta}^K(dxdy)\right).$$

*Proof.* This is a particular case of [163, Lemma A.18]. We give a short proof for completeness. Notice that  $ye^{-\beta|y|^2/2} = -\beta^{-1}\nabla e^{-\beta|y|^2/2}$ . Hence an integration by parts and Young's inequality yield, for any x,

$$\begin{split} &\int_{\mathbb{R}^d} y^2 g^2(x,y) e^{-\beta|y|^2/2} \mathrm{d}y \\ &= -\beta^{-1} \int_{\mathbb{R}^d} g^2(x,y) y \cdot \nabla e^{-\beta|y|^2/2} \mathrm{d}y \\ &= \beta^{-1} \int_{\mathbb{R}^d} \nabla \cdot (g^2(x,y)y) e^{-\beta|y|^2/2} \mathrm{d}y \\ &= d\beta^{-1} \int_{\mathbb{R}^d} g^2(x,y) e^{-\beta|y|^2/2} \mathrm{d}y + \beta^{-1} \int_{\mathbb{R}^d} 2g \nabla g(x,y) \cdot y e^{-\beta|y|^2/2} \mathrm{d}y \\ &\leqslant d\beta^{-1} \int_{\mathbb{R}^d} g^2(x,y) e^{-\beta|y|^2/2} \mathrm{d}y + \beta^{-2} \int_{\mathbb{R}^d} 2|\nabla g(x,y)|^2 e^{-\beta|y|^2/2} \mathrm{d}y \\ &\quad + \frac{1}{2} \int_{\mathbb{R}^d} y^2 g^2(x,y) e^{-\beta|y|^2/2} \mathrm{d}y, \end{split}$$

and thus

$$\int_{\mathbb{R}^d} y^2 g^2(x,y) e^{-\beta|y|^2/2} \mathrm{d}y \leqslant 2d\beta^{-1} \int_{\mathbb{R}^d} g^2(x,y) e^{-\beta|y|^2/2} \mathrm{d}y + 4\beta^{-2} \int_{\mathbb{R}^d} |\nabla g(x,y)|^2 e^{-\beta|y|^2/2} \mathrm{d}y.$$

Conclusion follows by integrating with respect to x.

The next two lemmas will be used in forthcoming computations to justify that some quantities are finite and therefore allowing to interchange differentiation and integration.

**Lemma 4.4.1.** Fix some K > 0 and any c > 0. Then for all  $\alpha > 0$  and initial condition  $f_0^K \in \mathcal{C}^{\infty}$  with compact support, there exists  $b_{\alpha,K}$  such that if  $\beta_0 \ge b_{\alpha,K}$ , then

$$t \mapsto \mathbb{E}_{f_0^K} \left( e^{(\beta_t - \alpha) H_K \left( Z_t^K \right)} \right)$$

is finite and locally bounded.

*Proof.* Write  $\tau_N = \inf \{t \ge 0, H_K(Z_t^K) \ge N\}$  and  $\phi_t(x, y) = e^{(\beta_t - \alpha)H_K(x, y)}$ . For  $\beta_0 > \alpha$ , we compute for  $t \in [0, T]$ :

$$\begin{split} \partial_t \phi_t + L_t \phi_t \\ &= \left( \beta'_t H_K + (\beta_t - \alpha) y. \nabla U^K - \gamma_t \left( \beta_t - \alpha \right) y. \nabla_y H_K \right. \\ &+ \beta_t^{-1} \gamma_t \left( (\beta_t - \alpha) \Delta_y H_K + (\beta_t - \alpha)^2 \left| \nabla_y H_K \right|^2 \right) \right) \phi_t \\ &\leqslant \left( \beta'_t \left( \| U^K \|_{\infty} + \frac{y^2}{2} \right) + (\beta_t - \alpha)^3 \| \nabla U^K \|_{\infty} + \frac{y^2}{\beta_t - \alpha} - \gamma_t \left( \beta_t - \alpha \right) y^2 \right. \\ &+ \beta_t^{-1} \gamma_t \left( d \left( \beta_t - \alpha \right) + \left( \beta_t - \alpha \right)^2 y^2 \right) \right) \phi_t \\ &\leqslant \left( \beta'_0 \| U^K \|_{\infty} + \left( \beta_T - \alpha \right)^3 \| \nabla U^K \|_{\infty} + Ld \left( \beta_T - \alpha \right) + \left( \frac{\beta'_t}{2} + \frac{1}{\beta_t - \alpha} - \frac{\gamma_t \alpha \left( \beta_t - \alpha \right)}{\beta_t} \right) y^2 \right) \phi_t. \end{split}$$

We can choose  $\beta_0$  great enough so that  $\frac{\beta'_t}{2} + \frac{1}{\beta_t - \alpha} - \frac{\gamma_t \alpha(\beta_t - \alpha)}{\beta_t} \leq 0$  for all  $t \geq 0$ . We then classically get for  $t \in [0, T]$ :

$$\mathbb{E}\left(\phi_{t\wedge\tau_{N}}\left(Z_{t\wedge\tau_{N}}^{K}\right)\right) \leqslant e^{C_{T}t}\mathbb{E}\left(\phi_{0}\left(X_{0},Y_{0}\right)\right)$$

where  $C_T = \beta'_0 ||U^K||_{\infty} + (\beta_T - \alpha)^3 ||\nabla U^K||_{\infty} + Ld(\beta_T - \alpha)$ . The fact that  $f_0^K$  has a compact support and Fatou's lemma then yield the result.

**Lemma 4.4.2.** Fix K, c > 0. Then there exists  $b_K$  such that if  $\beta_0 \ge b_K$ , the law  $f_t^K$  of the process defined by the Equation (4.7), with an initial condition  $f_0^K \in C^{\infty}$  with compact support, is smooth, bounded along with its derivative, and satisfies :

$$\exists \alpha > 0, \forall t \ge 0, \exists C_t \text{ such that } f_t^K + |\nabla f_t^K| \le C_t e^{-\alpha |y^2|}$$

and  $t \mapsto C_t$  is locally bounded.

*Proof.* First, by Ito's formula,  $f_t^K(x, y)$  is a weak measure solution of the forward equation

$$\partial_t f_t^K = -y \cdot \nabla_x f_t^K + \nabla U^K(x) \cdot \nabla_y f_t^K + \gamma_t \nabla_v \cdot \left(v f_t^K\right) + \gamma_t \beta_t^{-1} \Delta_v f_t^K$$
  
=  $\tilde{L}_t f_t^K + d\gamma_t f_t^K$ , (4.15)

where  $\tilde{L}_t$  is the generator of the process  $(\tilde{X}, \tilde{Y})$  solving

$$\mathrm{d}\tilde{X}_t = -\tilde{Y}_t \mathrm{d}t\,, \qquad \mathrm{d}\tilde{Y}_t = \nabla U^K(\tilde{X}_t)\mathrm{d}t + \gamma_t \tilde{Y}_t \mathrm{d}t + \sqrt{2\gamma_t \beta_t^{-1}}\mathrm{d}B_t\,.$$

Using again Ito's formula, we get that the function given by

$$(x, y, t) \mapsto e^{d \int_0^t \gamma_s \mathrm{d}s} \mathbb{E}_{x, y} \left( f_0^K \left( \tilde{X}_t, \tilde{Y}_t \right) \right) \,,$$

also solves equation (4.15). Uniqueness of the weak solution of (4.15) is ensured by [18, Theorem 9.8.7], hence

$$f_t^K(x,y) = e^{d\int_0^t \gamma_s \mathrm{d}s} \mathbb{E}_{x,y}\left(f_0^K\left(\tilde{X}_t, \tilde{Y}_t\right)\right) \,,$$

from which we immediately get that  $f_t^K$  is bounded uniformly over [0, T] for all T > 0. Thanks to [85, Theorem 1],  $f_t^K$  is smooth for all  $t \ge 0$ . We can always differentiate (4.15) in a weak sense (i.e. once integrated with respect to a smooth compactly supported function of time and space and formally integrating by parts), from which we get that  $\nabla f_t^K$  is a weak solution of

$$\partial_t \nabla f_t^K = \tilde{L}_t \nabla f_t^K + J_t \nabla f_t^K \,,$$

where  $\tilde{L}_t$  acts component-wise on  $\nabla f_t^K$  and

$$J_t(x,y) = \begin{pmatrix} d\gamma_t I_d & \nabla^2 U^K(x) \\ -I_d & 2d\gamma_t I_d \end{pmatrix},$$

which is bounded uniformly over [0, T] for all T > 0. For a fixed T > 0, considering  $(W_t)_{t \ge 0}$ the matrix-valued process solution of  $dW_t = W_t J_{T-t}(\tilde{X}_t, \tilde{Y}_t)$  with  $W_0 = I_d$  and using again the uniqueness of the weak solution of the PDE, we get the Feynman-Kac representation

$$\nabla f_t^K(x,y) = \mathbb{E}_{x,y} \left( W_t \nabla f_0^K(\tilde{X}_t, \tilde{Y}_t) \right) \,,$$

which can be obtained by Itô's formula for the process  $(\tilde{X}_t, \tilde{Y}_t, W_t)_{t\geq 0}$  applied to the test function

 $h(x, y, w, t) = w \nabla f_t^K(x, y)$ . Hence,  $\nabla f_t^K$  is uniformly bounded over [0, T], and the same argument applies to all derivatives of  $f_t^K$ .

Now, let us prove the Gaussian bound of the statement, starting with  $f_t^K$ .

For  $z \in \mathbb{R}^{2d}$ , let  $\epsilon(z) = \min(1, \frac{f_t^K(z)}{2\|\nabla f_t^K\|_{\infty}})$ . If  $z' \in B(z, \epsilon(z)), f_t^K(z') \ge f_t^K(z) - \|\nabla f_t^K\|_{\infty} \epsilon(z) \ge \frac{f_t^K(z)}{2}$ . Then for any  $z = (x_1, y_1) \in \mathbb{R}^{2d}$ :

$$\mathbb{E}(e^{H_{K}(Z_{t}^{K})/2}) = \int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} e^{H_{K}(x,y)/2} f_{t}^{K}(x,y) \mathrm{d}x \mathrm{d}y$$
$$\geqslant \int_{B(z,\epsilon(z))} e^{H_{K}(x,y)/2} f_{t}^{K}(x,y) \mathrm{d}x \mathrm{d}y$$
$$\geqslant e^{-\frac{\|U\|_{\infty}}{2}} e^{\frac{|y_{1}|^{2}-1}{4}} \frac{f_{t}^{K}(z)}{2} \omega_{2d} \epsilon(z)^{d}$$

where  $\omega_{2d}$  is the volume of the unit-sphere in dimension 2d. Then there are two possibilities. If  $f_t^K(x,y) \ge 2 \|\nabla f_t^K\|_{\infty}$ , then

$$f_t^K(x,y) \leqslant \left(2e^{\frac{1}{4}}e^{\frac{\|U\|_{\infty}}{2}}\frac{\mathbb{E}\left(e^{H_K(Z_t^K)/2}\right)}{\omega_{2d}}\right)e^{-\frac{y^2}{4}}.$$

Otherwise,  $f_t^K(x, y) \leq 2 \|\nabla f_t^K\|_{\infty}$  and

$$f_t^K(x,y) \leqslant 2 \|\nabla f_t^K\|_{\infty}^{\frac{d}{d+1}} \left( e^{\frac{1}{4}} e^{\frac{\|U\|_{\infty}}{2}} \frac{\mathbb{E}(e^{H_K(Z_t^K)/2})}{\omega_{2d}} \right)^{\frac{1}{d+1}} e^{-\frac{y^2}{4(d+1)}}$$

In any case, using the fact that  $t \mapsto \mathbb{E}\left(e^{H_K(Z_t^K)/2}\right)$  is locally bounded for  $\beta_0$  great enough, we have the result for  $f_t^K$  with  $\alpha = \frac{1}{4(d+1)}$ . Recall the definition of the approximation functions  $\eta_m$  from Proposition 4.3.1. We use them here with U replaced by  $U^K$  (or equivalently with H replaced by  $H_K$ ), but it doesn't change any of their properties since now x is in a compact set. We now turn to  $\nabla f_t^K$  by using the approximation functions  $\eta_m$  from Section 4.3.1 and integrating by parts:

$$\begin{split} \int_{\mathbb{R}^d \times \mathbb{R}^d} \eta_m e^{\frac{H_K}{4}} \left| \nabla f_t^K \right|^2 \mathrm{d}z &= -\int_{\mathbb{R}^d \times \mathbb{R}^d} \nabla \left( \eta_m e^{\frac{H_K}{4}} \nabla f_t^K \right) f_t^K \mathrm{d}z \\ &= -\int_{\mathbb{R}^d \times \mathbb{R}^d} \left( \nabla_y \eta_m \cdot \nabla f_t^K e^{\frac{H_K}{4}} + \frac{\eta_m}{2} e^{\frac{H_K}{4}} \nabla H_K \cdot \nabla f_t^K + \eta_m e^{\frac{H_K}{4}} \Delta f_t^K \right) f_t^K \mathrm{d}z \\ &\leqslant C \int_{\mathbb{R}^d \times \mathbb{R}^d} e^{\frac{H_K}{2}} f_t^K \mathrm{d}z = C \mathbb{E} \left( e^{H_K \left( Z_t^K \right)/2} \right) \end{split}$$

for some constant C > 0 independent from m, where we used uniform bounds on the two first derivatives of  $f_t^K$ . We can then conclude in the same way as for  $f_t^K$ .

Setting  $\phi_{\beta,\gamma}(h) = |(\nabla_x + \nabla_y)h|^2 + \sigma(\beta,\gamma)h^2$  with  $\sigma(\beta,\gamma) = \frac{1}{2} + 2\sqrt{\gamma^{-1}\beta}(1 + ||\nabla U^K||_{\infty} + \gamma)^2$ , we consider the modified  $H^1$ -norm (and its truncated version for  $m \ge 1$ ):

$$N(t,\beta,\gamma) = \int_{M_K \times \mathbb{R}^d} \phi_{\beta,\gamma} \left(\frac{f_t^K}{\mu_\beta^K}\right) \mathrm{d}\mu_\beta^K$$

#### 4.4. $L^2$ -hypocoercivity

$$N_m(t,\beta,\gamma) = \int_{M_K \times \mathbb{R}^d} \eta_m \phi_{\beta,\gamma} \left(\frac{f_t^K}{\mu_\beta^K}\right) \mathrm{d}\mu_\beta^K$$

as well as

$$I(t,\beta) = \int_{M_K \times \mathbb{R}^d} \left| \nabla \frac{f_t^K}{\mu_\beta^K} \right|^2 \mathrm{d}\mu_\beta^K$$
$$I_m(t,\beta) = \int_{M_K \times \mathbb{R}^d} \eta_m \left| \nabla \frac{f_t^K}{\mu_\beta^K} \right|^2 \mathrm{d}\mu_\beta^K$$

To study those functionals along the dynamic, write  $\tilde{H}_m(t) = H_m(t, \beta_t, \gamma_t)$ ,  $\tilde{H}(t) = H(t, \beta_t, \gamma_t)$ ,  $\tilde{I}_m(t) = I_m(t, \beta_t)$  and  $\tilde{I}(t) = I(t, \beta_t)$ .

The main technical tool that we will need in order to study the evolution of those functionals will be, given  $\phi : \mathcal{C}^{\infty} \mapsto \mathcal{C}^{\infty}$ , quantities of the form :

$$\Gamma_{\phi,L}(h) = L\left(\phi(h)\right) - D_h\left(\phi\right)(h)L(h),\tag{4.16}$$

where  $D_h(\phi)$  is the pointwide differential of  $\phi$ , see [135]. The reason is that for regular enough h, by writing  $L_{\beta,\gamma}^K$  the generator (4.6) on  $M_K$  with fixed  $\beta$  and  $\gamma$ , we have

$$\partial_t \int_{M_K \times \mathbb{R}^d} \phi\left(e^{tL_{\beta,\gamma}^K}h\right) \mathrm{d}\mu_{\beta}^K = -\int_{M_K \times \mathbb{R}^d} \Gamma_{\phi,L_{\beta,\gamma}^K}\left(e^{tL_{\beta,\gamma}^K}h\right) \mathrm{d}\mu_{\beta}^K.$$

This kind of quantities has been studied in [135], where the author showed among other things :

**Lemma 4.4.3.** Let  $\phi(h) = |(\nabla_x + \nabla_y)h|^2 + \sigma h^2$  where  $\sigma = \frac{1}{2} + \sqrt{\gamma\beta^{-1}}(1 + ||\nabla U||_{\infty} + \gamma)^2$ . Then, for all  $\beta > 0, \gamma > 0, h \in \mathcal{C}^{\infty}$ :

$$\Gamma_{\phi, L^{K,*}_{\beta,\gamma}}(h) \ge \frac{1}{2} |\nabla h|^2 + \Gamma((\nabla_x + \nabla_y)h),$$

where  $\Gamma((\nabla_x + \nabla_y)h) = \sum_{i=0}^d \Gamma((\partial_{x_i} + \partial_{y_i})h) = \sum_{i=0}^d \gamma_t \beta_t^{-1} |\nabla_y(\partial_{x_i} + \partial_{y_i})h|^2.$ 

*Proof.* This is [135, Example 3]. Let us simply recall the key idea (to alleviate notations we omit the generator in the subscript of  $\Gamma$ ). First,  $\Gamma$  is linear in  $\phi$ , hence for  $\phi$  as defined above:

$$\Gamma_{\phi} = \sigma \Gamma + \Gamma_{|(\nabla_x + \nabla_y) \cdot|^2},$$

where  $\Gamma$  is the classical "carré du champs"  $\Gamma(h) = \gamma_t \beta_t^{-1} |\nabla_y h|^2$ . We then have the following equality:

$$\Gamma_{|(\nabla_x+\nabla_y)\cdot|^2}(h)=\Gamma((\nabla_x+\nabla_y)h)+(\nabla_x+\nabla_y)h\cdot[L,\nabla_x+\nabla_y]h,$$

where the brackets denotes the commutators of two operators. An elementary computation concludes.  $\hfill \square$ 

Proof of Proposition 4.2.3. First,  $\beta_0$  must be great enough so that we can apply all previous lemmas of Section 4.4. Let's first justify why  $\tilde{N}_m$ ,  $\tilde{N}$ ,  $\tilde{I}_m$  and  $\tilde{I}$  are finite. They are all finite at time 0 because  $f_0$  has compact support. For  $m \in \mathbb{N}$ ,  $\tilde{N}_m$  and  $\tilde{I}_m$  are finite for all times as integrals of continuous functions with compact support.

Write  $\nabla_t^* = -\nabla + \beta_t \nabla H_K$  the dual of the gradient operator  $\nabla$  in  $L^2(\mu_{\beta_t}^K)$ . Integrating by

parts,

$$\begin{split} \tilde{I}_m &= \int_{M_K \times \mathbb{R}^d} \eta_m \left| \nabla \frac{f_t^K}{\mu_{\beta_t}^K} \right|^2 \mathrm{d}\mu_{\beta_t}^K \\ &= \int_{M_K \times \mathbb{R}^d} \nabla_t^* \left( \nabla \eta_m \nabla \frac{f_t^K}{\mu_{\beta_t}^K} \right) f_t^K \mathrm{d}z \\ &\leqslant \int_{M_K \times \mathbb{R}^d} C_t e^{(\beta_t - \alpha) H_K(z)} f_t^K(z) \mathrm{d}z = C_t \mathbb{E} \left( e^{(\beta_t - \alpha) H_K(Z_t^K)} \right) \end{split}$$

for some  $\alpha > 0$  and  $C_t > 0$  locally bounded and independent from m, using Lemma 4.4.2 and that the derivatives of  $f_t^K$  are bounded. Lemma 4.4.1 and the monotonous convergence of  $\tilde{I}_m$ towards  $\tilde{I}$  then implies that  $\tilde{I}$  is locally bounded. We conclude that  $\tilde{N}$  is also locally bounded thanks to the Poincaré inequality of Proposition 4.4.1. As in (4.12), we have that the dual in  $L^2(\mu_{\beta}^K)$  of the generator  $L_t$  is

$$L_t^* = -y \cdot \nabla_x + (-\gamma_t y + \nabla_x U^K) \cdot \nabla_y + \gamma_t \beta_t^{-1} \Delta_y$$

and that  $f_t^K$  solves  $\partial_t f_t^K = L_t^*(f_t^K e^{\beta_t H_K}) e^{-\beta_t H_K}$ . With the regularity result from Lemma 4.4.2, and the compactly supported approximation functions  $\eta_m$ , one can differentiate  $N_m$  and  $I_m$  to get for all m:

$$\begin{aligned} (\partial_t N_m)(t,\beta_t,\gamma_t) &= \int_{M_K \times \mathbb{R}^d} \eta_m D_h \phi_{\beta_t,\gamma_t} \left(h_t^K\right) \frac{\partial_t f_t^K}{\mu_{\beta_t}^K} \mathrm{d}\mu_{\beta_t}^K \\ &= \int_{M_K \times \mathbb{R}^d} \eta_m D_h \phi_{\beta_t,\gamma_t} \left(h_t^K\right) L_t^* \left(h_t^K\right) \mathrm{d}\mu_{\beta_t}^K - \int_{M_K \times \mathbb{R}^d} L_t^* \left(\eta_m \phi_{\beta_t,\gamma_t} \left(h_t^K\right)\right) \mathrm{d}\mu_{\beta_t}^K \\ &= -\int_{M_K \times \mathbb{R}^d} \eta_m \Gamma_{\phi_{\beta_t,\gamma_t}} \left(h_t^K\right) \mathrm{d}\mu_{\beta_t}^K - \int_{M_K \times \mathbb{R}^d} L_t^* (\eta_m) \phi_{\beta_t,\gamma_t} \left(h_t^K\right) \mathrm{d}\mu_{\beta_t}^K \\ &- 2\int_{M_K \times \mathbb{R}^d} \Gamma_t \left(\eta_m, \phi_{\beta_t,\gamma_t} \left(h_t^K\right)\right) \mathrm{d}\mu_{\beta_t}^K. \end{aligned}$$

From Lemma 4.4.3, we get  $\Gamma_{\phi_{\beta,\gamma}}(h) \ge \frac{1}{2} |\nabla h|^2 + \Gamma_t((\nabla_x + \nabla_y)h)$ , and since

$$\begin{split} \int_{M_K \times \mathbb{R}^d} \Gamma_t \left( \eta_m, \phi_{\beta_t, \gamma_t} \left( h_t^K \right) \right) \mathrm{d}\mu_{\beta_t}^K \\ &= -\frac{1}{2} \int_{M_K \times \mathbb{R}^d} \left( \eta_m L_t^* \phi_{\beta_t, \gamma_t} \left( h_t^K \right) + L_t^* \eta_m \phi_{\beta_t, \gamma_t} \left( h_t^K \right) \right) \mathrm{d}\mu_{\beta_t}^K \\ &= -\frac{1}{2} \int_{M_K \times \mathbb{R}^d} \left( L_t + L_t^* \right) \left( \eta_m \right) \phi_{\beta_t, \gamma_t} \left( h_t^K \right) \mathrm{d}\mu_{\beta_t}^K \end{split}$$

and  $L_t \eta_m \leq \frac{C\beta_t}{m}$ , we get :

$$\begin{aligned} (\partial_t N_m)(t,\beta_t,\gamma_t) \leqslant &-\frac{1}{2} \int_{M_K \times \mathbb{R}^d} \eta_m \left| \nabla h_t^K \right|^2 \mathrm{d}\mu_{\beta_t}^K - \int_{M_K \times \mathbb{R}^d} \Gamma_t \left( (\nabla_x + \nabla_y) h_t^K \right) \mathrm{d}\mu_{\beta_t}^K \\ &+ \int_{M_K \times \mathbb{R}^d} L_t(\eta_m) \phi_{\beta_t,\gamma_t} \left( h_t^K \right) \mathrm{d}\mu_{\beta_t}^K \end{aligned}$$

$$\leq -\frac{1}{2} \int_{M_K \times \mathbb{R}^d} \eta_m \left| \nabla h_t^K \right|^2 \mathrm{d}\mu_{\beta_t}^K - \int_{M_K \times \mathbb{R}^d} \Gamma_t \left( (\nabla_x + \nabla_y) h_t^K \right) \mathrm{d}\mu_{\beta_t}^K \\ + \frac{C\beta_t}{m} \int_{M_K \times \mathbb{R}^d} \phi_{\beta_t, \gamma_t} \left( h_t^K \right) \mathrm{d}\mu_{\beta_t}^K.$$

Now we look at the derivative in  $\beta$ , knowing that  $\partial_{\beta} Z \leq 0$  by writting :

$$N_m(t,\beta,\gamma) = \int_{M_K \times \mathbb{R}^d} \left( \eta_m \left| (\nabla_x + \nabla_x) \ln \left( \frac{f_t^K}{\mu_\beta^K} \right) \right|^2 \frac{f_t^K}{\mu_\beta^K} + \eta_m \sigma(\beta,\gamma) \frac{f_t^K}{\mu_\beta^K} \right) \mathrm{d}f_t^K.$$

This gives

$$\begin{split} (\partial_{\beta}N_m)(t,\beta_t,\gamma_t) \\ \leqslant \partial_{\beta}\sigma(\beta_t,\gamma_t) \int_{M_K \times \mathbb{R}^d} \eta_m \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K + \sigma(\beta_t) \int_{M_K \times \mathbb{R}^d} \eta_m (U^K(x) + \frac{y^2}{2}) \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K \\ &+ \int_{M_K \times \mathbb{R}^d} \eta_m (U^K(x) + \frac{y^2}{2}) \left| (\nabla_x + \nabla_y) h_t^K \right|^2 \mathrm{d}\mu_{\beta_t}^K \\ &+ \int_{M_K \times \mathbb{R}^d} 2\eta_m (\nabla_x + \nabla_y) \ln \left(h_t^K\right) \cdot (\nabla_x + \nabla_y) \partial_{\beta} (-\ln(\mu_{\beta_t}^K)) h_t^K \mathrm{d}f_t^K \end{split}$$

We now have to treat all those terms. We will use that  $N_m \leq N$ . First, since  $U^K$  is bounded,

$$\partial_{\beta}\sigma(\beta_{t},\gamma_{t})\int_{M_{K}\times\mathbb{R}^{d}}\eta_{m}\left(h_{t}^{K}\right)^{2}\mathrm{d}\mu_{\beta_{t}}^{K}+\sigma(\beta_{t})\int_{M_{K}\times\mathbb{R}^{d}}\eta_{m}U^{K}(x)\left(h_{t}^{K}\right)^{2}\mathrm{d}\mu_{\beta_{t}}^{K}\leqslant C(1+\beta_{t}^{2})\tilde{N}$$

for some C > 0 (in the rest of the proof we denote by C several constants which do not depend on t nor m). From Proposition 4.4.2, applied with  $g = h_t^K$ , we get :

$$\sigma(\beta_t, \gamma_t) \int_{M_K \times \mathbb{R}^d} \eta_m \frac{y^2}{2} \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K \leqslant C(1+\beta_t^2)\tilde{N}.$$

Again, because  $U^K$  is bounded :

$$\int_{M_K \times \mathbb{R}^d} U(x) \left| (\nabla_x + \nabla_y) h_t^K \right|^2 \mathrm{d}\mu_\beta^K \leqslant C \tilde{N}$$

Using again Proposition 4.4.2 we get :

$$\int_{M_K \times \mathbb{R}^d} \eta_m \frac{y^2}{2} \left| (\nabla_x + \nabla_y) h_t^K \right|^2 \mathrm{d}\mu_{\beta_t}^K \leqslant C \left( \tilde{N} + \sqrt{\gamma_t^{-1} \beta_t} \int_{M_K \times \mathbb{R}^d} \Gamma_t \left( (\nabla_x + \nabla_y) h_t^K \right) \mathrm{d}\mu_{\beta_t}^K \right).$$

Finally, using that  $a.b \leqslant \frac{a^2}{2} + \frac{b^2}{2}$  and  $-\partial_\beta \ln(\mu_\beta^K) = U^K(x) + \frac{y^2}{2} + \frac{\partial_\beta Z_\beta}{Z_\beta}$ , we get :

$$\int_{M_K \times \mathbb{R}^d} \eta_m (\nabla_x + \nabla_y) \ln \left(\frac{f_t^K}{\mu_\beta^K}\right) \cdot (\nabla_x + \nabla_y) \partial_\beta (-\ln(\mu_{\beta_t}^K)) \frac{f_t^K}{\mu_{\beta_t}^K} \mathrm{d} f_t^K$$
$$\leqslant C \int_{M_K \times \mathbb{R}^d} \eta_m \left| (\nabla_x + \nabla_y) h_t^K \right|^2 \left(\frac{f_t^K}{\mu_{\beta_t}^K}\right)^2 \mathrm{d} \mu_{\beta_t}^K$$

$$+ C \int_{M_K \times \mathbb{R}^d} \eta_m |(\nabla_x + \nabla_y)(U(x) + \frac{y^2}{2})|^2 \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K \\ \leqslant C \left(N_m + \|\nabla U\|_{\infty}^2 \int \eta_m \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K + \int y^2 \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K\right)$$

and we conclude as the second term. Finally, since  $\gamma$  only appears in the definition of  $\sigma$ , we have :

$$(\partial_{\gamma} N_m)(t, \beta_t, \gamma_t) = \partial_{\gamma} \sigma \int_{M_K \times \mathbb{R}^d} \eta_m \left(h_t^K\right)^2 \mathrm{d}\mu_{\beta_t}^K \leqslant C \gamma_t \tilde{N}$$

From the previous computation, and from the fact that  $\gamma_t'\leqslant L\beta_t',$  we get :

$$\begin{split} N'_{m}(t) &= (\partial_{t}N_{m})(t,\beta_{t},\gamma_{t}) + \beta'_{t}(\partial_{\beta}N_{m})(t,\beta_{t},\gamma_{t}) + \gamma'_{t}(\partial_{\gamma}N_{m})(t,\beta_{t},\gamma_{t}) \\ &\leqslant -\frac{1}{2}\tilde{I}_{m}(t) - \int_{M_{K}\times\mathbb{R}^{d}}\eta_{m}\Gamma_{t}\left(Ah_{t}^{K}\right)\mathrm{d}\mu_{\beta_{t}}^{K} \\ &+ C\beta'_{t}\beta_{t}\int_{M_{K}\times\mathbb{R}^{d}}\Gamma_{t}\left(Ah_{t}^{K}\right)\mathrm{d}\mu_{\beta_{t}}^{K} + \frac{C\beta_{t}}{m}\tilde{N}_{t} + C\beta'_{t}(1+\beta_{t}^{2})\tilde{N}_{t} \end{split}$$

By integration, we get for all  $0 \leq s \leq t$ :

$$\tilde{N}_{m}(t) - \tilde{N}_{m}(s) \leq \int_{s}^{t} \left( -\frac{1}{2} \tilde{I}_{m}(u) + \frac{C\beta_{u}}{m} \tilde{N}_{u} + C\beta_{u}'(1+\beta_{u}^{2})\tilde{N}_{u} - \int_{M_{K} \times \mathbb{R}^{d}} \eta_{m} \Gamma_{t} \left(Ah_{t}^{K}\right) \mathrm{d}\mu_{\beta_{u}}^{K} + C\beta_{u}'\beta_{u} \int_{M_{K} \times \mathbb{R}^{d}} \Gamma_{t} \left(Ah_{t}^{K}\right) \mathrm{d}\mu_{\beta_{u}}^{K} \right) \mathrm{d}u.$$

We consider  $\beta_0$  great enough so that  $1 - C\beta'_t\beta_t \ge 0$  for all t. Using monotonous convergence, the fact that  $\tilde{I} \ge \tilde{I}_m$ , and Fatou's lemma we get :

$$\tilde{N}(t) - \tilde{N}(s) \leqslant \int_{s}^{t} \left( -\frac{1}{2}\tilde{I}(u) + C\beta'_{u}(1+\beta_{u}^{2})\tilde{N}_{u} \right) \mathrm{d}u.$$

Now, from the Poincaré inequality of Proposition 4.4.1 and the definition of  $\sigma,$ 

$$\tilde{\lambda}(\beta)\tilde{N}(t) \leqslant \tilde{I}(t)$$

for some  $\tilde{\lambda}$  satisfying  $\frac{1}{\beta} \ln(\tilde{\lambda}) \to -c^*$ , so that we can find a  $\lambda_0$  such that

$$\tilde{\lambda}(\beta_t) \geqslant \lambda_0 e^{-\beta_t \frac{(c+c^*)}{2}} = \frac{\lambda_0}{(e^{c\beta_0} + t)^{1-\alpha}}$$

where  $\alpha = (c - c^*)/(2c)$ . Taking  $\beta_0$  large enough so that  $C\beta'_t(1 + \beta_t^2) \leq \lambda_0/4(e^{c\beta_0} + t)^{-1+\alpha}$ , we have finally obtained that, if  $\beta_0 \geq \tilde{b}_K$  for some  $\tilde{b}_K$ , then for all  $0 \leq s \leq t$ 

$$\tilde{N}(t) - \tilde{N}(s) \leqslant \int_{s}^{t} \left( -\frac{\lambda_{0}}{4(e^{c\beta_{0}} + u)^{1-\alpha}} \tilde{N}(u) \right) \mathrm{d}u$$

and this concludes (as explained at the beginning of this section).

## 4.5 $H^k$ -hypocoercivity

In all this section, whose goal is to prove Proposition 4.2.7, we use the definitions and notations of Section 4.2.4. In particular,  $f_t^K$  stands for the law of the process defined by Equation (4.10),  $\mu_{\beta}^K = e^{-\beta H_K(z)} dz / \mathcal{Z}_{\beta}^K$  where  $\mathcal{Z}_{\beta}^K$  makes  $\mu_{\beta}^K$  a probability density on  $M_K$ , and  $h_t^K = f_t^K / \mu_{\beta_t}^K$ . According to [159],  $f_t^K$  is a smooth function and solves

$$\partial_t f_t^K = L_t^{K,*} \left( \frac{f_t^K}{\mu_{\beta_t}^K} \right) \mu_{\beta_t}^K,$$

but here the dual in  $L^2(\mu_{\beta_t})$  of the generator of the process is

$$L_t^{K,*} = -\beta_t^{-1}\sigma(y)\nabla_x^*\nabla_x - \gamma_t\beta_t^{-1}\nabla_y^*\nabla_y + \beta_t^{-1}\left(\nabla_x^*\nabla_y - \nabla_y^*\nabla_x\right),$$

where the dual are taken in  $L^2(\mu_{\beta}^K)$ :

$$\nabla_x^* = -\nabla_x \cdot + \beta \nabla U^K \cdot \\ \nabla_y^* = -\nabla_y \cdot + \beta \nabla W \cdot .$$

Indeed, the additional diffusive part in x has been designed to be reversible. In order to prove Proposition 4.2.7, we introduce for  $m \in \mathbb{N}$  the classical  $H^m$ -Sobolev norms on  $(M_K^2, \mu_\beta^K)$  given by

$$\|h\|_{H^m(\mu_\beta^K)} = \sum_{\alpha \in \mathbb{N}^{2d}, |\alpha| \leqslant m} \int_{M_K^2} |\partial^\alpha h|^2 \mathrm{d} \mu_\beta^K$$

The general strategy is similar to the  $L^2$  case of Section 4.4, namely we will prove that  $||h_t^K - 1||_{H^m}$ goes to zero for all m, and conclude by a Sobolev embedding for m large enough. The constant in the Sobolev embedding depends on the time t which should be compensated by the fact  $||h_t^K - 1||_{H^m}$  goes to zero fast enough. As in  $L^2$  case, we need to introduce some modified Sobolev norm to deal with the lack of dissipativity in the x variable in some part of the space. Following [170], for  $m \in \mathbb{N}$ , we consider a modified  $H^m$ -Sobolev norm of the form

$$N_{m}(t,\beta) = \int_{M_{K} \times M_{K}} \left(\frac{f_{t}^{K}}{\mu_{\beta}^{K}} - 1\right)^{2} + \sum_{k=1}^{m} \left(\sum_{i=0}^{k} \omega_{i,k}(\beta) \left|\nabla_{x}^{i} \nabla_{y}^{k-i} \frac{f_{t}^{K}}{\mu_{\beta}^{K}}\right|^{2} + \omega_{k}(\beta) \nabla_{x}^{k-1} \nabla_{y} \frac{f_{t}^{K}}{\mu_{\beta}^{K}} \cdot \nabla_{x}^{m} \frac{f_{t}^{K}}{\mu_{\beta}^{K}}\right) \mathrm{d}\mu_{\beta}^{K}$$

$$(4.17)$$

for some weights  $\omega$  to be fixed later on. Recall the definition

$$\mathcal{M} = \left\{ y \in M_K; \nabla^2 W(y) = I_d \right\}.$$

If  $y \in \mathcal{M}$ , then we obtain  $\nabla_x$  in the derivative of

$$\int_{M_K \times M_K} \nabla_x^{k-1} \nabla_y \frac{f_t^K}{\mu_\beta^K} \cdot \nabla_x^m \frac{f_t^K}{\mu_\beta^K} \mathrm{d}\mu_\beta^K$$

using the commutator  $[\nabla_y, \nabla W \cdot \nabla_x]$ , and for  $y \notin \mathcal{M}$ , it comes from the  $\sigma(y) \nabla_x^* \nabla_x$  part of  $L_t^{K,*}$ . Here we used the notations

$$\begin{aligned} |\nabla_x^i \nabla_y^{k-i} h|^2 &= \sum_{|\alpha_1|=i; |\alpha_2|=k-i} |\partial_x^{\alpha_1} \partial_y^{\alpha_2} h|^2 \\ \nabla_x^{k-1} \nabla_y h \cdot \nabla_x^m h &= \sum_{|\alpha|=k-1} \sum_{j=1}^d \partial_x^{\alpha} \partial_{y_j} h \partial_x^{\alpha} \partial_{x_j} h \end{aligned}$$

We also define  $\tilde{N}_m(t) = N_m(t, \beta_t)$ . Since the process is now in a compact set, we do not need the approximation functions  $\eta_m$ , and the subscript here indicates the order of the Sobolev norm in contrast to the previous section. Besides, in Section 4.4, we had to keep track of the dependency of the constant in  $\beta_0$ , as some uniformity in time was necessary in Proposition 4.2.5 for the renewal argument of the proof of Proposition 4.2.6. This is no longer the case here.

In order to study the evolution of  $N_m$ , we need first to have some commutation results and control over the derivative of the  $L^2$ -norm of  $\nabla_x^{m_1} \nabla_y^{m_2} h$  as in Lemma 4.4.3. Here,  $\nabla_x^{m_1} \nabla_y^{m_2} h$ denotes the vector of all derivative of h of order  $m_1$  on x and  $m_2$  on y:

$$\nabla_x^{m_1} \nabla_y^{m_2} h = \left\{ \partial_x^{\alpha_1} \partial_y^{\alpha_2} h \mid |\alpha_1| = m_1, \, |\alpha_2| = m_2 \right\}$$

The method used here is adapted from [170] to take into account the time-inhomogeneity and the new dynamic. Recall the notation (4.16) for generalized  $\Gamma$  operators.

**Lemma 4.5.1.** Let  $m_1, m_2 \in \mathbb{N}$ , then there exists some constant  $\theta > 0$  such that for all  $h \in C^{\infty}(M_K^2)$ :

$$\begin{split} \Gamma_{t,|\nabla_x^{m_1}\nabla_y^{m_2}\cdot|^2}(h) &\geqslant \beta_t^{-1} \left(\gamma_t |\nabla_x^{m_1}\nabla_y^{m_2+1}h|^2 + \sigma |\nabla_x^{m_1+1}\nabla_y^{m_2}h|^2 \right) \\ &\quad -\theta(1+\gamma_t) \sum_{\substack{k=1,l=1\\k+l\leqslant m_1+m_2}}^{m_1+1,m_2+1} |\nabla_x^k\nabla_y^l h|^2 - \theta \, \mathbbm{1}_{m_2\geqslant 1} \sum_{l=1}^{m_2} |\nabla^l \sigma|^2 |\nabla_x^{m_1+2}\nabla_y^{m_2-l}h|^2. \end{split}$$

*Proof.* As in [137, Lemma 10], we have for smooth h:

$$\Gamma_{t,|\nabla_x^{m_1}\nabla_y^{m_2}\cdot|^2}(h) = \Gamma_t(\nabla_x^{m_1}\nabla_y^{m_2}h) + \nabla_x^{m_1}\nabla_y^{m_2}h \cdot \left[L_t^{K,*}, \nabla_x^{m_1}\nabla_y^{m_2}\right]h.$$

Then first :

$$\Gamma_t(\nabla_x^{m_1} \nabla_y^{m_2} h) = \beta_t^{-1} \left( \gamma_t |\nabla_x^{m_1} \nabla_y^{m_2+1} h|^2 + \sigma |\nabla_x^{m_1+1} \nabla_y^{m_2} h|^2 \right).$$

Then, we can write  $L_t^{K,*} = v \cdot \nabla + \beta_t^{-1} \sigma \Delta_x + \beta_t^{-1} \gamma_t \Delta_y$  with the drift

$$v(x,y) = \begin{pmatrix} -\nabla W(y) - \sigma(y)\nabla U^{K}(x) \\ \nabla U^{K}(x) - \gamma_{t}\nabla W(y) \end{pmatrix}$$

First,

$$\beta_t^{-1} \gamma_t \Delta_y, \nabla_x^{m_1} \nabla_y^{m_2} ] = 0.$$

Then, using Cauchy-Schwarz inequality,

$$\nabla_x^{m_1} \nabla_y^{m_2} h \cdot \left[\beta_t^{-1} \sigma \Delta_x, \nabla_x^{m_1} \nabla_y^{m_2}\right] h = \nabla_x^{m_1} \nabla_y^{m_2} h \cdot \beta_t^{-1} \sum_{l=1}^{m_2} \binom{m_2}{l} \nabla_y^l \sigma \cdot \nabla_y^{m_2-l} \nabla_x^{m_1} \Delta_x h$$

#### 4.5. $H^k$ -hypocoercivity

$$\geq -\beta_t^{-1} \left( |\nabla_x^{m_1} \nabla_y^{m_2} h|^2 + \theta \sum_{l=1}^{m_2} |\nabla^l \sigma|^2 |\nabla_x^{m_1+2} \nabla_y^{m_2-l} h|^2 \right).$$

Notice that if  $m_2 = 0$ , then  $\nabla_x^m h \cdot \left[\beta_t^{-1} \sigma \Delta_x, \nabla_x^m\right] h = 0$ . The next term is:

$$\begin{bmatrix} \left(-\nabla W - \sigma \nabla U^K\right) \nabla_x, \nabla_x^{m_1} \nabla_y^{m_2} \end{bmatrix} h = \sum_{l=1}^{m_2} \binom{m_2}{l} \nabla_y^{l+1} W \nabla_x^{m_1+1} \nabla_y^{m_2-l} h + \sum_{\substack{k \leq m_1 \\ l \leq m_2 \\ (k,l) \neq (0,0)}} \binom{m_1}{k} \binom{m_2}{l} \nabla_y^l \sigma \nabla_x^{k+1} U \nabla_x^{m_1-k+1} \nabla_y^{m_2-l} h$$

and

$$\begin{bmatrix} \left(-\gamma_t \nabla W + \nabla U^K\right) \nabla_y, \nabla_x^{m_1} \nabla_y^{m_2} \end{bmatrix} h = \gamma_t \sum_{l=1}^{m_2} \binom{m_2}{l} \nabla_y^{l+1} W \nabla_x^{m_1} \nabla_y^{m_2-l+1} h \\ - \sum_{k=1}^{m_1} \binom{m_1}{k} \nabla_x^{k+1} U^K \nabla_x^{m_1-k} \nabla_y^{m_2+1} h.$$

Putting those last two lines together and using Cauchy-Schwarz we get :

$$\begin{split} \nabla_x^{m_1} \nabla_y^{m_2} h. \left[ v \cdot \nabla, \nabla_x^{m_1} \nabla_y^{m_2} \right] h \\ \geqslant -\theta (1+\gamma_t) \left( \sum_{l=1}^{m_2} |\nabla_x^{m_1+1} \nabla_y^{m_2-l} h|^2 + \sum_{\substack{1 \leqslant k \leqslant m_1 \\ 1 \leqslant l \leqslant m_2}} |\nabla_x^{m_1-k+1} \nabla_y^{m_2-l} h|^2 \\ + \sum_{l=1}^{m_2} |\nabla_x^{m_1} \nabla_y^{m_2-l+1} h|^2 + \sum_{k=1}^{m_1} |\nabla_x^{m_1-k} \nabla_y^{m_2+1} h|^2 \right). \end{split}$$

This concludes the proof.

**Lemma 4.5.2.** For all  $m \in \mathbb{N}$  there exists some constant  $\theta > 0$  such that for all smooth h, denoting  $P^m(t,\beta) = \int_{M_K \times M_K} \nabla_x^{m-1} \nabla_y \frac{f_t^K}{\mu_\beta^K} \cdot \nabla_x^m \frac{f_t^K}{\mu_\beta^K} d\mu_\beta^K$ :

$$\begin{aligned} (\partial_t P^m)(t,\beta_t) &\leqslant \int_{M_K \times M_K} - \left(\frac{1}{2} \mathbbm{1}_{\mathcal{M}} - \theta \mathbbm{1}_{\mathcal{M}^c}\right) |\nabla^m_x h^K_t|^2 \\ &+ \theta(1+\gamma_t) \left(\sum_{k=1}^{m-2} |\nabla^k_x h^K_t|^2 + (|\nabla\sigma|^2 + \sigma) |\nabla^{m+1}_x h^K_t|^2 + \sum_{k=1}^{m-1} |\nabla^k_x \nabla_y h^K_t|^2\right) d\mu^K_{\beta_t}. \end{aligned}$$

*Proof.* The derivative of  $P^m$  is given by :

$$(\partial_t P^m)(t,\beta_t) = \int_{M_K \times M_K} \left( \nabla_x^{m-1} \nabla_y L_t^{K,*} h_t^K \cdot \nabla_x^m h_t^K + \nabla_x^{m-1} \nabla_y h_t^K \cdot \nabla_x^m L_t^{K,*} h_t^K \right) \mathrm{d}\mu_{\beta_t}^K.$$

We then have to study the two terms in the integral separately at first, by using commutators.
For any smooth h :

$$\begin{split} \nabla_x^{m-1} \nabla_y L_t^{K,*} h \cdot \nabla_x^m h &= \nabla_x^{m-1} \left[ \nabla_y, L_t^{K,*} \right] h \cdot \nabla_x^m h \\ &+ \sum_{k=1}^{m-1} \nabla_x^{k-1} \left[ \nabla_x, L_t^{K,*} \right] \nabla_x^{m-k} \nabla_y h \cdot \nabla_x^m h + L_t^{K,*} \nabla_x^{m-1} \nabla_y h \cdot \nabla_x^m h \\ &= \nabla_x^{m-1} \left( -\nabla W \nabla_x - \nabla \sigma \nabla U^K \cdot \nabla_x - \gamma_t \nabla W \nabla_y + \beta_t \nabla \sigma \Delta_x \right) h \cdot \nabla_x^m h \\ &+ \sum_{k=1}^{m-1} \nabla_x^{k-1} \left( \nabla^2 U^K \nabla_y - \sigma \nabla^2 U^K \nabla_x \right) \nabla_x^{m-k} \nabla_y h \cdot \nabla_x^m h + L_t^{K,*} \nabla_x^{m-1} \nabla_y h \cdot \nabla_x^m h \\ &\leqslant - \left( \frac{1}{2} \mathbbm{1}_{\mathcal{M}} - \theta \mathbbm{1}_{\mathcal{M}^c} \right) |\nabla_x^m h|^2 + \theta \sum_{k=1}^{m-1} |\nabla_x^k h|^2 + \theta (1+\gamma_t) \sum_{k=1}^{m-1} |\nabla_x^k \nabla_y h|^2 \\ &+ \theta |\nabla \sigma|^2 |\nabla_x^{m+1} h|^2 + L_t^{K,*} \nabla_x^{m-1} \nabla_y h \cdot \nabla_x^m h \end{split}$$

where we used that  $|\nabla \sigma \nabla U^K| \leq \frac{1}{2}$ . Similarly, we have :

$$\nabla_x^{m-1} \nabla_y h \cdot \nabla_x^m L_t^{K,*} h \leqslant \theta \sum_{k=1}^{m-1} |\nabla_x^k \nabla_y h|^2 + \theta \sum_{k=1}^{m-1} |\nabla_x^k h|^2 + \nabla_x^{m-1} \nabla_y h \cdot L_t^{K,*} \nabla_x^m h.$$

We conclude with the fact that :

$$\int_{M_K \times M_K} L_t^* \nabla_x^{m-1} \nabla_y h_t^K \cdot \nabla_x^m h_t^K + \nabla_x^{m-1} \nabla_y h_t^K \cdot L_t^* \nabla_x^m h_t^K d\mu_{\beta_t}^K$$

$$= \int_{M_K \times M_K} \Gamma_t (\nabla_x^{m-1} \nabla_y h_t^K, \nabla_x^m h_t^K) d\mu_{\beta_t}^K$$

$$\leqslant \int_{M_K \times M_k} \theta (1+\gamma_t) \left( |\nabla_x^m \nabla_y h_t^K|^2 + |\nabla_x^{m-1} \nabla_x h_t^K|^2 + |\nabla_x^m \nabla_y|^2 \right) + \beta_t^{-1} \sigma |\nabla_x^{k+1} h|^2 d\mu_{\beta_t}^K.$$

In order to state the main lemma, we introduce the following set :

$$\mathcal{P} = \left\{ g : \mathbb{R}_+ \mapsto \mathbb{R}_+ \text{ smooth, } \exists a, C > 0 \text{ s.t. } \beta^{-a} / C \leqslant g(\beta), |g'(\beta)| \leqslant C\beta^a \right\},\$$

where  $\mathcal{P}$  stands for polynomial. The main step in order to prove an analogous to Proposition 4.2.3 in higher Sobolev norms is then the following dissipation result.

**Lemma 4.5.3.** For all  $m \ge 1$ , there exist  $q_m$ ,  $r_m$ ,  $(\omega_{i,m})_{i \le m}$  and  $\omega_m$  some functions of  $\beta$ , all in  $\mathcal{P}$ , such that if  $f_0^K \in \mathcal{C}^{\infty}(M_K \times M_K)$  then, for all  $t \ge 0$ ,

$$\|h_t^K - 1\|_{H^m\left(\mu_{\beta_t}^K\right)} \leqslant q_m(\beta_t)\tilde{N}_m(t)$$

and:

 $(\partial_t N_m)(t,\beta_t)$ 

$$\leq -r_m(\beta_t) \int_{M_K \times M_K} \left( \sigma |\nabla_x^{m+1} h_t^K|^2 + \sum_{i=0}^m |\nabla_x^i \nabla_y^{m+1-i} h_t^K|^2 + \sum_{1 \leq i+j \leq m} |\nabla_x^i \nabla_y^j h_t^K|^2 \right) d\mu_{\beta_t}^K.$$

$$(4.18)$$

*Proof.* The proof is by induction. In fact we start at m = 0, setting simply

$$N_0(t,\beta) = \int_{M_K \times M_K} \left(\frac{f_t^K}{\mu_\beta^K} - 1\right)^2 \mathrm{d}\mu_\beta^K.$$

Hence,

$$\begin{aligned} (\partial_t N_0)(t,\beta_t) &= -\int_{M_K \times M_K} \Gamma_t \left(\frac{f_t^K}{\mu_{\beta_t}^K}\right) \mathrm{d}\mu_{\beta_t}^K \\ &= -\int_{M_K \times M_K} \beta_t^{-1} \left(\gamma_t |\nabla_y h_t^K|^2 + \sigma |\nabla_x h_t^K|^2\right) \mathrm{d}\mu_{\beta_t}^K \end{aligned}$$

which is the result for m = 0 and  $r_0(\beta_t) = \beta_t^{-1} \min(1, \kappa)$ . One could also have initialized the induction for m = 1 as in Proposition 4.2.3. Now, let's fix  $m \in \mathbb{N}$  and suppose we have the result for m - 1:

$$(\partial_t N_{m-1})(t,\beta_t) \leqslant \\ -r_{m-1}(\beta_t) \int_{M_K \times M_K} \left( \sigma |\nabla_x^m h_t^K|^2 + \sum_{i=0}^{m-1} |\nabla_x^i \nabla_y^{m-i} h_t^K|^2 + \sum_{i+j \leqslant m-1} |\nabla_x^i \nabla_y^j h_t^K|^2 \right) \mathrm{d}\mu_{\beta_t}^K.$$

We set

~

$$N_m = N_{m-1} + \int_{M_K \times M_K} \left( \sum_{i=0}^m \omega_{i,m}(\beta) \left| \nabla_x^i \nabla_y^{k-i} \frac{f_t^K}{\mu_\beta^K} \right|^2 + \omega_m(\beta) \nabla_x^{m-1} \nabla_y \frac{f_t^K}{\mu_\beta^K} \cdot \nabla_x^m \frac{f_t^K}{\mu_\beta^K} \right) \mathrm{d}\mu_\beta^K$$

with weights to be determined later on, so that

$$(\partial_t N_m)(t,\beta_t) = (\partial_t N_{m-1})(t,\beta_t) - \int_{M_K \times M_K} \sum_{i=0}^m \omega_{i,m}(\beta) \Gamma_{|\nabla_x^i \nabla_y^{k-i} \cdot|^2}(h_t^K) \mathrm{d}\mu_{\beta_t}^K + \omega_m(\beta)(\partial_t P^m)(t,\beta_t).$$

In this equality, using the two previous lemmas, the terms of order m + 1 are bounded by:

$$\int_{M_K \times M_K} \left( \omega_{m-1,m} |\nabla \sigma|^2 + \omega_m (\sigma + |\nabla \sigma|^2) \right) |\nabla_x^{m+1} h_t^K|^2 - \beta_t^{-1} \sum_{i=0}^m \omega_{i,m}(\beta_t) \left( \gamma_t |\nabla_x^i \nabla_y^{m+1-i} h_t^K|^2 + \sigma |\nabla_x^{i+1} \nabla_y^{m-i} h_t^K|^2 \right) \mathrm{d}\mu_{\beta_t}^K$$

In order to get (4.18), we would like this to be less than

$$-r_m(\beta_t) \int_{M_K \times M_K} \left( \sigma |\nabla_x^{m+1} h_t^K|^2 + \sum_{i=0}^m |\nabla_x^i \nabla_y^{m+1-i} h_t^K|^2 \right) \mathrm{d}\mu_{\beta_t}^K.$$

Using that  $|\nabla \sigma|^2 \leq \sigma$ , this is indeed the case if we impose the conditions  $\omega_{m,m}(\beta) \geq 4\beta \omega_m(\beta) +$  $\beta \omega_{m_1,m}$  and  $\omega_{i,m}(\beta) \ge r_m \beta \max(\kappa^{-1}, \frac{1}{2})$ . Next, the terms of order at most m-1 are bounded bv :

$$\int_{M_K \times M_K} \theta(1+\gamma_t) \max_{0 \leqslant i \leqslant m} \omega_{i,m} \sum_{i+j \leqslant m-1} |\nabla_x^i \nabla_y^j h_t^K|^2 - r_{m-1} \sum_{i+j \leqslant m-1} |\nabla_x^i \nabla_y^j h_t^K|^2 \mathrm{d}\mu_{\beta_t}^K$$

which means, similarly, in order to get (4.18), we want to impose the conditions

$$\max_{0 \leqslant i \leqslant m} \omega_{i,m} \theta(1+\gamma_t) \leqslant \frac{1}{2} r_{m-1}, \quad r_{m-1} \geqslant 2r_m.$$

Finally, the terms of order exactly m are bounded by

$$-\int_{M_K \times M_K} \left( r_{m-1} \sum_{i=1}^{m-1} |\nabla_x^i \nabla_y^{m-i} h_t^K|^2 + \left( r_{m-1}\sigma + \omega_m \frac{1}{2} \mathbb{1}_{\mathcal{M}} - \omega_m \theta \mathbb{1}_{\mathcal{M}^c} \right) |\nabla_x^m h_t^K|^2 - \max_i \omega_{i,m} \theta (1+\gamma_t) \sum_{i=0}^m |\nabla_x^i \nabla_y^{m-i} h_t^K|^2 \right) \mathrm{d}\mu_{\beta_t}^K$$

which leads to the conditions:  $r_m \leq \frac{1}{2}\omega_m, \omega_m \theta \leq \frac{1}{2}r_{m-1}\sigma_*$  and  $\frac{1}{2}r_{m-1}\sigma_* \geq r_m$ . Now, fix the  $\omega_{i,m}$  to be equal to  $\frac{r_{m-1}}{4\theta(1+L\beta)}$  except  $\omega_{m,m}$  equal to  $\frac{r_{m-1}}{2\theta(1+L\beta)}$ , then  $\omega_m = \min(\frac{\sigma_*r_{m-1}}{2\theta}, \frac{\omega_{m,m}}{4\beta}, \omega_{m,m})$ , and finally set  $r_m = \min(\frac{r_{m-1}}{2}, \frac{r_{m-1}\sigma_*}{2}, \frac{\omega_m}{2}, \beta^{-1}\min(\kappa, 2)\omega_{i,m})$ . These choices ensures that all the conditions in the computations above are met, which means that (4.18) holds. Moreover, all these functions are in  $\mathcal{P}$ , which concludes. 

Similarly to Proposition 4.4.1 but now in the fully (position and velocity) compact case, we have the following Poincaré inequality:

**Proposition 4.5.1.** If  $U^K : M_K \mapsto \mathbb{R}$  is a  $\mathcal{C}^{\infty}$  function, W is as constructed in Section 4.2.4, and  $\mu_{\beta}^{K}$  is the probability measure on  $M_{K} \times M_{K}$  proportional to  $e^{-\beta H_{K}}$ , then there exists  $\lambda$ :  $\mathbb{R}_+ \mapsto \mathbb{R}_+$  such that for all  $f \in \mathcal{C}^{\infty}(M_K \times M_K)$ :

$$\lambda(\beta) \int_{M_K \times M_K} \left( f - \int_{M_K \times M_K} f \, d\mu_\beta^K \right)^2 d\mu_\beta \leqslant \int_{M_K \times M_K} |\nabla f|^2 d\mu_\beta^K$$

and

$$\lim_{\beta \to \infty} \frac{1}{\beta} \ln(\lambda(\beta)) = -c^*(U^K)$$

where  $c^*(U^K)$  is defined as  $c^*$  with U replaced by  $U^K$ .

*Proof.* Using that  $c^*(H_K) = c^*(U^K)$  since W has only one minimum, this is [83, Theorem 1.14]. 

We are now ready to prove Proposition 4.2.7.

Proof of Proposition 4.2.7. Let m = d+1. We first show that there exist some constants  $C_1, C_2 > 0$  such that  $\tilde{N}_m(t) \leq C_1 e^{-C_2 t^{\alpha}}$  for all  $t \geq 0$ , where  $\alpha = \frac{c-c^*}{2c}$ . Indeed, we have :

$$\tilde{N}'_m(t) = \partial_t N_m(t,\beta_t) + \beta'_t \partial_\beta N_m(t,\beta_t)$$

From Lemma 4.5.3 and Proposition 4.5.1,

$$\partial_t N_m \leqslant -\frac{\hat{\lambda}(\beta_t)r_m}{2}N_m.$$

where  $\tilde{\lambda}$  satisfies  $\lim_{\beta\to\infty}\frac{1}{\beta}\ln(\tilde{\lambda}(\beta)) = -c^*(U^K)$  and  $r_m \geq \beta^{-a}/C$  for some a > 0. Hence, we have some constant C > 0 such that  $\lambda(\beta_t)r_m/2 \geq Ce^{-\frac{c+c^*}{2}\beta_t} = C(e^{c\beta_0} + t)^{-1+\alpha}$ . On the other hand, from the conditions on the  $\omega$ 's and since  $\partial_\beta e^{\beta H_K} \leq ||H_K||_{\infty} e^{\beta H}$ , there exist some a, C' > 0 such that  $\partial_\beta N_m \leq C'\beta_t^a N_m$  yielding  $\beta'_t \partial_\beta N_m \leq C'(1+t)^{-1+\frac{\alpha}{2}}N_m$  for some other constant C' > 0, and thus :

$$\tilde{N}'_m \leqslant \left( -\frac{C}{(1+t)^{1-\alpha}} + \frac{C'}{(1+t)^{1-\frac{\alpha}{2}}} \right) \tilde{N}_m,$$

yielding our first claim.

From the Sobolev embedding, there exists C > 0 such that for any smooth h on  $M_K \times M_K$ , we have  $\|h\|_{L^{\infty}} \leq C \|h\|_{H^m(\mu_{\infty}^K)}$ . Then we have :

$$\|h\|_{L^{\infty}} \leq 1 + \|h - 1\|_{L^{\infty}} \leq 1 + C\|h - 1\|_{H^{m}\left(\mu_{\beta_{0}}^{K}\right)} \leq 1 + Ce^{\beta_{t}\|H\|_{\infty}}\|h - 1\|_{H^{m}\left(\mu_{\beta_{t}}^{K}\right)}.$$

Applying this with  $h = h_t^K$  and using that  $\|h_t^K - 1\|_{H^m(\mu_{\beta_t}^K)} \leq C \beta_t^a \tilde{N}_m(t)$  for some a, C > 0, we get constants C, b > 0 such that :

$$\|h_t^K\|_{L^{\infty}} \leq 1 + C(1+t)^b e^{-C_2 t^{\alpha}},$$

yielding the result.

#### 4.6 Faster than logarithmic schedules

*Proof of Theorem 4.1.3.* Some parts of the proof follow the proof of Theorem 4.1.2, to which we refer for details, focusing on the new arguments in the present settings.

Up to a translation we assume without loss of generality that  $x_* = 0$ .

As in the proof of Theorem 4.1.2, it is in fact sufficient to prove that for all  $\delta > 0$ 

$$\mathbb{P}_{f_0}\left(|Z_t| \leqslant \min(\delta, \sqrt{\varepsilon_t}) \ \forall t \ge t_0\right) > 0$$

for a given fixed initial condition  $f_0$  and for  $t_0$  large enough and then use a controlability argument based on Lemma 4.2.8 and the Markov property to get the claimed result. Besides, it is sufficient to prove this result for  $\delta$  small enough and, since the times in  $[0, t_0]$  are treated with the controlability argument, focusing on the times larger than  $t_0$ , we are interested in an event under which the process stays in a small ball around 0, and we can thus modify U outside such a ball without modifying the result.

The Jacobian matrix of the drift  $b(z) = (y, -\nabla U(x) - \gamma y)$  of the process at z = (x, y) is

$$J(z) = \begin{pmatrix} 0 & I \\ -\nabla^2 U(x) & -\gamma I \end{pmatrix} \,.$$

Since  $\nabla^2 U(0)$  is positive semi-definite, a simple calculation shows that the eigenvalues of J(0) all have a negative real part, and thus there exist a positive definite symmetric matrix M of size 2d and r > 0 such that

$$u \cdot MJ(0)u \leqslant -2ru \cdot Mu$$

for all  $u \in \mathbb{R}^{2d}$ , see e.g. [4, Lemma 4.3]. Write  $||u||_M = \sqrt{u \cdot Mu}$  and let  $\delta > 0$  be sufficiently small so that

$$u \cdot MJ(z)u \leqslant -r \|u\|_M^2 \tag{4.19}$$

for all  $u \in \mathbb{R}^{2d}$  and all  $z \in \mathbb{R}^{2d}$  with  $||z||_M \leq \delta$ . As discussed above, up to a modification of U outside the ball  $\mathcal{B}(0, \delta)$ , without loss of generality we can modify the potential U outside this ball and assume that in fact (4.19) holds for all  $z \in \mathbb{R}^{2d}$ . Similarly we assume that  $\nabla^3 U$  is bounded.

Using that b(0) = 0, we get that for all  $z \in \mathbb{R}^{2d}$ 

$$z \cdot Mb(z) = z \cdot M(b(z) - b(0)) = \int_0^1 z \cdot MJ(pz) z dp \leqslant -r ||z||_M^2.$$
(4.20)

Let Z and  $\tilde{Z}$  be two solutions of (4.1) with the same initial condition and driven by the same Brownian motion, but with two different potentials, Z being associated to U and  $\tilde{Z}$  to  $\tilde{U}(z) = z \cdot \nabla^2 U(0) z/2$ . In particular,  $\tilde{Z}$  is a Gaussian process. Then

$$\begin{aligned} \mathbf{d} \| Z_t - \tilde{Z}_t \|_M^2 &\leqslant -2r \| Z_t - \tilde{Z}_t \|_M^2 \mathbf{d}t + 2 \| Z_t - \tilde{Z}_t \|_M |M^{1/2}| |\nabla U(\tilde{Z}_t) - \nabla^2 U(0) \tilde{Z}_t | \mathbf{d}t \\ &\leqslant -r \| Z_t - \tilde{Z}_t \|_M^2 \mathbf{d}t + \frac{1}{4r} \| \nabla^3 U \|_\infty^2 |M| |\tilde{Z}_t|^4 \mathbf{d}t \,. \end{aligned}$$

Let  $(\alpha_t)_{t \ge 0}$  be a positive non-increasing function, vanishing at infinity, to be chosen later on. Then the event  $\mathcal{G} = \{ | \tilde{Z}_t | \le \alpha_t \ \forall t \ge 0 \}$  implies that for all  $t \ge 0$ 

$$|Z_t| \leq |\tilde{Z}_t| + |M^{-1/2}| ||Z_t - \tilde{Z}_t||_M \leq \alpha_t + C \left(\int_0^t \alpha_s^2 e^{r(s-t)} \mathrm{d}s\right)^{1/2} := \tilde{\alpha}_t$$

with  $C^2 = |M| |M^{-1}| ||\nabla^3 U||_{\infty} / (2\sqrt{r})$ . Notice that  $\tilde{\alpha}_t$  vanishes at infinity, more precisely  $\tilde{\alpha}_t \leq \alpha_t + C(\alpha_{3t/4}/\sqrt{r} + \alpha_0\sqrt{t}e^{-rt/8})$ . As a consequence, it only remains to prove that  $\mathcal{G}$  has a positive probability for some suitable function  $t \mapsto \alpha_t$ . Again, following the proof of Theorem 4.1.2, we see that it is sufficient to prove that  $t \mapsto \mathbb{P}(|\tilde{Z}_t| \geq \alpha_t)$  is integrable for a given fixed initial condition. Since  $\tilde{Z}$  is a Gaussian process, we simply have to control its second moment. We chose an initial condition such that  $\mathbb{E}(\tilde{Z}_0) = 0$ , so that  $\mathbb{E}(\tilde{Z}_t) = 0$  for all  $t \geq 0$ .

Remark that the drift of  $\tilde{Z}_t$  also satisfies (4.20), since its Jacobian matrix at any  $z \in \mathbb{R}^{2d}$  is equal to J(0). Thus we get, for all  $t \ge 0$ ,

$$\partial_t \mathbb{E}\left(\|\tilde{Z}_t\|_M^2\right) \leqslant -r \mathbb{E}\left(\|\tilde{Z}_t\|_M^2\right) + \frac{d|M|}{\beta_t},$$

so that, writing  $C' = |M| |M^{-1}| (\mathbb{E} |\tilde{Z}_0|^2 + d),$ 

$$\mathbb{E}\left(|\tilde{Z}_t|^2\right) \leqslant |M^{-1}|\mathbb{E}\left(\|\tilde{Z}_t\|_M^2\right) \leqslant C'\left(e^{-rt} + \int_0^t \frac{e^{r(s-t)}}{\beta_s} \mathrm{d}s\right) := C'\kappa_t,$$

Using that the law of  $\tilde{Z}_t$  is Gaussian with zero mean, we get that there exist K, h > 0 such that for all  $t \ge 0$ 

$$\mathbb{P}(|\tilde{Z}_t| \ge \alpha_t) \le K \exp\left(-h\frac{\alpha_t^2}{\kappa_t}\right).$$

This is integrable in time if we chose  $\alpha_t^2 = \sup_{s \ge t} 2 \ln(s) \kappa_s / h$  (which is non-increasing). Since  $\ln(t) = o(\beta_t)$  and  $\kappa_t \le \sup_{s \ge 2t/3} 1/(r\beta_s) + e^{-rt/3}(1 + t \sup_{u \ge 0} 1/\beta_u)$ , we get that  $\kappa_t = o(1/\ln(t))$ , in other words  $\alpha_t \to 0$ . We conclude by using the previous bounds on  $\tilde{\alpha}_t$  and  $\kappa_t$  and the fact that

$$\sup_{s \ge 3t/4} \ln(s) \sup_{u \ge 2s/3} \beta_u^{-1} \leqslant \sup_{s \ge t/2} \ln(s) \beta_s^{-1}$$

to get the quantitative convergence speed  $\varepsilon_t$  stated in the theorem.

**Remark:** in order to adapt the previous proof in a case where  $\gamma_t$  is not constant, we can still find M and r such that (4.19) holds with the Jacobian matrix of the drift at time t, but they depend on  $\gamma_t$ , hence when differentiating  $\|Z_t - \tilde{Z}_t\|_M^2$  there is an additional term involving  $\partial_t M$ which has to be sufficiently small to be absorbed by the contraction at rate r. Moreover, r scales as  $\gamma_t$  when  $\gamma_t \to 0$  and as  $1/\gamma_t$  when  $\gamma_t \to +\infty$ , which means for the proof to be valid,  $\gamma_t$  or  $1/\gamma_t$ should not be too small depending on  $\beta_t$ . 132

### Chapter 5

# Switched diffusion processes for non-convex optimization and saddle points search

#### Outline of the current chapter

5.1	Introduction	133
	5.1.1 Overview	133
	5.1.2 Deterministic saddle point algorithms	135
	5.1.3 The stochastic processes $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	136
5.2	Theoretical analysis	138
	5.2.1 Noisy ISD and Noisy GAD	138
	5.2.2 Switched processes $\ldots$	143
5.3	Numerical experiments	143
	5.3.1 First 2D model: a mixture of Gaussians	144
	5.3.2 A 2D model with singular lines	148
	5.3.3 Lennard-Jones clusters	152

**Abstract**: We introduce and investigate stochastic processes designed to find local minimizers and saddle points of non-convex functions, exploring the landscape more efficiently than the standard noisy gradient descent. The processes switch between two behaviours, a noisy gradient descent and a noisy saddle point search. It is proven to be well-defined and to converge to a stationary distribution in the long time. Numerical experiments are provided on low-dimensional toy models and for Lennard-Jones clusters.

#### 5.1 Introduction

#### 5.1.1 Overview

This work addresses the issue of finding the local minimizers and the saddle points of a given non-convex potential  $U : \mathbb{R}^d \to \mathbb{R}_+$ . These questions are ubiquituous in a wide range of scientific

fields. In particular, in molecular dynamics, U being the energy of a molecular system, the minimizers are the metastable state of the system, while the saddle points are the transition states between the former. As we will see, we will design a stochastic algorithm which targets both local minimizers and saddle points, in such a way that each of this task benefits from the other. In particular, the algorithm is suitable even as an optimisation tool, when the goal is only to find the global minimizers of U.

In high dimension, this problem is challenging, as an exhaustive search is impossible and running local search algorithms started from random initial conditions only works when it is possible to sample starting points that have a reasonable probability to hit the basin of attraction of the points of interest (for a gradient descent or a saddle point search), which is often not the case when these basins are concentrated around an unknown manifold of dimension much smaller than d.

A classical solution in order to visit the whole space while taking into account the energy landscape (hence focusing on low-energy areas) is to follow the overdamped Langevin diffusion process, which is the Markov process solving

$$\mathrm{d}X_t = -\nabla U(X_t)\mathrm{d}t + \sqrt{2\varepsilon}\mathrm{d}B_t\,,\tag{5.1}$$

where B is a d-dimensional Brownian motion and  $\varepsilon > 0$  is a temperature parameter. When  $\varepsilon$  is small, this is a noisy perturbation of a deterministic gradient descent (contrary to the stochastic gradient descent algorithm, here the noise is voluntarily added, it has a constant intensity and does not come from a Monte Carlo approximation of  $\nabla U$ ). Under standard growth conditions on U at infinity, the process is ergodic with respect to the measure

$$\mu_{\varepsilon}(\mathrm{d}x) = \frac{e^{-U(x)/\varepsilon}\mathrm{d}x}{\int_{\mathbb{R}^d} e^{-U(y)/\varepsilon}\mathrm{d}y},$$

i.e. the time spent in any given domain is asymptotically proportional to its probability with respect to  $\mu_{\varepsilon}$ . Hence, the temperature  $\varepsilon$  has to be taken small enough to counteract the entropic effect due to the high dimension d and ensure that the vicinity of minimizers have a high probability. Then, one can use the point of the trajectory with the lowest energy as a starting point for a local gradient descent (or a variant, like the stochastic gradient descent). However, as  $\varepsilon$  vanishes, it is well known that (5.1) is metastable, since transitions between different potential wells occur in a time of order roughly  $e^{c/\varepsilon}$  for some c > 0, due to the energy barriers induced by low-probability regions interspersing high-probability ones. This makes the convergence of the law of the process towards its equilibrium, and thus the exploration, very slow, which makes this method fail in practice.

Besides, according to large deviations theory, more precisely to the Freidlin-Wentzell results on the low noise behaviour of diffusion processes, it is known that, as  $\varepsilon$  vanishes, the process (5.1) is likely to exit from a potential well through the saddle point with the lowest energy level on the boundary of the well. Instead of waiting for an unlikely deviation of the Brownian motion to lead the process (5.1) to this saddle point, this naturally motivates the definition of a process which actively looks for it. More precisely, the main object of this work is a process which switches randomly between two dynamics, the noisy gradient descent (5.1) and a noisy saddle points search.

The rest of this work is organized as follows. In the remainder of the introduction, we present two saddle point search algorithm in Section 5.1.2, and we define the stochastic processes we are going to investigate. In Section 5.2, we conduct a theoretical study of the processes, and in Section 5.3 a numerical study.

#### 5.1.2 Deterministic saddle point algorithms

Since saddle point search is less standard than optimization, let us first present some existing deterministic algorithms on this topic. A first class of methods, based on reaction paths, requires the knowledge of two local minimizers, and then finds a path of minimal elevation between them, which passes through a saddle point see e.g. [70, 53]. This doesn't correspond to our context, where the local minimizers are unknown. A second class of methods, considered e.g. in [54, 121, 66, 115, 151, 17, 75] and references within and which will be the one of interest for us, relies on local walkers, analogous of the gradient descent for optimization, i.e. solutions of some ODEs for which saddle points of U are stable equilibrium. Given some  $x \in \mathbb{R}^d$ , write:

$$\lambda_1(x) \leqslant \lambda_2(x) \leqslant \cdots \leqslant \lambda_d(x)$$

for the ordered eigenvalues of  $\nabla^2 U(x)$ , and suppose that we are given  $v_1(x), \ldots, v_d(x) \in \mathbb{R}^d$  such that  $v_i(x)$  is an eigenvector of  $\nabla^2 U(x)$  associated to  $\lambda_i(x)$  (in other words, when the eigenvalues are not all simple, we assume that we have an arbitrary rule to select a given eigenbasis, for instance we take the basis obtained as the limit of the Jacobi algorithm). The idealised saddle dynamic (ISD) is the solution of

$$\dot{x}_t = -\left(I - 2v_1(x_t)v_1(x_t)^T\right)\nabla U(x_t).$$

In other words,  $x_t$  follows a gradient descent, except in the direction  $v_1(x_t)$  where a reflection is performed, i.e. the process follows a gradient ascent in this direction. Notice that in general  $x \mapsto v_1(x)$  is not continuous so the existence of this process is unclear and may be restricted to some parts of the space or to a finite time interval.

A variation of the ISD is the so-called Gentlest Ascent Dynamic (GAD)  $(x_t, v_t) \in \mathbb{R}^d \times \mathbb{S}^{d-1}$ solving

$$\begin{cases} \dot{x}_t = -(I - 2v_t v_t^T) \nabla U(x_t) \\ \eta \dot{v}_t = -(I - v_t v_t^T) \nabla^2 U(x_t) v_t \end{cases}$$

where  $\eta > 0$  is a fixed parameter. If  $\nabla^2 U(x_t)$  were fixed in the second equation, this would be a gradient descent on  $\mathbb{S}^{d-1}$  for the Rayleigh quotient  $v \mapsto -v^T \nabla^2 U(x_t) v$ . As  $\eta$  vanishes, formally, we recover the ISD. Contrary to the later, the GAD is always well-defined for all initial conditions and all times.

The following results on the ISD and GAD are from [115, Theorems 2 and 3] and [54].

**Proposition 5.1.1.** • Any critical point of U is an equilibrium points for the ISD. It is stable if and only if it is a index-1 saddle point.

- Any  $(x, v) \in \mathbb{R}^d \times \mathbb{S}^{d-1}$  with x a critical point of U and v an eigenvalue of  $\nabla^2 U$  is an equilibrium point of the Gentlest Ascent dynamic. It is stable if and only if x is a index-1 saddle point and v is associated to the negative eigenvalue.
- If  $\Omega \subset \mathbb{R}^d$  satisfies that for all  $x \in \Omega$ , we have  $\lambda_1(x) < 0 < \lambda_2(x)$ , then there exists a unique saddle point  $z_*$  in  $\Omega$ , and for all  $x_0 \in \Omega$ , there exists a unique solution to the ISD which converges exponentially fast towards  $z_*$ .
- Under the same assumption, there is  $\delta > 0$  such that if  $(x_0, v_0) \in \Omega \times \mathbb{S}^{d-1}$ , and  $|v_0 v_1(x_0)| < \delta$ , then there exists a unique solution to the GAD which converges exponentially fast towards  $(z_*, v_1(z_*))$ .

The convergence results are only local, and one cannot expect more in general since the ISD may have attracting singularities [115]. In the case of the GAD, with a potential of the form

 $\sum_{i=1}^{d} U_i(x_i)$ , and an initial condition  $v_0 \in \text{vect}(e_1, \dots e_i)$ ,  $v_t$  will stay orthogonal to  $e_j$ , j > i, and hence the process will never get close to an equilibrium point with an eigenvector not in  $\text{vect}(e_1, \dots e_i)$ . As in the optimization case where the overdamped Langevin diffusion (5.1) allows for a global exploration while the gradient descent only converges locally, it is thus natural to add a small Brownian noise to the deterministic ISD and GAD. The resulting processes will explore eventually the whole space but, thanks to Proposition 5.1.1, will still be attracted locally by saddle points (notice that the Freidlin-Wentzell theory on random perturbation of ODE does not apply directly in the ISD case because of the discontinuity of  $x \mapsto v_1(x)$ ).

#### 5.1.3 The stochastic processes

For simplicity, from now on, we will restrict ourselves to the case of a function defined on compact manifold (specifically the periodic torus). Indeed, in practical situation, local minima and saddle points are located in a compact set. Most of the definitions and arguments below would be easily adapted to  $\mathbb{R}^d$  provided suitable modifications outside a given compact, in particular to ensure the stability of the processes.

Write  $\mathbb{T}^d$  for the d-dimensional torus and fix some function  $U \in \mathcal{C}^2(\mathbb{T}^d, \mathbb{R}_+)$ . As in the previous section, we write  $\lambda_i(x)$  the ordered eigenvalues of  $\nabla^2 U(x)$ , and we consider functions  $x \mapsto v_i(x)$  such that  $v_i(x)$  is an eigenvector of  $\nabla^2 U(x)$  associated to  $\lambda_i(x)$  for all  $x \in \mathbb{T}^d$  and  $i \in [\![1,d]\!]$ . Moreover, we assume that the functions  $v_i$  are measurable for all  $i \in [\![1,d]\!]$ , which is indeed possible:

**Lemma 5.1.1.** For all  $i \in [\![1,d]\!]$  there exists a measurable function  $v_i : \mathbb{T}^d \to \mathbb{S}^{d-1}$  such that  $v_i(x)$  is and eigenvector of  $\nabla^2 U(x)$  associated to  $\lambda_i(x)$  for all  $x \in \mathbb{T}^d$ .

Proof. Given a symmetric matrix H and  $(e_1, ..., e_d)$  the canonical basis of  $\mathbb{R}^d$ , the function  $H \mapsto (u_1, \mu_1, ..., u_d, \mu_d)$ , where  $u_i$  is the limit of the gradient descent for the Rayleigh quotient starting from  $e_i$  and  $\mu_i$  is the associated eigenvalue, is measurable. At least one of the  $\mu_i$  is the smallest eigenvalue of H, hence we may just select the smallest indices  $i_0$  such that it is the case and consider the associated vector. We can then iterate by applying the same procedure to  $(I - u_{i_0}u_{i_0}^T)H(I - u_{i_0}u_{i_0}^T) + (|H| + 1)u_{i_0}u_{i_0}^T$ , and similarly by induction. We conclude with the fact that  $x \mapsto \nabla^2 U(x)$  is continuous, hence measurable.

As in  $\mathbb{R}^d$ , for  $\varepsilon > 0$  the overdamped Langevin process is defined on  $\mathbb{T}^d$  as the solution of (5.1). It is ergodic with respect to  $\mu_{\varepsilon}$  the probability measure with density proportional to  $e^{-U/\varepsilon}$  on the torus. Similarly, we define the noisy ISD as the solution on  $\mathbb{T}^d$  of

$$dX_t = -\left(I - 2v_1(X_t)v_1(X_t)^T\right)\nabla U(X_t)dt + \sqrt{2\varepsilon}dB_t, \qquad (5.2)$$

which, contrary to its deterministic counterpart, is always well-defined (see Proposition 5.2.1 below). Concerning the noisy version of GAD, we may wonder whether it is necessary to add noise to both coordinates, and thus in general we can consider the noisy GAD as the solution on  $\mathbb{T}^d \times \mathbb{S}^{d-1}$  of

$$\begin{cases} dX_t = -\left(I - 2V_t V_t^T\right) \nabla U(X_t) dt + \sqrt{2\varepsilon} dB_t, \\ dV_t = \left(I - V_t V_t^T\right) \left[-\frac{1}{\eta} \nabla^2 U(X_t) V_t dt + \sqrt{2\varepsilon'} dB_t'\right] - d\varepsilon' dt, \end{cases}$$
(5.3)

where  $\eta > 0$ ,  $\varepsilon' \ge 0$  and B, B' are independent *d*-dimensional Brownian motions. The second line is such that if  $\nabla^2 U(X_t)$  is constant then this is an overdamped Langevin diffusion on the sphere with potential given by the Rayleigh quotient (see e.g. [110, Section 3.2.3] or [2]). In particular, according to [110, Lemma 3.18],  $V_t \in \mathbb{S}^{d-1}$  for all  $t \ge 0$ . In fact, by contrast with the deterministic case, since the noisy ISD is well defined, we will not insist much on the noisy GAD, and simply discuss the case  $\varepsilon' = 0$  in Section 5.2.1.

Next, we define the idealised switched process (ISP) as the Markov process  $(X_t, I_t)_{t \ge 0}$  on  $\mathbb{T}^d \times \{0, 1\}$  where  $(I_t)_{t \ge 0}$  is a Markov process on  $\{0, 1\}$  with constant jump rate  $\nu > 0$  and X solves

$$dX_t = H_{I_t}(X_t)dt + \sqrt{2\varepsilon}dB_t, \qquad (5.4)$$

with  $(B_t)_{t\geq 0}$  a Brownian motion independent from  $(I_t)_{t\geq 0}$  and, for all  $x\in\mathbb{T}^d$ ,

$$H_0(x) = -\nabla U(x), \qquad H_1(x) = -(I - 2v_1(x)v_1(x)^T)\nabla U(x)$$

In other words, the process alternates between the overdamped Langevin dynamics and the noisy ISD, the time being two switching events between independent and distributed according to an exponential law of parameter  $\nu$ . Likewise, we can define the gentle switched process (GSP) as the Markov process  $(X_t, V_t, I_t)_{t\geq 0}$  on  $\mathbb{T}^d \times \mathbb{S}^{d-1} \times \{0, 1\}$  where

$$\begin{cases} dX_t = \overline{H}_{I_t}(X_t, V_t) + \sqrt{2\varepsilon} dB_t, \\ dV_t = (I - V_t V_t^T) \left[ -\frac{1}{\eta} \nabla^2 U(X_t) V_t dt + \sqrt{2\varepsilon'} dB_t' \right] - d\varepsilon' dt, \end{cases}$$
(5.5)

where  $(I_t)_{t \ge 0}$  is as in the ISP, independent from B, B', and

$$\overline{H}_0(x,v) = -\nabla U(x), \qquad \overline{H}_1(x,v) = -(I - 2vv^T)\nabla U(x).$$

Finally, let us introduce slight variations of the processes defined above. Consider

$$\mathcal{S} = \left\{ x \in \mathbb{T}^d, \lambda_1(x) = \lambda_2(x) \right\},\$$

which is the set of singularities of the deterministic ISD flow. As pointed out in [115], some points of S may be attracting singularities for this flow, and thus they will also locally attract the noisy ISD when  $\varepsilon$  is small, or the noisy GAD when  $\varepsilon$  and  $\eta$  are small. This behaviour may be mitigated as follows. Fix some non-decreasing  $f : \mathbb{R}_+ \to [1, 2]$  such that f(0) = 1 and f(r) = 2for all r larger than some small threshold  $r_* > 0$ . Then, setting  $a(x) = f(\lambda_2(x) - \lambda_1(x))$  the drift  $H_1$  in the noisy ISD may be replaced by

$$\hat{H}_1(x) = -\left(I - a(x)v_1(x)v_1(x)^T - (2 - a(x))v_2(x)v_2(x)^T\right)\nabla U(x) + \frac{1}{2}\left(1 - \frac{1}{2}\right)\left(1 - \frac{1}{2$$

In other words, when  $\lambda_1(x)$  and  $\lambda_2(x)$  are clearly distinguished, then we recover the previous noisy ISD, however, on S, rather than undergoing an orthogonal reflection with respect to  $v_1(x)$ ,  $\nabla U(x)$  is orthogonally projected on the orthogonal of the space spanned by  $v_1(x), v_2(x)$ , which means that, in these two directions, the process behave like a Brownian motion. Natural choices for f would be piecewise constant (with f(r) = 1 for  $r < r_*$ ) or piecewise linear (with  $f(r) = 1 + r/r_*$  for  $r < r_*$ ). In the second case, the drift  $\overline{H}_1$  is continuous.

An analogous modification can be applied to the drift  $\overline{H}_1$  of the noisy GAD, in which case the process is then  $(X, V_1, V_2)$  where  $V_1, V_2$  are orthogonal vectors of  $\mathbb{S}^{d-1}$  whose evolution is given, in the case  $\varepsilon' = 0$ , by

$$dV_{1,t} = -\frac{1}{\eta} \left( I - V_{1,t} V_{1,t}^T \right) \nabla^2 U(X_t) V_{1,t} dt$$
  
$$dV_{2,t} = -\frac{1}{\eta} \left( I - V_{2,t} V_{2,t}^T - 2V_{1,t} V_{1,t}^T \right) \nabla^2 U(X_t) V_{2,t} dt.$$

(we refer to [110, Section 3.2.3] to derive the case  $\varepsilon' > 0$ , where the constraints are now that  $|v_1|^2 = |v_2|^2 = 1$  and  $v_1^T v_2 = 0$ ).

This variation can be generalised to any degree of degeneracy of the smallest eigenvalue, i.e. for  $k \ge 2$  modifying the drift  $H_1$  in the vicinity of the set where  $\lambda_1(x) = \cdots = \lambda_k(x)$  (generically these sets are empty for k > 2, but in many practical cases they are not, due to some symmetries in the problem).

#### 5.2 Theoretical analysis

Before turning to numerical experiment, we are interested in the existence of the different processes, and some of their properties, mainly in the long time limit. For the well-known Langevin process, we refer to [110, 112].

A valuable tool to study long time behavior of a process is the Doeblin condition. A Markov kernel  $\mathcal{P}: \mathbb{T}^d \to \mathcal{M}^1(\mathbb{T}^d)$  is said to satisfy a Doeblin condition if there exist c > 0 and a probability measure  $\mu$  such that for all  $x \in \mathbb{T}^d$ ,  $\mathcal{P}(x) \ge c\mu$ . This implies that  $(\mathcal{P}^n)_n$  converges in total variation to the unique probability measure  $\mu_\infty$  such that  $\mu_\infty \mathcal{P} = \mu_\infty$ , see for example [79]. In the continuous-time setting, this can be written as follow: there exists  $t_0, c > 0$  and a probability measure  $\mu$ , such that  $\delta_x P_{t_0} \ge c\mu$  for all  $x \in \mathbb{T}^d$ , with  $P_t$  the semi-group of the process, and  $\delta_x P_{t_0}$ is hence the law of the process starting from x a time  $t_0$ .

#### 5.2.1 Noisy ISD and Noisy GAD

Since we work on the compact torus and the drift of the noisy ISD (5.2) is measurable and bounded, the well-posedness and long-time behavior follows from classical results on diffusion processes. We summarize these results on the next proposition. Apart from the continuous-time equation (5.2), we are also interested in the corresponding Euler-Maruyama scheme. The noisy ISD is covered by the following:

**Proposition 5.2.1.** Consider  $\sigma > 0$ ,  $b : \mathbb{T}^d \to \mathbb{R}^d$  measurable and bounded, and X solution to

$$dX_t = b(X_t)dt + \sigma dB_t. \tag{5.6}$$

Strong existence: for any initial condition  $x_0 \in \mathbb{T}^d$ , and Brownian motion B, there exists a unique Markov process  $X^{x_0}$  solution to equation (5.6), with Brownian motion B and initial condition  $x_0$ .

Moreover, this process admits a unique invariant probability measure which admits a density with respect to the Lebesgue measure, and the law of the process converges exponentially fast towards this stationary measure in the total variation distance.

The Euler-Maruyama scheme defined for  $\delta > 0$  by

$$X_{n+1} = X_n + \delta b(X_n) + \sqrt{\delta G_n}$$

with  $(G_n)_n$  a sequence of standard Gaussian variables, admits as well a unique stationary measure  $\mu^{\delta}$ , and there exists  $c, C, \delta_0 > 0$  independent from  $\delta$  such that for all  $0 < \delta < \delta_0$ ,  $x_0 \in \mathbb{T}^d$ :

$$\|\mathcal{L}aw(X_n) - \mu^{\delta}\|_{TV} \leqslant Ce^{-c\delta n}.$$

*Proof.* The existence of a unique solution  $X^{x_0}$  to Equation (5.6) with initial condition  $x_0$  comes from [154, Corrolary 7.1.7 and 8.1.7]. By Itô's formula, the law  $M^{x_0}$  of  $(X_t^{x_0})_{t \in [0,T]}$  (a probability

measure on  $\mathcal{C}([0,T],\mathbb{T}^d))$  is a measure solution to the Kolmogorov equation

$$\partial_t M^{x_0} = L^* M^{x_0}$$

where  $L^*$  is the dual in  $L^2(\mathbb{R}^d)$  of the generator

$$L\varphi = b(x) \cdot \nabla \varphi + \frac{\sigma^2}{2} \Delta \varphi,$$

in the sense that for all function  $\varphi \in \mathcal{C}^2([0,T] \times \mathbb{T}^d, \mathbb{R})$ 

$$\int_{[0,T]\times\mathbb{T}^d} (\partial_t - L)\varphi(t,x) M^{x_0}(\mathrm{d}t,\mathrm{d}x) = 0.$$

Thus, according to [18, Proposition 6.5.1],  $\mathcal{L}aw(X_t^{x_0})$  (the time-marginal of  $M^{x_0}$ ) admits a density  $h^{x_0}(t, \cdot)$  with respect to the Lebesgue measure, for all t > 0, and for any compact interval  $J \subset \mathbb{R}_+$ 

$$h^{x_0} \in \mathbb{H}^{p,1}(J) := \left\{ u : \mathbb{R}_+ \times \mathbb{T}^d \to \mathbb{R}, \int_J \|u(t, \cdot)\|_{H^{p,1}} \mathrm{d}t < \infty \right\},$$

where  $\|\cdot\|_{H^{p,1}}$  is the classical Sobolev norm, for some p > d+2. Since we are in a compact set,  $h^{x_0} \in \mathbb{H}^{2,1}$ , and [18, Proposition 6.2.7] yields that  $h^{x_0}$  satisfies an Harnack inequality for t great enough: there exists  $C, \tau > 0$  such that for all  $x_0 \in \mathbb{T}^d$  and t great enough:

$$\sup_{\mathbb{T}^d} h^{x_0}(t-\tau,\cdot) \leqslant C \inf_{\mathbb{T}^d} h^{x_0}(t,\cdot).$$

Since  $h^{x_0}(t-\tau, \cdot)$  is a probability density, we have  $\sup_{\mathbb{T}^d} h^{x_0}(t-\tau, \cdot) \ge \operatorname{Leb}(\mathbb{T}^d)$ , where Leb stands for the Lebesgue measure. In particular, there exists  $t_0 > 0$  such that:

$$\inf_{x_0 \in \mathbb{T}^d} \inf_{\mathbb{T}^d} h^{x_0}(t_0, \cdot) > 0.$$

This is a Doeblin condition for the law of the process, with reference measure the Lebesgue measure, and in particular it classically implies the existence of a unique stationary measure for the process (5.2) and the exponential convergence of the law at time t toward this equilibrium.

For the Euler-Maruyama scheme, fix  $n = \lfloor t/\delta \rfloor$ , and write:

$$X_n^{x_0} = x_0 + \delta \sum_{k=0}^{n-1} b(X_k^{x_0}) + \sqrt{\delta} \sum_{k=0}^{n-1} G_k,$$

and

$$G = n^{-1/2} \sum_{k=0}^{n-1} G_k.$$

We have  $\delta \sum_{k=0}^{n-1} b(X_k^{x_0}) \leq t \|b\|_{\infty}$ . If  $A \subset \mathbb{T}^d$  is a measurable set, we are looking for a lower bound on  $\mathbb{P}(X_n^{x_0} \in A)$ . Let  $\overline{X}_n^{x_0}$  be the Euler scheme seen on  $\mathbb{R}^d$  rather than  $\mathbb{T}^d$  (considering *b* as a periodic function), with initial condition given as the representant of  $x_0$  in  $[0,1)^d$  and let  $\overline{A}$ be the intersection of  $[0,1)^d$  with the pre-image of A by the periodic quotient  $\mathbb{R}^d \to \mathbb{T}^d$ . Then,

 $\mathbb{P}(X_n^{x_0} \in A) \ge \mathbb{P}(\bar{X}_n^{x_0} \in \bar{A}), \text{ and }$ 

$$\mathbb{P}(\bar{X}_n^{x_0} \in \bar{A}) = \mathbb{P}\left(\sqrt{n\delta}G \in \bar{A} - x_0 - \delta\sum_{k=0}^{n-1} b(X_k)\right)$$
$$\geq \inf_{\substack{|a| \leq 2\pi d + t \|b\|_{\infty}}} \int_{\bar{A} - a} e^{-|y|^2/(t-\delta)} \mathrm{d}y$$
$$\geq e^{-2(4\pi d + t \|b\|_{\infty})/t} \mathrm{Leb}(A).$$

In other words, the *n*-step transition kernel of the Euler scheme satisfies a Doeblin condition with constant independent from the step size  $\delta$ . This implies the existence of a unique stationary measure  $\mu^{\delta}$  such that for all  $k \in \mathbb{N}$ ,

$$\|\mathcal{L}aw(X_{k|t/\delta|}) - \mu^{\delta}\|_{TV} \leqslant \alpha^{k} \|\mathcal{L}aw(X_{0}) - \mu^{\delta}\|_{TV}.$$

This gives for all  $n \in \mathbb{N}$ 

$$\begin{aligned} \|\mathcal{L}aw(X_n) - \mu^{\delta}\|_{TV} &\leq \|\mathcal{L}aw(X_{\lfloor t/\delta \rfloor \lfloor n/\lfloor t/\delta \rfloor \rfloor}) - \mu^{\delta}\|_{TV} \\ &\leq \alpha^{\lfloor n/\lfloor t/\delta \rfloor \rfloor} \|\mathcal{L}aw(X_0) - \mu^{\delta}\|_{TV} \leq Ce^{-c\delta n}, \end{aligned}$$

which concludes the proof.

We now turn to the analysis of the noisy GAD (5.3). When  $\varepsilon' > 0$ , we get an elliptic diffusion with bounded drift on a smooth compact manifold with no boundary, so essentially the adaptation of Proposition 5.2.1 in this more general context apply. In the following we focus on the case  $\varepsilon' = 0$ . Indeed, one may think that it is only necessary to have noise on the position in order to visit the whole space, and then the auxiliary vector V will go to the associated eigenvectors. The question is whether the process is hypoelliptic and controllable. However, it is clear that it cannot be the case in general, simply by considering counter-examples of the form

$$U(x) = \sum_{i=1}^{d} U_i(x_i)$$

Indeed, in these cases, if e.g.  $v_0 = (1, 0, ..., 0)$ , then  $v_t = v_0$  for all  $t \ge 0$ , hence there are no hope of having a unique stationary measure, and convergences towards it for all initial conditions. Let us discuss some some additional conditions that ensure this result. First, notice that if  $(X_t, V_t)_{t\ge 0}$ is a noisy GAD then so is  $(X_t, -V_t)_{t\ge 0}$ . In fact the relevant variable to define the process is not  $V_t$  but the class  $\{V_t, -V_t\}$  on the projective sphere  $\mathbb{P}^d = \mathbb{S}^{d-1}/\mathcal{R}$  with the equivalence relation given by  $v\mathcal{R}u$  iff u = -v or u = v, and we identify the process with its projection on  $\mathbb{T}^d \times \mathbb{P}^d$ (which is still a Markov process, solution of the same SDE (5.3) but where the drift of  $V_t$  is seen as a vector field on  $\mathbb{P}^d$ ).

**Proposition 5.2.2.** If U is  $C^3$ , then the noisy GAD is well-defined in the sense that there exists a unique strong solution to Equation (5.3). If in addition, we suppose that the potential U satisfies the following:

• There exists  $(\tilde{x}, \tilde{v}) \in \mathbb{T}^d \times \mathbb{P}^d$  such that:

$$\left(\partial_{x_i} \nabla^2 U(\tilde{x}) \tilde{v}\right)_{1 \leqslant i \leqslant d}$$

spans  $\tilde{v}^{\perp}$ .

• For all  $v \in \mathbb{P}^d$ , there exists H in the convex hull of the set

$$\left\{\nabla^2 U(x), x \in \mathbb{T}^d\right\}$$

such that, denoting by  $(\tilde{\lambda}_i)$  the ordered eigenvalues of H and  $(\tilde{v}_i)$  the associated eigenvectors,

$$\left\{ \begin{array}{rrr} v & = & \tilde{v}_1, \\ \tilde{\lambda}_1 & < & \tilde{\lambda}_2. \end{array} \right.$$

Then the noisy GAD admits a unique stationary measure  $\mu_{GAD}$  on  $\mathbb{T}^d \times \mathbb{P}^d$  with a positive density with respect to the Lebesgue measure. Moreover, for any initial condition, the law at time t of the process converges exponentially fast to  $\mu_{GAD}$  in total variation.

**Remark 5.2.1.** The second condition is implied by: For all  $v \in \mathbb{P}^d$ , there exists  $x \in \mathbb{T}^d$  such that

$$\begin{cases} v = v_1(x), \\ \lambda_1(x) < \lambda_2(x). \end{cases}$$

*Proof.* If U is  $C^3$ , then the noisy GAD is an SDE with Lipschitz coefficient, hence its well-posedness. The first additional assumption ensures that the generator of the noisy GAD

$$\mathcal{L} = -(I - 2vv^t)\nabla U(x) \cdot \nabla_x + \varepsilon \Delta_x - \frac{1}{\eta}(I - vv^t)\nabla^2 U(x)v \cdot \nabla_v$$

satisfies a weak Hörmander condition at  $(\tilde{x}, \tilde{v})$ , see [152].

Next, we want to establish the controllability of the process, in the following sense: fix  $(x_{ini}, v_{ini}), (x_f, v_f) \in \mathbb{T}^d \times \mathbb{P}^d, \delta > 0$ . We want to show that there exists T > 0 and a control  $u \in \mathcal{C}^0([0, T], \mathbb{R}^d)$  such that the solution of

$$\begin{cases} x'(t) = -(I - 2v(t)v(t)^T) \nabla U(x(t)) + u(t), \\ v'(t) = -\frac{1}{\eta} (I - v(t)v(t)^T) \nabla^2 U(x(t))v(t), \end{cases}$$
(5.7)

with  $(x(0), v(0)) = (x_{ini}, v_{ini})$ , satisfies  $||(x(1), v(1)) - (x_f, v_f)|| \leq \delta$ .

We first suppose that  $v_{ini}$  and  $v_f$  are not orthogonal. Fix H such that the second condition of Proposition 5.2.2 is satisfied with  $v = v_f$ , and  $x_1, \ldots, x_p$ , and  $\alpha_1, \ldots, \alpha_p$  such that

$$H = \sum_{i=1}^{p} \alpha_i \nabla^2 U(x_i), \quad \sum_{i=1}^{p} \alpha_i = 1.$$

Fix T > 0,  $n \in \mathbb{N}$ , and  $\gamma > 0$  such that  $T = n(\gamma + \gamma^2 + 2\gamma^4) + \gamma^2 + \gamma^4$ . Write  $\beta_i = \alpha_1 + \ldots + \alpha_i$ , and for  $k \in [0, n]$ :

$$u(t) = \begin{cases} (-x_{ini} + x_1)/(\alpha_1 \gamma^2) & \text{if} \quad 0 < t < \alpha_1 \gamma^2, \\ (I - 2v(t)v(t)^t) \nabla U(x_1) & \text{if} \quad \alpha_1(\gamma^2 + \gamma^4) < t < \alpha_1(\gamma + \gamma^2 + \gamma^4), \\ (x_{i+1} - x_i)/(\alpha_{i+1}\gamma^2) & \text{if} \quad (k + \beta_i)(\gamma + \gamma^2 + 2\gamma^4) \\ & < t < (k + \beta_i)(\gamma + \gamma^2 + 2\gamma^4) + \alpha_{i+1}\gamma^2, \\ (I - 2v(t)v(t)^t) \nabla U(x_i) & \text{if} \quad (k + \beta_i)(\gamma + \gamma^2 + 2\gamma^4) + \alpha_{i+1}(\gamma^2 + \gamma^4) \\ & < t < (k + \beta_{i+1})(\gamma + \gamma^2 + 2\gamma^4) - \alpha_{i+1}\gamma^4, \\ (x_f - x_p)/\gamma^2 & \text{if} \quad T - \gamma^2 < t < T, \end{cases}$$

and u is linear on the remaining intervals of size  $\gamma^4$ . Write as well

$$\bar{x}(t) = x_i$$
 for  $(k + \beta_{i-1})\gamma < t < (k + \beta_i)\gamma, \ k \in \mathbb{N}$ 

As  $\gamma$  goes to 0, v converges towards the solution to

$$\bar{v}'(t) = -\frac{1}{\eta} (I - \bar{v}(t)\bar{v}(t)^T) \nabla^2 U(\bar{x}(t))\bar{v}(t)$$

which is itself close to

$$w'(t) = -\frac{1}{\eta}(I - w(t)w(t)^T)Hw(t).$$

For clarity, we only show the second point which is the more complex one. We may write for all  $0 \leq u \leq T$ :

$$\begin{split} \left\| \int_{0}^{u} (I - \bar{v}_{s} \bar{v}_{s}^{t}) \nabla^{2} U(\bar{x}) \bar{v}_{s} \mathrm{d}s - \int_{0}^{u} (I - \bar{v}_{s} \bar{v}_{s}^{t}) H \bar{v}_{s} \mathrm{d}s \right\| \\ & \leq \left\| \int_{0}^{u} (I - \bar{v}_{s} \bar{v}_{s}^{t}) \nabla^{2} U(\bar{x}) \bar{v}_{s} \mathrm{d}s - \sum_{k=0}^{m-1} \sum_{i=1}^{p} (I - \bar{v}_{k\gamma} \bar{v}_{k\gamma}^{t}) \nabla^{2} U(x_{i}) \bar{v}_{k\gamma} \alpha_{i} \gamma \right\| \\ & + \left\| \sum_{k=0}^{m-1} \sum_{i=1}^{p} (I - \bar{v}_{k\gamma} \bar{v}_{k\gamma}^{t}) \nabla^{2} U(x_{i}) \bar{v}_{k\gamma} \alpha_{i} \gamma - \sum_{k=0}^{m-1} \sum_{i=1}^{p} (I - \bar{v}_{k\gamma} \bar{v}_{k\gamma}^{t}) H \bar{v}_{k\gamma} \gamma \right\| \\ & + \left\| \sum_{k=0}^{m-1} \sum_{i=1}^{p} (I - \bar{v}_{k\gamma} \bar{v}_{k\gamma}^{t}) H \bar{v}_{k\gamma} \gamma - \int_{0}^{u} (I - \bar{v}_{s} \bar{v}_{s}^{t}) H \bar{v}_{s} \mathrm{d}s \right\|. \end{split}$$

where  $m = \lfloor u/\gamma \rfloor$ . By hypothesis on H, the second term is 0.  $\bar{v}$  depends on  $\gamma$  by definition, but it is  $\|\nabla U\|_{\infty}$ -Lipschitz for all  $\gamma > 0$ , hence the first and third term are going to 0 as  $\gamma \to 0$ , uniformly on  $0 \leq u \leq T$ . The map  $v \mapsto (I - vv^t)Hv$  is  $\mathcal{C}^1$  on a compact set, hence is Lipschitz. Thus we have:

$$\|\bar{v}(u) - w(u)\| \leq L \int_0^u \sup_{r < s} \|\bar{v}(r) - w(r)\| \mathrm{d}s + \left\| \int_0^u (I - \bar{v}_s \bar{v}_s^t) \nabla^2 U(\bar{x}) \bar{v}_s \mathrm{d}s - \int_0^u (I - \bar{v}_s \bar{v}_s^t) H \bar{v}_s \mathrm{d}s \right\|,$$

and

$$\sup_{0\leqslant u\leqslant T} \|\bar{v}(u) - w(u)\| \leqslant e^{LT} \sup_{0\leqslant u\leqslant T} \left\| \int_0^u (I - \bar{v}_s \bar{v}_s^t) \nabla^2 U(\bar{x}) \bar{v}_s \mathrm{d}s - \int_0^u (I - \bar{v}_s \bar{v}_s^t) H \bar{v}_s \mathrm{d}s \right\| \underset{\gamma \to 0}{\to} 0.$$

Now, all it remains to do is to fix T > 0 great enough so that w(T) is in the  $\delta$ -neighborhood of  $v_f$ , then  $\gamma$  small enough so that v(T) is as well, and n accordingly.

If  $v_{ini}$  and  $v_f$  are orthogonal, writing  $\pi$  for the canonical projection from  $\mathbb{S}^{d-1}$  to  $\mathbb{P}^d$ , there exists an intermediate  $v_{inter}$  which is not orthogonal to neither  $v_{ini}$  or  $v_f$ , and the same control allows to go from  $(x_{ini}, v_{ini})$  to  $(x_{ini}, v_{inter})$  in a time T, and from  $(x_{ini}, v_{inter})$  to  $(x_f, v_f)$  in another time T.

We conclude with [152, Theorem 2.1] for the existence of the stationary measure and the convergence, and with [120, Theorem 5.2] for the positive density.  $\Box$ 

#### 5.2.2 Switched processes

When the mode  $(I_t)_{t\geq 0}$  is an autonomous Markov chain on a finite set [0, m-1],  $m \in \mathbb{N}$  (i.e., as it is the case in the processes we have considered in Section 5.1.3, if its jump rate does not depend on X) then the well-posedness of switched processes of the form

$$dX_t = H_{I_t}(X_t)dt + \sigma_{I_t}(X_t)dB_t,$$
(5.8)

with some drifts  $H_0, \ldots, H_{m-1}$  immediately follows from the well-posedness of the corresponding diffusion processes for each  $i \in [0, m-1]$ , since the process is then simply defined by induction along the jump times of I. In particular, in view of the previous section, the ISP (5.4) is well defined.

Moreover, in general, it is sufficient that one of the diffusion processes satisfies a Doeblin condition to imply the same for the switched process:

**Proposition 5.2.3.** If I is irreducible and there exists  $i_0$  such that the diffusion (5.8) with  $I_t = i_0$  satisfies a Doeblin condition (see the introduction of Section 5.2), then the switched process admits a unique invariant probability measure, and the law of the switched process converges exponentially fast towards this stationary measure.

*Proof.* The existence results from the previous construction. Fix  $t_0 > 0$  such that for all  $t > t_0$ ,  $x \in \mathbb{R}^d$ ,  $A \subset \mathbb{T}^d$  measurable,

$$\mathbb{P}_x(\bar{X}_t \in A) \ge cl(A)$$

where c > 0 and X is a solution to equation (5.8) with  $I_0 = i_0$ . Since I is an irreducible Markov process, we have for t > 0

$$\mathbb{P}(I_s = i_0, \ \forall s \in [t, 2t]) > 0.$$

Hence we have:

$$\mathbb{P}(X_{2t} \in A, I_{2t} = i_0) \geq \mathbb{P}(X_{2t} \in A, I_s = i_0, \forall s \in [t, 2t])$$
$$= \mathbb{E}\left(\mathbb{P}_{X_t}\left(\bar{X}_t \in A\right) \mathbb{1}_{I_s = i_0, \forall s \in [t, 2t]}\right)$$
$$\geq cl(A)\mathbb{P}(I_s = i_0, \forall s \in [t, 2t])$$
$$= \tilde{c}l(A),$$

which is a Doeblin condition with reference measure  $l \otimes \delta_{i_0}$ , and this concludes the proof.  $\Box$ 

**Corollary 5.2.1.** The ISP is well-defined, admit stationary measure, and its laws converge as t goes to infinity exponentially fast towards its unique stationary measure in the total variation distance.

#### 5.3 Numerical experiments

The main goal of the present work is to gain some empirical insights on the qualitative behavior of the processes introduced in Section 5.1.3 on simple models. Some questions that we have in particular are the following: what happens if the direction of the gentlest ascend starting from a local minimum does not correspond to the direction the process has to take eventually to find a saddle point? How does the process behave in front of attractive singularities of the ISD flow? How does the efficiency of the process (in terms of exploration) depend on the switching rate? How does it compare to a basic overdamped Langevin process? We will consider three models. The first one is a mixture of Gaussian, for which we can easily chose where are the minima and the saddle point and what are the gentlest ascend directions at the minima. The second one is the function studied in [115], for which the unique saddle point is separated from the two minima by singular lines. Finally, the third example, in higher dimension and closer to a genuine application, is the Lennard-Johns cluster model with 7 particles.

Contrary to the theoretical analysis, most processes here live on  $\mathbb{R}^d$ , which may raise a stability issue. This can be seen in dimension 1, where the process switch between a gradient descent and a gradient ascent, and thus can be non recurrent, or even explosive e.g. for potential of the form  $|x|^4$  at infinity. This can be solved by a suitable modification of the dynamics outside some compact set, for instance replacing  $H_1$  in (5.4) by  $\tilde{H}_1(x) = H_1(x) \mathbb{1}_{|x| \leq R} + H_0(x) \mathbb{1}_{|x| > R}$  or a smooth interpolation, or adding a deterministic jump from  $I_t = 1$  to  $I_t = 0$  when  $|X_t| \geq R$ . Then, classical Lyapunov conditions on U implies that the process remains stable. However, this will not be necessary in our simple numerical experiments.

#### 5.3.1 First 2D model: a mixture of Gaussians

We consider here a mixture of Gaussian in dimension 2. The potential is of the form:

$$U_1(x,y) = -\ln\left(\frac{1}{2}e^{-(x^2+y^2)} + \frac{1}{2}e^{-s_x(x-m_x)^2 - s_y(y-m_y)^2}\right),$$

with some parameters  $m_x, m_y \in \mathbb{R}$ ,  $s_x, s_y > 0$ . This is a classical model in statistics for multimodal problems. Here we take  $(m_x, m_y) = (4, 0)$  and consider two cases, either  $(s_x, s_y) = (3, 1)$ or  $(s_x, s_y) = (1, 3)$ , see Figure 5.1 and 5.2. In those figures, we represent a typical trajectory of the noisy ISD and of the ISP, with  $\varepsilon = 0.03$ , the final time T = 10, and  $\nu = 0.2$  in the case of the ISP, as well as the level lines of the potential. In these two cases, there are two local minimizers  $(\tilde{x}_1, 0)$  and  $(\tilde{x}_2, 0)$  separated by a unique saddle point  $(\tilde{z}, 0)$ , where  $\tilde{x}_1 < \tilde{z} < \tilde{x}_2$ , but they differ by the direction taken by the saddle search in the well from the left. The yellow part represent the trajectories of the noisy ISD, or the part of the ISP where  $I_t = 1$ , and in blue the part of the ISP where  $I_t = 0$  (corresponding to a gradient descent).

According to the Large Deviation Principle, for a small  $\varepsilon$ , if the initial condition corresponds to the right local minima ( $\tilde{x}_2, 0$ ), the overdamped Langevin process (5.1) will typically stays in the vicinity of this minimizer for a time exponentially long with  $\varepsilon$ , and will likely go from there to the other minimizer by following in reverse the trajectory of a gradient descent starting from a point arbitrarily close to the saddle point (which in the present case is a straight line from the minimum to the saddle point), see [65, Chapter 4, Theorem 2.1] as well as their computation of the quasi-potential.

In the case  $(s_x, s_y) = (3, 1)$ , with some initial condition  $(x_0, 0)$ ,  $\tilde{z} < x_0 < \tilde{x}_2$ , the noisy (and deterministic) ISD will follow the same reactive trajectory as the overdamped Langevin process, the gentlest way to go to the saddle point being here along the gradient, but it happens much earlier (since this is the behaviour of the deterministic system at  $\varepsilon = 0$  and not a large deviation from it). However, in the case,  $(s_x, s_y) = (1, 3)$ , the gentlest way to leave the right minima is to start along the directions (0, 1) and (0, -1), see figure 5.2. Nevertheless, independently from this difference in the beginning of the trajectory, in both cases the saddle point is found by the deterministic dynamics for initial condition between  $\tilde{x}_1$  and  $\tilde{x}_2$ .

For initial condition  $(x_0, y_0)$  with  $x_0 > \tilde{x}_2$ , the ISD goes to infinity. Hence, the dynamics has to be modified outside some compact set, as explained in the introduction of this section, so that the process does not escape to infinity. In that case, one only need to wait for the process to enter the domain of attraction of the saddle point for the deterministic ISD.





(d) Trajectory of the switched process.



Figure 5.2: Case  $(s_x, s_y) = (1, 3)$ .



Figure 5.3: Periodized potential.



Figure 5.4: Invariant measure of the ISP for  $\nu = 0.1$ ,  $\varepsilon = 0.05$ .

However, as an alternative way to enforce stability in order to get quantitative results, in the present case with a simple illustrative purpose, we simply use a periodized version of the potential. Denote

$$\tilde{U}_1(x,y) = -\ln\left(\frac{1}{2}e^{-L^2(\sin^2(x/L) + \sin^2(y/L))} + \frac{1}{2}e^{-L^2(s_x\sin^2((x-m_x)/L) + s_y\sin^2((y-m_y)/L))}\right)$$

where L > 0 is a parameter, see Figure 5.3. This periodized potential has two supplementary saddle point, but for all  $(x, y) \in \mathbb{R}^2$ ,  $\lim_{L\to\infty} \tilde{U}_1(x, y) = U_1(x, y)$ , and we may defined the ISP on  $(\pi L\mathbb{T})^2$ .

Using a 2-D histogram,  $\nu = 0.1$ ,  $\varepsilon = 0.05$ , we get a representation of the invariant measure in Figure 5.4. We see that the invariant measure charges both minima, as well as the saddle point between the two.

Now, if the goal is to find both minima, then one can find one of them using a gradient descent, and the second one using the ISP. Since the ISD allows to find the saddle point, then



Figure 5.5: Time to reach a minima from the other one on the periodized potential.

if the switching rate is low enough, the switched process will do as well, and thanks to the Brownian noise, after a first switch, will have a positive probability to be in the domain of attraction for the gradient descent of the second minima, and thus will find it. If it was not in the right domain of attraction, then the process will go back to the first well, and repeat the same kind of trajectories. The success of the algorithm is then the same as the one of a rigged coin flip. For a small  $\varepsilon$ , as long as the transitions from one well to another are driven by this switching behaviour, the transition time behaves much better than the exponentially long time of the overdamped Langevin dynamics (with Brownian-driven rare transitions). At a fixed  $\varepsilon$ , when the switching rate becomes too small, the process has to wait for switching events to cross the saddle (and eventually when it becomes very small the process behaves like the overdamped Langevin dynamics and the transitions are driven by the small Brownian noise); on the other hand, a large switching rate induces an averaging phenomenon (the drift along the gentlest ascend direction vanishes), which impairs the efficiency. Thus, there should be an optimal switching rate in terms of mean transition time. This is indeed what we observed, see Figure 5.5. The numerical experiment were conducted with  $\varepsilon = 0.03$ ,  $(x_0, y_0) = (4, 0)$ ,  $I_0 = 0$ , and the time was estimated by Monte-Carlo with n = 15 repetitions.

#### 5.3.2 A 2D model with singular lines

Now we are interested in the following potential:

$$U_2(x,y) = (1-x^2)^2 + 2y^2,$$

for  $(x, y) \in \mathbb{R}^2$ , see Figure 5.6.

The deterministic ISD and GAD have been studied for this potential in [115]. This potential has two minima, (-1,0) and (1,0), and one saddle point (0,0). However, there are two lines,  $\left\{x = \pm \sqrt{4/6}\right\}$ , for which the Hessian of U has two equal eigenvalues. The deterministic GAD and ISD cannot cross those lines, and go to infinity, see Figure 5.7. In the noisy case, the Brownian motion may allow the process to cross the line, see Figure 5.8.

As seen in figure 5.8, the Brownian motion allow to cross the set of singularities, but for small  $\varepsilon$ , this crossing may happen far from the set of minima and saddle point, because the deterministic process satisfies  $\lim_{t\to\infty} y_t = +\infty$  for  $x_0 > \sqrt{4/6}$  and  $y_0 > 0$ , where  $(x_t, y_t)$  is the



Figure 5.6: Level line of the potential  $U_2$ .







Figure 5.8:  $\varepsilon = 0.05$ .

ISD. The singular lines are repulsive, and for  $x > \sqrt{4/6}$ , the process is attracted to the line  $\{x = 1\}$ . Concretely, for the process to cross the singular line, the noise must be great enough so that this occurs before the simulation stops due to numerical limits.

Since the existence of this singularity results from a crossing between the eigenvalues of the Hessian of the potential  $U_2$ , we study the solution to solve the issue of singularities proposed in Section 5.1.3, to see in particular if it allows a faster convergences towards the saddle point in our example here. Recall the set of singularities:

$$\mathcal{S} = \left\{ (x, y) \in \mathbb{R}^2, \lambda_1(x, y) = \lambda_2(x, y) \right\},\$$

and the modified noisy ISD (writting z = (x, y) and  $Z_t$  the process):

$$Z_t = -\left(I - 2g(Z_t)v_1(Z_t)^t v_1(Z_t) - (2 - g(Z_t)v_2(Z_t)^t v_2(Z_t))\right) \nabla U_2(Z_t) dt + \sqrt{2\varepsilon} dB_t,$$

with  $g(z) = f(\lambda_2(z) - \lambda_1(z))$  for some function  $f : \mathbb{R}_+ \to \mathbb{R}_+$  such that f(0) = 1, and f(r) = 2 for  $r \ge r_*$ , and some small  $r_* > 0$ . As explained, the idea is to replace the reflection with respect to  $v_1(x)^{\perp}$  by a projection on  $vect(v_1(x), v_2(x))^{\perp}$ .

In dimension 2, the process simply becomes a Brownian motion near  $\mathcal{S}$ . For numerical study we consider:

$$f_1(r) = 1 + \mathbb{1}_{r > r_*},$$
$$f_2(r) = 1 + \frac{r}{2} \mathbb{1}_{r \le r_*} + \mathbb{1}_{r > r_*}$$

and

$$J_{2}(r) = r_{*} = r_{*} + r_{*},$$

where  $r_* > 0$  is a fixed parameter. An example of trajectory with  $f_1$ , is given in Figure 5.9.

We compare the speed at which the process reaches the saddle point with the different cut function  $f_0 = 2$ ,  $f_1$ , and  $f_2$ . We chose as parameter  $r_* = 2$ ,  $x_0 = (0.9, 0)$ , and n = 500 trials. The average of

$$\tau = \inf \{t \ge 0, |X_t| < 0.1\},\$$

over the *n* trials is displayed in Figure 5.10 for  $\varepsilon \in \{0.03, 0.04, 0.05, 0.06\}$ .

We can see that the regularised dynamics reach the saddle point faster than the initial dynamic with  $f_0$ . Moreover, as explained before, for smaller values of  $\varepsilon$ , there is a positive probability that the process doesn't reach the saddle point before hitting numerical limits. For a given temperature  $\varepsilon$ , this probability is lower for the processes defined from  $f_1$  or  $f_2$ . This possible failure was not observed in the experiments displayed in Figure 5.10, but for  $\varepsilon = 0.02$ , we



Figure 5.9:  $f_1$ -modified ISD,  $\varepsilon = 0.05$ .



Figure 5.10: Time to reach the saddle point.

estimated the probability of failure with 500 trials for each process. We get 0.394 for  $f_0$ , and 0.046 for  $f_2$ . Even at  $\varepsilon = 0.02$ , no failure was observed for  $f_1$ .

Here we haven't used any of the modifications discussed at the beginning of Section 5.3 to enforce stability, in order to study the effects of the singular line on the time needed to find the saddle point.

#### 5.3.3 Lennard-Jones clusters

We now study numerically the process for a potential in higher dimension. Consider the Lennard-Jones potential for N = 7 particles in dimension 2, given by

$$U_3(x_1, \dots x_N) = \sum_{i < j} W(|x_i - x_j|),$$

where  $x_i \in \mathbb{R}^2$  for all  $1 \leq i \leq N$ , and for r > 0:

$$W(r) = 4\left(\frac{1}{r^{12}} - \frac{1}{r^6}\right).$$

The potential U is invariant by rotation and translation of the full system and by permutation of the particles. Once these symmetries are ruled out, it has three non-global local minima, and an additional global one, represented in Figure 5.11. Lennard-Jones clusters in dimension 2 or 3 are classical models in physics and have been extensively, see e.g. [141, 40] and references within.

We use as initial condition the local minimum such that  $U \approx -11, 47$ , and we compute the time necessary for the process to visit all minima for different values of  $\nu$  and  $\varepsilon$ , estimated over 20 experiments. The average time is given in Figure 5.12. These results confirm the interpretation of Figure 5.5 discussed in Section 5.3.1. We see that the sensibility of the exploration time to the switching rate increases as the temperature  $\varepsilon$  decreases, as should be expected: in the high frequency regime, the drift along the gentlest ascent direction averages to zero so that, in that direction, the process moves at the speed of a Brownian motion with variance  $\varepsilon$ ; in the low frequency regime, the transitions between different wells are driven by the Brownian noise and thus follow an Eyring-Kramers formula, so that the mean transition time is exponentially large with  $\varepsilon^{-1}$ .

Since  $U_3$  is invariant by global translations and rotations of the system, 0 is always an eigenvalue of its Hessian, associated to eigenvectors which are orthogonal to  $\nabla U_3$ . As a consequence, at points x where  $\nabla^2 U_3(x)$  has no negative eigenvalue, the ISD dynamics is the same as the gradient descent dynamics. For a real application, these known symmetries should be taken into account in order to avoid this. However, as we saw, even without addressing this issue, the ISP still finds all minima faster than the Langevin process: even though the gentle ascend behaviour plays no role in the vicinity of minimizers, it still makes saddle points locally attractive for the dynamics, which is a key point for the efficiency of the exploration.



Figure 5.11: List of minimizers of  ${\cal U}$ 



Figure 5.12: Time to visit all minima of the Lennard-Jones cluster.

154

## Part III

# Fleming-Viot processes

## Chapter 6

# Convergence of a particle approximation for the quasi-stationary distribution of a diffusion process: uniform estimates in a compact soft case.

#### Outline of the current chapter

6.1	Introduction	158
	6.1.1 The problem	158
	Notations and conventions	159
	6.1.2 The algorithm and main result $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	160
6.2	Proofs	164
	6.2.1 The basic coupling	165
	6.2.2 Long-time convergence	167
	6.2.3 Propagation of chaos	169
	6.2.4 Discrete to continuous time	170
	6.2.5 Conclusion	174

**Abstract**: We establish the convergences (with respect to the simulation time t; the number of particles N; the timestep  $\gamma$ ) of a Moran/Fleming-Viot type particle scheme toward the quasistationary distribution of a diffusion on the *d*-dimensional torus, killed at a smooth rate. In these conditions, quantitative bounds are obtained that, for each parameter  $(t \to \infty, N \to \infty \text{ or } \gamma \to 0)$  are independent from the two others.

#### 6.1 Introduction

#### 6.1.1 The problem

Start from the diffusion on the *d*-dimensional periodic flat torus  $\mathbb{T}^d$ 

$$\mathrm{d}Z_t = b(Z_t)\mathrm{d}t + \mathrm{d}B_t \tag{6.1}$$

with  $b \in \mathcal{C}^1(\mathbb{T}^d)$ , where  $(B_t)_{t \ge 0}$  is a *d*-dimensional Brownian motion. Add a killing rate  $\lambda \in \mathcal{C}(\mathbb{T}^d)$ and, given a standard exponential random variable *E* independent from  $(Z_t)_{t \ge 0}$ , define the death time

$$T = \inf \left\{ t \ge 0, \ E \le \int_0^t \lambda(Z_s) \mathrm{d}s \right\}.$$
(6.2)

Then a probability measure  $\nu$  on  $\mathbb{T}^d$  is said to be a quasi-stationary distribution (QSD) associated to the SDE (6.1) and the rate  $\lambda$  if

$$\mathcal{L}aw(Z_0) = \nu \qquad \Rightarrow \qquad \forall t \geqslant 0, \ \mathcal{L}aw(Z_t \mid T > t) = \nu$$

In our case, there exists a unique QSD  $\nu_*$  and, whatever the initial distribution  $\eta_0$  of  $Z_0$ ,

$$\mathcal{L}aw(Z_t \mid T > t) \xrightarrow[t \to \infty]{} \nu_*$$

see e.g. [32, Theorem 2.1]) or Corollary 6.2.8 below.

The present work is dedicated to the proof of convergence of an algorithm designed to approximate  $\nu_*$ . This is classically done through a system of N interacting particles whose empirical measure converges to  $\mathcal{L}aw(Z_t \mid T > t)$  as  $N \to \infty$ , where killed particles are resurrected in a suitable way in order to keep constant the size of the system (while a naive Monte Carlo simulation would see the sample shrink along time). This question has already been addressed by many authors in various contexts, see the discussion in Section 1.3.3. Before introducing the algorithm, stating our results and comparing them with previous works, for now, let us simply highlight the main specificities of the present work.

The first novelty is that we take into account the time-discretization of the continuous-time diffusion. That way, we establish error bounds between the theoretical target QSD and the empirical measure indeed obtained with an actual implementation of the algorithm. There are three sources of errors: first, the continuous-time SDE (6.1) has to be discretized with some time-step parameter  $\gamma > 0$ . Second, as will be detailed below, a non-linearity in the theoretical algorithm has to be approximated by a system of N particles. This leads to the definition of an ergodic Markov chain whose invariant measure is close, in some sense, for large N, to the QSD of the time-discretization of the diffusion. But then this Markov chain is only run for a finite simulation time  $t = m\gamma$ ,  $m \in \mathbb{N}$ . A third error term then comes from the fact that stationarity is not fully achieved. We will obtained quantitative error bounds in  $\gamma$ , N and t.

A second specifity is that the bound obtained for each parameter will be uniform in the other two. For instance, the only other work in which the long-time convergence of the chain is proven to be, under some (restrictive) conditions, uniform in N, is [38] in a finite state space. Besides, our work is quite close in spirit to this work of Cloez and Thai.

Finally, although it was not the primary motivation of the present work, it seems that the particular definition of the system of interacting particles considered here, in particular the rebirth mechanism, was not considered in previous works (where, basically, killed particles are resurrected at the position of one of the other particles). Our variant is initially motivated by the property stated in Proposition 6.1.1 below, which has been indicated to the second author by Bertrand Cloez. Yet, this variant has the unintended advantage to be both discrete in time and non-failable, in the sense that it is well-defined for all times, even though all particles die simultaneously from time to time (see also [142] on this question).

Note that we restrict the study to a compact state space. Moreover, we only consider soft killing at some continuous rate, and no hard killing which would correspond to the case where T is the escape time from some sub-domain (see e.g. [11, 48]). Finally, as will be seen below, as far as the long-time behaviour of the process is concerned we will work in a perturbative regime, namely we will assume that the variations of  $\lambda$  are small with respect to the mixing time of the diffusion (6.1) (while  $\|\lambda\|_{\infty}$  itself is not required to be small). These very restrictive conditions, which rule out many cases of practical interest, have to be considered in light of our very strong results (Theorems 6.1.1 and 6.1.2 below and all the corollaries of Section 6.2.5, gathered in Figure 6.1). In fact, although already interesting by itself, this restricted framework can be thought as a toy model motivated in particular by the case that arises in the parallel replica algorithm [104]. In that case, T is the escape time for (6.1) from a bounded metastable domain, so that the lifespan of the process is expected to be larger than its mixing time (and to depend little from the initial condition, given it is far enough from the boundary). Hence, the compact and perturbative assumptions are consistent with this objective. The restriction to smooth killing rate, however, is made to avoid additional difficulties in the hard case where, even in the metastable case, the probability to leave the domain is high (and exhibits high variations) when the process is close to its boundary. The initial motivation of the present study was to test the general strategy of the proof (via coupling aguments) in a first simple case, with the goal of extending it later on to the metastable hard case by combining it with some Lyapunov arguments to control the variations of the killing rate near the boundary. This is postponed for future work.

This chapter is organized as follows. The algorithm and main results are presented in Section 6.1.2. Section 6.2 contains the proofs, and more precisely: a general coupling argument, which is the central tool for all our results, is presented in Section 6.2.1; the basic bounds in terms of  $t \to +\infty$ ,  $N \to +\infty$  and  $\gamma \to 0$  are then stated and proven respectively in Sections 6.2.2, 6.2.3 and 6.2.4; finally, these basic results are combined in Section 6.2.5, concluding the proofs of the main theorems and inducing a number of corollaries.

#### Notations and conventions

We respectively denote  $\mathcal{P}(F)$  and  $\mathcal{B}(F)$  the set of probability measures and of Borel sets of a Polish space F. Functions on  $\mathbb{T}^d$  are sometimes identified to  $[0,1]^d$ -periodic functions, and similar non-ambiguous identifications are performed, for instance if  $x \in \mathbb{T}^d$  and G is a d-dimensional standard gaussian random variable, x + G has to be understood in  $\mathbb{T}^d$ , etc. A Markov kernel Q on F is indiscriminately understood as, first, a function from F to  $\mathcal{P}(F)$ , in which case we denote  $Q: x \mapsto Q(x, \cdot)$  (where  $Q(x, \cdot)$  denotes the probability  $A \in \mathcal{B}(F) \mapsto Q(x, A) \in [0, 1]$ ); second, a Markov operator on bounded measurable functions on F, in which case we denote  $Q: f \mapsto Qf$  (where  $Qf(x) = \int f(w)Q(x, dw)$ ); third, by duality, a function on  $\mathcal{P}(F)$ , in which case we denote  $Q: \mu \mapsto \mu Q$  (so that  $\mu(Qf) = (\mu Q)f$ ). In particular,  $Q(x, \cdot) = \delta_x Q$  for  $x \in F$ . If  $\mu \in \mathcal{P}(F)$  and  $k \in \mathbb{N}_*$ , we denote  $\mu^{\otimes k} \in \mathcal{P}(F^k)$  the law of a k-uplet of independent random variables with law  $\mu$ . Similarly, if Q is a Markov kernel on F, we denote  $Q^{\otimes k}$  the kernel on  $F^k$  such that  $Q^{\otimes k}(x, \cdot) = Q(x_1, \cdot) \otimes \cdots \otimes Q(x_k, \cdot)$  for all  $x = (x_1, \ldots, x_k) \in F^k$ . We denote  $\mathcal{E}(1)$  the exponential law with parameter 1,  $\mathcal{U}(I)$  the uniform law on a set I and  $\mathcal{N}(m, \Sigma)$  the Gaussian law with mean m and variance matrix  $\Sigma$ . We use bold letters for random variables in  $\mathbb{T}^{dN}$  and decompose them in d-dimensional coordinates, like  $\mathbf{X} = (X_1, \ldots, X_N)$  with  $X_i \in \mathbb{T}^d$ , or  $\mathbf{X}_1 = (X_{1,1}, \ldots, X_{N,1})$ .

#### 6.1.2 The algorithm and main result

Starting from the diffusion (6.1) killed at time T given by (6.2), we introduce two successive approximations. The first is time discretization. For a given time step  $\gamma > 0$  and a sequence  $(G_k)_{k \in \mathbb{N}}$  of independent random variables with law  $\mathcal{N}(0, I_d)$ , we consider the Markov chain on  $\mathbb{T}^d$  given by  $\tilde{Z}_0 = Z_0$  and

$$\forall k \in \mathbb{N}, \qquad \tilde{Z}_{k+1} = \tilde{Z}_k + \gamma b(\tilde{Z}_k) + \sqrt{\gamma} G_k \tag{6.3}$$

and, given  $E \sim \mathcal{E}(1)$  independent from  $(G_k)_{k \in \mathbb{N}}$  and  $Z_0$ ,

$$\tilde{T} = \inf \left\{ t = n\gamma, \ n \in \mathbb{N}_*, \ E \leqslant \gamma \sum_{k=1}^n \lambda(\tilde{Z}_k) \right\}.$$

From classical results for Euler schemes of diffusions (see e.g. [131]), it is quite clear that, for any  $A \in \mathcal{B}(\mathbb{T}^d)$  and all  $t \ge 0$ ,

$$\mathbb{P}\left(\tilde{Z}_{\lfloor t/\gamma \rfloor} \in A, \ \tilde{T} < t\right) \xrightarrow[\gamma \to 0]{} \mathbb{P}\left(Z_t \in A, \ T < t\right),$$

from which, for all  $t \ge 0$ ,

$$\mathcal{L}aw\left(\tilde{Z}_{\lfloor t/\gamma \rfloor} \mid \tilde{T} < t\right) \xrightarrow[\gamma \to 0]{} \mathcal{L}aw\left(Z_t \mid T < t\right)$$

(we will prove this, see Corollary 6.2.4 below). Note that, from the memoryless property of the exponential law, given a sequence  $(U_k)_{k\in\mathbb{N}}$  of independent variables uniformly distributed over [0,1] and independent from  $(G_k)_{k\in\mathbb{N}}$  and  $Z_0$ , then  $((\tilde{Z}_n)_{n\in\mathbb{N}}, \tilde{T})$  has the same joint distribution as  $((\tilde{Z}_n)_{n\in\mathbb{N}}, \tilde{T})$  with

$$\hat{T} = \inf \left\{ t = n\gamma, \ n \in \mathbb{N}_*, \ U_n \leqslant p(\tilde{Z}_n) \right\}$$

where  $p(z) = 1 - \exp(-\gamma \lambda(z))$  is the probability that, arriving at state z, the chain is killed.

A naive Monte Carlo sampler for the QSD would be to simulate N independent copies of the chain (6.3) killed with probability  $z \mapsto p(z)$  and to consider after a large number of iterations the distribution of the copies that have survived. However, after a long time, most copies (possibly all) would have died and the estimator would be very bad. To tackle this issue, we have to introduce a rebirth mechanism to reincorporate dead particles in the system.

Denote  $K : \mathbb{T}^d \to \mathcal{P}(\mathbb{T}^d)$  the Markov kernel associated with the transition (6.3), i.e.

$$Kf(x) = (2\pi)^{-d/2} \int_{\mathbb{R}^d} f(x + \gamma b(x) + \sqrt{\gamma}y) e^{-\frac{1}{2}|y|^2} dy$$

For  $\mu \in \mathcal{P}(\mathbb{T}^d)$ , let  $Q_{\mu}$  be the Markov kernel such that, for all  $x \in \mathbb{T}^d$ ,  $Q_{\mu}(x, \cdot)$  is the law of the random variable X defined as follows. Let  $(X_k, U_k)_{k \in \mathbb{N}}$  be a sequence of independent random variables such that, for all  $k \in \mathbb{N}$ ,  $X_k$  and  $U_k$  are independent,  $U_k \sim \mathcal{U}([0, 1])$  and  $X_0 \sim K(x, \cdot)$ while, for  $k \ge 1$ ,  $X_k \sim \mu K$ . Let  $H = \inf\{k \in \mathbb{N}, U_k \ge p(X_k)\}$ , and set  $X = X_H$ . Since  $\lambda$  is bounded, p is uniformly bounded away from 1 and thus H is almost surely finite, so that  $Q_{\mu}$  is well-defined.

In other words, a random variable  $X \sim Q_{\mu}(x, \cdot)$  may be constructed through the following algorithm (in which *new* means: independent from all the variables previously drawned).

- 1. Draw  $X_0 \sim \mathcal{N}(x + \gamma b(x), \gamma I_d)$  and a new  $U_0 \sim \mathcal{U}([0, 1])$ .
- 2. If  $U_0 \ge p(X_0)$ , set  $X = X_0$  in  $\mathbb{T}^d$  (in that case, we say the particle has moved from x to  $X_0$  without dying).
- 3. If  $U_0 < p(X_0)$  then set i = 1 and, while X is not defined, do:
  - (a) Draw a new  $X'_i$  distributed according to  $\mu$ , a new  $X_i \sim \mathcal{N}(X'_i + \gamma b(X'_i), \gamma I_d)$  and a new  $U_i \sim \mathcal{U}([0, 1])$ .
  - (b) If  $U_i \ge p(X_i)$ , set  $X = X_i$  in  $\mathbb{T}^d$  (in that case, we say the particle has died, resurrected at  $X'_i$ , moved to  $X_i$  and survived).
  - (c) If  $U_i < p(X_i)$ , set  $i \leftarrow i + 1$  (in that case, we say the particle has died, resurrected at  $X'_i$ , moved to  $X_i$  and died again) and go back to step (a).

From this, we define a chain  $(Y_k)_{k\in\mathbb{N}}$  as follows. Set  $Y_0 = Z_0$  and suppose that  $Y_k$  has been defined for some  $k \in \mathbb{N}$ . Let  $\eta_k = \mathcal{L}aw(Y_k)$ , and draw a new  $Y_{k+1} \sim Q_{\eta_k}(Y_k, \cdot)$ . This somewhat intricate definition is motivated by the following results (whose proof is postponed to Section 6.2):

**Proposition 6.1.1.** For all  $n \in \mathbb{N}$ 

$$\eta_n = \mathcal{L}aw\left( ilde{Z}_n \mid ilde{T} > n\gamma
ight).$$

In particular, as  $n \to \infty$ , the law  $\eta_n$  of  $Y_n$  converges toward the QSD of  $\hat{Z}$ . Unfortunately, it is impossible to sample  $(Y_k)_{k\in\mathbb{N}}$  in practice since this would require to sample according to  $\eta_k$  for any  $k \in \mathbb{N}$ . This is a classical case of a time-inhomogeneous Markov chain which is interacting with its own law or, similarly, of a measure-valued sequence  $(\eta_k)_{k\in\mathbb{N}}$  with a non-linear evolution. Such processes arise in many applications, see e.g. [43, 42] and references within. Motivated by the Law of Large Numbers, we are lead to a second approximation, which is to use mean-field interacting particles. For a fixed  $N \in \mathbb{N}_*$  and for  $x = (x_i)_{i \in [1,N]} \in \mathbb{T}^{dN}$ , we denote

$$\pi(x) := \frac{1}{N} \sum_{i=1}^{N} \delta_{x_i} \in \mathcal{P}(\mathbb{T}^d)$$

the associated empirical distribution. Then we define the Markov operator R on  $\mathbb{T}^{dN}$  as

$$R(x,\cdot) = Q_{\pi(x)}(x_1,\cdot) \otimes \cdots \otimes Q_{\pi(x)}(x_N,\cdot).$$

In other words, a random variable  $\mathbf{Y} \sim Q(x, \cdot)$  is such that the  $Y_i$ 's are independent with  $Y_i \sim Q_{\pi(x)}(x_i, \cdot)$ . In order to specify the parameters involved, we will sometimes write  $R_{N,\gamma}$  for R.

Let us informally describe the transitions of such a Markov chain  $(\mathbf{X}_k)_{k\in\mathbb{N}}$ : the  $i^{th}$  particle follows the transition given by (6.3) independently from the other particles until it dies. If it dies at a step  $k \in \mathbb{N}_*$ , then it is resurrected on another particle  $X_{J,k-1}$  with J uniformly distributed over  $\llbracket 1, N \rrbracket$  (in particular and contrary to most previous works on similar algorithms, J = i is not excluded, although it doesn't change much since its probability vanishes as  $N \to \infty$ ) and
immediatly performs a step of (6.3); if it dies again after this unique step, it is resurrected again and performs a new step, and so on until it is not killed after a resurrection and an Euler scheme step. Then this is the new value  $X_{i,k}$  from which the particle follows again the transitions (6.3) until its next death, etc.

Note that there is no problem of simultaneous death since at step k the particles are resurrected on positions at step k - 1, which are well-defined even if all particles die at once at step k.

It is easily seen that R admits a unique invariant measure toward which the law of the associated Markov chain converges exponentially fast (in the total variation sense for instance), but a naive argument yields a convergence rate that heavily depends on N (and possibly  $\gamma$ ). Similarly, classical studies can be conducted for the limits  $N \to \infty$  and  $\gamma \to 0$  but again with estimates that are typically exponentially bad with respect to the total simulation time. In the following we will focus on a somewhat perturbative regime under which we will establish estimates for each of these limits that are uniform with respect to the other parameters. Even for the continuous-time process (corresponding to  $\gamma = 0$ , see Section 6.2.4 for the definition), such uniform results are new (see Corollaries 6.2.3 and 6.2.5).

Recall that the  $\mathcal{W}_1$  Wasserstein distance between  $\mu, \nu \in \mathcal{P}(\mathbb{T}^d)$  is defined by

$$\mathcal{W}_1(\mu, \nu) = \inf \{ \mathbb{E}(|X - Y|) : X \sim \mu, Y \sim \nu \}.$$

More generally, for  $\rho$  a distance on some Polish space F, denote  $\mathcal{W}_{\rho}$  the corresponding Wasserstein distance on  $\mathcal{P}(F)$ , defined by

$$\mathcal{W}_{\rho}(\mu,\nu) = \inf \left\{ \mathbb{E}\left(\rho(X,Y)\right) : X \sim \mu, Y \sim \nu \right\}.$$
(6.4)

If  $X \sim \mu$  and  $Y \sim \nu$ , we call (X, Y) a coupling of  $\mu$  and  $\nu$ . If (X, Y) is a coupling for which the infimum in (6.4) is attained, we say that it is an optimal coupling. From [164, Corollary 5.22], such an optimal coupling always exists.

Our first main result is a long-time convergence rate uniform in N:

**Theorem 6.1.1.** There exist  $c_1, c_2, \gamma_0 > 0$  and a distance  $\rho$  on  $\mathbb{T}^d$  equivalent to the Euclidean distance, that depend only on the drift b and the dimension d, such that, if  $\lambda$  is Lipschitz with a constant  $L_{\lambda}$  and

$$\kappa := c_1 - c_2 L_\lambda e^{\gamma \|\lambda\|_\infty} , \qquad (6.5)$$

then the following holds: for all  $\gamma \in (0, \gamma_0]$ ,  $N \in \mathbb{N}$  and all  $\mu, \nu \in \mathcal{P}(\mathbb{T}^{dN})$ , considering the distance  $\rho_N(x, y) = \sum_{i=1}^N \rho(x_i, y_i)$  for  $x, y \in \mathbb{T}^{dN}$ ,

$$\mathcal{W}_{\rho_N}(\mu R_{N,\gamma}, \nu R_{N,\gamma}) \leq (1 - \gamma \kappa) \mathcal{W}_{\rho_N}(\mu, \nu).$$

As a consequence, there exists C > 0 that depends only on b and d such that for all  $\gamma \in (0, \gamma_0]$ ,  $m, N \in \mathbb{N}$  and all  $\mu, \nu \in \mathcal{P}(\mathbb{T}^{dN})$ ,

$$\mathcal{W}_1\left(\mu R_{N,\gamma}^m, \nu R_{N,\gamma}^m\right) \leqslant CN \left(1 - \gamma \kappa\right)^m$$
.

This means that, with respect to the metric  $\rho_N$ ,  $R_{N,\gamma}$  has a Wasserstein curvature of  $\gamma \kappa$  in the sense of [93].

Theorem 6.1.1 is proven in Section 6.2.2. From this first result, similar bounds can be obtained for large N and small  $\gamma$  (see Sections 6.2.3 and 6.2.4). Combining all these results eventually yields a quantitative bound on the error made in practice by approximating  $\nu_*$  by the empirical distribution of the particular particle system:

**Theorem 6.1.2.** Under the conditions of Theorem 6.1.1, suppose that  $\kappa$  given by (6.5) is positive. There exists C > 0 such that for all  $N \in \mathbb{N}$ ,  $\gamma \in (0, \gamma_0]$ ,  $t \ge 0$  and  $\mu_0 \in \mathcal{P}(\mathbb{T}^{dN})$ , if  $(\mathbf{X}_k)_{k \in \mathbb{N}}$ is a Markov chain with initial distribution  $\mu_0$  and transition kernel  $R_{N,\gamma}$ ,

$$\mathbb{E}\left[\mathcal{W}_1\left(\pi(\mathbf{X}_{\lfloor t/\gamma \rfloor}), \nu_*\right)\right] \leqslant C\left(\sqrt{\gamma} + \alpha(N) + e^{-\kappa t}\right) \,,$$

where

$$\alpha(N) = \begin{cases} N^{-1/2} & \text{if } d = 1, \\ N^{-1/2} \ln(1+N) & \text{if } d = 2, \\ N^{-1/d} & \text{if } d > 2. \end{cases}$$

All the constants in Theorems 6.1.1 and 6.1.2 (and all other results stated in this work) are explicit. More precisely,  $c_1$  and  $\gamma_0$  come from [122, Corollary 2.2] (see Proposition 6.2.1 below) where an explicit value is given, and all the other constants involved in our results can be tracked by following the explicit computations.

In Theorem 6.1.2, the speeds of the different convergences (exponential in the simulation time, with the square-root of the timestep and with  $\alpha$  of the number of particles) are optimal since they are optimal for non-interacting diffusions (i.e. the case  $\lambda = 0$ ), see in particular [62] for the large N asymptotic.

Other intermediary results will be established in the rest of the paper that are interesting by themselves: propagation of chaos (i.e.  $N \to \infty$ ) and continuous-time limit at a fixed time (even without the condition  $\kappa > 0$ ) respectively in Propositions 6.2.4 and 6.2.5. From that, results for the continuous-time process ( $\gamma = 0$ ), the equilibria ( $t = \infty$ ) or the non-linear process ( $N = \infty$ ), or when two parameters among three are sent to their limits, are then simple corollaries, see Section 6.2.5. All these results are summarised in Figure 6.1 at the end of this work.

Note that  $\exp(-\gamma\lambda(x))$  is the probability that the chain is not killed when it arrives at state x. The time step  $\gamma$  should be chosen in such a way that this probability is relatively large, say at least one half. In that case,  $\exp(\gamma \|\lambda\|_{\infty})$  is typically close to 1. In other words, the positivity of  $\kappa$  given by (6.5) is mostly a condition about  $L_{\lambda}$  being small enough.

This perturbation condition is different from the one considered in [133], where  $\|\lambda\|_{\infty}$  rather than  $L_{\lambda}$  is supposed to be small (while our main arguments are a direct adaptation of the coupling arguments of [133]). This difference comes from the fact that, in the present study, we work with the  $\mathcal{W}_1$  distance rather than the total variation one (which is a Wasserstein distance but associated to the discrete metric  $d(x,y) = \mathbb{1}_{x\neq y}$ ). Indeed, in our coupling arguments, we need to control  $|\lambda(x) - \lambda(y)|$  the difference between the death rates of two processes at different locations, which is bounded here by  $L_{\lambda}|x-y|$  and in [133] by  $2\|\lambda\|_{\infty}\mathbb{1}_{x\neq y}$ . In fact our argument for the long-time convergence may easily be adapted to the total variation distance framework, following [133]. Nevertheless this would be more troublesome in the study of the limit  $N \to \infty$ . Then, one needs to couple  $\eta_k$  (that admits a density with respect to the Lebesgue measure) with  $\pi(\mathbf{X}_k)$  (which is a sum of Dirac masses), so that the total variation distance is not adapted. This may be solved by considering  $\mathcal{W}_1 \to \text{total variation regularization results for (Euler schemes of)}$ diffusions, that can be established by coupling arguments again. Nevertheless, in order to focus on the other difficulties of the problem and for the sake of clarity, we decided to stick to the  $\mathcal{W}_1$  distance in all the different results of this work. Similar Wasserstein coupling arguments have been used in [38] on a similar problem (see next section) and in [167] for a different kind of mean-field interacting particle system (also with a similar perturbative condition corresponding to the fact  $\sigma$  in [167, Proposition 3.1] has to be positive, i.e. the interaction should be small with respect to the independent mixing).

Notice that, among all possible discretization schemes, we only considered the explicit Euler-Maruyama one. This choice was made for simplicity, but the proofs could be extended to other usual schemes. The main ingredient required is a Wasserstein curvature of order  $\gamma$  for a modified  $W_1$  distance (see Proposition 6.2.1, based on [122, Corollary 2.2]). Similarly, we only considered the case of an elliptic diffusion process with a constant diffusion matrix for simplicity (since we use [122, Corollary 2.2] which covers this case), although a similar Wasserstein contraction certainly holds in a much more general framework (even hypoelliptic non-elliptic, as in continuous-time settings [56]). As stated in the introduction, the present paper does not aim at the broadest generality, and by avoiding technicalities we want to highlight the main issue (i.e. the question of the uniformity of bounds in the various parameters).

# 6.2 Proofs

Let us first establish the preliminary result stated in the introduction:

Proof of Proposition 6.1.1. For  $n \in \mathbb{N}$ , denote

$$\eta_n = \mathcal{L}aw(Y_n), \qquad \nu_n = \mathcal{L}aw\left(\tilde{Z}_n \mid \tilde{T} \ge n\gamma\right).$$

Since  $\nu_0 = \eta_0$ , suppose by induction that  $\nu_n = \eta_n$  for some  $n \in \mathbb{N}$ . Keeping the notations introduced in the definition of the kernel  $Q_{\mu}$ , consider the events  $B_k = \{U_k \ge p(X_k)\}$ . Then, for all bounded measurable f,

$$Q_{\mu}f(x) = \mathbb{E}(f(X))$$
  
=  $\mathbb{E}\left(f(X)\sum_{k\in\mathbb{N}} \mathbb{1}_{B_{k}\cap(\bigcap_{j=0}^{k-1}B_{j}^{c})}\right)$   
=  $\mathbb{E}\left(f(X_{0})\mathbb{1}_{U_{0}\geqslant p(X_{0})}\right) + \sum_{k\geqslant 1}\mathbb{E}\left(f(X_{k})\mathbb{1}_{U_{k}\geqslant p(X_{k})}\right)\prod_{j=0}^{k-1}\mathbb{P}\left(B_{j}^{c}\right)$   
=  $K\left[f(1-p)\right](x) + \sum_{k\geqslant 1}\mu K\left[f(1-p)\right](\mu Kp)^{k-1}Kp(x).$ 

In particular, integrating with respect to  $\mu$ , we obtain

$$\mu Q_{\mu} f = \mu K [f(1-p)] \sum_{k \in \mathbb{N}} (\mu K p)^{k} = \frac{\mu K [f(1-p)]}{\mu K [1-p]}.$$

Applied with  $\mu = \eta_n$ , this reads

$$\eta_{n+1}f = \mathbb{E}(f(Y_{n+1})) = \mathbb{E}(\mathbb{E}(f(Y_{n+1}) \mid Y_n)) = \eta_n Q_{\eta_n} f = \frac{\eta_n K[f(1-p)]}{\eta_n K[1-p]}$$

On the other hand,

$$\mathbb{E}\left(f(\tilde{Z}_{n+1})\mathbb{1}_{\tilde{T}>(n+1)\gamma}\right) = \mathbb{E}\left(f\left(\tilde{Z}_{n+1}\right)\mathbb{1}_{\tilde{T}>n\gamma}\mathbb{1}_{U_n \ge p(\tilde{Z}_{n+1})}\right) \\
= \mathbb{E}\left(f\left(\tilde{Z}_{n+1}\right)\left(1-p\left(\tilde{Z}_{n+1}\right)\right)\mathbb{1}_{\tilde{T}>n\gamma}\right)$$

$$= \mathbb{P}\left(\tilde{T} > n\gamma\right)\nu_n K\left[f(1-p)\right] \,,$$

frow which

$$\nu_{n+1}f = \frac{\mathbb{E}\left(f(\tilde{Z}_{n+1})\mathbb{1}_{\tilde{T}>(n+1)\gamma}\right)}{\mathbb{P}\left(\tilde{T}>(n+1)\gamma\right)} = \frac{\mathbb{P}\left(\tilde{T}>n\gamma\right)\nu_n K\left[f(1-p)\right]}{\mathbb{P}\left(\tilde{T}>n\gamma\right)\nu_n K\left[1-p\right]} = \frac{\nu_n K\left[f(1-p)\right]}{\nu_n K\left[1-p\right]},$$

which concludes.

# 6.2.1 The basic coupling

The long-time estimates needed to prove convergence toward equilibrium and uniform in time estimates in N and  $\gamma$  are based on the fact that, as long as particles don't die, they follow the chain (6.3) which, like its continuous-time counterpart (6.1), have some mixing properties. In order to quantify the latters, we start by stating [122, Corollary 2.2] in a suitable way in our context.

**Proposition 6.2.1.** There exists  $c_1, a, \gamma_0 > 0$  (that all depend only on the drift b of (6.1) and on the dimension d) such that, denoting  $\rho(x, y) = (1 - \exp(-a|x - y|))/a$  for  $x, y \in \mathbb{T}^d$ , then  $\rho$  is a metric on  $\mathbb{T}^d$  with

$$\forall \gamma \in (0, \gamma_0], \ \forall \mu, \nu \in \mathcal{P}(\mathbb{T}^d), \qquad \mathcal{W}_{\rho}(\mu K, \nu K) \leqslant (1 - c_1 \gamma) \mathcal{W}_{\rho}(\mu, \nu).$$

*Proof.* This is [122, Corollary 2.2] applied to a diffusion with smooth drift on the torus, in which case the distance on  $\mathbb{T}^d$  for which the contraction holds is  $\rho$ .

In the rest of the paper,  $\rho$  is the metric and  $c_1, a, \gamma_0$  are the constants given by Proposition 6.2.1. Remark that  $\rho$  is equivalent to the Euclidian metric, with

 $\beta |x-y| \leqslant \rho(x,y) \leqslant |x-y| \quad \text{for} \quad \beta = 2(1-e^{-a\sqrt{d}/2})/(a\sqrt{d}),$ 

where we used that the diameter of  $\mathbb{T}^d$  is  $\sqrt{d}/2$  and that  $r \mapsto (1 - \exp(-ar))/a$  is a concave function with derivative 1 at zero. In particular,  $\mathcal{W}_1$  and  $\mathcal{W}_{\rho}$  are equivalent.

Now, in this particle system, the contraction property of the chain (6.3) may be counterbalanced by the death/resurrection mechanism through which particles interact. Indeed, considering two systems of N interacting particles, for  $i \in [\![1, N]\!]$  the previous result means that we can couple the  $i^{th}$  particles of both systems to get closer one to the other (on average), as long as they don't die. But then, one of the two particle can die and resurrect far from the other, or even if they die simultaneously they may resurrect far apart one from the other. That being said, first, the closer they get, the easier it is to couple them in order to die simultaneously, and second, when they die simultaneously, keeping the particles close one to the other amount to do a suitable coupling of the laws from which the particles are resurrected. This is quantified in the following proposition.

In all the rest of the paper, we suppose that  $\lambda$  is  $L_{\lambda}$ -Lipschitz (but not necessarily that  $\kappa$  given by (6.5) is positive).

**Proposition 6.2.2.** Let  $\mu_0, \mu_1, \mu'_0, \mu'_1 \in \mathcal{P}(\mathbb{T}^d)$  and let  $(X_0, X'_0)$  (resp.  $(X_1, X'_1)$ ) be a coupling of  $\mu_0 K$  and  $\mu'_0 K$  (resp.  $\mu_1 K$  and  $\mu'_1 K$ ). Then

$$\mathcal{W}_{\rho}\left(\mu_{0}Q_{\mu_{1}},\mu_{0}'Q_{\mu_{1}'}\right) \leqslant h\left(\mathbb{E}\left(\rho(X_{0},X_{0}')\right)+\frac{q_{0}}{1-q_{1}}\mathbb{E}\left(\rho(X_{1},X_{1}')\right)\right)$$

where

$$h = 1 - \min p + (a\beta)^{-1} \gamma L_{\lambda}$$

and, considering  $U \sim \mathcal{U}([0,1])$  independent from  $(X_0, X'_0, X_1, X'_1)$ ,

$$q_i = \mathbb{P}(U < p(X_i) \land p(X'_i)), \quad i = 0, 1.$$

Proof. Let  $(X_k, X'_k, U_k)_{k \in \mathbb{N}}$  be a sequence of independent triplet of random variables such that, for all  $k \in \mathbb{N}$ ,  $U_k \sim \mathcal{U}([0, 1])$  is independent from  $(X_k, X'_k)$ , which are such as defined in the proposition for k = 0 and 1 and, for k > 1, have the same distribution as  $(X_1, X'_1)$ . Set  $H = \inf\{n \in \mathbb{N}, U_n < p(X_n)\}$  and  $H' = \inf\{n \in \mathbb{N}, U_n < p(X'_n)\}$ . Then, by considering the law of  $(X_k, U_k)_{k \in \mathbb{N}}$  alone, it is clear that  $X_H \sim \mu_0 Q_{\mu_1}$  and, similarly,  $X'_{H'} \sim \mu'_0 Q_{\mu'_1}$ , so that

$$\mathcal{W}_{\rho}\left(\mu_{0}Q_{\mu_{1}},\mu_{0}'Q_{\mu_{1}'}\right) \leq \mathbb{E}\left(\rho\left(X_{H},X_{H'}'\right)\right).$$

Different cases are distinguished depending on the value of H and H'. In the simplest case, none of the particles dies:

$$\mathbb{E}\left(\rho\left(X_{H}, X_{H'}'\right) \mathbb{1}_{H=H'=0}\right) = \mathbb{E}\left(\rho\left(X_{0}, X_{0}'\right) \mathbb{1}_{U_{0} \geqslant p(X_{0}) \lor p(X_{0}')}\right) \\ \leqslant \mathbb{E}\left(\rho\left(X_{0}, X_{0}'\right) \mathbb{1}_{U_{0} \geqslant \min p}\right) \\ \leqslant \left(1 - \min p\right) \mathbb{E}\left(\rho\left(X_{0}, X_{0}'\right)\right),$$

where we used the independence between  $U_0$  and  $(X_0, X'_0)$ . In the second case, only one particle dies: using that  $\|\rho\|_{\infty} \leq 1/a$ ,

$$\mathbb{E} \left( \rho \left( X_{H}, X_{H'}' \right) \mathbb{1}_{H \wedge H' = 0 < H \vee H'} \right) \leq a^{-1} \mathbb{P} \left( U_{0} \in [p(X_{0}) \wedge p(X_{0}'), p(X_{0}) \vee p(X_{0}')] \right) \\
= a^{-1} \mathbb{E} \left( |p(X_{0}) - p(X_{0}')| \right) \\
\leq a^{-1} \gamma L_{\lambda} \mathbb{E} \left( |X_{0} - X_{0}'| \right) \\
\leq (a\beta)^{-1} \gamma L_{\lambda} \mathbb{E} \left( \rho(X_{0}, X_{0}') \right).$$

In the third case, both particles die  $k \ge 1$  times:

$$\mathbb{E}\left(\rho\left(X_{H}, X_{H'}'\right) \mathbb{1}_{H=H'=k}\right) = \mathbb{E}\left(\rho\left(X_{k}, X_{k}'\right) \mathbb{1}_{U_{k} \geqslant p(X_{k}) \lor p(X_{k}')} \prod_{j=0}^{k-1} \mathbb{1}_{U_{j} < p(X_{j}) \land p(X_{j}')}\right) \\ \leqslant q_{0}q_{1}^{k-1} \mathbb{E}\left(\rho\left(X_{k}, X_{k}'\right) \mathbb{1}_{U_{k} \geqslant \min p}\right) \\ \leqslant q_{0}q_{1}^{k-1} \left(1 - \min p\right) \mathbb{E}\left(\rho\left(X_{1}, X_{1}'\right)\right).$$

Finally, combining the computations of the last two cases, the fourth one reads, for  $k \ge 1$ ,

$$\mathbb{E}\left(\rho\left(X_{H}, X_{H'}'\right) \mathbb{1}_{H \wedge H' = k < H \vee H'}\right) \leqslant a^{-1}q_{0}q_{1}^{k-1}\mathbb{P}\left(U_{k} \in [p(X_{k}) \wedge p(X_{k}'), p(X_{k}) \vee p(X_{k}')]\right) \\
\leqslant (a\beta)^{-1}q_{0}q_{1}^{k-1}\gamma L_{\lambda}\mathbb{E}\left(\rho\left(X_{1}, X_{1}'\right)\right).$$

Summing these four cases concludes.

#### 6.2.2 Long-time convergence

For  $N \in \mathbb{N}_*$  denote  $\rho_N$  the metric on  $\mathbb{T}^{dN}$  given by

$$\rho_N(x,y) = \sum_{i=1}^N \rho(x_i, y_i).$$

The following result is similar to the results of [133, 167, 38] and based on the same coupling argument.

**Proposition 6.2.3.** There exists  $c_2 > 0$  (that depends only on the drift b of (6.1) and on the dimension d) such that for all  $\gamma \in (0, \gamma_0]$   $N \in \mathbb{N}$ , and all  $\mu, \nu \in \mathcal{P}(\mathbb{T}^{dN})$ ,

$$\mathcal{W}_{\rho_N}\left(\mu R_{N,\gamma}, \nu R_{N,\gamma}\right) \leqslant (1 - \gamma \kappa) \mathcal{W}_{\rho_N}(\mu, \nu).$$

with  $\kappa$  given by (6.5).

*Proof.* It is in fact sufficient to prove this for  $\mu = \delta_x$  and  $\nu = \delta_y$  for any  $x, y \in \mathbb{T}^{dN}$ . Indeed, assuming the result proven for Dirac masses, in the general case, considering  $(\mathbf{X}_0, \mathbf{Y}_0)$  an optimal coupling of  $\mu$  and  $\nu$  and  $(\mathbf{X}_1, \mathbf{Y}_1)$  an optimal coupling of  $R(\mathbf{X}_0, \cdot)$  and  $R(\mathbf{Y}_0, \cdot)$ , then  $\mathbf{X}_1 \sim \mu R$  and  $\mathbf{Y}_1 \sim \nu R$ , so that

$$\begin{split} W_{\rho_N} \left( \mu R, \nu R \right) &\leqslant & \mathbb{E} \left( \rho_N \left( \mathbf{X}_1, \mathbf{Y}_1 \right) \right) \\ &= & \mathbb{E} \left( \mathbb{E} \left( \rho_N \left( \mathbf{X}_1, \mathbf{Y}_1 \right) \mid \left( \mathbf{X}_0, \mathbf{Y}_0 \right) \right) \right) \\ &= & \mathbb{E} \left( \mathbb{E} \left( \mathcal{W}_{\rho_N} \left( \delta_{\mathbf{X}_0} R, \delta_{\mathbf{Y}_0} R \right) \mid \left( \mathbf{X}_0, \mathbf{Y}_0 \right) \right) \right) \\ &\leqslant & \left( 1 - \gamma \kappa \right) \mathbb{E} \left( \rho_N \left( \mathbf{X}_0, \mathbf{Y}_0 \right) \right) \\ &= & \left( 1 - \gamma \kappa \right) \mathcal{W}_{\rho_N} \left( \mu, \nu \right) \,. \end{split}$$

Hence, in the following, we fix  $x, y \in \mathbb{T}^{dN}$ . Let  $(X_i, Y_i)_{i \in [\![1,N]\!]}$  be independent pairs of random variables in  $\mathbb{T}^d$  where, for all  $i \in [\![1,N]\!]$ ,  $(X_i, Y_i)$  is an optimal coupling of  $Q_{\pi(x)}(x_i, \cdot)$  and  $Q_{\pi(y)}(y_i, \cdot)$ . Then  $(\mathbf{X}, \mathbf{Y})$  is a coupling of  $R(x, \cdot)$  and  $R(y, \cdot)$ , so that

$$\mathcal{W}_{\rho_N}\left(\delta_x R, \delta_y R\right) \leqslant \mathbb{E}\left(\rho_N\left(\mathbf{X}, \mathbf{Y}\right)\right) = \sum_{i=1}^N \mathbb{E}\left(\rho\left(X_i, Y_i\right)\right)$$
$$= \sum_{i=1}^N \mathcal{W}_{\rho}\left(Q_{\pi(x)}(x_i, \cdot), Q_{\pi(y)}(y_i, \cdot)\right).$$

We want to apply Proposition 6.2.2 with  $\mu_1 = \pi(x)$ ,  $\mu_0 = \delta_{x_i}$ ,  $\mu'_1 = \pi(y)$  and  $\mu'_0 = \delta_{y_i}$ . To do so, for all  $i \in [\![1, N]\!]$ , we consider  $(\tilde{X}_i, \tilde{Y}_i)$  an optimal coupling of  $K(x_i, \cdot)$  and  $K(y_i, \cdot)$ . From Proposition 6.2.1,

$$\mathbb{E}\left(\rho\left(\tilde{X}_{i},\tilde{Y}_{i}\right)\right) \leqslant (1-c_{1}\gamma)\rho(x_{i},y_{i}).$$

$$(6.6)$$

Moreover, if  $J \sim \mathcal{U}(\llbracket 1, N \rrbracket)$  is independent from the  $(\tilde{X}_i, \tilde{Y}_i)$ 's, we remark that  $(\tilde{X}_J, \tilde{Y}_J)$  is a coupling of  $\pi(x)K$  and  $\pi(y)K$ . Proposition 6.2.2 applied with these couplings reads, for all

 $i\in [\![1,N]\!],$ 

$$\mathcal{W}_{\rho}\left(Q_{\pi(x)}(x_{i},\cdot),Q_{\pi(y)}(y_{i},\cdot)\right) \leqslant h\left(\mathbb{E}\left(\rho(\tilde{X}_{i},\tilde{Y}_{i})\right) + \frac{q_{i}}{1-q_{*}}\mathbb{E}\left(\rho(\tilde{X}_{J},\tilde{Y}_{J})\right)\right)$$
(6.7)

where, if  $U \sim \mathcal{U}([0,1])$  is independent from the previous variables,

$$q_i := \mathbb{P}\left(U < p(\tilde{X}_i) \land p(\tilde{Y}_i)\right)$$

and, conditioning on the value of J,

$$q_* := \mathbb{P}\left(U < p(\tilde{X}_J) \land p(\tilde{Y}_J)\right) = \frac{1}{N} \sum_{i=1}^N q_i.$$

Summing (6.7) over  $i \in [\![1, N]\!]$  and applying (6.6) yields

$$\mathcal{W}_{\rho_N}\left(\delta_x R, \delta_y R\right) \leqslant h\left(\left(1-c_1\gamma\right)\sum_{i=1}^N \rho(x_i, y_i) + \frac{Nq_*}{1-q_*}\mathbb{E}\left(\rho(\tilde{X}_J, \tilde{Y}_J)\right)\right).$$

Applying Proposition 6.2.1 again,

$$\mathbb{E}\left(\rho(\tilde{X}_J, \tilde{Y}_J)\right) = \frac{1}{N} \sum_{i=1}^N \mathbb{E}\left(\rho(\tilde{X}_i, \tilde{Y}_i)\right) \leqslant \frac{1}{N} \left(1 - c_1\gamma\right) \sum_{i=1}^N \rho(x_i, y_i),$$

and the previous inequality becomes

$$\mathcal{W}_{\rho_N}\left(\delta_x R, \delta_y R\right) \leqslant \frac{h\left(1-c_1\gamma\right)}{1-q_*}\rho_N(x,y).$$

Bounding  $1 - q_* \ge 1 - \max p \ge \exp(-\gamma \|\lambda\|_{\infty})$  and  $\max p - \min p \le \sqrt{d/2}\gamma L_{\lambda}$  yields

$$\frac{h\left(1-c_{1}\gamma\right)}{1-q_{*}} \leqslant \left(1-c_{1}\gamma\right) \frac{1-\max p+\max p-\min p+(a\beta)^{-1}\gamma L_{\lambda}}{1-\max p}$$
$$\leqslant \left(1-c_{1}\gamma+\gamma L_{\lambda}e^{\gamma\|\lambda\|_{\infty}}\left((a\beta)^{-1}+\sqrt{d}\right),\right)$$

which concludes.

As a direct consequence, Proposition 6.2.3 gives

$$\forall m \in \mathbb{N}, \qquad \mathcal{W}_{\rho_N}\left(\mu R^m, \nu R^m\right) \leqslant e^{-\kappa m \gamma} \mathcal{W}_{\rho_N}(\mu, \nu),$$

with  $\kappa$  that does not depends on N nor  $\gamma$ . Provided  $\kappa > 0$ , and since  $\mathcal{P}(\mathbb{T}^{dN})$  is complete for  $\mathcal{W}_1$ (hence for  $\mathcal{W}_{\rho_N}$ ) the Banach fixed-point theorem implies then that R admits a unique invariant measure toward which it converges at rate  $\gamma \kappa$ .

In the rest of the paper,  $\kappa$  is given by (6.5) (but is not necessarily assumed positive).

Proof of Theorem 6.1.1. The first part of the theorem has already been proven in Proposition 6.2.3. The last statement then follows from the first part, the equivalence between the Euclidean distance and  $\rho_N$  and the fact  $\mathcal{W}_{\rho_N}(\nu,\mu) \leq \|\rho_N\|_{\infty} = N\sqrt{d}/2$  for all  $\nu, \mu \in \mathcal{P}(\mathbb{T}^{dN})$ .  $\Box$ 

-	-	-	

### 6.2.3 Propagation of chaos

Recall that  $\eta_k$  is the law at time k of the non-homogeneous Markov chain  $(Y_k)_{k\in\mathbb{N}}$  on  $\mathbb{T}^d$  introduced in Section 6.1.2 with transition kernels  $Q_{\eta_k}$  and initial condition  $\eta_0$ , and that  $R = R_{N,\gamma}$ is the transition kernel of the Markov chain  $(\mathbf{X}_k)_{k\in\mathbb{N}}$  on  $\mathbb{T}^{dN}$ .

**Lemma 6.2.1.** There exist  $C_1 > 0$  such that for all  $N \in \mathbb{N}$ ,  $\gamma \in (0, \gamma_0]$ ,  $\eta \in \mathcal{P}(\mathbb{T}^d)$  and  $\mu \in \mathcal{P}(\mathbb{T}^{dN})$ ,

$$\mathcal{W}_{\rho_N}\left(\mu R, \mu Q_{\eta}^{\otimes N}\right) \leqslant \gamma N C_1 \int_{\mathbb{T}^{dN}} \mathcal{W}_{\rho}\left(\pi(x), \eta\right) \mu(dx)$$

*Proof.* Similarly to the proof of Proposition 6.2.3, we start with the case  $\mu = \delta_x$  for some  $x \in \mathbb{T}^{dN}$ . Let  $(X_i, Y_i)_{i \in [\![1,N]\!]}$  be N independent pairs of random variables such that for all  $i \in [\![1,N]\!]$ ,  $(X_i, Y_i)$  is an optimal coupling of  $Q_{\pi(x)}(x_i, \cdot)$  and  $Q_{\eta}(x_i, \cdot)$ . Then  $(\mathbf{X}, \mathbf{Y})$  is a coupling of  $R_N(x, \cdot)$  and  $Q_{\eta}^{\otimes N}(x, \cdot)$ , so that

$$\mathcal{W}_{\rho_N}\left(\delta_x R_N, \delta_x Q_\eta^{\otimes N}\right) \leqslant \mathbb{E}\left(\rho_N\left(X,Y\right)\right) \\ = \sum_{i=1}^N \mathbb{E}\left(\rho(X_i,Y_i)\right) = \sum_{i=1}^N \mathcal{W}_{\rho}\left(\delta_{x_i} Q_{\pi(x)}, \delta_{x_i} Q_\eta\right).$$

From Proposition 6.2.2 (bounding  $q_0 \leq \max p \leq \gamma \|\lambda\|_{\infty}$  and  $1 - q_1 \geq 1 - \max p \geq \exp(-\gamma_0 \|\lambda\|_{\infty})$ )

$$\mathcal{W}_{\rho}\left(\delta_{x_{i}}Q_{\pi(x)}, \delta_{x_{i}}Q_{\eta}\right) \leqslant \gamma \|\lambda\|_{\infty} \left(1 + (a\beta)^{-1}\gamma_{0}L_{\lambda}\right) e^{\gamma_{0}\|\lambda\|_{\infty}} \mathcal{W}_{\rho}\left(\pi(x), \eta\right)$$
$$:= \gamma C_{1}\mathcal{W}_{\rho}\left(\pi(x), \eta\right) .$$

Now in the general case where  $\mu$  is not a Dirac mass, considering  $Z_0 \sim \mu$ , and  $(Z_1, Z_2)$  an optimal coupling of  $R(Z_0, \cdot)$  and  $Q_{\eta}^{\otimes N}(Z_0, \cdot)$  and conditioning with respect to  $Z_0$ ,

$$\mathcal{W}_{\rho_N}\left(\mu R_N, \mu Q_{\eta}^{\otimes N}\right) \leqslant \mathbb{E}\left(\rho_N(Z_1, Z_N)\right) \leqslant \gamma N C_1 \mathbb{E}\left(\mathcal{W}_{\rho}\left(\pi(Z_0), \eta\right)\right).$$

**Proposition 6.2.4.** There exist  $C_2, C_3 > 0$  such that for all  $N \in \mathbb{N}$ ,  $\gamma \in (0, \gamma_0]$ ,  $m \in \mathbb{N}$  and  $\eta_0 \in \mathcal{P}(\mathbb{T}^d)$ , first,

$$\mathcal{W}_{\rho_N}\left(\eta_0^{\otimes N} R^m, \eta_m^{\otimes N}\right) \leqslant C_2 N \alpha(N) \gamma \sum_{s=1}^m (1 - \gamma \kappa)^{s-1}, \qquad (6.8)$$

and second, if  $(\mathbf{X}_k)_{k \in \mathbb{N}}$  is a Markov chain with initial distribution  $\eta_0^{\otimes N}$  and transition kernel R, then

$$\mathbb{E}\left(\mathcal{W}_{\rho}\left(\pi(\mathbf{X}_{m}),\eta_{m}\right)\right) \leqslant C_{3}\alpha(N)\left(1+\gamma\sum_{s=1}^{m}(1-\gamma\kappa)^{s-1}\right).$$
(6.9)

Remark that when  $\kappa > 0$ ,  $\gamma \sum_{s=1}^{m} (1 - \gamma \kappa)^{s-1} \leq 1/\kappa$  so that (6.8) and (6.9) yield uniform in time estimates. On the contrary, when k < 0, the estimates are exponentially bad in  $t = m\gamma$ .

*Proof.* We start with the proof of (6.8), for  $m \ge 1$  (the case m = 0 being trivial). From the triangular inequality, Proposition 6.2.3 and Lemma 6.2.1,

$$r_m := \mathcal{W}_{\rho_N} \left( \eta_0^{\otimes N} R^m, \eta_m^{\otimes N} \right) \leqslant \mathcal{W}_{\rho_N} \left( \eta_0^{\otimes N} R^m, \eta_{m-1}^{\otimes N} R \right) + \mathcal{W}_{\rho_N} \left( \eta_{m-1}^{\otimes N} R, \eta_{m-1}^{\otimes N} Q_{\eta_{m-1}}^{\otimes N} \right)$$

$$\leqslant \quad (1-\kappa\gamma)r_{m-1} + \gamma NC_1 \int_{\mathbb{T}^{dN}} \mathcal{W}_{\rho}\left(\pi(x), \eta_{m-1}\right) \eta_{m-1}^{\otimes N}(\mathrm{d}x) \,.$$

Since  $\mathcal{W}_{\rho} \leq \mathcal{W}_{1}$ , estimating the last term is a classical question, that is to bound the expected Wasserstein distance between the empirical measure of a sample of N independent and identically distributed random variables and their common law. From [62, Theorem 1] (and since on the torus the moments of probability measures are uniformly bounded), there exists some C' > 0independent from  $\eta_0$ , m, N and  $\gamma$  such that

$$\int_{\mathbb{T}^{dN}} \mathcal{W}_1(\pi(x), \eta_{m-1}) \eta_{m-1}^{\otimes N}(\mathrm{d}x) \leqslant C'\alpha(N).$$

Since  $r_0 = 0$ , a direct induction concludes the proof of (6.8). To prove (6.9), let  $(\mathbf{X}, \mathbf{Y})$  be an optimal coupling of  $\eta_0^{\otimes N} R^m$  and  $\eta_m^{\otimes N}$ . Considering  $J \sim \mathcal{U}(\llbracket 1, N \rrbracket)$  independent from  $(\mathbf{X}, \mathbf{Y})$  then, conditionally to  $(\mathbf{X}, \mathbf{Y}), (X_J, Y_J)$  is a coupling of  $\pi(\mathbf{X})$ and  $\pi(\mathbf{Y})$ , so that

$$\mathcal{W}_{\rho}\left(\pi(\mathbf{X}),\pi(\mathbf{Y})\right) \leqslant \mathbb{E}\left(\rho(X_J,Y_J) \mid (\mathbf{X},\mathbf{Y})\right) = \frac{1}{N}\rho_N(\mathbf{X},\mathbf{Y}).$$

Taking the expectation in

$$\mathcal{W}_{\rho}\left(\pi(\mathbf{X}),\eta_{m}\right) \leqslant \mathcal{W}_{\rho}\left(\pi(\mathbf{X}),\pi(\mathbf{Y})\right) + \mathcal{W}_{\rho}\left(\pi(\mathbf{Y}),\eta_{m}\right)$$

we conclude with (6.8) and [62, Theorem 1] again.

**Corollary 6.2.1.** With the notations of Proposition 6.2.4, for all  $k \in [1, N]$ ,

$$\mathcal{W}_{\rho_k}\left(\mathcal{L}aw(X_{1,m},\ldots,X_{k,m}),\eta_m^{\otimes k}\right) \leqslant C_2k\alpha(N)\gamma\sum_{s=1}^m (1-\gamma\kappa)^{s-1}.$$

*Proof.* Let  $(\mathbf{X}, \mathbf{Y})$  be an optimal coupling of  $\eta_0^{\otimes N} R^m$  and  $\eta_m^{\otimes N}$ , and let  $\sigma$  be uniformly distributed over the set of permutations of N elements, independent from  $(\mathbf{X}, \mathbf{Y})$ . Since the laws of **X** and **Y** are exchangeable,  $\mathbf{X}_{\sigma} = (X_{\sigma(1)}, \dots, X_{\sigma(N)})$  has the same law as **X**, in particular  $(\mathbf{X}_{\sigma(1)},\ldots,X_{\sigma(k)})$  has the same law as  $(X_1,\ldots,X_k)$ . The same goes for  $\mathbf{Y}_{\sigma}$ , and

$$\mathbb{E}\left(\sum_{i=1}^{k} \rho(X_{\sigma(i)}, Y_{\sigma(i)})\right) = k\mathbb{E}\left(\rho(X_{\sigma(1)}, Y_{\sigma(1)})\right) = \frac{k}{N}\mathbb{E}\left(\rho_{N}(\mathbf{X}, \mathbf{Y})\right)$$
$$= \frac{k}{N}\mathcal{W}_{\rho_{N}}\left(\eta_{0}^{\otimes N}R^{m}, \eta_{m}^{\otimes N}\right),$$

and Proposition 6.2.4 concludes.

Corollary 6.2.1 means that, for any fixed  $k \in \mathbb{N}_*$ , as N goes to infinity, the k-marginals of the system of particles converge toward the law of k independent non-linear chains, which is the so-called propagation of chaos phenomenon.

#### 6.2.4Discrete to continuous time

We start by defining  $(\overline{Y}_t)_{t \ge 0}$  and  $(\overline{\mathbf{X}}_t)_{t \ge 0}$  the continuous-time analoguous of the chains  $(Y_k)_{k \in \mathbb{N}}$ on  $\mathbb{T}^d$  and  $(\mathbf{X}_k)_{k\in\mathbb{N}}$  on  $\mathbb{T}^{dN}$  defined in Section 6.1.2. We start with the non-linear process. For

 $t \ge 0$ , let

$$\bar{\eta}_t = \mathcal{L}aw(Z_t \mid T > t)$$

where Z solves (6.1) with initial distribution  $\eta_0$  and T is given by (6.2). We define  $(\overline{Y}_t)_{t\geq 0}$  as follows. Set  $\overline{Y}_0 = Z_0 \sim \eta_0$ ,  $T_0 = 0$  and suppose that  $T_n$  and  $(\overline{Y}_t)_{t\in[0,T_n]}$  have been defined for some  $n \in \mathbb{N}$ . Let  $(B_t)_{t\geq 0}$  be a new Brownian motion on  $\mathbb{T}^d$  and  $E \sim \mathcal{E}(1)$ , independent one from the other. Let  $\tilde{Y}$  be the solution of

$$\mathrm{d}\tilde{Y}_t = b(\tilde{Y}_t)\mathrm{d}t + \mathrm{d}B_t$$

for  $t \ge T_n$  with  $\tilde{Y}_{T_n} = \overline{Y}_{T_n}$  and let

$$T_{n+1} = \inf \left\{ t > T_n, \ E \leqslant \int_{T_n}^t \lambda(\tilde{Y}_s) \mathrm{d}s \right\} .$$

For  $t \in (T_n, T_{n+1})$ , set  $\overline{Y}_t = \tilde{Y}_t$ . Finally, draw a new  $\overline{Y}_{T_{n+1}}$  according to  $\overline{\eta}_{T_{n+1}}$ . By induction  $T_n$  and  $(Y_t)_{t \in [0,T_n]}$  are then defined for all  $n \in \mathbb{N}$ . Since  $\lambda$  is bounded,  $T_n$  almost surely goes to infinity when  $n \to \infty$  so that  $(\overline{Y}_t)_{t \ge 0}$  is defined for all  $t \ge 0$ . Similarly to Proposition 6.1.1, it can be established that  $\mathcal{L}aw(\overline{Y}_t) = \overline{\eta}_t$  for all  $t \ge 0$ .

Now, as in Section 6.1.2, from the non-linear process  $(\overline{Y}_t)_{t\geq 0}$ , the interacting particles  $(\overline{\mathbf{X}}_t)_{t\geq 0}$  are obtained by replacing  $\overline{\eta}_t$  by the empirical distribution of the system when particles die and are resurrected.

More precisely, let  $(E_{i,k}, B_{i,k}, J_{i,k})_{i \in [\![1,N]\!], k \in \mathbb{N}}$  be a family of independent triplet of independent random variables where, for all  $i \in [\![1,N]\!]$  and  $k \in \mathbb{N}$ ,  $E_{i,k} \sim \mathcal{E}(1)$ ,  $J_{i,k} \sim \mathcal{U}([\![1,N]\!])$  (except if k = 0, in which case  $J_{i,k} = i$  almost surely) and  $B_{i,k} = (B_{i,k,t})_{t \ge 0}$  is a d-dimensional Brownian motion. From these variables, we simultaneously define by induction the process and its death times  $(T_{i,k})_{i \in [\![1,N]\!], k \in \mathbb{N}}$  as follows. First, set  $\overline{\mathbf{X}}_0 = x$  and  $T_{i,0} = 0$  for all  $i \in [\![1,N]\!]$ . For all  $i \in [\![1,N]\!]$ , set  $\hat{X}_{i,0,0} = x_i$  and for  $k \ge 1$ , assuming that  $\overline{\mathbf{X}}$  has been constructed up to time  $T_{i,k}$ , set

$$\hat{X}_{i,k,0} = \lim_{t \leq T_{i,k}} \overline{X}_{J_{i,k},t} \,. \tag{6.10}$$

For all  $k \in \mathbb{N}$ , for  $t \ge T_{i,k}$ , let  $\hat{X}_{i,k}$  solve

$$\mathrm{d}\hat{X}_{i,k,t} = b\left(\hat{X}_{i,k,t}\right)\mathrm{d}t + \mathrm{d}B_{i,k,t},$$

set

$$T_{i,k+1} = T_{i,k} + \inf\left\{t \ge 0, \ E_{i,k} \le \int_0^t \lambda\left(\hat{X}_{i,k,s}\right) \mathrm{d}s\right\}$$

and for all  $t \in [T_{i,k}, T_{i,k+1})$ , set  $\overline{X}_{i,t} = \hat{X}_{i,k,t-T_{i,k}}$ .

Then  $\overline{\mathbf{X}}_t = (\overline{X}_{1,t}, \dots, \overline{X}_{N,t})$  is well-defined for all  $t \ge 0$ . Indeed, it is well defined for all  $t < S_1 := \min\{T_{i,1}, i \in [\![1,N]\!]\}$  the first death time of some particle, and is equal on this interval to  $(\hat{X}_{1,0,t}, \dots, \hat{X}_{N,0,t})$ , which is continuous on  $[0, S_1]$ . Hence, the limits involved in (6.10) are well defined for k = 1 and all  $i \in [\![1,N]\!]$  such that  $T_{i,1} = S_1$ . Then the algorithm above similarly defines the process up to the second time some particles die, etc.

Remark that most of the times (6.10) simply reads  $\hat{X}_{i,k,T_{i,k}} = \overline{X}_{J_{i,k},T_{i,k}}$  (at its  $k^{th}$  death time, the  $i^{th}$  particle is resurrected at the current position of the  $J_{i,k}^{th}$  particle). Indeed, the only case when this is not true is when the  $J_{i,k}^{th}$  particle dies at time  $T_{i,k}$ . Since the probability that

two or more particles die simultaneously is zero, this almost surely only occurs if  $J_{i,k} = i$ , i.e. if the particle is resurrected at its own position.

Denote  $(P_t)_{t\geq 0}$  the Markov semi-group associated with  $(\overline{\mathbf{X}}_t)_{t\geq 0}$ , i.e. for all  $t\geq 0$ ,  $P_t$  is the Markov kernel given by

$$P_t f(x) = \mathbb{E} \left( f(\mathbf{X}_t) \mid \mathbf{X}_0 = x \right)$$

We sometimes write  $P_t = P_{N,t}$  to specify the number of particles.

**Lemma 6.2.2.** There exist  $C_4 > 0$  such that for all  $N \in \mathbb{N}$ ,  $\gamma \in (0, \gamma_0]$  and  $\mu \in \mathcal{P}(\mathbb{T}^{dN})$ ,

$$\mathcal{W}_{\rho_N}\left(\mu R_{N,\gamma}, \mu P_{N,\gamma}\right) \leqslant NC_4 \gamma^{3/2}.$$

Proof. As in the proof of Lemma 6.2.1, it is sufficient to treat the case  $\mu = \delta_x$  with a fixed  $x \in \mathbb{T}^{dN}$ . Let  $(\overline{\mathbf{X}}_t)_{t \ge 0}$  be defined as above from random variables  $(E_{i,k}, B_{i,k}, J_{i,k})_{i \in [\![1,N]\!], k \in \mathbb{N}}$ . In particular,  $\overline{\mathbf{X}}_{\gamma} \sim \delta_x P_{\gamma}$ .

To define  $\mathbf{X}_1 \sim \delta_x R$ , for all  $i \in [\![1, N]\!]$  and  $k \in \mathbb{N}$ , consider  $(\tilde{X}_{i,k,t})_{t \ge 0}$  the solution to  $\tilde{X}_{i,k,0} = x_{J_{i,k}}$  and

$$\mathrm{d}\tilde{X}_{i,k,t} = b\left(\tilde{X}_{i,k,0}\right)\mathrm{d}t + \mathrm{d}B_{i,k,t}.$$

Denoting

$$H_i = \inf \left\{ k \in \mathbb{N}, \ E_{i,k} \ge \gamma \lambda \left( \tilde{X}_{i,k,\gamma} \right) \right\},\$$

set  $\mathbf{X}_1 := (\tilde{X}_{1,H_1,\gamma}, \dots, \tilde{X}_{N,H_N,\gamma}).$ 

Then  $(\mathbf{X}_1, \overline{\mathbf{X}}_{\gamma})$  is a coupling of  $R(x, \cdot)$  and  $P_{\gamma}(x, \cdot)$ , so that

$$\mathcal{W}_{\rho_N}\left(R(x,\cdot), P_{\gamma}(x,\cdot)\right) \leqslant \mathbb{E}\left(\rho_N(\mathbf{X}_1, \overline{\mathbf{X}}_{\gamma})\right) = \sum_{i=1}^N \mathbb{E}\left(\rho(X_{i,1}, \overline{X}_{i,\gamma})\right).$$

We now distinguish four cases, considering the events

$$\begin{array}{lll} B_{i,1} &=& \left\{ H_i = 0 \ \text{and} \ T_{i,1} > \gamma \right\} \\ B_{i,2} &=& \left\{ H_i = 1 \ \text{and} \ T_{i,1} \leqslant \gamma < T_{i,2} \wedge T_{J_{i,0},1} \right\} \\ B_{i,3} &=& \left\{ H_i = 1 \ \text{and} \ T_{i,1} > \gamma \right\} \cup \left\{ H_i = 0 \ \text{and} \ T_{i,1} \leqslant \gamma \right\} \\ B_{i,4} &=& \left\{ H_i \geqslant 2 \right\} \cup \left\{ T_{i,2} \leqslant \gamma \right\} \cup \left\{ T_{i,1} \vee T_{J_{i,1},1} \leqslant \gamma \right\}, \end{array}$$

that is, respectively: none of the two  $i^{th}$  particles dies; both the  $i^{th}$  particles die exactly once; one particle dies but not the other; at least two deaths are involved for one of the two particle. For all  $i \in [\![1, N]\!]$ ,  $\Omega = \bigcup_{j=1}^{4} B_{i,j}$ , so that

$$\mathbb{E}\left(\rho(X_{i,1},\overline{X}_{i,\gamma})\right) \leqslant \mathbb{E}\left(\rho(X_{i,1},\overline{X}_{i,\gamma})\left(\mathbb{1}_{B_{i,1}} + \mathbb{1}_{B_{i,2}} + \mathbb{1}_{B_{i,3}} + \mathbb{1}_{B_{i,4}}\right)\right).$$

Conclusion follows by gathering the four cases.

Case 1. It reduces to the classical case of diffusions, since

$$\mathbb{E}\left(|X_{i,1} - \overline{X}_{i,\gamma}| \mathbb{1}_{B_{i,1}}\right) = \mathbb{E}\left(|\tilde{X}_{i,0,\gamma} - \hat{X}_{i,0,\gamma}| \mathbb{1}_{B_{i,1}}\right) \leq \mathbb{E}\left(|\tilde{X}_{i,0,\gamma} - \hat{X}_{i,0,\gamma}|\right).$$

Then

$$\begin{aligned} |\tilde{X}_{i,0,t} - \hat{X}_{i,0,t}| &= \left| \int_0^t \left( b(x_i) - b\left( \hat{X}_{i,0,s} \right) \right) \mathrm{d}s \right| \\ &\leqslant \|\nabla b\|_{\infty} \int_0^t \left( |\tilde{X}_{i,0,s} - \hat{X}_{i,0,s}| + |x_i - \tilde{X}_{i,0,s}| \right) \mathrm{d}s \end{aligned}$$

By the Gronwall Lemma, for all  $t \ge 0$ , almost surely,

$$\sup_{s \in [0,t]} |\tilde{X}_{i,0,t} - \hat{X}_{i,0,t}| \leq \|\nabla b\|_{\infty} e^{t \|\nabla b\|_{\infty}} \int_{0}^{t} |x_{i} - \tilde{X}_{i,0,s}| \mathrm{d}s.$$
(6.11)

Since  $\tilde{X}_{i,0,s}$  is a Gaussian variable with mean  $x_i + sb(x_i)$  and variance s,

$$\mathbb{E}\left(|x_i - \tilde{X}_{i,0,s}|\right) \leq sb(x_i) + \mathbb{E}\left(|x_i + sb(x_i) - \tilde{X}_{i,0,s}|\right) \leq \|b\|_{\infty}s + \sqrt{s}.$$
(6.12)

As a consequence, for  $\gamma \leq \gamma_0$ ,

$$\mathbb{E}\left(|\tilde{X}_{i,0,\gamma} - \hat{X}_{i,0,\gamma}|\right) \leqslant \|\nabla b\|_{\infty} e^{\gamma_0 \|\nabla b\|_{\infty}} \int_0^{\gamma} \mathbb{E}\left(|x_i - \tilde{X}_{i,0,s}|\right) \mathrm{d}s \leqslant c\gamma^{3/2}.$$
(6.13)

Case 2. We bound

$$\mathbb{E}\left(|X_{i,1}-\overline{X}_{i,\gamma}|\mathbb{1}_{B_{i,2}}\right) \leqslant \mathbb{E}\left(\left(|\tilde{X}_{i,1,\gamma}-x_{J_{i,0}}|+|\hat{X}_{i,1,\gamma}-x_{J_{i,0}}|\right)\mathbb{1}_{B_{i,2}}\right).$$

Similarly to (6.12),

$$\mathbb{E}\left(|\tilde{X}_{i,1,\gamma} - x_{J_{i,0}}|\mathbb{1}_{B_{i,2}}\right) \leqslant \mathbb{E}\left(|\tilde{X}_{i,1,\gamma} - x_{J_{i,0}}|\mathbb{1}_{E_{i,0}\leqslant\gamma\|\lambda\|_{\infty}}\right) \leqslant c\gamma^{3/2},$$

where we used the independence of  $E_{i,0}$  from  $J_{i,1}$  and  $(\tilde{X}_{i,1,t})_{t \ge 0}$ . Denote  $(X'_i)_{t \ge 0}$  the solution of

$$dX'_{i,t} = b(X'_{i,t}) dt + \begin{cases} dB_{J_{i,0},0,t} & \text{for } t < T_{i,0} \\ dB_{i,1,t} & \text{for } t \ge T_{i,0} \end{cases}$$

with  $X'_{i,0} = x_{J_{i,0}}$ . Under the event  $B_{i,2}$ ,  $\hat{X}_{i,1,\gamma} = X'_{i,\gamma}$ . Moreover,  $J_{i,0}$ ,  $B_{J_{i,0},0}$  and  $B_{i,1}$  are independent from  $T_{i,0}$  and thus, by the strong Markov property,  $(X'_{i,t})_{t\geq 0}$  is independent from  $T_{i,0}$  and conditionally to  $J_{i,0}$  it has the same distribution as  $\hat{X}_{J_{i,0},0,t}$  (namely it is a diffusion solving (6.1) with initial condition  $x_{J_{i,0}}$ ). Hence,

$$\mathbb{E}\left(|\hat{X}_{i,1,\gamma} - x_{J_{i,0}}|\mathbb{1}_{B_{i,2}}\right) \leqslant \mathbb{E}\left(|X'_{i,\gamma} - y_{J_{i,0}}|\mathbb{1}_{E_{i,0} \leqslant \gamma \|\lambda\|_{\infty}}\right) \leqslant c'\gamma^{3/2}.$$

Case 3. We bound

$$\mathbb{E}\left(\rho(X_{i,1},\overline{X}_{i,\gamma})\mathbb{1}_{B_{i,3}}\right) \leqslant \frac{1}{a}\mathbb{P}\left(B_{i,3}\right)$$
$$\leqslant \frac{1}{a}\mathbb{P}\left(\int_{0}^{\gamma}\lambda(\hat{X}_{i,0,s})\mathrm{d}s \wedge \left(\gamma\lambda(\tilde{X}_{i,0,\gamma})\right) \leqslant E_{i,0} \leqslant \int_{0}^{\gamma}\lambda(\hat{X}_{i,0,s})\mathrm{d}s \vee \left(\gamma\lambda(\tilde{X}_{i,0,\gamma})\right)\right)$$

174 CHAPTER 6. Particle approximation for the quasi-stationary distribution of a diffusion.

$$= \frac{1}{a} \mathbb{E} \left( \left| \exp \left( -\int_0^{\gamma} \lambda(\hat{X}_{i,0,s}) \mathrm{d}s \right) - \exp \left( -\gamma \lambda(\tilde{X}_{i,0,\gamma}) \right) \right| \right) \\ \leqslant \frac{1}{a} \mathbb{E} \left( \left| \int_0^{\gamma} \lambda(\hat{X}_{i,0,s}) \mathrm{d}s - \gamma \lambda(\tilde{X}_{i,0,\gamma}) \right| \right) \,.$$

Now,

$$\left| \int_0^\gamma \lambda(\hat{X}_{i,0,s}) \mathrm{d}s - \gamma \lambda(\tilde{X}_{i,0,\gamma}) \right| \leq L_\lambda \left( \int_0^\gamma |\hat{X}_{i,0,s} - x_i| \mathrm{d}s + \gamma |x_i - \tilde{X}_{i,0,\gamma}| \right).$$

Using (6.11) together with (6.12) yields

$$\mathbb{E}\left(\rho\left(X_{i,1},\overline{X}_{i,\gamma}\right)\mathbb{1}_{B_{i,3}}\right) \leqslant c_3\gamma^{3/2}.$$

Case 4. We bound

$$\mathbb{E}\left(\rho\left(X_{i,1}, \overline{X}_{i,\gamma}\right) \mathbb{1}_{B_{i,4}}\right) \leqslant \frac{1}{a} \mathbb{P}\left(B_{i,4}\right) \\
\leqslant \mathbb{P}\left(E_{i,0} \lor E_{i,1} \leqslant \gamma \|\lambda\|_{\infty}\right) + \mathbb{P}\left(E_{i,0} \lor E_{J_{i,0},0} \leqslant \gamma \|\lambda\|_{\infty}\right) \\
\leqslant 2\left(1 - e^{-\gamma \|\lambda\|_{\infty}}\right)^{2} \leqslant 2\gamma^{2} \|\lambda\|_{\infty}^{2}.$$

**Proposition 6.2.5.** There exist  $C_5 > 0$  such that for all  $N \in \mathbb{N}$ ,  $\gamma \in (0, \gamma_0]$  and  $\eta_0 \in \mathcal{P}(\mathbb{T}^d)$ ,

$$\mathcal{W}_{\rho_N}\left(\mu R_{N,\gamma}^m, \mu P_{N,m\gamma}\right) \leqslant \sqrt{\gamma} N C_5 \gamma \sum_{s=1}^m (1 - \gamma \kappa)^{s-1}$$

As for Proposition 6.2.4, when  $\kappa > 0$ ,  $\gamma \sum_{s=1}^{m} (1 - \gamma \kappa)^{s-1} \leq 1/\kappa$  so that (6.8) and (6.9) yield uniform in time estimates. On the contrary, when  $\kappa < 0$ , the estimates are exponentially bad in  $t = m\gamma$ .

*Proof.* The proof is similar to Proposition 6.2.4. Denoting  $\mu_m = \mu R^m$  and  $\nu_m = \mu P_{m\gamma}$ , from the triangular inequality, Proposition 6.2.3 and Lemma 6.2.2,

$$r_m := \mathcal{W}_{\rho_N} \left( \mu_m, \nu_m \right) \leqslant \mathcal{W}_{\rho_N} \left( \mu_m, \nu_{m-1} R \right) + \mathcal{W}_{\rho_N} \left( \nu_{m-1} R, \nu_{m-1} P_{\gamma} \right)$$
$$\leqslant (1 - \gamma \kappa) r_{m-1} + N C_4 \gamma^{3/2},$$

and an induction concludes.

### 6.2.5 Conclusion

In this section we use the notations of the previous ones, in particular  $\kappa$  is given by (6.5) and the constants  $C_2$ ,  $C_3$  and  $C_5$  are those of Propositions 6.2.4 and 6.2.5. We can now gather all these previous results.

Letting either  $\gamma$  vanish or N go to infinity in Proposition 6.2.3, we obtain long-time convergence for, respectively, the non-homogeneous self-interacting Markov chain  $(Y_k)_{k\in\mathbb{N}}$  introduced in Section 6.1.2 and the continuous-time Markov chain  $(\overline{\mathbf{X}}_t)_{t\geq 0}$  defined in Section 6.2.4.

**Corollary 6.2.2.** Let  $(\eta_n)_{n \in \mathbb{N}}$  be such as defined in Section 6.1.2, and  $(\tilde{\eta}_n)_{n \in \mathbb{N}}$  be similarly defined but with a different initial distribution  $\tilde{\eta}_0 \in \mathcal{P}(\mathbb{T}^d)$ . For all  $m \in \mathbb{N}$  and all  $\gamma \in (0, \gamma_0]$ ,

$$\mathcal{W}_{\rho}(\eta_m, \tilde{\eta}_m) \leq (1 - \gamma \kappa)^m \mathcal{W}_{\rho}(\eta_0, \tilde{\eta}_0).$$

**Corollary 6.2.3.** For all  $N \in \mathbb{N}_*$ ,  $t \ge 0$  and  $\mu, \nu \in \mathcal{P}(\mathbb{T}^{dN})$ ,

$$\mathcal{W}_{\rho_N}\left(\mu P_{N,t},\nu P_{N,t}\right) \leqslant e^{-\kappa t} \mathcal{W}_{\rho_N}\left(\mu,\nu\right)$$

Proof of Corollary 6.2.2. The proof is based on the simple equality: For all  $N \in \mathbb{N}$  and  $\mu, \nu \in \mathcal{P}(\mathbb{T}^d)$ ,

$$\mathcal{W}_{\rho_N}\left(\mu^{\otimes N}, \nu^{\otimes N}\right) = N \mathcal{W}_{\rho}\left(\mu, \nu\right).$$
(6.14)

Indeed, by considering N independent couplings  $(X_i, Y_i)_{i \in [\![1,N]\!]}$ ,

$$\mathcal{W}_{\rho_N}\left(\mu^{\otimes N},\nu^{\otimes N}\right) \leqslant \mathbb{E}\left(\rho_N(\mathbf{X},\mathbf{Y})\right) = \sum_{i=1}^N \mathbb{E}\left(\rho(X_i,Y_i)\right) = N\mathcal{W}_{\rho}\left(\mu,\nu\right).$$

Conversely, if  $(\mathbf{X}, \mathbf{Y})$  is an optimal coupling of  $\mu^{\otimes N}$  and  $\nu^{\otimes N}$ , then

$$\mathcal{W}_{\rho}\left(\mu,\nu\right) \leqslant \mathbb{E}\left(\rho(X_{1},Y_{1})\right) = \frac{1}{N}\mathbb{E}\left(\rho_{N}(\mathbf{X},\mathbf{Y})\right) = \frac{1}{N}\mathcal{W}_{\rho}\left(\mu^{\otimes N},\nu^{\otimes N}\right)$$

By the triangular inequality,

$$\mathcal{W}_{\rho}\left(\eta_{m}^{\otimes N}, \tilde{\eta}_{m}^{\otimes N}\right) \leq \mathcal{W}_{\rho}\left(\eta_{m}^{\otimes N}, \eta_{0}^{\otimes N}R^{m}\right) + \mathcal{W}_{\rho}\left(\eta_{0}^{\otimes N}R^{m}, \tilde{\eta}_{0}^{\otimes N}R^{m}\right) + \mathcal{W}_{\rho}\left(\tilde{\eta}_{0}^{\otimes N}R^{m}, \tilde{\eta}_{m}^{\otimes N}\right)$$
$$\leq (1 - \gamma\kappa)^{m} \mathcal{W}_{\rho}(\eta_{0}^{\otimes N}, \tilde{\eta}_{0}^{\otimes N}) + 2C_{2}N\alpha(N)\gamma\sum_{s=1}^{m} (1 - \gamma\kappa)^{s-1}.$$

where we applied Propositions 6.2.3 and 6.2.4. Using the equality 6.14, dividing by N and letting N go to infinity concludes the proof of Corollary 6.2.2.

Remark that the beginning of the proof also applies for  $\mu, \nu \in \mathcal{P}(\mathbb{T}^{dN})$  that are exchangeable (i.e. invariant by any permutation of the *d*-dimensional coordinates), in which case, denoting,  $\mu^{(1)}$  and  $\nu^{(1)}$  their *d*-dimensional marginals, we get that

$$\mathcal{W}_{\rho}\left(\mu^{(1)},\nu^{(1)}\right) \leqslant \frac{1}{N}\mathcal{W}_{\rho_{N}}\left(\mu,\nu\right).$$

*Proof of Corollary 6.2.3.* Similarly to the previous proof, Corollary 6.2.3 is a direct consequence of Propositions 6.2.3 and 6.2.5, letting m go to infinity at a fixed t and N in

$$\mathcal{W}_{\rho_N} \left( \mu P_{N,t}, \nu P_{N,t} \right) \leqslant \mathcal{W}_{\rho_N} \left( \mu P_{N,t}, \mu R_{N,t/m}^m \right) + \mathcal{W}_{\rho_N} \left( \mu R_{N,t/m}^m, \nu R_{N,t/m}^m \right) \\ + \mathcal{W}_{\rho_N} \left( \nu R_{N,t/m}^m, \nu P_{N,t} \right) .$$

We now turn to the continuous-time limit of the non-linear chain  $(Y_k)_{k \in \mathbb{N}}$ .

**Corollary 6.2.4.** There exists  $C_6 > 0$  such that for all  $\eta_0 \in \mathcal{P}(\mathbb{T}^d)$  and all  $\gamma \in (0, \gamma_0]$ , if  $(\eta_n)_{n \in \mathbb{N}}$  is such as defined in Section 6.1.2, and  $(\overline{\eta}_t)_{t \geq 0}$  is such as defined in Section 6.2.4 (with  $\overline{\eta}_0 = \eta_0$ ),

then

$$\mathcal{W}_1\left(\eta_1, \overline{\eta}_\gamma\right) \leqslant C_6 \gamma^{3/2},$$

and for all  $m \ge 1$ ,

$$\mathcal{W}_{\rho}\left(\eta_{m},\overline{\eta}_{m\gamma}\right) \leqslant \sqrt{\gamma}C_{6}\gamma\sum_{s=1}^{m}(1-\gamma\kappa)^{s-1}$$

*Proof.* For the first inequality, we could follow the proof of Lemma 6.2.2, but, using the notations of the introduction, we will rather use the fact that

$$\eta_1 = \mathcal{L}aw\left(\tilde{Z}_1 \mid \tilde{T} > \gamma\right), \qquad \overline{\eta}_{\gamma} = \mathcal{L}aw\left(Z_{\gamma} \mid T > \gamma\right),$$

where the gaussian variable  $G_0$  in (6.3) is equal to  $B_{\gamma}/\sqrt{\gamma}$  where  $(B_t)_{t\geq 0}$  is the Brownian motion involved in (6.1), and T and  $\tilde{T}$  are defined with the same  $E \sim \mathcal{E}(1)$ . Recall the estimate (6.13) for the error from an Euler scheme to its initial diffusion. Then we bound

$$\mathbb{E}\left(|\tilde{Z}_1 - Z_{\gamma}| \mid T > \gamma, \ \tilde{T} > \gamma\right) \leqslant \left(\mathbb{P}\left(T > \gamma, \ \tilde{T} > \gamma\right)\right)^{-1} \mathbb{E}\left(|\tilde{Z}_1 - Z_{\gamma}|\right) \\
\leqslant \left(1 - e^{-\gamma_0 \|\lambda\|_{\infty}}\right)^{-1} c \gamma^{3/2},$$

which concludes the first part of the corollary.

For the second part, denoting  $r_m = \mathcal{W}_{\rho}(\eta_m, \overline{\eta}_{m\gamma})$ , we bound

$$r_m \leqslant \mathcal{W}_{\rho} \left( \eta_m, \overline{\eta}_{m-1} Q_{\overline{\eta}_{\gamma(m-1)}} \right) + \mathcal{W}_{\rho} \left( \overline{\eta}_{m-1} Q_{\overline{\eta}_{m-1}}, \overline{\eta}_{m\gamma} \right)$$
  
$$\leqslant (1 - \gamma \kappa) r_{m-1} + C_6 \gamma^{3/2} ,$$

where we used the first part of the corollary and Corollary 6.2.2. An induction concludes.  $\Box$ 

We can now prove propagation of chaos results for the continuous-time process:

**Corollary 6.2.5.** For all  $N \in \mathbb{N}$ ,  $k \in [\![1, N]\!]$  and all  $t \ge 0$ , if  $(\overline{\mathbf{X}}_t)_{t\ge 0}$  is a Markov process with initial distribution  $\eta_0^{\otimes N}$  associated to the semigroup  $(P_{N,t})_{t\ge 0}$  then, first,

$$\mathcal{W}_{\rho_k}\left(\mathcal{L}aw(\overline{X}_{1,t},\ldots,\overline{X}_{k,t}),\overline{\eta}_t^{\otimes k}\right) \leqslant C_2k\alpha(N)\int_0^t e^{-\kappa s}ds,$$

and second,

$$\mathbb{E}\left(\mathcal{W}_{\rho}\left(\pi(\overline{\mathbf{X}}_{t}),\overline{\eta}_{t}\right)\right) \leqslant C_{3}\alpha(N)\left(1+\int_{0}^{t}e^{-\kappa s}\,ds\right)\,.$$

*Proof.* As shown in the proof of Proposition 6.2.4, if  $(\mathbf{X}, \mathbf{Y})$  is an optimal coupling of  $\mu$  and  $\nu$ ,

$$\mathbb{E}\left(\mathcal{W}_{\rho}\left(\pi(\mathbf{X}), \pi(\mathbf{Y})\right)\right) \leqslant \frac{1}{N} \mathcal{W}_{\rho_{N}}(\mu, \nu)$$

Thus, considering a time step  $\gamma = t/m, m \in \mathbb{N}$ , we decompose

$$\mathcal{W}_{\rho}\left(\pi(\overline{\mathbf{X}}_{t}),\overline{\eta}_{t}\right) \leqslant \mathcal{W}_{\rho}\left(\pi(\overline{\mathbf{X}}_{t}),\pi(\mathbf{X}_{m})\right) + \mathcal{W}_{\rho}\left(\pi(\mathbf{X}_{m}),\eta_{m}\right) + \mathcal{W}_{\rho}\left(\eta_{m},\overline{\eta}_{t}\right) \,,$$

take the expectation, apply Propositions 6.2.4 and 6.2.5 and Corollary 6.2.4 and let m go to

infinity. This proves the second point, and the proof of the first one is similar, with Corollary 6.2.1.

Up to now, we have sent either N or  $\gamma$  to their limit. When  $\kappa > 0$ , if we let  $t = m\gamma$  go to infinity at fixed N and  $\gamma$ , we recover results on the equilibria of the processes. Indeed, note that Corollary (6.2.2) together with the Banach fixed-point theorem imply that  $n \mapsto \eta_n$  admits a limit which is independent from  $\eta_0$ . Together with Proposition 6.1.1, this is the unique QSD of the Markov chain (6.3). Denote it  $\nu_{\gamma}$ . Similarly, Proposition 6.2.3 implies that  $R_{N,\gamma}$  admits a unique invariant measure. Denote it  $\mu_{\infty,N,\gamma}$ , and  $\mu_{\infty,N,\gamma}^{(k)}$  its first kd-dimensional marginal for  $k \in [\![1,N]\!]$  (i.e. the law of  $(X_1,\ldots,X_k)$  if  $\mathbf{X} \sim \mu_{\infty,N,\gamma}$ ). Third, Corollary 6.2.3 implies that  $(P_{N,t})_{t\geq 0}$  admits a unique invariant measure  $\overline{\mu}_{\infty,N}$ .

**Corollary 6.2.6.** If  $\kappa > 0$ , then for all  $N \in \mathbb{N}$  and  $\gamma \in (0, \gamma_0]$ 

$$\mathcal{W}_{\rho_N}\left(\mu_{\infty,N,\gamma},\overline{\mu}_{\infty,N}\right) \leqslant \sqrt{\gamma}N\kappa^{-1}C_5,$$

**Corollary 6.2.7.** If  $\kappa > 0$ , then for all  $N \in \mathbb{N}$ ,  $k \in [\![1, N]\!]$  and  $\gamma \in (0, \gamma_0]$ , first,

$$\mathcal{W}_{\rho_k}\left(\mu_{\infty,N,\gamma}^{(k)},\nu_{\gamma}^{\otimes k}\right) \leqslant \kappa^{-1}C_2k\alpha(N),$$

and second,

$$\mathbb{E}_{\mu_{\infty,N,\gamma}}\left(\mathcal{W}_{\rho}\left(\pi(\mathbf{X}),\nu_{\gamma}\right)\right) \leqslant \kappa^{-1}C_{3}\alpha(N).$$

Proofs of Corollaries 6.2.6 and 6.2.7. Considering any  $\eta_0 \in \mathcal{P}(\mathbb{T}^d)$  and  $m \in \mathbb{N}$ ,

$$\mathcal{W}_{\rho_{N}}\left(\mu_{\infty,N,\gamma},\overline{\mu}_{\infty,N}\right) \leqslant \mathcal{W}_{\rho_{N}}\left(\mu_{\infty,N,\gamma},\eta_{0}^{\otimes N}R^{m}\right) + \mathcal{W}_{\rho_{N}}\left(\eta_{0}^{\otimes N}R^{m},\eta_{0}^{\otimes N}P_{\gamma m}\right) \\ + \mathcal{W}_{\rho_{N}}\left(\eta_{0}^{\otimes N}P_{\gamma m},\overline{\mu}_{\infty,N}\right)$$

Apply Proposition 6.2.3 with  $\mu = \mu_{\infty,N,\gamma}$  and  $\nu = \eta_0^{\otimes N}$ , Corollary 6.2.3 with the same  $\nu$  and with  $\mu = \overline{\mu}_{\infty,N}$ , and Proposition 6.2.5. Letting *m* go to infinity concludes the proof of Corollary 6.2.6. The proof of Corollary 6.2.7 is similar (based on Proposition 6.2.4 and Corollary 6.2.1, like Corollary 6.2.5).

Next, we can send two parameters to their limit. Sending N to infinity and  $\gamma$  to zero, we get the long time convergence of the non-linear process  $(\overline{Y}_t)_{t\geq 0}$  introduced in Section 6.2.4 (or, equivalently, of the process Z solving (6.1) conditioned not to be dead):

**Corollary 6.2.8.** Let  $(\overline{\eta}_t)_{t\geq 0}$  be such as defined in Section 6.1.2, and  $(\hat{\eta}_t)_{t\geq 0}$  be similarly defined but with a different initial distribution  $\hat{\eta}_0 \in \mathcal{P}(\mathbb{T}^d)$ . For all  $t \geq 0$ ,

$$\mathcal{W}_{\rho}\left(\overline{\eta}_{t},\hat{\eta}_{t}
ight) \leqslant e^{-\kappa t}\mathcal{W}_{\rho}(\overline{\eta}_{0},\hat{\eta}_{0}).$$

*Proof.* Thanks to Corollary 6.2.4, let  $\gamma = t/m$  vanish in Corollary 6.2.2.

In particular, if  $\hat{\eta}_0$  is the QSD  $\nu_*$ , by definition,  $\hat{\eta}_t = \nu_*$  for all  $t \ge 0$ , so that Corollary 6.2.8 yields the uniqueness of the QSD and the exponential convergence of  $\mathcal{L}aw(Z_t \mid T > t)$  toward  $\nu_*$  (which is a result in the spirit of [32, 36, 47, 7]).

Now, at a fixed  $\gamma > 0$ , letting t and N go to infinity, we obtain an error bound between the QSD  $\nu_*$  of the continus process (6.1) and the QSD  $\nu_{\gamma}$  of the discrete scheme.

**Corollary 6.2.9.** If  $\kappa > 0$ , then for all  $\gamma \in (0, \gamma_0]$ 

$$\mathcal{W}_{\rho}\left(\nu_{\gamma},\nu_{*}\right) \leqslant \sqrt{\gamma}\kappa^{-1}C_{6},$$

*Proof.* Thanks to Corollaries 6.2.2 and 6.2.8 (applied with one of the initial condition being the equilibrium), let m go to infinity in Corollary 6.2.4.

Finally, letting  $\gamma$  vanish and t go to infinity at a fixed  $N \in \mathbb{N}$ , we obtain a propagation of chaos result at stationarity (as established first in [5], and more recently with a CLT in [108] in the case of a finite state space) for the continuous time system of interacting particle  $(\overline{\mathbf{X}}_t)_{t\geq 0}$  introduced in Section 6.2.4.

**Corollary 6.2.10.** If  $\kappa > 0$  and if  $\overline{\mathbf{X}}$  is a random variable with law  $\overline{\mu}_{\infty,N}$ , then for all  $N \in \mathbb{N}$  and  $k \in [\![1,N]\!]$ ,

$$\mathcal{W}_{\rho_N}\left(\mathcal{L}aw(X_1,\ldots,X_k),\nu_*^{\otimes k}\right) \leqslant \kappa^{-1}C_2k\alpha(N),$$

and second,

$$\mathbb{E}\left(\mathcal{W}_{\rho}\left(\pi(\overline{\mathbf{X}}),\nu_{*}\right)\right) \leqslant C_{3}\alpha(N)\left(1+\kappa^{-1}\right).$$

*Proof.* The proof is similar to Corollary 6.2.5, letting t go to infinity in Corollary 6.2.5 thanks to Corollaries 6.2.3 and 6.2.8.  $\Box$ 

Remark that our results at stationarity (Corollaries 6.2.6, 6.2.7, 6.2.9 and 6.2.10) all require the perturbative condition  $\kappa > 0$ . Yet, propagation of chaos at stationarity for the continuous time process follows from the works [5, 108, 46, 44, 149] in a much broader (non-perturbative) framework (and, although it doesn't seem to have been studied yet, the situation should be similar for error bounds in  $\gamma$  rather than N). As previously discussed, error bounds on N (and possibly  $\gamma$ ) that are uniform in time can be obtained thanks to the long-time convergence of the limit ( $N = +\infty$ ) non-linear process. In our case, when  $\kappa > 0$ , this long-time convergence follows from the (uniform in N) long-time convergence of the particle system (whether it is possible to obtain the latter from the former is unclear), but it holds in more general cases (see [32, 36, 47, 7] and references within) and, in those cases, results similar to Corollaries 6.2.6, 6.2.7, 6.2.9 and 6.2.10 should hold. The question of establishing propagation of chaos or discretization error bounds at stationarity in a more general case (i.e. without the condition  $\kappa > 0$ ) is out of the scope of the present work.

All our results are summarised in Figure 6.1.

Finally, we detail the proof of our main result.

Proof of Theorem 6.1.2. For  $\eta_0 \in \mathcal{P}(\mathbb{T}^d)$ , let  $(\mathbf{X}, \mathbf{Y})$  be an optimal coupling of  $\mu_0 R^{\lfloor t/\gamma \rfloor}$  and  $\eta_0^{\otimes N} R^{\lfloor t/\gamma \rfloor}$ . As in the proof of Proposition 6.2.4,

$$\mathbb{E}\left(\mathcal{W}_{\rho}\left(\pi(\mathbf{X}), \pi(\mathbf{Y})\right)\right) \leqslant \frac{1}{N} \mathcal{W}_{\rho_{N}}\left(\mu R^{\lfloor t/\gamma \rfloor}, \eta_{0}^{\otimes N} R^{\lfloor t/\gamma \rfloor}\right) \leqslant a e^{-\kappa(t-\gamma_{0})},$$

where we used Proposition 6.2.3 and the fact that  $\rho_N(x, y) \leq Na$  for all  $x, y \in \mathbb{T}^{dN}$ . Then, by the triangular inequality,

$$\mathcal{W}_{\rho}\left(\pi(\mathbf{Y}),\nu_{*}\right) \leqslant \mathcal{W}_{\rho}\left(\pi(\mathbf{Y}),\eta_{|t/\gamma|}\right) + \mathcal{W}_{\rho}\left(\eta_{|t/\gamma|},\nu_{\gamma}\right) + \mathcal{W}_{\rho}\left(\nu_{\gamma},\nu_{*}\right) + \mathcal{W}_{\rho}\left(\nu_{*},\nu_{*}\right) + \mathcal{W}$$



Figure 6.1: Summary of the different results. The label of each arrow indicates the corollary or proposition where the corresponding quantitative convergence is stated. Vertical, horizontal and diagonal arrows correspond respectively to t,  $\gamma$  and N going to their limit.

Taking the expectation, applying Proposition 6.2.4 and Corollaries 6.2.2 (applied with  $\tilde{\eta}_0 = \nu_{\gamma}$ ) and 6.2.9, the boundedness of  $\rho$  and the equivalence of  $W_{\rho}$  and  $W_1$  concludes.

180 CHAPTER 6. Particle approximation for the quasi-stationary distribution of a diffusion.

# Chapter 7

# Uniform convergence of the Fleming-Viot process in a hard killing metastable case

# Outline of the current chapter

7.1	Introduction	181
	7.1.1 Main Result	183
7.2	Preliminary results	186
	7.2.1 Coupling of the non-killed diffusion	186
	7.2.2 Lyapunov functional	188
	7.2.3 Particles near the boundary	192
7.3	Proofs of the main theorems	<b>19</b> 4
	7.3.1 Long time convergence	194
	7.3.2 Propagation of chaos	202
7.4	Establishing Assumption 7.1.2	206

**Abstract**: We study the long-time convergence of a Fleming-Viot process, in the case where the underlying process is a metastable diffusion killed when it reaches some level set. Through a coupling argument, we establish the long-time convergence of the Fleming-Viot process toward some stationary measure at an exponential rate independent of N, the size of the system, as well as uniform in time propagation of chaos estimates.

# 7.1 Introduction

Given some open bounded domain  $D \subset \mathbb{R}^d$ , and some potential  $U : \mathbb{R}^d \to \mathbb{R}_+$ , we are interested in the process:

$$dX_t = -\nabla U(X_t)dt + \sqrt{2\varepsilon}dB_t \tag{7.1}$$

with small  $\varepsilon > 0$ , killed when it reaches  $\partial D$  the boundary of D. More precisely, write:

$$\tau_{\partial D} = \inf \left\{ t \ge 0, X_t \notin D \right\}. \tag{7.2}$$

Denote by  $\mathcal{M}^1(D)$  the set of probability measures on D, and  $\mathbb{P}_{\mu}$  the law of the process (7.1), with initial condition  $\mu \in \mathcal{M}^1(D)$ . Then we say that  $\nu \in \mathcal{M}^1(D)$  is a quasi-stationary distribution (QSD) of the process (7.1) if for all  $t \ge 0$ :

$$\mathbb{P}_{\nu}\left(X_t \in \cdot \mid \tau_{\partial D} > t\right) = \nu$$

It is shown in [104] that, under some mild assumptions on U and D, the process (7.1) admits a unique QSD, that we will denote by  $\nu_{\infty}^{\varepsilon}$ . It is also proven that there is convergence for all initial condition of the law of the process conditioned on its survival toward this QSD, namely, for all  $\mu \in \mathcal{M}^1(D)$ ,

$$\mathbb{P}_{\mu}\left(X_t \in \cdot \mid \tau_{\partial D} > t\right) \xrightarrow[t \to +\infty]{} \nu_{\infty}^{\varepsilon}.$$

The fact that the process is killed when it exits a domain is classically referred to as a *hard* killing case, by contrast with the *soft killing* case where the process is killed according to a inhomogeneous Poisson process as in [94].

The present work is concerned with the question of sampling the QSD  $\nu_{\infty}^{\varepsilon}$ . More precisely, in practice, the QSD is approximated by the empirical measure of a system of interacting particles, called a Fleming-Viot (FV) process, at stationarity. This FV process is defined informally as follows: for a given N, let  $X^1, \ldots, X^N$  be N independent diffusions until one of them reaches  $\partial D$ . The diffusion that has been killed then branches onto one of the N-1 remaining ones, chosen uniformly at random. In very general settings, it is known that if the initial condition consists in N independent random variables distributed according to a common law  $\mu$ , then for any time  $t \ge 0$ , we have:

a.s. 
$$\pi^{N}(X_{t}^{1},\ldots,X_{t}^{N}) \xrightarrow[N \to \infty]{weak} \mathbb{P}_{\mu}(X_{t} \in \cdot \mid \tau_{\partial D} > t),$$
 (7.3)

(see Section 1.3.3) where

$$\pi^{N}(x) = \frac{1}{N} \sum_{i=1}^{N} \delta_{x_{i}}$$
(7.4)

stands for the empirical measure of a vector  $(x_1, \ldots, x_N) \in D^N$ . This would simply be the law of large numbers if the particles were independent, which they are not due to the resurrection mechanism. For mean-field interacting particle systems as the FV process, such a convergence is known as a propagation of chaos phenomenon.

Two questions are addressed in this work. First, the long-time relaxation of the FV process toward its invariant measure: a quantitative convergence in the total variation distance sense at a rate independent from N is stated in Theorem 7.1.1. Second, the propagation of chaos: Theorem 7.1.2 gives a quantitative version of (7.3), with a bound uniform in time. Combining both results yields a quantitative estimate for the convergence of the empirical measure of the FV process toward the QSD as  $N, t \to \infty$ .

As detailed below, these results are established under the condition that D is a metastable state for the diffusion (7.1), in the sense that the mixing time of (7.1) within D is shorter than the typical exit time from D. Mathematically speaking, this is reflected by the fact that  $c^*$ , the critical height within D, is smaller than  $U_0$ , the height of the boundary  $\partial D$  (see below for the definition of  $c^*$  and  $U_0$ ), and the temperature  $\varepsilon$  is small enough. This metastable context is typically the one where QSD are of interest, since in that case the (non-conditional) law of the process is close to the QSD for times in intermediary scales between the mixing time and the extinction time. Moreover, it is exactly the context of some algorithms in molecular dynamics, such as the parallel replica algorithm presented in [104], which involves the sampling of the QSD. In fact, for technical reasons, we will work under a stronger condition than simply  $c_* < U_0$  (see Assumption 7.1.3 below), which is also related to the metastability of D. While we haven't succeeded in this endeavor, we think that the proof may possibly be modified to work only with the condition  $c_* < U_0$ , without the additional condition.

The paper is organized as follows. The remainder of this introduction is dedicated to the statement of the main results. Some preliminary properties of the FV process are studied in Section 7.2, and the main theorems are proven in Section 7.3. Finally, we prove in Section 7.4 the technical lemma which involves the additional condition.

#### 7.1.1 Main Result

Define the critical height  $c^* = c^*(U)$  of U as  $c^* = \sup_{x_1, x_2 \in D} c(x_1, x_2)$  with

$$c(x_1, x_2) = \inf \left\{ \max_{0 \le t \le 1} U(\xi(t)) - U(x_1) - U(x_2) \right\},$$

where the infimum runs over  $\{\xi \in \mathcal{C}([0,1], D), \xi(0) = x_1, \xi(1) = x_2\}$ . The critical height  $c^*$  represents the largest energy barrier the process has to cross in order to go from any local minimum to any global one (within D).

The following conditions are enforced throughout all this work.

**Assumption 7.1.1.** •  $D \subset \mathbb{R}^d$  is open, bounded, connected and its boundary is  $C^2$ .

- $U: \mathbb{R}^d \to \mathbb{R}_+$  is smooth on some neighborhood of  $\overline{D}$ .
- $\min_D U = 0.$
- $U|_{\partial D}$  is constant, and

$$U_0 = U(\partial D) > c^*. \tag{7.5}$$

• For  $x \in \partial D$ , denote by n(x) the outward normal to D. For all  $x \in \partial D$ ,

$$n(x) \cdot \nabla U(x) > 0. \tag{7.6}$$

The condition  $\min_D U = 0$  is just a choice of normalisation since the process is unchanged if a constant is added to U. Under Assumption 7.1.1, neglecting sub-exponential terms, for small  $\varepsilon$ , the mixing rate of the non-killed process (7.1) is known to be of order  $e^{c_*/\varepsilon}$  (see [83]) while, according to the theory of Freidlin-Wentzell (see [65]), the exit time  $\tau_{\partial D}$  is of order  $e^{U_0/\varepsilon}$ . As already mentioned, the condition  $U_0 > c^*$  thus describes a difference of timescales between the mixing time and the death time. More precisely, it is known that for any neighborhood  $\mathfrak{B}_1$ of  $\partial D$  and any  $a < U_0$ ,  $\sup_{x \in D \setminus \mathfrak{B}_1} \mathbb{P}_x(\tau_{\partial D} < e^{a/\varepsilon})$  vanishes with  $\varepsilon$ . The assumption that U is constant over  $\partial D$ , which is consistent with the idea that D is a local potential well, could possibly be relaxed, with the condition (7.5) being replaced by  $\inf_{\partial D} U > c^*$ , however this adjustment would introduce additional technical complexities to the construction of a Lyapunov function, see Lemma 7.2.3. We will need an even stronger uniformity in terms of the initial condition. For now, let us state it as an assumption: Assumption 7.1.2. There exist  $a > c^*$  and a neighborhood  $\mathfrak{B}_1$  of  $\partial D$  such that the following holds:

- $\min_{\mathfrak{B}_1} U > a$ ,
- $\min_{x \in \mathfrak{B}_1} |\nabla U(x)| > 0$ ,
- for a fixed Brownian motion (B<sub>t</sub>)<sub>t≥0</sub>, denoting by (X<sup>x</sup><sub>t</sub>)<sub>t≥0</sub> the solution of (7.1) with initial condition x for all x ∈ D and by τ<sub>∂D</sub>(X<sup>x</sup>) its first exit time from D, we have

$$\bar{p}_{\varepsilon} := \mathbb{P}\left(\exists x \in D \setminus \mathfrak{B}_1, \tau_{\partial D}(X^x) < e^{a/\varepsilon}\right) \underset{\varepsilon \to 0}{\longrightarrow} 0.$$
(7.7)

We are able to prove that this is implied by the following condition, which in dimension d > 1 strengthens (7.5):

Assumption 7.1.3. One of the following is satisfied:

- d = 1.
- There exist a neighborhood  $\mathfrak{B}_1$  of  $\partial D$  such that  $\inf_{\mathfrak{B}_1} |\nabla U| > 0$  and

$$c^* < \frac{U_0 - U_1}{2}$$
, where  $U_1 = \min_{\mathfrak{V}_1} U$ .

As stated in Lemma 7.4.1 below, Assumptions 7.1.1 and 7.1.3 together imply Assumption 7.1.2. However we don't think that Assumption 7.1.3 is sharp, and thus we state our main results in terms of Assumption 7.1.2. Notice that Assumption 7.1.3 is always implied by Assumption 7.1.1 when  $c_* = 0$ .

We define the semi-group  $(P_t)_{t\geq 0}$  associated to a Markov process  $(X_t)_{t\geq 0}$  in  $\mathbb{R}^d$  by:

$$P_t f(x) = \mathbb{E}_x(f(X_t))$$

for any bounded measurable function  $f : \mathbb{R}^d \to \mathbb{R}$  and any  $t \ge 0$ , where  $\mathbb{E}_x$  stands for the expectation under  $\mathbb{P}_x = \mathbb{P}_{\delta_x}$ . We denote  $[\![1, N]\!] = \{1, \cdots, N\}$ .

Now, let us define rigorously the FV process, starting from some initial condition  $\mu \in \mathcal{M}^1(D^N)$ . Let  $(I_n^i)_{i \in [\![1,N]\!], n \in \mathbb{N}}$  be a family of independent random variables, where for  $i \in [\![1,N]\!]$ ,  $I_n^i$  is uniform on  $i \in [\![1,N]\!] \setminus \{i\}$ . Let  $(B^i)_{i \in [\![1,N]\!]}$  be N independent Brownian motions, and  $\mathbf{X}_0 = (X_0^1, \ldots, X_0^N)$  be distributed according to  $\mu$ . Define  $\bar{X}^i$  as the solution to:

$$\bar{X}_t^i = X_0^i - \int_0^t \nabla U(\bar{X}_s^i) \mathrm{d}s + \sqrt{2\varepsilon} B_t^i$$

and set

$$\tau_1 = \min_i \inf \left\{ t \ge 0, X_t^i \notin D \right\}.$$

Then, denote by  $i_1$  the index of the particle which exits the domain at time  $\tau_1$ . It is uniquely defined almost surely because, since the hitting time of the boundary has a density on  $\mathbb{R}_+$ , the probability that two particles hit the boundary at the same time is zero (this is true for the Brownian motion, and the general case follows from an application of the Girsanov theorem). For  $i \neq i_1$ ,  $0 \leq t \leq \tau_1$ , or  $i = i_1$  and  $1 \leq t < \tau_1$ , simply let:

$$X_t^i = \bar{X}_t^i$$
 and  $X_{\tau_1}^{i_1} = \bar{X}_{\tau_1}^{I_1^i}$ .

This defines the process between times 0 and  $\tau_1$ . The process is then defined on  $(\tau_1, \infty)$  by induction: if the process is defined up to time  $\tau_{n-1}$ , we define it between time  $\tau_{n-1}$  and  $\tau_n$  in the same way, with  $\mathbf{X}_0$  replaced by  $\mathbf{X}_{\tau_{n-1}}$ ,  $i_1$  by  $i_n$  the index of the particle that hits  $\partial D$  at time  $\tau_n$ , and  $I_1^{i_0}$  by  $I_n^{i_n}$ . Thus,  $(\tau_n)_n$  is the sequence of branching times of the process.

Under Assumption 7.1.1, the FV process  $\mathbf{X} = (X^1, \ldots, X^N)$  is well-defined and does not explode in finite time, meaning that  $\sup_n \tau_n = \infty$  almost surely, see [166, Theorem 2.1]. This defines a Markov process, and we denote by  $P^{N,\varepsilon} = (P_t^{N,\varepsilon})_{t \ge 0}$  the associated semi-group.

A law  $\mu \in \mathcal{M}^1(D^N)$  is said to be exchangeable if it is invariant by any permutation of the particles, i.e.  $(X^{\sigma(i)})_{i \in [\![1,N]\!]} \sim \mu$  if  $(X^i)_{i \in [\![1,N]\!]} \sim \mu$  for all permutations  $\sigma$  of  $[\![1,N]\!]$ . For  $k \in [\![1,N]\!]$ , we denote by  $\mu^k \in \mathcal{M}^1(D^k)$  the marginal law of the k first particles under  $\mu$  (which, for exchangeable laws, is thus the marginal law of any subset of k particles).

Our first main result concerns the long time behavior of the FV process.

**Theorem 7.1.1.** Under Assumptions 7.1.1 and 7.1.2, there exist  $\varepsilon_0, c, C > 0$  such that, for all  $\varepsilon \in (0, \varepsilon_0], N \in \mathbb{N}, t \ge 0$ , setting  $t_{\varepsilon} = e^{a/\varepsilon}$ , the following holds.

1. For all  $\mu, \nu \in \mathcal{M}^1(D^N)$ ,

$$\|\mu P_t^{N,\varepsilon} - \nu P_t^{N,\varepsilon}\|_{TV} \leqslant CN(1-c)^{t/t_{\varepsilon}}.$$

- 2. The semi-group  $P^{N,\varepsilon}$  admits a unique invariant measure  $\nu_{\infty}^{N,\varepsilon}$ , which is exchangeable.
- 3. For all exchangeable  $\mu, \nu \in \mathcal{M}^1(D^N)$ , for all  $k \in [\![1, N]\!]$ ,

$$\|(\mu P_t^N)^k - (\nu P_t^N)^k\|_{TV} \leqslant Ck(1-c)^{t/t_{\varepsilon}}.$$

Our second result is a uniform in time propagation of chaos estimate.

**Theorem 7.1.2.** Under Assumptions 7.1.1 and 7.1.2, there exists  $\varepsilon_0 > 0$  such that for all compact set  $K \subset D$ , all  $\varepsilon \in (0, \varepsilon_0]$ , there exist  $C_{\varepsilon}, \eta_{\varepsilon} > 0$  such that for all  $\mu_0 \in \mathcal{M}^1(D)$  satisfying  $\mu_0(K) \ge 1/2$ , all bounded  $f: D \to \mathbb{R}_+$  and all  $N \in \mathbb{N}$ ,

$$\sup_{t \ge 0} \mathbb{E}\left( \left| \int_D f d\pi^N(\mathbf{X}_t) - \mathbb{E}_{\pi^N(\mathbf{X}_0)} \left( f(X_t) | \tau_{\partial D} > t \right) \right| \right) \leqslant \frac{C_{\varepsilon} \|f\|_{\infty}}{N^{\eta_{\varepsilon}}},$$

where X solves (7.1),  $\tau_{\partial D}$  is defined in (7.2),  $\pi^N$  in (7.4), and **X** is a FV-process with initial condition  $\mu_0^{\otimes N}$ .

In Theorem 7.1.1, the dependency in N of the speed of convergence is the same as for N independent diffusion processes. Moreover, the dependency in  $\varepsilon$  is also sharp. Indeed, in Assumption 7.1.2, we can take a arbitrarily close to  $c^*$ , which means that we get a mixing time smaller than  $e^{(c^*+\delta)/\varepsilon}$  for any  $\delta > 0$ , which is the order of the mixing time for the non-killed process. However, as far as Theorem 7.1.2 is concerned, for independent processes, one would get from the Bienaymé-Chebyshev inequality the explicit rate  $1/\sqrt{N}$ . This is indeed what is proven in [165, Theorem 1] for the FV process, but with a bound that depends on time. In other words, we improve the result of [165] to a uniform in time bound, but at the cost of a loss in the rate in N. Notice that  $\eta_{\varepsilon}$  may be made explicit by carefully following the proofs.

# 7.2 Preliminary results

For any two probability measures  $\mu, \nu$ , we call (X, Y) a coupling of  $\mu$  and  $\nu$  if the law of X (resp. Y) is  $\mu$  (resp.  $\nu$ ). For any distance **d** on a set E (here E = D or  $E = D^N$ ), the associated Kantorovich distance on  $\mathcal{M}^1(E)$ , is defined by

$$W_{\mathbf{d}}(\mu, \nu) = \inf \left\{ \mathbb{E} \left( \mathbf{d}(X, Y) \right), (X, Y) \text{ coupling of } \mu \text{ and } \nu \right\}.$$

We say that (X, Y) is an optimal coupling if  $W_{\mathbf{d}}(\mu, \nu) = \mathbb{E}(\mathbf{d}(X, Y))$ . The existence of an optimal coupling results from [164, Theorem 4.1]. Given a Markov semi-group P, we call a coupling of  $(\mu P_t)_{t\geq 0}$  and  $(\nu P_t)_{t\geq 0}$  a stochastic process  $(X_t, Y_t)_{t\geq 0}$  such that  $(X_t)_{t\geq 0}$  and  $(Y_t)_{t\geq 0}$  are Markov processes of semi-group  $(P_t)_{t\geq 0}$  and initial condition  $\mu$  for X and  $\nu$  for Y. In particular, we have that for such a coupling and all  $t \geq 0$ ,

$$W_{\mathbf{d}}(\mu P_t, \nu P_t) \leq \mathbb{E}\left(\mathbf{d}(X_t, Y_t)\right).$$

We also say that the processes X and Y have coupled at time  $t \ge 0$  if  $X_t = Y_t$ . Finally, in the case where  $\mathbf{d}(x, y) = 2\mathbb{1}_{x \neq y}$ , we recover the total variation distance which we write  $W_{\mathbf{d}}(\mu, \nu) = \|\mu - \nu\|_{TV}$ .

The proof of our theorems relies on the construction of a coupling of  $(\mu P_t^{N,\varepsilon})_{t\geq 0}$  and  $(\nu P_t^{N,\varepsilon})_{t\geq 0}$ . This coupling will yield that  $P_t^{N,\varepsilon}$  is a contraction for some particular distance defined in Section 7.3.

In order to do this, we first need some preliminary results, which is the subject of this section. We start by studying the mixing properties of the non-killed process in Subsection 7.2.1 by embedding D into a torus. In Subsection 7.2.2, We construct a Lyapunov functional for each particle. In Subsection 7.2.3, using the Lyapunov functional, we study the number of particle that may stay near the boundary of the domain.

In the rest of the paper, we fix some  $a \in (c^*, U_0)$  satisfying assumption 7.1.2 and set  $t_{\varepsilon} = e^{a/\varepsilon}$ . Furthermore, bold letters will always denote particle systems, in the sense that **X** can always be written  $\mathbf{X} = (X^1, \ldots, X^N)$  where for all  $1 \leq i \leq N$ ,  $X^i \in D$ .

# 7.2.1 Coupling of the non-killed diffusion

In this section, we show that we are able to couple two diffusions solution of (7.1) on a torus in total variation distance in a time  $t_{\varepsilon}$  with a probability that goes to 1 as  $\varepsilon$  goes to 0, uniformly on D. Since we are studying a process killed at the boundary of D, we are not interested in what the potential might look like outside of D. Consider some torus  $\mathbb{T}^d = (\mathbb{R}/2L\mathbb{Z})^d$ , with L big enough so that as a subset of  $\mathbb{R}^d$  (meaning seeing  $\mathbb{T}^d$  as  $[-L, L[^d)$ ), we have that  $D \subset \mathbb{T}^d$ . Then consider some periodic potential  $\tilde{U} : \mathbb{R}^d \to \mathbb{R}_+$ , equal to U on D as a periodic function, and such that  $c^*(\tilde{U}) = c^*(U)$ , where  $c^*(\tilde{U})$  is defined as  $c^*(U)$  with U replaced by  $\tilde{U}$ . Such a function exists, as shown in [64, Section 4]. We still denote by  $\tilde{U}$  the associated function on  $\mathbb{T}^d$ , and this potential defines a diffusion on  $\mathbb{T}^d$  as:

$$\mathrm{d}\tilde{X}_t = -\nabla \tilde{U}(\tilde{X}_t)\mathrm{d}t + \sqrt{2\varepsilon}\mathrm{d}B_t.$$
(7.8)

We note  $\tilde{P}$  its semi-group. If we see  $\tilde{X}$  as a process in  $\mathbb{R}^d$ , then we have that  $X_t = \tilde{X}_t$  for all  $t \leq \tau_{\partial D}$ , where  $\tau_{\partial D}$  is the death time (7.2).

Now, we construct a coupling for the process  $\tilde{X}$ , for all initial condition  $(x, y) \in (\mathbb{T}^d)^2$ . To do this, we use Sobolev and Poincaré inequalities. The Sobolev inequality is used for ultracontractivity, whereas the Poincaré inequality is used to get an optimal convergence rate for the process (7.8). Let  $\mu_{\varepsilon}$  denote the probability measure on  $\mathbb{T}^d$ :

$$\mu_{\varepsilon}(\mathrm{d}x) = \mathcal{Z}e^{-U(x)/\varepsilon}\mathrm{d}x$$

where  $\mathcal{Z}$  is the normalization constant. Recall those inequalities:

**Lemma 7.2.1.**  $\mu_{\varepsilon}$  satisfies a Poincaré and a Sobolev inequality: there exist p > 2,  $C, \lambda_{\varepsilon} > 0$ , such that

$$\varepsilon \ln (\lambda_{\varepsilon}) \to -c^*$$

as  $\varepsilon \to 0$ , and for all smooth  $f : \mathbb{T}^d \mapsto \mathbb{R}$  with  $\int_{\mathbb{T}^d} f d\mu_{\varepsilon} = 0$ 

$$\lambda_{\varepsilon} \int_{\mathbb{T}^d} f^2 d\mu_{\varepsilon} \leqslant \int_{\mathbb{T}^d} |\nabla f|^2 d\mu_{\varepsilon} \quad (PI),$$
$$\left(\int_{\mathbb{T}^d} f^p d\mu_{\varepsilon}\right)^{\frac{2}{p}} \leqslant C e^{\|\tilde{U}\|_{\infty}/\varepsilon} \left(\int_{\mathbb{T}^d} f^2 d\mu_{\varepsilon} + \int_{\mathbb{T}^d} |\nabla f|^2 d\mu_{\varepsilon}\right) \quad (SI)$$

Moreover, for all t > 0, the law of  $\tilde{X}_t$  with initial condition x has a density  $h_t^{\varepsilon}(x, \cdot)$  with respect to  $\mu_{\varepsilon}$ , and both inequalities together imply the existence of some constant  $\tilde{C} > 0$  such that, for all  $t \ge 1$  and  $\varepsilon > 0$ ,

$$\|h_t^{\varepsilon}(\cdot, \cdot) - 1\|_{\infty} \leqslant \tilde{C} e^{C\varepsilon^{-1} - \lambda_{\varepsilon} t}.$$
(7.9)

*Proof.* The Poincaré inequality, as well as the asymptotic on  $\lambda_{\varepsilon}$ , have been proven in [83]. The uniform measure on  $\mathbb{T}^d$  satisfies a Sobolev inequality, see [6, section 6]. Then we can write:

$$\begin{split} \left(\int_{\mathbb{T}^d} f^p \mathrm{d}\mu_{\varepsilon}\right)^{\frac{2}{p}} &\leqslant \mathcal{Z}^{-\frac{2}{p}} \left(\int_{\mathbb{T}^d} f^p\right)^{\frac{2}{p}} \\ &\leqslant C \mathcal{Z}^{-\frac{2}{p}} \left(\int_{\mathbb{T}^d} f^2 + \int_{\mathbb{T}^d} |\nabla f|^2\right) \\ &\leqslant C \mathcal{Z}^{1-\frac{2}{p}} e^{\|\tilde{U}\|_{\infty}/\varepsilon} \left(\int_{\mathbb{T}^d} f^2 \mathrm{d}\mu_{\varepsilon} + \int_{\mathbb{T}^d} |\nabla f|^2 \mathrm{d}\mu_{\varepsilon}\right). \end{split}$$

 $\mathcal{Z}$  is bounded by the volume of  $\mathbb{T}^d$ , hence  $\mathcal{Z}^{1-\frac{2}{p}}$  is bounded uniformly on  $\varepsilon$  because p > 2 (and thus  $1-\frac{2}{p} > 0$ ). Therefore, we have the Sobolev inequality with the said constant. The last two points are [6, Theorem 6.3.1 and Proposition 6.3.4].

**Lemma 7.2.2.** Recall that  $t_{\varepsilon} = e^{a/\varepsilon}$  for some  $a > c^*$ . Under Assumption 7.1.1, there exists  $\varepsilon_0 > 0$ , such that for all  $0 < \varepsilon < \varepsilon_0$ , there exists  $c_{\varepsilon} > 0$  such that for all  $x, y \in D$ , there exists a coupling  $(\tilde{X}_t, \tilde{Y}_t)_{t \ge 0}$  of  $(\delta_x \tilde{P}_t)_{t \ge 0}$  and  $(\delta_y \tilde{P}_t)_{t \ge 0}$  such that:

$$\mathbb{P}\left(\tilde{X}_{t_{\varepsilon}} = \tilde{Y}_{t_{\varepsilon}}\right) \geqslant c_{\varepsilon}.$$

 $c_{\varepsilon} \rightarrow 1.$ 

Moreover, as  $\varepsilon \to 0$ ,

*Proof.* We start by bounding the total variation distance between the law of  $\tilde{X}_t$  and the equilibrium  $\mu_{\varepsilon}$ . Recall that  $h_t^{\varepsilon}(x, \cdot)$  denotes the density of the law of  $\tilde{X}_t$  with respect to  $\mu_{\varepsilon}$ . For  $x \in D$ ,

using (7.9), we have:

$$\|\delta_x \tilde{P}_t - \mu_{\varepsilon}\|_{TV} = \int_{R^d} |h_t^{\varepsilon} - 1| \mathrm{d}\mu_{\varepsilon} \leqslant \|h_t^{\varepsilon}(\cdot, \cdot) - 1\|_{\infty} \leqslant \tilde{C} e^{\tilde{C}\varepsilon^{-1} - \lambda_{\varepsilon}t}$$

Since  $t_{\varepsilon} = e^{a/\varepsilon}$ , and  $\varepsilon \ln(\lambda_{\varepsilon}) \to -c^*$  as  $\varepsilon \to 0$ , we have that  $\lambda_{\varepsilon} t_{\varepsilon} \ge e^{b/\varepsilon}$ , for  $\varepsilon$  small enough and all  $b < a - c^*$ . Then, we may fix some  $b < a - c^*$  and let:

$$c_{\varepsilon} = 1 - 2\tilde{C}e^{\tilde{C}\varepsilon^{-1} - e^{b/\varepsilon}},$$

so that  $\lim_{\varepsilon \to 0} c_{\varepsilon} = 1$  and for all  $x, y \in D$ ,

$$\|\delta_x \tilde{P}_t - \delta_y \tilde{P}_t\|_{TV} \leqslant \|\delta_x \tilde{P}_t - \mu_\varepsilon\|_{TV} + \|\delta_y \tilde{P}_t - \mu_\varepsilon\|_{TV} \leqslant 1 - c_\varepsilon.$$

The existence of the coupling of the trajectories results from the total variation distance properties, see for example [133, Lemma 9].  $\Box$ 

# 7.2.2 Lyapunov functional

In order to show the convergence of the particle system, we need first the construction of some Lyapunov function for each particle, that is to say some function of one particle that decreases in average over time, as long as it starts large enough. This is the part where we need U to be constant on the boundary of D. This part relies on a result first shown in [71], similar to Ito's formula. The process (7.1) has a generator  $\mathcal{L}$  defined for all smooth function  $f : \mathbb{R}^d \to \mathbb{R}$  with compact support as:

$$\mathcal{L}f = \varepsilon \Delta f - \nabla U \cdot \nabla f. \tag{7.10}$$

For a smooth function  $f: D^N \to \mathbb{R}_+$ , and  $1 \leq i \leq N$ , write  $\mathcal{L}_{x_i}$  for the generator  $\mathcal{L}$ , acting only on the i-th variable:

$$\mathcal{L}_{x_i}f = \sum_{j=1}^d \varepsilon \partial_{(x_i)_j}^2 f - \partial_j U(x_i) \partial_{(x_i)_j} f.$$

Write as well  $(\tau_n^i)_n$  for the sequence of death times of particle *i*:

$$\tau_0^i = 0, \quad \tau_{n+1}^i = \inf\left\{ t > \tau_n^i, X_{t-}^i \in \partial D \right\},$$
(7.11)

and  $R_i$  for the point process corresponding to the jumps of this particle:

$$R_i(t) = \sum_{n=1}^{\infty} \mathbb{1}_{\tau_n^i \leqslant t}.$$

For all  $\mathbf{x} \in D^N$ ,  $1 \leq i \neq j \leq N$ , write:

$$x_k^{ij} = \begin{cases} x_k \text{ if } k \neq i \\ x_j \text{ else} \end{cases}$$

**Proposition 7.2.1** ([71], Proposition 1). Let  $N \in \mathbb{N}$ ,  $f \in \mathcal{C}^0(\overline{D}^N) \cap \mathcal{C}^\infty(D^N)$ . Denote by

$$\mathbf{R}(f)(t) = \sum_{i=1}^{N} \frac{1}{N-1} \sum_{j \neq i} \int_{0}^{t} \left( f(\mathbf{X}_{s-}^{ij}) - f(\mathbf{X}_{s-}) \right) dR_{i}(s)$$

the finite variation process of the jump part, and

$$Q(f)(t) = \int_0^t \sum_{i=1}^N \mathcal{L}_{x_i} f(\mathbf{X}_s) ds$$

the finite variation process of the diffusion part. Then there exists a martingale  $\mathcal{M}(f)$  such that almost surely for all  $t \ge 0$ :

$$f(\mathbf{X}_t) - f(\mathbf{X}_0) = \mathbf{R}(f)(t) + Q(f)(t) + \mathcal{M}(f)_t.$$

Moreover,

$$\mathbf{R}(f)(t) + \mathcal{M}(f)_t = \sum_{i=1}^N \int_0^t \nabla_{x_i} f(\mathbf{X}_s) \cdot dB_s^i + \sum_{n, \tau_n \leqslant t} f(\mathbf{X}_{\tau_n}) - f(\mathbf{X}_{\tau_n-}),$$

where the  $(B^i)$ 's are the Brownian motion used in the definition of the FV process, and the  $(\tau_n)$  are the death times.

This decomposition will allow us to prove the existence of the Lyapunov functional. The idea is the following: the function V will be equal to U at the center of the domain, and near its boundary, V will be such that  $\nabla V$  is proportional to  $\nabla U$ , but with  $|\nabla V|$  possibly greater than  $|\nabla U|$ . Since  $n(x) \cdot \nabla U(x) > 0$  on the boundary, we will also have  $n(x) \cdot \nabla V(x) > 0$ , which means that, near the boundary, V tends to decrease along (7.1). If in addition V is maximal at the boundary, then it can only decrease at a jump time.

**Lemma 7.2.3.** Under Assumption 7.1.1, there exists  $C_1 > 0$  such that for all  $V_0 > 0$  with

$$V_0 \ge 4 \sup_D U + 1,$$

there exist  $\varepsilon_0 > 0$  and a smooth function  $V : \overline{D} \mapsto \mathbb{R}_+$  such that V is constant equal to  $V_0$  on  $\partial D$ ,  $\sup_D V = V_0$ ,  $4C_1 < V_0$ , and for all  $N \in \mathbb{N}$ ,  $0 < \varepsilon < \varepsilon_0$ ,  $\boldsymbol{x} \in D^N$ , and  $1 \leq i \leq N$ , we have:

$$\mathbb{E}_{\boldsymbol{x}}\left(V\left(X_{t_{\varepsilon}}^{i}\right)\right) \leqslant \gamma_{\varepsilon}V(x_{i}) + C_{1}(1-\gamma_{\varepsilon})$$

$$(7.12)$$

where  $\gamma_{\varepsilon} \in (0,1)$  is independent from  $\boldsymbol{x}$  and vanishes as  $\varepsilon \to 0$ .

The value  $V_0$  of V on the boundary is a fixed parameter that will be chosen in Section 7.3.

*Proof.* Fix some  $V_0 \ge 4 \sup_D U + 1 =: 4\theta + 1$ . Under Assumption 7.1.1, and in particular because of condition (7.6), there exists some set  $F \subset D$  such that  $\mathbb{R}^d \setminus F$  is a neighborhood of  $\partial D$ , there are no critical points of U on  $\overline{D \setminus F}$  (the closure of  $D \setminus F$ ), and

$$\sup_{D\setminus F} U \leqslant U_0.$$

Write

$$U_{min} = \min_{D \setminus F} U$$
,  $\zeta = \inf_{x \in D \setminus F} |\nabla U(x)|.$ 

Notice that  $\zeta > 0$  since  $|\nabla U|$  is continuous and positive over the compact set  $\overline{D \setminus F}$ . Let  $f : [0, U_0] \mapsto \mathbb{R}$  be some smooth function satisfying the following conditions:

• for all  $u \leq U_{min}, f(u) = u$ ,

- $f(U_0) = V_0,$
- $\min_{U_{min} \leq u \leq U_0} f'(u) > 0.$

Then, set V(x) = U(x) for all  $x \in F$ , and V(x) = f(U(x)) for all  $x \notin F$ . The function V is smooth, constant equal to  $V_0 = f(U_0)$  on  $\partial D$ , bounded above by  $V_0$ , and we have for all  $x \in D \setminus F$ :

$$\mathcal{L}V = f'(U)\left(\varepsilon\Delta U - |\nabla U|^2\right) + \varepsilon f''(U)|\nabla U|^2$$

Recall that  $|\nabla U|$  is positive on  $D \setminus F$ . We consider  $\varepsilon_0$  small enough so that on  $D \setminus F$  we have, for all  $\varepsilon < \varepsilon_0$ :

$$\varepsilon f'(U)|\Delta U| + \varepsilon |f''(U)||\nabla U| \leq 1/2f'(U)|\nabla U|^2$$

We then have for all  $x \in D \setminus F$ :

$$\mathcal{L}V(x) \leqslant -\omega V(x)$$

where

$$\omega = \frac{\zeta^2}{2V_0} \min_{U_1 \leqslant u \leqslant U_0} f'(u)$$

is independent of  $\varepsilon$ . Now, for  $x \in F$ :

$$\mathcal{L}V(x) \leqslant \varepsilon \sup_{D} |\Delta U| + \omega \theta - \omega V(x).$$

This inequality is thus true for all  $x \in D$ . Hence, we may plug it into the formula of Proposition 7.2.1 with  $f(\mathbf{x}) = V(x_i)$ . Recall the definition of  $(\tau_n^i)_{n \in \mathbb{N}}$  from (7.11). For  $n \in \mathbb{N}$ , using the fact that V is maximal on the boundary of D,  $\mathbf{R}(V) \leq 0$ , and hence for all  $0 \leq s \leq t$ :

$$\mathbb{E}\left(V\left(X_{t\wedge\tau_{n}^{i}}^{i}\right)\right) - \mathbb{E}\left(V\left(X_{s\wedge\tau_{n}^{i}}^{i}\right)\right) \leqslant \int_{s}^{t} \left(-\omega\mathbb{E}\left(V\left(X_{u\wedge\tau_{n}^{i}}^{i}\right)\right) + \varepsilon \|\Delta U\|_{\infty} + \omega\theta\right) \mathrm{d}u.$$

Writing  $g(t) = \mathbb{E}\left(V\left(X_{t \wedge \tau_n^i}^i\right)\right) - \theta - \varepsilon \|\Delta U\|_{\infty}/\omega$ , we have for  $t \ge 0$  and h > 0:

$$g(t+h) - g(t) \leqslant -\omega \int_{t}^{t+h} g(u) \mathrm{d}u.$$

We now suppose that g is continuous. Fix  $\eta > 0$  and write:

$$T_{\eta} = \min\left\{t \ge 0, g(t) \ge g(0)e^{\omega t} + \eta\right\}.$$

Suppose that  $T_{\eta} < \infty$  and write:

$$s_{\eta} = \max\left\{ 0 \leqslant s \leqslant T_{\eta}, g(s) \leqslant g(0)e^{\omega s} \right\}.$$

Then by continuity we have:

$$\begin{split} g(0)e^{-\omega T_{\eta}} + \eta - g(0)e^{-\omega s_{\eta}} &= g(T_{\eta}) - g(s_{\eta}) \leqslant -\omega \int_{s_{\eta}}^{T_{\eta}} g(u) \mathrm{d}u \\ &< -\omega \int_{s_{\eta}}^{T_{\eta}} g(0)e^{-\omega u} \mathrm{d}u = g(0)e^{-\omega T_{\eta}} - g(0)e^{-\omega s_{\eta}}, \end{split}$$

190

hence necessarily  $T_{\eta} = \infty$ , for all  $\eta > 0$ , and thus for all  $t \ge 0$ ,

$$g(t) \leqslant g(0)e^{-\omega t},$$

and

$$\mathbb{E}\left(V\left(X_{t_{\varepsilon}\wedge\tau_{n}^{i}}^{i}\right)\right)\leqslant\gamma_{\varepsilon}V(x_{i})+\frac{\varepsilon\|\Delta U\|_{\infty}+\omega\theta}{\omega}\left(1-\gamma_{\varepsilon}\right),$$

with  $\gamma_{\varepsilon} = e^{-\omega t_{\varepsilon}}$ . Because V is maximal at the boundary,  $t \mapsto V(X_t^i)$  is lower-semicontinuous. Hence, since  $V \ge 0$  and  $\tau_n^i \to \infty$  as  $n \to \infty$ , using Fatou's lemma,

$$\mathbb{E}\left(V\left(X_{t_{\varepsilon}}^{i}\right)\right) \leqslant \mathbb{E}\left(\liminf_{n \to \infty} V\left(X_{t_{\varepsilon} \wedge \tau_{n}^{i}}^{i}\right)\right)$$
$$\leqslant \liminf_{n \to \infty} \mathbb{E}\left(V\left(X_{t_{\varepsilon} \wedge \tau_{n}^{i}}^{i}\right)\right) \leqslant \gamma_{\varepsilon} V(x_{i}) + \frac{\varepsilon \|\Delta U\|_{\infty} + \omega\theta}{\omega} \left(1 - \gamma_{\varepsilon}\right).$$

Since  $V_0 > 4\theta + 1$ , we may take  $\varepsilon_0$  small enough so that, for all  $\varepsilon < \varepsilon_0$ :

$$\frac{\varepsilon \|\Delta U\|_{\infty} + \omega\theta}{\omega} < 4 \sup_{D} U + 1 \leqslant V_0,$$

and hence the result with  $C_1 = \sup_F U + 1/4$ .

We are left to show that, for all  $n \in \mathbb{N}$ ,  $t \mapsto \mathbb{E}(V(X_{t \wedge \tau_n^i}^i))$  is continuous. Write  $V^i(\mathbf{x}) = V(x_i)$ . Then from Proposition 7.2.1, we get that:

$$\mathbb{E}(V(X_{t\wedge\tau_n^i}^i)) = \mathbb{E}(V(X_0^i)) + \mathbb{E}(Q(V^i)(t\wedge\tau_n^i)) + \mathbb{E}(\mathbf{R}(V^i)(t\wedge\tau_n^i)).$$

Since  $\mathcal{L}_{x_i} V$  is bounded,  $t \mapsto \mathbb{E}(Q(V^i)(t \wedge \tau_n^i))$  is continuous. For  $0 \leq s \leq t$ , we may write

$$\mathbb{E}(\mathbf{R}(V^i)(t \wedge \tau_n^i)) - \mathbb{E}(\mathbf{R}(V^i)(s \wedge \tau_n^i)) = \frac{1}{N-1} \sum_{k=1}^n \sum_{j \neq i} \mathbb{E}\left(\left(V(X_{\tau_k^i}^j) - V(X_{\tau_k^i}^i)\right) \mathbb{1}_{s \leqslant \tau_k^i \leqslant t}\right),$$

and hence

$$\left|\mathbb{E}(\mathbf{R}(V^{i})(t \wedge \tau_{n}^{i})) - \mathbb{E}(\mathbf{R}(V^{i})(s \wedge \tau_{n}^{i}))\right| \leq 2n \|V\|_{\infty} \mathbb{P}(X^{i} \text{ dies between time } s \text{ and } t).$$

The law of the death times have a density with respect to the Lebesgue measure. Hence we have that

$$\lim_{x \to 0} P(X^i \text{ dies between time } s \text{ and } t) = 0,$$

and this implies the continuity of  $t \mapsto \mathbb{E}(\mathbf{R}(V^i)(t \wedge \tau_n^i)))$ , which concludes the proof.

**Remark 7.2.1.** When Assumption 7.1.2 also holds, we will additionally suppose that V is such that

$$\mathfrak{B}_1 \subset \{V > m\}$$

for some  $m > 3C_1$ , where  $\mathfrak{B}_1$  is the set from Assumption 7.1.2. Indeed, in the proof of Lemma 7.2.3, we may choose F such that  $U_{\min} < \min_{\mathfrak{B}_1} U$ . In this case, since  $V_0 > 4C_1$ , we may impose on f that  $f(\min_{\mathfrak{B}_1} U) > 3C_1$ .

We are now interested in the death probability of a particle.

**Proposition 7.2.2.** Under Assumption 7.1.1, denote by  $(X_t)$  the diffusion (7.1), and

$$\tau_{\partial D} = \inf \left\{ t \ge 0; X_t \notin D \right\}.$$

Consider any  $C_2 \in (2C_1, 4C_1)$ , where  $C_1$  is given in Lemma 7.2.3. Then we have:

$$p_{\varepsilon} := \sup_{x \in \{V \le C_2\}} \mathbb{P}_x(\tau_{\partial D} < t_{\varepsilon}) \to 0, \tag{7.13}$$

as  $\varepsilon \to 0$ .

*Proof.* Since  $t_{\varepsilon} = e^{a/\varepsilon}$  with  $a < U_0$ , this is the theory of Freidlin-Wentzell, see [65, Chapter 6, Theorem 6.2]. Freidlin and Wentzell didn't state the uniformity, but it follows from their proof.

# 7.2.3 Particles near the boundary

We want to control the number of particles which are close to the boundary of D after a time  $t_{\varepsilon}$ . Consider the neighborhood

$$\mathfrak{B} = \{V > 3C_1\}$$

of  $\partial D$ , where  $C_1$  is the constant from Lemma 7.2.3. For  $\mathbf{x} = (x_1, \cdots, x_N)$ , write:

$$A(\mathbf{x}) = \# \{ i \in [\![1, N]\!]; x_i \in \mathfrak{B} \},$$
(7.14)

where # stands for the cardinality of a set. We show that at time  $t_{\varepsilon}$ , the number of particles close to the boundary,  $A(\mathbf{x})$ , is a small fraction of N with high probability as N goes to infinity or as  $\varepsilon$  goes to 0.

**Lemma 7.2.4.** For all  $\alpha > 0$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon < \varepsilon_0$ , there exists  $q_{\varepsilon} > 0$  such that for all  $N \in \mathbb{N}$  and  $\mathbf{x} \in D^N$ :

$$\mathbb{P}_{\mathbf{x}}(A(\mathbf{X}_{t_{\varepsilon}}) > \alpha N) \leqslant q_{\varepsilon}^{N}$$
(7.15)

and  $q_{\varepsilon} \to 0$  as  $\varepsilon \to 0$ .

*Proof.* The idea for the proof is the following: we want to compare the evolution of  $V(X_i^i)$  and Ornstein-Uhlenbeck processes with small variance. If we had N independent diffusions, the result would derive from a simple enumeration. But then the interaction through jumps can only make the Lyapunov decrease. From Proposition 7.2.1 and the proof of Lemma 7.2.3, we have that almost surely for all  $1 \leq i \leq N$  and  $t \geq 0$ :

$$V\left(X_{t}^{i}\right) \leqslant V(x_{i}) + \int_{0}^{t} \left(-\omega V\left(X_{s}^{i}\right) + \omega C_{1}\right) \mathrm{d}s + \sqrt{2\varepsilon} \int_{0}^{t} \nabla V\left(X_{s}^{i}\right) \cdot \mathrm{d}B_{s}^{i}.$$

for some  $\omega > 0$  (independent from  $\varepsilon$ ). Now introduce for  $1 \leq i \leq N$  the process:

$$K_t^i = V(x_i)e^{-\omega t} + C_1(1 - e^{-\omega t}) + \sqrt{2\varepsilon} \int_0^t e^{\omega(s-t)} \nabla V(X_s^i) \cdot \mathrm{d}B_s^i$$

which solves

$$K_t^i = V(x_i) + \int_0^t \left( -\omega K_s^i + \omega C_1 \right) \mathrm{d}s + \sqrt{2\varepsilon} \int_0^t \nabla V\left(X_s^i\right) \cdot \mathrm{d}B_s^i.$$

From proposition 7.2.1, we have that:

$$V(X_{t}^{i}) - K_{t}^{i} = \int_{0}^{t} \left( \mathcal{L}V(X_{s}^{i}) + \omega K_{s}^{i} - \omega C_{1} \right) \mathrm{d}s + \sum_{n, \tau_{n}^{i} \leqslant t} V(X_{\tau_{n}^{i}}^{i}) - V(X_{\tau_{n}^{i}}^{i}),$$

where  $(\tau_n^i)$  are the death times of particle number *i* as defined in (7.11). Moreover,  $K^i$  is a continuous process, and  $V(X^i)$  is continuous between death times. Hence,  $V(X^i) - K^i$  is  $\mathcal{C}^1$  between death times. Let  $f(x,y) = ((x-y)_+)^2$ , so that *f* is differentiable, and is non-decreasing in the variable *x*. By construction of *V*, for all  $n \in \mathbb{N}$ ,  $V(X_{\tau_n^i}^i) \leq V(X_{\tau_n^i}^i)$ , hence we have for all  $t \geq 0$ :

$$\left(\left(V(X_t^i) - K_t^i\right)_+\right)^2 \leqslant 2 \int_0^t \omega \left(K_s^i - V\left(X_s^i\right)\right) \left(V\left(X_s^i\right) - K_s^i\right)_+ \mathrm{d}s \leqslant 0.$$

Thus, almost surely,  $V(X_t^i)$  is bounded by  $K_t^i$  for all  $t \ge 0$ , and we are left to show that with high probability, there are only a few  $K^i$ 's which are greater than  $3C_1$  at time  $t_{\varepsilon}$ . Write:

$$G_t^i = \sqrt{2\varepsilon} \int_0^t e^{\omega(s-t)} \nabla V(X_s^i) \cdot \mathrm{d}B_s^i$$

Fix some family of indexes  $(i_1, \ldots, i_k) \in \{1, \ldots, N\}^k$ . The  $G^i$ 's are  $L^2$ -martingales, hence for any  $\xi \in \mathbb{R}, \xi \sum_{j=1}^k G^{i_j}$  is a  $L^2$ -martingale, and:

$$\exp\left(\xi\sum_{j=1}^{k}G^{i_{j}}-\xi^{2}\left\langle\sum_{j=1}^{k}G^{i_{j}}\right\rangle\right)$$

is a local-martingale. We have that  $\langle G^i, G^j \rangle = 0$  for all  $i \neq j$  because the Brownian motions are independent, hence

$$\left\langle \sum_{j=1}^{k} G^{i_j} \right\rangle_t = \sum_{j=1}^{k} 2\varepsilon \int_0^t e^{2\omega(s-t)} |\nabla V(X_s^i)|^2 \mathrm{d}s \leqslant \frac{\varepsilon k \|\nabla V\|_{\infty}}{\omega}$$

and using Fatou's Lemma:

$$\mathbb{E}\left(\exp\left(\xi\sum_{j=1}^{k}G_{t}^{i_{j}}\right)\right)\leqslant\exp\left(\frac{\varepsilon\xi^{2}k\|\nabla V\|_{\infty}}{\omega}\right),$$

for all  $t \ge 0$ . Now we can write, using the Markov inequality:

$$\mathbb{P}\left(G_{t_{\varepsilon}}^{i_{1}} > C_{1}, \cdots, G_{t_{\varepsilon}}^{i_{k}} > C_{1}\right) \leqslant \mathbb{P}\left(\exp\left(\xi \sum_{j=1}^{k} G^{i_{j}}\right) > e^{\xi k C_{1}}\right) \leqslant e^{-\xi k C_{1}} e^{\frac{\varepsilon \xi^{2} k \|\nabla V\|_{\infty}}{\omega}}.$$

Taking  $\xi = C_1 \omega / (2\varepsilon \|\nabla V\|_{\infty})$ , one gets:

$$\mathbb{P}\left(G_{t_{\varepsilon}}^{i_{1}} > C_{1}, \cdots, G_{t_{\varepsilon}}^{i_{k}} > C_{1}\right) \leqslant \exp\left(-C_{1}^{2}\omega/2\varepsilon \|\nabla V\|_{\infty}\right)^{k} =: \tilde{q}_{\varepsilon}^{k}.$$

We chose  $\varepsilon_0$  small enough so that:

$$V_0 e^{-\omega t_{\varepsilon_0}} + C_1 (1 - e^{-\omega t_{\varepsilon_0}}) < 2C_1.$$

For all  $1 \leq i \leq N$ , we then have:

$$\left\{K_{t_{\varepsilon}}^{i} > 3C_{1}\right\} \subset \left\{G_{t_{\varepsilon}}^{i} > C_{1}\right\},\$$

and we have for all family of indexes  $(i_1, \ldots, i_k)$ :

$$\mathbb{P}\left(X_{t_{\varepsilon}}^{i_{1}} \in \mathfrak{B}, \dots, X_{t_{\varepsilon}}^{i_{k}} \in \mathfrak{B}\right) \leqslant \mathbb{P}\left(G_{t_{\varepsilon}}^{i_{1}} > C_{1}, \dots, G_{t_{\varepsilon}}^{i_{k}} > C_{1}\right) \leqslant \tilde{q}_{\varepsilon}^{k}.$$

Finally, we conclude with:

 $\mathbb{P}_{x}\left(A\left(\mathbf{X}_{t_{\varepsilon}}\right) > \alpha N\right) \leqslant \mathbb{P}\left(\text{There exist at least } \alpha N \text{ indexes } i \text{ such that } X_{t_{\varepsilon}}^{i} \in \mathfrak{B}\right)$ 

$$\leq \sum_{\alpha N \leq k \leq N} {n \choose k} \tilde{q}_{\varepsilon}^{k}$$
$$\leq (2 (\tilde{q}_{\varepsilon})^{\alpha})^{N} =: q_{\varepsilon}^{N}.$$

г		۰.
L		
L		
L		
L		

# 7.3 Proofs of the main theorems

Our goal is to construct a coupling of  $\delta_{\mathbf{x}} P^{N,\varepsilon}$  and  $\delta_{\mathbf{y}} P^{N,\varepsilon}$  for all  $\mathbf{x}, \mathbf{y} \in D^N$  in such a way that some distance  $\mathbf{d}(\mathbf{x}, \mathbf{y})$  is contracted on average by this coupling along time. The basic idea of the coupling is the following: particles are coupled by pair, namely we want the particle *i* of the system starting at  $\mathbf{x}$  to merge, after a time  $t_{\varepsilon}$ , with the particle *i* of the system starting at  $\mathbf{y}$ . However, contrary to the case of independent particles, here, even if two particles start at the same position (namely  $x_i = y_i$ ), they have a positive probability to decouple before time  $t_{\varepsilon}$ . This can be particularly bad for some initial conditions: for instance if most of the pairs start merged but close to the boundary while a decoupled pair is in the middle of the domain, then this will typically lead to a lot of decoupling as coupled pairs rebirth on the uncoupled pair. This will be tackled through the definition of the distance  $\mathbf{d}$ .

# 7.3.1 Long time convergence

We now construct the coupling of  $(\delta_{\mathbf{x}} P_t^{N,\varepsilon})_{t \ge 0}$  and  $(\delta_{\mathbf{y}} P_t^{N,\varepsilon})_{t \ge 0}$  for all  $\mathbf{x}, \mathbf{y} \in D^N$ , that will yield a bound on the distance between  $\delta_{\mathbf{x}} P_{t_{\varepsilon}}^{N,\varepsilon}$  and  $\delta_{\mathbf{y}} P_{t_{\varepsilon}}^{N,\varepsilon}$ . Fix  $\mathbf{x}, \mathbf{y} \in D^N$ , and a sequence  $(I_n^i)$  of independent random variable, where  $I_n^i$  is uniform on  $[\![1,N]\!] \setminus \{i\}$ .

For all  $1 \leq i \leq N$ , consider a coupling  $(\tilde{X}_t^i, \tilde{Y}_t^i)$  of the diffusion (7.8) starting from  $(x_i, y_i)$  such as the one constructed in Lemma 7.2.2 (with these processes being independent for two different values of the index *i*). Recall that  $\mathbb{T}^d = (\mathbb{R}/2L\mathbb{Z})^d$ , and *L* is great enough so that we may consider that  $D \subset \mathbb{T}^d$ . Hence we may write

$$\tilde{\tau}_1 = \inf\left\{t \ge 0, \exists i \in [\![1, N]\!], \tilde{X}_t^i \notin D \text{ or } \tilde{Y}_t^i \notin D\right\}.$$

Denote by  $i_1$  the index of the particles that exit the domain at time  $\tilde{\tau}_1$ . For all  $i \neq i_1$  and

 $0 \leq t \leq \tilde{\tau}_1$  or  $i = i_1$  and  $0 \leq t < \tilde{\tau}_1$ , let:

$$X_t^i = \tilde{X}_t^i \quad \text{and} \quad Y_t^i = \tilde{Y}_t^i,$$

in the sense that  $X_t^i$  (resp.  $Y_t^i$ ) is the only point of D whose projection is  $\tilde{X}_t^i$  (resp.  $\tilde{Y}_t^i$ ). Finally, if  $\tilde{X}_{\tilde{\tau}_1}^{i_1} \notin D$ , then set  $X_{\tilde{\tau}_1}^{i_1} = X_{\tilde{\tau}_1}^{I_1^{i_1}}$ , else set  $X_{\tilde{\tau}_1}^{i_1} = \tilde{X}_{\tilde{\tau}_1}^{i_1}$ . The same goes for  $Y_{\tilde{\tau}_1}^{i_1}$ : if  $\tilde{Y}_{\tilde{\tau}_1}^{i_1} \notin D$ , then set  $Y_{\tilde{\tau}_1}^{i_1} = Y_{\tilde{\tau}_1}^{I_1^{i_1}}$ , else set  $Y_{\tilde{\tau}_1}^{i_1} = \tilde{Y}_{\tilde{\tau}_1}^{i_1}$ . The coupling can then be constructed by induction, just as for the construction of the FV processes in the introduction.

Basically, the coupling is as follow: two particles with same index will be an optimal coupling of the diffusion as long as they don't die as constructed in Lemma 7.2.2, and if they die while coupled, then they resurrect using the same index. By this we mean that the uniform variable  $I_n^i$  used in the construction of the process in Section 7.1.1 must be the same for both systems.

We will show that this coupling yields a contraction for the Wasserstein distance associated to a particular distance on  $D^N$ , namely:

$$\mathbf{d}(\mathbf{x}, \mathbf{y}) = \sum_{i=1}^{N} \mathbb{1}_{x_i \neq y_i} \left( 1 + \beta V(x_i) + \beta V(y_i) \right) + \left( 1 + V_0 \right) N \left( \mathbb{1}_{A(\mathbf{x}) > \alpha N} + \mathbb{1}_{A(\mathbf{y}) > \alpha N} \right) \mathbb{1}_{\mathbf{x} \neq \mathbf{y}}, \quad (7.16)$$

where  $\beta, \alpha > 0$  are parameters that will be chosen small enough, and  $A(\mathbf{x})$  has been defined in (7.14). We define as well:

$$d^{1}(x_{i}, y_{i}) = \mathbb{1}_{x_{i} \neq y_{i}} \left( 1 + \beta V(x_{i}) + \beta V(y_{i}) \right).$$

The meaning of this distance, which follows the construction of Hairer and Mattingly in [79], is this: if  $x_i \neq y_i$  and  $V(x_i) + V(y_i) < C_2$ , where  $C_2$  is as in Proposition 7.2.2, then both particles of index i are in the center of the domain at initial time, and we are able to couple  $X^i$  and  $Y^i$ before time  $t_{\varepsilon}$  and before they die with high probability. If  $x_i \neq y_i$  and  $V(x_i) + V(y_i) \geq C_2$ , then we may not be able to couple them, but the Lyapunov functional will decrease on average. In any case, if  $x_i \neq y_i$ ,  $\mathbb{E}\left(d^1(X_t^i, Y_t^i)\right)$  will decrease between initial time and time  $t_{\varepsilon}$ . If  $x_i = y_i$ , then we cannot expect any contraction of  $\mathbb{E}\left(d^1(X_i^i,Y_i^i)\right)$ , since it is equal to zero at initial time, and the probability that  $X^i$  and  $Y^i$  decouple is positive (if they die and resurrect on an uncoupled pair). In this case, if  $x_i$  is in the center of the domain, then, as we will see below, the probability of decoupling is very small and won't be an issue. But in the case where there are many particles coupled at t = 0 close to the boundary, many of them will get separated. This is why we added the additional term  $N\left(\mathbb{1}_{A(\mathbf{x})>\alpha N} + \mathbb{1}_{A(\mathbf{y})>\alpha N}\right)\mathbb{1}_{\mathbf{x}\neq\mathbf{y}}$  in the definition of **d**. If we are in this case, this term is initially not zero but, according to Lemma 7.2.4, it will probably be zero at time  $t_{\varepsilon}$ , which will compensate for the non-zero terms that will appear with other parts of the distance. In other words, this term plays the role of a global Lyapunov function, by contrast with the pairwise Lyapunov function  $V(x_i) + V(y_i)$ .

Let's start by bounding from above the probability to decouple. This is the part where we use Assumption 7.1.2.

**Lemma 7.3.1.** Under Assumptions 7.1.1 and 7.1.2, there exists  $C_3$  such that for all  $0 < \alpha < 1/4$ , there exists  $\varepsilon_0 > 0$  such that for all  $0 < \varepsilon < \varepsilon_0$ , there exists  $m_{\varepsilon} > 0$ , such that for all  $N \in \mathbb{N}$ ,  $\mathbf{x}, \mathbf{y} \in D^N$  with  $A(\mathbf{x}), A(\mathbf{y}) \leq \alpha N$ , and all  $i \in [\![1, N]\!]$  such that  $x_i = y_i$ , we have:

$$\mathbb{P}_{\mathbf{x},\mathbf{y}}\left(X_{t_{\varepsilon}}^{i}\neq Y_{t_{\varepsilon}}^{i}\right) \leqslant \begin{cases} m_{\varepsilon}C_{3}d(\mathbf{x},\mathbf{y})/N & \text{if } x_{i}\notin\mathfrak{B}\\ C_{3}\overline{d}(\mathbf{x},\mathbf{y})/N & \text{if } x_{i}\in\mathfrak{B} \end{cases}$$

where  $\bar{d}(\mathbf{x}, \mathbf{y}) = \sum_{i=1}^{N} \mathbb{1}_{x_i \neq y_i}$ , and  $\lim_{\varepsilon \to 0} m_{\varepsilon} = 0$ .

An intermediate lemma is needed. The goal of this lemma is to get bounds on the number of death events. This next lemma is where we use the neighborhood  $\mathfrak{B}_1$  of  $\partial D$  from Assumption 7.1.2.

**Lemma 7.3.2.** Under Assumptions 7.1.1 and 7.1.2, let  $\mathfrak{B}_1$  be the neighborhood of  $\partial D$  from Assumption 7.1.2. Write the event:

$$\mathcal{A} = \left\{ \# \left\{ i \in \llbracket 1, N \rrbracket, \exists t \leqslant t_{\varepsilon}, X_t^i \in \mathfrak{B}_1 \right\} \ge 2\alpha N \right\}.$$

1. There exists  $\varepsilon_0 > 0$  such that for all  $0 < \varepsilon < \varepsilon_0$ , there exists  $\tilde{p}_{\varepsilon} > 0$ , such that for all  $0 < \alpha < 1/4$ ,  $N \in \mathbb{N}$ ,  $\mathbf{x} \in D^N$  with  $A(\mathbf{x}) \leq \alpha N$ ,

$$\mathbb{P}_{\mathbf{x}}(\mathcal{A}) \leqslant \left(2\tilde{p}_{\varepsilon}^{\alpha}\right)^{N},$$

and  $\lim_{\varepsilon \to 0} \tilde{p}_{\varepsilon} = 0$ .

2. Moreover, if T denote the number of rebirth in the system before time  $t_{\varepsilon}$ , there exists  $\varepsilon_0, \sigma > 0, 0 < q < 1$ , such that for all  $0 < \varepsilon < \varepsilon_0$  and  $0 < \alpha < 1/4$ :

$$\mathbb{P}(T > \sigma N, \mathcal{A}^c) \leqslant q^N.$$

3. Write  $T^i$  the number of rebirth of particle *i* before time  $t_{\varepsilon}$ . We have as well that there exist  $C, \varepsilon_0 > 0$  such that for all  $0 < \varepsilon < \varepsilon_0$ , for all  $0 < \alpha < 1/4$ ,  $\mathbf{x} \in D^N$  satisfying  $A(\mathbf{x}) < \alpha N$ :

$$\mathbb{E}_{\mathbf{x}}\left(T^{i}\mathbb{1}_{\mathcal{A}^{c}}\right) \leqslant \begin{cases} C\bar{p}_{\varepsilon} & \text{if } x_{i} \notin \mathfrak{B} \\ C & \text{if } x_{i} \in \mathfrak{B}, \end{cases}$$

where  $\bar{p}_{\varepsilon}$  is given in (7.7).

*Proof.* 1. At time t = 0, the condition on **x** implies that there are less than  $\alpha N$  particles in  $\mathfrak{B}$ . Under Assumption 7.1.2, we assumed in Remark 7.2.1 that

$$\min_{\mathfrak{B}_1} U > \max_{D \setminus \mathfrak{B}} U.$$

Hence  $\mathfrak{B}_1 \subset \mathfrak{B}$ , and this means that for  $\mathcal{A}$  to happen, namely for  $2\alpha N$  particles to visit  $\mathfrak{B}_1$  before time  $t_{\varepsilon}$ , at least  $\alpha N$  particles that were initially in  $D \setminus \mathfrak{B}$  must have reached  $\mathfrak{B}_1$  before time  $t_{\varepsilon}$ . Write:

$$\widetilde{p}_{\varepsilon} = \sup_{x \in D \setminus \mathfrak{B}} \mathbb{P}_x \left( \tau_{\mathfrak{B}_1} < t_{\varepsilon} \right),$$

where  $\tau_{\mathfrak{B}_1}$  is the first hitting time of the set  $\mathfrak{B}_1$  for the diffusion (7.1). Under Assumption 7.1.2,  $a < \min_{\mathfrak{B}_1} U$ . Together with the fact that  $\min_{\mathfrak{B}_1} U > \max_{D \setminus \mathfrak{B}} U$  and [65, Chapter 6, Theorem 6.2] yields that  $\tilde{p}_{\varepsilon} \to 0$  as  $\varepsilon \to 0$ . The fact that a particle reaches  $\mathfrak{B}_1$  only depends on the Brownian motion driving it, hence we have:

$$\mathbb{P}(\mathcal{A}) \leqslant \sum_{k \geqslant \alpha N} \binom{(1-\alpha)N}{k} \tilde{p}_{\varepsilon}^k \leqslant (2\tilde{p}_{\varepsilon}^{\alpha})^N.$$

2. In order to control the number of deaths of the  $i^{th}$  particle up to time  $t_{\varepsilon}$ , we are going to distinguish two types of rebirth events: either the particle is resurrected on a particle

which we know never reaches  $\mathfrak{B}_1$  (i.e. stays away from the boundary), in which case we can bound the probability that the  $i^{th}$  particle dies again, or it is resurrected on a particle for which we have no information, in which case it can be arbitrarily close to the boundary and the time before the next death of the  $i^{th}$  particle can be arbitrarily small.

For convenience, we consider in the rest of the proof that the FV process has been defined thanks to a construction similar to the one presented in Section 7.1.1 except that the Brownian motions driving the SDEs are changed at each death event, namely along with the variables  $(I_n^i)_{n \ge 0, i \in [\![1,N]\!]}$ , we consider a family of independent *d*-dimensional Brownian motions  $((B_t^{n,i})_{t\ge 0})_{n\ge 0,i\in [\![1,N]\!]}$ , so that after its  $n^{th}$  death and up to its  $(n+1)^{th}$  death the position of the particle *i* is given by  $X_{\tau_n^i+t}^i = \bar{X}_t^i$  where  $\bar{X}^i$  is the solution of (7.1) driven by  $B^{n,i}$  with initial condition  $\bar{X}_0^i = X_{\tau_n^i}^i$  (recall the notation  $\tau_n^i$  from (7.11)). Of course the law of the process is correct with this construction.

Denote:

$$\mathbf{S} = \left\{ i \in \llbracket 1, N \rrbracket, \exists t < t_{\varepsilon}, X_t^i \in \mathfrak{B}_1 \right\}.$$

Then the Markov inequality yields:

$$\mathbb{P}\left(T > \sigma N, \mathcal{A}^{c}\right) \leqslant e^{-\sigma N} \mathbb{E}\left(e^{T} \mathbb{1}_{\mathcal{A}^{c}}\right) = e^{-\sigma N} \sum_{\substack{S \in \mathcal{P}\left([\![1,N]\!]\right) \\ \#S \leqslant 2\alpha N}} \mathbb{E}\left(e^{\sum_{i=1}^{N} T_{i}} \mathbb{1}_{\mathbf{S}=S}\right).$$

Fix  $S \in \mathcal{P}(\llbracket 1, N \rrbracket)$ , such that  $\#S < 2\alpha N$ , and recall the definition of the variable  $I_n^i$  used in the construction of the FV process, which are independent uniform variables on  $\llbracket 1, N \rrbracket \setminus \{i\}$ . We define by induction  $P_0^i = 0$  and :

$$P_k^i = \inf\left\{n > P_{k-1}^i, I_n^i \notin S\right\}.$$

Notice that, under the event  $\{\mathbf{S} = S\}$ , if  $I_n^i \notin S$ , it means that at its  $n^{th}$  rebirth the particle i is resurrected on a particle which never reaches  $\mathfrak{B}_1$  before time  $t_{\varepsilon}$ .

Setting  $k_0(i) = 1$  if  $x_i \in \mathfrak{B}_1$  and  $k_0(i) = 0$  otherwise, we define as well

$$P^{i} = \inf \left\{ k \ge k_{0}(i), \forall x \in D \setminus \mathfrak{B}_{1}, \tau_{D}(X^{x,i,P_{k}^{i}}) > t_{\varepsilon} \right\},\$$

where for  $n \in \mathbb{N}$  the family of processes  $(X^{x,i,n})_{x \in D \setminus \mathfrak{B}_1}$  is as in Assumption 7.1.2 and are driven by the Brownian motion  $B^{n,i}$ . Since we have already observed that, for all k > 0, at its  $(P_k^i)^{th}$  death, the particle *i* is resurrected at a position in  $D \setminus \mathfrak{B}_1$ , the event  $\{\forall x \in D \setminus \mathfrak{B}_1, \tau_D(X^{x,i,P_k^i}) > t_{\varepsilon}\}$ , which is measurable with respect to the Brownian motion  $B^{P_k^i,i}$ , implies that the particle does not die again before time  $t_{\varepsilon}$ . For k = 0, it depends whether initially  $x_i \in \mathfrak{B}_1$ : if  $x_i \notin \mathfrak{B}_1$  (which is in particular the case if  $x_i \notin \mathfrak{B}$ ) then, again, the event  $\{\forall x \in D \setminus \mathfrak{B}_1, \tau_D(X^{x,i,0}) > t_{\varepsilon}\}$  implies that the particle doesn't die before time  $t_{\varepsilon}$ . This is not the case if  $x_i \in \mathfrak{B}_1$ . As a consequence, in any cases, under the event  $\{\mathbf{S} = S\}$ , we can bound the total number of death of the *i*<sup>th</sup> particle by

$$T^i \leqslant \sum_{k=1}^{P^i} (P^i_k - P^i_{k-1}).$$

The variables  $(P_k^i - P_{k-1}^i)_{k \ge 1,i}$  are independent geometric random variables of parameter
$1 - \#S > 1 - 2\alpha$ . Under Assumption 7.1.2, if  $x_i \in \mathfrak{B}_1$  (resp. if  $x_i \notin \mathfrak{B}_1$ ) then  $P^i$  (resp.  $P^i + 1$ ) is a geometric random variable of parameter  $1 - \bar{p}_{\varepsilon}$ . Moreover,  $(P^i)_{1 \leq i \leq N}$  is a family of independent random variables, independent from  $(P^i_k)_{k \geq 1, 1 \leq i \leq N}$ . We have:

$$\mathbb{E}\left(e^{\sum_{i=1}^{N}T_{i}}\mathbb{1}_{\mathbf{S}=S}\right) \leqslant \mathbb{E}\left(e^{\sum_{i=1}^{N}\sum_{k=1}^{P^{i}}(P_{k}^{i}-P_{k-1}^{i})}\right) = \left(\mathbb{E}\left(e^{\sum_{k=1}^{P^{i}}(P_{k}^{i}-P_{k-1}^{i})}\right)\right)^{N}$$

We are just left to show that  $\mathbb{E}\left(\exp\sum_{k=1}^{P^i}(P_k^i-P_{k-1}^i)\right)$  is finite and bounded uniformly in  $\varepsilon < \varepsilon_0$ . Conditioning with respect to  $P^i$  we get :

$$\mathbb{E}\left(e^{\sum_{k=1}^{P^{i}}P^{i}_{k}-P^{i}_{k-1}}\right) = \mathbb{E}\left(\mathbb{E}\left(e^{P^{i}_{0}}\right)^{P^{i}}\right) \leqslant \mathbb{E}\left(\left(\frac{e}{1-e\alpha}\right)^{P^{i}}\right)$$

hence the result if  $\varepsilon_0$  satisfies  $\bar{p}_{\varepsilon_0} < \frac{1-e\alpha}{e}$ , since we bound then

$$\mathbb{P}\left(T > \sigma N, \mathcal{A}^{c}\right) \leqslant \left(2e^{-\sigma}\mathbb{E}\left(\left(\frac{e}{1-e\alpha}\right)^{P^{i}}\right)\right)^{N}.$$

3. In the same spirit, fix  $i \in [1, N]$ , and write now:

$$P_k^i = \inf\left\{n > P_{k-1}^i, I_n^i \notin \mathbf{S}^i\right\}$$

where

$$\mathbf{S}^{i} = \left\{ j \in \left[\!\left[1, N\right]\!\right] \setminus \left\{i\right\}, \exists t < t_{\varepsilon}, X_{t}^{j} \in \mathfrak{B}_{1} \right\},\$$

and

$$\mathcal{A}^{i} = \left\{ \# \left\{ j \in \left[\!\left[1, N\right]\!\right] \setminus \left\{i\right\}, \exists t \leqslant t_{\varepsilon}, X_{t}^{j} \in \mathfrak{B}_{1} \right\} \geqslant 2\alpha N \right\},\$$

and the definition of  $P^i$  does not change. We have that  $(P_k^i)_k$ , and  $P^i$  are independent random variable, and  $P^i$  is independent of  $\mathcal{A}^i$  and  $\mathbf{S}^i$ . Indeed,  $\mathcal{A}^i$  and  $\mathbf{S}^i$  only depends on the Brownian motions that drive  $(X^j)_{j\neq i}$ . Under the event  $(\mathcal{A}^i)^c$ , the cardinality of  $\mathbf{S}^i$  is less than  $2\alpha N$ . Furthermore, we have that  $\mathcal{A}^i \subset \mathcal{A}$ , and hence, as in the previous step,

$$\mathbb{E}\left(T^{i}\mathbb{1}_{\mathcal{A}^{c}}\right) \leqslant \mathbb{E}\left(\sum_{k=1}^{P^{i}}\left(P_{k}^{i}-P_{k-1}^{i}\right)\mathbb{1}_{(\mathcal{A}^{i})^{c}}\right) \leqslant \mathbb{E}\left(P^{i}\mathbb{E}(P_{1}^{i}|\mathbf{S}^{i},\mathcal{A}^{i},P^{i})\right) \leqslant \frac{1}{1-2\alpha}\mathbb{E}(P^{i}),$$

and we conclude by bounding  $\mathbb{E}(P^i) \leq \bar{p}_{\varepsilon}(1-\bar{p}_{\varepsilon})^{-1}$  if  $x_i \notin \mathfrak{B}$  (since, then,  $x_i \notin \mathfrak{B}_1$ ) and, otherwise,  $\mathbb{E}(P^i) \leq (1-\bar{p}_{\varepsilon})^{-1}$ .

Proof of Lemma 7.3.1. Define the sets:

$$\begin{split} U_1(0) &= \left\{ i \in [\![1,N]\!], \ x_i \neq y_i \right\}, \\ U_2(0) &= \left\{ i \in [\![1,N]\!], \ x_i = y_i \right\}. \end{split}$$

Now, for  $t \ge 0$ , we want to define some sets  $U_1(t)$ ,  $U_2(t)$ , such that if  $X^i$  and  $Y^i$  decouple at

some time  $s \ge 0$ , then for all  $t \ge s$ ,  $i \in U_1(t)$ . For  $i \in U_2(0)$ ,  $n \in \mathbb{N}$ , write:

$$\tau_{n}^{i} = \inf \left\{ t > \tau_{n-1}^{i}, X_{t-}^{i} = Y_{t-}^{i} \in \partial D \right\},\$$

as in(7.11), and

$$\bar{\tau}_d^i = \inf\left\{t \ge 0, X_t^i \neq Y_t^i\right\}.$$

Since the FV-process is well-defined, almost surely, there is only a finite number of such events before time  $t_{\varepsilon}$ , for all  $1 \leq i \leq N$ . Then define the set  $U_1(t)$  and  $U_2(t)$  for  $t \in (\tau_{k-1}, \tau_k]$  by induction on  $k \geq 1$ . Assume that the sets have been defined up to the time  $\tau_{k-1}$  for some  $k \geq 1$ . Set  $U_j(t) = U_j(\tau_{k-1})$  for all  $t \in (\tau_{k-1}, \tau_k)$ . Let  $i \in [\![1, N]\!]$  be the index such that  $\tau_k \in \bigcup_{n \in \mathbb{N}} \{\bar{\tau}_n^i\}$ . Now we distinguish two cases. If  $\tau_k \neq \bar{\tau}_d^i$ , then  $U_j(\tau_k) = U_j(\tau_{k-1})$  for j = 1, 2. Else set:

$$U_1(\tau_k) = U_1(\tau_{k-1}) \cup \{i\}, \quad U_2(\tau_k) = U_2(\tau_{k-1}) \cap \{i\}^c.$$

It is immediate to check that  $U_1(t)$  and  $U_2(t)$  form a partition of  $\llbracket 1, N \rrbracket$  for all  $t \ge 0$ , and that  $U_1(t)$  is non-decreasing with t and such that  $\{i \in \llbracket 1, N \rrbracket, X_t^i \ne Y_t^i\} \subset U_1(t)$  for all  $t \ge 0$ . Recall from Lemma 7.3.2 the event:

$$\mathcal{A} = \left\{ \# \left\{ i \in \llbracket 1, N \rrbracket, \exists t \leqslant t_{\varepsilon}, X_t^i = Y_t^i \in \mathfrak{B}_1 \right\} \ge 2\alpha N \right\}.$$

For  $n \in \mathbb{N}$  and j = 1, 2, write  $u_k^j = \#U_j(\tau_k)$ . At each time  $\tau_k$ , the probability that a particle goes from  $U_2$  to  $U_1$  is less than  $u_{k-1}^1/N$ . Hence, we have that for all  $k \ge 1$ :

$$\mathbb{E}\left(u_{k+1}^{1}|\mathcal{F}_{\tau_{k}}\right) \leqslant u_{k}^{1}\left(1+\frac{1}{N}\right),$$

and thus

$$\mathbb{E}\left(u_{k}^{1}\right) \leqslant \bar{d}(x,y)\left(1+\frac{1}{N}\right)^{k}.$$

Using the notations of Lemma 7.3.2, in particular T to denote the total number of death event before time  $t_{\varepsilon}$ , using that  $u_n^1$  is non-decreasing, we bound

$$\mathbb{E}(u_T^1 \mathbb{1}_{\mathcal{A}^c}) \leq \mathbb{E}(u_{\sigma N}^1) + N\mathbb{P}(T > \sigma N, \mathcal{A}^c)$$
$$\leq e^{\sigma} \bar{d}(x, y) + Nq^N,$$

which is bounded uniformly on  $N \ge 1$  and  $\varepsilon$  small enough, by  $\tilde{C}_3 \bar{d}(x, y)$ , for some  $\tilde{C}_3 > 0$ , as soon as  $\bar{d}(x, y) \ge 1$  (while, if  $\bar{d}(x, y) = 0$  then the two processes remain equal for all times and thus the result is trivial). We get from all of this:

$$\mathbb{E}\left(\sup_{t\leqslant t_{\varepsilon}}\bar{d}\left(\mathbf{X}_{t},\mathbf{Y}_{t}\right)\mathbb{1}_{\mathcal{A}^{c}}\right)\leqslant\mathbb{E}(u_{T}^{1}\mathbb{1}_{\mathcal{A}^{c}})\leqslant\tilde{C}_{3}\bar{d}(x,y).$$

Now we can bound the probability to decouple starting from any  $x_i = y_i \in D$ , for a fixed *i* (recall the notation  $T^i$  from Lemma 7.3.2):

$$\mathbb{P}\left(\exists 0 < t < t_{\varepsilon}, X_{t}^{i} \neq Y_{t}^{i}\right) \leqslant \sum_{n \geq 1} \mathbb{P}\left(\bar{\tau}_{d}^{i} = \bar{\tau}_{n}^{i}, \mathcal{A}^{c}, T_{i} > n\right) + \mathbb{P}(\mathcal{A})$$
$$= \sum_{n \geq 1} \mathbb{E}\left(\bar{d}(\mathbf{X}_{\tau_{n}}, \mathbf{Y}_{\tau_{n}}) / N \mathbb{1}_{\mathcal{A}^{c}}\right) \mathbb{P}\left(T_{i} > n, \mathcal{A}^{c}\right) + (2\tilde{p}_{\varepsilon}^{\alpha})^{N}$$

$$\leq \frac{1}{N} \sum_{n \geq 1} \mathbb{E} \left( \sup_{t \leq t_{\varepsilon}} \bar{d} \left( \mathbf{X}_{t}, \mathbf{Y}_{t} \right) \mathbb{1}_{\mathcal{A}^{c}} \right) \mathbb{P} \left( T_{i} > n, \mathcal{A}^{c} \right) + (2\tilde{p}_{\varepsilon}^{\alpha})^{N}$$

$$\leq \tilde{C}_{3} \frac{\bar{d}(x, y)}{N} \mathbb{E} \left( T_{i} \mathbb{1}_{\mathcal{A}^{c}} \right) + (2\tilde{p}_{\varepsilon}^{\alpha})^{N}.$$

We conclude using Lemma 7.3.2.

We need to choose the parameters involved in the definition of the distance **d**. There are three of them:  $\alpha$ ,  $\beta$ , and  $V_0$ . We fix any  $V_0 > 4C_1$  (which is required in Lemma 7.2.3), any  $\beta < (2 \vee 4C_1)^{-1}$  and then  $\alpha$  small enough so that

$$\frac{1+2\beta C_1}{1+\beta C_2} \lor 4\beta C_1 + \alpha C_3 (1+2\beta V_0) < 1, \tag{7.17}$$

and

$$\frac{1+2\beta V_0}{1+V_0} < 1,\tag{7.18}$$

where we used in (7.17) that  $C_2 > 2C_1$  from Proposition 7.2.2. This is possible by fixing first some small  $\beta$ , and then taking any  $V_0 > 4C_1$ , and finally  $\alpha$  small enough.

**Lemma 7.3.3.** Let  $\mathbf{x}, \mathbf{y} \in D^N$  and  $1 \leq i \leq N$  such that  $x_i \neq y_i$  and  $V(x_i) + V(y_i) \leq C_2$ . Then with  $\kappa_{1,\varepsilon} = \gamma_{\varepsilon} \vee (1 - c_{\varepsilon} + 2p_{\varepsilon} + 4\beta C_1(1 - \gamma_{\varepsilon}))$ , where  $\gamma_{\varepsilon}$  has been defined in lemma 7.2.3, we have:

$$\mathbb{E}(d^1(X_{t_{\varepsilon}}^i, Y_{t_{\varepsilon}}^i)) \leqslant \kappa_{1,\varepsilon} d^1(x_i, y_i).$$

*Proof.* Let  $(\tilde{X}_t^i, \tilde{Y}_t^i)$  be the coupling of the diffusion (7.8) as in lemma 7.2.2, used in the construction of our coupling. Then,  $(\tilde{X}_t^i, \tilde{Y}_t^i) = (X_t^i, Y_t^i)$  until  $X^i$  or  $Y^i$  reaches  $\partial D$ . We have :

$$\begin{split} \mathbb{P}\left(X_{t_{\varepsilon}}^{i}=Y_{t_{\varepsilon}}^{i}\right) &\geqslant \mathbb{P}\left(X_{t_{\varepsilon}}^{i}=Y_{t_{\varepsilon}}^{i}, \tau_{x_{i}} > t_{\varepsilon}, \tau_{y_{i}} > t_{\varepsilon}\right) \\ &= \mathbb{P}\left(\tilde{X}_{t_{\varepsilon}}^{x_{i}}=\tilde{Y}_{t_{\varepsilon}}^{y_{i}}, \tau_{x_{i}} > t_{\varepsilon}, \tau_{y_{i}} > t_{\varepsilon}\right) \\ &\geqslant \mathbb{P}\left(\tilde{X}_{t_{\varepsilon}}^{x_{i}}=\tilde{Y}_{t_{\varepsilon}}^{y_{i}}\right) - \mathbb{P}\left(\tau_{x_{i}} > t_{\varepsilon}\right) - \mathbb{P}\left(\tau_{y_{i}} > t_{\varepsilon}\right) \\ &\geqslant c_{\varepsilon} - 2p_{\varepsilon}. \end{split}$$

Using the property of the Lyapunov function described in (7.12), we then have:

$$\mathbb{E}\left(d^{1}(X_{t_{\varepsilon}}^{i}, Y_{t_{\varepsilon}}^{i})\right) \leq 1 - c_{\varepsilon} + 2p_{\varepsilon} + 2\beta C_{1}(1 - \gamma_{\varepsilon}) + \gamma_{\varepsilon}\beta\left(V(x_{i}) + V(y_{i})\right)$$
$$\leq \kappa_{1,\varepsilon}d^{1}(x_{i}, y_{i}).$$

Now we focus on the particles near the boundary that are not coupled:

**Lemma 7.3.4.** Let  $\mathbf{x}, \mathbf{y} \in D^N$  and  $1 \leq i \leq N$  such that  $V(x_i) + V(y_i) \geq C_2$  and  $x_i \neq y_i$ . Then with  $\kappa_{2,\varepsilon} = \gamma_{\varepsilon} + (1 - \gamma_{\varepsilon}) \frac{1 + 2\beta C_1}{1 + \beta C_2}$ , we have:

$$\mathbb{E}(d^1(X_{t_{\varepsilon}}^i, Y_{t_{\varepsilon}}^i)) \leqslant \kappa_{2,\varepsilon} d^1(x_i, y_i).$$

*Proof.* Using the Lyapunov property and the fact that  $\gamma_{\varepsilon} \leq \kappa_{2,\varepsilon}$ , we have:

$$\mathbb{E}\left(d^{1}(X_{t_{\varepsilon}}^{i},Y_{t_{\varepsilon}}^{i})\right) \leq 1 + 2\beta C_{1}(1-\gamma_{\varepsilon}) + \beta\gamma_{\varepsilon}\left(V(x_{i})+V(y_{i})\right)$$
$$\leq \kappa_{2,\varepsilon}d^{1}(x_{i},y_{i}) + 1 + 2\beta C_{1}(1-\gamma_{\varepsilon}) - \kappa_{2,\varepsilon} + \beta(\gamma_{\varepsilon}-\kappa_{2,\varepsilon})\left(V(x_{i})+V(y_{i})\right).$$

The fact that  $V(x_i) + V(y_i) \ge C_2$  implies that

$$1 + 2\beta C_1(1 - \gamma_{\varepsilon}) - \kappa_{2,\varepsilon} + \beta(\gamma_{\varepsilon} - \kappa_{2,\varepsilon}) \left( V(x_i) + V(y_i) \right) \leqslant 0$$

and thus the result.

Proof of Theorem 7.1.1. Let  $\mathbf{x}, \mathbf{y} \in D^N$ ,  $\kappa_{\varepsilon} = \kappa_{1,\varepsilon} \vee \kappa_{2,\varepsilon}$ . First suppose that  $\mathbb{1}_{A(\mathbf{x}) > \alpha N} = \mathbb{1}_{A(\mathbf{y}) > \alpha N} = 0$ . We decompose:

$$\mathbb{E}\left(\mathbf{d}\left(\mathbf{X}_{t_{\varepsilon}}, \mathbf{Y}_{t_{\varepsilon}}\right)\right) = \sum_{i/x_{i} \neq y_{i}} \mathbb{E}\left(d^{1}\left(X_{t_{\varepsilon}}^{i}, Y_{t_{\varepsilon}}^{i}\right)\right) + \sum_{i/x_{i} = y_{i} \notin \mathfrak{B}} \mathbb{E}\left(d^{1}\left(X_{t_{\varepsilon}}^{i}, Y_{t_{\varepsilon}}^{i}\right)\right) + \sum_{i/x_{i} = y_{i} \in \mathfrak{B}} \mathbb{E}\left(d^{1}\left(X_{t_{\varepsilon}}^{i}, Y_{t_{\varepsilon}}^{i}\right)\right) + N\left(1 + V_{0}\right)\left(\mathbb{P}\left(A(\mathbf{X}_{t_{\varepsilon}}) > \alpha N\right) + \mathbb{P}\left(A(\mathbf{Y}_{t_{\varepsilon}}) > \alpha N\right)\right).$$
(7.19)

Thanks to Lemmas 7.3.3 and 7.3.4, we have that the first sum is less than  $\kappa_{\varepsilon} d(\mathbf{x}, \mathbf{y})$ . From Lemma 7.3.1, the second term is less than:

$$C_3 m_{\varepsilon} (1 + 2\beta V_0) \mathbf{d}(\mathbf{x}, \mathbf{y}),$$

and the third term is less than:

$$\alpha C_3(1+2\beta V_0)\mathbf{d}(\mathbf{x},\mathbf{y}).$$

Finally, thanks to Lemma 7.2.4, the last term is less than:

$$2N\left(1+V_0\right)q_{\varepsilon}^{N}\mathbf{d}(\mathbf{x},\mathbf{y}) \leqslant \frac{-2\left(1+V_0\right)}{e\ln(q_{\varepsilon})}\mathbf{d}(\mathbf{x},\mathbf{y}).$$

Putting all of this together we get:

$$\mathbb{E}(\mathbf{d}(\mathbf{X}_{t_{\varepsilon}}, \mathbf{Y}_{t_{\varepsilon}})) \leqslant s_{\varepsilon} \mathbf{d}(\mathbf{x}, \mathbf{y})$$

where

$$s_{\varepsilon} = \kappa_{\varepsilon} + C_3 \left( 1 + 2\beta V_0 \right) m_{\varepsilon} + \alpha C_3 (1 + 2\beta V_0) + \frac{-2\left( 1 + V_0 \right)}{e \ln(q_{\varepsilon})}.$$

As  $\varepsilon$  goes to 0,  $s_{\varepsilon}$  goes to  $\frac{1+2\beta C_1}{1+\beta C_2} \wedge 4\beta C_1 + \alpha C_3(1+2\beta V_0) < 1$  because of our choice of constants (7.17).

Now, consider the case where  $\mathbb{1}_{A(\mathbf{x}) > \alpha N} + \mathbb{1}_{A(\mathbf{y}) > \alpha N} > 0$ . Assume that  $\mathbf{x} \neq \mathbf{y}$ , the result being trivial otherwise since the processes stay equal for all times. In that case,  $\mathbf{d}(\mathbf{x}, \mathbf{y}) \ge N(1 + V_0)$  and we simply bound

$$\mathbb{E}\left(d\left(\mathbf{X}_{t_{\varepsilon}}, \mathbf{Y}_{t_{\varepsilon}}\right)\right) \leqslant N(1 + 2\beta V_{0} + (1 + V_{0})q_{\varepsilon}) \leqslant \left(\frac{1 + 2\beta V_{0}}{1 + V_{0}} + q_{\varepsilon}\right) \mathbf{d}(\mathbf{x}, \mathbf{y}),$$

for  $\varepsilon$  small enough. Since  $r_{\varepsilon} := \frac{1+2\beta V_0}{1+V_0} + q_{\varepsilon}$  is strictly less than 1 with our choice of constants (7.18) for  $\varepsilon$  small enough, we conclude that, with  $c = 1 - \sup_{\varepsilon < \varepsilon_0} s_{\varepsilon} \wedge r_{\varepsilon} > 0$  where  $\varepsilon_0$  is small

enough, we have for all  $\mathbf{x}, \mathbf{y} \in D^N$ :

$$\mathbb{E}_{\mathbf{x},\mathbf{y}}\left(\mathbf{d}\left(\mathbf{X}_{t_{\varepsilon}},\mathbf{Y}_{t_{\varepsilon}}\right)\right) \leqslant (1-c)\mathbf{d}(\mathbf{x},\mathbf{y})$$

By conditioning with respect to the initial condition we get that:

$$W_{\mathbf{d}}\left(\mu P_{t_{\varepsilon}}^{N}, \nu P_{t_{\varepsilon}}^{N}\right) \leqslant (1-c)W_{\mathbf{d}}\left(\mu, \nu\right),$$

for all probability measures  $\mu, \nu$  in  $\mathcal{M}^1(D^N)$ , and by iteration:

$$\begin{split} W_{\mathbf{d}}\left(\mu P_{t}^{N},\nu P_{t}^{N}\right) &\leqslant (1-c)^{\lfloor t/t_{\varepsilon} \rfloor} W_{\mathbf{d}}\left(\mu P_{t-\lfloor t/t_{\varepsilon} \rfloor t_{\varepsilon}}^{N},\nu P_{t-\lfloor t/t_{\varepsilon} \rfloor t_{\varepsilon}}^{N}\right) \\ &\leqslant (1+2(\beta+1)V_{0})N(1-c)^{\lfloor t/t_{\varepsilon} \rfloor}. \end{split}$$

We conclude the first point of the theorem using  $1_{x \neq y} \leq d(x, y)$ .

Let's now prove the second point of Theorem 7.1.1.  $\mathcal{M}^1(D^N)$  endowed with the distance **d** is a complete space. The contraction of  $P_{t_{\varepsilon}}^N$  yield the existence and uniqueness of the stationary measure  $\nu_{\infty}^{N,\varepsilon}$ , as well as the exponential convergence of  $\mu P_t^{N,\varepsilon}$  towards  $\nu_{\infty}^{N,\varepsilon}$ . If  $\mu$  is exchangeable, then  $\mu P_t^{N,\varepsilon}$  is exchangeable for all  $t \ge 0$ . The convergence of  $\mu P_t^{N,\varepsilon}$  towards  $\nu_{\infty}^{N,\varepsilon}$  implies that  $\nu_{\infty}^{N,\varepsilon}$  is exchangeable. Now consider an optimal coupling for the distance d of  $\mu P_t^{N,\varepsilon}$  and  $\nu_{\infty}^{N,\varepsilon} P_t^{N,\varepsilon}$ . Using the exchangeability property we have:

$$\begin{aligned} \|\mu P_t^{N,k} - \nu_{\infty}^{N,\varepsilon,k}\|_{TV} &\leq \mathbb{E} \left( \mathbb{1}_{\left(X_t^1,\dots,X_t^k\right) \neq \left(Y_t^1,\dots,Y_t^k\right)} \right) \\ &\leq \sum_{i=1}^k \mathbb{E} \left( \mathbb{1}_{X_t^i \neq Y_t^i} \right) \\ &= k \mathbb{E} \left( \mathbb{1}_{X_t^1 \neq Y_t^1} \right) \\ &\leq \frac{k}{N} \sum_{i=1}^N \mathbb{E} \left( \mathbb{1}_{X_t^i \neq Y_t^i} \right), \end{aligned}$$

and we conclude with the first point of the theorem.

#### 7.3.2 Propagation of chaos

Recall the definition of the empirical measure  $\pi^N$  in (7.4). As said in the introduction, the goal is to get a uniform in time propagation of chaos result. We start from a propagation of chaos result, with a time dependency, from [165]. Their result reads as follows:

**Proposition 7.3.1** ([165], Theorem 1). For all  $\mu_0 \in \mathcal{M}^1(D^N)$ , considering  $(X_t^i)_{t \ge 0}$  the FV process with initial condition  $(X_0^i)$  which is a random variable of law  $\mu_0$ , and  $(X_t)$  the diffusion (7.1), then, for all bounded  $f: D \to \mathbb{R}_+$ , all  $\varepsilon > 0$  and all  $t \ge 0$ :

$$\mathbb{E}\left(\left|\int_{D} f d\pi^{N}(\boldsymbol{X}_{t}) - \mathbb{E}_{\pi^{N}(\boldsymbol{X}_{0})}\left(f(X_{t})|\tau_{\partial D} > t\right)\right|\right) \leq \frac{2(1+\sqrt{2})\|f\|_{\infty}}{\sqrt{N}}\sqrt{\mathbb{E}\left(\frac{1}{\mathbb{P}_{\pi^{N}(\boldsymbol{X}_{0})}\left(\tau_{\partial D} > t\right)^{2}}\right)},$$

where  $\tau_{\partial D}$  is defined in (7.2).

We also need a result on the convergence of the law of the diffusion (7.1), conditioned on survival, towards the QSD  $\nu_{\infty}^{\varepsilon}$ . This is from [35], although the statement is slightly modified to fit our setting.

**Proposition 7.3.2.** Under Assumption 7.1.1, there exists  $\varepsilon_0 > 0$  such that for all compact  $K \subset D$ , and all  $0 < \varepsilon < \varepsilon_0$ , there exists  $C_{\varepsilon}, \tilde{C}_{\varepsilon}, \lambda_{\varepsilon}, \chi_{\varepsilon} > 0$  such that for all  $\mu_0 \in \mathcal{M}^1(D)$  satisfying  $\mu_0(K) \ge 1/4$ :

$$\|\mathbb{P}_{\mu_0}\left(X_t \in \cdot | \tau_{\partial D} > t\right) - \nu_{\infty}^{\varepsilon}\|_{TV} \leqslant C_{\varepsilon} e^{-\chi_{\varepsilon} t},$$

and

$$\mathbb{P}_{\mu_0}\left(\tau_{\partial D} > t\right) \geqslant \tilde{C}_{\varepsilon} e^{-\lambda_{\varepsilon} t}$$

Proof. The process (7.1) satisfies equation (4.7) of [35], with  $D_0 = F$ , where F was defined in the proof of Lemma 7.2.3, some  $\lambda_1$  independent from  $\varepsilon$ , and  $\varphi = V$ . The constant  $\lambda_0$  defined in equation (4.4) of [35] goes to zero as  $\varepsilon$  goes to zero, hence we may chose  $\varepsilon_0$  such that for all  $\varepsilon < \varepsilon_0, \lambda_1 > \lambda_0$ , and assumption of [35, Corollary 4.3] hold true. From [35, Theorem 4.1], this yields the existence of the  $QSD \nu_{\infty}^{\varepsilon}$  and of some function  $\phi : D \to \mathbb{R}^+_+$ , uniformly bounded away from 0 on all compact subsets of D such that for all  $\mu_0 \in \mathcal{M}^1(D)$ :

$$\|\mathbb{P}_{\mu_0}\left(X_t \in \cdot | \tau_{\partial D} > t\right) - \nu_{\infty}^{\varepsilon} \|_{TV} \leqslant C_{\varepsilon} e^{-\chi_{\varepsilon} t} \mu_0(V) / \mu_0(\phi).$$

Since V is bounded, if  $\mu_0(K) \ge 1/4$ , then  $\mu_0(\phi) \ge 1/4 \min_K \phi$ , and we get that

$$\|\mathbb{P}_{\mu_0}\left(X_t \in \cdot | \tau_{\partial D} > t\right) - \nu_{\infty}^{\varepsilon}\|_{TV} \leqslant C_{\varepsilon} e^{-\chi_{\varepsilon} t} \frac{4\|V\|_{\infty}}{\inf_K \phi}.$$

For the second point, write:

$$\mathbb{P}_{\mu_0}\left(\tau_{\partial D} > t\right) \ge \frac{1}{4} \inf_{x \in K} \mathbb{P}_x(\tau_{\partial D} > t).$$

Now fix  $0 < \tilde{U}_1 < \tilde{U}_0$  such that  $K \subset F \cup \left\{ U \leqslant \tilde{U}_0 \right\} =: \tilde{F}$ , and

$$\tilde{\omega}:=\inf_{(F\cup\left\{U\leqslant\tilde{U}_1\right\})^c}|\nabla U|>0$$

. Fix T > 0 such that  $\tilde{U}_0 - \tilde{\omega}^2 T < \tilde{U}_1$ , and some  $\delta > 0$  such that:

$$\delta < \min\left(dist\left(F \cup \left\{U \leqslant \tilde{U}_1\right\}, (F \cup \left\{U \leqslant \tilde{U}_0\right\})^c\right), dist\left(F \cup \left\{U \leqslant \tilde{U}_0\right\}, \mathbb{R}^d \setminus D\right)\right).$$

Write:

$$\mathcal{E} = \left\{ \tau_{\partial D} > T, X_T \in \tilde{F} \right\},\,$$

as well as  $\varphi_t(x)$  for the flow

$$\begin{cases} \varphi_t' = -\nabla U(\varphi_t) \\ \varphi_0(x) = x \end{cases}$$

With our choice of T,  $\tilde{U}_0$  and  $\tilde{U}_1$ , we have that  $\varphi_T \in F \cup \left\{ U \leq \tilde{U}_1 \right\}$  for all  $x \in \tilde{F}$ . We get using Gronwall's Lemma:

$$\sup_{0 \leqslant t \leqslant T} |X_t - \varphi_t| \leqslant e^{\|\nabla^2 U\|_{\infty} T} \sqrt{2\varepsilon} \sup_{0 \leqslant t \leqslant T} |B_t|.$$

If W is a one-dimensional Brownian motion,  $\sup_{0 \le t \le T} W_t$  has the law of |G| where G is a standard Normal random variable. Hence:

$$\mathbb{P}\left(\sup_{0\leqslant t\leqslant T}|X_t-\varphi_t|\geqslant \delta\right)\leqslant 4d\mathbb{P}\left(G\geqslant \frac{\delta e^{-\|\nabla^2 U\|_{\infty}T}}{T\sqrt{2\varepsilon}}\right)\leqslant 4de^{-\frac{b}{\varepsilon}}$$

where  $b = \frac{\delta e^{-\|\nabla^2 U\|_{\infty} T}}{2\sqrt{2T}}$ Now write

$$\mathcal{E}_i = \left\{ \tau_{\partial D} > (i+1)T \text{ and } X_{(i+1)T} \in \tilde{F} \right\}$$

We showed that for  $\varepsilon$  small enough,  $\mathbb{P}(\mathcal{E}_{i+1}|\mathcal{E}_i) \ge 1 - e^{-b/\varepsilon}$ . We also have for our choice of  $\delta$ :

$$\{\tau_{\partial D} < t_{\varepsilon}\} \subset \mathcal{E}^c_{\lceil t/T \rceil}.$$

Hence for all  $x \in \tilde{F}$ :

$$\mathbb{P}_x\left(\tau_{\partial D} > t\right) \geqslant \left(1 - e^{-b/\varepsilon}\right)^{t/T}$$

and thus the result.

In our metastable setting, this already yields propagation of chaos at equilibrium. Indeed, if the FV process starts from its stationary measure, its law won't change. But then from Proposition 7.3.1, the empirical measure of this process is close to the law of the process (7.1)conditioned on survival at time  $t \ge 0$  starting from  $\nu_{\infty}^{N,\varepsilon}$ , which is itself close to the QSD if t is large enough.

**Lemma 7.3.5.** Under Assumption 7.1.1 and 7.1.2, there exists  $\varepsilon_0 > 0$  such that for all  $0 < \varepsilon < \varepsilon_0$  $\varepsilon_0$ , there exists  $C_{\varepsilon}, \eta_{\varepsilon,1} > 0$  such that if  $X_{\infty}$  is a random vector of law  $\nu_{\infty}^{N,\varepsilon}$  on  $D^N$ , then for all bounded function  $f: D \to \mathbb{R}_+$ :

$$\mathbb{E}\left(\left|\int_{D} f d\pi^{N}(\boldsymbol{X}_{\infty}) - \int_{D} f d\nu_{\infty}^{\varepsilon}\right|\right) \leqslant \frac{C_{\varepsilon} \|f\|_{\infty}}{N^{\eta_{\varepsilon,1}}}$$

*Proof.* Assumptions 7.1.1 and 7.1.2 yield the existence of  $\nu_{\infty}^{N,\varepsilon}$ . Introduce the FV process  $(\mathbf{X}_t)_{t\geq 0}$  with initial condition  $\mathbf{X}_{\infty}$ . By definition, the law of  $\mathbf{X}_t$  is  $\nu_{\infty}^{N,\varepsilon}$  for all  $t \geq 0$ . Recall the definition of  $\mathfrak{B}$  and A from (7.14), and set  $K = D \setminus \mathfrak{B}$ . Since  $\nu_{\infty}^{N,\varepsilon}$  is the stationary measure of the FV process. process, we get from Lemma 7.2.4 applied with  $\alpha = 1/2$  that for all  $t \ge 0$ :

$$\mathbb{P}\left(A\left(\mathbf{X}_{t}\right) > N/2\right) \leqslant q_{\varepsilon}^{N}$$

where  $q_{\varepsilon} > 0$  goes to zero as  $\varepsilon$  goes to zero. Recall the definition of  $\lambda_{\varepsilon}$  from Proposition 7.3.2, let  $t = b \ln(N)$  for some  $0 < b < 1/(2\lambda_{\varepsilon})$ , and write  $\mathcal{A} = \{G_1(\mathbf{X}_t) > N/2\}$ . We have that:

$$\mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{\infty}) - \int_{D} f \mathrm{d}\nu_{\infty}^{\varepsilon}\right|\right) = \mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{t}) - \int_{D} f \mathrm{d}\nu_{\infty}^{\varepsilon}\right|\right),$$

and

$$\mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{\infty}) - \int_{D} f \mathrm{d}\nu_{\infty}^{\varepsilon}\right|\right) = \mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{t}) - \int_{D} f \mathrm{d}\nu_{\infty}^{\varepsilon}\right|\right)$$
$$\leq \mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{t}) - \mathbb{E}_{\pi^{N}(\mathbf{X}_{\infty})}(f(X_{t})|\tau_{\partial D} > t)\right|\right)$$

$$+ \mathbb{E}\left(\left|\mathbb{E}_{\pi^{N}(\mathbf{X}_{\infty})}(f(X_{t})|\tau_{\partial D} > t) - \int_{D} f \mathrm{d}\nu_{\infty}^{\varepsilon}\right|\right).$$

On  $\mathcal{G}_1^c$ , we have that  $\pi^N(\mathbf{X}_\infty)(K) \ge 1/2$ . Hence, from Proposition 7.3.1 and 7.3.2, there exists C > 0 such that:

$$\mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{t}) - \mathbb{E}_{\pi^{N}(\mathbf{X}_{\infty})}(f(X_{t})|\tau_{\partial D} > t)\right|\right) \leqslant \frac{C\|f\|_{\infty}}{\sqrt{N}} \frac{1}{N^{-b\lambda_{\varepsilon}} - q_{\varepsilon}^{N}} = \frac{C\|f\|_{\infty}}{N^{1/2 - b\lambda_{\varepsilon}}}.$$

From Proposition 7.3.2, we get that:

$$\mathbb{E}\left(\left|\mathbb{E}_{\pi^{N}(\mathbf{X}_{\infty})}(f(X_{t})|\tau_{\partial D} > t) - \int_{D} f \mathrm{d}\nu_{\infty}^{\varepsilon}\right|\right) \leqslant \left(Ce^{-\chi_{\varepsilon}t} + q_{\varepsilon}^{N}\right) \|f\|_{\infty} = \frac{C\|f\|_{\infty}}{N^{b\chi_{\varepsilon}}}.$$

The fact that we chose  $b < 1/2\lambda_{\varepsilon}$  concludes, as soon as  $\varepsilon$  is small enough so that  $q_{\varepsilon} < 1$ .

Proof of Theorem 7.1.2. Fix some  $\varepsilon > 0$ , some compact  $K \subset D$ ,  $\mu_0 \in \mathcal{M}^1(D)$  such that  $\mu_0(K) \ge 1/2$ , and a random variable  $\mathbf{X}_0$  of law  $\mu_0^{\otimes N}$ . Write:

$$\mathcal{G}_2 = \left\{ \pi^N(\mathbf{X}_0)(K) \ge 1/4 \right\}.$$

We have that  $\mathbb{E}(\pi^{N}(\mathbf{X}_{0})(K)) = \mu_{0}(K)$ , and  $\operatorname{Var}(\pi^{N}(\mathbf{X}_{0})(K)) = \mu_{0}(K)(1 - \mu_{0}(K))/N$ . Hence we have:

$$\mathbb{P}(\mathcal{G}_2^c) \leqslant 4/N.$$

We now fix  $0 < b < 1/(2\lambda_{\varepsilon})$ . For all  $t \leq b \ln(N)$ , from Propositions 7.3.1 and 7.3.2, we get that there exists C > 0 such that:

$$\mathbb{E}\left(\left|\int_{D} f \mathrm{d}\pi^{N}(\mathbf{X}_{t}) - \mathbb{E}_{\pi^{N}(\mathbf{X}_{0})}\left(f\left(X_{t}\right)|\tau_{\partial D} > t\right)\right|\right) \leqslant \frac{C\|f\|_{\infty}}{\sqrt{N}}\sqrt{\frac{1}{N^{-b\lambda_{\varepsilon}} - CN^{-1}}} \leqslant \frac{C\|f\|_{\infty}}{N^{1/2 - b\lambda_{\varepsilon}/2}}.$$

If  $t \ge b \ln(N)$ , we consider a random variable  $(\mathbf{X}_t, \mathbf{X}_\infty)$  which is an optimal coupling of  $\mu_0^{\otimes N} P_t^{N,\varepsilon}$ and  $\nu_\infty^{N,\varepsilon}$  for the distance **d** defined in (7.16). We then bound as follow:

$$\mathbb{E}\left(\left|\int_{D} f d\pi^{N}(\mathbf{X}_{t}) - \mathbb{E}_{\pi^{N}(\mathbf{X}_{0})} \left(f\left(X_{t}\right) | \tau_{\partial D} > t\right)\right|\right) \leq \mathbb{E}\left(\left|\int_{D} f d\pi^{N}(\mathbf{X}_{t}) - \int_{D} f d\pi^{N}(\mathbf{X}_{\infty})\right|\right) \\
+ \mathbb{E}\left(\left|\int_{D} f d\pi^{N}(\mathbf{X}_{\infty}) - \int_{D} f d\nu_{\infty}\right|\right) \\
+ \mathbb{E}\left(\left|\int_{D} f d\nu_{\infty}^{\varepsilon} - \mathbb{E}_{\pi^{N}(\mathbf{X}_{0})} \left(f\left(X_{t}\right) | \tau_{\partial D} > t\right)\right|\right).$$

From the proof of Theorem 7.1.1, the first term is bounded by:

$$\frac{\mathbb{E}(d(\mathbf{X}_t, \mathbf{X}_\infty))}{N} \|f\|_{\infty} \leqslant C(1-c)^{t/t_{\varepsilon}} \|f\|_{\infty} \leqslant CN^{\ln(1-c)/t_{\varepsilon}} \|f\|_{\infty}$$

From Lemma 7.3.5, the second term is bounded by:

$$\frac{C}{N^{\eta_{\varepsilon,1}}} \|f\|_{\infty}.$$

Finally, by Proposition 7.3.2, the third term is bounded by:

$$\left(Ce^{-\chi_{\varepsilon}t} + 2\mathbb{P}(\mathcal{G}_2)\right) \|f\|_{\infty} \leqslant CN^{-\chi_{\varepsilon}b} \|f\|_{\infty},$$

and thus the result.

## 7.4 Establishing Assumption 7.1.2

This section is devoted to the proof of the following:

Lemma 7.4.1. Under Assumptions 7.1.1 and 7.1.3, Assumption 7.1.2 holds.

Proof. 1. Suppose first that d = 1. Under Assumption 7.1.1,  $D = (x_1, x_2)$  for some  $x_1 < x_2$ . Set  $\mathfrak{B}_1 = (x_1, x_1 + \theta) \cup (x_2 - \theta, x_2)$  with  $\theta > 0$  sufficiently small so that  $U' \neq 0$  on  $\mathfrak{B}_1$  (as a consequence of Assumption 7.1.1, U is thus decreasing (resp. increasing) on  $(x_1, x_1 + \theta)$  (resp.  $(x_2 - \theta, x_2)$ )) and so that  $\min(U(x_1 + \theta), U(x_2 - \theta)) > c_*$  (which is possible since  $U(x_1) = U(x_2) = U_0 > c_*$ ). Take any  $a \in (c^*, \min(U(x_1 + \theta), U(x_2 - \theta)))$ . Using the uniqueness of the solution to equation (7.1) we have:

$$\left\{ \exists x \in D \setminus \mathfrak{B}_1, \tau_{\partial D}(X^x) < t_{\varepsilon} \right\} = \left\{ \tau_{\partial D}(X^{x_1 + \theta}) < t_{\varepsilon} \right\} \cup \left\{ \tau_{\partial D}(X^{x_2 - \theta}) < t_{\varepsilon} \right\}.$$

We can conclude with [65, Chapter 6, Theorem 6.2].

2. We now suppose the second condition of Assumption 7.1.3. We set

$$\mathfrak{B}_1 = \{U > (U_0 + U_1)/2\} \cap \mathfrak{B}_1,$$

where  $\tilde{\mathfrak{B}}_1$  is the set from Assumption 7.1.3, and we may chose any  $a \in (c^*, (U_0 - U_1)/2)$ . Fix some

$$\begin{split} &U_2 \in (a + (U_0 + U_1)/2, U_0), \\ &U_3 \in (U_1 + a, (U_0 + U_1)/2), \\ &0 < \gamma < \min(U_3 - U_1 - a, U_2 - (U_0 + U_1)/2 - a), \end{split}$$

see Figure 7.1.

Fix as well some T > 0 (to be chosen large enough below, independently from  $\varepsilon$ ) and for all  $x \in D \setminus \mathfrak{B}_1$  write the event:

$$\mathcal{E}^x = \{ U(X_T^x) < U_3 \text{ and } \forall 0 < t < T, U(X_t^x) < U_2 \}.$$

Let's first show that there exists C > 0 such that  $\mathbb{P}((\mathcal{E}^x)^c) \leq Ce^{-(a+\gamma)/\varepsilon}$  for  $\varepsilon$  small enough and T great enough, using Large Deviation results. In other words, in a fixed time interval, with high probability, starting from a medium energy level at most  $(U_1+U_0)/2$ , the process will stay below the high energy level  $U_2$  and will end below the medium energy level  $U_3$ . According to [65, Chapter 4, Theorem 1.1], the action function of the process (7.1) is:

$$I(\varphi) = \frac{1}{4\varepsilon} \int_0^T |\varphi'_s + \nabla U(\varphi_s)|^2 ds$$
  
=  $\frac{1}{4\varepsilon} \int_0^T \left( |\varphi'_s|^2 + |\nabla U(\varphi_s)|^2 \right) ds + \frac{U(\varphi_T) - U(\varphi_0)}{2\varepsilon}$  (7.20)



Figure 7.1: First,  $U_0$  is the energy level on the boundary of  $\mathcal{D}$ . Then  $U_1$  is taken as low as possible with the constraint that Assumption 7.1.3 holds, namely it is such that there exist a neighborhood  $\mathfrak{B}_1$  of  $\partial \mathcal{D}$  over which  $U \ge U_1$  and  $|\nabla U|$  is lower bounded by a positive constant. Then  $U_2$  (resp.  $U_3$ ) is taken slightly below  $U_0$  (resp.  $(U_1 + U_0)/2$ ) so that the length of the two grey lines in the figure is larger than a, which is itself a constant larger than the critical depth  $c_*$ . The condition that  $c_* < (U_0 - U_1)/2$  means that it is possible to find  $U_2, U_3$  and a to fulfill these constraints.

$$= \frac{1}{4\varepsilon} \int_0^T |\varphi'_s - \nabla U(\varphi_s)|^2 \mathrm{d}s + \frac{U(\varphi_T) - U(\varphi_0)}{\varepsilon}.$$
 (7.21)

For all function  $\varphi : [0,T] \to D$  such that  $\varphi_0 \notin \mathfrak{B}_1$  and there exists  $t \in (0,T)$  such that  $U(\varphi_t) \ge U_2$ , using (7.21), we have:

$$\varepsilon I(\varphi) \ge \frac{1}{4} \int_0^t |\varphi'_s + \nabla U(\varphi_s)|^2 \mathrm{d}s \ge U(\varphi_t) - U(\varphi_0) \ge U_2 - (U_0 + U_1)/2 > a + \gamma.$$

From this we deduce that for  $\varepsilon$  small enough, for all  $x \notin \mathfrak{B}_1$ :

$$\mathbb{P}\left(\exists t \in [0,T], U(X_t^x) \ge U_2\right) \leqslant e^{-(a+\gamma)/\varepsilon}.$$

Second, to bound the probability that  $U(X_T^r) \ge U_3$ , we consider two possible events: either the process stays during the whole interval [0, T] above the energy level  $U_1$  (which is unlikely because it would mean it stays in an unstable region where  $\nabla U$  is non-zero) or the process goes down to  $U_1$  but then climbs back in a time less than T to the level  $U_3$  (which is also unlikely). More precisely, for all functions  $\varphi : [0,T] \to D$  such that  $\varphi_t \notin D \setminus \mathfrak{B}_1$  for all  $t \in [0,T]$ , using (7.20), we have that

$$\varepsilon I(\varphi) \ge T \inf_{\tilde{\mathfrak{B}}_1} |\nabla U|^2 / 4 - U_3 / 2.$$

We chose T great enough so that for all such functions,  $\varepsilon I(\varphi) > a + \gamma$ . Then for  $\varepsilon$  small enough we have that for all  $x \notin \mathfrak{B}_1$ ,

$$\mathbb{P}\left(\exists t \in [0,T], X_t^x \notin \tilde{\mathfrak{B}}_1\right) \ge 1 - e^{-(a+\gamma)/\varepsilon}$$

Next, for all functions  $\varphi: [0,T] \to D$  such that  $\varphi_0 \notin \mathfrak{B}_1$  and  $U(\varphi_T) > U_3$ ,

$$\varepsilon I(\varphi) > U_3 - U_1.$$

Hence, for all x such that  $x \notin \tilde{\mathfrak{B}}_1$  and  $\varepsilon$  small enough :

$$\mathbb{P}(\exists t \in [0,T], U(X_t^x) > U_3) \leqslant e^{-(a+\gamma)/\varepsilon}.$$

From those last two bounds we get for all  $x \notin \mathfrak{B}_1$ :

$$\begin{split} \mathbb{P}\left(U(X_T^x) > U_3\right) \\ \leqslant \mathbb{P}\left(\forall t < T, X_t^x \in \tilde{\mathfrak{B}}_1\right) + \mathbb{P}\left(\exists T > s > t > 0, X_t^x \notin \tilde{\mathfrak{B}}_1, U(X_s^x) > U_3\right) \\ \leqslant 2e^{-(a+\gamma)/\varepsilon}. \end{split}$$

Finally we get for all  $x \notin \mathfrak{B}_1$ :

$$\mathbb{P}\left((\mathcal{E}^x)^c\right) \leqslant 3e^{-(a+\gamma)/\varepsilon}.$$

Up to now, we only have a control for a fixed initial condition x. To tackle simultaneously all initial conditions in  $D \setminus \mathfrak{B}_1$ , we use that, in a time T, two processes which start close

stays close (deterministically). More precisely, fix

$$\delta < \min\left(dist(\left\{U \leqslant U_2\right\}, \mathbb{R}^d \setminus D), dist(\left\{U \leqslant U_3\right\}, \mathfrak{B}_1)\right),$$

and  $\delta' > 0$  such that  $\delta' e^{\|\nabla^2 U\|_{\infty}T} < \delta$ . Fix a family of point  $z_1, \ldots, z_k \notin \mathfrak{B}_1$  such that  $D \setminus \mathfrak{B}_1 \subset \bigcup_{i=1}^k B(z_i, \delta')$ , where B(z, r) is the ball of center z and radius r. Write the event:

$$\mathcal{E} = \{ \forall x \notin \mathfrak{B}_1, \tau_{\partial D}(X^x) > T \text{ and } X_T^x \notin \mathfrak{B}_1 \}.$$

If  $x \notin \mathfrak{B}_1$ , there exists *i* such that  $|x - z_i| < \delta'$ . Gronwall's lemma then classically yields that

$$\sup_{0 \leqslant t \leqslant T} |X_t^x - X_t^{z_i}| \leqslant \delta' e^{\|\nabla^2 U\|_{\infty} T} < \delta.$$

In particular,  $\tau_{\partial D}(X^x) < T$  implies that  $U(X_t^{z_i}) \ge U_2$  for some  $t \in [0,T]$ . Hence we have that:

$$\bigcap_{i=1}^{\kappa} \mathcal{E}^{z_i} \subset \mathcal{E},$$

and for  $\varepsilon$  small enough

$$\mathbb{P}(\mathcal{E}^c) \leqslant 3ke^{-(a+\gamma)/\varepsilon}.$$

Now write

$$\mathcal{E}_{i} = \left\{ \forall x \notin \mathfrak{B}_{1}, \tau_{\partial D}(X^{x}) > (i+1)T \text{ and } X^{x}_{(i+1)T} \notin \mathfrak{B}_{1} \right\}$$

We showed that for  $\varepsilon$  small enough,  $\mathbb{P}(\mathcal{E}_{i+1}|\mathcal{E}_i) \ge 1 - 3ke^{-(a+\gamma)/\varepsilon}$ . We also have that

$$\{\exists x \in D \setminus \mathfrak{B}_1, \tau_{\partial D}(X^x) < t_{\varepsilon}\} \subset \mathcal{E}^c_{\lfloor t_{\varepsilon}/T \rfloor}.$$

Hence:

$$\mathbb{P}\left(\forall x \in D \setminus \mathfrak{B}_1, \tau_{\partial D}(X^x) > t_{\varepsilon}\right) \ge \left(1 - e^{-(a+\gamma)/\varepsilon}\right)^{e^{a/\varepsilon}/T}$$

and thus this probability goes to 1 as  $\varepsilon$  goes to 0.

- 1		1
- 1		
1		
- L		_

,

# Chapter 8

# The Fleming-Viot process in a high-killing regime: the soft killing case.

#### Outline of the current chapter

8.1	Introduction	212
	8.1.1 Motivation	212
	8.1.2 Mathematical setting and notation	213
	8.1.3 Mathematical setting and notation	213
	8.1.4 Fixed number of particles	214
	8.1.5 Increasing number of particles	216
8.2	Proof of Theorem 8.1.1	217
	8.2.1 A time-scale separation result	219
	8.2.2 The initial condition	221
	8.2.3 The Poisson equation	224
	8.2.4 Proof of Theorem 8.1.1	229
8.3	Proof of Theorem 8.1.2	<b>233</b>
8.4	Proof of Theorem 8.1.3	237

**Abstract**: We study the Fleming–Viot particle system in the limit where the selection (or killing) rate tends to infinity. In this regime, and under two different sets of assumptions, we show that the empirical measure of this particle system converges towards a Dirac distribution with a moving mass. This stands in contrast to the mean-field limit, where the empirical measure exhibits a deterministic limit.

# 8.1 Introduction

#### 8.1.1 Motivation

*Fleming–Viot particle systems* refer to population dynamics models in which n individuals are subjected to two mechanisms:

- individuals evolve in some state space as independent copies of a given continuous-time stochastic process (called the *mutation* process);
- they are killed at random times with a rate which depends on their current position, and are then instantaneously duplicated at the position of a uniformly chosen individual among the n-1 remaining ones (this is the *selection* mechanism).

When the mutation process is a random walk on  $\mathbb{Z}^d$  and the killing rate  $\lambda > 0$  is uniform on this space, the empirical measure of this particle system is known to converge, in the  $n \to \infty$ limit and under a diffusive space-time rescaling for the mutation process, to the *Fleming-Viot* superprocess [61], which is a measure-valued process on  $\mathbb{R}^d$  introduced to describe the distribution of alleles in a population with a large number of possible genetic states [57]. In particular, in this regime, the selection mechanism is sped up by a factor n with respect to the mutation process.

The asymptotic behaviour of the system is also of interest without space-time rescaling. Indeed, denoting by  $(X_t)_{t\geq 0}$  the mutation process and  $\tau$  the killing time, it is known that the empirical measure of the system generally converges, when  $n \to \infty$ , to the deterministic quantity  $\mathbb{P}(X_t \in \cdot | \tau > t)$ , which is the law of  $X_t$  conditioned on survival up to time t. This observation, which also holds true for *hard* killing mechanism, that is to say instantaneous killing when the process reaches some subset of the state space, has motivated numerous developments on the study of propagation of chaos and hydrodynamic limit [25, 71, 119, 5, 165, 142, 94] as well as fluctuations [48, 108], for both soft and hard killing mechanisms, in both discrete and continuous spaces.

From a numerical point of view, Fleming–Viot particle systems can be seen as Interacting Particle Systems [46, 42, 149] allowing to approximate the conditional distribution  $\mathbb{P}(X_t \in \cdot | \tau > t)$ , whose evolution is given by a nonlinear Fokker–Planck equation and is therefore nontrivial to study. The  $t \to \infty$  limit of this conditional distribution is called a *quasistationary distribution* (QSD) for the process  $(X_t)_{t\geq 0}$  [39], and plays a prominent role in population dynamics [124], molecular dynamics [104, 49] or Monte Carlo methods [147]. For recent works on the existence and uniqueness of QSDs, as well as convergence of the conditional distribution  $\mathbb{P}(X_t \in \cdot | \tau > t)$ towards the QSD, we refer to [35, 36, 8].

While the numerical interest of Fleming–Viot particle systems lies in the consistency of their  $n \to \infty$  limit, in practice they can only be implemented with finitely many particles. In this context, one may expect their efficiency to depend on the intensity of the killing rate. More precisely, to approximate the QSD of the mutation process by the stationary distribution of the Fleming–Viot particle process, one needs the latter to reach stationarity on a shorter time scale than the typical killing time. This time scale separation is referred to as the *metastability* of the system [143]; in this situation, quantitative estimates for Fleming–Viot particle systems associated with diffusions with hard killing were recently obtained in [97]. The first motivation of the present article is to study the behaviour of Fleming–Viot particle systems in *non-metastable* systems, namely for which killing occurs on a faster time scale than convergence to stationarity for the mutation process.

We shall therefore consider the Fleming–Viot particle system for a mutation process  $(X_t)_{t\geq 0}$ on a countably infinite space D, subjected to killing with a rate  $\lambda : D \to [0, +\infty)$  which may depend on the current position of the process, but which grows to  $+\infty$  uniformly in the position. Before describing our results, let us mention that such a situation may also be relevant from the population dynamics point of view: for instance, the recent work [91] introduced a coupled PDE/ODE model for collective motion of cells driven by nutrient consumption and showed that in the limit where nutrient consumption is infinitely faster than the motion of cells, the cell distribution forms a Dirac mass whose motion is then determined by the nutrient distribution. The situation may also be compared with the well-known *condensation* phenomenon of zerorange processes with decreasing jump rates [74], although the interaction mechanism between particles is different and, in the present situation, leads to a cluster which concentrates *all* the system's mass, and not only a macroscopic fraction.

While our model is mathematically different, our main results show that in the high-killing regime, the Fleming–Viot particle system exhibits a similar behaviour: all particles concentrate in a single cluster whose evolution remains random and generally depends on both the mutation dynamics and the asymptotic ratio of the killing rates between neighbouring sites. This type of results for Fleming–Viot particle system and related models already appears in [33] (see also reference therein). In this work, it was proved that in the small mutation limit, after a time rescaling, the empirical measure of a Fleming–Viot particle process on  $\mathbb{R}$  converges to a Dirac mass, concentrated on the solution to some ODE. Their proof follows an averaging method developed by Kurtz in [103], different from the one used in our present work, see Section 8.2.

#### 8.1.2 Mathematical setting and notation

#### 8.1.3 Mathematical setting and notation

Let D be a finite or countable set,  $q: D \times D \to \mathbb{R}_+$  and  $\lambda: D \to \mathbb{R}_+$ . We suppose that

$$Q:=\sup_{x\in D}\sum_{y\in D}q(x,y)<\infty,\quad \sup_{x\in D}\lambda(x)<\infty.$$

We consider the Fleming-Viot *n*-particle process  $\mathbf{X} = (\mathbf{X}_t)_{t \ge 0}$  in  $D^n$  for a mutation mechanism with jump rates q(x, y), and a selection mechanism with killing rates  $\lambda(x)$ . We are interested in the evolution of the empirical measure  $\pi(\mathbf{X}_t)$  of the system, where  $\pi$  is defined by:

$$\pi(\mathbf{x}) = \frac{1}{n} \sum_{i=1}^{n} \delta_{x_i}.$$
(8.1)

We write  $\mathcal{M}^1(D)$  for the set of probability measures on D and by  $\mathcal{M}^1_n(D) \subset \mathcal{M}^1(D)$  the set of empirical measures of n particles in D:

$$\mathcal{M}_n^1(D) = \left\{ \left(\frac{k_x}{n}\right)_{x \in D} \middle| \forall x \in D, k_x \in \mathbb{N} \text{ and } \sum_{x \in D} k_x = n \right\}.$$

Let us also introduce the spaces:

$$\ell^{1}(\mathcal{M}_{n}^{1}(D)) = \left\{ u : \mathcal{M}_{n}^{1}(D) \to \mathbb{R}, \|u\|_{1} < \infty \right\}, \quad \|u\|_{1} = \sum_{\xi \in \mathcal{M}_{n}^{1}(D)} |u(\xi)|,$$

and

$$\ell^{\infty}(\mathcal{M}^1_n(D)) = \left\{ u : \mathcal{M}^1_n(D) \to \mathbb{R}, \|u\|_{\infty} < \infty \right\}, \quad \|u\|_{\infty} = \sup_{\xi \in \mathcal{M}^1_n(D)} |u(\xi)|.$$

The empirical measure  $(\pi(\mathbf{X}_t))_{t\geq 0}$  of the Fleming-Viot process is a Markov process on the set  $\mathcal{M}_n^1(D)$ , whose infinitesimal generator L acts on  $\ell^{\infty}(\mathcal{M}_n^1(D))$  and reads

$$L = L_m + L_s,$$

where  $L_m$  corresponds to the mutation process: for all  $f \in \ell^{\infty}(\mathcal{M}^1_n(D))$ , for all  $\xi \in \mathcal{M}^1_n(D)$ ,

$$L_m f(\xi) = \sum_{x,y \in D} n\xi(x)q(x,y) \left( f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - f(\xi) \right),$$
(8.2)

and  $L_s$  corresponds to the selection mechanism: for all  $f \in \ell^{\infty}(\mathcal{M}_n^1(D))$ , for all  $\xi \in \mathcal{M}_n^1(D)$ ,

$$L_s f(\xi) = \sum_{x,y \in D} \frac{n^2}{n-1} \xi(x) \lambda(x) \xi(y) \left( f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - f(\xi) \right).$$

For  $u \in \ell^1(\mathcal{M}^1_n(D))$  and  $v \in \ell^\infty(\mathcal{M}^1_n(D))$ , we define the duality bracket by:

$$\langle u, v \rangle = \sum_{\xi \in \mathcal{M}_n^1(D)} u(\xi) v(\xi)$$

The adjoint operators  $L^*$ ,  $L_m^*$  and  $L_s^*$  are defined through this duality bracket and therefore act on  $\ell^1(\mathcal{M}_n^1(D))$ .

As explained in Section 8.1.1, we are interested here in the high killing regime, namely

$$\inf_{D} \lambda \to \infty. \tag{8.3}$$

If the death rates are large with respect to the jump rates and the number of particles, the dynamics of the empirical measure will be dominated by the selection mechanism. If there was only this part, then the empirical measure would become a Dirac measure in finite time. In presence of mutation, and in the limit (8.3), the empirical measure of the Fleming-Viot system will still form a Dirac mass, but this Dirac mass will exhibit some dynamics, in the space

$$\Delta = \{\delta_x, x \in D\} \tag{8.4}$$

of Dirac masses on D.

We present in the next two subsections two results which confirm this intuition, respectively when the number of particles n is fixed, and when the number of particles also tends to infinity together with  $\lambda$ .

#### 8.1.4 Fixed number of particles

In this section, the number of particles n is fixed, and we show that, in the limit (8.3), the trajectories of the empirical process  $(\pi(\mathbf{X}_t))_{t\geq 0}$  converge towards trajectories in the set of Dirac measure. This limiting dynamics can be precisely described in terms of a Markov kernel on D.

In order to make precise the limiting regime (8.3), let us consider a family  $(\lambda_r)_{r \in \mathbb{N}}$  of functions  $D \to \mathbb{R}_+$ , and introduce the notation

$$\underline{\lambda}_r = \inf_{x \in D} \lambda_r(x). \tag{8.5}$$

We set

$$\alpha_{x,y,r} = \frac{\lambda_r(y)}{\lambda_r(x)},\tag{8.6}$$

and we denote by  $\mathbf{X}^r = (\mathbf{X}_t^r)_{t \ge 0}$  the underlying Fleming-Viot *n*-particle processes. We work in this section in the following setting:

**Assumption 8.1.1.** The number n of particles is fixed,  $\lim_{r\to\infty} \underline{\lambda}_r = +\infty$ , and for all  $x, y \in D$ , there exists  $\alpha_{x,y,\infty} \in [0,\infty]$  such that:

$$\lim_{r \to \infty} \alpha_{x,y,r} = \alpha_{x,y,\infty}.$$
(8.7)

As explained in Section 8.1.1, it is expected that  $(\pi(\mathbf{X}_t^r))_{t\geq 0}$  converges to a process  $(\delta_{Y_t})_{t\geq 0}$ where  $(Y_t)_{t\geq 0}$  is a continuous-time Markov chain on D with a jump rate function which is a perturbation of q. Intuitively, the evolution of  $(Y_t)_{t\geq 0}$  is driven by the following mechanism. If all the particles are in  $x \in D$  at a given time t (i.e.  $Y_t = x$ ), then the rate at which one particle goes from x to y is given by nq(x, y). After such a move of one particle from x to y, since the selection mechanism occurs on a much faster time scale than the mutation mechanism, all the particles will concentrate on one of the two states x or y before any other jumps occurs. The probability that a new Dirac mass at y is observed is given by a Gambler's ruin problem, and is equal to  $(\alpha_{x,y,r} - 1)/(\alpha_{x,y,r}^n - 1)$  (this is a consequence of Lemma 8.2.6 below). This justifies the introduction of the new jump rates:

$$\tilde{q}_r(x,y) = \begin{cases} nq(x,y) \frac{\alpha_{x,y,r} - 1}{\alpha_{x,y,r}^n - 1} & \text{if } \alpha_{x,y,r} \neq 1, \\ q(x,y) & \text{otherwise,} \end{cases}$$
(8.8)

as well as their  $r \to \infty$  limit:

$$\tilde{q}_{\infty}(x,y) = \begin{cases} nq(x,y)\frac{\alpha_{x,y,\infty}-1}{\alpha_{x,y,\infty}^n-1} & \text{if } \alpha_{x,y,\infty} \neq 1, \\ q(x,y) & \text{otherwise.} \end{cases}$$
(8.9)

In the above,  $\tilde{q}_{\infty}(x,y) = 0$  in the case  $\alpha_{x,y,\infty} = \infty$ .

The first theorem concerns the limit of the time-marginal distribution of the empirical measure process under Assumption 8.1.1. For any random variable  $\pi$  in  $\mathcal{M}_n^1(D)$ , we identify the probability measure  $\mathcal{L}aw(\pi)$  as an element of  $\ell^1(\mathcal{M}_n^1(D))$  and therefore use the  $\ell^1$  norm to measure distances between such probability measures.

**Theorem 8.1.1.** Under Assumption 8.1.1, for all  $\eta \in \mathcal{M}_n^1(D)$ , there exists a (deterministic) probability measure  $\eta_{\infty} \in \mathcal{M}^1(D)$  described in Lemma 8.2.3, a random variable  $Y_0$  of law  $\eta_{\infty}$ , such that if  $(Y_t)_{t\geq 0}$  is a Markov process on D with jump rate  $(\tilde{q}_{\infty}(x,y))_{(x,y)\in D}$  and initial condition  $Y_0$ , then: for all t > 0,

$$\lim_{t \to \infty} \|\mathcal{L}aw\left(\pi(\mathbf{X}_t^r)\right) - \mathcal{L}aw\left(\delta_{Y_t}\right)\|_1 = 0.$$

Moreover, if  $\pi(\mathbf{X}_0) = \delta_z$  for some  $z \in D$ , then for all T > 0, there exists C > 0 such that if  $(Y_t^r)_{t \ge 0}$  is a Markov process on D with jump rate  $(\tilde{q}_r(x, y))_{(x,y)\in D}$  and initial condition  $Y_0^r = z$ , then:

$$\sup_{t \in [0,T]} \|\mathcal{L}aw\left(\pi(\mathbf{X}_{t}^{r})\right) - \mathcal{L}aw\left(\delta_{Y_{t}^{r}}\right)\|_{1} \leqslant \frac{C}{\underline{\lambda}_{r}}.$$

In the first statement of the theorem, the marginal distribution is taken at time t > 0. Indeed, if the Fleming-Viot particle system is initialized with a distribution which is not a Dirac distribution, namely with particles spread among several sites, then on the short time scale  $1/\underline{\lambda}_r$ , the selection mechanism makes all particles condensate in a single site, whose distribution is precisely the probability measure  $\eta_{\infty}$ . This shows that, in general, the convergence of  $\mathcal{L}aw(\pi(\mathbf{X}_t^r))$  to  $\mathcal{L}aw(\delta_{Y_t})$  cannot hold uniformly on [0, T]. However, if the initial distribution of the Fleming-Viot particle system is already condensed on a single site, then the second statement of the theorem shows that the distance between  $\mathcal{L}aw(\pi(\mathbf{X}_t^r))$  and  $\mathcal{L}aw(\delta_{Y_t^r})$  may be controlled uniformly on [0, T], with a quantitative bound which is obtained by using averaging methods, as presented in [144, Chapter 16]. The remaining distance, between  $\mathcal{L}aw(\delta_{Y_t^r})$  and  $\mathcal{L}aw(\delta_{Y_t})$ , also vanishes (see Lemma 8.2.7), with a rate which depends on the convergence of  $(\alpha_{x,y,r})_{x,y\in D}$ toward  $(\alpha_{x,y,\infty})_{x,y\in D}$ . The proof of Theorem 8.1.1 is given in Section 8.2.

We now state a convergence result for the Fleming-Viot particle system towards the  $\Delta$ -valued dynamics  $(\delta_{Y_t})_{t\geq 0}$  in the space of sample paths. For reasons which are detailed below, the Skorohod topologies for cadlag processes is not adapted to this system, and therefore we need to introduce another topology on the space of sample paths. To proceed, we first introduce the total variation norm  $\|\cdot\|_{TV}$ , and defined by: for  $\nu, \mu \in \mathcal{M}^1(D)$ ,

$$\|\mu - \nu\|_{TV} = \sum_{x \in D} |\mu(x) - \nu(x)|.$$

Note that, since D is a finite or countably infinite set, the weak topology on  $\mathcal{M}^1(D)$  is metrized by the total variation norm. Next, for any T > 0, let  $\mathcal{D}([0,T], \mathcal{M}^1(D))$  be the set of *cadlag* functions from [0,T] to  $\mathcal{M}^1(D)$ . For  $\pi, \tilde{\pi} \in \mathcal{D}([0,T], \mathcal{M}^1(D))$ , we define the  $L_{TV}^1$  distance by:

$$\|\pi - \tilde{\pi}\|_{L^{1}_{TV}} = \int_{0}^{T} \|\pi_{t}(x) - \tilde{\pi}_{t}(x)\|_{TV} dt = \int_{0}^{T} \sum_{x \in D} |\pi_{t}(x) - \tilde{\pi}_{t}(x)| dt.$$
(8.10)

This defines a distance on  $\mathcal{D}([0,T], \mathcal{M}^1(D))$ , and the associated topology coincides in our case with the so-called Meyer-Zheng topology [127]. We then have:

**Theorem 8.1.2.** Under Assumption 8.1.1, for all T > 0, the process  $(\pi(\mathbf{X}_t^r))_{t \in [0,T]}$  converges in distribution, in  $L_{TV}^1$ , to the process  $(\delta_{Y_t})_{t \in [0,T]}$ .

The proof of this theorem is given in Section 8.3. Let us conclude this section with two remarks. First, due to the nature of the process, the convergence in Theorem 8.1.2 does not hold in the Skorokhod topology based on  $\ell^{\infty}$  norms, but in the  $L_{TV}^1$  topology which is a weaker topology. Indeed, when a particle jumps from x to y, one observes motions from x to y and then back to x due to the selection mechanism, over a time-scale of order  $1/\underline{\lambda}_r$ . This implies jumps of non vanishing heights (multiple of 1/n) in the process  $\pi(\mathbf{X}_t^r)$ , and therefore, one cannot expect convergence to hold in the Skorokhod topology. Second, since the mapping  $\pi \in \mathcal{D}([0,T], \mathcal{M}^1(D)) \mapsto \pi_t \in \mathcal{M}^1(D)$  is not continuous with respect to the  $L_{TV}^1$  distance, the convergence in Theorem 8.1.2 does not imply the convergence of the time-marginal distribution from Theorem 8.1.1.

#### 8.1.5 Increasing number of particles

In this section, we are interested in a mean-field limit, namely an asymptotic regime where both the number of particles and the killing rates go to infinity. Therefore, in addition to the *r*-indexed family  $(\lambda_r)_{r\in\mathbb{N}}$  of death rates, we introduce  $(n_r)_{n\in\mathbb{N}} \subset \mathbb{N}^{\mathbb{N}}$  a sequence of numbers of particles, and  $(\mathbf{X}_0^r)_{r\in\mathbb{N}}$  a sequence of initial conditions, such that  $\mathbf{X}_0^r \in D^{n_r}$ . We will again denote by  $\mathbf{X}^r = (\mathbf{X}_t^r)_{t\geq 0}$  the associated Fleming-Viot particle processes.

Compared to the previous setting, we restrict ourselves to a situation where  $\lambda_r(x)$  is a bounded perturbation of  $\underline{\lambda}_r$ . In this regime, the quantities  $\alpha_{x,y,\infty}$  defined in the previous section are all equal to 1, and therefore the jump rates  $\tilde{q}_{\infty}(x, y)$  in (8.9) are equal to q(x, y) whatever the value of n. As a consequence, letting  $n \to \infty$ , one may expect the limiting jump rates to be simply q(x, y), and this will indeed be the case as stated in Theorem 8.1.3 below. Moreover, in order to keep a moving Dirac mass as a limiting process, we impose the rates to go to infinity much faster than the number of particles. Here is the precise setting of this section:

#### Assumption 8.1.2.

$$\sup_{r \in \mathbb{N}} \sup_{x, y \in D} |\lambda_r(x) - \lambda_r(y)| < \infty, \quad \lim_{r \to \infty} \frac{\underline{\lambda}_r}{n_r} = \infty, \quad and \quad \exists \eta \in \mathcal{M}^1(D), \ \pi(\mathbf{X}_0^r) \xrightarrow[r \to \infty]{} \eta \ weakly.$$

**Theorem 8.1.3.** Let us assume that Assumption 8.1.2 holds. Let  $(Y_t)_{t\geq 0}$  be a Markov process on D with jump rate  $(q(x,y))_{(x,y)\in D}$  and initial condition  $\eta$ . Then, the following convergence in law holds: for all t > 0,

$$\pi(\mathbf{X}_t^r) \xrightarrow[r \to \infty]{} \delta_{Y_t}$$

where the convergence in law is considered with respect to the topology defined by the total variation distance on  $\mathcal{M}^1(D)$ . Moreover, if  $\pi(\mathbf{X}_0^r) = \delta_z$  for some  $z \in D$  and for all  $r \in \mathbb{N}$ , then for all T > 0, there exists C > 0 such that for all  $F : \mathcal{M}^1(D) \to \mathbb{R}$  bounded, and for all  $t \in [0,T]$ :

$$\left|\mathbb{E}\left(F\left(\pi(\mathbf{X}_{t}^{r})\right)\right)-\mathbb{E}\left(F\left(\delta_{Y_{t}}\right)\right)\right| \leq C\|F\|_{\infty}\sqrt{\frac{n_{r}}{\underline{\lambda}_{r}}}.$$

The proof is completely different in nature from the one of Theorem 8.1.1, and relies on the Kolmogorov equation associated with the dynamics on  $(\pi(\mathbf{X}_t^r))_{t>0}$ , see Section 8.4.

Let us conclude this section with a few remarks. First, let us emphasize that if the killing rate is not large enough with respect to the number of particles, then there might not be creation of a Dirac mass. For instance, as already mentioned in Section 8.1.1, in the case  $\underline{\lambda}_r = cn_r$ ,  $\pi(\mathbf{X}_t^r)$ converges under a proper rescaling to the Fleming-Viot super-process, see [57]. Second, it is not difficult to check that if D is finite and the initial condition is a Dirac mass, then under Assumption 8.1.2, the convergence is uniform in time. Third, the speed of convergence in Theorem 8.1.3 is not optimal. Indeed, choosing  $n_r = n$  constant (which is allowed in Assumption 8.1.2), Theorem 8.1.3 yields a rate of convergence of order  $1/\sqrt{\Delta_r}$  whereas we know from Theorem 8.1.1 that it is of order  $1/\underline{\lambda}_r$ . Fourth, it would be interesting to generalize the results to the more general situation of arbitrary values of  $\alpha_{x,y,\infty}$  in (8.7), and also to prove in the setting of this section a convergence result in path space similar to Theorem 8.1.2 (these are left for future work).

## 8.2 Proof of Theorem 8.1.1

We assume that Assumption 8.1.1 holds in all this section. For all  $x \in D$ , write:

$$\alpha_r(x) = \frac{\lambda_r(x)}{\underline{\lambda}_r}.$$
(8.11)

For all  $x \in D$ ,  $\alpha_r(x) \ge 1$ , and we write:

$$L = L_m + \underline{\lambda}_r L_0^r, \tag{8.12}$$

where for  $f \in l^{\infty}(\mathcal{M}^1_n(D))$  we denoted:

$$L_0^r f(\xi) = \sum_{x,y \in D} \frac{n^2}{n-1} \xi(x) \alpha_r(x) \xi(y) \left( f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - f(\xi) \right) = \frac{1}{\underline{\lambda}_r} L_s f(\xi).$$

For simplicity, we drop the dependence in r of  $L_0^r$  and we simply write  $L_0$ .

The proof of Proposition 8.1.1 uses averaging methods [144, Chapter 16]. Let us first present the overall idea. Recall the notation (8.1) for the empirical measure. Denote for all  $t \ge 0$  and  $z \in D$ 

$$\pi_t(z) = \pi(\mathbf{X}_t^r)(z), \tag{8.13}$$

and let u be the law of the empirical measure process  $(\pi_t)_{t>0}$ :

$$u(t,\xi) = \mathbb{P}(\pi_t = \xi), \quad \forall \xi \in \mathcal{M}_n^1(D).$$

For all  $t \ge 0$ ,  $u(t, \cdot) \in \ell^1(\mathcal{M}^1_n(D))$ , and we have the Kolmogorov equation:

$$\partial_t u = L^* u,$$

where  $L^*$  denotes the adjoint of the generator L, and acts on  $\ell^1(\mathcal{M}^1_n(D))$ .

Let us present a formal argument in order to identify the limiting dynamics. If we assume an expansion of the form

$$u = u_0 + \frac{1}{\underline{\lambda}_r} u_1 + o\left(\frac{1}{\underline{\lambda}_r}\right),$$

with  $||u_0||_1$  and  $||u_1||_1$  bounded uniformly in  $r \in \mathbb{N}$  and  $t \ge 0$ , then, using (8.12), a formal identification at order  $(\underline{\lambda}_r)^1$  yields that  $u_0$  satisfies

$$L_0^* u_0 = 0$$

which shows that one expects the limiting dynamics to only take Dirac mass values. Indeed, for all  $r \in \mathbb{N}$ ,  $L_0$  is the generator of a Markov process that corresponds to a Fleming-Viot process with no mutation mechanism and only selection dynamics. We will refer in the sequel to this dynamics as the selection dynamics. Under this dynamics, each particle of the Fleming-Viot process is killed at rate  $\alpha_r$ , and then branches uniformly at random over the surviving particles. Let  $(\pi_t^0)_{t\geq 0}$  be a Markov process with this generator  $L_0$ . Every Dirac measure is an absorbing point of such dynamics, and thus  $\text{Span}(\{\mathbb{1}_x, x \in D\}) \subset \text{Ker}L_0^*$ , where

$$\mathbb{I}_x = \delta_{\delta_x} \tag{8.14}$$

is a probability measure on  $\mathcal{M}_n^1(D)$ , and can also be seen as an element of  $\ell^1(\mathcal{M}_n^1(D))$ . Actually, writing (remember the notation (8.4))

$$\tau = \inf\left\{t \ge 0, \pi_t^0 \in \Delta\right\},\tag{8.15}$$

we will show that  $\tau < \infty$  almost surely (as an application of Lemma 8.2.1 below), and this will allow us to prove that, as a bounded operator from  $\ell^1(\mathcal{M}_n^1(D))$  to itself,  $L_0^*$  satisfies

$$\operatorname{Span}\left(\{\mathbb{1}_x, x \in D\}\right) = \operatorname{Ker} L_0^*,$$

see Lemma 8.2.4.

We will then postulate the limiting dynamics on  $u_0$  using our intuition on how Dirac masses

evolve. Let us recall the intuition to derive these limiting dynamics: if the Dirac mass is at  $x \in D$ , at rate nq(x, y) one particle will go from x to y. Since the selection dynamics is on a much faster time scale, we only have to consider this part of the dynamics, and the probability that the Dirac mass goes from x to y is given by the probability  $p_{x,y}$  of a Gambler's ruin, see Lemma 8.2.6. Therefore, the rates at which the Dirac mass will move from x to y is simply  $nq(x, y)p_{x,y}$ .

To formalize this reasoning, after having introduced  $u_0$  following the expected limiting dynamics described above, we will define  $u_1$  as a solution to the Poisson equation

$$\partial_t u_0 - L_m^* u_0 = L_0^* u_1 \tag{8.16}$$

which is the equation formally obtained at order  $(\underline{\lambda}_r)^0$ . The proof will then consist in proving that

$$u - u_0 - \frac{1}{\underline{\lambda}_r} u_1$$

goes to 0 as  $r \to \infty$ .

#### 8.2.1 A time-scale separation result

In the selection/mutation mechanism, there exists a cascade of time-scales. The largest time scale is the mutation one, which is independent of  $r \in \mathbb{N}$ . Besides, we may define an equivalence relation on D by  $x \sim y$  if and only if  $0 < \alpha_{x,y,\infty} < \infty$ . Then, each equivalence class associated with this equivalence relation gives a different time scale, as, if  $\alpha_{x,y,\infty} = \infty$ , then all particles in y will jump away from y to go to a site with a lower death rate before any particle in x have time to jump or move, with probability going to 1 as r goes to infinity. Lemma 8.2.1 below describes this phenomenon, and is a cornerstone in the proofs of the results under Assumption 8.1.1. In particular, if at a given time, the empirical process is not a Dirac mass, it will give us information on the time taken by the process to reach the set of Dirac measures and on how the particles will be redistributed to form a Dirac mass.

**Lemma 8.2.1.** Let  $L_r = a_r L_r^1 + b_r L_r^2$  be a jump Markov process generator on the state space  $\mathcal{M}_n^1(D)$ , such that  $L_r^1$  is a selection dynamics with partial rates:

$$\forall f \in l^{\infty}(\mathcal{M}_{n}^{1}(D)), \ L_{r}^{1}f(\eta) = \sum_{x \in \Theta, y \in D} \frac{n^{2}}{n-1} \eta(x)\gamma_{r}(x)\eta(y) \left( f\left(\eta + \frac{\delta_{y} - \delta_{x}}{n}\right) - f(\eta) \right),$$

for some  $\Theta \subset D$  and  $\gamma_r \ge 1$ , and such that the coefficient of  $L^2_r$  are bounded from above in r:

$$\forall f \in l^{\infty}(\mathcal{M}_{n}^{1}(D)), L_{r}^{2}f(\eta) = \sum_{\xi \in \mathcal{M}_{n}^{1}(D)} \beta_{r}^{2}(\eta,\xi) \left(f(\xi) - f(\eta)\right), \qquad \sum_{\xi \in \mathcal{M}_{n}^{1}(D)} \beta_{r}^{2}(\eta,\xi) \leqslant \bar{\mu},$$

for some  $\bar{\mu} > 0$  and all  $\eta \in \mathcal{M}^1_n(D)$ . Let us assume that

$$\lim_{r \to \infty} \frac{b_r}{a_r} = 0,$$

and let us introduce  $c_r > 0$  such that

$$\lim_{r \to \infty} \frac{c_r}{a_r} = \lim_{r \to \infty} \frac{b_r}{c_r} = 0.$$

Write

$$\mathbb{M}^1 = \left\{ \xi \in \mathcal{M}_n^1(D), L_r^1 f(\xi) = 0, \, \forall f \in l^\infty(\mathcal{M}_n^1(D)), \, \forall r \in \mathbb{N} \right\},\$$

and

$$\tau_1^r = \inf \left\{ t \ge 0, \, \tilde{\pi}_t^r \in \mathbb{M}^1 \right\},\,$$

where  $\tilde{\pi}^r$  is a Markov process with generator  $L_r$ . Denote by  $S_1$  the first jump of the process coming from the  $b_r L_r^2$  part of the generator. Then the following holds:

$$\lim_{r \to \infty} \inf_{\xi \in \mathcal{M}_{n}^{1}(D)} \mathbb{P}_{\xi} \left( \tau_{1}^{r} < c_{r}^{-1} < S_{1} \right) = 1.$$
(8.17)

Moreover, if  $L_r^2 = 0$ , then there exist C, c > 0 such that

$$\sup_{r \in \mathbb{N}} \sup_{\xi \in \mathcal{M}_n^1(D),} \mathbb{P}_{\xi} \left( \tau_1^r \ge t \right) \leqslant C e^{-ca_r t}.$$
(8.18)

This lemma will be applied to the Fleming-Viot generator (8.12), either with  $\Theta = D$  or with  $\Theta$  a subset of D such that the killing rates on  $\Theta$  are much greater than the ones on  $D \setminus \Theta$ , in the sense that for all  $x \in \Theta$ ,  $y \notin \Theta$ ,  $\alpha_{x,y,\infty} = 0$ . In the latter case, the set  $\mathbb{M}^1$  contains the Dirac measures, as well as the empirical measures of n particles with support in  $D \setminus \Theta$ .

*Proof.* Let  $\bar{\pi}$  be a Markov process with generator  $L_r^1$ , defined using the same jump events as  $\tilde{\pi}$ , and

$$\bar{\tau}_1^r = \inf \left\{ t \ge 0, \bar{\pi}_t^r \in \mathbb{M}^1 \right\}$$

On the event  $\tau_1^r < S_1$ , we have  $\tau_1^r = a_r^{-1} \overline{\tau}_1^r$ . Let us show that there exist C, c > 0 such that

$$\sup_{r \in \mathbb{N}} \sup_{\xi \in \mathcal{M}_n^1(D),} \mathbb{P}_{\xi} \left( \bar{\tau}_1^r \ge t \right) \leqslant C e^{-ct}.$$
(8.19)

Fix  $\xi \in \mathcal{M}_n^1(D)$ , write  $D_0 = supp(\xi)$  and

$$\gamma_{x} = \min\left\{\gamma_{r}(x), x \in D_{0} \cap \Theta\right\}$$

Since  $\gamma_r(x) \ge 1$  for all  $x \in D$ ,  $\underline{\gamma}_r \ge 1$  for all  $r \in \mathbb{N}$ . Let  $\underline{x}_r \in D$  be such that  $\underline{\gamma}_r = \gamma_r(\underline{x}_r)$  if  $D_0 \subset \Theta$ , and let  $\underline{x}_r \in D_0 \setminus \Theta$  otherwise. Let **X** be a process whose empirical measure is given by  $\overline{\pi}$ , and let  $i_0 \in [\![1, n]\!]$  be such that  $X_0^{i_0} = \underline{x}_r$ . Denote

$$I = \left\{ i, X_0^i \in D_0 \setminus \Theta \right\} \cup \left\{ i_0 \right\}, \quad n_0 = \left| I \setminus \{i_0\} \right|,$$

and write:

 $\mathcal{A}_r = \left\{ \text{On } [0, 1/\underline{\gamma}_r], X^i, i \in I, \text{ do not die, and all other die once and branch on } X^{i_0} \right\}.$ 

We have that  $\mathcal{A}_r \subset \{\bar{\tau}_1^r < 1\}$ . Since the number of death events for  $(X_s^i)_{0 \leq s \leq 1/\underline{\gamma}_r}$  is upper (resp. lower) bounded by a Poisson random variable with parameter  $\underline{\gamma}_r$  if  $i \in I$  (resp.  $i \notin I$ ), we have that:

$$\mathbb{P}_{\xi}\left(\mathcal{A}_{r}\right) \geqslant \frac{e^{-n}\left(1-e^{-1}\right)^{n-n_{0}}}{n^{n-n_{0}}},$$

and hence, we get that

$$\inf_{r \in \mathbb{N}} \inf_{\xi \in \mathcal{M}_n^1(D)} \mathbb{P}_{\xi} \left( \bar{\tau}_1^r < 1 \right) \ge 1 - \omega,$$

for some  $\omega \in [0,1)$ . Using the Markov property at time t-1, this implies that for all  $r \in \mathbb{N}$ ,  $\xi \in \mathcal{M}_n^1(D)$ , and  $t \ge 1$ :

$$\mathbb{P}_{\xi}\left(\bar{\tau}_{1}^{r} \ge t\right) \leqslant \omega \mathbb{P}_{\xi}\left(\bar{\tau}_{1}^{r} \ge t-1\right),$$

and this implies

$$\sup_{r \in \mathbb{N}} \sup_{\xi \in \mathcal{M}_n^1(D)} \mathbb{P}_{\xi} \left( \bar{\tau}_1^r > t \right) \leqslant \omega^{\lfloor t \rfloor} \leqslant C e^{-ct}$$

with  $c = -\ln(\omega) > 0$  and  $C = \omega^{-1}$ . Writing  $\beta_r^2(\eta) = \sum_{\xi \in \mathcal{M}_n^1(D)} \beta_r^2(\eta, \xi)$ ,  $S_1$  is defined as:

$$S_1 = \inf\left\{t \ge 0, E_1 \leqslant \int_0^t b_r \beta_r^2(\tilde{\pi}_s^r) \mathrm{d}s\right\} \ge \inf\left\{t \ge 0, E_1 \leqslant b_r \bar{\mu}t\right\} = b_r^{-1} \tilde{S}_1,$$

where  $E_1$  is an exponential random variable with parameter 1. Hence,  $S_1$  is bounded from below by  $b_r^{-1}\tilde{S}_1$ , where  $\tilde{S}_1$  is an exponential random variable with parameter  $\bar{\mu}$ . Finally, using the bound (8.19) we have:

$$\mathbb{P}\left(\tau_{1}^{r} < c_{r}^{-1} < S_{1}\right) = \mathbb{P}\left(a_{r}^{-1}\bar{\tau}_{1}^{r} < c_{r}^{-1} < S_{1}\right)$$

$$\geqslant 1 - \mathbb{P}\left(\bar{\tau}_{1}^{r} \geqslant \frac{a_{r}}{c_{r}}\right) - \mathbb{P}\left(S_{1} < c_{r}^{-1}\right)$$

$$\geqslant 1 - \mathbb{P}\left(\bar{\tau}_{1}^{r} \geqslant \frac{a_{r}}{c_{r}}\right) - \mathbb{P}\left(\tilde{S}_{1} < \frac{b_{r}}{c_{r}}\right) \xrightarrow[r \to \infty]{} 1,$$

and hence the limit (8.17). In the case where  $L_r^2 = 0$ , Equation (8.19) yields:

$$\mathbb{P}_{\xi}\left(\tau_{1}^{r} \geqslant t\right) = \mathbb{P}_{\xi}\left(\bar{\tau}_{1}^{r} \geqslant a_{r}t\right) \leqslant Ce^{-ca_{r}t},$$

which concludes the proof.

#### 8.2.2 The initial condition

Now, we aim to address the issue of the initial condition. Given some initial condition  $\eta \in \mathcal{M}_n^1(D)$ , write  $D_0 = supp(\eta)$ , where

$$supp(\eta) = \left\{ x \in D, \eta(x) > 0 \right\},\$$

and

$$\Lambda = \left\{ x \in D_0, \min_{y \in D_0} \alpha_{x,y,\infty} > 0 \right\}.$$
(8.20)

This set  $\Lambda$  correspond to the subset of  $D_0$  where, asymptotically in r, the order of magnitude of the killing rate is minimal. Thanks to Lemma 8.2.1, we will be able to show that in the limit (8.3), all particles from  $D_0 \setminus \Lambda$  will be redistributed instantaneously (ie on a time scale faster than  $1/\inf_{D_0} \lambda_r$ ) in  $\Lambda$ , with no movement from the particles in  $\Lambda$ . Moreover, the law of the configuration obtained in  $\Lambda$  after this instantaneous redistribution is given by a Polya's urn problem, with  $|\Lambda|$  possible colors and initial condition  $n\eta|_{\Lambda}$ , after  $n\eta(D_0 \setminus \Lambda)$  draws. On the time scale  $1/\inf_{D_0} \lambda_r$ , this measure will then form a Dirac mass according to the probability given by commitor functions, that we are now going to define.

Recall that  $\pi^0$  denotes a Markov process with generator  $L_0$ , and the definition (8.15) of  $\tau$ . Let us apply Lemma 8.2.1, with  $L_r^2 = 0$ ,  $a_r = 1$ ,  $\Theta = supp(\pi_0^0)$ ,  $\gamma_r = \alpha_r$ , so that  $L_r^1 = L_0$  and

-		

 $\tau_r^1 = \tau$ . Equation (8.18) then yields that  $\tau < \infty$  almost surely. Denote for  $\eta \in \mathcal{M}_n^1(D)$ :

$$\psi_x^r(\eta) = \mathbb{P}_\eta \left( \pi_\tau^0 = \delta_x \right). \tag{8.21}$$

The functions  $\psi_x^r$  are the so called committor functions, and [140, Theorem 3.3.1.] yields that they satisfy:

$$\psi_x^r(\delta_x) = 1, \ \psi_x^r(\delta_y) = 0 \ \forall y \neq x, \ L_0 \psi_x^r(\xi) = 0, \ \forall \xi \in \mathcal{M}_n^1(D) \setminus \Delta.$$
(8.22)

Remark that for all  $\xi \in \Delta$ ,  $f \in \ell^{\infty} (\mathcal{M}_n^1(D))$ ,  $L_0 f(\xi) = 0$ , so that we actually have  $L_0 \psi_x^r(\xi) = 0$ , for all  $\xi \in \mathcal{M}_n^1(D)$ . The next lemma proves that the committor functions are the unique solution to (8.22) in  $\ell^{\infty} (\mathcal{M}_n^1(D))$ .

**Lemma 8.2.2.** Let  $\Theta \subset D$  and  $\gamma : \Theta \to [1, +\infty)$ . Let  $L^1$  be defined by:  $\forall f \in \ell^{\infty} (\mathcal{M}^1_n(\Theta))$ ,

$$L^{1}f(\eta) = \sum_{x,y\in\Theta} \frac{n^{2}}{n-1} \eta(x)\gamma(x)\eta(y) \left( f\left(\eta + \frac{\delta_{y} - \delta_{x}}{n}\right) - f(\eta) \right).$$

For all  $x \in \Theta$ , if  $f \in \ell^{\infty} \left( \mathcal{M}_n^1(\Theta) \right)$  satisfies

$$f(\delta_x) = 1, \ f(\delta_y) = 0 \ \forall y \neq x, \ L^1 f(\xi) = 0, \ \forall \xi \in \mathcal{M}_n^1(\Theta) \setminus \Delta,$$
(8.23)

then for all  $\xi \in \mathcal{M}_n^1(\Theta)$ 

$$f(\xi) = \mathbb{P}_{\xi} \left( \pi_{\tau_1}^1 = \delta_x \right),$$

where  $\tau_1 = \inf \{t \ge 0, \pi_t^1 \in \Delta\}$  and  $\pi^1$  is a Markov process with generator  $L^1$  and values in  $\mathcal{M}_n^1(\Theta)$ .

In particular,  $\psi_x^r$  defined by (8.21) are the unique solution to (8.22) in  $\ell^{\infty}(\mathcal{M}^1_n(D))$ .

*Proof.* Let f be a solution to Equation (8.23). Let  $\xi \in \mathcal{M}_n^1(\Theta)$  and  $\pi^1$  be a Markov process with generator  $L^1$  and initial condition  $\xi$ . The map  $t \mapsto f(\pi_{\tau_1 \wedge t}^1)$  is a martingale, and hence for all  $t \ge 0$ ,

$$\mathbb{E}_{\xi}\left(f(\pi^{1}_{\tau_{1}\wedge t})\right) = f(\xi). \tag{8.24}$$

Equation (8.18) from Lemma 8.2.1 yields that  $\tau_1 < \infty$  almost surely, and since f is bounded, letting t go to infinity in (8.24) yields that  $f(\xi) = \mathbb{P}_{\xi} \left( \pi_{\tau_1}^1 = \delta_x \right)$ .

We are now able to prove

**Lemma 8.2.3.** For all  $\eta \in \mathcal{M}_n^1(D)$ , there exists  $\eta_\infty \in \mathcal{M}^1(D)$  such that if  $\pi^0$  is a Markov process with generator  $L^0$  and initial condition  $\eta$ , and  $Y_0$  is a random variable with law  $\eta_\infty$ , then

$$\lim_{r \to \infty} \|\mathcal{L}aw(\pi^0_{\tau}) - \mathcal{L}aw(\delta_{Y_0})\|_{TV} = 0.$$

In the statement of Theorem 8.1.1,  $\eta_{\infty}$  is the law of the initial condition  $Y_0$ .

*Proof.* Fix  $\eta \in \mathcal{M}^1_n(D)$  and write  $\bar{\eta}_r = \mathcal{L}aw(\pi^0_\tau) \in \mathcal{M}^1(\Delta)$ . Denote  $D_0(\eta) = supp(\eta)$  and

$$\underline{\alpha}_r = \inf_{x \in D_0} \alpha_r(x),$$

where  $\alpha_r(x)$  is defined in (8.11). Recall the definition of  $\Lambda$  from (8.20). We distinguish two cases: if  $\Lambda = D_0$ , then we can find the limit of  $\bar{\eta}_r$  when  $r \to \infty$  using the fact that  $\bar{\eta}_r(\delta_x) = \psi_x^r(\eta)$  and that  $(\psi_x^r(\eta))_{\eta \in \mathcal{M}_n^1(\Lambda)}$  is the unique solution to a Dirichlet equation on a finite set. If  $\Lambda \neq D_0$ , then using Lemma 8.2.1, we show that all particles from  $D_0 \setminus \Lambda$  will first be distributed on  $\Lambda$ , without any particles from  $\Lambda$  moving, and then we can apply the argument from the case  $D_0 = \Lambda$ .

Let us suppose that  $\Lambda = D_0$ . Write  $L_0^{D_0,r}$  the re-normalised generator of the selection dynamics

$$L_0^{D_0,r}f(\xi) = \sum_{x,y\in D_0} \frac{\alpha_r(x)}{\underline{\alpha}_r} \xi(x)\xi(y) \left( f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - f(\xi) \right),$$

defined for  $f: \mathcal{M}^1_n(D_0) \to \mathbb{R}$ . Denote also

$$\alpha_{D_0,\infty}(x) = \lim_{r \to \infty} \alpha_r(x) / \underline{\alpha}_r = \frac{1}{\min_{y \in D_0} \alpha_{x,y,\infty}}$$

which is well defined and lies in  $[1, \infty)$  in the case  $D_0 = \Lambda$ , as well as

$$L_0^{D_0,\infty}f(\xi) = \sum_{x,y\in D_0} \alpha_{D_0,\infty}(x)\xi(x)\xi(y)\left(f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - f(\xi)\right).$$

 $L_0^{D_0,\infty}$  is the generator of a selection dynamics with death rates  $\alpha_{\infty,D_0}$ . We have that

$$\bar{\eta}_r(\delta_x) = \psi_x^r(\eta),$$

and for all  $x \in D_0$  and  $\xi \in \mathcal{M}^1_n(D_0)$ ,

$$L_0^{D_0,r}\psi_x^r(\xi) = 0. ag{8.25}$$

Since  $D_0 \times \mathcal{M}_n^1(D_0)$  is a finite set, we may consider  $(\psi_x^{\infty}(\xi))_{x \in D_0, \xi \in \mathcal{M}_n^1(D_0)}$  a sub-sequential limit of  $(\psi_x^r(\xi))_{x \in D_0, \xi \in \mathcal{M}_n^1(D_0)}$ . Passing to the limit in equation (8.25), we get that for all  $x \in D_0$  and  $\xi \in \mathcal{M}_n^1(D_0)$ :

$$L_0^{D_0,\infty}\psi_x^\infty(\xi) = 0,$$

as well as  $\psi_x^{\infty}(\delta_y) = \mathbb{1}_{x \neq y}$ . Lemma 8.2.2 then implies that

$$\psi_x^{\infty}(\xi) = \mathbb{P}_{\xi} \left( \pi_{\tau}^{0,\infty} = \delta_x \right),$$

where  $\pi^{0,\infty}$  is a Markov process with generator  $L_0^{D_0,\infty}$ . Hence there is a unique sub-sequential limit, and  $\psi_x^r(\xi)$  converge for all  $\xi \in \mathcal{M}_n^1(D_0)$  and  $x \in D_0$ , and this is particularly true for  $\xi = \eta$ .

We now suppose that  $\Lambda \neq D_0$ . In this case, for all  $x \in D_0 \setminus \Lambda$ ,  $\alpha_{D_0,\infty} = \infty$  and  $L^{D_0,\infty}$  is ill-defined. To resort to the previous case, write

$$S_1^{\Lambda} = \inf \{ t \ge 0, \text{ one particle starting from } \Lambda \text{ dies} \},\$$

and

$$\tau_{\mathcal{M}_n^1(\Lambda)} = \inf \left\{ t \ge 0, \pi_t^0 \in \mathcal{M}_n^1(\Lambda) \right\}.$$

Using the notation of Lemma 8.2.1, write

$$\Theta = D_0 \setminus \Lambda, \quad a_r = \inf \left\{ \alpha_r(x) / \underline{\alpha}_r, x \in D_0 \setminus \Lambda \right\}, \quad b_r = 1, \quad L_r^2 = L_0^{D_0, r}.$$

In this case, we have  $\mathbb{M}^1 = \Delta \cup \mathcal{M}^1_n(\Lambda)$ , and because  $\Lambda \neq D_0$ ,  $a_r \xrightarrow[r \to \infty]{} \infty$ , and Lemma 8.2.1

(with  $c_r = \sqrt{a_r}$ ) yields that:

$$\lim_{r \to \infty} \mathbb{P}_{\eta} \left( S_1^{\Lambda} \leqslant \tau_{\mathcal{M}_n^1(\Lambda)} \right) = 0.$$

Now, using the strong Markov property at time  $\tau_{\mathcal{M}_n^1(\Lambda)}$ , write:

$$\mathbb{P}_{\eta}\left(\pi_{\tau}^{0}=\delta_{x}\right)=\sum_{\xi\in\mathcal{M}_{n}^{1}(\Lambda)}\mathbb{P}_{\eta}\left(S_{1}^{\Lambda}>\tau_{\mathcal{M}_{n}^{1}(\Lambda)}\right)\mathbb{P}_{\eta}\left(\pi_{\tau_{\mathcal{M}_{n}^{1}(\Lambda)}}^{0}=\xi\left|S_{1}^{\Lambda}>\tau_{\mathcal{M}_{n}^{1}(\Lambda)}\right)\mathbb{P}_{\xi}\left(\pi_{\tau}^{0}=\delta_{x}\right)+R_{r},$$

where a rest satisfying

$$R_r \leqslant \mathbb{P}_\eta \left( S_1^\Lambda \leqslant \tau_{\mathcal{M}_n^1(\Lambda)} \right) \xrightarrow[r \to \infty]{} 0.$$

From the case  $D_0 = \Lambda$  we get that for all  $\xi \in \mathcal{M}_n^1(\Lambda)$ ,  $\mathbb{P}_{\xi} \left( \pi_{\tau}^0 = \delta_x \right)$  admits a limit as r goes to infinity. In addition,

$$\mathbb{P}_{\eta}\left(\pi^{0}_{\tau_{\mathcal{M}_{n}^{1}(\Lambda)}}=\pi\left|S_{1}^{\Lambda}>\tau_{\mathcal{M}_{n}^{1}(\Lambda)}\right.\right)$$

corresponds to the law of a Polya urn Markov chain with with color indexed by  $\Lambda$ , initial condition  $n\eta|_{\Lambda}$ , after  $|D_0 \setminus \Lambda|$  draws, and is hence independent of r. This concludes the convergence of  $\mathbb{P}_{\eta}(\pi_{\tau}^0 = \delta_x)$  towards some  $\bar{\eta}_{\infty} \in \mathcal{M}^1(\Delta)$ . If  $Y_0 \in D$  is such that  $\mathcal{L}aw(\delta_{Y_0}) = \bar{\eta}_{\infty}, \eta_{\infty}$  is then the law of  $Y_0$ .

#### 8.2.3 The Poisson equation

Let us now consider the Poisson equation (8.16). Before showing existence and uniqueness of a solution to this equation, let us describe the kernel of  $L_0^*$  and  $L_0$ . Although this is not strictly necessary for the proof of Theorem 8.1.1, this makes the functional setting clearer. To proceed, let us recall that  $L_0$  is defined as an operator of  $\ell^{\infty}(\mathcal{M}_n^1(D))$ , and  $L_0^*$  as an operator of  $l^1(\mathcal{M}_n^1(D))$ . Define the weak-\* topology on  $\ell^{\infty}(\mathcal{M}_n^1(D))$  by the fact that a sequence of elements  $(w_k)_{k\geq 1}$  weak-\* converges to w if  $\langle v, w_k \rangle \to \langle v, w \rangle$  for any  $v \in \ell^1(\mathcal{M}_n^1(D))$ . Then  $l^1(\mathcal{M}_n^1(D))$  equipped with the (strong) topology induced by the norm  $\|\cdot\|_1$ , and  $\ell^{\infty}(\mathcal{M}_n^1(D))$  equipped with the weak-\* topology, are each other's topological dual. Let us now write

$$(\mathrm{Ker}L_0)^{\perp} = \left\{ v \in \ell^1 \left( \mathcal{M}_n^1(D) \right), \, \langle v, w \rangle = 0, \, \forall w \in \mathrm{Ker}L_0 \right\},\$$

recall the definition of the commitor functions from (8.21), and denote by  $\overline{A}$  the closure of a set A for the topology at hand.

Lemma 8.2.4. We have

$$(\operatorname{Ker} L_0)^{\perp} = \{ v \in \ell^1 \left( \mathcal{M}_n^1(D) \right), \, \langle v, \psi_x^r \rangle = 0, \, \forall x \in D \}$$

as well as

$$\overline{\operatorname{Span}(\mathbb{1}_x, x \in D)} = \operatorname{Ker} L_0^*$$

However, the inclusion

$$\operatorname{Span}(\psi_x, x \in D) \subset \operatorname{Ker}L_0$$

is strict if D is infinite, and the kernel of  $L_0$  is given by

$$\overline{\operatorname{Span}(\psi_x, x \in D)}^{w*} = \operatorname{Ker} L_0,$$

where  $\overline{A}^{w*}$  denotes the closure for the weak-\* topology of a set  $A \subset \ell^{\infty}(\mathcal{M}^{1}_{n}(D))$ .

*Proof.* For all  $x \in D$ , Lemma 8.2.2 yields that  $\psi_x^r \in \text{Ker}L_0$ , so that:

$$(\operatorname{Ker} L_0)^{\perp} \subset \{ v \in \ell^1 \left( \mathcal{M}_n^1(D) \right), \, \langle v, \psi_x^r \rangle = 0, \, \forall x \in D \}$$

Let  $v \in \ell^1(\mathcal{M}_n^1(D))$  such that  $\langle v, \psi_x^r \rangle = 0$  for all  $x \in D$ , and  $w \in \operatorname{Ker} L_0$  (and thus  $w \in \ell^\infty(\mathcal{M}_n^1(D))$ ). The map  $t \mapsto w(\pi_{t\wedge\tau}^0)$  is a martingale, and hence for all  $t \ge 0$ ,

$$w(\xi) = \mathbb{E}_{\xi} \left( w(\pi_{\tau \wedge t}^0) \right).$$

Since w is bounded, and  $\tau < \infty$  almost surely, letting t go to infinity and the dominated convergence theorem yields that for all  $\xi \in \mathcal{M}_n^1(D)$ :

$$w(\xi) = \sum_{x \in D} w(\delta_x) \psi_x(\xi), \qquad (8.26)$$

and the sum is actually finite because  $\psi_x(\xi) = 0$  if  $x \notin supp(\xi)$ . We have:

$$\sum_{\xi \in \mathcal{M}_n^1(D)} \sum_{x \in D} |w(\delta_x)v(\xi)\psi_x(\xi)| = \sum_{\xi \in \mathcal{M}_n^1(D)} |v(\xi)| \sum_{x \in D} |w(\delta_x)|\psi_x^r(\xi) \leqslant ||v||_1 ||w||_\infty < \infty.$$

Hence, the following computation holds:

$$\langle v, w \rangle = \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \sum_{x \in D} w(\delta_x) \psi_x(\xi) = \sum_{x \in D} w(\delta_x) \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \psi_x(\xi) = \sum_{x \in D} w(\delta_x) \langle v, \psi_x^r \rangle = 0,$$

which yields that  $v \in (\text{Ker}L_0)^{\perp}$ , and concludes the first point.

We already saw in the introduction of Section 8.2 that  $\text{Span}(\mathbb{1}_x, x \in D) \subset \text{Ker}L_0^*$ . Since  $L_0^*$  is a bounded operator,  $\text{Ker}L_0^*$  is closed and

$$\operatorname{Span}(\mathbb{1}_x, x \in D) \subset \operatorname{Ker} L_0^*.$$

For  $v \in \text{Ker}L_0^*$ , the map

$$t \mapsto \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi} \left( \pi_t^0 = \eta \right)$$

is constant for all  $\eta \in \mathcal{M}_n^1(D)$ . Write for all  $\eta \in \mathcal{M}_n^1(D)$ :

$$\mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) - \mathbb{P}_{\xi} \left( \pi_{\tau}^0 = \eta \right) = \mathbb{P}_{\xi} \left( \pi_s^0 = \eta, \tau > s \right) + \mathbb{P}_{\xi} \left( \pi_s^0 = \eta, \tau < s \right) - \mathbb{P}_{\xi} \left( \pi_{\tau}^0 = \eta \right) \\ = \mathbb{P}_{\xi} \left( \pi_s^0 = \eta, \tau > s \right) - \mathbb{P}_{\xi} \left( \pi_{\tau}^0 = \eta, \tau > s \right),$$

which yields

$$\sum_{\eta \in \mathcal{M}_n^1(D)} \left| \mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) - \mathbb{P}_{\xi} \left( \pi_\tau^0 = \eta \right) \right| \leq 2 \mathbb{P}_{\xi} \left( \tau > s \right).$$
(8.27)

Using the notations of Lemma 8.2.1, write:

$$\Theta = D, \quad a_r = 1, \quad \gamma_r = \alpha_r, \quad L_r^2 = 0.$$

Lemma 8.2.1 then yields that there exist C, c > 0 such that for all  $s \ge 0$ 

$$\sup_{r \in \mathbb{N}} \sup_{\xi \in \mathcal{M}_n^1(D),} \mathbb{P}_{\xi}(\tau > s) \leqslant Ce^{-cs}.$$
(8.28)

The bound (8.27) and (8.28) yields that we may let t go to infinity and get:

$$v = \sum_{x \in D} v(\delta_x) \mathbb{1}_x.$$

Since this sum converges in  $l^1(\mathcal{M}^1_n(D))$  as soon as  $v \in l^1(\mathcal{M}^1_n(D))$ ,  $v \in \overline{\operatorname{Span}(\mathbb{1}_x, x \in D)}$ , which concludes the second point.

Define for all  $\xi \in \mathcal{M}^1_n(D)$ ,

$$f(\xi) = \sum_{x \in D} \psi_x(\xi).$$

In this sum, there are at most n non-vanishing terms, and thus  $f \in \ell^{\infty}(\mathcal{M}_{n}^{1}(D))$ . It is easily seen that  $L_{0}f = 0$ . If  $g = \sum_{x \in D_{0}} a_{x}\psi_{x}(\xi)$  for some finite subset  $D_{0} \subset D$  and  $a_{x} \in \mathbb{R}$ , we have that  $\|f - g\|_{\infty} \ge 1$ , yielding that

$$f \in \operatorname{Ker} L_0 \setminus \overline{\operatorname{Span}(\mathbb{1}_x, x \in D)}.$$

However, for all  $w \in \text{Ker}L_0$ , the sum (8.26) converges for the weak-\* topology, which proves the last point.

We are now ready to solve the Poisson equation (8.16), which relies on the next statement. For any  $v \in \ell^1(\mathcal{M}^1_n(D))$  and  $s \ge 0$ , let us introduce the notation

$$ve^{sL_0}(\eta) = \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi}(\pi_s^0 = \eta),$$
(8.29)

which defines an element of  $\ell^1(\mathcal{M}^1_n(D))$ .

**Lemma 8.2.5.** For all  $v \in (\text{Ker}L_0)^{\perp} (\subset \ell^1(\mathcal{M}_n^1(D)))$ , there exists a unique  $w \in (\text{Ker}L_0)^{\perp}$  such that

$$L_0^* w = v, (8.30)$$

given by:

$$w = -\int_0^\infty v e^{sL_0} \, ds$$

Moreover, there exists C > 0 independent of r and v such that

 $||w||_1 \leq C ||v||_1.$ 

*Proof.* Let us start with the existence. Write, for  $\eta \in \mathcal{M}^1_n(D)$ 

$$w(\eta) = -\left(\int_0^\infty v e^{sL_0} \mathrm{d}s\right)(\eta) = -\int_0^\infty \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi}\left(\pi_s^0 = \eta\right) \mathrm{d}s.$$

We first show that w is well-defined. Recall the stopping time  $\tau$  defined in (8.15). Since  $v \in (\text{Ker}L_0)^{\perp}$ , Lemma 8.2.4 yields that  $\langle v, \psi_x^r \rangle = 0$  for all  $x \in D$ , and we have

$$\sum_{\xi\in\mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi} \left( \pi_{\tau}^0 = \delta_x \right) = \sum_{\xi\in\mathcal{M}_n^1(D)} v(\xi) \psi_x^r(\xi) = 0.$$

Since, if  $\eta \notin \Delta$ ,  $\mathbb{P}_{\xi} (\pi_{\tau}^0 = \eta) = 0$ , we have that

$$\sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) = \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \left( \mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) - \mathbb{P}_{\xi} \left( \pi_\tau^0 = \eta \right) \right).$$

The bounds (8.27) and (8.28) imply that w is well-defined, because  $v \in \ell^1(\mathcal{M}^1_n(D))$ , and we have

$$||w||_1 \leq 2C \int_0^\infty e^{-cs} \mathrm{d}s ||v||_1 = \frac{2C}{c} ||v||_1.$$

The Kolmogorov equations yield that  $L_0^* \mathbb{P}_{\xi} \left( \pi_s^0 = \cdot \right) = \partial_s \mathbb{P}_{\xi} \left( \pi_s^0 = \cdot \right)$ , and hence for all  $\eta \in \mathcal{M}_n^1(D)$ :

$$\begin{split} L_0^* w(\eta) &= -\int_0^\infty \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) L_0^* \mathbb{P}_{\xi} \left( \pi_s^0 = \cdot \right) (\eta) \, \mathrm{d}s \\ &= -\int_0^\infty \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \partial_s \mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) \, \mathrm{d}s \\ &= -\int_0^\infty \partial_s \left( \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) \right) \, \mathrm{d}s. \end{split}$$

We already saw that

$$\lim_{s \to \infty} \sum_{\xi \in \mathcal{M}_n^1(D)} v(\xi) \mathbb{P}_{\xi} \left( \pi_s^0 = \eta \right) = 0,$$

and hence  $L_0^*w(\eta) = v(\eta)$ , which concludes the existence.

To prove uniqueness, fix  $w \in \text{Ker}L_0^* \cap (\text{Ker}L_0)^{\perp}$ . The fact that  $L_0^*w = 0$  yields that for all  $\eta \in \mathcal{M}_n^1(D), t \ge 0$ :

$$w(\eta) = \sum_{\xi \in \mathcal{M}_n^1(D)} w(\xi) \mathbb{P}_{\xi} \left( \pi_t^0 = \eta \right).$$

Letting  $t \to \infty$ , because  $w \in \ell^1(\mathcal{M}^1_n(D))$ , Inequalities (8.27), and (8.28) then imply that

$$w(\eta) = \sum_{\xi \in \mathcal{M}_n^1(D)} w(\xi) \mathbb{P}_{\xi} \left( \pi_{\tau}^0 = \eta \right) \, \forall \eta \in \mathcal{M}_n^1(D),$$

and thus we may write  $w = \sum_{x \in D} w(\delta_x) \mathbb{1}_x$ , where  $\mathbb{1}_x$  was defined in (8.14), and where the equality holds in  $\mathcal{M}_n^1(D)$ . In particular,  $w(\eta) = 0$  for all  $\eta \in \mathcal{M}_n^1(D) \setminus \Delta$ . We have for all  $x \in D$ :

$$\sum_{\eta \in \mathcal{M}_n^1(D)} \sum_{y \in D} |w(\delta_y) \mathbb{1}_{\eta = \delta_y} \psi_x^r(\eta)| = \sum_{y \in D} |w(\delta_y)| \psi_x^r(\delta_y) \leqslant ||w||_1 < \infty,$$

and hence the following computation holds:

$$0 = \langle w, \psi_x^r \rangle = \sum_{\eta \in \mathcal{M}_n^1(D)} w(\eta) \psi_x^r(\eta)$$
$$= \sum_{\eta \in \mathcal{M}_n^1(D)} \left( \sum_{y \in D} w(\delta_y) \mathbb{1}_{\eta = \delta_y} \right) \psi_x^r(\eta) = \sum_{y \in D} w(\delta_y) \psi_x^r(\delta_y) = w(\delta_x).$$

Thus, w = 0, which concludes for the uniqueness.

In the proof of Theorem 8.1.1, in order to apply Lemma 8.2.5, we need to show that for all  $x \in D$ ,

$$\langle \partial_t u_0 - L_m^* u_0, \psi_x^r \rangle = 0$$

To this end, let us introduce

$$\delta_{x;y} = \frac{n-1}{n}\delta_x + \frac{1}{n}\delta_y \in \mathcal{M}^1_n(D).$$

The next lemma provides an explicit formula for  $\psi_x^r(\delta_{x;y})$ . The fact that we only need the values of the committor functions on the set  $\Delta \cup \{\delta_{x;y}; x, y \in D\}$  is a consequence of the fact that, as explained above, in the limiting regime, the selection dynamics is much faster than the mutation dynamics, so that the support of  $\pi_t$  is concentrated on  $\{x, y\}$  in the transition from  $\delta_x$  to  $\delta_y$ .

**Lemma 8.2.6.** For all  $x, y \in D$ ,  $x \neq y$ , the committor functions  $\psi_x^r$  satisfy:

$$\psi_x^r(\delta_{x;y}) = \alpha_{x,y,r} \frac{\alpha_{x,y,r}^{n-1} - 1}{\alpha_{x,y,r}^n - 1}, \qquad \psi_x^r(\delta_{y;x}) = \alpha_{x,y,r}^{n-1} \frac{\alpha_{x,y,r} - 1}{\alpha_{x,y,r}^n - 1},$$

where  $\alpha_{x,y,r}$  is defined in (8.6).

*Proof.* The selection dynamics cannot extend the support of a measure, hence if  $supp \pi_0^0 = \{x, y\}$ , then  $supp \pi_t^0 \subset \{x, y\}$  for all  $t \ge 0$  and  $Z_t = \pi_t^0(x)$  completely determines  $\pi_t^0$ . The process Z corresponds to a Gambler's ruin problem, namely a Markov process on  $\{0, ..., n\}$  with jump rates

$$p(k, k+1) = \alpha_r(y) \frac{k(n-k)}{n-1}, \quad p(k, k-1) = \alpha_r(x) \frac{k(n-k)}{n-1},$$

where  $\alpha_r$  is given by (8.11). Denote by R its generator,

 $\tau_i(Z) = \inf \{t \ge 0, Z_t = i\}, \quad g(k) = \mathbb{P}_k(\tau_n(Z) < \tau_0(Z)).$ 

Then, g is solution to the problem:

$$Rg = 0, g(0) = 0, g(n) = 1.$$

We have:

$$Rg(k) = \alpha_r(y) \frac{k(n-k)}{n-1} \left( g(k+1) - g(k) \right) + \alpha_r(x) \frac{k(n-k)}{n-1} \left( g(k-1) - g(k) \right).$$

Hence:

$$g(k+1) - g(k) = \alpha_{x,y,r}^{-1} \left( g(k) - g(k-1) \right)$$

 $\Leftrightarrow$ 

$$g(k) - g(k-1) = \alpha_{x,y,r}^{-(k-1)} \left( g(1) - g(0) \right) = \alpha_{x,y,r}^{-(k-1)} g(1)$$

 $\Leftrightarrow$ 

$$g(k) = \frac{\alpha_{x,y,r}^{-k} - 1}{\alpha_{x,y,r}^{-1} - 1} g(1).$$

The condition g(n) = 1 yields:

$$g(1) = \frac{\alpha_{x,y,r}^{-1} - 1}{\alpha_{x,y,r}^{-n} - 1},$$

and finally:

$$\psi_x^r(\delta_{x;y}) = g(n-1) = \frac{\alpha_{x,y,r}^{-(n-1)} - 1}{\alpha_{x,y,r}^{-n} - 1}, \quad \psi_x^r(\delta_{y;x}) = g(1) = \frac{\alpha_{x,y,r}^{-1} - 1}{\alpha_{x,y,r}^{-n} - 1},$$

which concludes the proof.

#### 8.2.4 Proof of Theorem 8.1.1

Let  $Y^r$  and Y be jump Markov processes on D with rates  $\tilde{q}_r$  and  $\tilde{q}_{\infty}$  respectively. The last needed result is the convergence in total variation distance of  $Y^r$  towards Y. This is a general result on jump Markov processes whose jump rates converge.

Lemma 8.2.7. Under Assumption 8.1.1

$$\lim_{r \to \infty} \|\mathcal{L}aw(Y_0^r) - \mathcal{L}aw(Y_0)\|_{TV} = 0 \implies \forall t \ge 0, \ \lim_{r \to \infty} \|\mathcal{L}aw(Y_t^r) - \mathcal{L}aw(Y_t)\|_{TV} = 0.$$

*Proof.* We have that for all  $x, y \in D$ ,

$$\lim_{r \to \infty} \tilde{q}_r(x, y) = \tilde{q}_\infty(x, y)$$

and for all  $x \in D$ ,

$$\sum_{y \in D} \sup_{r \in \mathbb{N}} \tilde{q}_r(x, y) < \infty.$$

Hence, [99, Theorem 19.25, p. 385] concludes.

We are now in position to prove Theorem 8.1.1.

Proof of Theorem 8.1.1. Recall the jump rates:

$$\tilde{q}_r(x,y) = \begin{cases} nq(x,y)\frac{\alpha_{x,y,r}-1}{\alpha_{x,y,r}^n-1} & \text{if } \alpha_{x,y,r} \neq 1, \\ q(x,y) & \text{otherwise,} \end{cases}$$

and let  $Y^r$  be the Markov process with jump rates  $(\tilde{q}_r(x, y))_{x,y\in D}$  and initial condition  $Y_0^r$  such that the law of  $\delta_{Y_0^r}$  is  $\mathcal{L}aw(\pi_{\tau}^0)$ . Write  $p_x^r(t) = \mathbb{P}(Y_t^r = x)$ , for  $x \in D$ . The  $(p_x^r)_{x\in D}$  satisfy:

$$\sum_{x \in D} p_x^r = 1$$

and the Kolmogorov equations:

$$\partial_t p_x^r = \sum_{y \neq x} \left( \tilde{q}_r(y, x) p_y^r - \tilde{q}_r(x, y) p_x^r \right).$$

Denote by  $u(t) = (\mathbb{P}(\pi_t = \xi))_{\xi \in \mathcal{M}_n^1(D)}$  the law of the empirical measure of the Fleming-Viot process (which depends on r, even though this is not indicated explicitly for the sake of notation)

and define the conjectured equivalent by:

$$u_0(t) = \sum_{x \in D} p_x^r(t) \mathbb{1}_x,$$

which corresponds to the law of  $\delta_{Y_t^r}$ . Lemma 8.2.4 yields that for all  $t \ge 0$ :

$$L_0^* u_0(t) = 0.$$

The goal now is to solve (8.16) using Lemma 8.2.5. In order to do so, let us show that

$$\partial_t u_0 - L_m^* u_0 \in (\operatorname{Ker} L_0)^{\perp}$$

Fix  $x \in D$ . First:

$$\langle \partial_t u_0, \psi_x^r \rangle = \partial_t p_x^r.$$

For all  $f \in \ell^1(\mathcal{M}^1_n(D))$ :

$$L_m^*f(\xi) = \sum_{x \neq y} \left( \left( n\xi(y) + 1 \right) q(y, x) f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - n\xi(x)q(x, y)f(\xi) \right),$$

hence for all  $x \neq y, n \ge 3$ :

$$L_m^* u_0(\eta) = \begin{cases} -n \sum_{z \neq x} q(x, z) p_x^r & \text{if } \eta = \delta_x, \\ nq(x, y) p_x^r & \text{if } \eta = \delta_{x;y}, \\ 0 & \text{otherwise.} \end{cases}$$

In the case n = 2, the formula is adapted by  $L_m^* u_0(\delta_{x;y}) = nq(x,y)p_x^r + nq(y,x)p_y^r$ . Using Lemma 8.2.6, this yields that:

$$\begin{split} \langle L_m^* u_0, \psi_x^r \rangle \\ &= L_m^* u_0(\delta_x) \psi_x^r(\delta_x) + \sum_{y \neq x} \left( L_m^* u_0(\delta_y) \psi_x^r(\delta_y) + L_m^* u_0(\delta_{x;y}) \psi_x^r(\delta_{x;y}) + L_m^* u_0(\delta_{y;x}) \psi_x^r(\delta_{y;x}) \right) \\ &= -n \sum_{z \neq x} q(x, z) p_x^r + \sum_{y \neq x} nq(x, y) p_x^r \alpha_{x,y,r} \frac{\alpha_{x,y,r}^{n-1} - 1}{\alpha_{x,y,r}^n - 1} + nq(y, x) p_y^r \alpha_{x,y,r}^{n-1} \frac{\alpha_{x,y,r} - 1}{\alpha_{x,y,r}^n - 1} \\ &= \sum_{y \neq x} \left( -\tilde{q}_r(x, y) p_x^r + \tilde{q}_r(y, x) p_y^r \right). \end{split}$$

Finally we get that for all  $x \in D$ :

$$\langle \partial_t u_0 - L_m^* u_0, \psi_x^r \rangle = \partial_t p_x^r - \sum_{y \neq x} \left( \tilde{q}_r(y, x) p_y^r - \tilde{q}_r(x, y) p_x^r \right) = 0.$$

Lemma 8.2.4 then yields that  $\partial_t u_0 - L_m^* u_0 \in (\text{Ker}L_0)^{\perp}$ . Thus, Lemma 8.2.5 yields that, for any  $t \ge 0$ , there exists  $u_1(t) \in (\text{Ker}L_0)^{\perp}$  such that

$$\partial_t u_0(t) - L_m^* u_0(t) = L_0^* u_1(t),$$

with the explicit representation

$$u_1(t) = \int_0^\infty \left( \partial_t u_0(t) - L_m^* u_0(t) \right) e^{sL_0} \mathrm{d}s$$

From this representation and the regularity of  $u_0(t)$  provided by its definition, one gets that  $u_1$  is a  $C^1$  function of t, and that  $\partial_t u_1$  solves the Poisson equation

$$\partial_t^2 u_0(t) - L_m^* \partial_t u_0(t) = L_0^* \partial_t u_1(t).$$

As a conclusion, using Lemma 8.2.5 again, we deduce that

$$||u_1(t)||_1 \leq C ||\partial_t u_0(t) - L_m^* u_0(t)||_1, \quad ||\partial_t u_1(t)||_1 \leq C ||\partial_t^2 u_0(t) - L_m^* \partial_t u_0(t)||_1.$$

The Kolmogorov equations yield that  $\|\partial_t u_0\|_1 \leq Q$ , and  $\|\partial_t^2 u_0\|_1 \leq Q^2$ , and since  $L_m^*$  is a bounded operator, we have that:

$$\sup_{r\in\mathbb{N}}\sup_{t\geq 0}\|u_1(t)\|_1<\infty,\quad \sup_{r\in\mathbb{N}}\sup_{t\geq 0}\|\partial_t u_1(t)\|_1<\infty.$$

Write

$$u = u_0 + \frac{1}{\underline{\lambda}_r}u_1 + v.$$

Then v satisfies the following equation:

$$\partial_t v = L^* v + \frac{1}{\underline{\lambda}_r} \left( L_m^* u_1 - \partial_t u_1 \right), \ v(0) = u(0) - u_0(0) - \frac{1}{\underline{\lambda}_r} u_1(0).$$

In the case where there exists  $z \in D$  such that  $\pi(\mathbf{X}_0) = \delta_z$ , then  $\tau = 0$  so  $Y_0^r = z$ , and we have  $v(0) = -u_1(0)/\underline{\lambda}_r$ . Thus we can write:

$$v(t) = -\frac{1}{\underline{\lambda}_r} e^{tL^*} u_1(0) + \frac{1}{\underline{\lambda}_r} \int_0^t e^{(t-s)L^*} \left( L_m^* u_1 - \partial_t u_1 \right) \mathrm{d}s, \tag{8.31}$$

where for any  $w \in \ell^1(\mathcal{M}^1_n(D)), \eta \in \mathcal{M}^1_n(D)$ 

$$e^{tL^*}w(\eta) = \sum_{\xi \in \mathcal{M}_n^1(D)} w(\xi) \mathbb{P}_{\xi} \left( \pi_t = \eta \right).$$

Thanks to the bounds on  $||u_1||_1$  and  $||\partial_t u_1||_1$ , Equation (8.31) yields that for any T > 0, there exists C > 0 such that:

$$\sup_{t \in [0,T]} \|v(t)\|_1 \leqslant C/\underline{\lambda}_r.$$

As a conclusion, for any  $t \in [0, T]$ ,

$$\|\mathcal{L}aw\left(\pi(\mathbf{X}_t)\right) - \mathcal{L}aw\left(\delta_{Y_t^r}\right)\|_1 \leqslant \|v(t)\|_1 + \frac{\|u_1(t)\|_1}{\underline{\lambda}_r},$$

which concludes the proof of the second statement of Theorem 8.1.1.

To prove the first statement, with an initial condition which is not necessarily a Dirac mass,

we first write

$$\|\mathcal{L}aw\left(\pi(\mathbf{X}_{t})\right) - \mathcal{L}aw\left(\delta_{Y_{t}}\right)\|_{1} \leq \|\mathcal{L}aw\left(\pi(\mathbf{X}_{t})\right) - \mathcal{L}aw\left(\delta_{Y_{t}^{r}}\right)\|_{1} + \|\mathcal{L}aw\left(\delta_{Y_{t}^{r}}\right) - \mathcal{L}aw\left(\delta_{Y_{t}}\right)\|_{1},$$

where Y is the continuous-time Markov chain with initial condition  $\eta_{\infty}$  given by Lemma 8.2.3 and jump rates  $\tilde{q}_{\infty}(x, y)$ . For the first term of the right-hand side, we integrate the equation on v between time  $1/\sqrt{\Delta_r}$  and t to get:

$$v(t) = e^{(t-1/\sqrt{\underline{\lambda}_r})L^*} v\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) + \frac{1}{\underline{\lambda}_r} \int_{\frac{1}{\sqrt{\underline{\lambda}_r}}}^t e^{(t-s)L^*} \left(L_m^* u_1 - \partial_t u_1\right) \mathrm{d}s$$

The second term of this last equality still goes to 0 at speed  $1/\underline{\lambda}_r$ , and we are just left with showing that:

$$v\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) = u\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) - u_0\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) - \frac{1}{\underline{\lambda}_r}u_1\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) \underset{r \to \infty}{\to} 0,$$

in the  $\ell^1$  distance. Since  $u_1$  is bounded in  $\ell^1(\mathcal{M}^1_n(D))$ , uniformly in  $t \ge 0$  and  $r \in \mathbb{N}$ , we have that:

$$\lim_{r \to \infty} \left\| \frac{1}{\underline{\lambda}_r} u_1\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) \right\|_1 = 0.$$

Moreover,  $Y^r$  is a continuous-time Markov chain with initial condition  $Y_0^r$ , where  $Y_0^r$  is such that  $\mathcal{L}aw(\delta_{Y_0^r}) = \mathcal{L}aw(\pi_{\tau}^0)$ , and jump rates uniformly bounded with respect to r, hence  $\underline{\lambda}_r \to \infty$  yields that:

$$\lim_{r \to \infty} \left\| \mathcal{L}aw(\pi_{\tau}^0) - u_0\left(\frac{1}{\sqrt{\underline{\lambda}_r}}\right) \right\|_1 = 0.$$

Let us now show that

$$\lim_{r \to \infty} \|u\left(1/\sqrt{\underline{\lambda}_r}\right) - \mathcal{L}aw(\pi_\tau^0)\|_1 = 0,$$

even though  $u(0) = \eta$  for all  $r \in \mathbb{N}$ . Write

$$\tau_1 = \inf\left\{t \ge 0, \pi_t \in \Delta\right\},\tag{8.32}$$

and  $S_1$  for the first jump of a particles due to the mutation mechanism. Using the notation of Lemma 8.2.1, write  $\Theta = supp(\eta)$ ,  $b_r = 1$ ,  $L_r^2 = L_m$ ,  $L_r^1 = L_0^{supp(\eta)}$ , where

$$L_0^{supp(\eta)} f(\xi) = \sum_{x,y \in supp(\eta)} \alpha_r(x)\xi(x)\xi(y) \left( f\left(\xi + \frac{\delta_y - \delta_x}{n}\right) - f(\xi) \right)$$

as well as  $a_r = \underline{\lambda}_r$ , and  $c_r = \sqrt{\underline{\lambda}_r}$ . In this case, we have  $\mathbb{M}^1 = \Delta$ , and Lemma 8.2.1 yields that:

$$\lim_{r \to \infty} \mathbb{P}_{\eta} \left( \tau_1 < \frac{1}{\sqrt{\underline{\lambda}_r}} < S_1 \right) = 1.$$
(8.33)

For all  $\xi \in \mathcal{M}_n^1(D)$ , we may write:

$$u\left(\frac{1}{\sqrt{\Delta_r}}\right)(\xi) = \mathbb{P}_\eta\left(\pi_{\frac{1}{\sqrt{\Delta_r}}} = \xi, \tau_1 < \frac{1}{\sqrt{\Delta_r}} < S_1\right) + \mathbb{P}_\eta\left(\pi_{\frac{1}{\sqrt{\Delta_r}}} = \xi, \left(\tau_1 < \frac{1}{\sqrt{\Delta_r}} < S_1\right)^c\right)$$
$$= \mathbb{P}_\eta\left(\pi_\tau^0 = \xi, \tau_1 < \frac{1}{\sqrt{\Delta_r}} < S_1\right) + \mathbb{P}_\eta\left(\pi_{\frac{1}{\sqrt{\Delta_r}}} = \xi, \left(\tau_1 < \frac{1}{\sqrt{\Delta_r}} < S_1\right)^c\right),$$

and thus

$$\|u\left(1/\sqrt{\underline{\lambda}_r}\right) - \mathcal{L}aw(\pi_{\tau}^0)\|_1 \leq 2\mathbb{P}_{\eta}\left(\left(\tau_1 < \frac{1}{\sqrt{\underline{\lambda}_r}} < S_1\right)^c\right) \underset{r \to \infty}{\to} 0.$$

Finally, Lemma 8.2.3 and Lemma 8.2.7 imply the convergence of  $\|\mathcal{L}aw(\delta_{Y_t^r}) - \mathcal{L}aw(\delta_{Y_t})\|_1$  towards 0, and conclude the proof of Theorem 8.1.1.

# 8.3 Proof of Theorem 8.1.2

We are now interested in the convergence of the trajectories. As stated in the introduction, there is no hope of having convergence in the Skorohod topology. Instead, we use the  $L_{TV}^1$  topology. As the projection

$$\pi \in \mathcal{D}([0,T], \mathcal{M}^1(D)) \mapsto \pi_t \in \mathcal{M}^1(D)$$

is not continuous for this topology, it is not enough to show compactness of the probability measures of the trajectories, because an accumulation point would not be characterised by its time marginal. Instead, we proceed as follows. First, let  $F : \mathcal{D}([0,T], \mathcal{M}^1(D)) \to \mathbb{R}$  be bounded and Lipschitz continuous for the  $L_{TV}^1$  distance (8.10). We decompose

$$\begin{split} \mathbb{E}\left(F\left((\pi(\mathbf{X}_{t}^{r}))_{t\in[0,T]}\right)\right) &- \mathbb{E}\left(F\left((\delta_{Y_{t}})_{t\in[0,T]}\right)\right) \\ &= \left(\mathbb{E}\left(F\left((\pi(\mathbf{X}_{t}^{r}))_{t\in[0,T]}\right)\right) - \mathbb{E}\left(F\left((\delta_{Y_{t}^{r}})_{t\in[0,T]}\right)\right)\right) \\ &+ \left(\mathbb{E}\left(F\left((\delta_{Y_{t}^{r}})_{t\in[0,T]}\right)\right) - \mathbb{E}\left(F\left((\delta_{Y_{t}})_{t\in[0,T]}\right)\right)\right), \end{split}$$

with  $\mathcal{L}aw(Y_0^r) = \mathcal{L}aw(\pi_\tau^0)$  and  $\mathcal{L}aw(Y_0) = \eta_\infty$ . Thus, the conclusion of Theorem 8.1.2 follows from the combination of Lemmas 8.3.1 and 8.3.2.

**Lemma 8.3.1.** The process  $(\delta_{Y_t})_{t \in [0,T]}$  converges in distribution, in  $\mathcal{D}([0,T], \mathcal{M}^1(D))$ , to the process  $(\delta_{Y_t})_{t \in [0,T]}$ .

Proof. Let us endow the set D with the distance  $\rho(x, y) = \mathbb{1}_{x \neq y}$ , and the set  $\mathcal{D}([0, T], D)$  of cadlag trajectories  $[0, T] \to D$  with the associated Skorohod  $J_1$ -topology. By [99, Theorem 19.25, p. 385] (see the proof of Lemma 8.2.7), the process  $(Y_t^r)_{t \in [0,T]}$  converges in distribution to  $(Y_t)_{t \in [0,T]}$  in this space. To conclude, we now check that the mapping

$$\begin{array}{rcl} \mathcal{D}([0,T],D) & \to & \mathcal{D}([0,T],\mathcal{M}^1(D)) \\ (y_t)_{t\in[0,T]} & \mapsto & (\delta_{y_t})_{t\in[0,T]} \end{array}$$

is continuous. Let  $((y_t^k)_{t \in [0,T]})_{k \ge 1}$  be a sequence of elements of  $\mathcal{D}([0,T],D)$  which converges to
$(y_t)_{t \in [0,T]}$  in the J<sub>1</sub>-topology. We have

$$\|(\delta_{y_t^k})_{t \in [0,T]} - (\delta_{y_t})_{t \in [0,T]}\|_{L^1_{TV}} = 2\int_0^T \mathbb{1}_{y_t^k \neq y_t} \mathrm{d}t = 2\int_0^T \rho(y_t^k, y_t) \mathrm{d}t.$$

By [16, p. 125], the convergence of  $(y_t^k)_{t \in [0,T]}$  in the  $J_1$ -topology implies the convergence of  $\rho(y_t^k, y_t)$  to 0 for almost every  $t \in [0,T]$ , and therefore the conclusion follows from the dominated convergence theorem.

**Lemma 8.3.2.** There exists a coupling of the processes  $(\pi(\mathbf{X}_t^r))_{t \in [0,T]}$  and  $(\delta_{Y_t^r})_{t \in [0,T]}$  such that

$$\lim_{r \to \infty} \mathbb{E}\left[\int_0^T \|\pi(\mathbf{X}_t^r) - \delta_{Y_t^r}\|_{TV} dt\right] = 0.$$

*Proof.* Fix T > 0. The proof of Lemma 8.3.2 is divided in five steps. The first one is dedicated the definition of the coupling.

Step 1. Write  $\sigma_0 = 0$ ,

$$\tau_k = \inf \left\{ t \ge \sigma_k, \pi_t \in \Delta \right\},\,$$

and

$$\sigma_{k+1} = \inf \left\{ t \ge \tau_k, \pi_t \notin \Delta \right\}$$

for all  $k \in \mathbb{N}$ . We define an intermediate process, which is not Markov:

$$\bar{\eta}_t = \pi_{\tau_{N(t)}}, \quad N(t) = \max\left\{k, \tau_k \leqslant t\right\},$$

and for  $t \in [0, \tau_0]$ , write  $\bar{\eta}_t = \pi_{\tau_0}$ . This is a process living in  $\Delta$ , and we will get a Markov process by a time re-scaling and conditioning. Define the change of time:

$$s(t) = \inf \left\{ s \ge 0, \int_0^s \mathbb{1}_{\pi_u \in \Delta} \mathrm{d}u > t \right\}.$$

Write  $\tilde{N}(t) = \max\{k, \sigma_k \leq t\}$ , and

 $\mathcal{A}_r(T) = \left\{ \text{For all } k \in [\![0, \tilde{N}(T)]\!], \text{ there are no jumps from the mutation mechanism between } \sigma_k \text{ and } \tau_k \right\}.$ 

If  $\pi_{\tau_k} = \delta_x$ , then  $\sigma_{k+1} - \tau_k$  is an exponential random variable with rate  $n \sum_{y \neq x} q(x, y)$ . Additionally, on  $\mathcal{A}_r(T)$ , Lemma 8.2.6 yields that the Dirac mass goes from x to y with probability

$$\frac{q(x,y)}{\sum_{y\neq x}q(x,y)}\alpha_{y,x,r}^{n-1}\frac{\alpha_{y,x,r}-1}{\alpha_{y,x,r}^n-1}.$$

Hence, writing

$$\delta_{Y^{r,1}} = \bar{\eta}_{s(t)},$$

we have that on  $\mathcal{A}_r(T)$ ,  $Y^{r,1}$  is exactly a continuous-time Markov chain with jump rates  $\tilde{q}_r$  and initial condition  $\pi_{\tau_0}$ . Let  $Y^{r,2}$  be another continuous-time Markov chain on D with jump rates  $\tilde{q}_r$ , initial condition with distribution  $\mathcal{L}aw(\pi^0_{\tau})$ , independent of the Fleming-Viot process, and write:

$$Y^r = Y^{r,1} \mathbb{1}_{\mathcal{A}_r(T)} + Y^{r,2} \mathbb{1}_{\mathcal{A}_r^c(T)}.$$

The process  $Y^r$  is a jump Markov process with jump rates  $\tilde{q}_r$  and initial condition with distribution  $\mathcal{L}aw(\pi_\tau^0)$ . In Step 2, we show that the probability of  $\mathcal{A}_r(T)$  converges to 1. In Step 3, we show that  $\pi$  is close to  $\bar{\eta}$ , and finally in Steps 4 and 5, we show that  $\bar{\eta}$  is close to  $Y^{r,1}$  in the  $L^1$  distance defined in (8.10), and thus to  $Y^r$ .

Step 2. Let us show that for all T > 0,

$$\lim_{r \to \infty} \mathbb{P}\left(\mathcal{A}_r(T)\right) = 1.$$

For any  $k \ge 0$ , the events  $\{\tilde{N}(T) \ge k\}$  and {there is a mutation jump between  $\sigma_k$  and  $\tau_k$ } are independent. Thus we have:

$$\mathbb{P}\left(\mathcal{A}_{r}(T)^{c}\right)$$

$$\leqslant \sum_{k=0}^{\infty} \mathbb{P}\left(\tilde{N}(T) \ge k, \text{ there is a mutation jump between } \sigma_{k} \text{ and } \tau_{k}\right)$$

$$\leqslant \sup_{k \in \mathbb{N}} \mathbb{P}\left(\text{there is a mutation jump between } \sigma_{k} \text{ and } \tau_{k}\right) \sum_{k=0}^{\infty} \mathbb{P}\left(\tilde{N}(T) \ge k\right).$$

Because  $\sigma_{k+1} - \tau_k$  is an exponential random variable with parameter bounded by nQ,  $\tilde{N}$  is bounded by a Poisson random variable with parameter nQT, and hence:

$$\sum_{k=0}^{\infty} \mathbb{P}\left(\tilde{N}(T) \ge k\right) = \mathbb{E}\left(\tilde{N}(T)\right) \le nQT.$$

The convergence (8.17) and the strong Markov property at time  $\sigma_k$  yield that

$$\lim_{r \to \infty} \sup_{k \in \mathbb{N}} \mathbb{P} (\text{there is a mutation jump between } \sigma_k \text{ and } \tau_k) = 0,$$

which concludes the proof.

Step 3. Let us now prove that the  $L_{TV}^1$  distance between  $\pi$  and  $\bar{\eta}$  goes to 0 when r goes to infinity. We have:

$$\|\pi - \bar{\eta}\|_{L^1_{TV}} \mathbb{1}_{\mathcal{A}_r(T)} \leqslant 2\mathbb{1}_{\mathcal{A}_r(T)} \sum_{k=0}^{\tilde{N}(T)} \bar{\tau}_k - \sigma_k \leqslant 2\sum_{k=0}^{\infty} \left(\bar{\tau}_k - \sigma_k\right) \mathbb{1}_{\sigma_k \leqslant T},$$

where

$$\bar{\tau}_k = \inf\left\{t \geqslant \sigma_k, \bar{\pi}_t^k \in \Delta\right\},\tag{8.34}$$

and  $\bar{\pi}^k$  is equal to  $\pi$  on  $[0, \sigma_k]$ , then the particles are subject to the same jump, but only from the selection part. The application of Lemma 8.2.1 to  $(\bar{\pi}_{t+\sigma_k}^k)_{t\geq 0}$  yields that there exist C, c > 0, independent of  $k \in \mathbb{N}$ , such that for all  $t \geq 0$ :

$$\mathbb{P}\left(\bar{\tau}_k - \sigma_k > t | \sigma_k \leqslant T\right) \leqslant C e^{-c\underline{\lambda}_r t},$$

yielding that

$$\lim_{r \to \infty} \sup_{k \in \mathbb{N}} \mathbb{E} \left( \bar{\tau}_k - \sigma_k | \sigma_k \leqslant T \right) = 0.$$

Using again that  $\tilde{N}(T)$  is bounded by a Poisson random variable with parameter nQT we get:

$$\mathbb{E}\left(\|\pi - \bar{\eta}\|_{L^{1}_{TV}}\right) \leq 2T\mathbb{P}\left(\mathcal{A}_{r}(T)^{c}\right) + \sup_{k \in \mathbb{N}} \mathbb{E}\left(\bar{\tau}_{k} - \sigma_{k}|\sigma_{k} \leq T\right) \sum_{k \geq 0} \mathbb{P}\left(\tilde{N}(T) \geq k\right)$$
$$= 2T\mathbb{P}\left(\mathcal{A}_{r}(T)^{c}\right) + \sup_{k \in \mathbb{N}} \mathbb{E}\left(\bar{\tau}_{k} - \sigma_{k}|\sigma_{k} \leq T\right) \mathbb{E}(\tilde{N}(T))$$
$$\leq 2T\mathbb{P}\left(\mathcal{A}_{r}(T)^{c}\right) + nQT \sup_{k \in \mathbb{N}} \mathbb{E}\left(\bar{\tau}_{k} - \sigma_{k}|\sigma_{k} \leq T\right) \xrightarrow{r \to \infty} 0.$$

Step 4. In order to show that the distance between  $\bar{\eta}$  and  $(\delta_{Y_t^{r,1}})_{0 \leq t \leq T}$  goes to zero as well, let us define the event

$$\mathcal{B}_r(T) = \left\{ \sum_{k=0}^{N(T)} \tau_k - \sigma_k < \min_{0 \le k \le \tilde{N}(T)} \sigma_{k+1} - \tau_k \right\},\,$$

and show that

$$\lim_{r \to \infty} \mathbb{P}\left(\mathcal{A}_r(T), \mathcal{B}_r(T)\right) = 1.$$

Fix  $\varepsilon > 0$ , and  $N_0 \in \mathbb{N}$  such that  $\mathbb{P}\left(\tilde{N}(T) > N_0\right) < \varepsilon$ . Then

$$\mathbb{P}\left(\mathcal{B}_r(T)^c, \mathcal{A}_r(T)\right) \leqslant \varepsilon + \mathbb{P}\left(\sum_{k=0}^{N_0} \tau_k - \sigma_k > \min_{0 \leqslant k \leqslant N_0} \sigma_{k+1} - \tau_k, \mathcal{A}_r(T)\right).$$

On the event  $\mathcal{A}_r(T)$ , the random variable  $\min_{0 \leq k \leq N_0} \sigma_{k+1} - \tau_k$  is lower bounded by a exponential random variable E with parameter  $N_0 n Q$ , independent of the family of random variable  $(\tau_k - \sigma_k)_{0 \leq k \leq N_0}$ . This implies that

$$\mathbb{P}\left(\sum_{k=0}^{N_0} \tau_k - \sigma_k > \min_{0 \leqslant k \leqslant N_0} \sigma_{k+1} - \tau_k, \mathcal{A}_r(T)\right) \leqslant \mathbb{P}\left(\exists 1 \leqslant k \leqslant N_0, \tau_k - \sigma_k > E/N_0\right)$$
$$\leqslant N_0 \sup_{k \in \mathbb{N}} \mathbb{P}\left(\tau_k - \sigma_k > E/N_0\right)$$

Lemma 8.2.1 then yields that

$$\limsup_{r\to\infty} \mathbb{P}\left(\mathcal{A}_r(T), \mathcal{B}_r(T)^c\right) \leqslant \varepsilon,$$

for all  $\varepsilon > 0$ , which concludes Step 4.

Step 5. Now, using Step 4, all is left to do is show that

$$\lim_{r \to \infty} \|\bar{\eta} - (\delta_{Y_t^r})_{0 \leqslant t \leqslant T}\|_{L_{TV}^1} = 0.$$

First notice that for all  $s \ge 0$ 

$$\int_0^s \mathbb{1}_{\pi_u \in \Delta} \mathrm{d}u = s - \sum_{k=0}^{\bar{N}(s)} s \wedge \tau_k - \sigma_k,$$

so that

$$s(t) = t + \sum_{k=0}^{\tilde{N}(s(t))} \tau_k - \sigma_k \ge t.$$

Additionally,  $\delta_{Y_t^r} \neq \bar{\eta}_t$  only if there is  $p \in \mathbb{N}$  such that  $t \leq \sigma_p \leq s(t)$ , and on the event  $\mathcal{B}_r(T)$ ,  $t \leq \sigma_p < s(t)$  implies that  $\tilde{N}(s(t)) = p$ . Hence

$$t \ge \sigma_p - \sum_{k=0}^p \tau_k - \sigma_k.$$

Hence the total amount of time where  $\delta_{Y^r} \neq \bar{\eta}$  is bounded as follow

$$\mathbb{1}_{\mathcal{A}_{r}(T),\mathcal{B}_{r}(T)} \int_{0}^{T} \mathbb{1}_{\delta_{Y_{t}^{r}} \neq \bar{\eta}_{t}} \mathrm{d}t \leqslant \mathbb{1}_{\mathcal{A}_{r}(T),\mathcal{B}_{r}(T)} \sum_{p=0}^{\tilde{N}(T)} \sum_{k=0}^{p} \tau_{k} - \sigma_{k} \leqslant \sum_{k=0}^{\tilde{N}(T)} (\tilde{N}(T) - k) \left(\bar{\tau}_{k} - \sigma_{k}\right).$$

where  $\bar{\tau}_k$  was defined in (8.34). We have:

$$\|\delta_{Y^{r}} - \bar{\eta}\|_{L^{1}_{TV}} \mathbb{1}_{\mathcal{A}_{r}(T)} \mathbb{1}_{\mathcal{B}_{r}(T)} = \|\delta_{Y^{r,1}} - \bar{\eta}\|_{L^{1}_{TV}} \mathbb{1}_{\mathcal{A}_{r}(T)} \mathbb{1}_{\mathcal{B}_{r}(T)} \leqslant 2 \sum_{k=0}^{\infty} \tilde{N}(T) \left(\bar{\tau}_{k} - \sigma_{k}\right) \mathbb{1}_{\sigma_{k} \leqslant T}.$$

Using Cauchy-Schwarz inequality, one gets:

$$\mathbb{E}\left(\sum_{k=0}^{\infty} \tilde{N}(T) \left(\bar{\tau}_{k} - \sigma_{k}\right) \mathbb{1}_{\sigma_{k} \leqslant T}\right) \leqslant \sum_{k=0}^{\infty} \mathbb{E}\left(\tilde{N}(T)^{2}\right) \mathbb{E}\left(\left(\bar{\tau}_{k} - \sigma_{k}\right)^{2} \mathbb{1}_{\sigma_{k} \leqslant T}\right)$$
$$\leqslant \mathbb{E}\left(\tilde{N}(T)^{2}\right) \mathbb{E}\left(\tilde{N}(T)\right) \sup_{k \in \mathbb{N}} \mathbb{E}\left(\left(\bar{\tau}_{k} - \sigma_{k}\right)^{2} \middle| \sigma_{k} \leqslant T\right).$$

Hence:

$$\mathbb{E}\left(\|\delta_{Y^r} - \bar{\eta}\|_{L^1_{TV}}\right) \leq 2T\mathbb{P}\left(\mathcal{A}_r(T) \cup B_r(T)^c\right) + 2\sup_{k \in \mathbb{N}} \mathbb{E}\left(\left(\bar{\tau}_k - \sigma_k\right)^2 \middle| \sigma_k \leq T\right) \mathbb{E}\left(\tilde{N}(T)^2\right) \mathbb{E}\left(\tilde{N}(T)\right).$$

yielding that

$$\lim_{r \to \infty} \mathbb{E}\left( \|\delta_{Y^r} - \bar{\eta}\|_{L^1_{TV}} \right) = 0.$$

which concludes the proof.

### 8.4 Proof of Theorem 8.1.3

The aim of this section is to prove Theorem 8.1.3, and we therefore suppose that Assumption 8.1.2 holds for the remainder of this work. In this case, we can write:

$$\lambda_r = \underline{\lambda}_r + m_r,$$

where  $\underline{\lambda}_r = \inf_D \lambda_r, \ m_r \ge 0$  and

$$||m||_{\infty} := \sup_{r \in \mathbb{N}} \sup_{x \in D} m_r(x) < \infty.$$

We start by showing that  $\pi_t$  converges to a Dirac mass. Denote still for  $z \in D$ :

$$\pi_t(z) = \pi(\mathbf{X}_t^r)(z). \tag{8.35}$$

**Lemma 8.4.1.** Under Assumption 8.1.2, we have for all  $t \ge 0$ :

$$\lim_{r \to \infty} \sum_{x \neq y \in D} \mathbb{E}(\pi_t(x)\pi_t(y)) \to 0, \qquad \lim_{r \to \infty} \mathbb{E}(\|\pi_t\|_{\infty}) = 1.$$

Moreover, if  $\pi(\mathbf{X}_0) = \delta_z$  for some  $z \in D$  and all  $r \in \mathbb{N}$ , then:

$$\sum_{x \neq y \in D} \mathbb{E}(\pi_t(x)\pi_t(y)) \leqslant (Q+2\|m\|_{\infty})\frac{n_r}{\underline{\lambda}_r}, \qquad \|\pi_t\|_{\infty} \ge 1 - (Q+2\|m\|_{\infty})\frac{n_r}{\underline{\lambda}_r}$$

*Proof.* Write, for a given probability measure  $\xi$  on D,

$$g_2(\xi) = \sum_{x \in D} \xi(x)^2.$$
(8.36)

We have:

$$\begin{split} Lg_{2}(\xi) &= \sum_{x \neq y} n_{r}\xi(x) \left( q(x,y) + \lambda_{r}(x) \frac{n_{r}}{n_{r}-1}\xi(y) \right) \\ &\qquad \left( \left( \xi(x) - \frac{1}{n_{r}} \right)^{2} - \xi(x)^{2} + \left( \xi(y) + \frac{1}{n_{r}} \right)^{2} - \xi(y)^{2} \right) \\ &= 2 \sum_{x \neq y} n_{r}\xi(x) \left( q(x,y) + \lambda_{r}(x) \frac{n_{r}}{n_{r}-1}\xi(y) \right) \left( \frac{1}{n_{r}^{2}} + \frac{\xi(y) - \xi(x)}{n_{r}} \right) \\ &\geqslant \frac{2}{n_{r}} \sum_{x \neq y} \lambda_{r}(x)\xi(x)\xi(y) - 2 \sum_{x \neq y} \xi(x)^{2}q(x,y) + \frac{2n_{r}}{n_{r}-1} \sum_{x \neq y} \lambda_{r}(x)\xi(x)\xi(y)(\xi(y) - \xi(x)) \\ &\geqslant \frac{2\lambda_{r}}{n_{r}} (1 - g_{2}(\xi)) - 2Qg_{2}(\xi) - \frac{n_{r}}{n_{r}-1} \sum_{x \neq y} (\lambda_{r}(y) - \lambda_{r}(x))\xi(x)\xi(y)(\xi(y) - \xi(x)) \\ &\geqslant \frac{2\lambda_{r}}{n_{r}} (1 - g_{2}(\xi)) - (2Q + 4 \|m\|_{\infty}). \end{split}$$

Using the Kolmogorov equation, this yields that for all  $t \ge 0$ :

$$\partial_t \mathbb{E}(g_2(\pi_t)) \ge \frac{2\underline{\lambda}_r}{n_r} (1 - \mathbb{E}(g_2(\pi_t))) - (2Q + 4||m||_{\infty})$$

and hence

$$\mathbb{E}(g_2(\pi_t)) \ge 1 - e^{-\frac{2\lambda_r}{n_r}t} (1 - \mathbb{E}(g_2(\pi_0))) - (2Q + 4||m||_{\infty}) \frac{n_r}{2\underline{\lambda}_r}.$$
(8.37)

Now, since  $\sum_{x \in D} \xi(x) = 1$  for all  $\xi \in \mathcal{M}^1(D)$ , we have:

$$g_2(\pi_t) \leqslant \|\pi_t\|_{\infty},$$

which implies the result on  $\|\pi_t\|_{\infty}$ . We also have:

$$\sum_{x \neq y \in D} \mathbb{E}(\pi_t(x)\pi_t(y)) = 1 - \mathbb{E}(g_2(\pi_t)),$$

which concludes the proof.

The preceding Lemma showed that  $\pi_t$  converges towards a Dirac measure as  $r \to \infty$ . The goal of the next Lemma is to determined the law of this limiting Dirac mass.

**Lemma 8.4.2.** Recall that  $\pi_t(x) = \pi(\mathbf{X}_t^r)(x)$ . Under Assumption 8.1.2, we have:

$$\lim_{r \to 0} \sum_{x \in D} \left| \mathbb{E}(\pi_t(x)) - \mathbb{P}\left(Y_t = x\right) \right| = 0,$$

where  $(Y_t)_{t\geq 0}$  is the Markov process with jump rates  $(q(x, y))_{x,y\in D}$  and initial condition  $\eta$ . Moreover, if  $\pi(\mathbf{X}_0^r) = \delta_z$  for some  $z \in D$  and all  $r \in \mathbb{N}$ , then for all T > 0, there exists C > 0 such that for all  $t \in [0, T]$ :

$$\sum_{x \in D} |\mathbb{E}(\pi_t(x)) - \mathbb{P}(Y_t = x)| \leq C \frac{n_r}{\underline{\lambda}_r}.$$

*Proof.* The Kolmogorov equation yields that for all  $x \in D$ :

$$\begin{split} \partial_t \mathbb{E}(\pi_t(x)) &= \sum_{y \neq x} \mathbb{E}(\pi_t(y)) q(y, x) - \mathbb{E}(\pi_t(x)) \sum_{y \neq x} q(x, y) \\ &+ \frac{n_r}{n_r - 1} \sum_{y \neq x} \mathbb{E}(\pi_t(x) \pi_t(y)) \left(\lambda_r(y) - \lambda_r(x)\right). \end{split}$$

Let  $(\nu_t)_{t\geq 0}$  be the unique solution to the following system:

$$\partial_t \nu_t(x) = \sum_{y \neq x} \nu_t(y)_t q(y, x) - \nu_t(x) \sum_{y \neq x} q(x, y) = L_m^*(\nu_t)(x),$$

with initial condition  $\nu_0 = \eta$ . We can identify  $\nu_t(x) = \mathbb{P}_{\eta}(Y_t = x)$ . Besides,  $L_m^*$  is a linear operator, and we have:

$$\partial_t \left( \mathbb{E}(\pi_t) - \nu_t \right) = L_m^* \left( \mathbb{E}(\pi_t) - \nu_t \right) + g_t,$$

where

$$g_t(x) = \sum_{y \neq x} \frac{n_r}{n_r - 1} \mathbb{E}(\pi_t(x)\pi_t(y)) \left(\lambda_r(y) - \lambda_r(x)\right),$$

for all  $x \in D$ . We get:

$$\mathbb{E}(\pi_t) - \eta_t = e^{tL_m^*} \left( \pi \left( \mathbf{X}_0^r \right) - \eta \right) + \int_0^t e^{sL_m^*} g_{t-s} \mathrm{d}s.$$

We have that for all  $t \ge 0$ :

$$||g_t||_1 \leq \frac{n_r}{n_r - 1} ||m||_{\infty} \sum_{x \neq y \in D} \mathbb{E}(\pi_t(x)\pi_t(y)) \leq 2||m||_{\infty} (1 - \mathbb{E}(g_2(\pi_t))),$$

where  $g_2$  is defined in (8.36). Since  $L_m$  is a Markov generator, Lemma 8.4.1 (and more precisely

equation (8.37)) yields that

$$\lim_{r \to \infty} \|\mathbb{E}(\pi_t) - \nu_t\|_1 = 0,$$

as well as, if  $\eta = \pi(X_0)$ :

$$\|\mathbb{E}(\pi_t) - \nu_t\|_1 \leqslant C \frac{n_r}{\underline{\lambda}_r},$$

for all 0 < t < T and some C > 0, which concludes the proof.

We are now in position to conclude the proof of Theorem 8.1.3.

Proof of Theorem 8.1.3. Let  $F: \mathcal{M}^1(D) \to \mathbb{R}$  bounded, and  $\varepsilon > 0$ . Write

$$\left| \mathbb{E}(F(\pi(\mathbf{X}_t^r))) - \sum_{z \in D} F(\delta_z) \mathbb{P}\left( \| \pi(\mathbf{X}_t^r) - \delta_z \|_{TV} \leqslant \varepsilon \right) \right| \leqslant \sum_{z \in D} \varepsilon \|F\|_{\infty} \mathbb{P}\left( \| \pi(\mathbf{X}_t^r) - \delta_z \|_{TV} \leqslant \varepsilon \right) + R_{r,t},$$

where

$$R_{r,t} \leq ||F||_{\infty} \mathbb{P}\left(d(\pi(\mathbf{X}_{t}^{r}), \Delta) > \varepsilon\right).$$

Using the fact that  $\|\pi_t - \delta_z\|_{TV} = 2(1 - \pi_t^z)$ , we get

$$R_{r,t} \leq \|F\|_{\infty} \mathbb{P}\left(\|\pi(\mathbf{X}_{t}^{r})\|_{\infty} < 1 - \varepsilon/2\right) \leq \frac{2\|F\|_{\infty}}{\varepsilon} \mathbb{E}\left(1 - \|\pi(\mathbf{X}_{t}^{r})\|_{\infty}\right),$$

and finally

$$\left| \mathbb{E}(F(\pi(\mathbf{X}_t^r))) - \sum_{x \in D} F(\delta_x) \mathbb{P}\left( \| \pi(\mathbf{X}_t^r) - \delta_x \|_{TV} \leqslant \varepsilon \right) \right| \leqslant \|F\|_{\infty} \left( \varepsilon + \frac{2}{\varepsilon} \mathbb{E}\left( 1 - \| \pi(\mathbf{X}_t^r) \|_{\infty} \right) \right).$$
(8.38)

Using Markov inequality, we have:

$$\mathbb{P}\left(\|\pi(\mathbf{X}_t^r) - \delta_z\|_{TV} \leqslant \varepsilon\right) = \mathbb{P}\left(\pi_t^z \ge 1 - \varepsilon/2\right) \leqslant (1 - \varepsilon/2)^{-1} \mathbb{E}(\pi_t(z)) \leqslant (1 + \varepsilon) \mathbb{E}(\pi_t(z))$$

as soon as  $\varepsilon < 1.$  On the other side, we have that

$$\mathbb{E}(\pi_t(z)) = \mathbb{E}\left(\pi_t(z)\mathbb{1}_{\pi_t(z) \ge 1-\varepsilon/2} + \mathbb{1}_{\pi_t(z) \le 1-\varepsilon/2}\right) \le \mathbb{P}\left(\pi_t^z \ge 1-\varepsilon/2\right) + \mathbb{E}\left(\pi_t(z)\mathbb{1}_{\pi_t(z) \le 1-\varepsilon/2}\right).$$

Moreover,

$$\sum_{x \in D} \mathbb{E} \left( \pi_t(z) \mathbb{1}_{\pi_t(z) \leq 1-\varepsilon/2} \right) = \sum_{x \in D} \mathbb{E} \left( \pi_t(x) \mathbb{1}_{\pi_t(x) \leq 1-\varepsilon/2} \left( \mathbb{1}_{\|\pi\|_{\infty} \leq 1-\varepsilon/2} + \mathbb{1}_{\|\pi\|_{\infty} > 1-\varepsilon/2} \right) \right)$$
$$\leq \mathbb{P} \left( \|\pi(\mathbf{X}_t)\|_{\infty} < 1-\varepsilon/2 \right) + \frac{\varepsilon}{2},$$

yielding that

$$\left|\sum_{x\in D} F(\delta_x)\mathbb{P}\left(\|\pi(\mathbf{X}_t^r) - \delta_x\|_{TV} \leqslant \varepsilon\right) - \sum_{x\in D} F(\delta_x)\mathbb{E}\left(\pi_t(x)\right)\right| \leqslant \|F\|_{\infty} \left(\varepsilon + \frac{2}{\varepsilon}\mathbb{E}(1 - \|\pi(\mathbf{X}_t^r)\|_{\infty})\right).$$
(8.39)

Lastly, we have that:

$$\left|\sum_{x\in D} F(\delta_x)\mathbb{E}\left(\pi_t(x)\right) - \sum_{x\in D} F(\delta_x)\mathbb{E}\left(\nu_t(x)\right)\right| \le \|F\|_{\infty} \sum_{x\in D} |\mathbb{E}(\pi_t(x)) - \mathbb{P}\left(Y_t = x\right)|.$$
(8.40)

Equations (8.38), (8.39) and (8.40) together yield

$$\left| \mathbb{E}(F(\pi(\mathbf{X}_{t}^{r}))) - \sum_{x \in D} F(\delta_{x}) \mathbb{E}(\nu_{t}(x)) \right|$$
  
$$\leq 2 \|F\|_{\infty} \left( \varepsilon + \frac{2}{\varepsilon} \mathbb{E}(1 - \|\pi(\mathbf{X}_{t}^{r})\|_{\infty}) \right) + \|F\|_{\infty} \sum_{x \in D} |\mathbb{E}(\pi_{t}(x)) - \mathbb{P}(Y_{t} = x)|, \quad (8.41)$$

and Lemma 8.4.1 and Lemma 8.4.2 yield

$$\lim_{r \to \infty} \sup_{t \to \infty} \left| \mathbb{E}(F(\pi(\mathbf{X}_t^r))) - \sum_{x \in D} F(\delta_x) \mathbb{E}(\nu_t(x)) \right| \leq 2 \|F\|_{\infty} \varepsilon.$$

Letting  $\varepsilon \to 0$  yields the first part of the theorem. In the case where the initial condition is a Dirac mass, plugging  $\varepsilon = \sqrt{n_r/\underline{\lambda}_r}$  in equation (8.41) and Lemma 8.4.2 yields for all T > 0 and 0 < t < T:

$$\begin{split} \left| \mathbb{E}(F(\pi(\mathbf{X}_t^r))) - \sum_{x \in D} F(\delta_x) \mathbb{E}\left(\nu_t(x)\right) \right| \\ \leqslant \|F\|_{\infty} \left( \sqrt{\frac{n_r}{\underline{\lambda}_r}} + \sqrt{\frac{\underline{\lambda}_r}{n_r}} \left(Q + 2\|m\|_{\infty}\right) \frac{n_r}{\underline{\lambda}_r} + C \frac{n_r}{\underline{\lambda}_r} \right) \leqslant C' \sqrt{\frac{n_r}{\underline{\lambda}_r}}, \end{split}$$

for some C' > 0, and this concludes the proof.

L		L	
		L	
L		L	
-			

# Bibliography

- D. Albritton, S. Armstrong, J.-C. Mourrat, and M. Novack. "Variational methods for the kinetic Fokker-Planck equation". In: arXiv: Analysis of PDEs (2019).
- [2] A. Alfonsi, R. Coyaud, and V. Ehrlacher. "Constrained overdamped Langevin dynamics for symmetric multimarginal optimal transportation". In: *Mathematical Models and Methods in Applied Sciences* 32.03 (2022), pp. 403–455. eprint: https://doi.org/10.1142/ S0218202522500105.
- [3] L. Angeli, D. Cristan, and M. Ottobre. "Uniform in time convergence of numerical schemes for stochastic differential equations via Strong Exponential stability: Euler methods, Split-Step and Tamed Schemes". In: arXiv e-prints, arXiv:2303.15463 (2023), arXiv:2303.15463. arXiv: 2303.15463 [math.NA].
- [4] A. Arnold and J. Erb. "Sharp entropy decay for hypocoercive and non-symmetric Fokker-Planck equations with linear drift". In: arXiv e-prints, arXiv:1409.5425 (2014), arXiv:1409.5425. arXiv: 1409.5425 [math.AP].
- [5] A. Asselah, P. A. Ferrari, and P. Groisman. "Quasistationary distributions and Fleming-Viot processes in finite spaces". In: J. Appl. Probab. 48.2 (2011), pp. 322–332.
- [6] D. Bakry, I. Gentil, and M. Ledoux. Analysis and geometry of Markov diffusion operators. Vol. 348. Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer, Cham, 2014, pp. xx+552.
- [7] V. Bansaye, B. Cloez, and P. Gabriel. "Ergodic Behavior of Non-conservative Semigroups via Generalized Doeblin's Conditions". In: Acta Applicandae Mathematicae 166.1 (2019), pp. 29–72.
- [8] V. Bansaye, B. Cloez, P. Gabriel, and A. Marguet. "A non-conservative Harris ergodic theorem". English. In: J. Lond. Math. Soc., II. Ser. 106.3 (2022), pp. 2459–2510.
- [9] F. Baudoin. "Bakry-Émery meet Villani". English. In: J. Funct. Anal. 273.7 (2017), pp. 2275–2291.
- [10] F. Baudoin, M. Gordina, and D.-P. Herzog. "Gamma calculus beyond Villani and explicit convergence estimates for Langevin dynamics with singular potentials". In: Arch. Ration. Mech. Anal. 241.2 (2021), pp. 765–804.
- [11] M. Benaïm, N. Champagnat, and D. Villemonais. "Stochastic approximation of quasistationary distributions for diffusion processes in a bounded domain". In: Annales de l'Institut Henri Poincaré, Probabilités et Statistiques 57.2 (2021), pp. 726-739.
- [12] M. Benaïm and B. Cloez. "A stochastic approximation approach to quasi-stationary distributions on finite spaces". In: *Electron. Commun. Probab.* 20 (2015), 13 pp.

- [13] M. Benaïm, B. Cloez, and F. Panloup. "Stochastic approximation of quasi-stationary distributions on compact spaces and applications". In: Ann. Appl. Probab. 28.4 (2018), pp. 2370–2416.
- [14] M. Bieniek, K. Burdzy, and S. Finch. "Non-extinction of a Fleming-Viot particle model". In: Probability Theory and Related Fields 153 (2012), pp. 293–332.
- [15] M. Bieniek, K. Burdzy, and S. Pal. "Extinction of Fleming-Viot-type particle systems with strong drift". In: *Electronic Journal of Probability* 17.none (2012), pp. 1–15.
- [16] P. Billingsley. Convergence of probability measures. English. 2nd ed. Wiley Ser. Probab. Stat. Chichester: Wiley, 1999.
- [17] J. M. Bofill, W. Quapp, and E. Bernuz. "Some remarks on the model of the extended gentlest ascent dynamics". English. In: J. Math. Chem. 53.1 (2015), pp. 41–57.
- [18] I. Bogachev, N. V. Krylov, and S. V. Röckner M.and Shaposhnikov. Fokker-Planck-Kolmogorov equations. Vol. 207. Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2015, pp. xii+479.
- [19] S. Bonfanti and W. Kob. "Methods to locate saddle points in complex landscapes". In: *The Journal of Chemical Physics* 147.20 (2017), p. 204104.
- [20] L. Bottou, F. E. Curtis, and J. Nocedal. "Optimization methods for large-scale machine learning". English. In: SIAM Rev. 60.2 (2018), pp. 223–311.
- [21] N. Bou-Rabee. "Time Integrators for Molecular Dynamics". In: Entropy 16 (2014), pp. 138– 162.
- [22] A. Bouchard-Côté, S. J. Vollmer, and A. Doucet. "The bouncy particle sampler: a nonreversible rejection-free Markov chain Monte Carlo method". English. In: J. Am. Stat. Assoc. 113.522 (2018), pp. 855–867.
- [23] P. Bras and G. Pagès. "Convergence of Langevin-simulated annealing algorithms with multiplicative noise. II: Total variation". English. In: Monte Carlo Methods Appl. 29.3 (2023), pp. 203–219.
- [24] K. Burdzy, R. Holyst, D. Ingerman, and P. March. "Configurational transition in a Fleming - Viot-type model and probabilistic interpretation of Laplacian eigenfunctions". In: *Journal of Physics A Mathematical General* 29.11 (1996), pp. 2633–2642. arXiv: condmat/9603064 [cond-mat].
- [25] K. Burdzy, R. Holyst, and P. March. "A Fleming-Viot particle representation of the Dirichlet Laplacian". In: Comm. Math. Phys. 214.3 (2000), pp. 679–703.
- [26] E. Camrud, A. Durmus, P. Monmarché, and G. Stoltz. Second order quantitative bounds for unadjusted generalized Hamiltonian Monte Carlo. 2023. arXiv: 2306.09513 [math.PR].
- [27] E. Cancès, F. Legoll, and G. Stoltz. "Theoretical and numerical comparison of some sampling methods for molecular dynamics". In: *ESAIM: M2AN* 41.2 (2007), pp. 351–389.
- [28] P. Cattiaux and L. Mesnager. "Hypoelliptic non-homogeneous diffusions". English. In: Probab. Theory Relat. Fields 123.4 (2002), pp. 453–483.
- [29] F. Cerou, B. Delyon, A. Guyader, and M. Rousset. "A Central Limit Theorem for Fleming-Viot Particle Systems with Soft Killing". In: arXiv e-prints, arXiv:1611.00515 (2016), arXiv:1611.00515. arXiv: 1611.00515 [math.PR].
- [30] M. Chak. "Regularity preservation in Kolmogorov equations for non-Lipschitz coefficients under Lyapunov conditions". In: arXiv e-prints, arXiv:2209.05436 (2022), arXiv:2209.05436. arXiv: 2209.05436 [math.PR].

- [31] M. Chak, N. Kantas, and G. A. Pavliotis. "On the Generalised Langevin Equation for Simulated Annealing". In: arXiv e-prints, arXiv:2003.06448 (2020), arXiv:2003.06448. arXiv: 2003.06448 [math.PR].
- [32] N. Champagnat, K. Coulibaly-Pasquier, and D. Villemonais. "Criteria for exponential convergence to quasi-stationary distributions and applications to multi-dimensional diffusions". In: arXiv e-prints, arXiv:1603.07909 (2016), arXiv:1603.07909. arXiv: 1603.07909 [math.PR].
- [33] N. Champagnat and V. Hass. Convergence of population processes with small and frequent mutations to the canonical equation of adaptive dynamics. 2024. arXiv: 2303.09936 [math.PR].
- [34] N. Champagnat and D. Villemonais. "Convergence of the Fleming-Viot process toward the minimal quasi-stationary distribution". In: arXiv e-prints, arXiv:1810.06849 (2018), arXiv:1810.06849. arXiv: 1810.06849 [math.PR].
- [35] N. Champagnat and D. Villemonais. "General criteria for the study of quasi-stationarity". In: arXiv e-prints, arXiv:1712.08092 (2017), arXiv:1712.08092. arXiv: 1712.08092 [math.PR].
- [36] N. Champagnat and D. Villemonais. "Practical criteria for *R*-positive recurrence of unbounded semigroups". In: *Electron. Commun. Probab.* 25 (2020), 11 pp.
- [37] T.-S. Chiang, C.-R. Hwang, and S. Sheu. "Diffusion for Global Optimization in ℝ<sup>n</sup>". In: SIAM Journal on Control and Optimization 25.3 (1987), pp. 737–753. eprint: https: //doi.org/10.1137/0325042.
- [38] B. Cloez and M.-N. Thai. "Quantitative results for the Fleming–Viot particle system and quasi-stationary distributions in discrete space". In: *Stochastic Processes and their Applications* 126.3 (2016), pp. 680–702.
- [39] P. Collet, S. Martínez, and J. San Martín. Quasi-stationary distributions. Probability and its Applications (New York). Markov chains, diffusions and dynamical systems. Springer, Heidelberg, 2013, pp. xvi+280.
- [40] M. D. and B. H. "Conquering the hard cases of Lennard-Jones clusters with simple recipes". In: Computational and Theoretical Chemistry 1107 (2017). Structure prediction of nanoclusters from global optimization techniques: computational strategies, pp. 7–13.
- [41] A. Debussche and E. Faou. "Weak backward error analysis for SDEs". English. In: SIAM J. Numer. Anal. 50.3 (2012), pp. 1735–1752.
- [42] P. Del Moral. Feynman-Kac Formulae. Genealogical and Interacting Particle Systems with Applications. Springer-Verlag New York, 2004.
- [43] P. Del Moral. Mean Field Simulation for Monte Carlo Integration. New York: Chapman and Hall/CRC, 2013.
- [44] P. Del Moral and A. Guionnet. "On the stability of interacting processes with applications to filtering and genetic algorithms". en. In: Annales de l'I.H.P. Probabilités et statistiques 37.2 (2001), pp. 155–194.
- [45] P. Del Moral and L. Miclo. "A Moran particle system approximation of Feynman-Kac formulae". In: Stochastic Processes and their Applications 86.2 (2000), pp. 193 –216.
- [46] P. Del Moral and L. Miclo. "Particle approximations of Lyapunov exponents connected to Schrödinger operators and Feynman-Kac semigroups". In: ESAIM Probab. Stat. 7 (2003), pp. 171–208.

- [47] P. Del Moral and D. Villemonais. "Exponential mixing properties for time inhomogeneous diffusion processes with killing". In: *Bernoulli* 24.2 (2018), pp. 1010–1032.
- [48] B. Delyon, F. Cérou, A. Guyader, and M. Rousset. "A Central Limit Theorem for Fleming-Viot Particle Systems with Hard Killing". In: *To appear in Annales de l'IHP (Probability and Statistics)*, arXiv:1709.06771 (2017), arXiv:1709.06771. arXiv: 1709.06771 [math.PR].
- [49] G. Di Gesù, T. Lelièvre, D. Le Peutrec, and B. Nectoux. "Jump Markov models and transition state theory: the quasi-stationary distribution approach". In: *Faraday Discussions* 195 (2017), pp. 469–495.
- [50] J. Dolbeault, C. Mouhot, and C. Schmeiser. "Hypocoercivity for linear kinetic equations conserving mass". English. In: Trans. Am. Math. Soc. 367.6 (2015), pp. 3807–3828.
- [51] J. Duncan, Q. Wu, K. Promislow, and G. Henkelman. "Biased gradient squared descent saddle point finding method". In: *The Journal of chemical physics* 140.19 (2014), p. 194102.
- [52] A. Durmus, A. Guillin, and P. Monmarché. "Geometric ergodicity of the bouncy particle sampler". In: Ann. Appl. Probab. 30.5 (2020), pp. 2069–2098.
- [53] W. E, W. Ren, and E. Vanden-Eijnden. "String method for the study of rare events". In: *Physical Review B* 66.5 (2002).
- [54] W. E and X. Zhou. "The gentlest ascent dynamics". English. In: Nonlinearity 24.6 (2011), pp. 1831–1842.
- [55] A. Eberle. "Reflection couplings and contraction rates for diffusions". English. In: Probab. Theory Relat. Fields 166.3-4 (2016), pp. 851–886.
- [56] A. Eberle, A. Guillin, and R. Zimmer. "Couplings and quantitative contraction rates for Langevin dynamics". In: *The Annals of Probability* 47.4 (2019), pp. 1982 –2010.
- [57] A. M. Etheridge. An introduction to superprocesses. Vol. 20. University Lecture Series. American Mathematical Society, Providence, RI, 2000, pp. xii+187.
- [58] S. N. Ethier and T. G. Kurtz. Markov processes. Characterization and convergence. English. Wiley Ser. Probab. Math. Stat. John Wiley & Sons, Hoboken, NJ, 1986.
- [59] P. Ferrari and N. Maric. "Quasi Stationary Distributions and Fleming-Viot Processes in Countable Spaces". In: *Electron. J. Probab.* 12 (2007), pp. 684–702.
- [60] G. Ferré, M. Rousset, and G. Stoltz. "More on the long time stability of Feynman-Kac semigroups". English. In: Stoch. Partial Differ. Equ., Anal. Comput. 9.3 (2021), pp. 630– 673.
- [61] W. H. Fleming and M. Viot. "Some Measure-Valued Markov Processes in Population Genetics Theory". In: Indiana University Mathematics Journal 28.5 (1979), pp. 817–843.
- [62] N. Fournier and A. Guillin. "On the rate of convergence in Wasserstein distance of the empirical measure". English. In: Probab. Theory Relat. Fields 162.3-4 (2015), pp. 707–738.
- [63] N. Fournier, P. Monmarché, and C. Tardif. "Simulated annealing in R<sup>d</sup> with slowly growing potentials". In: Stochastic Processes and their Applications 131 (2021), pp. 276–291.
- [64] N. Fournier and C. Tardif. "On the Simulated Annealing in ℝ<sup>n</sup>". In: J. Funct. Anal. 281 (2021).
- [65] M. Freidlin and A. Wentzell. Random Perturbations of Dynamical Systems. Springer, 2012.

- [66] W. Gao, J. Leng, and X. Zhou. "An Iterative Minimization Formulation for Saddle Point Search". In: SIAM Journal on Numerical Analysis 53.4 (2015), pp. 1786–1805. eprint: https://doi.org/10.1137/130930339.
- [67] N. J. Gerber, F. Hoffmann, and U. Urbain Vaes. Mean-field limits for Consensus-Based Optimization and Sampling. 2023. arXiv: 2312.07373 [math.PR].
- [68] M. Giles. "An introduction to multilevel Monte Carlo methods". English. In: Proceedings of the international congress of mathematicians 2018, ICM 2018, Rio de Janeiro, Brazil, August 1–9, 2018. Volume IV. Invited lectures. Hackensack, NJ: World Scientific; Rio de Janeiro: Sociedade Brasileira de Matemática (SBM), 2018, pp. 3571–3590.
- [69] N. Gouraud, P. Le Bris, A. Majka, and P. Monmarché. *HMC and underdamped Langevin united in the unadjusted convex smooth case*. 2023. arXiv: 2202.00977 [math.PR].
- [70] H. Graeme, J. Gísli, and J. Hannes. "Methods for finding saddle points and minimum energy paths". In: *Theoretical methods in condensed phase chemistry*. Springer, 2002, pp. 269–302.
- [71] I. Grigorescu and M. Kang. "Hydrodynamic limit for a Fleming-Viot type system". In: Stochastic Process. Appl. 110.1 (2004), pp. 111–143.
- [72] P. Groisman and M. Jonckheere. "Front propagation and quasi-stationary distributions: two faces of the same coin". English. In: Sojourns in probability theory and statistical physics. III. Interacting particle systems and random walks, a festschrift for Charles M. Newman. Singapore: Springer; Shanghai: NYU Shanghai, 2019, pp. 242–254.
- [73] P. Groisman and M. Jonckheere. "Simulation of quasi-stationary distributions on countable spaces". In: arXiv e-prints, arXiv:1206.6712 (2012), arXiv:1206.6712. arXiv: 1206.
   6712 [math.PR].
- [74] S. Groß kinsky, G. M. Schütz, and H. Spohn. "Condensation in the zero range process: stationary and dynamical properties". In: J. Statist. Phys. 113.3-4 (2003), pp. 389–410.
- S. Gu and X. Zhou. "Simplified gentlest ascent dynamics for saddle points in non-gradient systems". In: *Chaos: An Interdisciplinary Journal of Nonlinear Science* 28.12 (2018), p. 123106.
- [76] A. Guillin, P. Le Bris, and P. Monmarché. "Convergence rates for the Vlasov-Fokker-Planck equation and uniform in time propagation of chaos in non convex cases". In: *Electronic Journal of Probability* 27.none (2022), pp. 1–44.
- [77] M. Hairer. "On Malliavin's proof of Hörmander's theorem". English. In: Bull. Sci. Math. 135.6-7 (2011), pp. 650–666.
- [78] M. Hairer, M. Hutzenthaler, and A. Jentzen. "Loss of regularity for Kolmogorov equations". In: Ann. Probab. 43.2 (2015), pp. 468–527.
- [79] M. Hairer and J. Mattingly. "Yet another look at Harris' ergodic theorem for Markov chains". English. In: Seminar on stochastic analysis, random fields and applications VI. Centro Stefano Franscini, Ascona (Ticino), Switzerland, May 19–23, 2008. Basel: Birkhäuser, 2011, pp. 109–117.
- [80] B. Hajek. "Cooling schedules for optimal annealing". In: Math. Oper. Res. 13.2 (1988), pp. 311–329.
- [81] F. Hérau. "Short and long time behavior of the Fokker-Planck equation in a confining potential and applications". English. In: J. Funct. Anal. 244.1 (2007), pp. 95–118.

- [82] D.-P. Herzog and J.-C. Mattingly. "Ergodicity and Lyapunov functions for Langevin dynamics with singular potentials". In: Comm. Pure Appl. Math. 72.10 (2019), pp. 2231– 2255.
- [83] R. A. Holley, S. Kusuoka, and D. W. Stroock. "Asymptotics of the spectral gap with applications to the theory of simulated annealing". In: J. Funct. Anal. 83.2 (1989), pp. 333– 347.
- [84] R. A. Holley and D. W. Stroock. "Simulated annealing via Sobolev inequalities". In: Comm. Math. Phys. 115.4 (1988), pp. 553–569.
- [85] R. Höpfner, E. Löcherbach, and M. Thieullen. "Strongly degenerate time inhomogeneous SDEs: densities and support properties. Application to Hodgkin-Huxley type systems". English. In: *Bernoulli* 23.4A (2017), pp. 2587–2616.
- [86] K. Hu, Z. Ren, D. Siška, and Ł. Szpruch. "Mean-field Langevin dynamics and energy landscape of neural networks". English. In: Ann. Inst. Henri Poincaré, Probab. Stat. 57.4 (2021), pp. 2043–2065.
- [87] M. Hutzenthaler and A. Jentzen. "On a perturbation theory and on strong convergence rates for stochastic ordinary and partial differential equations with nonglobally monotone coefficients". In: Ann. Probab. 48.1 (2020), pp. 53–93.
- [88] M. Hutzenthaler, A. Jentzen, and P.-E. Kloeden. "Strong and weak divergence in finite time of Euler's method for stochastic differential equations with non-globally Lipschitz continuous coefficients". In: Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci. 467.2130 (2011), pp. 1563–1576.
- [89] M. Hutzenthaler, A. Jentzen, and P.-E. Kloeden. "Strong convergence of an explicit numerical method for SDEs with nonglobally Lipschitz continuous coefficients". In: Ann. Appl. Probab. 22.4 (2012), pp. 1611–1641.
- [90] M. Hutzenthaler, A. Jentzen, and X. Wang. "Exponential integrability properties of numerical approximation processes for nonlinear stochastic differential equations". English. In: Math. Comput. 87.311 (2018), pp. 1353–1413.
- [91] P.-E. Jabin and B. Perthame. "Collective motion driven by nutrient consumption". In: Asymptot. Anal. 133.4 (2023), pp. 483–497.
- [92] S. Jin, L. Li, Z. Xu, and Y. Zhao. "A random batch Ewald method for particle systems with Coulomb interactions". English. In: SIAM J. Sci. Comput. 43.4 (2021), b937–b960.
- [93] A. Joulin and Y. Ollivier. "Curvature, concentration and error estimates for Markov chain Monte Carlo". In: Ann. Probab. 38.6 (2010), pp. 2418–2442.
- [94] L. Journel and P. Monmarché. "Convergence of a particle approximation for the quasistationary distribution of a diffusion process: uniform estimates in a compact soft case". English. In: ESAIM, Probab. Stat. 26 (2022), pp. 1–25.
- [95] L. Journel and P. Monmarché. "Convergence of the kinetic annealing for general potentials". English. In: *Electron. J. Probab.* 27 (2022). Id/No 159, p. 37.
- [96] L. Journel and P. Monmarché. "Switched diffusion processes for non-convex optimization and saddle points search". English. In: *Statistics and Computing* (2023).
- [97] L. Journel and P. Monmarché. "Uniform convergence of the Fleming-Viot process in a hard killing metastable case". In: Arxiv (2022). arXiv: 2207.02030 [math.PR].
- [98] A. Juditsky, P. Rigollet, and A. B. Tsybakov. "Learning by Mirror Averaging". In: The Annals of Statistics 36.5 (2008), pp. 2183–2206.

- [99] O. Kallenberg. Foundations of modern probability. Second. Probability and its Applications (New York). Springer-Verlag, New York, 2002, pp. xx+638.
- [100] M. Kopec. "Weak backward error analysis for Langevin process". English. In: BIT 55.4 (2015), pp. 1057–1103.
- [101] M. Kopec. "Weak backward error analysis for overdamped Langevin processes". English. In: IMA J. Numer. Anal. 35.2 (2015), pp. 583–614.
- [102] H. Kunita. Stochastic flows and stochastic differential equations. English. Vol. 24. Camb. Stud. Adv. Math. Cambridge etc.: Cambridge University Press, 1990.
- [103] T. G. Kurtz. "Averaging for martingale problems and stochastic approximation". In: Applied Stochastic Analysis. Ed. by I. Karatzas and D. Ocone. Berlin, Heidelberg: Springer Berlin Heidelberg, 1992, pp. 186–209.
- [104] C. Le Bris, T. Lelièvre, M. Luskin, and D. Perez. "A mathematical formalization of the parallel replica dynamics". In: *Monte Carlo Methods and Applications* 18.2 (2012), pp. 119– 146.
- [105] B. Leimkuhler and C. Matthews. "Rational construction of stochastic numerical methods for molecular sampling". English. In: AMRX, Appl. Math. Res. Express 2013.1 (2013), pp. 34–56.
- [106] B. Leimkuhler and C. Matthews. Robust and efficient configurational molecular sampling via Langevin Dynamics. 2013. arXiv: 1304.3269 [physics.comp-ph].
- [107] B. Leimkuhler, C. Matthews, and G. Stoltz. "The computation of averages from equilibrium and nonequilibrium Langevin molecular dynamics". In: *IMA J. Numer. Anal.* 36.1 (2016), pp. 13–79.
- [108] T. Lelièvre, L. Pillaud-Vivien, and J. Reygner. "Central limit theorem for stationary Fleming-Viot particle systems in finite spaces". In: ALEA Lat. Am. J. Probab. Math. Stat. 15.2 (2018), pp. 1163–1182.
- [109] T. Lelièvre, M. Ramil, and J. Reygner. "A probabilistic study of the kinetic Fokker-Planck equation in cylindrical domains". In: arXiv e-prints, arXiv:2010.10157 (2020), arXiv:2010.10157. arXiv: 2010.10157 [math.PR].
- [110] T. Lelièvre, M. Rousset, and G. Stoltz. Free energy computations: A mathematical perspective. Imperial College Press, 2010.
- [111] T. Lelièvre and G. Stoltz. "Partial differential equations and stochastic methods in molecular dynamics". English. In: Acta Numerica 25 (2016), pp. 681–880.
- [112] T. Lelievre and G. Stoltz. "Partial differential equations and stochastic methods in molecular dynamics". In: Acta Numerica 25 (2016), pp. 681–880.
- [113] V. Lemaire, M. Thieullen, and N. Thomas. "Exact simulation of the jump times of a class of piecewise deterministic Markov processes". English. In: J. Sci. Comput. 75.3 (2018), pp. 1776–1807.
- [114] G. Leung and A. Barron. "Information Theory and Mixing Least-Squares Regressions". In: *IEEE Transactions on Information Theory* 52.8 (2006), pp. 3396–3410.
- [115] A. Levitt and C. Ortner. "Convergence and cycling in Walker-type saddle search algorithms". English. In: SIAM J. Numer. Anal. 55.5 (2017), pp. 2204–2227.
- [116] P. A. W. Lewis and G. S. Shedler. "Simulation of nonhomogeneous Poisson processes by thinning". English. In: Nav. Res. Logist. Q. 26 (1979), pp. 403–413.

- [117] T. Lindvall and L. C. G. Rogers. "Coupling of multidimensional diffusions by reflection". English. In: Ann. Probab. 14 (1986), pp. 860–872.
- [118] L. Liu, M. B. Majka, and P. Monmarché. L<sup>2</sup>-Wasserstein contraction for Euler schemes of elliptic diffusions and interacting particle systems. 2023. arXiv: 2310.15897 [math.PR].
- [119] J.-U. Löbus. "A stationary Fleming-Viot type Brownian particle system". In: Math. Z. 263.3 (2009), pp. 541–581.
- [120] A. Ludwig and K. Wolfgang. "On unique ergodicity for degenerate diffusions". In: Stochastics 21.1 (1987), pp. 41–61. eprint: https://doi.org/10.1080/17442508708833450.
- [121] Y. Luo, X. Zheng, X. Cheng, and L. Zhang. "Convergence analysis of discrete high-index saddle dynamics". English. In: SIAM J. Numer. Anal. 60.5 (2022), pp. 2731–2750.
- [122] M. B. Majka, A. Mijatović, and L. Szpruch. "Non-asymptotic bounds for sampling algorithms without log-concavity". In: arXiv e-prints, arXiv:1808.07105 (2018), arXiv:1808.07105. arXiv: 1808.07105 [math.PR].
- [123] J.-C. Mattingly, A.-M. Stuart, and D.-J. Higham. "Ergodicity for SDEs and approximations: locally Lipschitz vector fields and degenerate noise". In: *Stochastic Process. Appl.* 101.2 (2002), pp. 185–232.
- S. Méléard and D. Villemonais. "Quasi-stationary distributions and population processes". In: Probab. Surv. 9 (2012), pp. 340–410.
- [125] A. Menegaki. "Quantitative rates of convergence to non-equilibrium steady state for a weakly anharmonic chain of oscillators". English. In: J. Stat. Phys. 181.1 (2020), pp. 53– 94.
- [126] G. Menz and A. Schlichting. "Poincaré and logarithmic Sobolev inequalities by decomposition of the energy landscape". English. In: Ann. Probab. 42.5 (2014), pp. 1809–1884.
- [127] P. A. Meyer and W. A. Zheng. "Tightness criteria for laws of semimartingales". English. In: Ann. Inst. Henri Poincaré, Probab. Stat. 20 (1984), pp. 353–372.
- [128] S. P. Meyn and R. L. Tweedie. Markov chains and stochastic stability. English. Commun. Control Eng. Berlin: Springer-Verlag, 1993.
- [129] L. Miclo. "Recuit simulé sur  $\mathbb{R}^n$ . Étude de l'évolution de l'énergie libre". In: Ann. Inst. H. Poincaré Probab. Statist. 28.2 (1992), pp. 235–266.
- [130] L. Miclo. "Remarks on the ergodicity of simulated annealing algorithms on a graph". French. In: Stochastic Processes Appl. 58.2 (1995), pp. 329–360.
- [131] G. Milstein and M. V. Tretyakov. Stochastic numerics for mathematical physics. English. 2nd edition. Sci. Comput. Cham: Springer, 2021.
- [132] P. Monmarché. "Almost sure contraction for diffusions on ℝ<sup>d</sup>. Application to generalised Langevin diffusions". In: arXiv e-prints, arXiv:2009.10828 (2020), arXiv:2009.10828. arXiv: 2009.10828 [math.PR].
- [133] P. Monmarché. "Elementary coupling approach for non-linear perturbation of Markov processes with mean-field jump mechanisms and related problems". In: ESAIM: PS 27 (2023), pp. 278–323.
- [134] P. Monmarché. "Generalized Γ calculus and application to interacting particles on a graph". English. In: *Potential Anal.* 50.3 (2019), pp. 439–466.
- [135] P. Monmarché. "Generalized Γ calculus and application to interacting particles on a graph". In: Potential Analysis 50 (2019), pp. 439–466.

- [136] P. Monmarché. "High-dimensional MCMC with a standard splitting scheme for the underdamped Langevin diffusion". English. In: *Electron. J. Stat.* 15.2 (2021), pp. 4117–4166.
- [137] P. Monmarché. "Hypocoercivity in metastable settings and kinetic simulated annealing". In: Probability Theory and Related Fields (2018).
- [138] P. Monmarché, M. Rousset, and P.-A. Zitt. "Exact targeting of Gibbs distributions using velocity-jump processes". English. In: Stoch. Partial Differ. Equ., Anal. Comput. 11.3 (2023), pp. 908–947.
- [139] P. A. P. Moran. "Random processes in genetics". In: Mathematical Proceedings of the Cambridge Philosophical Society (1958).
- [140] J. R. Norris. Markov chains. English. Reprint. Vol. 2. Camb. Ser. Stat. Probab. Math. Cambridge: Cambridge Univ. Press, 1998.
- [141] J. A. Northby. "Structure and binding of Lennard-Jones clusters:  $13 \le N \le 147$ ". In: The Journal of Chemical Physics (1987).
- [142] W. Oçafrain and D. Villemonais. "Convergence of a non-failable mean-field particle system". In: Stoch. Anal. Appl. 35.4 (2017), pp. 587–603.
- [143] E. Olivieri and M. E. Vares. Large deviations and metastability. Vol. 100. Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 2005, pp. xvi+512.
- [144] G. A. Pavliotis and A. M. Stuart. Multiscale methods. Averaging and homogenization. English. Vol. 53. Texts Appl. Math. New York, NY: Springer, 2008.
- [145] E. K. Peter and P. Eckhard. Numerical Solution of Stochastic Differential Equations. 1st ed. Springer Berlin, Heidelberg, 1992, pp. XXXVI, 636.
- [146] P. P. Poier, L. Lagardère, and J.-P. Piquemal. "O(N) Stochastic Evaluation of Many-Body van der Waals Energies in Large Complex Systems". In: *Journal of Chemical Theory and Computation* 8.3 (Feb. 2022), pp. 1633–1645.
- [147] M. Pollock, P. Fearnhead, A. M. Johansen, and G. O. Roberts. "Quasi-stationary Monte Carlo and the ScaLE algorithm". In: J. R. Stat. Soc. Ser. B. Stat. Methodol. 82.5 (2020). With discussions followed by a rejoinder by the authors, pp. 1167–1221.
- [148] G. Rotskoff, S. Jelassi, J. Bruna, and E. Vanden-Eijnden. Global convergence of neuron birth-death dynamics. 2019. arXiv: 1902.01843 [stat.ML].
- [149] M. Rousset. "On the control of an interacting particle estimation of Schrödinger ground states". In: SIAM J. Math. Anal. 38.3 (2006), pp. 824–844.
- [150] G. Royer. "A remark on simulated annealing of diffusion processes". In: SIAM J. Control Optim. 27.6 (1989), pp. 1403–1408.
- [151] A. Samanta and E. Weinan. "Atomistic simulations of rare events using gentlest ascent dynamics". In: *The Journal of Chemical Physics* (2012).
- [152] A. Shirikyan. "Controllability implies mixing I. Convergence in the total variation metric". In: Russian Mathematical Surveys 72 (Oct. 2017).
- [153] G. Stoltz and Z. Trstanova. "Langevin dynamics with general kinetic energies". In: Multiscale Model. Simul. 16.2 (2018), pp. 777–806.
- [154] D. W. Stroock and S. R. S. Varadhan. Multidimensional diffusion processes. Classics in Mathematics. Reprint of the 1997 edition. Springer-Verlag, Berlin, 2006, pp. xii+338.

- [155] D. Stroock and S. Varadhan. "On the Support of Diffusion Processes with Applications to the Strong Maximum Principle". In: Proc. Sixth Berkeley Symp. Math. Statist. Prob. III (1972), pp. 333–359.
- [156] D. Talay. Efficient numerical schemes for the approximation of expectations of functionals of the solution of A.S.D.E., and applications. English. Filtering and control of random processes, Proc. Colloq., Paris 1983, Lect Notes Control Inf. Sci. 61, 294-313 (1984). 1984.
- [157] D. Talay. "Stochastic Hamiltonian systems: exponential convergence to the invariant measure, and discretization by the implicit Euler scheme". In: *Markov Process. Related Fields* 8.2 (2002). Inhomogeneous random systems (Cergy-Pontoise, 2001), pp. 163–198.
- [158] D. Talay and L. Tubaro. "Expansion of the global error for numerical schemes solving stochastic differential equations". In: Stochastic Anal. Appl. 8.4 (1990), 483–509 (1991).
- [159] S. Taniguchi. "Applications of Malliavin's calculus to time-dependent systems of heat equations". English. In: Osaka J. Math. 22 (1985), pp. 307–320.
- [160] O. Tough. Selection principle for the Fleming-Viot process with drift -1. 2023. arXiv: 2306.03585 [math.PR].
- [161] M. E Tuckerman. Statistical mechanics: theory and molecular simulation. Oxford university press, 2023.
- [162] M. Tuckerman, B. Berne, and G. Martyna. "Reversible multiple time scale molecular dynamics". In: J. Chem. Phys. 97.3 (1992), pp. 1990–2001.
- [163] C. Villani. "Hypocoercivity". In: Mem. Amer. Math. Soc. 202.950 (2009), pp. iv+141.
- [164] C. Villani. Optimal transport. Vol. 338. Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Old and new. Springer-Verlag, Berlin, 2009, pp. xxii+973.
- [165] D. Villemonais. "General approximation method for the distribution of Markov processes conditioned not to be killed". In: ESAIM Probab. Stat. 18 (2014), pp. 441–467.
- [166] D. Villemonais. "Interacting Particle Systems and Yaglom Limit Approximation of Diffusions with Unbounded Drift". In: *Electronic Journal of Probability* 16.none (2011), pp. 1663 –1692.
- [167] D. Villemonais. "Lower bound for the coarse Ricci curvature of continuous-time pure-jump processes". English. In: J. Theor. Probab. 33.2 (2020), pp. 954–991.
- [168] H. Vroylandt, L. Goudenège, P. Monmarché, F. Pietrucci, and B. Rotenberg. "Likelihoodbased non-Markovian models from molecular dynamics". In: *Proceedings of the National Academy of Sciences* 119.13 (2022).
- [169] K. Yosida. Functional analysis. English. Repr. of the 6th ed. Berlin: Springer-Verlag, 1994.
- [170] C. Zhang. "Hypocoercivity and global hypoellipticity for the kinetic Fokker-Planck equation in H<sup>k</sup> spaces". In: arXiv e-prints, arXiv:2012.06253 (2020), arXiv:2012.06253. arXiv: 2012.06253 [math.AP].
- [171] P.-A. Zitt. "Annealing diffusions in a potential function with a slow growth". In: Stochastic Process. Appl. 118.1 (2008), pp. 76–119.

## Long-time behavior of some Markov processes, and application to stochastic algorithms

### Abstract

This thesis is devoted to the study of the long-time behavior of some Markov processes arising in stochastic algorithms. These algorithms are variations of Monte Carlo methods, which constitute a class of techniques for sampling probability measures based on Markov processes. The thesis is structured into three parts, each addressing a different type of algorithm:

- 1. Chapters 2 and 3 focus on sampling Gibbs measures using two distinct kinetic processes.
- 2. Chapter 4 explores non-convex and high-dimensional optimization through simulated annealing, while Chapter 5 investigates optimization using a switched process.
- 3. Chapters 6, 7, and 8 delve into sampling quasi-stationary distributions based on the Fleming-Viot process, a Moran-type particle system.

The motivation for these algorithms, and thus much of the research conducted for this thesis, stems from molecular dynamics and statistics applications. By applying coupling and partial differential equation methods, we aim to establish convergence guarantees for these algorithms.

**Keywords:** probability theory, stochastic calculus, coupling methods, long time behavior of markov processes, poincaré inequality, stochastic algorithm, hypocoercivity.

### Résumé

Cette thèse est consacrée à l'étude du comportement à long terme de certains processus de Markov apparaissant dans les algorithmes stochastiques. Ces algorithmes sont des variations autour des méthodes de Monte Carlo, qui constituent une classe de techniques d'échantillonnage de mesures de probabilité basées sur des processus de Markov. La thèse est structurée en trois parties, chacune traitant d'un type d'algorithme différent :

- 1. Les chapitres 2 et 3 portent sur l'échantillonnage des mesures de Gibbs à l'aide de deux processus cinétiques différents.
- 2. Le chapitre 4 traite de l'optimisation non-convexe en grande dimension par recuit simulé, tandis que le chapitre 5 étudie l'optimisation à l'aide d'un processus commuté.
- 3. Les chapitres 6, 7 et 8 traitent de l'échantillonnage de mesures quasi-stationnaires basé sur le processus de Fleming-Viot, un système de particules de type Moran.

La motivation pour ces algorithmes, et donc pour la plupart des recherches menées dans le cadre de cette thèse, provient de dynamique moléculaire et d'applications en statistiques. En appliquant des méthodes de couplage et d'équations aux dérivées partielles, nous établirons des garanties de convergence pour ces algorithmes.

Mots clés : probabilités, calcul stochastique, méthodes de couplage, comportement en temps long de processus de markov, inégalité de poincaré, algorithme stochastique, hypocoercivité.



Sorbonne Université – Campus Pierre et Marie Curie – 4 place Jussieu – 75005 Paris – France